palgrave macmillan

Finance and Trade in Africa

Macroeconomic Response in the World Economy Context

Alemayehu Geda



Finance and Trade in Africa

International Finance and Development Series

Published in association with the Institute of Social Studies

General Editor: E. V. K. FitzGerald, Professor of Economics, Institute of Social Studies, The Hague, and Director, Finance and Trade Policy Research Centre, University of Oxford

The International Finance and Development Series reflects the research carried out at The Hague and associated centres in Europe, Asia, Africa and Latin America on the relationship between international capital flows and the process of structural adjustment in less-developed economies. The studies in this series share a common analytical approach based on the use of advanced social accounting techniques and the explicit modelling of the economic behaviour of institutional sectors, which in turn permit a new approach to macroeconomic policy design.

Titles include:

Alemayehu Geda FINANCE AND TRADE IN AFRICA Macroeconomic Response in the World Economy Context

Philomen Harrison THE IMPACT OF MACROECONOMIC POLICIES IN TRINIDAD AND TOBAGO The Firm under Adjustment

Karel Jansen EXTERNAL FINANCE IN THAILAND'S DEVELOPMENT

Karel Jansen and Rob Vos (*editors*) EXTERNAL FINANCE AND ADJUSTMENT

Luis Carlos Jemio DEBT, CRISIS AND REFORM IN BOLIVIA Biting the Bullet

Joke Luttik ACCOUNTING FOR THE GLOBAL ECONOMY

Rob Vos DEBT AND ADJUSTMENT IN THE WORLD ECONOMY

Rob Vos and Josef T. Yap THE PHILIPPINE ECONOMY East Asia's Stray Cat?

Howard White (*editor*) AID AND MACROECONOMIC PERFORMANCE

Finance and Trade in Africa

Macroeconomic Response in the World Economy Context

Alemayehu Geda Assistant Professor of Economics Addis Ababa University Ethiopia





Institute of Social Studies

© Institute of Social Studies 2002



All rights reserved. No reproduction, copy or transmission of this publication may be made without written permission.

No paragraph of this publication may be reproduced, copied or transmitted save with written permission or in accordance with the provisions of the Copyright, Designs and Patents Act 1988, or under the terms of any licence permitting limited copying issued by the Copyright Licensing Agency, 90 Tottenham Court Road, London W1T 4LP.

Any person who does any unauthorised act in relation to this publication may be liable to criminal prosecution and civil claims for damages.

The author has asserted his right to be identified as the author of this work in accordance with the Copyright, Designs and Patents Act 1988.

First published 2002 by PALGRAVE MACMILLAN Houndmills, Basingstoke, Hampshire RG21 6XS and 175 Fifth Avenue, New York, N. Y. 10010 Companies and representatives throughout the world

PALGRAVE MACMILLAN is the global academic imprint of the Palgrave Macmillan division of St. Martin's Press, LLC and of Palgrave Macmillan Ltd. Macmillan® is a registered trademark in the United States, United Kingdom and other countries. Palgrave is a registered trademark in the European Union and other countries.

ISBN 0-333-97722-X

This book is printed on paper suitable for recycling and made from fully managed and sustained forest sources.

A catalogue record for this book is available from the British Library.

Library of Congress Cataloging—in—Publication Data Alemayehu Geda Fole.

Finance and trade in Africa: macroeconomic response in the world economy context / Alemayehu Geda.

p. cm. — (International finance and development series) Includes bibliographical references and index.

ISBN 0-333-97722-X (cloth)

1. Africa—Economic conditions. 2. International finance– Econometric models. 3. International trade—Econometric models. 4. Economic policy—Econometric models—Evaluation. I. Title. II. Series.

HC800 .A694 2002 339'.096—dc21

2001055685

10 9 8 7 6 5 4 3 2 1 11 10 09 08 07 06 05 04 03 02

Printed and bound in Great Britain by Antony Rowe Ltd, Chippenham and Eastbourne for Selamawit Asfaw

Contents

| List | of Tables and Diagrams | xi |
|------|---|-------|
| List | of Acronyms and Abbreviations | xv |
| Seri | ies Editor's Preface | xviii |
| Ack | nowledgements | xxiii |
| 1 | Introduction | 1 |
| 1.1 | Introduction and Overview | 1 |
| 1.2 | Background to African Debt and Macro Policy Debates | 8 |
| | 1.2.1 Background to African Debt | 8 |
| | 1.2.2 Africa's Economic Crisis: What Caused It? | 10 |
| 1.3 | The Historical Origin of Africa's Economic Linkage with the | |
| | North | 16 |
| | 1.3.1 Pre-colonial Trade in Africa | 17 |
| | 1.3.2 The Formation of a Commodity-exporting and | |
| | External Finance-constrained Economy | 19 |
| | 1.3.3 The Three Macro Regions of Colonial Africa: | |
| | The Amin-Nzula Category | 25 |
| 1.4 | Some Notes on Methodology | 29 |
| 1.5 | Organization of the Book | 31 |
| 2 | Trade, Finance and Development in a | |
| | Global Context | 34 |
| 2.1 | Introduction | 34 |
| 2.2 | Finance Theories, North-South Modelling and | |
| | the African Context | 35 |
| | 2.2.1 Foreign Direct Investment | 35 |
| | 2.2.2 Theories of Bank Lending and Optimal Borrowing | 38 |
| | 2.2.3 Models of Sovereign Lending and the | |
| | Institutional Response | 39 |

| | 2.2.4 Official Capital Flows | 44 |
|-----|--|-----|
| | 2.2.5 Conclusion | 50 |
| 2.3 | International Trade Theories and their Relevance | |
| | in Modelling Africa | 52 |
| | 2.3.1 A Brief Overview of Trade Theories and their Implication | s |
| | for Africa | 52 |
| | 2.3.2 Non-orthodox Models and Trade in Primary Commodities | 54 |
| | 2.3.3 Quantitative Models of Trade in Primary Commodities | 59 |
| 2.4 | Conclusion | 85 |
| 3 | The Macroeconomics of Africa: | |
| | Import Compression and External Finance | 87 |
| 3.1 | Introduction | 87 |
| 3.2 | A Cursory Look at Some Macro Models of Africa | 88 |
| 3.3 | Some Basic Issues in African Macroeconomics | 100 |
| | 3.3.1 Import Compression | 101 |
| | 3.3.2 Macroeconomic Effects of External Finance | 103 |
| | 3.3.3 The Accounting Framework: An Overview | 105 |
| 3.4 | Conclusion | 109 |
| 4 | Determinants of Foreign Exchange Supply | 110 |
| 4.1 | Introduction | 110 |
| 4.2 | Determinants of Financial Flows to Africa: | |
| | Econometric Results | 111 |
| | 4.2.1 Determinants of Allocation of FDI to Africa | 111 |
| | 4.2.2 Determinants of Allocation of Official Capital Flows to | |
| | Africa | 114 |
| 4.3 | Determinants of African Export Supply | 119 |
| | 4.3.1 Introduction | 119 |
| | 4.3.2 The Literature | 120 |
| | 4.3.3 The Data | 129 |
| | 4.3.4 Estimation | 134 |
| | 4.3.5 Summing up | 157 |
| 4.4 | African Exports in a Global Context | 162 |
| | 4.4.1 Introduction | 162 |
| | 4.4.2 The Global Commodity Model | 163 |
| | 4.4.3 Estimation of the Model | 165 |
| | 4.4.4 Locating African Commodity Trade in the Global Market | 171 |
| 45 | Conclusion | 172 |

| 5 | External Finance and the African Economy: | |
|-----|---|-----|
| | A Macro Approach | 173 |
| 5.1 | Introduction | 173 |
| 5.2 | The Fiscal Response to External Finance | 173 |
| | 5.2.1 Introduction | 173 |
| | 5.2.2 The Model | 178 |
| | 5.2.3 Estimation Results | 181 |
| | 5.2.4 Summing Up | 196 |
| 5.3 | Aid and the 'Dutch Disease' in Africa | 198 |
| | 5.3.1 Aid and the 'Dutch Disease': An Overview | 198 |
| | 5.3.2 Aid, the 'Dutch Disease' and the African Context | 200 |
| | 5.3.3 Estimation Results | 205 |
| | 5.3.4 Summing Up | 219 |
| 5.4 | Private Investment and Import Compression: | |
| | Econometric Results | 220 |
| 5.5 | Conclusion | 225 |
| 6 | Modelling Africa Within a Global Economic | |
| | Framework | 227 |
| 6.1 | Introduction | 227 |
| 6.2 | Basic Theoretical Models | 228 |
| | 6.2.1 The Mundell-Fleming Model | 228 |
| | 6.2.2 The Theoretical Basis of Recent North-South Models | 230 |
| 6.3 | Standard North-South Linkages | 242 |
| | 6.3.1 Financial Linkages | 243 |
| | 6.3.2 Trade Linkages | 250 |
| 6.4 | Summing Up | 257 |
| 6.5 | Africa within a North-South Framework: | |
| | A Neostructuralist Global Econometric Model (AFRIMOD) | 258 |
| | 6.5.1 The North | 259 |
| | 6.5.2 The South: Africa | 264 |
| | 6.5.3 Major Oil Exporting Countries (OPEC) | 269 |
| | 6.5.4 International Banking | 270 |
| | 6.5.5 Identities Used | 272 |
| | 6.5.6 Some Partial Equilibrium-based Analytical Solutions | 273 |
| | 6.5.7 Parameters of the Model | 278 |
| | Glossary of symbols used | 289 |

| 7 | The Effect on Africa of External Shocks Generated in Developed Countries | 292 | | | | |
|------|---|-----|--|--|--|--|
| 7.1 | Introduction | 292 | | | | |
| 7.2 | 2 World Accounting Matrices (WAM) as Database for | | | | | |
| | AFRIMOD | 292 | | | | |
| 7.3 | Model Solution Algorithm and Stability Conditions | | | | | |
| | 7.3.1 Model Solution | 295 | | | | |
| | 7.3.2 Stability Conditions | 298 | | | | |
| 7.4 | Analysis of External Shocks and Policy Simulations Using | | | | | |
| | AFRIMOD | 301 | | | | |
| | 7.4.1 An Increase in the Level of Aid to the South | 304 | | | | |
| | 7.4.2 Debt Cancellation to Africa | 310 | | | | |
| | 7.4.3 An Increase in Public Investment | 313 | | | | |
| | 7.4.4 An Increase in the World Interest Rate | 315 | | | | |
| | 7.4.5 An Increase in Oil Price | 317 | | | | |
| | 7.4.6 An Increase in Supply of Commodity Exports by | | | | | |
| | Other South | 320 | | | | |
| 7.5 | Conclusion | 321 | | | | |
| 8 | Concluding Remarks | 324 | | | | |
| Ap | pendices | 331 | | | | |
| 1 | Country Classification | 332 | | | | |
| 4.1 | Econometric Approach Followed | 334 | | | | |
| 4.2 | Stationarity Condition of the Expected Profit Indicator | 340 | | | | |
| 4.3 | Results Using First Difference Values in the | | | | | |
| | Global Commodity Model | 343 | | | | |
| 5.1 | The Structure of Heller's Model | 348 | | | | |
| 5.2 | Estimation Using Other Approaches | 350 | | | | |
| 6.1 | OLS Estimation Results | 352 | | | | |
| No | tes | 353 | | | | |
| Refe | rences | 380 | | | | |
| Inde | x | 414 | | | | |

x

List of Tables and Diagrams

| Table 1.1 | Major debt indicators for Africa | 4 |
|------------|---|-----|
| Table 1.2 | Major debt indicators for Africa | 6 |
| Diag. 2.1 | The debt relief Laffer curve | 43 |
| Diag. 3.1 | Number of African macro models | 88 |
| Diag. 3.2 | Type or structure of African macro models | 89 |
| Diag. 3.3 | African macro models by source | 89 |
| Diag. 4.1 | The modified and actual UNCTAD price series (WCA) | 131 |
| Diag. 4.2 | The modified and actual UNCTAD price series (ESA) | 132 |
| Diag. 4.3 | The modified and actual UNCTAD price series (NA) | 133 |
| Table 4.1 | Unit root analysis: an ADF test at 1% Mackinnon | |
| | critical value for East and Southern Africa | 137 |
| Table 4.2 | Co-integration test for East and Southern Africa | 138 |
| Table 4.3 | Estimation with current prices for East and | |
| | Southern Africa | 140 |
| Table 4.4 | Estimation with real exchange rate for East | |
| | and Southern Africa | 141 |
| Table 4.5 | Estimation with capital inflow for East and | |
| | Southern Africa | 142 |
| Table 4.6 | Unit root analysis: an ADF test at 1% Mackinnon | |
| | critical value for North Africa | 143 |
| Table 4.7 | Co-integration test for North Africa | 144 |
| Table 4.8 | Estimation with current prices for North Africa | 148 |
| Table 4.9 | Estimation with real prices for North Africa | 149 |
| Table 4.10 | Estimation with capital inflow for North Africa | 150 |
| Table 4.11 | Unit root analysis: an ADF test at 1% Mackinnon | |
| | critical value for West and Central Africa | 151 |
| Table 4.12 | Co-integration test for West and Central Africa | 152 |
| Table 4.13 | Estimation with current prices for West and | |
| | Central Africa | 154 |
| Table 4.14 | Estimation with real exchange rate for West and | |
| | Central Africa | 155 |

| Table 4.15 | Estimation with capital inflow for West and | |
|-------------|---|-----|
| | Central Africa | 156 |
| Table 4.16 | Summary table of elasticities: All Africa | 158 |
| Table 4.17 | A unit root analysis of the world commodity model | |
| | and ADF test | 166 |
| Table 4.18 | Mackinnon critical values for the ADF test | 166 |
| Table 4.19 | Co-integration test for world commodity model | 167 |
| Table 4.20 | An ECM estimation result of the model | 168 |
| Diag. 4.4 | The structure of the commodity sub-bloc | 171 |
| Table 5.1a | A unit root analysis of the fiscal response model: | |
| | an ADF test | 182 |
| Table 5.1b | Mackinnon critical values for the ADF test | 182 |
| Table 5.1c | Co-integration test for fiscal response model | 183 |
| Table 5.2 | Direct taxes and capital inflows | 185 |
| Table 5.3 | Variation across financial instruments | 187 |
| Table 5.4 | Indirect taxes and capital inflow | 191 |
| Table 5.5 | Government current expenditure and capital inflow | 193 |
| Table 5.6a | A unit root analysis of the 'Dutch disease' model: | |
| | an ADF test | 207 |
| Table 5.6b | Mackinnon critical values for the ADF test | 207 |
| Table 5.6c | Co-integration test for the 'Dutch disease' model | 208 |
| Table 5.7 | First-order ECM results for Equation 5 | 210 |
| Table 5.8 | Country (constant) dummies used in the estimation of | |
| | Table 5.7 | 212 |
| Table 5.9 | Results of second-order ECM | 213 |
| Table 5.10a | A unit root analysis of the private investment model: | |
| | an ADF test | 223 |
| Table 5.10b | Mackinnon critical values for the ADF test | 223 |
| Table 5.10c | Co-integration test for the private investment model | 223 |
| Diag. 6.1 | The Meade-Mundel-Fleming model | 229 |
| Diag. 6.2 | The Solow-Lewis model | 235 |
| Diag. 6.3 | The Kalecki-Lewis model | 237 |
| Diag. 6.4 | Neo-Marxian North and Lewisian South | 240 |
| Table 6.1 | Summary of financial linkages | 249 |
| Table 6.2 | Summary of trade linkages | 252 |
| Table 6.3 | Regional capital-output ratio parameters | 279 |
| Table 6.4 | Estimated parameters of the model | 282 |
| Diag. 7.2a | A World Accounting Matrix for n countries | 293 |
| Table 7.2b | Balanced World Accounting Matrix for 1990 | 296 |
| Table 7.3 | Zellner and Peck symmetry and linearity test | 301 |

| xiii |
|------|
| |

| Diag. 7.4.1a | The effect of a 100 per cent increase in bond and budget financed aid to the South on commodity exports from | |
|--------------|---|--------------|
| | Africa | 307 |
| Diag. 7.4.1b | The effect of a 100 per cent increase in bond and budget financed aid to the South on African commodity prices | 308 |
| Diag. 7.4.1c | The effect of a 100 per cent increase in bond and budget financed aid to the South on African commodity prices | 308 |
| Diag. 7.4.1d | The effect of a 100 per cent increase in budget-financed | 500 |
| | aid to the South on government revenue in Africa | 309 |
| Diag. 7.4.1e | The effect of a 100 per cent increase in bond and budget financed aid to the South on government expenditure | |
| | in Africa | 309 |
| Diag. 7.4.1f | The effect of a 100 per cent increase in bond and budget | |
| Diag. 7.4.2a | financed aid to the South on real output from Africa The effect of a 66.7 per cent bilateral and multilateral | 310 |
| | debt cancellation on Africa's debt service ratio | 312 |
| Diag. 7.4.2b | The effect of a 66.7 per cent bilateral and multilateral | 210 |
| Diag 7420 | The affect of an annual 10 per cent avecanous increase | 312 |
| Diag. 7.4.5a | in (public revenue financed) public investment on | 214 |
| Diag 712h | The affect of an annual 10 per cent evogenous increase | 314 |
| Diag. 7.4.30 | in (public revenue-financed) public investment on | 214 |
| Diam 7420 | The effect of an annual 10 new cost every support in annual | 314 |
| Diag. 7.4.3c | in (public revenue financed) public investment on | |
| | real exports | 315 |
| Diag. 7.4.4a | The effect of a 1-point permanent exogenous rise in | |
| U | the world interest rate, 3 months LIBOR on the | |
| | debt service ratio | 316 |
| Diag. 7.4.4b | The effect of a 1-point permanent exogenous rise in | |
| | the world interest rate, 3 months LIBOR on | |
| | commodity price | 316 |
| Diag. 7.4.5a | The effect of a 175 per cent permanent increase in | |
| | oil price on interest rate, oligopolistic lending and | |
| | Northern output | 318 |
| Diag. 7.4.5b | The effect of a 175 per cent permanent increase in | a 1 o |
| Diag. 7.45 | oil price on imports to Africa | 319 |
| Diag. /.4.50 | oil price on imports to East Africa | 319 |

| Finance and Trade in Africa |
|-----------------------------|
|-----------------------------|

| .4.6a | The effect of a 10 per cent increase in supply by | |
|-------|---|---|
| | other (non-Africa) South on world commodity price | 321 |
| .4.6b | The effect of a 10 per cent increase in supply by | |
| | other (non-Africa) South on real commodity exports | |
| | from Africa | 322 |
| .4.6c | Vulnerability to debt: The effect of a supply-induced | |
| | decline in world commodity price on flows of | |
| | official finance to Africa | 322 |
| | .4.6a .4.6b .4.6c | 4.6a The effect of a 10 per cent increase in supply by other (non-Africa) South on world commodity price 4.6b The effect of a 10 per cent increase in supply by other (non-Africa) South on real commodity exports from Africa 4.6c Vulnerability to debt: The effect of a supply-induced decline in world commodity price on flows of official finance to Africa |

xiv

List of Acronyms and Abbreviations

| Two Stage Least Square |
|---|
| Three Stage Least Square Iterative |
| Three Stage Least Square |
| Africa |
| African Alternative Framework to Structural Adjustment |
| Programs |
| Addis Ababa University |
| European Union Designated 'Africa Carribean and Pacific' |
| States |
| World Bank, African Development Indicators |
| African Economic Research Consortium |
| Africa within a North-South Framework: A Neostructuralist |
| Global Econometric Model |
| World Bank's Global Economic model |
| The Central African Republic |
| Council for Mutual Economic Assistance |
| Constant Elasticity of Substitution |
| Computable General Equilibrium Model |
| Consumer Price Index |
| Centre for the Study of African Economies, University of |
| Oxford |
| Development Assistance Committee of the OECD |
| Debt Relief Laffer Curve |
| European Community |
| Economic Commission for Africa |
| Economic Commission for Latin American and the Caribbean |
| Error Correction Model |
| Total External Debt Stock |
| European Economic Community |
| East and Southern Africa |
| Enhanced Structural Adjustment Facility |
| |

| Finance | and | Trade | in A | frica |
|---------|-----|-------|------|-------|
| | | | | |

xvi

| ESPLA | Inter-American Development Bank's Economic and Social |
|-----------|--|
| | Progress in Latin America dataset |
| EU | European Union |
| EWTM | IMF Research Department's Expanded World Trade Model |
| FOB | Free on Board |
| CIF | Cost Insurance Freight |
| FDI | Foreign Direct Investment |
| FIML | Full-Information Maximum Likelihood |
| GDP | Gross Domestic Product |
| GFS | IMF's Government Financial Statistics |
| GK | Goldstein and Khan model |
| GNP | Gross National Product |
| GS | Henderey's General to Specific methodology |
| HIPC | Highly Indebted Poor Countries |
| IDS | Institute for Development Studies, Sussex University |
| IMF | International Monetary Fund |
| INTERLINK | OECD Model [International Linkages Model] |
| ISS | Institute of Social Studies, The Hague |
| IV | Instrumental Variables |
| LDC | Least Developed Countries |
| LIBOR | London Inter Bank Rate |
| LM | Money Market |
| LSE | The London School of Economics and Political Science |
| MNE | Multinational Enterprises |
| MSG2 | McKibbin and Sachs Global Model, version 2 |
| MULTIMOD | Multicountry Econometric Model (of the IMF) |
| N | North |
| NA | North Africa |
| NIESR | National Institute of Economic and Social Research, London |
| N-S | North/South |
| OAU | Organization of African Unity |
| ODA | Overseas Development Aid/Assistance |
| OECD | Organization for Economic Cooperation and Development |
| OLI | Ownership-specific Location and Internalisation |
| OLS | Ordinary Least Squares |
| OPEC | Organization of Petroleum Exporting Countries |
| OT | Other South (Excluding Africa) |
| QEH | Queen Elizabeth House, University of Oxford |
| RER | Real Exchange Rate |
| RW | Rest of the World |
| S | South |

| Structural Adjustment Facility |
|---|
| Social Accounting Matrix |
| Structural Adjustment Programs |
| System Dynamics Models |
| Spatial Equilibrium and Programming Models |
| UN Standard International Trade Classification |
| System of National Accounts |
| School of Oriental and African Studies, University of |
| London |
| Sub-Saharan Africa |
| A Structuralist North-South Trade Model |
| External Terms of Trade |
| United Nations Conference on Trade and Development |
| United Nations Development Programme |
| United Nations Economic Commission for Africa |
| Union of Soviet Socialist Republics |
| World Accounting Matrices |
| West And Central Africa |
| First World War |
| Second World War |
| |

Series Editor's Preface

At the start of a new millennium and after almost half a century of independence, the African economy seems trapped in a low-level disequilibrium. This not only condemns millions of people to a life of poverty but also blocks the process by which human and material capital can be accumulated. Many of Africa's economic problems are to be found within her own state and civil society, but it is too easy – and too convenient for the rest of the world – to attribute all economic problems to 'governance'. The fact is that Africa is an integral part of the world economy, open to financial and trade shocks and without the means to protect itself effectively against these shocks.

There are 53 countries in Africa, which accounts for 12 per cent of the world population. Yet, according to IMF estimates, the region generates only 3.3 per cent of world GDP at 1998 purchasing power parity and 1.8 per cent of world exports of goods and services. By comparison, Italy alone accounts for 3.4 per cent of world output and 5.7 per cent of world trade. This apparent marginalization is, of course, asymmetric. Although the world may not 'need' Africa in an economic sense, what happens in the world economy does have profound effects on Africa through commodity prices, interest rates, foreign investment flows, migrant remittances and aid budgets. African policymakers – and indeed ordinary Africans – are well aware of this asymmetry but seem powerless to affect it, or even to reframe the agenda of international debate.

Indeed, the prevailing orthodoxy in Washington and Brussels tends to be that this is an out-of-date socialist view, based on discredited 'dependency' theories, originating in the independence struggle. The importance of the world economy to Africa is not denied, of course, but national governments are enjoined to adjust to 'economic reality' and 'market discipline' in order to stimulate exports and promote foreign investment. The danger that the efforts of individual countries trying to adjust will lead to a 'race to the bottom' in labour standards, environmental safeguards and tax concessions is ignored; as is the 'fallacy of composition' inherent in the small-country assumption, leading to the overexpansion of commodity supplies and declining prices.

The problem is thus one of economic *perception* as well as one of economic *reality*. What has been missing, perhaps, is a modern view of the world economy from an African standpoint. Alemayehu Geda is perhaps uniquely qualified to undertake this task, since, possibly alone among young African economists, he possesses the technical skills to see beyond the platitudes of the 'North-South' debate. Thus, he examines external economic relations as a variable, rather than as a given. Hence, domestic economic events become endogenous to the model rather than simply at the behest of policy makers. This shows, on the one hand, how limited the options really are for domestic policymakers (especially if they act individually and only for short periods) and, on the one hand, the crucial importance of changing international arrangements – particularly trade and investment rules rather than aid.

In this book, Alemayehu Geda constructs a rigorous model of the world economy. This is based on earlier work by The Hague based 'Finance and Development' research group on global economic accounting and modelling. Some of this work is reported in two previous volumes in this series: Rob Vos' *Debt and Adjustment in the World Economy* (1994) and Joke Luttik's *Accounting for the World Economy* (1998). These set the foundations of how the world economy can be modelled, in a tractable fashion that allows the effect of trade and investment flows upon macroeconomic balances in any one developing region to be identified. In this way, the global framework required to analyse the full effect of aid and other capital flows from 'North' to 'South', as well as the effect of northern macroeconomic policies on the 'South' can be set up in such a way as to be compatible with the variables used in the analysis of structural adjustment.

The book opens with an outline of the historical origins of African debt and external finance problems, arguing that the debt problem results from the structure of its trade, in general, and commodity trade in particular. That the twin effects of low income elasticities and low price elasticities in northern markets have lead to declining terms of trade and high price volatility is generally agreed. The issue, then, seems to be why Africa has not switched to other export products – manufactures, services or processed raw materials – which offer better growth prospects. However, such a switch requires capital (infrastructure and plant) and skills (or 'human capital') which Africa does not currently possess. Debt and aid have both been used in an effort to overcome this constraint – but to no avail. The book then moves on to examine what existing economic theory has to say about this issue, providing a useful survey of theories of international finance, primary commodity trade models, and macroeconomic frameworks coming from an African perspective. In particular, the incisive analysis of macromodels currently being used in Africa is valuable for the insight it provides into import compression, fiscal response and 'Dutch disease' from the point of view of a domestic 'consumer' rather than an international 'producer' of knowledge. Indeed, in this book, Alemayehu Geda provides the basis for a textbook for advanced African students of economics that would take them beyond the one-country technical and dependency political perspectives.

The empirical part of the book explores the patterns of capital flows to the region and their relationship to foreign exchange earnings from commodities, where the declining terms of trade are not balanced by aid or debt flows in the short run, while in the long run these latter lead to unsustainable liability positions. Fiscal response to these external shocks and the consequences for the real exchange rate are then analysed, closing the circle, so to speak, as we return to the chronic trade deficit. On the basis of these data and the theoretical framework discussed above, Alemayehu Geda builds a simulation model of the region, which is a *tour de force*, not only in technical terms, but also as an alternative interpretation of reality. This empirical work demonstrates what should be the conventional wisdom but unfortunately is usually overlooked: that investment is the key to the problem and its possible solution.

What lessons does this book hold for the economic future of Africa? The author is too modest to put forward strong policy recommendations himself, although his main conclusions do have serious policy implications. Firstly, his view that the debt problem in Africa is essentially a commodity problem is amply confirmed. In part, this implies that the HIPC process led by the UK and the World Bank – which, while limited in scope to bilateral debt, and thus hardly reducing the bulk of the debt which is now multilateral, at least indicates a political commitment to act – will have little lasting effect unless export capacity and prices are raised. Secondly, his findings on the mixed effects of aid on

growth – both through the 'Dutch disease' effect of overvalued currencies on exports, and through the diversion of public investment away from production – imply that (a) more aid should be channelled towards small export farmers, so as to promote exports and reduce poverty; and (b) that aid should be accompanied by expansionary policies in order to keep the exchange rate competitive.

Thirdly, his conclusion that Africa is highly vulnerable to changes in world interest rates – due not to capital market effects but rather to their impact on commodity prices resulting from the activities of speculators – implies that action can only be taken at an international level as part of the construction of the new 'global financial architecture'. Fourthly, his finding that fiscal deficits are largely exogenously determined by aid flows has important implications in terms of the need for donors to co-ordinate their actions in order to ensure macroeconomic sustainability, rather than leaving this task to the IMF.

These various findings imply that trade relations will need to be improved. The question, then, is how? Improved access to northern markets for processed primary commodities, and, in particular, the replacement of the Lome system with improved access to the European market, would be a first and important step. Commodity price stabilization schemes are currently out of favour, and would require the full co-operation of the major importing transnational corporations in order to work at all. However, this is a problem of price volatility around the trend, as well as the declining trend itself. Reducing this volatility would benefit both importers and exporters and thus should not be impossible to achieve through a properly administered buffer stock system. The market mechanism alone cannot achieve this result, since hedging ranges are so short, so this would have to be a form of public intervention. However, the long-term downward direction of the terms of trade is difficult. It wouldn't matter so much if volume was increasing fast enough to raise the income terms of trade (as is happening with labour-intensive manufactures), but this is not, in fact, the case. The market for tropical commodities is oligopsonistic and riddled with restrictive practices - examples being sugar and cotton in the US, bananas and coffee in Europe as well as minerals. Therefore, a producer's cartel may be the only theoretically viable solution. However, in spite of the recent success of OPEC in driving up oil prices, Africa is unlikely to be able to organize such a cartel, in view of the worldwide competition in those commodities, which it supplies.

In fact, Africa needs to change the mix (or at the very least upgrade the quality) of its primary export products, with the continent unlikely to able to compete within the foreseeable future. This requires investment, and investment not so much by foreign companies (which is in fact already fairly high as a proportion of the total by international standards) but by domestic investors such as firms, flight capital and households. More than savings, risk is the main problem here, since there is plenty of capital held overseas and also plenty of liquidity within the banking system. However, this cannot be undertaken by each African country in isolation, but rather requires an international agreement on investment rules and stabilisation of commodity prices. In other words, the *orderly* insertion of Africa into the global market.

Valpy FitzGerald Oxford and The Hague

Acknowledgements

This study would not have been possible but for the assistance of a number of people. My heartfelt thanks go to my PhD study advisors Prof. E.V.K. FitzGerald, formerly of the Institute of Social Studies (ISS), The Hague, and now of University of Oxford, and to Prof. Rob Vos at ISS. I have learned a great deal from both. Indeed, without wishing to implicate either men with any responsibility, I have reached a point where their ideas, and those of mine, have become so intermingled that it becomes difficult to say which are theirs, and which mine. In addition to providing academic supervision, both men lent me invaluable support when my continued presence in the Netherlands, both financial and legal, was in jeopardy.

I also owe special thanks to a number of other staff members of the ISS. Firstly, to Prof. Marc Wuyts who not only taught me the importance of careful reading of data, and methodological rigour in general, but also helped me to support myself when I was in need of such help. I have benefited greatly from my frequent visits to his office, particularly when writing Chapter 4. I would also like to thank Dr Howard White who read the entire first draft of this book and gave me many constructive comments. His insight has been of great value. I have also learned a great deal from Dr Karel Jansen, who was particularly supportive at the initial stages of my work. Niek de Jong helped me to familiarise myself with the data of the ISS 'World Accounting Matrix' and provided invaluable assistance with many related issues. I have also benefited from the useful comments of Prof. John Weeks and Prof. Paul Mosley.

Dr Susan Harton (Canada), Dr Abebayehu Tegene (Washington) and the participants of the internet discussion group 'H-Africa' also deserve special thanks. George Mavrota, Alfred Maizels and Ann Roche of the Finance and Trade Policy Research Centre, Queen Elizabeth

House (OEH). University of Oxford, deserve special appreciation, (The Centre, which was effectively my second school, is directed by Prof. FitzGerald.) I have also benefited from discussions with T. Palaskas at OEH and P. Sarkar. Thanks are also due to the staff of the Food Research Unit, University of Oxford, the libraries of The London School of Economics (LSE). Institute for Development Studies (IDS). Sussex. and School of Oriental and African Studies (SOAS), as well as to D. Evans at IDS. Ato Desalegne Rahamato and Dr Abdulahamid Bedri at The Institute of Development Research, Addis Ababa University (AAU), together with my friend Kasahun Berhanu at PSIR, AAU, facilitated my fieldwork stay in Addis. My undergraduate instructors, Dr Tave Mengiste and the late Ato Mekonnen Tadesse of The Department of Economics, AAU, helped me in many ways. I would also like to thank Dr Befekadu, the late Dr Eshetu and Dr Dejene of The Department of Economics at AAU for their encouragement. Prof. O. Teriba and Dr C.T. Mwalwanda, both at the Economic Commission for Africa (ECA), the Library of ECA, Dr Henock Kifle and the Research Department of The African Development Bank, and Uri Dadush, T.G. Srivasan and M. Brahmbhtt at The International Economic Analysis and Prospects Division of the World Bank have all been helpful during the course of my study. My stay in Washington was made comfortable thanks to Rob and family, as well as Gigi, Lencha and their family. Thanks are also due to my colleague, Elias Kedir, for his help in updating part of the data upon which this study is based.

Many thanks to my colleagues at ISS and to Harm Zebregs at Erasmus University, Rotterdam. My dear friends Alex, Ashesh, Mona, Daniel, Kalpana, Karen, Maha and Phil deserve special thanks. Without the values we shared, life away from home would have been much more difficult. From them I learned to stand for the under-class. Alex, with whom I shared a room, and discussed economics, for over half a decade, deserves special mention. I would also like to thank my friend Wangu, who read the first chapter and gave me valuable comments, as well as my Ethiopian friends and colleagues Daniel H., Yeshitela G.M., Awed J., Menbere T., Meron B., Hayat A., Alemayehu S. and Getachewu Y. who lent me encouragement of various kinds. My lifelong friends Thomas, Daniel and Endrias deserve special mention. I have been cherished to enjoy their unreserved love and care since our high school days at Gelawdious, nearly a decade and a half ago. I would also like to thank the staff of various ISS departments for their help. In the Student Office, Ank van de Berg, Miss Nijhof and Marien, Jeff, Ton and Henry in the Computer Department, Dita Walenkamp and Els van der Weele in the PhD Secretariat, Joy Misa, Paula Bownas and Gary Debus in the Publications Department, Martin Blok in the Welfare Office, John and Martin in the financial office, and A. Voorvelt at ECD all deserve special thanks. My heartfelt thanks go to Mike Warner, who did an excellent job of editing this book. I would also like to extend my gratitude to the Netherlands Ministry of Development and Co-operation and the African Economic Research Consortium (AERC) for financing this study, through their generous grant. Special thanks are also due to J.C. Fine, Prof. W. Lyakurwa and Prof. B. Ndulu of AERC. (However, views expressed in the book are solely my own and do not reflect the views of these institutions.)

During my stay in the Netherlands, I was fortunate enough to become friends with Dr Gebru Mersha. He was a teacher, a father and a friend. I would also like to thank Abera A., Argawi B., Melse W.A., Emmanuel T. and Eskedar in The Hague and Nebiat in Arnhem, all of whom helped me in various ways. My family, W/o Bruke, Ato Ababu and the entire extended family have all been very supportive. And, last but surely not least, I owe a great debt to Selam, who has been a source of inspiration throughout. Her useful comments, love and care have been instrumental in bringing this work to a successful conclusion.

Alemayehu Geda Addis Ababa

1 Introduction

1.1 INTRODUCTION AND OVERVIEW

Notwithstanding the recent optimism about African economies, the performance of these economies since the time of political 'independence' can only be described as dismal. There has been a secular deterioration in their terms of trade (particularly from the mid-1970s); the level of external debt has grown enormously, leading to near insolvency; dependency on foreign aid has grown at an alarming rate (and this has been exacerbated by stagnation in exports); and, finally, levels of investment have been extremely low. Partly as a result of the latter, physical and social infrastructures have deteriorated rapidly. Political instability, frequent wars, and natural disasters have further aggravated this situation. The major question to be asked then is 'why?'

In this book an attempt to address only one part of this question – the impact of the external sector - will be made. Specifically I will focus on understanding the major causes of Africa's external finance and debt problem. It will be argued that Africa's external finance problem is a consequence of the structure of its trade in the world economy (and, in particular, its position as a commodity producer). It will also be argued, that this is compounded by the adverse effect of the macroeconomic policies adopted by 'Northern' (industrialized/ developed) countries. The purpose of this book, therefore, is to underline the existence of an economic structure, primarily built in the colonial era, which continues to shape the external economic conditions of the African continent today. Thus, it will be argued that any analysis of the external finance problem of Africa, as well as policy proposals based on such an analysis, will be incomplete if this fails to take account of such issues. The book addresses these issues using historical information, and employing both partial and general equilibrium analyses. The following are some of the major findings of the study.

First, with regard to capital flows to Africa, these are found to depend on the relative (to other 'South') position of Africa, in terms of the economic, political and strategic self-interest of donors. Humanitarian considerations are not found to explain aid flows.

Second, with regard to trade, the study shows that there is a clear difference between UNCTAD's world commodity prices and the regional export prices constructed in the context of this study. Hence, previous studies, using the former set of prices, could have biased elasticities. In order to examine determinants of exports of primary commodities, three competing models of export supply are applied to three regions of Africa for four commodity categories. Although the result is mixed, the model based on a real exchange rate is found to be comparatively the best one. This model underscores the importance of both relative prices and capital formation indicators. However, relative prices are found to affect capacity utilization but not capacity creation.

Third, two macroeconomic effects of external finance are examined, namely, fiscal response and 'Dutch disease' effects. With respect to fiscal response, the result of the empirical analysis is mixed. However, the following major points may be stated. Firstly, the impact of capital inflows on taxes varies, depending on the type of capital inflows, nature of taxes and region. In general, bilateral flows are found to have a negative impact on direct taxes. Capital inflows, in general, are also found to have a strong positive impact on current government expenditure in all regions. Capital inflows to Africa, generally, were disbursed conditional on deficit reduction. However, these inflows have an inherent tendency of aggravating the deficit and, consequently, may result in governments drifting away from sustainable self-financing behaviour. The study also shows that there is evidence of the 'Dutch disease' effect in all regions. Interestingly, in almost all cases, government spending on the non-traded sector is found to have a statistically significant negative elasticity. Thus, insofar as part of this spending is financed by foreign inflows, the 'Dutch disease' effect may be observed indirectly. This is entirely plausible since most capital flows to Africa are directed to the public sector.

The last important sets of findings are derived using the global model developed in the book. From simulation of different policies and external shocks, the following major results are obtained. Firstly, the effect of aid on Africa depends on the manner of its financing in the 'North'. Compared to budget financing, deficit financing of aid is found to have an adverse effect on commodity prices and, hence, export revenue. If bond/deficit financing is deemed the preferred means of financing, there may emerge asymmetric relations in policy choice between the 'South' and the 'North'. Secondly, the exact effects of this policy will vary, depending on the specific dynamics of each commodity. Thus, depending on the composition of exports, different regions may experience different effects, for the same policy shock. Thirdly, unlike some global models, where aid leads to a rise in South's GDP (e.g., MULTIMOD, see Jamshidi 1989: 38), the impact of such aid in Africa is not only found to vary across regions but also to be generally negligible. The latter is the result of both the degree of disaggregation and the inclusion of the macroeconomic effect of aid in the model developed within this book. Thus, although aid might initially result in a rise in the level of imports, and hence output, the macroeconomic ('Dutch disease' and 'fiscal response') and terms of trade effects of such aid work against this initial positive impact, resulting in no noticeable effect on output. Finally, the study shows that debt cancellation is not a panacea to the African debt problem, since these problems are intricately linked with trade problems. It is also shown that its impact can vary drastically between regions. However, debt cancellation may alleviate the pressure on the fiscal balance of governments, both by reducing the expenditure burden and by providing positive signals to private lenders.

The remainder of this chapter is concerned with providing a solid conceptual background to the various questions and findings noted above. In section 1.2 the external finance problems of Africa are described. The section concludes by highlighting a number of economic implications of the debt problem for Africa. Section 1.3 describes how the debt issue has been examined during the course of a recent policy debate about African economic problems in general. Section 1.4 examines how the structure of African economies may usefully be seen as a legacy of its colonial history. Section 1.5 highlights a number of important methodological issues. Finally, section 1.6 sets out the organization of the remainder of this book.

| ଚ |
|----------|
| ate |
| Ste |
| ő |
| Ň |
| ē |
| ħ |
| S |
| es |
| <u>I</u> |
| 5 |
| a S |
| ilo. |
| 0 |
| S |
| 5 |
| ŝ |
| õ |
| liic |
| n L |
| 0 |
| ŝ. |
| 4fr |
| S |
| ŝ |
| ŏ |
| ät |
| ij |
| 2 |
| \$pt |
| g |
| õ |
| /aj |
| < |
| - |
| - |
| ble |
| ~ |

| Table 1.1 Major debt indicators for A | Africa (in t | illions of | US dollar | s, unless | otherwis | e stated) | |
|---|--------------|------------|-----------|-----------|----------|-----------|-------|
| | 1971 | 1975 | 1980 | 1985 | 1990 | 1991 | 1992 |
| Total external debt stock (EDT) | | | | | | | |
| East and Southern Africa (ESA) | 4.9 | 11.4 | 28.3 | 55.4 | 85.2 | 88.8 | 91.1 |
| North Africa (NA) | 5.1 | 12.9 | 51.3 | 75.0 | 93.0 | 90.9 | 88.7 |
| West and Central Africa (WCA) | 3.9 | 8.2 | 32.2 | 51.7 | 91.7 | 94.2 | 90.9 |
| Sub-Saharan Africa | 8.8 | 19.7 | 60.8 | 107.1 | 176.9 | 183.4 | 182.7 |
| All Africa | 13.9 | 32.6 | 112.1 | 182.1 | 269.8 | 274.2 | 271.4 |
| Long-term external debt (total, All Africa) Mutrilateral (DOD) | 11.9 | 27.2 | 84.6 | 140.1 | 227.7 | 231.7 | 227.2 |
| East and Southern Africa | 0.5 | 1.4 | 4.0 | 8.8 | 18.9 | 20.7 | 21.5 |
| North Africa | 0.2 | 0.6 | 4.1 | 7.2 | 12.4 | 13.8 | 14.2 |
| West and Central Africa | 0.5 | 1.2 | 3.6 | 7.8 | 19.4 | 20.9 | 21.7 |
| Sub-Saharan Africa | 1.0 | 2.6 | 7.6 | 16.7 | 38.2 | 41.6 | 43.2 |
| All Africa | 1.2 | 3.2 | 11.7 | 23.9 | 50.6 | 55.4 | 57.4 |
| Bilateral | | | | | | | |
| East and Southern Africa | 2.6 | 5.0 | 11.1 | 23.7 | 36.6 | 37.3 | 37.2 |
| North Africa | 3.0 | 6.0 | 17.5 | 31.9 | 36.6 | 37.8 | 38.2 |
| West and Central Africa | 2.0 | 3.2 | 6.9 | 10.5 | 33.9 | 36.7 | 37.1 |
| Sub-Saharan Africa | 4.6 | 8.3 | 18.1 | 34.2 | 70.5 | 74.0 | 74.4 |
| All Africa | 7.6 | 14.3 | 35.6 | 66.1 | 107.2 | 111.7 | 112.6 |
| Private creditors (DOD) | | | | | | | |
| East and Southern Africa | 1.1 | 2.6 | 5.4 | 8.6 | 13.3 | 13.3 | 13.0 |
| North Africa | 1.2 | 4.8 | 21.0 | 23.8 | 34.5 | 30.8 | 29.0 |
| West and Central Africa | 0.7 | 2.3 | 10.8 | 17.6 | 22.1 | 20.2 | 14.5 |
| Sub-Saharan Africa | 1.9 | 4.9 | 16.3 | 26.3 | 35.4 | 33.9 | 28.2 |
| All Africa | 3.0 | 9.7 | 37.3 | 50.1 | 69.9 | 64.7 | 57.2 |
| Interest & principal arrears (% of total external | debt) | | | | | | |
| East and Southern Africa | 0.3 | 3.6 | 10.3 | 16.4 | 21.1 | 25.2 | 28.4 |
| North Africa | 1.5 | 0.2 | 0.0 | 6.9 | 6.3 | 0.9 | 0.3 |
| West and Central Africa | 2.8 | 3.9 | 1.6 | 3.1 | 9.6 | 10.3 | 14.1 |
| Sub-Saharan Africa | 1.4 | 3.7 | 5.6 | 10.0 | 15.2 | 17.5 | 21.2 |
| All Africa | 1.4 | 2.3 | 3.5 | 8.7 | 12.1 | 12.0 | 14.4 |

(Continued)

4

| Continued) | |
|------------|--|
| 1.1 | |
| able | |

| Table 1.1 (Continued) | | | | | | |
|---|-------|-------|-------|-------|-------|-------|
| | 1993 | 1994 | 1995 | 1996 | 1997 | 1998 |
| Total external debt stock (EDT) | | | | | | |
| East and Southern Africa (ESA) | 93.4 | 101.4 | 106.0 | 104.0 | 103.2 | 107.3 |
| North Africa (NA) | 86.7 | 94.1 | 9.66 | 98.0 | 92.2 | 94.4 |
| West and Central Africa (WCA) | 94.7 | 98.2 | 103.9 | 101.7 | 94.5 | 98.1 |
| Sub-Saharan Africa | 194.8 | 221.3 | 235.4 | 231.8 | 223.1 | 230.1 |
| All Africa | 281.5 | 315.4 | 335.0 | 329.9 | 315.3 | 324.5 |
| Long-term external debt (total, All Africa) | 234.0 | 253.0 | 266.9 | 260.9 | 249.2 | 256.3 |
| | : | | | | | |
| East and Southern Africa | 23.4 | 25.9 | 27.7 | 28.0 | 28.0 | 29.8 |
| North Africa | 15.4 | 17.0 | 18.4 | 18.2 | 17.2 | 18.5 |
| West and Central Africa | 22.8 | 25.3 | 27.1 | 26.7 | 25.6 | 27.3 |
| Sub-Saharan Africa | 46.1 | 51.2 | 54.7 | 54.7 | 53.6 | 57.0 |
| All Africa | 61.4 | 68.2 | 73.0 | 72.9 | 70.7 | 75.6 |
| Bilateral | | | | | | |
| East and Southern Africa | 36.8 | 39.2 | 40.0 | 39.5 | 38.2 | 39.8 |
| North Africa | 38.1 | 44.4 | 49.8 | 50.7 | 48.0 | 49.1 |
| West and Central Africa | 37.0 | 40.4 | 42.8 | 41.2 | 38.7 | 40.7 |
| Sub-Saharan Africa | 73.8 | 79.6 | 82.7 | 80.7 | 77.0 | 80.5 |
| All Africa | 111.9 | 124.0 | 132.6 | 131.4 | 125.0 | 129.7 |
| Private creditors (DOD) | | | | | | |
| East and Southern Africa | 13.0 | 12.9 | 13.5 | 12.6 | 12.6 | 12.3 |
| North Africa | 27.3 | 26.1 | 24.5 | 21.5 | 18.7 | 17.6 |
| West and Central Africa | 14.6 | 13.9 | 13.4 | 12.2 | 10.6 | 10.6 |
| Sub-Saharan Africa | 33.4 | 34.6 | 36.8 | 35.1 | 34.7 | 33.6 |
| All Africa | 60.7 | 60.8 | 61.3 | 56.6 | 53.4 | 51.1 |
| Interest & principal arrears (% of total external | debt) | | | | | |
| East and Southern Africa | 32.5 | 34.0 | 36.7 | 35.0 | 34.2 | 35.4 |
| North Africa | 0.3 | 0.3 | 0.4 | 0.5 | 0.5 | 0.5 |
| West and Central Africa | 19.2 | 19.5 | 21.9 | 23.8 | 22.7 | 25.6 |
| Sub-Saharan Africa | 25.0 | 24.2 | 26.3 | 26.2 | 25.5 | 27.5 |
| All Africa | 17.4 | 17.1 | 18.6 | 18.6 | 18.2 | 19.6 |
| | | | | | | |

Source: World Bank Global Development Finance (2000).

| 5 |
|--|
| ă |
| 1 |
| τ <u>α</u> |
| ŝ |
| ~ |
| × |
| .2 |
| ~ |
| < |
| d) |
| Ē |
| 4 |
| 0 |
| ŝ |
| õ |
| ٥Ś. |
| ž |
| S |
| 3 |
| |
| 8 |
| 2 |
| Ľ, |
| õ |
| ž |
| |
| S |
| 5 |
| \sim |
| 5 |
| 0 |
| ŝ |
| ř |
| 2 |
| .≚ |
| 3 |
| õ |
| - |
| .5 |
| 5 |
| G |
| ö |
| |
| ē |
| ţ, |
| Afri |
| r Afri |
| or Afri |
| for Afri |
| s for Afri |
| rs for Afri |
| ors for Afri |
| ators for Afri |
| ators for Afri |
| licators for Afri |
| dicators for Afri |
| indicators for Afri |
| indicators for Afri |
| of indicators for Afri |
| ebt indicators for Afri |
| tebt indicators for Afri |
| debt indicators for Afri |
| r debt indicators for Afri |
| ior debt indicators for Afri |
| ajor debt indicators for Afri |
| Aajor debt indicators for Afri |
| Major debt indicators for Afri |
| Major debt indicators for Afri |
| Major debt indicators for Afri |
| 2 Major debt indicators for Afri |
| .2 Major debt indicators for Afri |
| 1.2 Major debt indicators for Afri |
| 9.1.2 Major debt indicators for Afri |
| le 1.2 Major debt indicators for Afri |
| ble 1.2 Major debt indicators for Afri |

| Table 1.2 Major debt indicators | for Africa | (in billior | is of US c | Iollars, un | less other | wise stated) | |
|---|-------------|----------------|----------------|-------------|------------|--------------|--------|
| | 1971 | 1975 | 1980 | 1985 | 1990 | 1991 | 1992 |
| Net transfer on debt | | | | | | | |
| East and Southern Africa | 1.09 | 2.19 | 2.48 | 2.49 | 2.07 | 0.80 | 2.45 |
| North Africa | 0.54 | 4.68 | 2.25 | -0.06 | -5.64 | -3.39 | -5.71 |
| West and Central Africa | 0.40 | 1.12 | 4.01 | -1.89 | -0.27 | -2.22 | -2.02 |
| Sub-Saharan Africa | 1.50 | 3.31 | 6.56 | 0.55 | 1.78 | -1.18 | 0.74 |
| All Africa | 2.04 | 7.99 | 8.81 | 0.49 | -3.86 | -4.57 | 4.98 |
| Aggregate net transfer | | | | | | | |
| East and Southern Africa | 1.08 | 2.30 | 4.18 | 5.37 | 7.62 | 7.57 | 7.55 |
| North Africa | 0.46 | 5.19 | 2.95 | 2.60 | 3.08 | 0.54 | -0.75 |
| West and Central Africa | 0.29 | 1.14 | 1.16 | -1.28 | 2.32 | 1.80 | 1.49 |
| Sub-Saharan Africa | 1.43 | 3.41 | 5.11 | 4.44 | 9.23 | 8.97 | 8.83 |
| All Africa | 1.90 | 8.60 | 8.06 | 7.04 | 12.31 | 9.51 | 8.08 |
| Technical co-operation grants (as % of total | grants) | | | | | | |
| East and Southern Africa | 61.81 | 36.39 | 36.77 | 32.06 | 24.31 | 25.61 | 26.44 |
| North Africa | 38.70 | 16.33 | 58.53 | 39.71 | 20.31 | 22.50 | 31.49 |
| West and Central Africa | 51.54 | 46.90 | 54.37 | 36.61 | 30.04 | 31.78 | 35.62 |
| Sub-Saharan Africa | 61.36 | 49.23 | 42.60 | 35.52 | 29.76 | 31.79 | 32.91 |
| All Africa | 50.03 | 32.78 | 50.57 | 37.61 | 25.04 | 27.14 | 32.20 |
| Debt (EDT)/GNP (%)(simple arithmetic mean, | based on co | ountries that | have relevar | it data) | | | |
| East and Southern Africa | 19.87 | 23.12 | 44.32 | 77.66 | 99.45 | 94.71 | 115.66 |
| North Africa | 29.33 | 31.56 | 57.07 | 84.27 | 71.98 | 75.11 | 68.30 |
| West and Central Africa | 21.35 | 27.61 | 59.61 | 106.68 | 121.16 | 127.89 | 147.63 |
| Sub-Saharan Africa | 14.55 | 15.52 | 23.45 | 56.37 | 63.01 | 63.68 | 62.91 |
| All Africa | 21.94 | 23.54 | 40.26 | 70.32 | 67.50 | 69.39 | 65.61 |
| Debt service ratio (%)(simple arithmetic mean | 1, based on | countries that | at have releva | ant data) | | | |
| East and Southern Africa | 8.54 | 9.63 | 14.23 | 21.89 | 20.87 | 21.73 | 18.79 |
| North Africa | 21.77 | 8.52 | 22.27 | 30.24 | 32.96 | 34.58 | 37.21 |
| West and Central Africa | 5.28 | 6.88 | 13.16 | 22.08 | 19.29 | 18.12 | 15.83 |
| Sub-Saharan Africa | : | : | 7.20 | 17.58 | 12.92 | 12.46 | 12.28 |
| All Africa | : | : | 14.73 | 23.91 | 22.94 | 23.52 | 24.74 |

⁽Continued)

| 193 1994 1995 1997 1998 1997 1998 Net transfer on debt East and Southern Africa 0.84 0.04 0.47 0.45 1.65 -1.32 North Africa 5.85 2.80 0.57 -1.33 0.31 -3.28 North Africa 5.85 2.80 -0.57 -1.33 0.31 -3.28 North Africa 0.84 0.04 0.47 -0.45 1.65 -1.32 North Africa 5.85 2.80 -0.57 -3.33 0.31 -3.28 Sub-Saharan Africa 0.83 2.80 -0.57 -3.33 0.31 -3.28 Aggregate net transfer 6.46 5.66 -1.32 -6.15 -4.71 -12.46 Sub-Saharan Africa 0.121 11.76 15.36 3.05 2.72 0.23 -6.48 Sub-Saharan Africa 10.21 11.76 15.36 3.05 2.72 0.25 2.49 Sub-Saharan Africa 10.21 15.38 3.66 | lable 1.2 (Continued) | | | | | | |
|--|--|---------------|-----------------|--------------|---------|--------|--------|
| Net transfer on debt Net transfer on debt Bast and Southern Africa 0.84 -0.04 0.47 -0.45 1.65 -1.32 North Africa 0.88 -0.04 0.47 -0.45 1.65 -1.32 North Africa 0.88 -0.04 0.47 -0.45 1.65 -1.32 West and Central Africa 0.83 -1.32 -6.15 -4.71 -12.45 Sub-Saharan Africa 0.83 -1.32 -6.15 -4.71 -12.45 Aggregate and central Africa -3.32 2.88 -1.32 -6.15 -4.71 -12.45 Aggregate and central Africa -1.80 0.39 -1.54 -0.23 0.22 -1.54 Sub-Saharan Africa -1.80 0.39 -1.54 -4.71 -12.45 Aggregate and Southern Africa 10.21 11.75 15.36 2.77 2.52 1.54 Sub-Saharan Africa 10.21 11.75 15.36 2.07 2.23 2.86 Sub-Saharan Africa 5.13 | | 1993 | 1994 | 1995 | 1996 | 1997 | 1998 |
| East and Southern Africa 0.84 -0.04 0.47 0.45 1.65 -1.32 West and Cartral Africa 0.83 -2.80 -3.33 -3.33 -3.33 -3.28 West and Cartral Africa 0.83 -2.80 -3.33 -3.33 -3.33 -3.28 Sub-Saharan Africa 0.83 -2.80 -3.32 -5.86 -5.76 -1.57 0.15 -5.48 Aggregate net transfer -3.32 2.66 5.76 -3.77 5.37 4.13 Aggregate net transfer -3.32 2.66 5.76 3.77 5.37 4.14 North Africa -1.180 0.39 -1.54 -0.22 0.29 -220 North Africa 10.21 11.75 15.36 8.66 5.76 -1.245 Sub-Saharan Africa 10.21 11.75 15.45 -3.77 5.37 4.18 North Africa 10.21 11.75 15.36 8.66 5.76 3.126 Sub-Saharan Africa 5.01 2.318 </td <td>Net transfer on debt</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> | Net transfer on debt | | | | | | |
| North Africa -5.85 -2.80 -3.33 -3.58 -4.86 -5.97 West and Central Africa 0.83 -2.80 -3.33 -3.58 -0.57 -1.33 0.31 -3.28 Sub-Saharan Africa 2.83 -2.80 -3.33 -5.86 -1.32 -5.15 -4.71 -12.45 Algregate net transfer 5.46 5.66 5.76 3.77 5.37 4.13 Agregate net transfer 6.46 5.66 5.76 3.77 5.37 4.13 Bast and Southern Africa 1.10 1.54 0.23 0.25 1.246 North Africa 10.21 11.175 15.36 8.66 13.33 6.68 North Africa 10.21 11.175 15.36 3.125 2.49 All Africa 10.21 11.175 15.36 3.136 4.48 All Africa 3.125 13.64 4.00 4.332 2.522 1.95 North Africa 3.166 6.173 3.265 3.126 <td>East and Southern Africa</td> <td>0.84</td> <td>-0.04</td> <td>0.47</td> <td>-0.45</td> <td>1.65</td> <td>-1.32</td> | East and Southern Africa | 0.84 | -0.04 | 0.47 | -0.45 | 1.65 | -1.32 |
| West and Central Africa 0.83 2.80 0.57 -1.33 0.31 -3.28 Sub-Saharan Africa 3.32 5.48 2.01 -5.57 0.15 -4.71 -12.45 Aggregate net transfer 3.32 5.48 2.01 -5.57 0.15 -5.48 Aggregate net transfer 3.77 5.37 4.13 -5.37 4.13 Aggregate net transfer 6.46 5.56 5.76 3.77 5.37 4.13 North Africa 10.21 11.75 15.36 8.66 13.361 4.48 North Africa 10.21 11.75 15.36 8.68 13.361 4.48 Sub-Saharan Africa 10.21 11.76 13.81 8.43 13.61 4.48 North Africa 10.21 11.76 13.81 8.43 13.61 4.48 Sub-Saharan Africa 52.21 38.85 51.19 30.01 32.55 148 North Africa 52.21 38.85 51.19 30.01 3 | North Africa | -5.85 | -2.80 | -3.33 | -3.58 | 4.86 | -5.97 |
| Sub-Saharan Africa 2.53 5.48 2.01 -2.57 0.15 -6.48 All Africa -3.32 2.68 -1.32 -6.15 -1.71 -12.45 Aggregate net transfer -3.32 2.68 5.76 3.77 5.37 -4.13 North Africa -1.80 0.39 -1.54 -0.23 0.29 -2.20 North Africa -1.80 0.39 -1.54 -0.23 0.29 -2.20 North Africa -1.80 0.39 -1.54 -0.23 0.29 -2.20 Sub-Saharan Africa 10.1 1.7.56 15.36 8.66 13.361 4.48 Sub-Saharan Africa 10.1 1.7.56 15.36 8.66 13.32 2.49 Sub-Saharan Africa 10.1 1.7.61 1.2.45 3.05 2.49 3.73 3.05 Mit Africa Sub-Saharan Africa 37.18 2.8.40 4.0.94 3.3 3.0.5 2.4.94 Nest and Central Africa 37.18 2.8.40 | West and Central Africa | 0.83 | -2.80 | -0.57 | -1.33 | 0.31 | -3.28 |
| All Africa -3.32 2.68 -1.32 -6.15 -4.71 -12.45 Aggregate net transfer 6.46 5.66 5.76 3.77 5.37 -4.13 Rest and Southern Africa 6.46 5.66 5.76 3.77 5.37 -4.13 Next and Central Africa -1.80 0.39 -1.54 -0.23 0.29 -2.20 West and Central Africa -1.175 15.36 8.66 13.33 6.68 Sub-Saharan Africa 0.021 11.75 15.36 8.66 13.33 6.68 Sub-Saharan Africa 0.021 11.75 13.81 8.43 13.61 4.48 Technical Comperation grants (as % of total grants) 8.41 12.14 13.81 8.43 13.61 4.48 North Africa 3.01 3.85 5.1.19 4.004 4.33 3.255 Sub-Saharan Africa 3.7.18 3.4.5 3.0.70 3.2.68 3.7.31 3.0.53 East and Southern Africa 3.7.18 3.2.26 3.2. | Sub-Saharan Africa | 2.53 | 5.48 | 2.01 | -2.57 | 0.15 | -6.48 |
| Aggregate net transfer Aggregate net transfer Aggregate net transfer 6.46 5.66 5.76 3.77 5.37 4.13 East and Southern Africa -1.80 0.39 -1.54 0.23 -2.20 Worth Africa -1.80 0.39 -1.54 0.23 -2.22 Worth Africa -1.80 0.39 -1.54 0.23 -2.22 Worth Africa -1.175 15.36 8.66 13.33 6.68 Sub-Saharan Africa 10.21 11.75 15.36 8.66 13.33 6.88 Sub-Saharan Africa 10.21 11.75 15.36 8.66 13.33 6.88 All Africa 8.41 12.14 13.81 8.43 13.61 4.48 Technical co-operation grants (as % of total grants) 28.94 20.06 33.36 24.94 North Africa 52.11 13.61 4.18 31.36 24.94 Not Africa 53.21 38.85 51.19 30.33 32.325 < | All Africa | -3.32 | 2.68 | -1.32 | -6.15 | 4.71 | -12.45 |
| East and Southern Africa 6.46 5.66 5.76 3.77 5.37 4.13 North Africa -1.80 0.39 -1.54 -0.23 0.29 -2.20 North Africa 10.21 11.75 15.36 8.66 13.33 6.68 Sub-Staran Africa 10.21 11.75 15.36 8.66 13.33 6.68 Sub-Staran Africa 10.21 11.21 13.81 8.43 13.61 4.48 Sub-Sahara Africa 10.21 11.21 12.14 13.81 8.43 33.55 5.49 Sub-Sahara Africa 28.94 24.26 20.03 33.03 31.25 24.94 West and Southerm Africa 28.94 24.5 30.70 32.69 33.53 West and Central Africa 37.18 28.45 30.70 32.69 37.31 30.53 Sub-Saharan Africa 57.19 23.65 40.94 35.36 5.81 30.53 Sub-Saharan Africa 57.13 26.84 33.65 40.94 | Aggregate net transfer | | | | | | |
| North Africa -1.80 0.39 -1.54 -0.23 0.29 -2.20 West and Central Africa 1.175 15.45 3.05 2.77 2.52 1.95 Sub-Saharan Africa 1.175 15.35 8.66 1.33 6.68 Sub-Saharan Africa 8.41 12.14 1.381 8.43 13.61 4.48 Technical co-operation grants 8.41 12.14 1.381 8.43 13.61 4.48 East and Southern Africa 8.41 12.14 1.381 8.43 13.61 4.48 North Africa 8.45 30.70 33.88 51.19 40.04 43.32 28.86 Sub-Saharan Africa 33.65 28.45 30.70 32.69 31.30 27.82 North Africa 33.65 41.51 133.18 14.43 30.53 37.31 30.53 Dist Africa 52.1 133.18 14.595 141.51 30.53 37.31 30.53 30.53 North Africa 167.20 | East and Southern Africa | 6.46 | 5.66 | 5.76 | 3.77 | 5.37 | 4.13 |
| West and Central Africa 4.19 5.45 3.05 2.77 2.52 1.95 Sub-Saharan Africa 10.21 11.75 15.36 8.66 13.33 6.68 All Africa 10.21 11.75 15.36 8.66 13.33 6.68 Technica 28.94 8.41 12.14 13.81 8.43 13.61 4.48 Technica 28.94 28.94 28.94 28.95 51.19 40.04 43.32 33.25 North Africa 52.21 38.85 51.19 40.04 43.32 33.25 Sub-Saharan Africa 37.18 28.45 30.70 32.53 38.85 Sub-Saharan Africa 33.68 28.45 30.70 32.53 30.53 Sub-Saharan Africa 33.66 33.06 34.00 111.55 56.81 North Africa 67.02 69.68 68.75 61.36 57.14 57.14 East and Southern Africa 16.72 141.51 133.16 14.627 163 | North Africa | -1.80 | 0.39 | -1.54 | -0.23 | 0.29 | -2.20 |
| Sub-Saharan Africa 10.21 11.75 15.36 8.66 13.33 6.68 All Africa 8.41 12.14 13.81 8.43 13.61 4.48 Technical co-operation grants 8.41 12.14 13.81 8.43 13.61 4.48 Technical co-operation grants 8.41 12.14 13.81 8.43 13.61 4.48 Technical co-operation grants 8.41 12.14 13.81 8.43 13.61 4.48 Rest and Central Africa 28.94 24.26 29.06 33.98 31.25 24.94 West and Central Africa 37.18 28.81 32.06 30.01 32.53 28.86 Netst and Central Africa 43.04 33.86 28.45 30.70 32.65 31.30 55.81 Debt (EDT)/GNP (%) (simple arithmetic mean, based on countries that have relevant data) 114.93 103.40 111.55 55.81 North Africa 65.05 145.95 145.95 144.93 103.40 111.55 55.81 | West and Central Africa | 4.19 | 5.45 | 3.05 | 2.77 | 2.52 | 1.95 |
| All Africa 8.41 1.2.14 1.3.81 8.43 1.3.61 4.48 Technical co-operation grants 8.41 1.2.14 1.3.81 8.43 1.2.14 1.3.81 8.43 1.4.8 Technical co-operation grants 8.94 1.2.14 1.3.81 8.4.9 1.2.14 1.3.81 8.4.9 1.4.8 East and Southern Africa 2.8.94 2.8.94 2.8.06 30.01 32.55 24.94 West and Central Africa 3.7.18 2.8.45 30.70 32.66 31.30 27.82 North Africa 3.7.16 3.3.65 2.8.45 30.70 32.65 31.30 27.82 Debt (EDT)/GNP (%) (simple arithmetic mean, based on countries that have relevant data) 114.13 103.40 111.55 North Africa 16.7.02 69.68 68.75 61.36 67.14 72.32 Sub-Saharan Africa 16.7.20 158.63 145.95 145.27 72.32 North Africa 16.7.20 133.18 63.16 67.14 72.32 | Sub-Saharan Africa | 10.21 | 11.75 | 15.36 | 8.66 | 13.33 | 6.68 |
| Technical co-operation grants (as % of total grants) 28.94 24.26 29.06 33.36 31.25 24.94 North Africa 58.94 24.26 29.06 33.36 31.25 24.94 North Africa 57.19 30.01 32.53 33.25 33.25 North Africa 57.19 28.45 30.01 32.53 33.25 Nubst and Central Africa 33.86 28.45 30.70 32.56 37.31 30.53 Nubst and Central Africa 33.86 28.45 30.70 32.69 37.31 30.53 Debt (EDT)/GNP (%) (simple arithmetic mean, based on countries that have relevant data) 111.55 133.18 114.93 103.40 111.55 North Africa 67.02 69.06 77.16 133.18 114.93 103.40 111.55 North Africa 167.20 69.06 77.57 73.32 66.107 Sub-Saharan Africa 167.20 158.65 145.55 145.27 72.32 Sub-Saharan Africa 167.20 158.65 1 | All Africa | 8.41 | 12.14 | 13.81 | 8.43 | 13.61 | 4.48 |
| East and Southern Africa 28.94 24.26 29.06 33.98 31.25 24.94 North Africa 52.21 38.85 51.19 40.04 43.32 33.25 West and Central Africa 57.18 26.81 32.06 30.01 32.53 33.25 West and Central Africa 37.18 26.81 32.06 30.01 32.53 33.25 SubSaharan Africa 37.86 31.06 30.01 32.53 33.25 All Africa 37.31 30.53 27.05 30.69 31.30 27.82 Debt (ED7)/GNP (%) (simple arithmetic mean, based on countries that have relevant data) 111.15 103.40 111.155 North Africa 67.02 69.68 68.75 61.36 56.81 Sub-saharan Africa 167.12 141.51 133.16 114.93 103.40 111.55 North Africa 67.02 69.68 68.75 61.36 56.81 56.81 Sub-saharan Africa 167.12 141.51 133.16 67.24 72.32 <td>Technical co-operation grants (as % of tota</td> <td>ll grants)</td> <td></td> <td></td> <td></td> <td></td> <td></td> | Technical co-operation grants (as % of tota | ll grants) | | | | | |
| North Africa 52.21 38.85 51.19 40.04 43.32 33.25 West and Central Africa 37.18 26.81 32.06 30.01 32.53 28.86 West and Central Africa 37.18 26.81 32.06 30.01 32.53 28.86 Sub-Saharan Africa 37.18 28.45 40.04 43.32 33.35 All Africa 3.3.65 40.07 35.36 31.30 37.31 30.53 Debt (EDT)/GNP (%) (simple arithmetic mean, based on countries that have relevant data) 36.36 31.31 30.53 37.31 30.53 North Africa 116.12 141.51 133.18 114.93 103.40 111.55 North Africa 67.02 69.68 68.75 61.36 58.15 55.81 West and Central Africa 16.72 158.63 145.95 146.27 163.02 West and Southern Africa 67.02 69.68 68.75 61.36 57.74 72.32 Netst and Southern Africa 167.20 158.63 | East and Southern Africa | 28.94 | 24.26 | 29.06 | 33.98 | 31.25 | 24.94 |
| West and Central Africa 37.18 2.6.81 3.2.06 30.01 32.53 28.86 Sub-Saharan Africa 33.86 2.8.45 30.70 32.69 31.30 27.82 Alt Africa 33.86 2.8.45 30.70 32.69 31.30 27.82 Debt (EDT)/GNP (%) (simple arithmetic mean, based on countries that have relevant data) 31.30 31.30 27.82 Debt (EDT)/GNP (%) (simple arithmetic mean, based on countries that have relevant data) 31.30 31.30 31.53 North Africa 67.02 69.68 68.75 61.36 58.15 55.81 North Africa 16.72 158.63 145.95 146.27 163.02 23.32 North Africa 67.02 69.68 68.75 61.36 57.74 72.32 Nest and Central Africa 167.20 158.63 145.95 146.27 163.02 Sub-Saharan Africa 67.00 77.57 73.82 67.74 72.32 North Africa 167.20 158.63 145.95 146.27 164.07 | North Africa | 52.21 | 38.85 | 51.19 | 40.04 | 43.32 | 33.25 |
| Sub-Saharan Africa 33.86 28.45 30.70 32.69 31.30 27.82 All Africa 43.04 33.65 40.94 36.36 37.31 30.53 Debt (EDT)/GNP (%) (simple arithmetic mean, based on countries that have relevant data) 116.12 111.61 133.18 113.33 113.33 113.36 55.81 30.53 North Africa 67.02 69.68 68.75 61.36 56.15 55.81 North Africa 16.72 158.63 145.95 146.27 133.18 56.15 55.81 North Africa 67.02 69.68 68.75 61.36 56.15 55.81 West and Central Africa 16.72 158.63 145.95 146.27 72.32 Sub-Saharan Africa 77.16 77.57 73.82 67.74 73.32 Bebt service ratio (%)(simple arithmetic mean, based on countries that have relevant data) 73.46 77.46 72.32 Bebt service 741 73.56 145.52 145.52 15.92 17.69 Best and Sou | West and Central Africa | 37.18 | 26.81 | 32.06 | 30.01 | 32.53 | 28.86 |
| All Africa 43.04 33.65 40.94 36.36 37.31 30.53 Debt (EDT)/GNP (%) (simple arithmetic mean, based on countries that have relevant data) 30.36.3 37.31 30.53 Debt (EDT)/GNP (%) (simple arithmetic mean, based on countries that have relevant data) 133.18 114.93 103.40 111.55 East and Southern Africa 67.02 69.87.5 61.36 55.81 111.55 North Africa 167.20 69.68.7.5 145.95 146.27 163.02 23.2 Sub-Saharan Africa 167.20 158.63 145.95 146.27 153.02 23.2 Sub-Saharan Africa 167.20 158.63 145.95 146.27 153.02 23.3 Both Africa 167.20 158.63 145.95 146.27 153.02 23.3 Nuch Africa 167.20 158.63 145.95 146.27 153.02 23.3 Both Africa 77.46 67.59 62.94 64.07 23.2 24.9 27.49 07.3 Both Africa 73.46 | Sub-Saharan Africa | 33.86 | 28.45 | 30.70 | 32.69 | 31.30 | 27.82 |
| Debt (EDT)/GNP (%) (simple arithmetic mean, based on countries that have relevant data) 111.55 East and Southern Africa 116.12 141.51 133.18 114.93 103.40 111.55 North Africa 67.02 69.68 68.75 61.36 55.81 North Africa 67.20 158.65 145.27 163.02 55.81 North Africa 67.20 158.65 145.27 163.02 55.81 North Africa 70.99 82.57 145.27 153.02 55.81 Sub-Saharan Africa 70.99 82.146 57.32 67.74 72.32 All Africa 69.00 76.13 73.16 67.59 62.94 64.07 East and Southern Africa 17.55 14.86 21.26 15.26 17.49 North Africa 36.49 29.86 24.48 21.15 20.10 22.38 North Africa 17.55 14.86 21.26 15.92 17.49 North Africa 22.24 22.112 19.88 17.69 | All Africa | 43.04 | 33.65 | 40.94 | 36.36 | 37.31 | 30.53 |
| East and Southern Africa 116.12 141.51 133.18 114.93 103.40 111.55 North Africa 67.02 69.68 68.75 61.36 58.15 55.81 Next and Central Africa 167.20 158.63 145.95 61.36 58.15 55.81 West and Central Africa 167.20 158.63 145.95 146.27 163.02 55.81 Sub-Saharan Africa 167.20 158.63 145.95 145.27 163.02 56.14 72.32 All Africa 69.00 76.13 73.16 67.59 62.94 64.07 Debt service ratio (%)(simple arithmetic mean, based on countries that have relevant data) 17.55 14.86 21.26 15.26 15.49 North Africa 36.49 29.86 28.48 21.15 20.10 22.38 North Africa 17.55 14.86 21.12 18.39 15.92 17.08 North Africa 36.49 22.112 18.98 17.69 17.41 18.53 North Africa | Debt (EDT)/GNP (%) (simple arithmetic mear | 1, based on c | ountries that h | ave relevant | data) | | |
| North Africa 67.02 69.68 68.75 61.36 58.15 55.81 West and Central Africa 167.20 158.63 145.95 146.27 153.02 55.81 West and Central Africa 167.20 158.63 145.95 146.27 153.02 55.81 Sub-Saharan Africa 77.57 73.82 67.74 72.32 All Africa 70.99 82.57 73.16 62.94 64.07 Debt service ratio (%)(simple arithmetic mean, based on countries that have relevant data) 62.94 64.07 23.16 12.39 15.49 North Africa 17.55 14.86 21.26 15.26 12.39 15.49 North Africa 36.49 29.86 24.48 21.15 20.10 22.38 North Africa 14.52 21.12 18.98 18.39 15.92 17.69 Sub-Saharan Africa 22.04 22.11 19.88 17.69 17.41 18.53 | East and Southern Africa | 116.12 | 141.51 | 133.18 | 114.93 | 103.40 | 111.55 |
| West and Central Africa 167.20 158.63 145.95 146.27 163.02 Sub-Saharan Africa 70.99 82.57 77.57 73.82 67.74 72.32 Sub-Saharan Africa 70.99 82.57 77.57 73.82 67.74 72.32 All Africa 70.99 82.57 77.57 73.82 67.74 72.32 Debt service ratio (%)(simple arithmetic mean, based on countries that have relevant data) 62.94 64.07 East and Southern Africa 17.55 14.86 21.26 12.39 15.49 North Africa 36.49 29.86 24.48 21.15 20.10 22.38 North Africa 14.52 14.86 21.12 18.39 15.92 17.49 Sub-Saharan Africa 22.112 18.98 17.62 17.61 14.68 Mit Africa 22.112 19.88 17.69 17.41 18.53 | North Africa | 67.02 | 69.68 | 68.75 | 61.36 | 58.15 | 55.81 |
| Sub-Saharan Africa 70.99 82.57 77.57 73.82 67.74 72.32 All Africa 69.00 76.13 73.16 67.59 62.94 64.07 Debt service ratio (%)(simple arithmetic mean, based on countries that have relevant data) 73.16 67.59 62.94 64.07 Debt service ratio (%)(simple arithmetic mean, based on countries that have relevant data) 73.66 21.15 12.39 15.49 Cast and Southern Africa 17.55 14.86 21.15 20.10 22.38 North Africa 14.52 21.12 18.93 15.92 17.08 West and Central Africa 9.20 14.52 21.12 14.72 14.68 Sub-Saharan Africa 22.84 22.21 19.88 17.41 18.53 | West and Central Africa | 167.20 | 158.63 | 145.95 | 146.27 | 163.02 | |
| All Africa 69.00 76.13 73.16 67.59 62.94 64.07 Debt service ratio (%)(simple arithmetic mean, based on countries that have relevant data) 73.16 67.59 62.94 64.07 Debt service ratio (%)(simple arithmetic mean, based on countries that have relevant data) 75.55 14.86 21.26 15.26 12.39 15.49 North Africa 17.55 14.86 21.12 18.26 17.15 20.10 22.38 North Africa 36.49 29.29.86 24.48 21.15 20.10 23.38 North Africa 14.52 21.12 18.98 18.23 15.92 17.08 Sub-Saharan Africa 22.84 22.21 19.88 17.41 18.53 | Sub-Saharan Africa | 70.99 | 82.57 | 77.57 | 73.82 | 67.74 | 72.32 |
| Debt service ratio (%)(simple arithmetic mean, based on countries that have relevant data) East and Southern Africa 17,55 14,86 21.26 15.26 15.49 East and Southern Africa 17,55 14,86 21.26 15.26 15.49 North Africa 36,49 29,86 24.48 21.15 20.10 22.38 North Africa 14,52 21.12 18.98 15.92 17.08 Sub-Saharan Africa 9.20 14.52 14.22 14.72 14.68 All Africa 22.84 22.21 19.88 17.41 18.53 | All Africa | 69.00 | 76.13 | 73.16 | 67.59 | 62.94 | 64.07 |
| East and Southern Africa 17.55 14.86 21.26 15.26 12.39 15.49 North Africa 36.49 29.86 24.48 21.15 20.10 22.38 North Africa 36.49 29.86 24.48 21.15 20.10 22.38 Sub-stand Central Africa 14.52 21.12 18.39 15.92 17.08 Sub-Saharan Africa 9.20 14.52 14.72 14.76 14.68 All Africa 22.84 22.21 19.88 17.41 18.53 | Debt service ratio (%)(simple arithmetic mea | an, based on | countries that | have relevan | t data) | | |
| North Africa 36.49 29.86 24.48 21.15 20.10 22.38 West and Central Africa 14.52 21.12 18.98 18.39 15.92 17.08 Sub-Saharan Africa 9.20 14.57 15.29 14.72 14.68 All Africa 22.84 22.21 19.88 17.41 18.53 | East and Southern Africa | 17.55 | 14.86 | 21.26 | 15.26 | 12.39 | 15.49 |
| West and Central Africa 14.52 21.12 18.98 18.39 15.92 17.08 Sub-Saharan Africa 9.20 14.57 15.29 14.72 14.68 All Africa 22.84 22.21 19.88 17.41 18.53 | North Africa | 36.49 | 29.86 | 24.48 | 21.15 | 20.10 | 22.38 |
| Sub-Saharan Africa 9.20 14.57 15.29 14.22 14.72 14.68 All Africa 22.84 22.21 19.88 17.41 18.53 | West and Central Africa | 14.52 | 21.12 | 18.98 | 18.39 | 15.92 | 17.08 |
| All Africa 22.84 22.21 19.88 17.69 17.41 18.53 | Sub-Saharan Africa | 9.20 | 14.57 | 15.29 | 14.22 | 14.72 | 14.68 |
| | All Africa | 22.84 | 22.21 | 19.88 | 17.69 | 17.41 | 18.53 |

Source: Based on World Bank Global Development Finance (2000). Net transfer = Loan disbursements less amortization and interest payment [as defined in World Debt Tables] Aggregate net transfer = Aggregate net resource flows (Loan disbursements less amortization) plus official grants (non-technical) and foreign direct investment (FDI) less interest payment and FDI profit [as defined in World Debt Tables].

1.2 BACKGROUND TO AFRICAN DEBT AND MACRO POLICY DEBATES

1.2.1 Background to African Debt

African economic problems can be seen as a complex of interrelated factors of an internal and external nature. As noted in section 1.1, this study focuses on the latter. The external finance problem – and the debt crisis in particular – represents one of the major problems facing African nations today. As can be seen from Table 1.1, the total external debt of Africa has risen nearly 25-fold from a relatively low level of US\$13.9 billion, in 1971, to over \$300 billion in 1998. The major component of this burden comprises outstanding long-term debt. During the late 1970s, and early 1980s, IMF credits were increasingly used, with 'Structural Adjustment' and 'Enhanced Structural Adjustment' facilities comprising an ever-important component of flows to Africa.

Changes in the structure of African debt can also be described in terms of creditor patterns. From Table 1.1, it can be seen that bilateral debt comprises the largest component of Africa's total debt. This is followed by multilateral debt, with private inflows showing a decline. Generally, it may be observed that a larger share of official debt is now disbursed on concessional terms. Finally, it is interesting to note that the debt problem is being aggravated by capitalization of interest and principal arrears, which constitute nearly a quarter of the external debt burden.

Although the share of African debt as a proportion of the total debt of developing countries is low, the relative debt burden borne by African nations remains high. As can be seen from Table 1.2, the debt to GNP and debt service ratios rose from 22 per cent and around 9 per cent, respectively, in 1971, reaching a high of 67 per cent and 23 per cent during the late 1980s. In 1998, the last year for which we have data, these ratios stood at 64 per cent and 18 per cent, respectively.¹ Africa's burden of debt may also be assessed by examining net transfers to the sub-regions. Thus, if we exclude grants and net foreign direct investment inflows from Table 1.2, it can be seen that net transfers since 1990 have, in fact, flowed *from* Africa to the developed nations. Further, it is noteworthy that the level of such transfers has increased, from US\$ 3.86 billion in 1985 to nearly US\$ 12.45 billion in 1998. Finally, it is worth pointing out that in the 1990s nearly 35 per cent of grants to Africa, in fact went to 'technical experts' coming from the North.

In summary, the last three decades have witnessed an unprecedented increase in the level of African debt. This debt is characterized by its predominant long-term character, the growing importance of debt owed to bilateral and multilateral creditors, a trend away from concessionality to nonconcessionality and an increase in the importance of interest and principal arrears (usually capitalized through the Paris and London clubs) as a component of long-term debt. Indicators of the debt burden also indicate that African debt is extremely heavy compared to the capacity of the African economies, and, in particular their export sectors. Moreover, most African nations have been subjected to net financial outflows in the period since the mid-1980s. The performance of these economies, coupled with a mounting debt burden, surely indicates that African countries are incapable of simultaneously servicing their debt and attaining a reasonable level of economic growth, let alone addressing issues of poverty alleviation.

The actual size of indebtedness does not usually represent an economic problem in itself, since this debt may usually be mitigated by rescheduling and similar short-term arrangements. However, the size of accumulated debt, relative to capacity level, and subsequent impacts on the economy, do represent a serious problem for African nations. In this respect, three interrelated implications of the debt problem deserve mention. First, servicing of the external debt erodes foreign exchange reserves, which might otherwise be available for purchase of imports. This has led to the 'import compression problem', in which a shortage of foreign exchange adversely affects levels of public and private sector investments. The import compression problem represents one of the major macroeconomic issues facing Africa today. Second, the accumulation of a debt stock results in a 'debt overhang' problem, which tends to undermine the confidence of private investors, both foreign and domestic. A decline in levels of private investment as a share of GDP, from the late 1970s onwards, may partly be attributed to this factor. Finally, servicing of debt is placing an enormous fiscal pressure on many African nations. Such pressure has had an adverse effect on public investment. This finding is reflected in a reduction in the share of public investment in GDP from late 1970s onwards. Naturally, a reduction in
levels of public investment will tend to have negative consequences for physical and social infrastructure. To sum up, the debt issue is a crucial element of the overall economic crisis facing Africa. How, then, has this crisis come about? The following sections will briefly summarize some of the general arguments surrounding this issue.

1.2.2 Africa's Economic Crisis: What Caused It?

There are three sets of contending explanations for Africa's economic crisis. The first is set out in World Bank (1981) – also known as 'the Berg Report' – and a number of subsequent World Bank publications. An alternative explanation for Africa's economic problems, associated with the United Nations' 'Economic Commission for Africa' (ECA) is outlined in *African Alternative Framework to Structural Adjustment Programmes*, AAF-SAP (ECA 1989a). Finally, there exists a third view, which is less clearly associated with any particular institution and largely held by academics of a Marxist orientation. This latter position is often offered as a critique to the other two explanations. Although the scope of all three sets of explanations is general, encompassing every aspects of the African economic crisis, we focus mainly on how problems in the external sector of the economy are explained. Nevertheless, by referring to this wider debate, we aim to locate the problems and the role of the external sector in a wider context.

The World Bank's *Agenda for Action* (1981) argues that Africa's problems relate to underdeveloped human resources, political fragility, problems of restructuring colonial institutions, inheritance of poorly shaped economies, climate, geography and population growth. Set in the context of these problems, disappointing performance of the external sector is, perhaps, a little more understandable. The Bank argues that, in spite of external shocks, associated particularly with a rise in oil prices in the periods 1973–74 and 1978–80 and a decline in world demand for primary commodities, the balance of payments problems experienced by most African nations since the 1970s cannot generally be attributed to a deterioration in terms of trade.² With the exception of mineral exporters, it is suggested that terms of trade for most African nations have, in fact, either been favourable or neutral.³

The main cause of the balance of payments problem, according to the Bank, has been a decline in the volume of exports. The decline in terms of trade faced by African nations is attributed to three factors. Firstly, structural changes in the composition of world trade, with trade

in commodities growing at a slower rate than that of manufactures, has resulted in a decline in the African share of total world trade. Secondly, drought and civil strife has negatively affected Africa's supply capacity. And thirdly, trade restrictions and agricultural subsidy policies of industrial countries represent a barrier to African trade.⁴ The Bank goes on to argue that the failure of Africa's export sector may be explained in terms of three main factors. Firstly, government policy has tended to be biased against agricultural and export production. Secondly, increased consumption associated with rapid population growth has placed a burden on resources, which might otherwise have been used by the export sector. And, thirdly, inflexibilities in African economies are seen as representing an obstacle to diversification. The Bank's insistence that policy failure represents the main explanation for Africa's economic crisis, and consequent emphasis on the need for reforms, has continued with the publication of its long-term perspective study (World Bank 1989). Moreover, as recently as 1994, the Bank continues to argue that orthodox macroeconomic management represents the road to economic recovery in Africa and, hence, that more adjustment, not less, is required (World Bank 1994). This assertion has been the subject of various criticisms, coming from a host of different angles (see inter alia Adam 1995, Mosley et al. 1995, Lall 1995).5

A number of other analysts have arrived at conclusions, in line with those of the Bank. Van Arkadie (1986), while sympathetic to the problems posed by external shocks, argues that stagnating or falling output has had an important impact on export earnings. On the latter point the World Bank (1989) argues, rather vigorously, that declining export volumes, rather than declining prices, account for Africa's poor export revenue. Grier and Tullock's (1989) analysis supports this view. Based on their survey of empirical studies into the causes of the African economic crisis, Elbadawi et al. (1992), also found domestic policies to be important. White (1996b), citing the case of Zambia, argues that economic decline following Zambia's independence may largely be attributed to economic mismanagement. Using a simple pooled multiple regression equation for 33 African countries, Ghura (1993) also found significant support for the Bank/IMF viewpoint. Easterly and Levine (1996) suggest political instability, low levels of schooling, deterioration in infrastructure, as well as policy failures as representing possible causes of Africa's growth problems. They conclude, however, that policy improvements alone are likely to boost growth substantially (see

also Collier & Gunning 1999 for a similar argument). Although the above survey is not exhaustive, the aforementioned works would tend to lend strong support to the Bank/Fund's viewpoint. The logical conclusion to be drawn from this survey, therefore, is that the remedy to Africa's economic problems is to implement structural adjustment programmes (SAPs).

In contrast, the ECA (1989a) prefers to explain Africa's problems in terms of deficiencies in basic economic and social infrastructure (especially physical capital), research capability, technological know-how and human resource development, compounded by problems of sociopolitical organization. The ECA sees inflation, balance of payments deficit, a rising debt burden and instability of exports as resulting from a lack of structural transformation, unfavourable physical and sociopolitical environment, as well as an excessive outward orientation and dependence. The ECA study suggests that weaknesses in Africa's productive base, the predominant subsistence and exchange nature of the economy and its openness (to international trade and finance) have all conspired to perpetuate the external dependence of the continent. Hence, one of the striking features of the African economy is the dominance of the external sector. This has the effect of rendering African countries guite vulnerable to exogenous shocks.⁶ Consequently, according to the ECA viewpoint, perceiving African problems in terms of internal and external balance problems and seeking a solution within that framework (most notably, through the implementation of structural adjustment programmes) implies not only the wrong diagnosis but also the wrong treatment. The ECA study argues that '...both on theoretical and empirical grounds, the conventional SAPs are inadequate in addressing the real causes of economic, financial and social problems facing African countries that are of a structural nature' (ECA 1989a: 25).

Based on this alternative diagnosis, and the major objectives of the 'Lagos Plan of Action' (OAU 1981), the ECA formulated an African alternative framework to the Bank/Fund's policy recommendations. The ECA framework focuses on three dynamically interrelated aspects, which need to be taken into account. First, *the operative forces* (political, economic, scientific and technological, environmental, cultural and sociological),⁷ second *the available resources* (human and natural resources, domestic saving and external financial resources) and third *the needs to be catered for* (i.e., focusing on vital goods and services as opposed to luxuries and semi-luxuries). The adoption of this general

framework would allow the different categories of operative force to influence not only the level and structure of what is produced but also the distribution of wealth. Moreover, these forces may then influence the nature of needs to be catered for and the degree of their satisfaction. At a concrete level this is envisaged as taking a number of policy directions. Firstly, improving production capacity and productivity, mobilization and efficient use of resources, human resource development, strengthening the scientific and technological base and vertical and horizontal diversification. Secondly, improving the level and distribution of income, adopting a pragmatic balance between the public and private sectors, putting in place 'enabling conditions' for sustainable development (particularly economic incentives and political stability). shifting of (non-productive) resources, and improving income distribution among various groups. And, finally, focusing on the required needs, particularly in relation to food self-sufficiency, reducing import dependence, re-alignment of consumption and production patterns and managing of debt and debt servicing.

Just as many have argued in favour of the Bank/IMF view, so too, many analysts have come out in support of the ECA's position. Thus, various studies have emphasized Africa's extreme dependence on primary commodity exports (see Ngwenya & Bugembe 1987, Fantu 1992, Adedeij 1993). Siting this discussion in a broader historical context. these studies have highlighted the impact of colonialism in establishing the rules by which Africa might participate in the world economy. According to these rules. African nations produced raw materials and agricultural goods for Europe's industries. Further, it is argued that this pattern of trade has changed very little since the time of political 'independence' (Fantu 1992: 497-500, Adedeji 1993: 45). Indeed, Stefanski (1990) argues that, understood in the context of direct continuum with the colonial experience, Africa's economy still depends on external factors to a much greater degree than any other developing region. As a result of this dependence. Africa's economic crisis is seen as being intricately interconnected with external factors such as falling terms of trade, declining demand for African exports and related external shocks (Stefanski 1990: 68-77, Adedeji 1993: 45). Collier (1991) also argues that abrupt external shocks (be they negative or positive) have represented important causes of the poor long-term economic performance of Africa.⁸ Ali (1984) has touched on another dimension of the problem. He argues that, for most African nations, the mitigation of their

problems depends not only on the characteristics of the commodities they export (and specifically their elasticities) but also on the presence or absence of the necessary market staying power. Wheeler (1984) has made an exploratory econometric analysis of the sources of stagnation and suggests that 'environmental' factors (especially terms of trade and international conditions of demand) have had a greater impact on growth than policy variables. Indeed, based on Ghura's (1993) recent econometric analysis, world interest rates represent a further significant variable, which should be added to Wheeler's list of adverse 'environmental' factors.⁹

The negative impact of dependence on exports of primary commodities is reflected in three interdependent phenomena. Firstly, a decline in prices faced by exporters ('terms of trade'). Secondly, instability of export earnings. And, thirdly an absolute decline in levels of demand and supply. Attempts to compensate for a deterioration in the exchange rate facing exporters, by increasing supply, have resulted in a further decline in prices (Fantu 1992: 502, Stefanski 1990, Stein 1977). Stein (1977) examined export trends in East Africa (Uganda, Kenya and Tanzania) in order to determine the causes of the divergence of each country's export growth from that of the world. He found that unfavourable commodity composition,¹⁰ as opposed to the favourable/ unfavourable nature of its market and increased competitiveness went a long way in accounting for this divergence. Because African countries depend on a few commodities, whose prices swing cyclically and may decline over time, these countries face export-earning instability. Naturally, such instability adversely affects their economies. However, Fosu (1991) examining the evidence for sub-Saharan Africa, argues that export instability per se is less important than fluctuation in capital formation (capital instability) in affecting economic growth. Yet, as his own work shows, in sub-Saharan Africa, high export instability may render export proceeds a relatively unreliable source for funding for investment projects (Fosu 1991: 74-75). This usually forces countries to depend on external finance (discussed, at length, in section 1.3).

The third view differs from the other two in its understanding of what crisis means in the African context. For these analysts crisis '...has a connotation of systemic breakdown, but more generally it can refer to a moment or a specific time period in the history of a system at which various developments of a negative character combine to generate a serious threat to its survival' (Lawrence 1986: 2). Sutcliffe (1986),

Introduction

for instance, argues that the African crisis represents the continuation of a complex process of polarization trends. It emanates from Africa's economic dependence. For him, the African crisis is best understood in terms of the combined result of long-term secular effects of imperialism suddenly aggravated by the impact of the world capitalist crisis. Thus, according to these viewpoints. Africa's problems are best understood as resulting from long-term underdevelopment, following dependency theory,¹¹ and short-term vulnerability, following international aspects of crisis theory (Amin 1974a, 1974b; Ake 1981, cited in Ofuatey-Kadjoe 1991; Sutcliffe 1986: 19-20; Harris 1986: 93; Onimode 1988: 13; Moyo et al. 1992: 210). In general, these writers are against the view that there is a 'norm' from which African countries are in a temporary deviation, with associated implications that these countries may return to that norm given a particular adjustment measure (Harris 1986: 84). Harris (1986) and Mamdani (1994) for instance, argue that the IMF and Bank's ultimate objective is not to correct distortions in a free market international system, but to construct such a system (Harris 1986: 88). In so doing, these institutions may undermine any attempt to create an independent, integrated and self-sustained (African) economy (Mamdani 1994: 129).

While there are areas where the first two approaches both converge and diverge, the third explanation for Africa's economic crisis stands firmly in opposition to both. Thus, the core of the disagreement between the bank and ECA views centres on 'the role of the market' mechanism¹² (Oskawe, quoted in Asante 1991: 179). While the Bank believes in the market mechanism as representing the fundamental instrument of resource allocation and income distribution, the ECA questions this viewpoint. Thus, while the bank focuses mainly on financial balances, the ECA considers a much broader transformation as an enabling condition for the former. While the Bank emphasizes the export sector, the ECA strategy advocates selectivity (see also Asante 1991: 180). While the Bank expresses concern about anti-export bias and population policy, the ECA prefers to emphasize the need to ensure total structural transformation and food self-sufficiency. While the Bank places more emphasis on short-term policies than on Africa's long-term needs, the ECA strategy, as defined in the Lagos plan of action, stresses the importance of also addressing issues of long-term transformation, alongside these short-term policies.¹³ However, these institutions do agree on some major issues, such as the need for human resource development, improving the efficiency of parastatals, and sound debt management. The ECA analysis is quite comprehensive in addressing the causes of the crisis and in suggesting not only short-run solutions but also a framework for long-term transformations.¹⁴ Thus, the analysis of the external sector of Africa, adopted in this study, will be conducted within this broader context. Within this perspective, it is not difficult to show that the African debt crisis has developed as part of the broader external economic problem of the continent.

The literature on the origins of the African debt crisis lists a number of factors as its cause. The oil price shocks of 1973–74 and 1978–79, the expansion of the Eurodollar, a rise in public expenditure by African governments following increases in commodity prices during the early 1970s, recession in the industrialized nations and subsequent fall in commodity prices, as well as rises in real world interest rate are all mentioned as major factors. Surprisingly, almost all of this literature focuses on the post-independence period, with a greater part of the analysis contained therein relating to the 1970s, 1980s and 1990s. The main argument set out in this book is that we need to extend this analysis to the pre-independence period if we are adequately to explain the current debt crisis, as well as propose possible solutions for its resolution. From this point of departure, the following section traces the historical formation of an African economic structure incapable of handling the current debt crisis.

1.3 THE HISTORICAL ORIGIN OF AFRICA'S ECONOMIC LINKAGE WITH THE NORTH

Following Amin (1972), African economic history may be classified into: (i) the 'pre-mercantilist period' (from pre-history to the beginning of the 17th century); (ii) the 'mercantilist period'¹⁵ (from the 17th century to 1800), characterized by the operation of the slave-trade; (iii) the 'third period' (from 1800 to 1880) characterized by attempts to set up a European dependent African economy; and finally, (iv) the 'period of colonization' in which the dependent African economy became fully established (Amin 1972: 106). This section will not pretend to discuss the details of Amin's periodization. Rather, after briefly reviewing the economic history of the other periods, it will focus mainly on the colonial period, during which time the economic structure African countries inherited at the time of independence became established.

1.3.1 Pre-colonial Trade in Africa

African interactions with the rest of the world, and especially Europe, date back many centuries, before culminating in full-fledged colonization in the latter part of the 19th century.¹⁶ During the first part of this period, Africa had autonomy in its linkages with the rest of the world¹⁷ (Amin 1972: 107–10). However, during the 16th century, African trade centres moved from the savannah hinterland to the coast, in reaction to changes in European trade, which shifted increasingly from the Mediterranean to the Atlantic (Hopkins 1973: 87).

Various studies have documented how pre-colonial Africa was characterized by production of diversified agricultural products (see, for instance, Rodney 1972: 257). The internal trade of the continent was distinguished by regional complementarities, with a broad natural resource base. Thus, a dense and integrated network was set in place, dominated by African traders, which included, inter alia, trade among herdsmen and crop farmers, supply of exports and distribution of imports. This was dominated by trade in salt, West African 'spices', perfumes, resins and kola nuts, of which the latter was the most important (Amin 1972: 117, Hopkins 1973: 51-86; Neumark 1977: 128-30, Vansina 1977: 237–48, Austen 1987: 36). Brooks' account of the economic conditions prevailing in this period provides an impressive insight into African trade at the time (Brooks 1993). Specifically, one is struck by: (a) the extent of local and long distance trade; (b) the range of goods traded; and, (c) the degree of processing of commodities (for instance in textile manufacturing, dyeing and metal working), particularly in West Africa. According to his account, the major commodities traded among West Africans in pre-colonial times include salt, iron, gold, kola, and malaguetta pepper and cotton textile. Of these, kola and malaguetta pepper were important, not only in West Africa, but also in the trans-Saharan trade. Indeed, this trade was so extensive that Europeans were able to obtain malaguetta pepper at inflated prices from Maghreb¹⁸ middlemen from at least the 14th century onwards (Brooks 1993: 51-121). Moreover, in this period, Europeans were able to purchase cloth from Morocco, Mauritania, Senegambia, Ivory Coast, Benin, Yorubaland and Loango for resale elsewhere (Rodney 1972: 113, Hopkins 1973: 48). (It is curious to note that, in a geographic and economic sense, North Africa was connected, rather than separated, by the Sahara to other parts of Africa.)¹⁹ It is also worth noting that the quality of many of these processed goods was quite comparable with products

originating in other parts of the world. For instance the level of manufacturing of textiles in pre-colonial West Africa was so sophisticated that these textiles were not only traded in West, North and Central Africa but also in the European market (see Hopkins 1973: 48 for details). Moreover, none of the goods brought by Europeans supplied any of the basic or unfulfilled needs of African societies. Indeed, similar commodities and/or substitutes were obtainable through West African commercial networks. Specifically, African artisans of the time manufactured high-quality iron, cotton, textiles, beers, wines and liquors (Brooks 1993: 56). Austin argues that this trade, sometimes referred to as the 'Sudanic economy', represents 'an ideal African development pattern: continuous and pervasive regional growth with a minimum of dependence upon foreign partners for provision of critical goods and services' (Austen 1987: 48). However, this autonomy in traditional industries was to be undermined by subsequent events (Koncazcki 1990: 24).

The early development pattern of Africa varies between regions. In contrast to West Africa. East and Southern Africa (ESA) were characterized by a well-established economic interaction with the Arabian and Asian countries, long before the arrival of the Europeans. More specifically, this part of Africa supplied a range of products, such as gold, copper, grain, millet and coconut to the Middle East and Indian Ocean economies. There also existed a dynamic caravan trade and commercial plantations long before the onset of European colonial rule. According to Austen, the towns in this part of Africa degenerated into little more than entrepôts for raw material exports and manufactured imports, rendering them dependent on the external economy (Austen 1987: 67–74). However, as documented by Kjekshus, during the mid-19th century, prior to the onset of the colonial period, the interior of what is now mainland Tanzania carried an estimated 4.5 million head of cattle. Indeed, the entire coastal region also supported a rich agricultural and pastoral economy (quoted in Levs 1996; 111). Further, Nzula et al. (1979)²⁰ argued that the region was characterized by peasant production, which was mainly a natural and closed economy, with a substantial number of people leading a nomadic existence (Nzula et al. 1979: 38). The existence of an independent and autonomous economy, dating back to antiquity, is also well documented in Ethiopian history.²¹ Amin also notes that the African societies of the pre-colonial period developed autonomously (Amin 1972: 107-8). Thus, one may reasonably

conclude that, although its economy was not as complex as that of West Africa, nevertheless, that the ESA region had some degree of autonomy in its economic activity, and, hence, was not as dependent on the export of commodities, particularly to Europe.

To sum up, there would appear to be a long history of integrated and autonomous economic activity in most regions of Africa with local and long-distance trade playing a linking role. This is not an attempt to paint a 'golden past' for Africa. Rather, it is meant to underline the fact that Africa had a healthy and fairly independent economic system, before colonialism intervened to force a structural interaction with Europe.

1.3.2 The Formation of a Commodity-exporting and External Finance-constrained Economy

The period leading up to the industrial revolution, and the 16th and 17th centuries, in particular, witnessed the beginning of the shaping of the African economy by European demand. A clear example is the pressing demand for gold coin in Europe, and the subsequent search for gold in West and Central Africa (WCA).²² Indeed, demand for labour, required in the American gold search, was instrumental in the formation of the European slave trade (Rodney 1972: 86–87). Thus, the shaping of the African economy by Europe began, even before the onset of the formal colonial period.

With the onset of the industrial revolution in Europe, Africa lost its remaining autonomy and was reduced to being a supplier of slave labour for the plantations of America (Amin 1972: 107-10). The European slave trade, and the so-called 'triangular trade', both of which are beyond the scope of this book, are widely discussed issues in the economic history of Africa. Any resistance to the slave trade was silenced, not only by the co-opting of local chiefs, but also by sheer force. Such use of force has been documented in what is now Angola, Guinea and various other parts of the continent (Rodney 1972: 90-91, see also Bernstein et al. 1992 for a brief summary of the triangular trade). Moreover, this era witnessed a widespread expansion of European control. This expansion was undertaken with the dual aims of: (a) incorporating new areas under primary crop production, using African land and labour (which were priced below world market prices); and, (b) increasing the level of production of existing primary commodities. On the import side, cheaper and purer iron bars, and implements such as knives

and hoes were made available, displacing some of the previous economic activities undertaken by local blacksmiths. This had knock-on effects in terms of a reduction in levels of iron smelting and even a decline in the mining of iron-ore (Wallerstein 1976: 34–36, Baran 1957: 141-43).²³

Within the ESA region, cloves grown in Zanzibar and Pemb islands, for export to the Asian and European markets, were the first cash crops successfully produced prior to European colonialism. Mainland estates, dominated initially by Arab and Asian traders, were involved in externally-oriented production through sales of copra, sesame seed and oilvielding materials, for which France was the principal market (Munro 1976: 55). Following colonization, peasant cash-cropping developed in East Africa. However, unlike the WCA region, this was mainly as a consequence of a combination of political injunction and regulation. Such imposition from above was usually resisted, the Maji-Maji uprising, in today's Tanzania, being a case in point. In other instances cash cropping simply failed to take hold, as in the case of a cotton scheme proposed for Nyanza province, Kenya (Munro 1976: 116). However, in spite of these initial setbacks, eventually the colonial powers were successful in implementing their policy of introducing cash cropping to the region.

As described above, there existed a reasonable degree of trade linkage with Europe in the pre-colonial period. Leaving aside the slave trade, the main feature of this trade was the export of primary commodities by African nations to Europe. Thus, even before the onset of the colonial era, the seeds of Africa's subsequent role (as a supplier of raw materials and foodstuffs for Europe, and a market for European manufactures) as well as its dependence on external finance had already been sown.²⁴ Or, to take a slightly different perspective, a move from the production of primary products to processing of these products (by Africans and in Africa) was interrupted. This represents the first pre-designed attempt to articulate African economic activity to the requirements of the outside world. This development was vigorously followed up upon during the colonial period as a consequence of: (i) the so-called imperial self-sufficiency in raw materials scheme; (ii) the impact of the First and Second World Wars; and, (iii) financing requirements for the creation of public utilities designed to serve (i) and (ii).

The imperial self-sufficiency scheme

As noted above, the export structure associated with colonialism did not arise by accident. Rather, it was preceded by various experiments to produce agricultural products demanded by the developing European industries. A French experiment to produce crops similar to those produced in America, the establishments of plantations in Senegal, during the 1820s, British experiments with 'model farms' in Niger, during the 1840s and cotton experiments²⁵ in Senegal, Nigeria and the Gold Coast (Ghana) all represent cases in point (Hopkins 1973: 137). In Germany, Bismarck, initially reluctant to create a colonial empire, was persuaded by German commercial interests that overseas territories could provide raw materials for German industries, as well as markets for their products (Longmire 1990: 202). This growing demand for raw materials, the search for a market for finished products from Europe, inter-European competition, and a number of other factors conspired to form the basis upon which colonialism was to evolve.²⁶

During the colonial period, one of the main phenomena, which strengthened primary commodity exports from European colonies in Africa, was the so-called 'imperial self-sufficiency' scheme. Thus, British, French and Belgian textile industries sought to obtain cotton from Africa, and invested accordingly. A similar scheme was also developed for tobacco. This was administered both by colonial governments and by some European-based companies (Munro 1976: 128-37) and resulted in an expansion in colonial trade. With the onset of colonialism, the centre of African trade shifted from the hinterland to the coast, and the composition of this trade also changed in response to the demands of the increasing external orientation of the economy (Amin 1972: 117). For example, expansion in the production of palm products and groundnuts in Africa was directly linked with increased demand for inputs required in soap and candle factories, lubricants (particularly for the railways) and European economic growth in general (Hopkins 1973: 129).

At the same time, the processing of such primary products in Africa, except in white settler colonies was actively discouraged. Indeed, this was the case even when factories were owned by Europeans. For example, in Senegal, the proportion of groundnuts, which could be processed, prior to their export to France, was strictly controlled (Fieldhouse 1986: 48, Fyfe quoted in Wallerstein 1976: 36, Onimode 1988: 177). In Angola the Portuguese prevented the operation of flour mills,

with the country exporting wheat to Portugal and importing wheat flour back (Koncazcki 1977: 81). According to Austen, the fact that colonial governments, (with the possible exception of the Union of South Africa), saw themselves primarily as representatives of the 'mother' (colonial) country, which was benefiting from the existing pattern of trade, explains why they pursued policies which were directly and indirectly designed to block efforts at local industrialization (Austen 1987: 133).

In order to achieve these dual objectives, of inducing the colonies to be suppliers of inputs, and markets for manufactured goods, various methods of coercion were employed. Africans were forced, by superior firepower, to abandon small-scale manufacturing industries and trade with rival European nations (Dickson 1977: 142). At the same time, large European firms were encouraged to concentrate on growing and trading in agricultural products. This was easily achieved for a variety of reasons. Specifically, African peasants moved into cash cropping: (a) to ensure access to European goods, to which they had become accustomed, in a limited way, in the pre-colonial era; (b) to earn cash. which was required to pay various taxes; and, finally: (c) as a result of force.²⁷ In other cases, Africans were simply exterminated to pave the way for settlers.²⁸ In other parts of Africa Europeans directly controlled the production of commodities such as cotton, sugarcane and tobacco. (Amin 1972: 112–13). Indeed, in areas such as British East Africa the law required that farmers grow a minimum acreage of cash crops. However, these peasants were not wholly dependent on cash crop production. Rather, they also produced food for own consumption, this being in the interests of the big firms, since it enabled them to pay only minimal wages, which did not have to cover maintenance of the labourer and his family (Rodney 1972: 172). Nevertheless, the colonial authorities ensured that the extent of such food production was not large enough to ensure self-sufficiency. For instance, in British Guinea it was a criminal offence to grow rice (at a time when it was imported from India and Burma) because it was feared that rice growing would lead to the diversion of labour from the sugar plantations (Frankel 1977: 236). Thus, in this manner, Africa's economic role, basically as a producer of primary commodities, continued to be shaped to serve Europe's industrial and commercial interests.

The First and Second World Wars (W.W.I and II)

The impact of the First World War on African colonies was devastating. Although trade was disrupted during the period, nevertheless African colonies were forced to supply commodities to finance the war. The end of the war was followed by a surge in major commodity prices and hence high export earnings for the African colonies (Munro 1976: 119-23). Similarly, The Second World War also resulted in an increased demand for primary commodities, and especially those with military strategic importance such as vegetable oils, metals and industrial diamonds (ibid.: 170, Burdette 1990: 84.). This had the effect of reinforcing the commodity producing and exporting role of the European colonies in Africa. In addition to the direct effects of the war, the post-war reconstruction of Europe, rising levels of European incomes and removal of restrictions on consumer demand and commodity stockpiling, engendered by the outbreak of the Korean war in 1950, resulted in the price of African exports surging to unprecedented heights (Munro 1976: 177). Thus, when war erupted or was expected to erupt in the colonizing countries, commodity production and exports by African colonies was boosted by non-price mechanisms. Further, the end of the war was usually also followed by a commodity price boom and associated increase in the level of the commodity exported, this time through the operation of the price mechanism. In the process, the specialization of European colonies in Africa as producers and exporters of primary commodities became firmly established.

Financing public utilities and commodity exports

In general, in the pre-1929 international financial order, which was dominated by government bonds (i.e., portfolio investment), Asian and African colonies had little choice in relation to the nature of their involvement in international financial systems. Political considerations were at the heart of regulating access to capital markets (Bacha & Alejandro 1982: 2–3). Besides, such inflows to Africa were generally negligible (UN 1949: 26–28). Capital inflows from W.W.II onwards increasingly came in the form of foreign direct investment (FDI). There was a moderate flow of such capital from the United States and Britain to Africa. However, such investment that did come (especially that originating in the United States, which was the largest supplier) was concentrated mainly in South Africa, Egypt and Liberia, the latter relating to the introduction of a shipping line by the United States (UN 1954: 15–16). In almost all cases, the investment went into plantations and mineral extraction (UN 1949: 32–33).

The colonial period also witnessed a flow of loans and grants from European centres to the African colonies. In almost all cases, these funds were spent on public infrastructure development such as railways and roads to link ports to export production sites, and, to a lesser extent, on schools and health facilities. This was undertaken with the aim of developing the primary commodity exporting capacity of the colonies (see UN 1954: 32-33). In some circumstances the colonial powers were also motivated by military-strategic considerations. It is estimated that, from the mid-1940s to 1960, only 15-20 per cent of such inflows were allocated for social and production sectors, while the rest went into infrastructural development (Munro 1976: 183). The nature of these financial flows to the colonies also differs before and after W.W.II. In general, it can be said that the pre-W.W.II flows came mainly in loan form, while the post-W.W.II flows, and especially those from France, increasingly incorporated a grant element (see also Austen 1987: 197-202 for details). However, the repayment of this debt by colonial administrators created serious difficulties.

These financial difficulties were exacerbated by instability in the world commodity market and the vulnerability of the African colonies to this. Indeed, various analogies may be drawn between the current debt crisis and the situation in this period. For instance, after the great depression (1929–32), African exports declined by about 42 per cent. The depression also resulted in contraction of credit flowing to the colonies. These events led to a serious incapacity to service debt owed to the 'mother' (colonizing) country. Since colonies were not in a position to default on these debts, there was effectively no way out for them. This had repercussions for every African economy, with widespread bank failures, retrenchment programmes in colonial administrations and liquidation of businesses (see Munro 1976: 150-53 for details).

Setting in place a vicious cycle, the financial difficulties being experienced by colonial governments forced the colonies to vigorously follow a policy of producing export commodities, at the expense of other alternatives (Munro 1976: 155, Austen 1987: 127). Peasant cropping, with its attractive minimum cost for colonial governors, was chosen as a convenient vehicle to address this problem. This, the so-called 'peasant path' to financial solvency, became a universal phenomenon

Introduction

throughout the colonies, and especially in the present-day WCA. It was attained by forced involvement of ordinary peasants in the primary commodity export sector. Indeed, this coercion was sometimes so harsh that the ordinary peasants were paid not in cash, but in bills of credit to the administration's head tax (Munro 1976: 156). In the British colonies of East Africa a similar emphasis to the 'peasant path' was also followed (ibid.: 156–57).

In summary, through the process discussed above, the foundations for the existing economic structure of African countries were laid during the colonial period. This was achieved through two channels. Firstly, by directly contributing to the expansion of an enclave of primary commodity-exporting economies. And, secondly, by bringing about a situation of indebtedness, it further accentuated the importance of these activities as sources of foreign exchange required for settling of this debt. Although this general pattern was applied throughout the African colonies, some variations existed across the regions. The following section addresses this issue.

1.3.3 The Three Macro Regions of Colonial Africa: The Amin-Nzula Category

Although colonialism shaped the production structure in a similar way across Africa, nevertheless one may observe certain variations in this general pattern between different macro regions. Leaving aside North Africa, Nzula et al. (1979)²⁹ and Amin (1972) divide the continent into three distinct regions, based on their colonial structure. Firstly, *Africa of the labour reserves* (Nzula et al. 1979 label this 'East and Southern Africa'); secondly, *Africa of the colonial economy* (Nzula et al. 1979 label this the region 'British and French West Africa'); and, thirdly, *Africa of the concession-owning companies* (Nzula et al. 1979 label this 'Belgian Congo and French Equatorial Africa'). The fundamental distinction between these regions is derived from the manner in which the colonial powers settled the 'land question' (Nzula et al. 1979: 36).

In West Africa, commodity production did not take a plantation form. Besides, until quite recently the mineral wealth of the region remained largely untapped (Amin 1972: 115). The amount of African peasant land expropriated was also negligible (Nzula et al. 1979). However, in spite of this, the control and growth of the commodity sector was governed by European interests, while land remained in the hands of small peasants. The mechanisms for this control were as much po-

litical as economic (Amin 1972: 115). Hopkins lists a number of reasons why plantation-based production never became fully established in West Africa. Firstly, some traders were opposed to plantations for fear that they might compete with the export sector for scarce capital. (Such objections were voiced, for example, by businessmen such as Lever and Verdier.) Secondly, a few plantations, which were established, failed because of lack of capital and ignorance about tropical conditions. The third, and perhaps most important, reason why plantations failed to became fully established in West Africa was that small African peasants had already succeeded in forming an export economy by their own efforts. Moreover, establishing plantations would have created conflicts with traditional land rights. Indeed, some crops, such as groundnuts, would not have been suited to plantation agriculture (Hopkins 1973: 213–14). Finally, it is worth pointing out that it was not necessary to develop formal plantation agriculture, since it was possible to influence the nature of production and control the export supply of peasants through monopolistic trading practices, customs restrictions, fiscal controls and appropriate credit arrangements (Nzula et al. 1979: 38).30

In much of today's Central Africa, and part of Southern Africa, concessionaire companies, usually supported by their European state, dominated the entire economic structure through their involvement in mining, fishing, public works and communication, and even taxation (see Seleti 1990: 40). In these regions, the indigenous population were reduced to semi-slavery, and exploited by open and non-economic forms of coercion on the plantations and mines (Nzula et al. 1979: 37, Austen 1987: 140-42). The establishment of such concessionaire companies was further facilitated by the indigenous population fleeing and seeking refuge in the more inaccessible parts of the region. Discouraged by this population exodus, the colonial authorities encouraged adventurer companies to 'try to get something out of the region' (Amin 1972: 117). The activities of these companies were organized in line with demand in the 'mother country'. One example of this was the demand for raw materials required in the European war effort. Thus, the mining companies, in co-operation with colonial officials, designed and determined the nature of their enclave activity to meet the increased demand for copper and other base metals required by the European war industries (Burdette 1990: 84).

Introduction

In Southern and Eastern Africa both systems referred to above were intricately interwoven with a number of specific features (Nzula et al. 1979: 36). In this region the extraction of mineral and settler agriculture was accompanied by the creation, often by force, of a small, and often insufficient, reserve of labour comprising land-owning peasants and the urban unemployed. This was undertaken with the labour demands of mineral extraction and settler agriculture firmly in mind (Amin 1972: 114. Nzula et al. 1979: 37). This labour was further supplemented by interregional migration. Other economic instruments, such as taxation, were also used to create reserve labour for European plantations and mining (Seleti 1990: 34, Koncazcki 1977: 82). The reduction of the cost of labour in such regions to mere subsistence levels rendered the exports of the colonies competitive, in comparison to similar goods produced in Europe. Clearly, the formulation of such a structure was 'as much political as economic'³¹ (Amin 1972: 115, Seleti 1990: 47). However, since the focus of this book is on the economic, we do not go further into such political considerations here. Rather, we would simply observe that, during this period, an economic structure was set in place. characterized by the export of primary commodities.

By the end of the colonial period, what had been achieved in all these macro regions was the creation of a commodity-exporting economy and virtual monopoly of African trade (both import and export) by Europe (see Hopkins 1973: 174). The commodity export-led strategy was vigorously followed during this period. As a result, not only did production for overseas markets expand at a high rate, but also several new items (especially foodstuffs) began to appear on the import list (Hopkins 1973: 178). In some cases, European business interests were so pervasive that they created a protected market, on which to dump their manufactured goods.³² Summarizing the stylized facts in the colonial period, Koncazcki described the economic pattern of what is called 'matured' colonialism³³ as having three distinct components. Firstly, both imports (which were mainly manufactured goods), and exports (mainly raw materials), were fixed with the 'mother' country. Secondly, capital investment in the colony was determined by the trading interest of the 'mother' country, and concentrated in exporting enclaves. Finally, a supply of cheap labour was ensured through a variety of mechanisms (legal, monopolistic employment and through other economic instruments; Koncazcki 1977: 75-76). Indeed, it is worth noting that this pattern has not changed fundamentally, even today. Another important characteristic of this period relates to technological change. For example, if one focuses on cotton production, during the colonial era, Africa '...was concentrating almost entirely on export of raw cotton and the import of manufactured cotton cloth. This remarkable reversal [compared to the pre-colonial period] is tied to technological advance in Europe and to stagnation of technology in Africa owing to the very trade with Europe' (Rodney 1972: 113). Colonialism further exacerbated this situation. Thus, as Amin notes, when we speak of the exchange of agricultural products against imported manufacture (i.e., the terms of trade), 'the concept is much richer: it describes analytically the exchange of agricultural commodities provided by a peripheral society shaped in this [colonial] way against the product of a central capitalist industry (imported or produced on the spot by European enterprises)' (Amin 1972: 115).

To sum up, it has been shown that African nations were in possession of an integrated and autonomous economic structure prior to their intensive interactions with Europeans during the colonial period. It is hard to speculate what the future of such a structure might have been. in the absence of colonialism. However, it goes without saving that it would not have been what it is now, since clearly the present is the result of specific historical process. More specifically, historical interaction with today's developed countries has shaped the structure of the economic activity of African nations, particularly in the areas of international trade and finance. Indeed, economic domination, accompanied by colonization, has further cemented this structure. Thus, given such historical process it is not surprising to find that almost all African nations had become exporters of a limited range of primary products, and importers of manufactured goods, by the time of independence, in the 1960s.³⁴ This was further accompanied by a demand for external finance, when export earnings were not sufficient to finance the level of public expenditures required for maintaining and expanding the commodity-exporting economy. This structure has not changed in any meaningful way in the post-colonial era. Thus, when one examines the financial problems of Africa (which I am arguing relate to its role as a primary commodity exporter) one is, compelled to conclude that these problems are a direct outcome of historical process.

1.4 SOME NOTES ON METHODOLOGY

Methodological discussions in economics are usually problematic. Mainstream economists, usually follow the Popperian approach (the theory-hypothesis-critical test/evidence-falsification or corroboration chain) (see Blaug 1992). However, it could be argued that this approach is more relevant for physics than it is for economics. Thus, some of the problems a researcher in developing countries might face while following the Popperian approach include the possibility of excluding rival explanations ex-hypothesis as well as the difficulty of obtaining 'evidence' or 'facts'. Moreover, as noted by Feyerabend (1975), a method, which adheres to a binding principle, stands in contradiction to the history of research/science. Indeed, openness in research, and Feyerabend's principle of 'anything goes', may be defended under all circumstance. However, as Dutt (1990) notes, the problem, which Feyerabend does not address, is how a researcher may become versed in all the relevant theories pertaining to a particular problem. Dutt's answer to this question is, 'by specializing in areas or problems' (Dutt 1990: 6). And, one might add, by explicit recognition of the fact that the researcher is dealing with an aspect of a problem which is presumed to fit the overall structure, not as a jigsaw puzzle, but as an integral part of it. This implies an obvious trade-off between depth (in the sense of deeply focusing on the particular) and breadth (which entails focusing on the overall picture).

The approach adopted in this study departs from the Popperian one in favour of a realist approach. It is our view that the adoption of such an approach represents a much more fruitful avenue of research in developing countries. This methodological framework is informed by the works of Lipton (1991), Mukhrjee and Wuyts (1991), Wuyts (1992a, 1992b) and Lawson (1989). The overall framework adopted in this study is Lipton's 'inference to the best explanation' (contrastive inference), which looks for residual differences in similar histories of facts and foils as a fruitful method for determining a likely cause (Lipton 1991: 78). This approach entails testing competing hypotheses in the process of research.

At a practical level, this general approach is narrowed down to a more refined one proposed by Mukhrjee and Wuyts (1991) in which a *working hypothesis*, is confronted with the evidence and various rival explanations. Wuyts (1992b) argues 'the best way to test an idea (wrapped up as a hypothesis) is not merely to confront it with its own evidence, but to compare it with rival explanations. It then becomes easier to detect which explanation has more loose ends or will need to resort to *ad hoc* justifications to cope with criticism' (Wuyts 1992b: 4). Once a working hypothesis has been arrived at, the dialogue between data and alternative explanations may best be handled by exploratory data analysis, which comprises graphical display, techniques of diagnostic analysis and transformation of data (Mukherjee & Wuyts 1991: 1). This does not imply that theory has no role to play. Rather, that theory is important 'as a guide to pose interesting questions which we shall explore with data' (Wuyts 1992a: 2).

The generation of working hypotheses, and the subsequent examination of these, may be pursued along Kaldorian lines (Lawson 1989, Lawson et al. 1996). In this realist approach to economic analysis, the researcher is free to start from 'stylized' facts – broader tendencies ignoring individual details – and to construct a working hypothesis, which fits with these facts. The final stage of the analysis entails subjecting the entities postulated at the modelling or explanatory stage to further scrutiny (Lawson 1989).

Building on this methodological background, the study is divided into three main sections. In the first of these, a theoretical literature study, in line with the research problem, will be undertaken. This will help to shape alternative theoretical explanations, in order that the questions and problems posed might be more clearly defined. In Part II (comprising Chapters 4 and 5) the dialogue between the data and alternative explanations will be explored. At this stage in the analysis, the researcher is faced with the practical problem of being open to all conceivable explanations for a particular phenomenon. Thus, economists might differ on their view on a particular economic phenomenon (Dutt 1990: 6). Based on this line of thinking, the underlying view adopted in this study is that the African economy has its own peculiarity. Moreover, it is suggested that different institutions and agents, both in the North and South have different behavioural rules by which they operate. This, in turn, affects the functioning of the economy. Since such structures are explicitly incorporated into this study, the book may be seen as lying within the structural macroeconomics school (see, for example, Taylor 1983 & 1991, FitzGerald & Vos 1989, FitzGerald 1993). Thus, the amalgamation of the view of an economy with the 'inference to best explanation' leads one to work under a specific paradigm à la

Khun. The wider context of 'inference to best explanation' is not lost, however, because, as research progresses, the view about the economy, and judgement about theories, follows a dynamic process of learning. Once the empirical exploration is conducted within the framework of this background, the final step is to depict the stylized facts, which emerge from the dialogue between data and theory, using modelling techniques. This is undertaken in Part III, which comprises the last two chapters of this study.

1.5 ORGANIZATION OF THE BOOK

As set out above, this book is organized in three main sections. Part I, comprising the current chapter and Chapter 2, explores some background issues and reviews a number of relevant theories. Part II, which comprises Chapters 4 and 5, presents an empirical analysis of the major issues and theories explored in Part I. In Part III (Chapters 6 and 7) these empirical issues are brought together within a North-South modelling framework. Chapter 7, in particular, is devoted to an analysis of external shocks and policies, using the model developed in Chapter 6. Finally, Chapter 8 highlights some of the major policy implications arising from this study.

In this introductory chapter, I have argued that Africa's external finance problem is the result of the structure of its trade in the context of the world economy, in general, and its place as a commodity producer in particular. I have also argued that this is compounded by the adverse effects of macroeconomic policies adopted by the Northern (industrialized) nations. The major purpose of this background discussion is to underline the existence of an economic structure, primarily built in the colonial era, which continues to shape the external economic conditions of African nations today. This discussion sets the ground for subsequent chapters, which will examine the questions posed from both a theoretical and empirical perspective.

Chapter 2 is devoted to an in-depth analysis of the key theoretical explanations for the external finance problems of Africa. This analysis is carried across financial instruments which comprise the external finance to Africa (FDI and official and private flows). The theoretical discussion presented in this chapter is used to specify the model developed in the book. The chapter will also attempt to identify theories, which may be used to explain the linkage between Africa's trade and finance problems. Cognizant of the importance of the primary commodity trade in Africa, the chapter attempts to identify theories to explain the functioning of the commodity market, in general, and terms of trade, in particular. The chapter also reviews the literature relating to commodity modelling strategies, with the aim that this might inform the commodity modelling exercise undertaken in this book.

Chapter 3 focuses on theoretical issues relating to African macroeconomics. In this chapter, a cursory look at existing African macro models is undertaken. Major theoretical issues in African macroeconomics, such as import compression, 'the fiscal response' and 'Dutch disease' effects are discussed. The chapter also describes the accounting framework, which is used to organize the macro database of the 21 sample countries, used in the study.

Chapter 4, which begins the empirical analysis section of the book, focuses on the determinants of foreign exchange supply to Africa. The chapter addresses two broad themes. Firstly, capital flows to Africa are examined. The chapter then moves on to examine other sources of foreign exchange, and, in particular, earnings from commodity exports. An attempt to understand the determinants of African export supply is conducted by locating these exports in the context of the world commodity market.

Chapter 5 examines some macroeconomic ramifications of the foreign exchange earnings discussed in Chapter 4. This is undertaken by extending the discussion presented in Chapter 2 to take account of a number of theoretical and empirical issues. Again, the chapter covers two broad themes. Firstly, the 'fiscal response' to external finance is examined, and an alternative framework for analysing the relevant issues is set out. Secondly, the 'Dutch disease' effect of such flows is described. The results obtained in this analysis feed into the global model developed in Chapter 6.

Chapter 6 represents a synthesis of the preceding chapters. Specifically, the theoretical discussion and empirical analysis set out in these chapters is formalized in a North-South economic model. The chapter begins by examining the analytical framework developed by Dutt (1990) with the aim of developing a taxonomy of the major global models currently in use. By critically examining existing models, the framework lays the foundation for the model developed in this book. The main focus of the model, which is based, among others, on Taylor (1981) Darity and FitzGerald (1982) and Vos (1994), is an elaborated modelling of Africa using a Kalecki-Lewis set-up. Most of the model's parameters are econometrically estimated for the three African subregions, and for the North. In Chapter 7, this model is applied in the analysis of various policy issues and external shocks. Finally, Chapter 8 summarizes the major implications arising from the policy and external shock simulation undertaken in Chapter 7.

2 Trade, Finance and Development in a Global Context

2.1 INTRODUCTION

In Chapter 1, the external finance problems of Africa were discussed. This chapter will focus on providing a theoretical perspective to this analysis. More specifically, the chapter will attempt to identify key analytical elements, which may be used to explain Africa's external finance problems. With this objective in mind, the theoretical determinants of capital flows to Africa, and the 'South', in general, will be identified. The major systemic explanations found to be relevant in explaining Africa's external finance problems will form the basis of attempts to develop a formal model. The development of such a model will be set out in subsequent chapters. The general approach chosen will be to examine the issue of Africa's external finance problems across different financial instruments, each of which is assumed to have its own specific features. With these broad objectives in mind, the remainder of this chapter is organized as follows. In section 2.2, international finance theories, which may have relevance for the modelling of Africa, within a North-South framework will be discussed. Section 2.2.1 will examine the determinants of foreign direct investment (FDI) with a view to arriving at an explanation for such flows, which might have relevance for Africa. Section 2.2.2 focuses on theories of bank lending and their relevance for Africa. Section 2.2.3 examines the theoretical underpinnings of official capital flows, which are becoming increasingly important in Africa. In section 2.3, relevant trade theories and models are explored. Finally, section 2.4 highlights some major strands of the literature, which underpin the modelling work set out in subsequent chapters.

2.2 FINANCE THEORIES, NORTH-SOUTH MODELLING AND THE AFRICAN CONTEXT

2.2.1 Foreign Direct Investment

Foreign direct investment does not comprise a major component of external finance flows to low-income countries of Africa. Nevertheless. this section will briefly examine various theories on the determinants of foreign direct investment, in the hope that this may help to explain why this type of investment has not been important in that continent. The early neoclassical approach, summarized in an article by MacDougall (1960), hypothesized that capital flows across countries are governed by differential rates of return (within the neoclassical market setting).¹ It is argued that such capital inflows are welfare-enhancing² to both parties engaged in the capital movement. The MacDougall model assumes perfect competition, risk free capital movement, mobility in factors of production and no risk of default. The portfolio approach to FDI, presented in reaction to The MacDougall model emphasizes not only return differential, but also risk (Iversen 1935 and Tobin 1958, both cited in Agarwal 1980). This is strengthened by a theory which emphasizes the positive relationship between FDI and output (sales in host country) along the lines of Jorgenson's (1963) model (see Agarwal 1980).

A major criticism of these theories relates to the question of perfection in markets. Hymer (1960, published in 1976) and Kindleberger (1969) argue that, if foreign firms were able to compete and succeed in the host country, then they must be in possession of a specific and transferable competitive advantage, both over local firms, and other potential entrants into the local market. This analysis also focuses on the micro foundations of FDI, by moving from a simple capital movement/ portfolio theory to a broader production and industrial organizational theory.

Indeed, this school of thought has formed the basis for a whole strand of the literature. According to this line of thinking, some advantages of the competitive foreign firm include cheaper sources of financing, the use of brand names and patent rights, technological, marketing and managerial skills, economies of scale, and entry and exit barriers (Kindleberger 1969, Agarwal 1980).

A related micro-based theory of FDI has also emerged with the development of the Vernon's product cycle theory (Vernon 1966). The product cycle theory represents an advance on previous theories, in that it incorporates an analysis of oligopoly and strategic market considerations. Based on Vernon's theory of 'product cycle', and the existence of 'new' and 'old' goods, Krugman (1979a) further develops this theoretical avenue for explaining FDI flows. Specifically, he extends the analysis to a North-South framework with innovation (in the 'North') and technology transfer (to the 'South') representing its crucial aspects. Krugman notes that technological progress raises the marginal product of capital and provides an incentive for foreign direct investment. On the other hand, this process may be reversed through technology transfer (Krugman 1979a: 263–65). Mainstream trade theories usually underlie this type of analysis. Indeed, recent theories of trade, such as that of the 'economies of specialization' which emphasizes the existence of intra-industry (as well as intra-firm) trade, also provides scope for analysis of FDI (see, for instance, Ocampo's survey, 1986: 152–55).

Notwithstanding Vernon's contribution, a second wave of refinements to the neoclassical capital movement/portfolio theory of FDI, building upon Hymer's original contribution came with the emergence of explanations based on the ideas of 'international firm' and 'industrial organization'. The fact that decision making about FDI takes place within the context of oligopolistic firm structures - and that such investment includes a package of other inputs, such as intermediate imports and capital flows - has led to the development of alternative explanations grounded in the theory of industrial organization (see Agarwal 1980, Helleiner 1989: 1452, Dunning 1993). In this approach, as set out by Hymer, foreign firms are seen as having an advantage over local ones. The foreign firms pursuit of FDI is explained by the theory of internalization. This is characterized by the desire to minimize transaction costs, à la Coase (1937), to tackle risk and uncertainty, increase control and market power, achieve economies of scale, and ensure advantageous transfer pricing (Hymer 1976, Buckley & Casson 1976). In this approach, oligopoly power is seen as mitigating, rather than creating, market imperfection (Helleiner 1989). The recent works of Dunning (1993), which he terms the 'eclectic paradigm', represents a culmination of this trend towards a refinement of theories of FDI. Without departing much from the Heckscher-Ohlin-Samuelson theory of trade, in explaining spatial distribution of multinational firms, Dunning's paradigm summarizes this strand of theory under an 'ownershipspecific, location and internalisation' (OLI) framework (see Dunning

1993). Helleiner notes that 'this "eclectic" theory of direct investment ... drawing on firm-specific attributes, location advantages and internalisation advantages – is widely accepted' (Helleiner 1989: 1253). There also exists an international trade version of FDI determination (termed the macro approach) which is associated with the works of Kojima (Kojima 1982). The Kojima model argues that FDI may be explained by the 'comparative disadvantage' of industry within the investing countries. According to Kojima's theory, this may be mitigated by investing in foreign industry, which may be able to achieve comparative advantage in the production of a particular product and potentially, even export back to the home country. Naturally, this type of FDI will also have the effect of increasing trade volumes (Kojima 1982).

Another strand of literature, which is often overlooked in mainstream analysis, are Marxist theories of FDI determination. Citing historical and other empirical evidence from Britain and the United States, Baran and Sweezy (1966) argue that FDI represents an outlet for investment-seeking surplus, resulting from stagnation in the centres of capitalism. Indeed, according to Marxist theories, FDI also represents a mechanism for extracting surplus from underdeveloped areas (Baran & Sweezy 1966: 110-11). Thus, Magdoff (1992) argues that the 1970s and 1980s exhibited a slow down in economic activity - itself an inherent feature of capitalism, according to Marxist theory - and that this slow down spurred capital to seek and create new profit opportunities. Thus, the speeded-up flow of direct investment from one country to another is itself seen as a reaction to stagnation in the capitalist centres (Magdoff 1992: 9-13).³ According to Magdoff, the 1980s witnessed world FDI growing at an average annual rate of 29 per cent, and the pattern of such investment increasingly switching to finance and insurance, real estate, advertising and the media (as opposed to the traditional sectors of manufacturing and raw material extraction).⁴ In spite of such changing patterns. Magdoff sees stagnation in the centre and the search for profit as representing the main reasons for FDI.

In sum, the theory of determinants of FDI covers a range of explanations: the pure capital movement, product cycle, industrial organization, the stagnation thesis as well as other political considerations. In the African context, the pure capital theory does not work since the assumptions simply do not hold. Neither is Krugman's hypothesis workable, since it is more relevant to countries with a good industrial

base and infrastructure. As discussed in Chapter 1, the deterioration in terms of trade, combined with the debt crisis, will greatly undermine the relevance of this theory, in the African context. The most probable theoretical explanation seems to be found in the Marxist version, as well as 'industrial organization' and 'the international firm' (the 'eclectic') explanations. The Marxist version focuses primarily on the consequence of FDI, which is not the prime focus of our empirical study. Besides, its stagnation thesis may not fully explain FDI destination as much as its source and might also be inferred from the industrial organization and international firm-based theories. On the other hand, the concentration of multinational corporations in the mining sector of most African countries and, to a good degree, the importance of the colonial history in determining their spatial pattern might be taken as lending support to the importance of the 'eclectic' approach. This theoretical insight will be used in identifying the determinants of FDI in the empirical analysis and construction of the model undertaken in Chapters 4 and 6.

2.2.2 Theories of Bank Lending and Optimal Borrowing

Bank lending also represents an important source of finance for Africa. There are a number of systemic explanations for such lending. These include static/dynamic capital movement theories (MacDougall 1960, Bardhan 1967), the portfolio theory of capital movement (see Williamson 1983: 182-92, Solink 1974) and theories of credit rationing (see Sachs 1984 and Stiglitz & Weiss 1981; for an empirical study of credit rationing, see Feder & Uy 1985 and Lee 1993). All of these are essentially microeconomic explanations, tailored to a neoclassical framework. The first explains how capital is allocated across countries and generations in order to equate return differentials, with the aim of achieving optimal borrowing. The second approach adopts a portfolio choice methodology, owing to the emphasis it places on the evaluation of risk and uncertainty. However, the portfolio theory is essentially governed by the same motives as the first approach. The credit-rationing theory builds on both of the above approaches, but also incorporates a constraint based on market imperfection. A common problem in applying these approaches is their implicit belief in the workings of a competitive market, albeit with some reservations in the case of the credit-rationing theory. A further weakness is the assumption that theories based on micro behaviour may readily be carried over to the macro/international level. This theoretical underpinning greatly limits the relevance of these approaches for countries in Africa where risk, uncertainty and instability all represent significant factors, which need to be taken into account. Neither is it reasonable to assume that fully developed financial markets are available in all African countries. Indeed, most African countries are sovereign borrowers. In the following section, therefore, we will examine a number of alternative theories, which seek to explain borrowing in Africa.

2.2.3 Models of Sovereign Lending and the Institutional Response

The important distinction between lending domestically and lending to other sovereign countries has led to the emergence of a number of theories, which are summarized in this sub-section. At the heart of all of these stands the issue of how the risk of default issue is dealt with. More specifically, theorists seek to explain why creditors continue to lend to countries which do not have collateral and, hence, where risk of default is a real possibility. One of the first explanations for this phenomenon was set out in a model developed by Eaton and Gersovitz (1981). They argued that private creditors could take a number of retaliatory actions, such as exclusion from future borrowing, in order to penalize defaulting debtors by rationally anticipating the inherently dishonest nature of such debtors. For the debtor, the major cost of default is the possibility of endangering future borrowing which might be required to smooth the variability (and growth) in its income (Eaton & Gersovitz 1981).

According to some authors such as Cooper and Sachs (1985), countries' capacity to borrow, and pay their international debt is likely to depend on the nature of their crisis. This may take three general forms. Firstly, they may become insolvent, and so, incapable of servicing their debt in the long run. Secondly, they may become illiquid, and so not have cash on hand to pay outstanding debt obligations. Or, thirdly, they may be insouciant, that is, unwilling to pay, being conscious of the fact that there are economic gains to be had by repudiating debt (Smith & Cuddington 1985: 21–26, Eaton & Taylor 1986: 217–20). Cooper and Sachs (1985) also raise the possibility that, if debtors should become illiquid, but not insolvent, that this may lead to rescheduling or moratorium mainly as a consequence of panic and failure of individual banks to co-ordinate their actions.

Building on this last suggestion, Sachs (1984) attempts to develop a simple model of debt-repayment feasibility. Thus, he argues that, for debtor nations to be able to repay their debts, requires that their indebtedness at any particular time should not exceed the 'national productive wealth' (defined as the discounted value of GDP net of investment). In other words, that debtor nations should not become insolvent (Sachs 1984: 6-7: see also Smith & Cuddington 1985). In Sachs' model, optimal borrowing is arrived at by maximizing consumption, subject to a production function, capital stock, level of consumption and the maximum possible debt (Sachs 1984: 6-8).⁵ Sachs also analyses the possibility of default and considers this to depend on the benefit and costs of repudiation. Thus, the borrower is considered to be a rational economic agent who will default when the cost of default is comparable to the level of relief from debt that would be obtained by defaulting. Sachs also notes attempts to diversify this risk (especially by American banks) as well as the formation of syndicated banks to tackle such problems. Indeed, he argues that a borrower confronted with problems of liquidity (as opposed to insolvency) can more easily and safely be provided with additional loans by syndicated, than individual banks. Syndication may also help to avoid the 'free rider' problem, in which an individual lender may demand full repayment at the time that other lenders reschedule (Sachs 1984: 33). Similar arguments have also been developed in an earlier work of Cline (1983).

The first major weakness in both Cline's (1983) projection model and Sachs' (1984) approach is that they consider debtor countries as representing rational atomistic agents in a perfect market setting, capable of assessing the benefits and costs of repudiation. However, in practice, the structure of capital markets, combined with trade sanctions, denies any such choice to debtors. Thus, Bulow and Rogoff (1989) have developed a model, which explicitly addresses this issue. They argue that, although collateral is insignificant and limited (except possibly in aircraft financing), debtors' desire to maintain a good reputation, taken alongside lenders capacity to impose sanctions that will impede trade and financial market transactions, together represent effective incentives to ensure repayment. Of these, the latter is likely to be the more important. Moreover, as Bulow and Rogoff point out, previous models invoke penalties without exploiting the possibility of negotiation. However, such negotiation *is* possible and bargaining-theoretic approaches may be applied in explaining these (Bulow & Rogoff 1989). The second weakness of the Sachs- and Cline-type models is that, by focusing primarily on the individual lenders' perspective, they downplay the potential impact of efforts towards syndication. Some obvious examples of the potential effects of syndication include mark-up pricing, commissions, and, market segmentation. Another extreme example is where borrowers are forced by such syndicated lenders to accept a policy, such as structural adjustment programmes, in line with the objectives of these lenders.

It should be noted that the earlier theories are essentially neoclassical since they assume the existence of optimizing agents, acting in a competitive capital market with the possibility of imperfection (as in the case of credit rationing). One problem with these theories is that the neoclassical assumptions clearly do not apply in a situation characterized by 'sovereign borrowers' and syndicated lenders. Hence, the importance of such 'concerted' lending is emphasized in the works of Cline (1983) and Sachs (1984). However, it is not reasonable to assume that the institutional response to the debt crisis evolved from the atomistic actions of borrowers and creditors making decisions based on a marginal assessment of risks and returns within a neoclassical market setting. Rather, the importance of analysing the institutional response to the debt problem is increasingly being recognized.

In Folkerts-Landau's (1985) view, it is precisely the problem of 'sovereign risk'⁶ and the simplicity of tackling this through syndicated international bank lending which has led to a shift in the nature of development finance. Thus, from the use of bond finance and direct investment, common in the 1920s and 1930s (and to which the above theories relate), international bank lending became increasingly important during the 1970s and 1980s (see also Bacha & Alejandro 1982, Kletzer 1984, Sachs & Cohen 1984 cited in Eaton & Taylor 1986). Folkerts-Landau proposes two explanations for this phenomenon. Firstly, 'financial authorities were perceived as being increasingly willing to protect the deposit liabilities of their large money-centre banks'. Secondly, international banking proved to be effective in tackling the risk of lending to sovereign borrowers by taking collective action such as limiting access to capital markets. This led to restructuring of debt, through rescheduling and stabilization programmes (Folkerts-Landau 1985: 317-18).⁷ As part of a logical effort to address the issue of com-

petitive market failure, Folkerts-Landau argues for an institutional evolution of banks, who are seen as moving 'away from pricing to offset the expected risk of outright default to a new system where rates on syndicated international credits reflect only the (lower) expected cost of rescheduling'. This has resulted in a lowering, and reduction in the variability, of interest rates on loans, even compared with those of the well-developed domestic financial markets of the North (Folkerts-Landau 1985: 332). Folkerts-Landau's approach is rooted in treating the bank as a firm, rather than simply as an investor, as is the case in portfolio theory. Thus, the bank, acting in its capacity as firm, is seen as pursuing a strategy, which entails more than one goal. These include, inter alia, market share, stability, as well as a host of goals other than profit maximization (see Vos 1994). A related explanation is the concept of 'loan pushing and revulsion'. The essence of this argument is that commercial banks lent to LDCs in a syndicated fashion, in reaction to a weakening in demand for loans from sources in the developed world. However, as loan levels built up, and perceived risk grew, banks began to feel pressure to retreat from this type of lending. The preferred tactic adopted in such a situation was usually a shortening in the maturation period on loans (see Eaton & Taylor 1986: 239-41).8

Finally, the late 1980s and early 1990s witnessed the inability of debtors to pay back their debt and consequent debt overhang problems. This led to a number of theoretical advances relating specifically to the development of efficient, usually market-based, methods of debt forgiveness (see Cline 1995, Cohen 1994). This is well summarized by the debt Laffer curve, associated with Krugman (1989; see Diagram 2.1). On this, the vertical axis measures the expected payment and the horizontal axis the nominal value of the country's debt. The relationship between the two is depicted by the curve CD. At lower levels, nominal debt may be expected to be fully repaid (OC equals the 45°-line). The ray OL gives the ratio of expected to nominal debt. Assuming risk and transaction costs away, this ray gives an idea of the secondary-market price of debt. A movement away from C is associated with a declining secondary-market price: The curve DRLC, the *debt relief Laffer curve*. shows that, in a similar way that government tax rate reduction might increase the total level of tax revenue collected, so too creditors may increase expected payment by forgiving part of a country's debt. This type of incentive effect usually becomes significant when the debt is very large, such as beyond point R, in Diagram 2.1.

Diagram 2.1 The debt relief Laffer curve (Krugman 1989: 264)



To sum up, we have noted that the assumptions, upon which the static/dynamic and portfolio theories of capital movement are based, ignore both the institutional framework, within which the market operates, and the evolution of institutions, which market problems give rise to. Folkerts-Landau's approach and the 'loan pushing' theories are just some of the attempts to address these issues. As the empirical analysis on credit rationing also shows, as far as African economies are concerned, the capital market is clearly a segmented one. This segmentation of the capital market, and resulting credit rationing, gives a powerful insight into how African economies are positioned within the global financial market (see Vos 1994). Such institutional development can also shape the nature of official capital flows, influencing, for example whether these are likely to take bilateral, multilateral or some other form. Such flows, which have different institutional features, are particularly important for Africa. Therefore, the next section will examine the theoretical underpinnings of these flows.

2.2.4 Official Capital Flows

Official capital flows – comprising bilateral, multilateral, concessional and non-concessional loans, and foreign grants – represent an important component of financial flows to African countries. The basic questions which we set out to answer in this chapter are 'what determines official capital flows to the South?' and 'why such flows?' The search for an answer to these questions leads one to different, sometimes conflicting, theoretical explanations. Thus, one school of thought maintains that capital flows are determined by the *economic and geo-political* interests of donors. Indeed, this suggestion finds support in a number of studies (see, for example, Mikesell 1968, OECD 1985, Mosley 1985, Ruttan 1992, McGillivray & White 1993). Another major explanation for these aid flows relates to humanitarian or developmental considerations (studies supporting this viewpoint include Streeten 1976, cited in Gasper 1992; Riddell 1987; and the 'aid as a public good' literature of Mosley 1985, Dudley & Montmarquette 1976 and Frey 1984).⁹

A number of studies investigate one or both of these explanations empirically. Indeed, Beenstock (1980), Mosley (1985) and White and McGillivray (1992, 1993) have gone so far as to portray these empirically-based studies as representing a distinct approach, devoid of theory. However, I prefer to view these simply as empirical manifestations of the first two theoretical explanations.

These three strands of the literature will now be discussed, and an attempt to relate this discussion to Africa's economic conditions, as discussed in the previous chapter, will be made. I begin by examining the literature, which sets out to explain capital flows to Africa in terms of *economic and geo-political considerations*, before moving on to examine *aid as public goods* as well as *empirical explanations* for such flows.

2.2.4.1 Economic and geo-political factors

Here, again, we are seeking to answer the basic question 'why such flows?' However, we will also examine the interrelated issue of how these flows are distributed among countries. It could be argued that this issue is simply an empirical manifestation of the first question. Hence, creating a dichotomy between these, as White and McGillivray (1993), and others, have done, may be analytically problematic (unless this is considered as simply representing a problem of 'operationalizing' within a quantitative studies framework).

In an attempt to answer the first of the above questions, Ruttan (1992) examines the philosophical underpinnings of foreign aid. Ruttan maintains that there are two arguments for such aid. Firstly, 'economic and strategic self-interest', and, secondly 'ethical¹⁰ responsibility' of donors. According to the author, the former entails an inherent contradiction since 'self-interest' goals may usually be achieved at a cost to the recipient country. As far as the 'ethical argument is concerned', Ruttan maintains that 'in a society of free people the concept of social or distributive justice has no meaning'. Stated differently, 'justice is a function of the rules or processes that govern individual and group behaviour and not of the outcome generated by the rules'. Citing Bansfled (1963), Ruttan notes that the extreme argument can be 'our [the North's] political philosophy does not give our government any right to do good for foreigners' (Ruttan 1992: 219). However, Ruttan questions the validity of the ethical argument when writing 'I would personally prefer strong behavioural foundation on which to rest conviction about moral responsibility for assistance to poor countries' (ibid.: 220). Notwithstanding Ruttan's doubts, the ethical or moral argument, as set out by Streeten (1976), is founded on a belief; (a) that there exists a 'brotherhood of men' *[sic]*; (b) that the poor are not fully responsible for their poverty; and, (c) that the rich are partly responsible for this poverty. Opeskin (1996), Riddell (1987) have also pursued a moral case for inter-governmental foreign aid (see also Gasper 1992: 6-8).

According to the OECD's (1985) explanation, its aid in 1950s was governed by strategic consideration (*vis-à-vis* the USSR). During the 1960s, the level of aid declined, due to economic problems in donor countries. In the 1970s and 1980s, the oil crisis brought new demand and supply into the market. However, in general, OECD countries maintained aid flows, in line with the long-run political and economic interests of the industrialized countries, and the general humanitarian leanings of Western civilization (OECD 1985:13–14). Hence, the exact geographic distribution of such aid is governed by the bilateral donors' perceptions of these driving forces.

Mosley et al. (1987) summarize the argument by identifying three 'universal' reasons for aid. Firstly, aid may have a *redistributive* function, being motivated by a sense of responsibility for the poor. Secondly, aid may have an *allocative* function, given perceived imperfec-
tions in capital markets. And, finally, aid may have a *stabilization* function, being disbursed with the aim of augmenting world aggregate demand and reducing unemployment. Further, at what Mosley et al. terms a 'particular' level, bilateral aid may be used to buy political support and increase exports from the donor country (Mosley et al. 1987: 12). However, this approach has been criticized since the weight attached to Mosley et al.'s 'universal' reasons are very controversial and, thus, give rise to a number of empirical questions, which we discuss below. It can also be argued that Mosley et al.'s 'particular' reasons are incomplete, since they ignore syndicated efforts to achieve those bilateral motives, which are increasingly being made 'universal' among donors. It is worth noting that Mosley et al.'s (1987) work is recent enough that it could have taken account of such developments.¹¹

Magdoff (1969, 1992) argues that capital flows from the North to South are best understood as part of the historical development of capitalism, which he sees as born in the process of creating a world market. Hence, growth in the core industrialized countries was associated with a centuries-long spread in the sphere of influence of these countries, through conquest and economic penetration. (According to this viewpoint, the drive for accumulation and control of raw materials is believed to push business beyond political boundaries.) He observes that growth of industrial production has slowed from decade to decade (between 1960 and 1990), and that this has spurred capital to seek and create new profit opportunities. Thus, for Magdoff, it is within this general context that flows of capital are best understood (Magdoff 1992: 45–50). With regard to the specific issue of aid, for Magdoff, this is simply an instrument of control to ensure the political, economic and military interests of the donor nations (Magdoff 1969: 117).

This broad theoretical/philosophical discussion is manifested in specific cases. Thus, Miksell (1968) attempts to explain official capital flows by examining these from the US and European perspective. For the former, such flows represent an instrument of foreign policy as well as a mechanism for securing national security, particularly in the context of the cold war. Capital flows were also used to obtain economic benefit, particularly through the use of export credits and the promotion of direct foreign investment. However, Miksell questions whether these flows are best explained by business, national interest or humanitarian¹² motives. For Europe, these flows are seen as a continuation of the colonial/dependency relationship (Miksell 1968: 1–14).

2.2.4.2 Aid as a public good

The other attempt to explain the underlying reasons for aid is articulated around the theory of 'aid as public good'. Mosley (1985) distinguishes between studies, which attempt to explain aid in terms of the theory of public goods, and those, which adopt an empirical approach. without attempting to apply explanatory theory. Vos (1994) terms the latter a 'descriptive and empiricist approach' and the former as 'theoretical approaches' (Vos 1994). Mosley's approach is to treat foreign aid as a public good in an imperfect market with ignorant consumers (namely, donor country taxpayers). Citizens are assumed to have a demand for this public good and they are able to adjust their actual consumption to the desired level not simply through the market but also through political action. However, it is worth pointing out that both governments and electors may be ignorant about the end product of aid. and, indeed that voter's demand for aid could be altruistic. These facts effectively allow Mosley to exclude the 'price' of aid (in the form of donor country taxes) from his aid demand equation. This leaves the relative level of donor's income, together with the quality of aid (the degree of concessionality), as determining factors. Supply is seen as being determined by the previous year's budget, which is marginally influenced by the treasury and a reflection of the state of domestic economy. White and McGillivray (1992) have termed this the administrative/incremental approach. Other factors seen as influencing supply are the relative performance of other donor countries (a sort of 'bandwagon effect') as well as an adjustment parameter between supply, as determined by the above factors and the electorates demand for such aid.

Based on this empirical approach, Mosley concludes that, past disbursements, as well as other countries' disbursements, represent significant determinants of aid supply. On the other hand, international aid is found to be mildly responsive to the state of the donor nation's domestic economy. This last result leads Mosley to conclude 'either that my original hypothesis about the public demand for aid was wrong or simply that the government of that country has not chosen to respond to public demand'. However, there are three basic problems with this approach. Firstly, there is no reason to expect that electors in the North will act effectively in support of increased aid flows by voting, where aid represents a negligible issue in the overall political process. Secondly, factors posited as influencing supply, are likely to explain only the marginal effect and not the central cause. Thirdly, and perhaps related to the second point, one can hypothesize that the demand-side finding (aid's insensitivity to the economic conditions in the donor country) might be attributed to the view that aid is a necessity (not a luxury) to the donor. It might also be taken as showing that, in general, the assumption of no difference between the objective of the consumers (electors) and the government of countries in the North is too weak to theorize upon.

Dudley and Montmaguette (1976) adopt a similar approach to that, later adopted by Mosley, in that they see aid as a public good, consumed indirectly by donors, or, alternatively, donated with the expectation of receiving something, intangible and difficult to quantify, in return. Thus, based on this line of thinking, it could be argued that donors' supply is determined by their own demand. The authors attempted to model supply at two levels: with and without administrative costs. In the latter model the probability of grant is related to the income and population size of the recipient. The empirical conclusion of the model is that granting aid is a decreasing function of recipient per capita income, as well as economic, political and 'bandwagon' (relative to other donor) considerations. Frey (1984) also attempted to explain aid flows. by considering these as a public good, even though his empirical evidence suggests that aid is not a public good from which all nations benefit. Frey's approach differs slightly from the others, in that he combines the concept of aid as a public good with the use of game theory. In his approach the recipient country is characterized as maximizing aid, or maintaining its independence, in a situation where donors are seen as competing with each other. Frey's main conclusion is that such interaction is quite complex, and hence that 'the behaviour of each actor has to be identified before any conclusion can be drawn' (Frey 1984: 95).

2.2.4.3 Empirical studies

A further branch of the literature comprises empirical studies which seek to answer the question 'why aid?' In one of the earliest studies, McKinley and Little (1979) develop a 'recipient need and donor interest model' which sets out to examine the 'humanitarian' versus 'national (donor) security' explanations for aid flows. According to this 'recipient need model', aid should be allocated in proportion to the economic and welfare needs of the recipient, otherwise, such aid is simply fulfilling the donor's interests, either as a commitment, or as leverage strategy. After discussing, at length, how these propositions might be operationalized, McKinley and Little conclude that the recipient need argument is likely to be a secondary one. However, while this paper focuses on bilateral issues, syndicated efforts are more widely followed today by international and regional organizations, in their effort to realize donor interests. In this study, we do not examine this phenomenon. However, we would refer the interested reader to Anyadike-Danes and Anyadike-Danes (1992) review of evidence relating to European Community (EC) aid to the African Caribbean and Pacific (ACP) region. They conclude that such aid is strongly associated with the pre-Lome association.

However, most econometric studies of recipient need models are not so robust. Having surveyed such models, White and McGillivray (1993) note that the separate estimation of recipient need/donor interest models suffers from specification error due to the omission of relevant variables, which are usually not orthogonal, (i.e., correlated with all included variables). This, they suggest, leads to OLS estimates with bias. They argue that this problem is inherent in the very methodology of this approach. White and McGillivray illustrate how different results may be obtained if one allows for correction of such specification errors (White & McGillivray 1993: 36–41). To this, one might add the observation that such time series studies might as well suffer from spurious regression.

In his study, which sets out to explain what determines total aid volumes, Beenstock (1980) starts from the assumption that political factors affect the geographical distribution and not the total volume of aid. He points out that, whatever the objective of aid, its volume is constrained by GNP (or GNP per capita), the balance of payment, levels of unemployment and the size of net budget surplus of the donor state. At a statistical level, all signs are as expected and, with the exception of the budget term, all are significant at a 5 per cent significance level (Beenstock 1980: 142). Using a time trend, Beenstock also tested whether there was a tendency for ODA to increase over time. He concludes that that there is a positive trend. Although Beenstock's analysis focuses on the supply side of the issue, this analysis does not explain the central reason for aid. Further, it is quite difficult to envisage a situation where politics is used solely in allocation, and not also in the determination of total supply.

Finally, this section will conclude by summarizing the findings of a recent survey of the literature on allocation of aid, undertaken by White and McGillivray (1992, 1993). White and McGillivray adopt two broad classification schemes: firstly, descriptive measures, which are evaluative in their nature, while measuring donor performance (White & Mc-Gillivray 1992: 1); and secondly, explanatory studies which trace their origin to political-economy theories, and base their explanation on 'political, strategic, commercial and (albeit often begrudgingly) humanitarian motives¹³ (White & McGillivray 1993: 2). These surveys raise a number of important issues, which are useful at various stages in the subsequent analysis. They conclude by maintaining that models should approximate the actual practice of aid determination process. Perhaps more importantly, these surveys also highlight how aid allocation is the outcome of a bureaucratic decision-making process, economic, political and other relations between the donor and the recipient (White & Mc-Gillivrav 1993: 68). This line of thought will be re-examined in subsequent sections of this book, in which issues of empirical analysis will be addressed.

2.2.5 Conclusion

To sum up, this section has attempted to examine a number of systemic explanations for the determinants of capital flows, with the aim of understanding the external finance problems of Africa. This analysis has been carried across three categories of capital flows, namely FDI, bank lending and official flows, each of which has its own peculiar features. A number of implications of each of these types of capital flow for the rest of this study may briefly be outlined.

Firstly, looking at FDI, it is noteworthy that the level of this type of capital flow to African countries is extremely low. This may broadly be attributed to low levels of infrastructure, poorly developed markets and absence of a skilled workforce in many African countries. (Indeed, for similar reasons, levels of portfolio investment in most African countries are also insignificant.) Since such factors are implicitly assumed to be similar in many of the earlier capital movement and technology-based FDI theories, their relevance to Africa is likely to be limited. However, an eclectic approach could be important in demonstrating empirically why levels of FDI to Africa are so low. Indeed, such an exercise is undertaken in Chapter 4, and the results of this analysis are used to construct the model developed in Chapter 6.

Bank lending in Africa largely falls under the 'sovereign lending' category. This is particularly the case since most loans are either public, or publicly guaranteed. However, for Africa, accessing this market may not always be straightforward. The empirical evidence also supports the segmented nature, and prevalence of credit rationing in this market, as far as Africa is concerned. Thus, the segmented capital market and credit-rationing framework gives an important insight into the analysis of bank lending to Africa. Indeed, this insight is used in building the international banking sub-block of the model developed in Chapter 6.

Nonetheless, whatever the nature of lending, the sustainability of debt, as discussed in Chapter 1, requires a reduction in vulnerability in the export market, as well as sustained investment, especially in infrastructure. These fundamental problems are not addressed, and in fact are aggravated, by the policy of stabilization and structural adjustment being implemented in Africa. In the recent past, these polices have also led to an increase in the importance of official flows, often on concessional terms. If one is to fully understand how these policies operate, therefore, one not only needs to understand their impact (as discussed in Chapter 5), but also the determinants of such official flows. Indeed, this information will prove extremely useful in attempting to model the patterns of such official flows. Again, based on the literature discussed above, the economic, strategic and political self-interest and developmental considerations arguments could give us an important handle on this type of flows. Hence, in combination with the empirical analysis discussed in Chapter 4, this insight will be used in building the official capital flow sub-block of the model developed in Chapter 6.

To sum up, although the discussion, so far, has been helpful in sharpening our understanding of the financial side of the external finance problems of Africa, as argued in Chapter 1, this only represents one side of the inquiry. Rather a more fundamental issue is the impact of trade on the external finance problem of the continent. In the next section, therefore, we will attempt to give a theoretical underpinning to this issue.

2.3 INTERNATIONAL TRADE THEORIES AND THEIR RELEVANCE IN MODELLING AFRICA

2.3.1 A Brief Overview of Trade Theories and their Implications for Africa

In the African context, the implications of the classical trade theories (Smith 1776, Ricardo 1817) relate to Africa's role as specialist producers of primary commodities. The rigidity of the classical school's assumptions, particularly in relation to the structure of costs, led to the evolution of the neoclassical (or orthodox) trade theories developed at the turn of this century (Heckscher 1919, Ohlin 1933, Stopler & Samuelson 1941, Samuelson 1948). The emphasis on factor endowment differentials as the source of trade in these theories, in turn strengthened the policy implications of the classical theories about Africa, and in particular the continent's role as a specialist producer of primary commodities and tropical products.

By late 1950s and 1960s, the importance of technology in explaining world trade patterns led to the emergence of trade theories based on technological gaps (see Posner 1961, Hufbauer 1966 and Vernon 1966). These theories – for instance, the approach advocated by Vernon - helped in the development of a North-South model based on the importance of technology and innovation (see Krugman 1979a). Although coming from a slightly different perspective, Emmanuel had also put forward a similar proposition, in the early 1970s. Both Emmanuel and Krugman questioned the factor mobility assumptions of traditional trade theories. However, they differ in that, while the former focuses on labour, the latter is more concerned with technology. In both cases, however, the South is placed in a relatively disadvantageous position. In general, technology gap models within a North-South perspective require a high level of infrastructure, with cheap, semi-skilled labour, and are based mainly on trade in manufactured goods. However, such a structure may apply only to a fraction of the countries in the South, and is certainly not relevant for Africa (see also Stewart 1984: 92).

The apparent failure of the above theories in explaining the evolving trade patterns, especially among developed countries, has led to the development of other theories based on imperfect competition and scale economies. Partly inspired by the technological gap models of the 1960s (see, for instance, Posner 1961: 329), as well as dynamic and industrial organization theory, which their exposition implied, writers such as Krugman (1979b, 1980, 1981), Dixit and Norman (1980), Lancaster (1980), Helpman (1981), Ethier (1982), Lall (1973) and Helleiner (1981) have developed what might be called the 'new'¹⁴ trade theories.

According to Krugman's excellent summary, there are at least two distinct features of these theories which distinguish them from the traditional ones (Krugman 1992). Firstly, the importance of *increasing* returns/scale economies as a cause of trade: and secondly, the need to model international markets as *imperfectly competitive*. Two distinctive features are usually emphasized here: firstly, 'intra-industry' trade¹⁵ (see Balassa 1967; Kravis 1971; Grubel 1970; Dixit & Stiglitz 1977; Krugman 1979b, 1980, 1981; Dixit & Norman 1980; Lancaster 1980; Ethier 1982; Falvey 1981; Helpman 1981); and secondly, 'intra-firm' trade (see, among others, Lall 1973, Helleiner 1981). In the former case, economies of scale, and trade in differentiated products are emphasized. The latter, on the other hand, emphasizes the importance of transfer pricing in the operation of multinational enterprises (MNEs), the irrelevance of the nation state as a unit of analysis as well as the importance of joint profit maximization by all the subsidiaries of such enterprises.

What, then, are the implications of these theories in the African context? According to Stewart (1984), the new trade theories are designed primarily to explain trade patterns in the North, in which there exists a great deal of preference similarity (see Linder 1961). When these theories are observed, within a North-South perspective, the South is generally found to have a comparative disadvantage, since it does not have the market size (with appropriate income) to exploit economies of scale. Moreover, attempts to move into larger markets would entail a transportation cost, which would further accentuate this disadvantage. Apart from this fact, owing to income differential, the combination of characteristics of preferred consumer and producer goods differs between the two regions. This implies lower welfare for the South, since it has no alternative but to consume Northern products. Besides, free trade at the initial stages, under such a set-up, effectively eliminates local production in the South. The logical implication of this theory is that South-South trade should take place (Stewart 1984). However, Stewart warns that further evaluation is required, in order to determine whether South-South trade is best seen as a substitute for, or a complement to, North-South trade.

Based on his analysis of intra-firm trade, Lall (1973) notes that the welfare impact of such trade has resulted from a bargaining process between MNEs and the South. The existence of intra-firm trade acts 'as a powerful bargaining counter in the MNEs' favour, enabling it to conceal from the government a crucial item of information' (Lall 1973: 190). Similarly, Helleiner (1981) stresses that transfer pricing by MNEs adversely affects gains by the South through tax evasion, transfers, and legitimate share of profit of joint domestic owners. In terms of the structure of Southern economies, he argues that intra-firm trade has the effect of creating an enclave sector and ill-formed import substitution industries in the South. This also affects the direction of volumes of trade towards the interest areas of the MNEs, at the expense, say, of regional integration (Heleiner 1981: 14-15). In a nutshell, he concludes that the intra-firm framework implies that the future expansion of the South's trade could be managed as part of the long-term investment planning of MNEs, in order to minimize possible later disruption or surprises (Helleiner 1981: 65). To sum up, the relevance of the 'new' trade theories in explaining trade between North and South, as opposed to North-North trade, is highly limited. This implies that a search is needed for other systemic explanations, which might help to explain North-South trade, in general, and trade in primary commodities in particular. This is undertaken in the following section.

2.3.2 Non-orthodox Models and Trade in Primary Commodities

In parallel with the above theories, a critical (non-orthodox) school has provided an alternative analysis about the gains from trade. Analysts from this school have focused particularly on North-South trade. This is discussed in the next two sub-sections. Since primary commodities represent the most important component of the South's trade, in general, and Africa's trade, in particular, in this section we will briefly survey a number of issues relating to the modelling of primary commodities. This focus on modelling stands in line with the general objective of this book, which is to develop a formal model for this market and integrate this into a global macro framework. Such an approach helps in the analysis of external shocks, and is also useful in terms of setting policy options within a wider framework.

2.3.2.1 Unequal exchange

The distinction between the structuralist school and Emmanuel's unequal exchange theory is a blurred one. This is particularly the case in relation to terms of trade. Although, chronologically speaking, the ideas of Prebisch and Singer precede those of Emmanuel, for various reasons we begin by examining the unequal exchange theory. Firstly, the unequal exchange theory sets a general framework for a discussion of deterioration in terms of trade. Secondly, the alternative, structuralist school's view was still being debated as recently as the 1990s. And, finally, such non-chronological ordering is helpful, in terms of linking the discussion to current issues concerning primary commodity production and trade by African countries.

Emmanuel's (1972) unequal exchange theory is informed by an attempt to understand why there has been a secular deterioration in the terms of trade in primary commodities produced by developing countries. More specifically, the theory questions whether this results from the fact that these commodities originate from poor countries (Emmanuel 1972: xxx).¹⁶ Bacha (1978) and Ocampo (1986) beautifully summarized the main argument of the unequal exchange theory using a few equations. Thus, if we let w_i be the real wage in country *i* in terms of Southern commodity, *q* average productivity of labour, *P* price, and employ the Ricardian assumption of capital as wage fund, on which a uniform international average rate of profit *r* is made, owing to capital mobility, then the equilibrium price is given by:

$$P = (1 + r) w_N / q_N$$
 [1a]

$$1 = (1+r) w_S / q_S$$
 [1b]

Under these conditions, the barter (P) and factoral (f) terms of trade are given by:

$$P = (w_N / w_S) / (q_N / q_S)$$
[2]

$$f = w_N / w_S$$
^[3]

Emmanuel maintains that 'it [is] clear that inequality of wages as such, all other things being equal, is alone the cause of inequality of exchange' (Emmanuel 1972).¹⁷ Gibson (1980) labels this casual association of relative wages and the barter terms of trade the 'Fundamental Theorem of Unequal Exchange' (cited in Ocampo 1986: 132). However,

the unequal exchange theory has been severely criticized from a host of different angles (see, for instance, Bettelheim 1972, Amin 1974a & 1974b, Mack 1974, Kay 1975, Sau 1978 and Johnstone 1980).

2.3.2.2 The structuralist school and the Prebisch-Singer hypothesis

The discussion about terms of trade has taken concrete form in the works of the 'structuralist' economists of the Economic Commission for Latin American and the Caribbean, ECLAC (Prebisch 1950 reprinted 1962, and Singer 1950).¹⁸ Having highlighted the enclave nature of export sectors, and the role of industry within a protected domestic market as a dynamic force for growth, Singer goes on to argue against specialization in the export of primary commodities. This stands in contradiction to traditional trade theories. The main reasons he gives for this are twofold. Firstly, that it removes the secondary and cumulative effects of investment, since such investment usually comes in the form of foreign capital, which is not reinvested. And, secondly, that it diverts investment in developing countries to areas where the scope for technical progress is limited and worsening terms of trade prevail (Singer 1950: 477).

For Singer and Prebisch, the root cause of the terms of trade deterioration is the fact that the increase in productivity of manufactures in the North raises the income of producers, as opposed to lowering prices, which would have benefited consumers. On the other hand, productivity improvements in primary commodity production, benefits consumers through lower prices. Thus, developed countries benefit, both as consequence of higher income and lower prices, while no such benefits are reaped by the developing countries. In Kaleckian or Hicksian terms, the essential point is that the North sells in an oligopolist fix-price market, while the South operates within a flexi-price one. This results in a logical asymmetry in favour of the former (see Spraos 1983 for a critical review of this). The demand elasticity for food, and negative impact of technical progress on raw material demand further aggravates the problem (Singer 1950: 478–79, Prebisch 1962: 4–6).

In addition to the elasticity argument noted above, Singer and Prebisch have also emphasized the adverse impact of cyclical commodity prices. Singer argues that the developing countries could face a dilemma in that they could fail 'to industrialize in boom because things are as good as they are, and [they fail] to industrialize in a slump because things are as bad as they are' (Singer 1950: 482). Similarly, Prebisch explains how the business cycles, which arise from imbalance between aggregate demand and supply in the centre, could pass to the primary producers of the periphery, and so result in ups and downs in prices. Thus, Prebisch argues that rising profits are bound in with economic upswing in the center. This has the effect of curtailing excess demand by increasing prices, with the opposite effect taking place in a downswing. Thus, as prices rise, profits are transferred to the South, since prices of primary commodities tend to rise more sharply than those of finished goods. However, the prices of primary products 'fall more in downswing, so that in the course of the cycles the gap between prices of the two [finished and primary] is progressively widened'. The main reason which Prebisch gives for this fact being the downward inflexibility of wages, for organizational reasons, as well as profit, arising from profit earners prominent role in production in the North.¹⁹ Demonstrating the deterioration in terms of trade of the South using priceindex data, Prebisch maintains a similar view to that of Singer (Prebisch 1962: 5-7).

Patnaik (1996) notes that the disadvantage of specializing in primary commodity exports is usually assessed based on an analysis of demand side or market structure. However, he argues that the problem can also be seen from the supply side. Thus, Patnaik constructs a model in Kaleckian line of supply-constrained agriculture as well as a demand-constrained manufacturing sector. Under such a framework, exports may be seen in terms of external demand for primary commodities and a source of finance for the import of manufactured goods. Such trade, even assuming the small country assumption, non-diminishing returns, and a balanced trade, entails a decline in demand for the domestic manufacturing sector and, hence, output and employment, for a given level of real wages, or, indeed, vice-versa. This de-industrialization will affect agriculture retrogressively, and this, in turn, affects the manufacturing sector in the familiar Kaleckian way of bringing about a shortage of wage goods.²⁰

2.3.2.3 The Prebisch-Singer hypothesis and empirical evidence

The hypothesis of a secular deterioration in the terms of trade of primary commodities *vis-à-vis* manufactured goods, termed the *Prebisch-Singer hypothesis*,²¹ has generated a number of empirical studies. The debate relating to these studies is summarized in Sarkar (1986). Based on this review, Sarkar basically concludes that the Prebisch-Singer hypothesis is valid (Sarkar 1986).²² Similarly, debates about the quality of data problem would appear to be resolved²³ following reconstruction of the 'G-Y series' data, upon which recent work has been based, by Grilli and Yang themselves (Grilli & Yang 1988). Indeed, based on this reconstruction, Grilli and Yang confirm the validity of the Prebisch-Singer hypothesis, although at a smaller magnitude than originally claimed by Prebisch (1962).²⁴

Recently, with developments in time-series econometrics, there has been renewed interest to re-examine the Prebisch-Singer hypothesis. Thus, Cuddington and Urzua (1989) and Cuddington (1992) have reexamined the hypothesis using modern time series econometric techniques. More specifically, they have questioning the earlier trend stationary approach, conducted without a formal stationarity test. Claiming to find non-stationarity in the G-Y series, they concluded that 'secular decline' in the price of primary commodities should not be interpreted as a 'stylized fact'. However, neither of these studies challenges the underlying theory, nor, indeed the statistical validity of this theory. In re-examining Cuddington and Urzua's work, Ardeni and Wright (1992) highlight three methodological problems. Firstly, Ardeni and Wright question the use of a simple correlogram of the series as a guide for their specification. Secondly, they note that Cuddington and Urzua ignored the low power of the Dickey-Fuller test, in the case of closer alternatives. And, thirdly, they highlight a number of problems relating to the use of the dummy. Perhaps more importantly, they also cast doubt over the non-stationarity assumption of the series upon which Cuddington and Urzua's analysis is based. However, use of a structural time series approach to overcome the above problems would appear to lend support to the validity of the hypothesis (Ardeni & Wright 1992). Helg (1990) also reports on attempts to apply the new time series analysis to the reconstructed G-Y series. Sarkar (1992) and Sapsford and Balasubramanyam (1994) have also critically reviewed recent empirical debates and conclude that the Prebisch-Singer hypothesis is still valid. Indeed, Sarkar (1992) and Reinhart and Wickham (1994) note that the deterioration is not simply the effect of a temporary shock but a rather a deterministic one. Thus, Sarkar (1992) concludes, 'the secular decline of the terms of trade of primary commodities is not a myth but a reality'.

2.3.3 Quantitative Models of Trade in Primary Commodities

The preoccupation of the structuralist school with trade in primary commodities, and, in particular, the empirical studies discussed in the previous section, underscore the importance of understanding the functioning of commodity markets. This is hardly surprising, since (as discussed in Chapter 1), for many countries in Africa the role of commodities is central. An insight into such markets may be obtained from a discussion of both theoretical and practical modelling of commodities. With this objective in mind, this section will examine the underlying theory behind commodity modelling, and present some recent examples of commodity models.

2.3.3.1 Structure of theoretical primary commodity models

Based on the works of, *inter alia*, Labys and Pollak (1984), Labys (1978), Behrman (1978), Adams (1978) and Guvenen et al. (1991), three broad classes of commodity models may be identified: firstly, *econometric* models (which include 'market' models, 'process' models and 'system dynamics and optimization' models); secondly, *spatial equilibrium and programming* models; and finally, *input-output* models. We begin by examining econometric models, before moving on to describe the other two broad classes of models.

(a) Econometric models

Perhaps the most basic econometric model is the 'market' model. This model describes a competitive market, in which the demand for, and the supply of, commodities, including inventories, interact. Equilibrium is arrived at through price clearing. The approach is based fundamentally on an assumption of perfect competition (Labys & Pollak 1984, Labys 1978, Behrman 1978, Adams 1978). An important subset of market models are 'process' models. The focus of these is on supply and demand within a particular industry, rather than across markets. In the process model, the demand for a commodity is a derived demand from the product within the production process (see Labys & Pollak 1984, Labys 1978). Thus, it could be argued that process models focus mainly on demand for a particular commodity.

In sharp contrast, system dynamics models (SDMs) focus mainly on the supply side of the commodity in question. In SDMs, adjustment of actual levels of inventory, production and consumption of a commodity, towards a desired level is central. In these models, the level of stock adjusts to the differences between rates of production and consumption. Such stock is usually given as a certain proportion of expected consumption. The latter, in turn, is determined as a certain proportion of the desired stock. This formulation, termed 'relative stock coverage', provides a mechanism for the determination of prices for the particular commodity in question. This price, in turn, will determine the desired level of production capacity, which forms the basis for actual production, usually through an adaptive expectation scheme (see Labys & Pollak 1984: 61–63 for formal treatment).

The third important type of econometric model are 'optimization' models. In contrast to the other varieties of model, optimization models focus on the structure of markets. This has the effect of shifting the emphasis away from perfect competition towards an oligopolistic market structure. This, when taken in combination with the assumptions underlying the economics of exhaustible resources, will tend to imply a backward-sloping supply curve. Such models are usually employed in the analysis of cartel behaviour and are given in an optimization equation which maximizes the sum of discounted profits, with the aim of picking the price trajectory of the commodity in question (Labys & Pollak 1984, Behrman 1978).

(b) Spatial equilibrium and programming models (SEPMs)

'Spatial equilibrium and programming models' represent a further branch of commodity modelling, widely cited in the literature. The major objective of such models is to determine the spatial flows of demand and supply, as well as to identify equilibrium conditions, by explicitly considering the configuration of transport networks. These are usually formulated as a mathematical program with the aim of maximizing sectoral/regional revenues or minimizing costs. In other words, the model operates in such a way that transportation costs are minimized, with commodities left free to transfer between markets until demand equals supply in every spatially-separated market (Labys & Pollak 1984). Depending on the degree of complexity, such models may take either a linear, quadratic, recursive or mixed form. The solution for the objective function of the model, given its particular constraint, is arrived at using mathematical programming techniques, such as Simplex.

(c) Input-output models

Input-output models are based on Leontief's pioneering work on interindustry relations. Although these are not commodity-market models, as such, nevertheless they may be used to study how the demand and supply of commodities relates to inter-industry structure and macroeconomic variables at a national economy level (Labys 1978, Labys & Pollak 1984). Though static, the technical coefficients specified in input-output models can help to determine the demand and supply for a commodity. It is worth pointing out, however, that the input-output model is a demand-driven model, such that it may not always be helpful in explaining supply determination. Moreover, since technical progress is a central feature of commodity markets, the fact that this model is static, represents its major weakness.

To sum up, the basic methodologies outlined above form the basis for most commodity models. However, the above description represents an enormous simplification upon reality, since, in practice, modellers may use a mixed bag of overlapping methodologies. However, this description is sufficient to give an idea of the underlying theory upon which most commodity modelling work is based. Further, based on this description, one might conclude that recent developments in commodity modelling have been promising (see Guvenen et al. 1991). In general, most commodity modelling and non-linear programming now takes account of risk and uncertainty, intertemporal linkages and adjustment processes. Most of these also attempt to capture the structure of the market, especially market imperfections, address issues of data quality, incorporating new techniques for estimation, particularly in relation to time series data and, finally, link in with other macro and policy models (Guvenen et al. 1991). These theoretical discussions, combined with an overview of some recent models, are used to arrive at the approach employed in Chapters 4 to 6.

2.3.3.2 A brief overview of recent commodity models of 'the South'

Having set out some background on the methodology of modelling, this section will examine some recent commodity models that emphasize North-South trade in primary commodities. The intention is not so much to exhaustively examine all recent models, but rather to focus on those modelling approaches, widely cited in the literature, and deemed relevant in informing the commodity modelling pursued later in this book. We examine six such approaches here, beginning with a model developed by Bond (1987).

(a) M.E. Bond (1987) and Goldstein & Khan (1978)

Bond's (1987) work follows Goldstein and Khan (1978). I have used her work because it is relatively recent and also representative of other work, using this approach. In Bond's model, the demand for primary commodities depends on a number of factors. Specifically, economic activity within the industrial countries, the price and composition of commodities, the geographical location of the exporting nation, and, finally, the trade and agricultural policies pursued by industrial countries. Supply, on the other hand, is seen as depending on weather conditions, factor endowment (together with investment), relative prices, technology, domestic market characteristics and population growth.

The author emphasizes the need to specify demand and supply equations for different commodity groups as well as regions of the developing world. Thus, five commodity groups are identified for the study, namely, food, beverages and tobacco, agricultural raw materials, minerals and energy. These classifications of commodity supply and demand equations are set out for Africa, Asia, the Middle East, Europe and the Western hemisphere. Adjustment in export demand to changing market conditions is assumed to occur within a period of one year, while supply is allowed to have delayed adjustment possibilities, due to production lag. The four major commodity groups, with the exception of energy, are modelled as follows.

Export demand

The world demand for exports of commodity k from developing country region R is specified in log-linear form as follows,

$$\ln XD_R^K = a_0 + a_1 \ln\left(\frac{PX_R^K}{PW^k}\right) + a_2 \ln YW$$
[1]

where: XD_{R}^{K} is the quantity of exports of commodity k demanded from region R, PX_{R}^{K} the export price of commodity X from region R, PW^{K} the average price of commodity K in the international markets and YW the real income in importing countries.

It is noted that, owing to the logarithmic specification, a_1 is the (relative) elasticity of world demand for region *R*'s export of the k^{th} commodity with respect to the divergence between region *R*'s export price of the k^{th} commodity and the average world price. a_2 is the income elasticity of demand. The former is expected to be negative while the latter positive. It is implicitly assumed that there is no, or constant, elasticity of substitution.

Export supply

The supply of exports of commodity k from region $R(XS^{K}_{R})$ is specified as a log-linear function of the current and lagged ratios of the export price of commodity $k(PX^{K}_{R})$ to domestic price level in the producing countries in region $R(P_{R})$, multiplied by the exchange rate of currencies of the producing countries, i.e., US\$ per unit of local currency (E_{R}) , an index of productive capacity in region $R(Y_{R})$, supply shocks (SS_{R}) and trend value, t:

$$\ln XS_R^K = \beta_0 + \beta_1 \ln\left(\frac{PX_R^K}{P_R E_R}\right) + \beta_2 \ln\left(\frac{PX_{R-1}^K}{P_{R-1} E_{R-1}}\right) + \beta_3 \ln \overline{Y}_R + \beta_4 \ln SS_R + \beta_5 t \qquad [2]$$

This equation allows for a positive relationship between supply and export price, relative to domestic price. It also allows for the operation of a lag supply response.²⁵ Normalizing this equation for the price of exports in region R yields the following equation, which together with equation [1] can be estimated simultaneously to obtain the estimates of the structural parameters.

$$\ln PX_{R}^{K} = b_{0} + b_{1}XS_{R}^{K} + b_{2}\ln P_{R}E_{R} + b_{3}\ln\left(\frac{PX_{R-1}^{K}}{P_{R-1}E_{r-1}}\right) + b_{4}\ln\overline{Y}_{R} + b_{5}\ln SS_{R} + b_{6}t$$
[3]

Finally, the equations for energy exports are treated differently. This follows two steps, first the world demand for energy is specified assuming that the supply is exogenously fixed $(XS^{en}_{W} = XS^{en}_{W} \text{ bar})$. Then the world demand for energy (XD^{en}_{W}) is allocated across exporting developing regions with reference to trend term. The world demand is

taken as a function of energy price (PW^{en}) relative to world price (PW) and world income (YW) as,

$$\ln XD_{W}^{en} = \alpha_{0} + \alpha_{1} \ln\left(\frac{PW^{en}}{PW}\right) + \alpha_{2} \ln YW$$
[4]

World energy demand is allocated using the region's share in the world energy market and a trend variable given by,

$$XD_R^{en} = w_R X D_w^{en} e^t$$
^[5]

By substituting equation [4] into [5], the demand for energy exports from region R is given by,

$$\ln XD_R^{en} = \alpha_0 + \alpha_1 \ln\left(\frac{PW^{en}}{PW}\right) + \alpha_2 \ln YW + \alpha_3 t$$
 [6]

Equation [6], combined with the fixed regional supply of energy (arising from the oligopolistic pricing practices of OPEC member states), determines the estimating equation. The author undertakes empirical investigation based on the set of equations specified above.

Some notes on the initial model

The original model set out by Goldstein and Khan (1978) is divided into two parts: firstly, an 'equilibrium' model, in which no lag is assumed, and adjustment is deemed as taking place instantaneously; and secondly, a 'disequilibrium' model, in which the above assumption is relaxed. In Bond's work, discussed above, the first of these models is employed.²⁶ I will now discuss the disequilibrium model, in order that these two approaches may be brought into perspective.

In order to introduce the disequilibrium mechanism Goldstein and Khan (1978) use the method adopted by Taylor and Houthakker (1970, cited in Goldstein & Khan 1978). Thus, they assume that exports (X) adjust to the difference between demand for exports in period t and the actual flow in the previous period, which is given as,

$$\Delta \ln X_t = \gamma \ (\ln X_t^d - \ln X_{t-1}) \tag{A}$$

where: γ is the coefficient of adjustment, *d* demand, *t* time and Δ denotes change. By substituting equation [1] into [A] we can obtain an equation for estimating exports (minor symbolic difference between the

two papers is ignored; I have used symbols as reported in the earlier model of Goldstein & Khan 1978):

$$\ln X_{t} = c_{0} + c_{1} \ln \left(\frac{PX}{PXW}\right) + c_{2} \ln YW_{t} + c_{3} \ln X_{t-1}$$
[B]

where: $c_0 = \gamma a_0$ $c_1 = \gamma a_1$ $c_2 = \gamma a_2$ $c_3 = 1 - \gamma$

The mean time lag in the adjustment of exports is equal to γ^{-1} and may be calculated from parameters of equation [B] as $(1-c_3)^{-1}$. Since the quantity of exports is specified as adjusting to excess demand, the price of exports adjusts to conditions of excess supply:

$$\Delta \ln PX_t = \lambda \left(\ln X_t - \ln X_t^s \right), \ \lambda > 0$$
[C]

Substituting equation [2], taking note of the footnote remark under equation [2], into equation [C] and solving for PX_t we get

$$\ln PX_{t} = d_{0} + d_{1} \ln X_{t} + d_{2} \ln P_{t} + d_{3}Y_{t}^{*} + d_{4} \ln PX_{t-1}$$
[D]

where:

$$d_0 = \frac{-\lambda\beta_0}{1+\lambda\beta_1} \quad d_1 = \frac{\lambda}{1+\lambda\beta_1} \quad d_2 = \frac{\lambda\beta_1}{1+\lambda\beta_1}$$

$$d_3 = \frac{-\lambda\beta_2}{1+\lambda\beta_1} \quad d_4 = \frac{1}{1+\lambda\beta_1}$$

The reduced form equations obtained from equations [B] and [D] are:

$$\ln X_{t} = \frac{c_{0} + c_{1}d_{0}}{D'} - \frac{c_{1}}{D'}\ln PXW + \frac{c_{2}}{D'}\ln YW + \frac{c_{1}d_{3}}{D'}Y_{t}^{*}$$
[E]

$$+ \frac{c_1 d_2}{D'} \ln P_t + \frac{d_4}{D'} \ln P X_{t-1} + \frac{c_3}{D'} \ln X_{t-1}$$

$$\ln PX = \frac{d_0 + d_1 d_0}{D'} - \frac{c_1 d_1}{D'} \ln PX W_t + \frac{c_2 d_1}{D'} \ln Y W_t \qquad [F]$$
$$+ \frac{d_3}{D'} Y_t^* + \frac{d_2}{D'} \ln P_t + \frac{d_4}{D'} \ln PX_{t-1} + \frac{c_3 d_1}{D'} \ln X_{t-1}$$

where $D' = 1 - c_1 d_1$.

After developing these two forms of the model – taking an equilibrium and disequilibrium form – the reduced equations are then estimated using the 'Full-Information Maximum Likelihood' (FIML) approach (see Chapter 6 for further details).

Bond (1987) and Goldstein & Khan's (1978) models are interesting in that they recognize the importance of various factors in modelling trade in primary commodities. The author's emphasis on the use of functions specified by product type and region, and the method of solving this, simultaneously, are both worth noting. Moreover, Goldstein and Khan's distinction between equilibrium and disequilibrium models is its strength. However, when it comes to applying this model, they fail to specify all the factors that were previously mentioned as likely to affect the market. Indeed, even the suggested classification between products is not reflected in the fully specified model, except in the case of energy. More importantly, the trade sector in their model is seen as an enclave, which does not affect, and is not affected by the other sectors of the economy. This problem is addressed in this book.

(b) Ke-Young Chu and T.K. Morrison (1986)

Previously, Chu and Morrison had analysed the demand-side factors underlying the short-run fluctuation of non-oil primary commodity prices (see Chu & Morrison 1984).²⁷ In the study summarized below (i.e., Chu & Morrison 1986), a model with a medium-term framework, in which supply-price dynamics²⁸ plays a significant role is specified. Moreover, the role of supply shocks in the short-run fluctuation of commodity prices was also underscored.

The authors developed their model following a moderately-sized survey of the literature on supply-price dynamics. We discuss this survey, in detail, in Chapter 4. In developing this model, they show how the simple 'cobweb theorem' of Ezekiel (1938) has evolved into Ner-

66

love's (1958) model (Chu & Morrison 1986: 140–44). Chu and Morrison begin by making a number of assumptions. World non-oil primary commodity markets are assumed to be price clearing. Supplies are considered to be based on profit maximization, and thus are affected by supply shocks. Demand is assumed to depend on world economic activity and relative price of commodities. Supply may change as a result either of changes in medium- and long-run supply or in short- and medium-term utilization.

Supply determination

Thus, in Chu and Morrison's model, production is specified, in the short run, as a function of prices, supply shocks and potential production, as follows:

$$q_t^s = u_t + qc_t \tag{[1]}$$

$$u_{t} = \alpha_{0} + \alpha_{1} rps_{t} + \alpha_{2} rps_{t-1} - \alpha_{3} SS_{t}$$
^[2]

$$rps_t = p_t + es_t - ps_t$$
[3]

where, in logarithms,

 q_{t}^{s} = world production of a commodity

- u_t = utilization of potential production
- qc_t = potential production
- p_t = output price (the international price of a commodity, in US\$)
- es_t = exchange rates (relative to US\$) of the currencies of exporting countries
- ps_t = domestic price levels in exporting countries
- rps_t = real price of commodities, as defined in equation [3], approximated by output to domestic price
- ss_t = supply shocks (resulting, say, from weather conditions in food production and approximated by stochastic error term).

The first equation defines world production of the commodity. The second specifies the utilization ratio, as a function of a distributed lag of real output prices faced by producers and as a consequence of supply shocks. And finally, the third equation defines real prices faced by producers. The change in potential production (supply) is specified as a function of average excess profits in recent years.

$$\Delta qct = \beta_0 + \beta_1^{k-1} \sum_{i=1}^{k} erps_{t-1}$$

$$erps = rps_t - \overline{rps}$$
[4]

where: rps (bar) is the long-run average of rps_t , $erps_t$ is the excess profit and k is a parameter (k>0).

Equations [4] and [5] specify the change in potential production as a function of the average excess profits. (In other words, the average real price faced by producers in excess of their long-run average in recent years.) Average excess profit, as defined in equation [5], may cause an increase in potential production (as given in equation 4) either through expansion of production or entry of new firms. This increase in potential production may result from excess profits arising from the distant past, for example, in the case of the maturation of newly planted perennial crops. Alternatively, such excess profits may have arisen in the immediate past, as in the case of the reactivation of existing trees. Based on the above, the entire system of equations for *change* is given as

Supply

$$\Delta q_i^s = \Delta u_i + \delta q c_i \tag{[1a]}$$

$$\Delta u_{t} = \alpha_{0} + \alpha_{1} \Delta r p s_{t} + \alpha_{2} \Delta r p s_{t-1} - \alpha_{3} S S_{t}$$
^[2a]

$$\Delta rps_{t} = \Delta p_{t} + \Delta es_{t} - \Delta ps_{t}$$
[3a]

$$\Delta qc_{t} = \beta_{0} + \beta_{1} k^{-1} \sum_{i=1}^{\kappa} erps_{t-i}$$
[4]

$$erps = rps - \overline{rps}$$
 [5]

Demand

$$\Delta q_i^d = \gamma_0 - \gamma_1 \Delta r p d_i + \gamma_2 \Delta y_i$$
[6]

$$\Delta rpd_{i} = \Delta p_{i} + \Delta ed_{i} - \Delta pd_{i}$$
^[7]

Equilibrium condition

$$\Delta q_t^s = \Delta q_t^d \tag{8}$$

where:

 q_t^d = quantity demanded ed_t = exchange rates of importing countries (in relation to the US\$) y_t = industrial production (economic activity) in importing countries pd_t = domestic price of importing countries; *r* stands for real.

Equations [6] and [7] are a simplified form of the demand equation given in Chu and Morrison (1984).

The above system of equations explains both the short-run and medium-term fluctuation in prices. It also captures the dynamic interaction between supply and price, over a number of years, where prices affect potential production. With potential production taken as exogenous, equations [1a] to [3a] and [6] to [8] capture the short-run determination of prices. Using this general approach, the authors modify the basic model, using relevant assumptions in relation to the different commodity groups selected. For instance, lagged prices are assumed to play a strong role in determining the supply function for food, while prices of industrial raw materials are seen as being determined mostly by shifts in demand. The study concludes that potential production responds to medium-term fluctuation in prices, while the utilization rate responds to short-run fluctuations. Industrial production and exchange rate fluctuations are also found to have been major factors underlying commodity price fluctuations, since the early 1970s.

The strength of this model lies in its attempt to explicitly capture supply price dynamics for primary commodity exports. It also brings into the picture the role of exchange rates, in affecting primary commodity markets. At a specification level, the nature of each primary commodity also plays a role. Although not explicitly an objective of this modelling exercise, the impact of this sector on the rest of the economy, and vice versa, is relatively neglected. Such an approach is tantamount to setting the function of the utilization rate (say equation 2a) and insulating it from the repercussions of, say, financial flows.

(c) Ramanujam and Vines (1990): A structural rational expectation model

This model is applied to four commodity groups, namely, food, beverage, agricultural raw materials and metals.²⁹ Unlike the previous models, this and Hwa's (1985) model emphasize the importance of commodity stockholding. Indeed, it is this aspect of the model, which serves as justification for their inclusion in this review.

Ramanujam and Vines (1990) have developed supply, demand and price equations incorporating stockholding identities. Their model assumes that commodity stockholders who are seen as having forward-looking – rational or model-consistent – expectations willingly hold stocks. The assumption in relation to forward-looking expectations is justified on the grounds that speculators are assumed to equate the risk-adjusted yields on commodities and on financial assets. Indeed, the theoretical model, based on the works of Adams and Behrman (1976) and Labys (1978),³⁰ also assumes that there is an instantaneous market clearing. This may be formally specified by the following set of equations:

Supply

$$Q_t = Q_t \left(P_t , \sum_{i=1}^t P_{t-i} , Y \right)$$
[1]

where supply is assumed to be a function of current price P_t , (lagged) series of past prices, due to costs of adjusting output, as well as other exogenous variables, Y.

Demand

$$C_{t} = C_{t} \left(P_{t} , \sum_{i=1}^{m} P_{t-m} , X \right)$$
[2]

where demand is assumed to be a function of current price P_t , a short series of past prices, due to habit, persistent, and other exogenous variables, X.

Price

$$P_{t} = P_{t} \left(P_{t+1_{t}}^{*}, f(H_{t}), Z_{t} \right)$$
[3]

The price equation is an inverted stock demand function. It is a function of expected price $(p^e_{t+1},t)^{31}$ and either the change in stock or the level of stock (*H*) together with other exogenous variables (*Z*) particularly the interest rate.

Stock identity

$$H_t = H_{t-1} + Q_t - C_t$$

The vectors of exogenous variables X, Y and Z vary between the four different commodity categories. The price variable is taken as the real price of the commodity group, with the deflator being the unit value of manufactured exports. All the variables, except for the interest rate (which is used as a proxy for the opportunity cost of holding stocks), as well as the trend term, are expressed as a natural logarithm. The estimation is predominantly by OLS, and in some cases, also by IV (instrumental variables).

The four supply equations are estimated after tests about stationary and co-integration have been made (see pp. 511-12). Thus, for 'food', estimates are based on current price, five-year lagged price, lagged stock and price of fertilizer (as representing cost of production). For 'beverage', short-term price effect (with a two-year lag) and long-term gestation period price effect (with 4 to 8-year lag combined price). price of fertilizer and lagged stock holding are all used as explanatory variables. For 'agricultural raw materials' both current and lagged (5year) prices are used, since this category also covers perennial crops. In relation to 'processing costs', lagged oil prices are used. Stock holdings are also used in this model, but found not to be significant. The 'metals and minerals' supply function is estimated using current and lagged (4-8 years) prices, as well as the difference between the current and the average price, in order to reflect the cost of extraction, which increases according to the amount extracted. Finally, stock holding and interest rates have been used, with the aim of reflecting capital cost.

The authors found it difficult to distinguish between demand by households, for direct consumption, and demand by firms, for further processing. Hence, the demand function for each commodity group is postulated to be a composite one. Demand is assumed to be directly related to income (taken as the weighted real GDP of the industrial countries) and inversely related to prices. Prices of substitutes are not considered. For food, lagged prices with a weighted exchange rate be-

[4]

tween OECD currencies and US dollars were also considered. Finally, price is estimated using an equation derived from models with costs, including handling costs such as storage, as well as interest rates (representing the opportunity cost of holding financial assets) adjusting to the commodity holding structure. The model is closed by estimating the stock identity using a logarithmic form (equation 4) in order that this might be comparable with the other equations of the model, which are also in logarithmic form. The full model, using the single equation estimations, may then be solved using a forward-looking model.

By estimating the model using different groups of primary commodities within a forward-looking market, this model brings a new insight into the modelling of commodities. The strength of this model is its effort to make each of commodity groups' specifications unique by allowing for variation in the exogenous factor, as well as through the incorporation of lags. In common with most models, the model assumes instantaneous market clearing. Moreover, in defining the real price it is not clear why the current price is deflated by unit value of manufactured exports. Further, it could be argued that the use of rough proxies (such as cost of fertilizer, for food and beverage, and cost of oil, for raw materials) rather than the actual cost of imported inputs as intermediate costs may not accurately describe the situation of, for example, smallholder producers in Africa. Similarly, on the demand side, although there is scope to disaggregate existing data, in fact, no attempts have been made to undertake such an exercise. Thus, the stock holding equation, which describes a crucial relationship within the model, doesn't actually provide information on who holds such stocks. and how the data has been generated. And, in spite of the authors' finding that the series are non-stationary, most estimations are in fact, undertaken in the traditional way, without any discussion as to whether these are derived from a co-integration vector or an ECM model. Finally, as with previous models this model does not connect the commodity market with the rest of the economy.

(d) E. Hwa (1985)

Hwa's model is a further variant on recent models having a disequilibrium structure. The model is based on two broad assumptions: firstly, that quantity adjustments may be relatively 'sticky', as a consequence of supply lag, and, hence, that the adjustment mechanism lies in price; secondly, as a disequilibrium model, it assumes away instantaneous market clearing. However, the modelling begins from an equilibrium model of storable primary commodities given by

$$c_i = f_1(p, xc_i) + u_i$$
 [1]

$$q_i = f_2(p_i, xq_i) + v_i$$
 [2]

$$hd_t = f_3(p_t, xh_t) = w_t$$
[3]

$$h_{t} = h_{t-1} + q_{t} - c_{t} - x_{t}$$
[4]

$$hd_i = h_i \tag{5}$$

The first three equations describe consumption (c), production (q) and stock demand (hd), respectively. These are postulated to be a function of real price (p) and other relevant predetermined values (xc, xq, xh). The fourth equation describes stock supply (h), in which x represents intervention by a buffer stock agency or national government. The last equation is an equilibrium (stock demand and supply) condition.

Within the model, in order to explicitly consider price adjustments, a disequilibrium condition in price and quantity is introduced. Thus, if functions f_i , i=1,2,3 are linear, the following relation will hold true,

$$P_t^e - Pt = (f_1' + f_2' + f_3')^{-1} (hd_t - h_t)$$
^[6]

 $P^{e_{t}}$ is the equilibrium price that can be derived from [5], the \dot{f}_{i} 's are the partial derivative (with respect to price) of *c*, *q* and *hd*, respectively. This equation shows that the actual (observed) price is not equal to the equilibrium price, unless the market is in equilibrium. (The same can be said about stock demand and supply.) The important question is how equilibrium may be attained from such an initial disequilibrium condition. Although, theoretically, this may be attained through price and quantity adjustment, the latter is perceived not to hold in the short run. Thus, Hwa assumes that prices will gradually adjust towards their equilibrium value. This is approximated by a distributed lag function [*T*(*L*)] in which *L* is a lag operator.

$$p_t = \omega(L)p_t^e \tag{7}$$

Believing that equation [6], rather than [5], is, in general, the one that holds and inserting [7] into [6], yields the following dynamic price equation:

$$P_{t} = \phi \left[\frac{1}{\omega(L)} - 1 \right]^{-1} (hd_{t} - h_{t})$$
[8]

where: $\phi = (f_1' + f_2' + f_3')^{-1} (hd_t - h_t), \quad \phi > 0$

Hwa admits that there can be as many dynamic price equations as there are hypotheses about the distributed lag function. In his study two hypotheses which can be handled with annual data are discussed. The first [9a] is a geometric lag hypothesis, which implies a monotonic increasing or decreasing path towards the equilibrium price. The second [9b] implies oscillations around the equilibrium price.

$$\omega(L) = \frac{1-\lambda}{1-\lambda L} , \quad 0 < \lambda < 1$$
 [9a]

$$\omega(L) = \frac{1 - \lambda_1 - \lambda_2}{1 - \lambda_1 L - \lambda_2 L^2} , \qquad [9b]$$

$$\lambda_1 + \lambda_2 < 1, \quad \lambda_2 - \lambda_1 < 1 - -1 < \lambda_2 < 1$$

Substituting [9a] and [9b] into [8], respectively, yields,

$$Dp_{t} = \frac{\phi(1-\lambda)}{\lambda} (hd_{t} - h_{t})$$
[10a]

$$Dp_{t} = \frac{\lambda 2}{\lambda_{1} + \lambda_{2}} DP_{t-1} + \frac{\phi (1 - \lambda_{1} - \lambda_{2})}{\lambda} (hd_{t} - h_{t})$$
[10b]

where: $Dp_t = P_t - P_{t-1}$

Both [10a] and [10b] show that price changes are a positive function of excess demand for stock. Note that the, λ 's dictate both the size and speed of adjustment. (For instance, in [10a], if λ approaches 0, the price adjustment would be an instantaneous one, similar to the equilibrium system.)

In general, the competitive disequilibrium model consists of equations [1]–[4] and either [10a] or [10b]. In this model the voluntary accumulation of stocks $(h_t - h_{t-1})$ is not the same as the actual accumulation of stock $(h_t - h_{t-1})$. The involuntary part being $(h_t - hd_t)$. This can be given as $h_{i} - h_{i-1} = (h_{i} - hd_{i}) + (hd_{i} - h_{i-1})$

It is noted that the observed accumulation will be equal to voluntary accumulation of stock, only if the market is in equilibrium (i.e., $h_t - hd_t = 0$). Hwa states that, from an estimation point of view the demand for stock equation may be estimated by inserting [3] into either [10a] or [10b]. By estimating the demand for stock equation as a function of consumption (i.e., transaction demand for stocks), the expected price spread and interest costs and risk premium (i.e., speculative demand) and inserting this into [10a] and [10b], the price adjustment equations for a disequilibrium market may be estimated. For an equilibrium market, this demand equation may be directly estimated (see Hwa 1985: 309–11).

Finally, the models are estimated using OLS and IV methods, with annual data for six primary commodity groups. Following estimation, a dynamic simulation is undertaken using the full (simultaneous) model. It is reported that the model traces endogenous variables fairly well. Besides, when the model is 'shocked' (via exogenous variables) the commonly observed boom-and-bust-cycles are observed.

In keeping with Ramanujam and Vines model, the basic structure of this model is similar to that of Adams and Behrman (1976). However, the model represents a step forward in the modelling of primary commodities, since its focus is on disequilibrium structure. Moreover, the stock demand and supply function, which is a key relationship in the model, is specified realistically by taking into account a number of conditions of the financial market. More specifically, the speculative demand for stocks is assumed to be at a level where the expected price spread equals the storage cost plus the interest cost and risk premium. One weakness of the model is the use of identical specification for all commodity groups. Although the disequilibrium structure is realistic, eventually it will be price, rather than quantity, or indeed both, that will adjust. This effectively makes it a short-run model and so limits its generalizability. A further major weakness of the model is its failure to accommodate the role of supply factors in determining the equilibrium path. This is further aggravated by the structure of the model, in which supply effects are buried within stock demand³² and supply balance. which are implicitly held constant. This has the consequence of allowing price to adjust, instead, as per equations [10a] and [10b]. Finally, as with most models, the repercussions on, and feedback from, the wider economy are not considered.

[11]

(e) CGE model of Dick et al. (1983)

This and subsequent models are based on a computable general equilibrium (CGE) framework, with export share matrices. This makes them somewhat different from those discussed above. Dick et al.'s (1983) model is designed to study the short-run impact of fluctuating primary commodity prices on three economies. As the authors point out, most commodity models suffer from 'casual empiricism' failing to explain the underlying casual mechanism (Dick et al. 1983: 405). This has led to the use of simpler econometric models, not only with the aim of overcoming such problems in the literature, but also to rule out the implicit assumption that instability in income is solely the result of export earnings instability. This approach is further developed to include sectoral and economy-wide effects.

Although sufficient allowance is made to accommodate the specificity of each country, the three countries in this study are modelled using a similar multisectoral general equilibrium approach. The use of a similar macro-modelling framework for all of the countries allows one to attribute the response to exogenous shocks to the specific nature of each country rather than to differences in model specification. The analytical framework adopted runs as follows. Producers are assumed to minimize the cost of producing a given activity level, subject to a production function³³ of a three-level, or nested, form. The average household is assumed to maximize its utility, subject to a function describing substitution prospects in consumption between imports and domestic products as well as an aggregate budget constraint. The solution to this constrained optimization results in a system of commodity-demand equations for production, capital creation and consumption, as well as a system of factor demand equations.

Government demand is linked to aggregate domestic consumption. Commodity export demand is linked to world demand facing a particular commodity and to world export price. A system of demand and supply equations ensures market-clearing equilibrium for domestic commodities, occupational labour, fixed capital and land. Finally, a host of miscellaneous identities and/or definitions are used. In a summarized form the model is framed as follows.

Output is given by:

 $Y = Y(F_1, F_2, F_3)$

[1]

Factor marginal product, for each of *i* factors

| dY | _ | <u>ri</u> | [2-4] |
|-----|----|-----------|-------|
| dFi | Fi | Р | L7 |

Income

$$Y = ABS + B$$
^[5]

Trade balance

$$B = P^e X - P^m M \tag{6}$$

Imports

$$M = \left(ABS, \frac{P^e}{P}\right)$$
[7]

Exports

$$X = X\left(\frac{P^{*}}{P}\right)$$
[8]

<u>Numerior</u> P = 1

where:

Y is the real output and F_i are aggregate factor inputs (1 = labour, 2 = capital and 3 = land).

Thus, based on the above, competitive behaviour implies [2]–[4], in which the marginal factor $i (dY/dF_i)$ is equated to its real factor price $(r_i$ is factor *i*'s money price and *P* the price of output). Equation [5] shows national income identity, where *ABS* is real domestic absorption (aggregate real consumption, investment and government expenditure) and *B* the balance of trade. *B* is defined in equation [6] as the difference between foreign currency value of exports (P^eX) and imports (P^mM) . Commodity import demand (M) in [7] is defined as a function of *ABS* and the domestic price level relative to the world price of imports. Finally, equation [8] specifies export demand.

The model is solved, first, by converting the equations to linear form by logarithmic differentiation. Since the number of variables (14)

is greater than the number of equations (9), five variables are exogenously set, to close the system. These exogenous variables are selected to reflect the short-run nature of the model (F_2 , capital; F_3 , land; and the money price of labour, r_1) and by targeting domestic absorption (i.e., by making ABS exogenous and B endogenous) or vice versa (i.e., making ABS endogenous with targeted B). Based on this framework the authors proceed to investigate the impact of primary commodity price fluctuations as the only source of fluctuation in export earning. The conclusion arrived at, following a simulation exercise, is that the price shocks may be successfully isolated from affecting the relevant commodity sector by fixing domestic absorption. The problem with this option is that it requires a large sum of foreign exchange. The second conclusion is that a targeted balance of trade may be maintained, but that it requires a change in exchange rate to facilitate a switch in industrial composition. However, as the authors point out, since such adjustment takes time, compensatory financing schemes which extend the time required for restructuring, may be necessary.

This model is a good example of an attempt to relate the external market to the macroeconomic conditions of a particular country. Notwithstanding its merits, the model rests almost entirely on optimization solutions and instantaneous market adjustment, at the expense of differential pricing rules, the possibility of lagged responses and other modelling techniques. A further weakness of the model is a lack of specification plausibility, with government demand, for instance, being related only to total consumption, and with factor mobility within a competitive market assumed. Although the model attempts to explain the impact of fluctuating prices, nevertheless it fails to trace the source of that fluctuation. This fact would explain the authors' presumption that developing countries should adjust. Not surprisingly, the simulation results are identical to the IMF/World Bank policy proposals, in which price adjustments (such as devaluation) are emphasized, at the expense of structural problems related to cyclical markets, supply problems and market bottlenecks.

(f) G.B. Taplin (1973): A Model of World Trade

Taplin's model is one of the earlier models, which emphasizes a wider inter-country/region framework. It is based on a model developed by the Research Department of the International Monetary Fund, known as the 'Expanded World Trade Model (EWTM)', designed for short-term forecasting of trade flows and analysing economic policies. The version of the model surveyed below treats the world as being divided into 27 countries and regions, comprising 25 developed countries, the CEMA (former socialist countries) countries and the Rest of the World (RW). The RW comprises all of the less developed countries. The imports of each developed country are determined by an import function, which takes into account economic activity, relative prices and other relevant factors. As a consequence of lack of information, economic activity and exports are used to determine the imports of the CEMA region. RW's imports are a function of current and past foreign exchange receipts. Each country and region's exports are obtained by distributing forecast imports through an export share matrix.

Consistent sets of imports and exports are obtained using a programme for solving simultaneous non-linear equations by iteration. As with project LINK,³⁴ export share matrices stand at the heart of the system. Based on these points, two version of the model are presented. The first of these deals with total merchandise trade flows, and is termed the 'total trade model'. The second model divides imports into four commodity groups, which are used by Project LINK. Since the two versions of the model are essentially identical, the following discussion focuses on the total trade model, supplemented by comments on the commodity version, when necessary.

The total trade model

The model comprises a total of 84 equations (27 equations each for imports, exports and trade balance) given as,

$$M_i = f_i \left(X_i, \, Z_i \right) \tag{1}$$

$$X_{i} = \sum_{j=1}^{27} a_{ij} \left(\frac{M_{j}}{\delta_{j}} \right)$$
[2]

$$B_i = X_i - M_i \tag{3}$$

In addition,³⁵ three world trade identities are given as

$$M_{28} = \sum_{i=1}^{27} M_i$$
 [4]

$$X_{28} = \sum_{j=1}^{27} X_j$$
 [5]

[6]

$$B_{28} = X_{28} - M_{28}$$

where:

| i,j | {125 | = | the individual developed countries |
|-----|------|---|---|
| | 26 | = | CMEA countries |
| | 27 | Ξ | RW, the developing countries grouped together |
| | 28 | = | total world |
| | | | |

- M_i = total import of region *i*, c.i.f., in current prices in US\$
- X_i = total exports of region *i*, f.o.b. in current prices in US\$
- B_i = merchandise trade balance of region *i*
- Z_i = explanatory factors other than exports, and assumed independent of exports, determining the imports of country *i*.
- i = the f.o.b.-c.i.f. adjustment factor
- $a_{ij} = X_{ij}/E_iX_{ij}$, the share of country *i* in the total of all countries' exports to market *j*.

If imports are quoted either in c.i.f. or f.o.b. the last equation should yield a value of zero. This result is essentially the same when imports are disaggregated into commodity groups. Given this system of equations, exports and trade balances may be obtained from a set of import forecasts (M_j) , knowledge of the market share (a_{ij}) and the f.o.b.-c.i.f. adjustment factor (i). Each of these factors are examined below.

Import equations

The specification of imports may vary depending on the nature of imports. Raw-material imports may be related to industrial activity, consumer imports to consumption demand, and capital imports to conditions of investment and return. In the absence of such a detailed account, the alternative is to use reduced form equations relating imports to exogenous variables. This procedure is relatively easy to compute and understand. However, such a procedure inevitably leads to a loss of knowledge on structural interactions as well as an inability to apply 'superior' statistical properties. In pursuit of simplicity, the import equation adopted in the Taplin's model is given by,

$$\frac{M}{P_M} = a_0 W A E^{a_1} P^{a_2}$$
^[7]

where:

M= import value Рм = import unit value index. $WAE = 0.5 AE_t + 0.5 AE_{t-1}$; average of autonomous expenditure. = G + I + X; autonomous expenditure in constant prices defined AE as the sum of government expenditure, gross fixed capital formation and export of goods and services. = domestic price index divided by import unit value index. Р = elasticity with respect to WAE. a_1 elasticity with respect to relative prices (P). a_2 =

The use of autonomous expenditure as an activity indicator is justified on two grounds. First, the simultaneity problem that would arise because GNP includes imports could be avoided. Second, in a simple income determination model where GNP includes consumption, investment, exports less imports and autonomous components, such a functional relation would be expressed as a reduced form equation, since imports are related either to consumption, income or both. Consumption in turn is always considered as a function of income.³⁶ In EWTM, increases in demand for imports leads to an increase in demand for exports of the supplying countries. Increase in demand for a country's exports also leads to increases in that country's demand for imports through the income generation process.

Prices also play an important role in import demand determination. In order to account for substitution possibilities, relative prices (price ratios) are used. The role of prices is believed not to be simple. Thus, domestic prices are seen as related to import prices and demand for imports may affect the export prices of another region. Although the ideal solution is recognized to be the joint determination of export supplies and import demand, this is not applied, owing to lack of data.³⁷ The method used for estimating the import function is OLS on annual data.

Export shares matrix

Market shares $(a_{ij}$'s) may vary due to changes in relative prices, exogenous change in taste or other factors. For each year and each commodity there are 729 (27 times 27) shares which must add to one in any
market. The author adopts a method developed by Armington (1969) in his study of demand functions for n goods from m suppliers to estimate the market share values (Taplin 1973: 190). Armington specifies the demand function of country j for the k^{th} type of good supplied by country (or group of countries) i after assuming (1) preference for one good is independent of the other and (2) that elasticity of substitution among suppliers is constant. This takes the form:

$$X_{ij}^{k} = b_{ij}^{\delta kj} X_{j}^{k} \left(\frac{P_{ij}^{k}}{\overline{P}_{j}^{k}}\right)^{-\delta kj}$$
[8]

and,

$$P_{ij}^{k} X_{ij}^{k} = a_{ij}^{\delta kj} \left(\overline{P}_{j}^{k} X_{j}^{k}\right) \left(\frac{P_{ij}^{k}}{\overline{P}_{j}^{k}}\right)^{1-\delta kj}$$
[9]

where:

$$X_j^k$$
 = quantity demanded δ_{kj} = elasticity of substitution
 P_{ij}^k = import price \overline{P}_j^k = average price level

By dividing equation [8] by X_j and [9] by $P_j^k X_j^k$ we may obtain an expression for market share given as

$$\frac{X_{ij}^k}{X_j^k} = b_{ij}^{\delta kj} \left(\frac{P_{ij}^k}{\overline{P}_j^k}\right)^{-\delta kj}$$
[10]

$$\frac{P_{ij}^{k} X_{ij}^{k}}{\overline{P}_{j}^{k} X_{j}^{k}} = b_{ij}^{\delta kj} \left(\frac{P_{ij}^{k}}{\overline{P}_{j}^{k}}\right)^{1-\delta kj}$$
[11]

Assuming further that bilateral prices may be replaced by export unit-value indices for the exporting country, and that all imports are grouped together for the total trade model (or into the commodity categories), the model may help to handle share matrices. P_j^k was approximated by a linearly weighted average of export unit-value indices, with the previous year's share of country *i* in country *j* being the weight for the current period. Finally, constant market share instead of constant elasticity of substitution is assumed. The market share is given by,

$$\frac{X_{ij}^{k}}{\sum_{i} X_{ij}^{k}} = a_{ij}^{k} = A_{ij}^{k} \left(P_{ij}^{k}\right)^{-\delta kj}$$
[12]

where: $P_{ij}^{k} = \frac{P_{ij}^{k}}{\overline{P}_{i}^{k}}$

$$\overline{P}_{j}^{k} = \sum_{m} a_{mj}^{k}, \quad \dots \quad P_{m}^{k}$$
[13]

Exports are in constant price (in volume) and all variables, except the weighting, are for the time period t. Thus the market share (a_{ij}) is based on volume flows. P_i^k is the export unit-value index of country i for commodity k. P_j^k (bar) is the average export unit-value index in market j obtained by weighting each supplying country's export unitvalue index by that country's share in the previous period. The market share elasticity $\delta_i kj$, and Aij is a constant term.³⁸ For the purpose of modelling, a matrix of dimension 27 by 27 is constructed for total trade and each commodity group. The share matrices are computed by dividing the value of exports f.o.b. from country i to j of commodity k by all countries' exports of commodity k to j. Exports to the RW (the developing countries) are obtained as a residual, subtracting the exports to countries from 1 to 26, as well as to unallocated groupings (created to record miscellaneous trade), from total exports.

The f.o.b.-c.i.f. problem

Since exports by one country could be recorded as import c.i.f., and at f.o.b. by other, there could be a problem of consistency at total world trade level (for instance, USA, Canada and Australia record imports at f.o.b.). In the EWTM, the ratio of recorded imports of a country to the sum of all countries' exports to that country $(M_j/EX_{ij} = F_j)$ were calculated and termed the 'f.o.b.-c.i.f. adjustment factor'. This reflects not only the f.o.b.-c.i.f. discrepancy, but also statistical errors, geographical factors, timing of shipment and other factors. Thus, caution should be exercised, in order to avoid systematic error, which may create biased market share figures. This share is used to correct (deflate) the total imports of each country, as is shown in equation [2].

Following the c.i.f.-f.o.b. correction, the final model contains 27 import equations, 27 export equations, 27 trade balance equations and 3

83

identities about total imports, exports and trade balance. The 84 equations have M_i , X_i , B_i , a_{ij} and F_j as endogenous variables. All price measures and Z_i 's are exogenous. The model is then solved by iteration.

The strength of this model lies in its simplicity and limited data requirements. The f.o.b.-c.i.f. adjustment factor is also a useful innovation, which may be applied elsewhere. However, the simplicity of this model does not come without cost. As is usually the case, the focus of the model is on the industrialized countries, with the 'rest of the world' (i.e., the developing countries) being obtained as a residual. Although an attempt is made to immunize this residual from miscellaneous and statistical problems, there is no guarantee (within the model) that this will be successful. The import equation, which is the pillar of the model, suffers from huge aggregation problems and, thus, is highly simplified. Supply is totally neglected. Exports are not specified in their own right, but as derived from the import function, with all associated setbacks, which I pointed out earlier). Finally, as is usually the case, the external market is modelled as an enclave of the macro set-up of importing/exporting countries, precluding the possibility of analysing repercussion and feedback effects. This is an important criticism, since one of the objectives of this model is policy analysis.

To sum up, in this section I have examined some recent efforts to model trade in primary commodities. Since summary and comment for each of the surveyed models is given at the end of each section, we need not repeat them here. Naturally, surveying this type of model brings with it a host of difficulties, not least relating to the sheer number of models which have been developed. For instance, Labys (1978) provides a list of some 337 commodity models.³⁹ Notwithstanding such difficulties, I have summarized some basic features of the models using the synoptic table below (Table 2.1). Based on this survey, and the theoretical discussion that precedes it, the approach in this book (Chapters 4 and 6) is informed by:

 Explicit consideration of the short-, medium- and long-run impacts of demand and supply factors. Thus, the discussion about equilibrium and disequilibrium models in the works of Chu and Morisson (1986), Goldstein and Khan (1978) and Hwa (1985) will be used.

- The emphasis accorded to estimation across different commodities. In this regard the studies of Hwa (1985), Ramanujam and Vines (1990) and Bond (1987) are informative.
- The importance of focusing on factors, which affect the market owing to the macroeconomic interaction of the two economies (say North and South). The studies of Dick et al. (1983), Taplin (1973) and Bond (1987) represent good starting points, from which to proceed. However, these could be improved by placing the commodity market within a global macroeconomic framework.

Considering these points, short-, medium- and long-run supply, demand, stockholding and prices will be taken as the key relationships required in modelling African trade. As pointed out earlier in this chapter, a common weakness of these models is the lack of connection with macroeconomic balances and the neglect of their feedback effects (although we note that Dick et al. have attempted to take account of such effects). In this book, existing modelling works will be built upon by focusing, firstly, on the importance of macroeconomic interactions and placing the commodity model within a global macro framework, and, secondly, by emphasizing the supply side of commodity modelling and the impact of interest rate on the demand for commodities. The latter, in particular, will serve as an important mechanism for linking the commodity market with international finance and Northern macro policy.

2.4 CONCLUSION

This chapter, and particularly the last sub-section, has examined the interrelated questions of why countries trade with each other, and who gains from this trade. This has been set in a historical framework, with the discussion on these questions being traced back over the past two centuries. The main objective of this survey has been to find a systemic explanation to help illuminate the trade and external finance problems of Africa, as outlined in Chapter 1. The main conclusion of this survey has been that mainstream trade theories, from Ricardo to the present, serve to justify the specialization of African countries as producers of primary, and particularly tropical, commodities. However, as demonstrated in Chapter 1, this specialization represents precisely the root cause of Africa's financial problems.

This finding, in turn, justifies the search for other systemic explanations about African trade and finance problems. Thus, in this chapter it is shown that even the 'new' trade theories, including the technological gap models, are better suited to explaining North-North, than North-South trade. This effectively limits our choice to those theories, which focus on primary commodities and secular deterioration in terms of trade. These theories, informed by the structuralist economists of ECLAC, focus mainly on the structure of markets and on North-South interaction. This may be formally captured through a commodity modelling approach. Thus, the theoretical discussion and the overview of current models presented in this section will be used to inform the empirical analysis undertaken in subsequent chapters, and particularly Chapters 4 and 6. However, since the overall objective of this book is to examine the interaction of international finance, trade in primary commodities and Africa's place in the world economy from a macro perspective, we will now move on to examine some wider issues relating to the macroeconomics of Africa. This analysis, which will be set out in Chapter 3, will form the basis for the actual modelling work presented in subsequent chapters.

The Macroeconomics of Africa: 3 Import Compression and External Finance

3.1 INTRODUCTION

This chapter will examine some of the main features which might usefully be included in a macro model for Africa. It is hoped that this discussion will help in formulating an actual prototype African macro model, which might form a working component of the North-South model developed in Chapter 6. Although the relevant African macroeconomic framework for this should broadly be similar to that outlined in Trap (1993), there are, nevertheless, a number of other specific features not properly dealt with in Trap (1993), which an African macroeconomic framework might usefully include. The first such feature is an 'import compression argument', as discussed in Ndulu (1986, 1991) and Rattsø (1992b). Two other features, which I will argue should also be incorporated into macro models of Africa, are the 'fiscal response to external finance' and 'Dutch disease' phenomena. Hence, an attempt will be made to integrate these features into the prototype African macro component of the North-South model developed in Chapter 6.

The discussion presented in this chapter will be organized as follows. In section 3.2, a number of published macro models of Africa will briefly be reviewed, with the aim of examining the main features emphasized in these models. In section 3.3, the import compression argument, as well as a number of other important theoretical issues relevant for the construction of an African macro model, will be discussed. Section 3.4 will outline the accounting framework, which will then be used to construct a consistent macro database for use in this study. Finally, section 3.5 will attempt to draw out a number of conclusions, based on the preceding discussion.

3.2 A CURSORY LOOK AT SOME MACRO MODELS OF AFRICA

As a recent survey by Harris (1985) shows, macro modelling in Africa is still in its infancy. Harris surveys most of the macro models of Africa which have been constructed by North American universities and international institutions, as well as published models of the African economy. He reports on a total of 184 macro models, 120 of which he evaluates further (see Diagram 3.1). Harris classifies the 184 models based on their underlying structure. Thus, models are categorized as (1) demand-driven Keynesian-type models; (2) supply-driven, general equilibrium with price adjusting to clear the market; (3) reduced form monetary-driven; and, (4) models based on consistency checking, without formal closure (see Diagram 3.2). Harris also applies a classification based on publication and sponsorship. Thus, models are grouped as: (1) Ph.D. dissertations, (2) university-based professional work, (3) international agency-sponsored, and (4) models developed by African governmental agencies (Diagram 3.3). Harris' data is summarized below.







Diagram 3.2 Type or structure of African macro models (Harris 1985)

Diagram 3.3 African macro models by source (Harris 1985)



Naturally, these models emphasize different aspects of African economies. Thus, in order to understand how these work, a number of different types of African macroeconomic models are examined below. While some of these study the impact of foreign borrowing or, indeed, a sudden onset of oil or other mineral revenue on the economy, others focus more on fiscal conditions. However, in the discussion which follows, I will not primarily be concerned with the objectives of each model, although I am cognizant of the fact that this objective may

shape the type of modelling undertaken. Rather, my own focus is on how these various approaches have been applied in building models of African economies. From this, I hope to be able to draw out some lessons, which may help in developing a prototype macroeconomic model for Africa.

I will begin by examining the work of Oshikoya (1989), who has developed a macro-sectoral model for Nigeria, with the aim of studying the impact of expanded domestic spending on externally-borrowed funds. His model is interesting in that it integrates a macroeconometric approach with an input-output analysis. This integration allows one to determine final demand from equations of economic behaviour, and to allocate this over sectoral production activities, by means of an inputoutput table. Oshikoya's model comprises five production sectors (bloc 1). Each sector's value added is determined by transformation of the input-output type production function and by linking final demand components with the sectoral value added. The residuals, in each of these equations are modelled by econometrically relating these, alongside others, to (lagged) capital stock, which is believed to show the capital constraint in such economies (and hence represent a supply factor in the long run). At the same time, this capital stock augments demand, in the short run. This bloc is related to the external sector, which basically takes foreign borrowing as exogenous (bloc 2), as well as to final demand, in which the components of final demand are estimated (bloc 3). A major strongpoint of this modelling approach is that it takes account of the impact of supply and demand. A further interesting feature of this model is that its production function is made realistic by the use of inter-industry coefficients. Although the structure of the lags does not follow the theoretical formulation, the estimation of the supply components of blocs 1 and 2 is its unique and insightful feature.

Olafin and Iyaniwura (1983) also report on a macroeconomic model of *Nigeria* developed by the University of Ibadan, Nigeria.¹ The model consists in 25 stochastic equations, of which 16 are exogenous and lagged endogenous, including output of petroleum and export of agricultural products, as well as four identities. GDP is decomposed into six sectors, which are then specified within the model. These are (1) agriculture, (2) mining and quarrying, (3) manufacturing and crafts, (4) transport and communication, (5) construction, and (6) services. Agriculture is taken as a function of exports of agricultural products, domestic consumption of food, non-food agricultural and other products, and investment in agriculture and mining sectors. Within the mining and quarrying sector, gross output is explained by output of crude petroleum, as well as by capital formation within the sector. Manufactured output is explained by urban population (taken as a proxy for demand) as well as investment in machinery and equipment (taken as a proxy for productive capacity). Export of petroleum is related to its output and to government revenue. The consumer price index (as a proxy for inflation) is related to money supply, government expenditure and the price of imports. Finally, money supply is related to the government deficit, government securities, rate of change and stock of reserves. These sets of equations are then closed using four identities given in the model. These identities sum up the GDP component, government revenue, total population (urban and rural) and total imports.

Pleskovi (1989) has also constructed a computable general equilibrium model for $Egypt^2$ with the objective of studying fiscal incidence in that country. Pleskovi's model is based on Harberger's (1962) model and a SAM framework. The extended version of Pleskovi's model includes a nine-sector input-output table. This model is a very simple one with a number of restrictive assumptions. These include perfect mobility of labour and capital, a closed economy, full employment, fixed technology and an elastic supply of factors. The SAM comprises seven accounts, with twelve rows and columns. For the purposes of the model, the economy is assumed to have two goods. On the supply side, net output is represented by a Cobb-Douglas production function, while gross output, including intermediate output, is represented by an inputoutput production function. The value-added relationships for each good are derived from the SAM. On the demand side, final demand is equated with consumption and intermediate demand. The income of urban and rural households is derived from their endowment of capital and labour, to which is added transfer income. Finally, expenditure (by activity) and prices (given by factory-get price plus tax) are defined by a set of equations derived from the SAM. The model leaves prices and output of the two goods, wages and returns to capital to be determined endogenously. Tax rates and allocation of tax revenue are assumed exogenous, while most parameters are taken from the SAM and assumed to be constant. The model is based on a number of heuristic assumptions about Egypt's economy. Thus, it deliberately avoids addressing issues of accumulation and, in fact, is very limited in its coverage of the economy in general. Nevertheless, the attempt to maintain consistency

through use of a SAM, as well as its focus on tax (as both revenue and transfer) represent interesting features of this model.

Benjamin et al. (1989)³ constructed a CGE model for Cameroon, which sets out to study the economic impacts of an oil boom. This model is an 11-sector version of a model of 'Dutch disease' in a developing economy. Its underlying mechanism may be grasped from the sector version, which is presented as the theoretical underpinning of the model. Thus, according to this version, the economy produces output, which is either domestically consumed or exported. Imports are assumed to be imperfect substitutes for domestic production, with consumers having constant elasticity of substitution (CES) over the two goods. Based on these assumptions, demand is taken as being determined by the relative price of these two goods. World price is also taken as exogenous (i.e., the small-country assumption). This is converted to the domestic price of imports through application of an exchange rate. Export demand is considered to be a downward-sloping schedule in terms of its domestic, relative to the world, market price, Finally, the system is closed by equating national income – augmented by inflows of foreign exchange - with total spending on domestic consumption and imports. Since this system generates five variables with four equations, foreign exchange inflow is assumed to be exogenous. However, the parameter of this variable is changed with the aim of simulating an oil boom, and thus examining how this affects other variables within the system. Within this broader framework, the authors assume that foreign savings are exogenous, that investment is determined by total savings, and that capital is fixed and sector-specific. This type of specification renders 'Dutch disease' type models, with different implications in developing countries, where, for example not all the traded sector may contract. However, the model has a number of obvious limitations. For instance, the small-country assumption is held for imports but, rather unrealistically, is left out when it comes to exports. It is also interesting to note the implications of the imperfect substitution of imports for domestic production assumption employed in this model.

Lipumba et al. (1988)⁴ constructed a macro model for *Tanzania*, containing 89 equations, of which 32 are behavioural. The model, which is supply-constrained, sets out to study the role of capital formation in constraining economic activity in that country. In so doing, seven major blocs are defined: (1) GDP by sectors of production, (2)

employment, (3) expenditure on GDP, (4) the government sector, (5) the monetary sector, (6) the external sector, and (7) prices and productivity. Exports and other foreign exchange earnings are taken as the major determinants of ability to afford inputs required by the non-agricultural sector. For each of the eight sectors which are defined, output is modelled using a production function, with capital and labour as arguments. In some of these sectors, intermediate input is also taken as a constraint. Employment is endogenously determined in two of the sectors, through real wage and sectoral output, while it is assumed to be exogenous on the others. Within the expenditure bloc, private consumption is related to disposable income, while capital formation is assumed to be an 'accelerator' type, which will depend on net inflow and availability of foreign exchange. These, in turn, are taken as determined by previous year's exports. Government deficits are specified as affecting the money supply, leaving interest rates as relatively unimportant within the monetary sector. Demand for money is seen as passively adapting to supply, which, in turn, affects price level. Tax revenue is related to the non-agricultural sector, private consumption as well as imports and exports.

Within this model, the external trade sector is one of the most elaborated. This serves to underscore the role of imports and exports in constraining supply. Five main agricultural export crops are endogenized. As far as exports are concerned, the small-country assumption is maintained. Finally, prices are taken as determined by import price levels, wages and labour productivity. Account is also given to the role of producer prices in affecting export crop production. In turn, these producer prices are seen as depending on world market prices, exchange rate, marketing margin and the non-food consumer price index. World market price is likely to be particularly important in influencing the price of perennial crops, while the prices of cotton and tobacco are likely to depend, to a considerable extent, on the non-food consumer price index. Weather dummies, as well as a time trend (which shows institutional changes), are also included within this model. Imports (which are subdivided into 'consumption', 'intermediate' and 'capital') are related to demand, consumption, GDP and capital formation. Foreign exchange availability, rather than prices, is considered as representing the most important constraint to imports. Finally, price deflators are related to labour productivity and import prices, with a built-in lag, with money supply also given a role in determining GDP deflator and CPI (Consumer Price Index).

This attempt to describe the Tanzanian economy represents the most elaborated model so far specified. Much attention is paid to depicting the role of imports in constraining supply (i.e., capacity utilization). The impact of external sectors is also basically modelled through its impact on imports and prices. Although not utilized in the model, a sort of 'Dutch disease' phenomenon is observed for some of these exports. However, no role is given to domestic capital formation, which represents a shortcoming of this model, since such considerations might have been useful in the search for policy options. Foreign finance is also considered in a general way and hence, this analysis is kept less detailed. However, different financial flows are recognized as having different impacts. In summary, overall this is a useful model, which captures most of the salient features of the Tanzanian economy.

Van Frausum and Sahn (1993) have constructed an econometric model of *Malawi*, which sets out to measure the effects of external shocks and policies in that country. The model is composed of five blocs with a number of stochastic equations and identities. The production bloc consists in equations that predict sectoral output. Agriculture is disaggregated into cash crop and maize production by smallholders and export crop production by the state. Due to lack of data on labour and cultivated land, output is related to relative prices, weather, as well as secular changes captured by the time trend. Industrial production is explained by imports of intermediate inputs, capital stock of equipment and infrastructure. Value added in construction is regressed onto gross fixed capital formation of the government and the private sectors.

A further bloc examined within this model relates to balance of payment (BOP). This bloc attempts to show how the current account problem is ultimately transformed into external debt. Imports of goods and non-factor services are determined by the availability of foreign exchange, net of foreign borrowing, debt service payment and other factor payments. This is then set, together with creditworthiness indicators, as a constraint on imports. Total external debt is modelled as the sum of lagged debt level as well as balance on current account. The model also examines the government finance bloc. Thus, government revenue is determined by imports and GDP, with official transfers taken as exogenous. Government debt service and interest payments on domestic borrowing are taken as a function of outstanding external and domestic debt, with a one-period lag. Government expenditure is derived as an identity defined by a financing constraint. The final bloc within the model examines the price and monetary sectors of the economy. With regard to prices, most essential prices are allowed to be determined endogenously. In relation to the monetary sector, this is examined in a simplified form. Thus, the advance from commercial banks is modelled as a function of foreign exchange availability, through the monetary base and real interest rates, with the advance to the private sector taken as residual, once the government's share has been removed.

In summary, this is a very detailed model, which makes a good attempt to depict the Malawian economy. Its strong points are its attempts to link the current account balance with the debt condition, as well as to separate export and non-export agricultural production. On the negative side, some of the assumptions upon which the model is based may be questionable. For instance, the model assumes that government expenditure is determined by a financial constraint. This assumption ignores the inelastic nature of government spending, as well as the inflation that such spending generates in most developing countries. Moreover, variation of investment across public and private institutions have not been properly addressed, although some attempt to take account of these has been undertaken for the construction sector.

Kavizzi-Mugerwa (1990) formulated a model with the aim of studying the impacts of copper prices on the economy of Zambia. The underlying theoretical formulation for this is the 'Dutch disease' literature, in which the spending effect and real exchange rate are given a key role. The economy is divided into tradable⁵ and non-tradable sectors. The ratio of the price of tradables to non-tradables determines the real exchange rate. Within the model, government revenue is related to GDP, expenditure to government revenue, and both of these to their respective lagged value, this indicating lack of instant equilibrium. The real exchange rate is set to depend on price of copper, both through its impact on the non-traded sector, and by affecting the level of credit which, itself, affects demand. The rise in non-tradable prices is determined by excess demand for money, which itself is related to income and to relative prices. Real private consumption is related to real income as well as the stock of money in the preceding period. Real government expenditure, private consumption and relative prices determine the volume of imports. Finally, the model is closed using an income

identity, as well as a definition of changes in the nominal money supply.⁶

In summary, the working of this model is clear and simple. Fiscal and trade disequilibria resulting from a change in copper price affects the domestic credit and volume of money. These, in turn, have an impact on non-traded prices, the real exchange rate and consumption, with the last two of these having a knock-on effect on import levels. The strength of this model is its attempt to describe the domestic sectoral adjustment process. Its weakness is that it overlooks the impact of the other aspects of the external sector. Moreover, the production structure of major sectors is also overlooked.

Berhanu (1994) has built a Kalecki-FitzGerald-inspired macroeconomic model of *Ethiopia*. The model is very small, having only six equations. In the first of these equations, the manufactured price is formulated, à la Kalecki, while, in the second equation, marketed surplus is formulated as a function of internal terms of trade. In the third equation, the total output of the manufacturing sector is assumed to equal the marketed surplus, non-food consumption of wage earners in the manufacturing and export sectors and the unsaved profit of capitalists. The remaining equations define total profits in the economy, the demand and supply of basic goods (including imported ones), the external balance, and, finally, investment (which is constrained by balance of payments).

The model assumes, *inter alia*, that the agricultural output market is not flexi-price, and that investment (not output) adjusts to the foreign exchange constraint. These assumptions neither follow the standard structuralist assumptions upon which the model is based, nor tally with the stylized facts in Ethiopia. Rather, peasants in that country had been marketing a good portion of their output in the flexi-price market, most industries were extremely import dependent, and import compression – with its adverse effect on capacity utilization and output – was the rule. Moreover, the ('socialist') state, as centre of accumulation, is not well integrated into the model. The model also contains a number of other limiting assumptions. Manufactured goods are assumed to be necessities within rural household consumption, the model abstracts from the debt problem and hence, balance of payments is essentially assumed to be a trade balance. These assumptions seriously limit the relevance of the model. Thus, if we assume that import compression takes place, then some of the arguments and conclusions set out in this model may no longer be sustained.

Notwithstanding such shortcomings, the model gives an interesting analysis of the distribution of income, process of accumulation/growth and the policy dilemma facing different actors. Its importance should also be seen in the context of a virtual absence of macro models for Ethiopia.⁷ This analysis would have been more illuminating if mark-up pricing and its effect on accumulation had not been assumed to have a foreign exchange constraint. Indeed, it could be argued that this foreign exchange constraint, rather than the author's emphasis on the share of profit, should have been the focus of analysis, in relation to accumulation.

Davies et al. (1994) have also developed a CGE model for Zimbabwe. The model contains 25 supply-side equations. Five of these related to a fixed wage demand-determined labour market; eight equations specify income generation, saving and consumption; three define public sector accounting; and four define foreign exchange rationing. Finally, one equation is set as a closure that equates the 25 supply-side equations with total demand, while another ensures consistency by maintaining the investment-saving balance. Although these are the basic features, the fully specified model consists in 113 equations, all but one of which are independent, with the remaining equation dealing with the saving-investment balance.

One of the notable features of this model is the inclusion of an import compression aspect within a CGE model framework. The model identifies five sectors, each of which has a different adjustment mechanism. Thus, the agricultural sector will adjust through a flexi-price mechanism, services by way of a demand-determined one, construction through supply-demand adjustment, exportables derived as residual and, finally, importables through supply-demand adjustment. Imports are basically rationed among these sectors, with priority given to the exportable and construction sectors. The explicit inclusion of import compression and its linkage with food production (as shown in the counterfactual analysis) represent an interesting feature of this model. However, being a CGE model, it suffers from the use of parameters, which essentially show a 'snap-shot' picture of the economy. Moreover, exogeneity assumptions in relation to the agricultural sector, as well as similar assumptions in relation to investment and its impact on capital formation, could set a limit to the wider use of this model.⁸

Kwack (1989) also developed a prototype model for *Africa*, which is applied to different African countries. The model's underlying assumption is that the economy is 'small and open in the sense that activities and prices in the country are affected by conditions in world markets where its own influence is minimal' (Kwack 1989: 137). Thus, African countries are assumed to be 'price takers' in the international market. Within the model consumption is specified as a function of disposable income and inflation. Investment is basically an accelerator type (where change in output and lagged capital stock are arguments). This is augmented by the inclusion of a financial constraint (credit). A further aspect of the aggregate demand bloc comprises government expenditure and revenue. Thus, direct taxes are related to nominal income, and indirect taxes to nominal private consumption, nominal imports and exports. Finally, government consumption expenditure is considered as a policy variable.

On the supply side, GDP is assumed to equal the sum of value added in (1) agriculture, forestry and fishing; (2) mining and quarrying, and (3) manufacturing, utilities, construction and services. Output of mining and agricultural sectors are assumed to depend on world price relative to domestic cost and capital stock, while services and manufacturing outputs are assumed to depend on domestic demand.

Merchandise trade is disaggregated into four commodity groups. Imports are set as determined by relative prices, including tariffs and real income. On the other hand, real exports in manufacturing are specified as determined by foreign income and export prices relative to overseas price. Exports of primary commodities are assumed to depend on domestic supply, with producers of these exports assumed to be price takers. Exports and imports of services are assumed to depend on the level of exports and imports of goods as well as on overseas prices. It is also postulated that factor income payments depend on the USbased LIBOR (London Inter Bank Rate) interest rates as well as on outstanding net external assets. An attempt is also made to combine the monetary sector with balance of payments, by defining the demand for money as a function of real GNP and rate of inflation.

The prototype model explained above is estimated for ten African countries. As is explicitly acknowledged by the authors, the method suffers from simultaneous equation bias, since each equation is estimated using OLS. The author also notes theoretical and data limitations in modelling Africa. The former relates to such factors as the absence of financial markets and failure of the neo-classical production function, which may arise from natural disturbances, and other related causes. In relation to data quality, gaps in data, comparability across countries, underestimation of the subsistence sector, as well as lack of information as to the method of estimation of the acquired data, are all mentioned. However, in spite of these shortcomings, the author claimed that his model performed reasonably well. More generally, he stresses the importance of policy modelling in Africa.⁹ Within this model, output in certain sectors, such as manufacturing, is assumed to be demanddetermined. This assumption overlooks the stylized fact of import compression. Moreover, the model ignores the role of Northern demand as well as the commercial policies pursued by Northern nations, in affecting Africa's export sector. By emphasizing relative prices as determinants of imports, the model further undermines the importance of capacity to import, and hence capacity utilization, as well as creation.

Based on the above review, one might reasonably conclude that most of the published macro models of African nations developed in the past decade have been designed with the aim of studying the impact of external factors on the economy. Although a rigid classification would be difficult to justify, since overlapping objectives are common, the models reviewed nevertheless may broadly be grouped into those which:

- (1) focus on the impact of foreign capital and foreign exchange earnings on major macro variables (Lipumba et al. 1988, Oshikoya 1989, Davies et al. 1994),
- (2) stress the impact of oil revenue (Olafin & Iyaniwura 1983, Benjamin et al. 1989, Benjamin 1990),
- (3) focus on the impact of external shocks in general (Lipumba et al. 1988, van Frausum & Sahn 1993, Kayizzi-Mugerwa 1990), and
- (4) focus on domestic macroeconomic conditions and policy (Boutros-Ghali & Taylor 1980, Asmerom & Kocklaeuner 1985, Pleskovi 1989, Harton & McLaren 1989, Lemma 1993, Berhanu 1994, Decaluwé & Nsengiyumva 1994, Davies et al. 1994).

Based on the above review, a number of lessons may be drawn from these various efforts to depict African economies. Firstly, African macro models should focus more on the supply-constrained nature of the economy. This may be done by concentrating more on the role of both intermediate imports (in the short to medium term) and on capital formation (as representing a longer-run concern), as well as on their mechanism of financing. Secondly, sectoral adjustment (between traded and non-traded sectors) could represent an important focus for future efforts. And finally, the government fiscal posture and the monetary sector need to be linked and should be left open to the influence of external sector effects. In addition to these general lessons, a number of technical recommendations (for example on combining input-output analysis with econometrics) may be drawn from these models. These recommendations, taken alongside a number of theoretical issues which we examine in the following section, will form the basis for the construction of the prototype model for Africa detailed later in this book.

3.3 SOME BASIC ISSUES IN AFRICAN MACRO-ECONOMICS

In this section a modest attempt is made to identify some salient features of macroeconomic theory as it relates to Africa. Trap (1993) provides an excellent assessment of the consistency of the macro framework advocated by the IMF and World Bank, as well as the theoretical underpinnings of the 'Stabilization' and 'Structural Adjustment Programs' sponsored by these institutions. However, notwithstanding his discussion on alternative macroeconomic theoretical frameworks, Trap does not actually go so far as to propose an alternative model for Africa.¹⁰

A further effort to understand how the African economy functions is associated with the economists of the Centre for the Study of African Economies (CSAE) at the University of Oxford. A number of studies undertaken by the CSAE have been helpful in highlighting which are likely to be the most productive channels through which policy might be propagated. Methodologically, these studies adopt a basic neoclassical approach, dealing with structural factors on only an *ad hoc* basis.¹¹ However, in common with the IMF and World Bank models, the African economic crisis is more often characterized as essentially a policy problem by CSAE economists. Thus, it is argued, those nations who achieved the best growth rates were those who were boldest in liberalizing (see Collier & Gunning 1999 for an excellent summary of the CSAE's thinking on this).¹²

Many of the models surveyed above, as well as those studies falling under the auspices of the African Economic Research Consortium (AERC), have been undertaken individually, and hence, have not been organized into any kind of coherent framework. Indeed, it could be argued that the time for some degree of abstraction from these various models is long overdue. This would allow for the identification of salient features which might be included in an alternative economic model for Africa. Hence, in the remainder of this section exactly such an exercise will be undertaken. Issues highlighted in this discussion will then be empirically examined in the subsequent two chapters. Based on this analysis, a prototype economic model for Africa will be developed and applied in Chapter 6.

3.3.1 Import Compression

The balance of payment crisis discussed in Chapter 1 has brought about a policy response in most African countries. This response may be termed 'import compression' (Ndulu 1986, 1991; Rattsø 1994). The import compression model works as follows. The balance of payments crisis had set a limit on the level of imports. This, in turn, led to direct government control over the allocation of such imports. Given such constrained import capacity, governments were forced to set priorities between imports of intermediate goods and of investment goods. This sets in place a trade-off between capacity utilization and capacity expansion (Rattsø 1992b, Ndulu 1986, 1991). Understood in the context of the gap models of Bacha (1984) and its extension in Taylor (1991), the inclusion of intermediate imports as determinants of capacity utilization has the effect of relaxing the fixed import coefficient assumed in these models (Rattsø 1994: 36).

Rattsø's (1994) model is interesting in that it explicitly emphasizes the need to incorporate an import compression component into African macro models. However, its weakness is that it relies on the movement of real exchange rate and the assumed flexi price¹³ structure in this protected, import-dependent sector. In fact, this pricing and market structure would appear to contradict the stylized fact of mark-up pricing in such sectors. Indeed, the fact that these sectors are protected by states or by their shear size in many African countries would appear to justify this assertion. Thus, incorporating this latter set of assumptions into Rattsø's (1994) model will result in a different set of conclusions.

Ndulu (1991) has extended his original contribution – which was based on Tanzania's experience (Ndulu 1986) – to a number of other African countries. Thus, as Ndulu notes, the process of growth entails

not only the trade-off between current consumption and capacity growth, which was the focus of most labour surplus-based models, but also between capacity growth and capacity utilization, in an importcompressed situation (Ndulu 1991: 288). Ndulu argues that idleness of capacity in industry has resulted from structural bottlenecks as well as a mismatch between rate of expansion of capacity and the availability of operational resources (Ndulu 1986: 14). This, combined with a 'cost-plus' (i.e., mark-up) pricing system, implies that output growth in the short run will depend on capacity utilization, which in turn will depend on the composition of intermediate imports or, more generally, of total imports (Ndulu 1986: 15–19).

Import compression occurs either when domestic goods are imperfect substitutes for imported ones, or as a consequence of low import capacity of African countries. The latter will usually result from low levels of export earnings and foreign savings, which in turn will depend, *inter alia*, on external terms of trade, net foreign resource inflows, weather vagaries in economies dominated by rain-fed agriculture, and discovery and exhaustion of mineral resources. Hence, the possibility of exogeneity of growth in sub-Saharan Africa is likely to be a real one¹⁴ (Ndulu 1991: 290). Ndulu further expands this framework to include the public sector in his analysis, along a three-gap line of thinking. Basing this analysis on empirical evidence of public sector crowding-in in Africa, Ndulu underlines the existence of a trade-off between growth, through public investment and inflation. The exact form that this takes will depend on whether this deficit is financed through borrowing or monetization (Ndulu 1991: 291–92).

The formalization of these ideas into a simple and elegant framework allows Ndulu to arrive at the important conclusion that 'in economies with excess capacity, the increase in net exports will not crowd out investment since domestic saving increases with the rise in capacity utilization'. He also notes that depreciation in the real exchange rate, together with a higher level of inflows and lower levels of debt servicing will play an important positive role in relieving the import compression problem. Similarly, such flows are also likely to ease the fiscal constraint to growth, both by raising tax revenue through growth of output and by reducing monetization (Ndulu 1991: 293–96).

Ndulu's analysis represents an important contribution to African macroeconomics. However, there are certain features within this analysis which could be elaborated upon and improved. For instance, within this model, the public sector deficit requirement is essentially just an accounting rule, or an 'identity'. However, if this is expanded to include behavioural rules, along the lines of the fiscal response literature, then some of these conclusions will need to be modified. Likewise, it could be argued that the impact of capital inflows on the trade and non-trade sector (i.e., the 'Dutch disease' effect), common in many African countries, should be explicitly modelled within such a macro framework. This addition is, in particular, important to accommodate the 'import compression argument' in the more liberalized context of the 1990s, where exchange rate increasingly became an important (adjusting) macro variable. The inclusion of these two related sets of concerns can only represent a positive development in relation to the macro modelling of African economies. Hence, the following section will outline how such considerations might be incorporated within a macro framework and this will be re-examined empirically in Chapter 5.

Arguably, a further shortcoming of Ndulu's framework is the modelling of inflation and the monetary sector using the 'Quantity Theory' of money. This implies that inflation is a purely monetary phenomenon. This is likely to have arisen because of the neglect of the non-export agricultural sector in Ndulu's formulation. Indeed, the linking of this sector to the import-compressed sector, along Kaleckian lines, is likely to have rendered a more fruitful analysis. This should be obvious in African economies where more than 50 per cent of the weight in the consumer price index (CPI) is constituted by the food category.

3.3.2 Macroeconomic Effects of External Finance

The modelling procedure pursued in the existing North-South models by and large fails to take account of the macroeconomic impact of external finance on the South. Nevertheless, theories of the macroeconomic impact of external finance, and especially of aid, which is particularly relevant in Africa, have grown in their own right. Such theories include the 'two-gap model', the 'Dutch disease' and the 'recipient fiscal response'. These are areas that have been studied in their own right, and hence have not been considered within a global framework or in a wider African macroeconomic framework.

The two-gap model (Chenery & Strout 1966) basically links a simple Harrod-Dommar growth model with flows of external assistance. The essential point of the model is that growth is constrained by skill supply, organizational ability, supply of domestic savings and the supply of imported commodities and services. The constraints to growth are examined in three phases. In phase I, growth is limited by the ability to invest and the savings gap determines capital inflows. Phase I ends when investment reaches the level adequate to sustain the target growth rate. In phase II, growth is set by the target set, and capital inflow is required to fill the saving gap. The tapering off of the capital inflow in this phase requires exports to grow faster than imports. However, this is found to be difficult for most LDCs and hence, a new set of restrictions, which is binding in this event, is introduced in phase III. Using this model the authors conclude that, although its impact varies in different phases, foreign economic assistance is generally productive, either in supplementing domestic savings or in relieving foreign exchange constraints.

The Chenery-Strout dual-gap approach has been criticized from a host of different angles. Griffin (1970),¹⁵ Griffin and Enos (1970) and Papanek (1972) argue that aid will displace, rather than supplement, domestic savings. A further argument against the two-gap model is that it adopts an extreme disequilibrium framework, and hence, neglects relative price effects. These effects are explicitly covered in the 'Dutch disease' literature, which is becoming an important aspect of the discussion on the macroeconomic impact of aid. The argument is simple. Revenue is obtained from a booming sector. If part of this is spent on non-traded goods ('the spending effect'), this leads to a real appreciation, that is a rise in the relative price of non-tradables to tradables. This, in turn, draws resources out of the booming sector into the nontraded sector ('the resource movement effect') (Corden 1984, van Wijnbergen 1984). The growth effect of such a resource shift is considered to be negative, since traded sectors are characterized by 'learning by doing', dynamic externalities which have a higher and positive effect on growth (van Wijnbergen 1984, 1986a, 1986b; Edwards & van Wijnbergen 1986). The similarity between this and aid underscores the importance of the 'Dutch disease' approach in explaining the macroeconomic impact of external finance. Hence, in Chapter 5, I will return to examine this issue and its relevance in the African context. The empirical findings of Chapter 5 will then be employed in the North-South model developed in Chapter 6.

A further aspect of the macroeconomic impact of external finance which could usefully be considered within African macroeconomics is 'the recipients fiscal response' to foreign inflows. Critics of the twogap model (Griffin 1970, Papanek 1972) have highlighted how foreign capital inflows may reduce government efforts to raise revenue, especially through taxation. This concern has been formalized by Heller (1975), in a model that assumes public sector decision-makers maximize a utility function. By maximizing this function, subject to the constraint that government expenditure should be equal to its revenue, he solves the model for a set of first order conditions that yield structural equations. The latter is estimated using data from a number of African countries. Mosley et al. (1987) have further extended this analysis by including aid's impact on growth of output as well as minor changes in the presentation of the public sector variables' specification. Chapter 5 will present an extensive treatment of the theoretical underpinnings of this literature, and will also attempt to set out an alternative means of thinking and estimation in relation to this. The results of this analysis will be incorporated into the model developed in Chapter 6.

To sum up, the macroeconomic impacts of external finance, and especially of the 'Dutch disease' and 'fiscal response', represents an important area of development within the literature. However, this has been a neglected aspect of North-South macroeconomic modelling, and also, to a certain extent, of macro modelling of Africa. Therefore, following an empirical examination of this issue in Chapter 5, I will explicitly incorporate this important literature within the specification of the North-South model developed in Chapter 6.

3.3.3 The Accounting Framework: An Overview

In this section I will not dwell upon the details of the relevant accounting framework for macroeconomic analysis in Africa. Harvey (1985), Trap (1993) and Lensink (1996) provide a good discussion of such accounting framework in African context. Rather, I will outline the procedure by which a consistent macro database might be built for 21 African countries selected for the purpose of this study, from three geographical regions: North Africa, West and Central Africa, and East and Southern Africa. This database, comprising panel data for the period 1970–90,¹⁶ will form the basis for all the empirical analysis undertaken in this book. Sample selection is guided largely by the availability of data, as well as by the economic and demographic characteristics of African countries. Thus, one of the features of the global model developed in Chapter 6 is the use of econometrically estimated parameters in almost all sub-blocs of the model.¹⁷ However, a major problem in constructing a macro database for each of the sample countries has been lack of consistency between the international data reporting systems of different institutions. In order to overcome this problem, I have used a data reconciliation procedure, which facilitates consistency in data reported by different international institutions. This procedure builds on earlier work discussed in Alemayehu et al. (1992) and FitzGerald (1993), and is formalized into a formal accounting framework, outlined below.

The accounting framework

Since the emergence of Keynes' macroeconomics in 1930s,¹⁸ the Keynesian version of national income accounting has become the *de facto* accounting framework for macroeconomics. Thus, much of the data used in macroeconomic analysis today is theory- (and more specifically Keynesian theory-) laden. However, this is not a static phenomenon, with change and progress in theories having repercussions on data. Hence, the accounting framework relevant to open-economy macroeconomics has undergone a constant process of improvement, although its essential features remain unchanged. In the following discussion, I will outline the framework that has been adopted in constructing the sample database of this study, and detail the improvements made to this. Chapter 7 will outline how this effort might be extended, with the aim of improving the accounting framework of open-economy macroeconomics within a global context. The latter will define the accounting framework, which will be used in solving the model.

The cornerstone of an open-economy macroeconomics accounting framework is the identity that links the internal balance with the external balance. Thus, in relation to the United Nations System of National Accounts, this is the link between national accounts and the balance of payment. However, one problem which one might encounter in using such an accounting framework is the lack of institutionally disaggregated detailed data which could, to some degree, be resolved by resorting to various multinational data sources (see Alemayehu et al. 1992 for details on this).

A major macro problem in the case of Africa - and developing countries in general, for that matter - is how to finance this investment. This may be addressed by way of the accumulation balance, which may be defined as,

I = S + F

where: I is gross domestic investment, S national savings and F net capital inflows. The latter is defined as the net change in assets and liability position of the country, and is equal to the deficit of the current account of the balance of payments (i.e., the external balance), which is given as,

$$F = M - X + N \tag{2}$$

where: M and X are imports and exports of goods and non-factor services, respectively, and N is net factor payment and current transfer to abroad. Combining these, disaggregated into public (g) and private (p)sectors and rearranging [1] and [2] vields.

$$(I_g - S_g) + (I_p - S_p) = M - X + N = F_g + F_p$$
[3]

This yields the basic identity which links the domestic investment and savings gap with the current account deficit or surplus, and hence the resulting capital inflow or outflow. Further disaggregation of each of the variables in [3] may then be carried out by consolidating the various multinational sources of data. For instance N and capital inflows may be disaggregated using the balance of payment (BOP) statistics and public savings and investment may be derived using government financial statistics (Alemavehu et al. 1992: 7-14).

In the reconciliation of data from different sources, the major adjustment lies in dealing with the discrepancy between the national accounts estimate for the net factor payment and current transfers from abroad $(S_n - S_d)$, and that derived from the BOP statistics. Depending on what assumption one makes about the accuracy of national accounts data vis-à-vis the balance of payments, there are two options in dealing with this discrepancy. If one assumes that the N computed from national accounts is accurate, then this value can be imposed on the BOP statistics. The discrepancy on the BOP may then be accounted for in any variable deemed residual or assumed to be less accurately recorded. A database based on such an adjustment mechanism is reported in Alemayehu et al. (1992).

The second alternative is to assume that the net factor payment and current transfer computed from balance of payment is correct. In such a situation, the net factor payment and current transfer obtained from the balance of payments may be imposed upon domestic savings, in order to arrive at national savings. Assuming further that savings in the pub-

[1]

lic sector are relatively accurately recorded, then private sector savings may be chosen to account for the discrepancy between the net factor payment and current transfer, as computed from national accounts and that of the balance of payment.¹⁹ The choice between the two alternatives should depend on the nature of the analysis to be undertaken, as well as the relative faith one places in these two sources.

In relation to this study, the most relevant data is that relating to balance of payments. This is a logical consequence of the fact that the focus of this study is on international finance and trade. As a consequence, the adjustment is undertaken by imposing the net factor payment and current transfer obtained from the BOP (which is assumed to be correct) onto the national account data. Private saving is then taken to be the adjusting variable. This choice is based on two assumptions. Firstly, although, in principle, domestic saving may be set as an adjusting variable, in fact, it is more logical to choose national savings for this purpose, since the problem arises from net factor payment and current transfer, which is related more to national than to domestic savings. Secondly, among the components of the national savings, it is highly probable that public saving in Africa could be recorded more accurately than private ones. This is a consequence of the public sector's relative efficiency in data recording.

A further important dimension of the database is the disaggregation of trade and financial data. Firstly, a disaggregated stock of financial data is set out. That is, the stock data is disaggregated according to whether creditors are bilateral, multilateral, concessional, non-concessional or private (i.e., banks, portfolio and other commercial suppliers). Hence, instead of using direct flows reported in IMF balance of payments statistics, the flow counterpart is derived from the change in stocks. This allows consistency between the stock and the flow data. In relation to trade data, the total exports are disaggregated by the SITC classification chosen for the purpose of the study. Since this classification tallies with the one used by UNCTAD in its Annual Commodity Yearbook, this allows one not only to arrive at aggregated historic data by SITIC, but also to assign UNCTAD-based relevant prices for each SITIC classification. Consistency of total exports, between that reported in the national accounts and the disaggregated UNCTAD data, is maintained by introducing a category of 'other exports' as an adjusting variable. A similar approach may also be employed for imports.

3.4 CONCLUSION

Based on the above discussion, one might reasonably conclude that efforts to develop macroeconomic models for Africa have remained relatively unsophisticated. However, there have been a number of promising developments in depicting the actual macro dynamics of these countries. Further, based on the models examined in this chapter, it would appear that modellers have generally emphasized the supplyconstrained nature of African economies as well as the vulnerability of these economies to external shocks, both positive and negative.

Indeed, this has taken a particular form in relation to the analysis of import compression and the resulting trade-off between capacity growth and utilization on the one hand, and inflation and growth on the other. However, this focus is at the expense of other related concerns, such as the macroeconomic impacts of external finance and, more specifically, the fiscal response and 'Dutch disease' effects. Moreover, it could be argued that the integration of this approach within a monetarist model for inflation would limit its realism. Hence, these and a number of other shortcomings of these approaches will be further examined in Chapter 5, and the findings of this analysis will be used to inform the prototype African model developed later in this book. However, before moving on to examine the impact of external finance, we begin by empirically examining some of its determinants.

4 Determinants of Foreign Exchange Supply

4.1 INTRODUCTION

In previous chapters we examined the trade and external finance problems of Africa from a historical and theoretical perspective. In this, and the subsequent chapter, I will attempt to investigate this issue empirically. The econometric results obtained in these two chapters will then be used in the construction of a North-South model, which will be detailed in Chapter 6. The empirical analysis which follows will be split between two chapters. In this chapter, some determinants of foreign exchange supply to African economies will be examined. In Chapter 5, a number of macroeconomic impacts of such flows will be noted.

The analysis contained in this chapter will be organized as follows. Section 4.2 will report some econometric results obtained using the 'eclectic' approach to FDI. This analysis will attempt to build upon the 'economic, strategic and political self-interest and developmental considerations' arguments for official flows, discussed in Chapter 2.¹ In section 4.3 the determinants of export supply will be examined, since exports represent the other main source of foreign exchange for Africa. (Again, this analysis will build upon the theoretical discussion presented in Chapter 2.) Section 4.4 will aim to further complement this analysis by placing the supply of commodities from Africa within a global commodity-market framework. Finally, section 4.5 will bring the chapter to a close, by attempting to draw out some conclusions, based on the preceding discussion.

4.2 DETERMINANTS OF FINANCIAL FLOWS TO AFRICA: ECONOMETRIC RESULTS

4.2.1 Determinants of Allocation of FDI to Africa

The level of FDI flows to African countries is extremely low. The existing flows are also concentrated within a small number of export enclaves, such as the mining sector. Although FDI does not represent an important financial flow to Africa, there has nevertheless been a tendency for such flows to increase in recent years (Kasekende et al. 1995, Fernandez-Arias & Montiel 1996). Indeed, by examining the pattern that this has taken in the recent past, we may be able to make useful inferences about the future of these flows. Thus, a simple model of FDI determination, based on the major features of the 'eclectic' approach, will now be formally estimated. This estimation is undertaken for the same three African regions, described in previous chapters, yielding the following final form for the estimated model:

$$FDI_{A} = \beta_{0} + \beta_{1} \frac{GDP_{ol}}{GDP_{A}} + \beta_{2} \frac{Ming_{ol}}{Ming_{A}} + \beta_{3} [FDI_{A}]_{l-1} + \beta_{4} Dum_{SAP}$$

where: the subscript *ot* refers to Other South (World Bank definition of South Asia and Latin America) and *A* indicates the African region in question (East & Southern Africa [ESA], North Africa [NA] or West & Central Africa [WCA]).

The first argument (GDP ratio) is included, with the aim of showing the relative size of the market. The higher this ratio, the lower the relative size of African markets and hence, the lower the level of FDI flows to these markets. The second argument (Ming) gives the ratio of the value added of the mining sector. The lagged level of FDI aims to show the historical pattern of FDI flows (taking account, for example, of colonial influences). This is inferred from the error correction part in the Error Correction Model (ECM). Before arriving at this parsimonial form, I have experimented with similar ratios for imports, as representing an alternative indicator of market size, but have not found this to be statistically significant.

The ratios used as regressors are computed from their respective index (1990=100). The data is obtained from World Tables (1994, electronic) as well as the Inter-American Development Bank's 'Economic and Social Progress in Latin America (ESPLA)' (various years). All of the variables in each of the equations are found to be an I(1) series using the ADF test at a 5 per cent significance level. A co-integration test is conducted for the ESA and WCA regions. The null hypothesis of no co-integration is rejected. Denoting the log of FDI by I, the log of the GDP ratio by Y, the log of the mining ratio by M, and the dummy by Dum_{SAP} , the result of the Error Correction Model (ECM) for the ESA and WCA regions, and the OLS for the NA region is given below.²

East and Southern Africa (ESA)

 $\Delta I = 0.73 Dum_{SAP} - 5.5 \Delta Y - 1.24 \Delta M - (1.9)^{+} (-2.6)^{*} (-0.9)$ [4.1-ESA] $0.99(I_{t-1} + 1.65Y_{t-2} + 1.51M_{t-2})$ (-4)* (-1.95)⁺ (-1.29) $R^{2}-adj = 0.45 \qquad \text{Jarque-Bera } 4.5^{+}$ Reset 1.76(0.21) LM 1.3(0.26) $BG=2.4(0.12) \qquad \text{Sample } 1970-92$

Notes:

Values in bracket are *t*-values, while for diagnostic test *P*-values are given (see Appendix 4.1).

* implies 1%. + 5% and ^10% level of significance.

LM is a hetroscedasticity test while BG is the Breuch-Godfery test for serial correlation.

The 5% (2 degrees of freedom) value for Jarque-Bera is 5.99.

North Africa (NA)

 $I = -0.656Y_{t-1} - 1.15M_{t-1} + 0.711I_{t-1}$ (-1.5) (-1.8)^{\lambda} (2.8)* $R^{2}\text{-adj} = 0.61 \qquad \text{Jarque-Bera } 2.4^{+}$ Reset 0.01(0.92) LM 0.14(0.71)
BG = 0.09(0.77) \qquad \text{Sample 1970-92 (of which only 11 observations are used)}

Other notes as given in equation [4.1-ESA].

West and Central Africa (WCA)

 $\Delta I = 1.08\Delta Y - 0.078\Delta M - (0.98) \quad (-0.08) \quad [4.3-WCA]$ $0.816(I_{t-1} + 0.237Y_{t-2} + 0.107M_{t-2}) \quad (-3.2)^{*} \quad (-1.9)^{+} \quad (-0.29)$ $R^{2}\text{-adj} = 0.31 \qquad \text{Jarque-Bera } 4.7^{+}$ Reset 0.01(0.92) LM 0.30 (0.60) $BG = 4(0.07) \qquad \text{Sample } 1970-92$ Other notes as given in equation [4.1-ESA].

However, these results throw up a number of problems, the major one relating to the absence of a sufficiently long time series, which severely limits the ECM model adopted. Nevertheless, the results remain better than might be arrived at through a simple OLS-based estimation. This is particularly the case since such OLS estimations might, in fact, be spurious, as can be inferred from the non-stationarity observed here.

For equations relating to the ESA region, we have experimented with a number of different estimations. Of these, the result given above is found to be relatively the best. Based on this equation, the GDP ratio for the ESA region failed to show a long-run relationship, using a standard co-integration analysis. However, if the GDP *per capita* figure is used instead, then such a relationship may be identified. A further important implication arising from this analysis of the ESA region is that adjustment towards equilibrium levels of FDI flow is instantaneous (99 per cent). This might be explained by the extreme enclave nature of FDI within this region, investment in mining representing a case in point. Finally, it is worth noting that only within this region is the structural adjustment dummy found to be important.

The estimation for the NA region is the most difficult and the least reliable. Firstly, the sample is very small, which limits the scope for use of a co-integration analysis. However, it is essential to undertake such an analysis, since the series are I(1). Therefore, the estimation is based on a simple OLS, which may suffer from spurious regression unless the assumed, untested co-integration is true. This basically hinges upon an assumption that a long-term relationship would be identified if it were possible to run a co-integration test. In contrast to the other two regions

- where net FDI is found to be equivalent to FDI within the home country, coming in the form of credit – levels of FDI within the NA region are, in fact, found to be negative. This is entirely plausible, since the region is an oil exporter and consequently is in possession of resources which may be invested elsewhere. Hence, FDI within the home country (or credit) is used as a dependent variable. In contrast, the estimation for the WCA region is comparatively more sound. Nevertheless, the length of available time series remains short and so, at best, should be taken as representing only a medium-term approximation. The structural adjustment dummy is not found to be statistically significant, and hence is dropped from the estimation for this region.

In all cases, possible multicolinearity due to the correlation between mineral value added and GDP ratios are checked. It is found that, except within the NA region, the degree of multicolinearity is reasonably low, the simple correlation coefficient between the two being 0.34, 0.77 and 0.32 for ESA, NA and WCA, respectively. In general, the results summarized in this section suggest that relative market size, mining activity and the historical pattern of FDI together determine the flow of FDI to Africa, as well as its allocation among Southern nations. Although this is not a robust result, since such flows to Africa are extremely small, the limitations of the above equations may nevertheless be tolerated. In the following section, equations will be estimated for official aid, since this represents the most important type of capital flow to Africa.

4.2.2 Determinants of Allocation of Official Capital Flows to Africa

This section details some econometric results relating to the determinants of the allocation of official capital flows³ to Africa. Data relating to such flows are generated by examining the change in the stock of such debt. This derivation has the advantage of making stocks and flows consistent. On the other hand, debt restructuring may entail overstatement of such flows by the amount of what are termed 'inferred flows'. These inferred flows – which are to be distinguished from cash flows – may result, say, from principal arrears. However, up to the mid-1980s, rescheduling of principal was negligible in most African countries.⁴ To a degree, this overstatement may also be offset by debt forgiveness. The remaining data is obtained from the World Bank's World Tables and World Debt Tables (1994, electronic) as well as from the Inter-American Development Bank's ESPLA dataset (various years).

A simple model of the determinants of allocation of official capital flows between Africa and the rest of the South (defined here, in keeping with the World Bank definition, as South Asia and Latin America) is given below. The specification is based on the theoretical discussion presented in Chapter 2. Thus, official flows to Africa are assumed to be determined by economic, political and strategic self-interest arguments, as well as by developmental and humanitarian considerations. Unlike other studies, these arguments are explicitly defined, in order to indicate Africa's position relative to other South regions. The estimation is then undertaken for the three regions in Africa, yielding the following final general form of the estimated model:

$$Fbm_{A} = \beta_{0} + \beta_{1} \frac{M_{ol}}{M_{A}} + \beta_{2} \frac{PY_{ol}}{PY_{A}} + \beta_{3} \frac{FDI_{ol}}{FDI_{A}} + \beta_{4} \frac{DSR_{ol}}{DSR_{A}} + \beta_{5} Dum_{SAF}$$

where: *Fbm* is bilateral and multilateral flows to Africa; M is import; *PY* per capita income; *FDI* is foreign direct investment; *DSR* is the debt service ratio and *Dum_{SAP}* is the structural adjustment dummy active since 1985. The subscripts *ot* and *A* are as defined before.

The import ratio (denoted below as M) and the FDI ratio (denoted below as I) are assumed to support the economic self-interest argument. The assumption behind the former argument is that most imports come from the North. Thus, aid inflows from the North are geared towards strengthening the export sector of the supplying countries. The FDI ratio is also used, based on the same line of thinking. The relative share of the mining sector in GDP of Africa (MIN_A) vis-à-vis other regions in the South (MIN_{OT}) is also used. This choice is made based on the assumption that most mining sectors in South are dominated by Northern companies, which are likely to shape the flow of resources from the 'mother' country. A related economic self-interest argument is the debt service ratio (denoted below as D), which may also indicate repayment capacity of borrowers. The dummy (Dum_{SAP}) shows the structural adjustment-related flows, which may also be interpreted as the political/strategic self-interest argument. This takes the value 1 if SAP policies are being applied in the country in question and zero if they are not. The developmental/humanitarian argument (denoted below as PY) is assumed to reflect the ratio of per-capita income in other regions in

the South relative to that of Africa.⁵ The *Fbm* is denoted by *F*. All parameters, except β_0 , β_2 and β_4 , are expected to be negative, since a higher ratio is the result of a higher relative value of the numerator. The latter is hypothesized to imply lower levels of flows to Africa.

As noted in Chapter 2, a further important feature of official capital flows to Africa relates to the fact that the budgetary considerations in the North will have an influence on such flows. Indeed, because proper evaluation of aid's impact from the donor's perspective is difficult, at least in the short run, the underlying determinants of budgetary allocation for the previous period may well influence the current budgetary process, with due consideration given to making corrections for previous mistakes. This is captured by the ECM model, which also tackles the non-stationarity problem of the series adopted.

In common with the FDI equations, the estimation which follows should, at best, be taken as a medium-term indicator, since the relevant time series (1970–92) is too short to undertake a fully fledged co-integration analysis. Nevertheless, this estimation is comparatively better than existing empirical studies, which are based either on a time-series estimation, without a formal test for non-stationarity (see the survey by McGillivray & White 1993), or on a cross-sectional analysis only (see Anyadike-Danes & Anyadike-Danes 1992). I have identified that most series are I(1). This should cast doubt on simple OLS time-series studies. Through an examination of a simple correlation coefficient, the degree of multicolinearity is also found to be low for all estimations. Thus, following an exhaustive search process, the best estimated equations for the three regions which are the focus of this study are given below. These equations have also been tested for co-integration and a null hypothesis of no co-integration is rejected for all.

East and Southern Africa (ESA)

$$\Delta F = 0.612 \Delta M_{t-1} - 0.136 \Delta I_{t-1} - (0.36) \quad (-0.67) \quad [4.4 - ESA]$$

$$0.89(F_{t-1} + 2.138 M_{t-2} + 0.512 I_{t-2} - 1.972) \quad (-3.6)^{*} \quad (-2.5)^{*} \quad (-1.72)^{\wedge} \quad (2.5)^{*}$$

 $\label{eq:R2-adj} \begin{array}{ll} R^2 \text{-adj} = 0.32 & Jarque-Bera \ 0.3^* \\ Reset \ 0.01(0.93) & LM \ 0.26(0.62) \\ BG = 1.2(0.30) & Sample \ 1970-92 \\ Notes as defined under equation \ [4.1-ESA]. \end{array}$

In the estimation for equation [4.4-ESA] I have experimented with the ratio of value added from the mining sector as well as a structural adjustment dummy. However, neither of these was found to be statistically significant, and so were excluded. The debt service ratio is also found to be highly correlated with the import ratio, having a simple correlation coefficient of -0.70. Hence, its inclusion would result in multicolinearity. However, this ratio has been omitted since an alternative equation to [4.4-ESA], using the debt service rather than the import ratio, performed comparatively less well, having a lower level of significance. Two other estimations, which separately use bilateral and multilateral flows as dependent variables, were also attempted. Signs remained unchanged from the aggregate form. However, the potency of the parameters decreased when we used bilateral flows and increased with multilateral flows. The estimation of [4.4-ESA], based on aggregate flows, yields an average value and hence, this estimation has been chosen.

In common with the FDI equation, the interpretation of this equation is relatively straightforward (see Appendix 4.1). Thus, if we take the first ECM, the 'long-run' co-integration relationship is given by,

 $F_t = 1.972 - 2.138M_{t-1} - 0.512I_{t-1}$

89 per cent of any deviation from this long-run equilibrium in the previous period is made up for in the current period. The short-run coefficients are not found to be statistically significant. However, in the long run, the relative (to other South) level of imports and FDI flows within the ESA region determine the official flows to that region. Of these, the import ratio has relatively the more important influence. Moreover, within this region, humanitarian and structural adjustment related arguments are not found to be statistically significant. The statistical significance of the adjustment coefficient may, in all cases, be taken as showing the importance of the past budgetary practice of donors in influencing current allocations.
North Africa (NA)

 $F = 1.05 - 2.05M_{t-1} - 0.133D_{t-1}$ (6.0) * (-5.9) * (-0.38) $R^{2}-adj = 0.64$ F = 1.064 F = 1.1 + 1.1 F = 1.064 F = 1.3(0.27) F = 1.1 + 1.1 F = 1.1 + 1.1

The individual series within the NA region are found to be stationary. Hence, a simple OLS estimation is made. Per capita income and debt service ratios are found to be statistically insignificant. Indeed, a further estimation, using bilateral and multilateral flows, separately, yields a nearly identical result.

West and Central Africa (WCA)

$$\begin{split} \Delta F &= 2.02 Dum_{SAP} - 1.564 \Delta M_{t-1} + 3.135 \Delta PY_{t-1} - \\ &(1.84)^{\wedge} &(-1.11) &(1.16) \\ &0.86(F_{t-1} + 2.657 M_{t-2} + 1.319 PY_{t-2} - 2.591) \\ &(1.9)^{\wedge} &(-2.1)^{+} &(1.01) &(-1.01) \\ R^{2}\text{-adj} &= 0.17 & \text{Jarque-Bera } 3.2^{\wedge} \\ \text{Reset } 3.91(0.07) & \text{LM } 0.00(0.97) \\ \text{BG} &= 4(0.07) & \text{Sample } 1972-92 \\ \text{Notes as defined under equation } [4.1\text{-ESA}]. \end{split}$$

Within the WCA region the individual series are I(1). The estimation [4.6-WCA] is undertaken once the null hypothesis of no co-integration has been rejected. The variables I, YP and D are found not to be statistically significant. Neither is the per-capita ratio found to be significant. Hence this ratio could be excluded from the estimation. However, since its *t*-value is closer to one, it is nevertheless included within the estimation. Neither does the separate use of bilateral and multilateral flows significantly change the result. In fact, the inclusion of these flows actually improves the RESET value, based on a specification test, to an acceptable level of around 30 per cent, although some serial correlation problems are exhibited. In general, equation [4.6-WCA], which summarizes aggregate official flows, is found to be preferable when all properties are taken into account.

To sum up, this section has reported on some econometric results relating to the determinants of the allocation of both FDI and official flows to Africa. Although the results of this analysis are not robust, because of the relatively short available time series, nevertheless they do provide a useful second-best indicator of the determinants of the allocation of flows. In any case, the use of this estimation, no matter how flawed, is preferable to using spurious results arising from existing studies. Besides, a regionally-based estimation for Africa is not available. In relation to the actual results of these estimations, these may be summed up as follows. Firstly, there exists a long-run equilibrium relationship between Africa's relative economic performance and flows of capital to that continent. Secondly, humanitarian and developmental considerations are found to be largely negligible in influencing such flows. This result would tend to lend support to the apparently obvious notion that capital flows are associated with the level of development, and particularly involvement in trade. Indeed, this remains true even when these flows chiefly comprise aid.

4.3 DETERMINANTS OF AFRICAN EXPORT SUPPLY

4.3.1 Introduction

Much of the analysis relating to the export problems faced by producers of primary commodities in developing countries would lead one to conclude that the problems faced by these producers are mainly pricerelated in nature. More specifically, overvaluation of the exchange rate and marketing board intervention are found to represent particular problems for these producers. Although the literature on commodity export supply functions begins by formulating structural equations which accommodate other factors, the reduced form of these equations, which is used for estimation purposes, is characterized only by explanatory variables for current or lagged (relative) prices. In this section, an attempt is made to underline the existence of other equally, or even more important, factors, which will influence the export of primary commodities. These factors will then be considered explicitly within the estimation. This is undertaken for two fundamental reasons. Firstly, exclusion of variables can only be interpreted as an omission, and hence, represents poor econometric practice. Secondly, such omission

has the effect of depriving countries of policy handles other than price. Thus, the analysis which follows will first set about exploring pricefocused estimations. Subsequent estimations will then be specified by adding other relevant explanatory variables, which are rare in the literature.

Thus, the discussion will be organized as follows. Section 4.3.2 will attempt to build upon the theoretical discussion presented in Chapter 2, by reviewing some literature relating to the supply of commodities. This discussion will be useful in developing the modelling approach set out later in this study. In section 4.3.3 a number of salient points relating to the nature of the data used in this study will be highlighted. Section 4.3.4 details the estimation of the aggregate export supply functions. Finally, in section 4.3.5 a number of conclusions, arising out of the preceding discussion, will be outlined.

4.3.2 The Literature

A number of studies have attempted to answer the question of what determines the supply of primary commodity exports. Broadly, a range of such factors has been identified. These include cost and accessibility of consumer goods, farm subsidies and taxes, research and extension, the existence and quality of road infrastructure as well as the availability of services such as marketing and credit (Binswanger 1992). Naturally, these various factors will not operate in isolation. Thus, the quality of road infrastructure is likely to affect the availability of services, while the availability of such services as credit⁶ may constrain the effective demand for fertilizer. Agro-climatic conditions and considerations of human capital, such as rural population density and literacy rates, are also likely to represent important factors having an influence on the export supply of primary commodities. Further, these studies show that the short-run supply elasticities for primary commodities are generally low, simply because the basic factors of production (land, labour and capital) remain fixed in the short run. The latter constitute 70-85 per cent of the cost of agricultural production (Binswanger 1992: 151). Attempts to specify supply functions represent a logical outcome arising from this broad idea of supply-determining factors. However, most writers end up specifying supply equations in terms of relative prices. In general, these studies may be categorized under two themes. Firstly, price-focused models, which use prices of different complexity as explanatory factors, and secondly, mixed (heterodox) factor-based models.

Price-focused models

Price-focused supply models began from the simple 'cobweb theorem' of Ezekiel (1938), which states that output is determined by the level of price in the previous period. Over time, this basic theorem has been developed upon. Thus, Nerlove (1958) modifies the assumption relating to previous period prices. He maintains that producers are influenced by their perception of 'normal' price, which could be captured by the adaptive expectation scheme developed by Cagan (1956). In terms of the classification set out in Chapter 2, these fall under the category of econometric market models. However, this original formulation of a supply and price relationship has undergone numerous changes. Thus, a number of studies emphasize the distinction between the long-run (potential supply) and the short-run (a proportion of potential supply) responses (Wickens & Greenfield 1973, Chu & Morrison 1986). For instance, as outlined in Chapter 2, Chu and Morrison (1986) define the structural equations of supply as the sum of utilization of potential output (i.e., the utilization rate approach) and potential output (i.e., the potential supply approach). However, in the final form of this model. supply is specified (in reduced form) as a function of current and lagged prices, exchange rate and a supply shock indicator. Indeed, a similar approach is used in the earlier work of Wickens and Greenfield (1973). Such a classification is typically used for perennial crops and mineral extraction.

Other studies have focused on the optimization strategy of agents under a range of different assumptions. These strategies may be classified, according to whether they are based on: (1) maximization of average return per unit of cost (Gray 1914); (2) optimizing the amount of total deposit to be exploited, given first decreasing and then increasing cost structure over the expected life of the mine (Carlisle 1954: 3), maximizing the present value of discounted future net profit (Hotelling 1931); (4) optimizing expected profit under free competition, at both a firm and industry level (Herfindahl 1955); and finally, (5) relating higher short-term rate of extraction under uncertainty with a higher interest rate (Parish 1938). In terms of the theoretical classification set out in Chapter 2, these models fall broadly under the category of econometric optimization models. Again, this type of model is widely used in explaining mineral extraction.

Following Shu (1975) and Askari and Cummings (1976), supply response functions may be categorized according to whether they relate to annual crops, marketed surplus or perennial crops. For annual crops, six different models may be fitted: (1) a simple Koyck distribution lag or Nerlovian expectation model, (2) a complex Nerlovian expectation model, (3) a Kovck second-order lag model, (4) a Nerlovian adjustment model, (5) an expectations-adjustment model, and (6) the simple model (Shu 1975: 27). The Kovck model uses lagged prices (with a geometric lag assumption between these prices) as the only explanatory variable. In Nerlove this is replaced by expected prices. The complex form of the Nerlove model adds to this other expected values, such as expected vield. Kovck's second-order lag function uses a lagged dependent as a regressor, under the assumption of slow response due to institutional factors. Similarly, the Nerlovian adjustment model employs a lagged dependent variable by assuming that farmers adjust by learning from their past expectation mistakes. On the other hand, the simple model, which includes neither adjustment nor expectation variables, usually includes lagged price, lagged yield and trend variables to capture 'other' factors. Estimation is invariably undertaken using simple OLS. Indeed, the theoretical aspect of these models is ahead of the relevant estimation techniques, which accommodates their basic idea of adjustment to past disequilibrum (the relevance of the error correction model in this context is discussed below).

The first attempt to model supply response in perennial crops is Bateman's (1965) model of cocoa supply in Ghana. The model uses expected producer prices for cocoa and a competing crop, such as coffee. These prices are used since it is believed that they show expected profit as regressors. Other non-price factors are ignored completely. Within this model, the dependent variable is the additional acreage under the planted crop for each year in question (Shu 1975: 55–69, Askari & Cummings 1976: Ch. 7). Other models, such as those developed by Beherman, French⁷ and Matthew (detailed in Shu 1975 and Askari & Cummings 1976), are fundamentally no different in terms of the explanatory variables used. A similar approach to Bateman's is to use the stock of trees rather than acreage under planted crops as the dependent variable. Differentiating such a specification yields equations similar to that of Bateman except for the fact that the explanatory variable, lagged planted trees, is replaced by lagged stock of planted trees.⁸ The latter models, although fundamentally price-based, show a trend towards the inclusion of other factors in specifying the supply function. However, in general, price-focused models use functional forms, which either explicitly exclude non-price factors, or structural equations which include other non-price factors but which ultimately can be explained by price. The end result in both cases is a price-based estimation.

Mixed (heterodox) factor-based models

The Ady (1968) model, which was developed for perennial crops in Ghana, Nigeria and Uganda, represents an improvement on Bateman's model in that, in addition to price, it includes the existing acreage (i.e., the stock) of crop for the previous period. Another strand of supply function modelling which focuses on heterodox factors is what is known as the 'liquidity model'. This model includes farmers' incomes as an additional explanatory variable indicating his or her capacity to invest. However, in all other respects, this model is similar to Bateman's. Its basic feature is an attempt to relate investment to the difference between desired and actual level of capital. Wickens & Greenfield (1973), Palaskas (1986) and Chu & Morrison (1986) summarize such models by characterizing them as essentially being based on capital and investment behavior theory, as presented in the Nerlovian adjustment model.⁹ Alternative forms of this theory arise in specifying the various factors, which determine the desired level of capital stock. These factors include capacity utilization (in capacity utilization theory), net output or returns to capital (based on a neo-classical approach), internal cash flow (in liquidity theory) and, finally, an expected profit-based approach. A different degree of emphasis is placed on each of these factors by different authors (Palaskas 1986: 16-18, Chu & Morrison 1986: 142-43, Wickens & Greenfield 1973). In some studies, supply is also considered as a function of expected price, expected opportunity cost. production costs, stock of output (trees, and specifically perennial), potential output of the industry and tax considerations (Kalaitzandonakes et al. 1992). In terms of our classification, set out in Chapter 2, such models represent a hybrid of process and market models. Finally, a further important explanatory variable, reported in the literature, is domestic demand (Pal 1992).

As discussed in Chapter 2, based on the original works of Goldstein and Khan (1978), Bond (1987) uses exchange rate, current and lagged

price, an index of productivity, a vaguely defined supply shock factor and a time trend to estimate her model. These are far fewer than the supply factors mentioned in the text of her discussion, indicating either the difficulty of quantification or other data problems. In Ramanujam and Vines (1990), supply is specified as a function of current price, a lagged series of past prices, taking account of the cost of adjusting output, and exogenous factors to capture 'others'.¹⁰ This general form is estimated for different commodity groups. Hwa (1985) emphasizes the understanding of the effect of supply and demand through stock demand and supply conditions. Palaskas and Gilbert (1990) have reviewed Haw's model and argue that storage disequilibrium is an implausible basis for a price adjustment theory. Firstly, as they note, correct specification/ estimation of Haw's model shows that the model performs poorly. Secondly, the model essentially relates price changes to the disturbance term within the stock demand equation, which, for them, represents the least important source of price variation (Palaskas & Gilbert 1990: 1424). However, Palaskas and Gilbert's review may be criticized in that it focuses almost exclusively on Haw's reduced form equation, while neglecting his main structural equations, in which supply and demand are allowed to vary.

Later developments in commodity supply modelling do not differ markedly from the principles set out in the above discussion. Thus, the emphasis remains on relative prices, expectations and types of equilibrium across different commodity groups.¹¹ In other words, it is hard to find well-elaborated supply functions. Indeed, a recent supply management study for tropical beverages (Maizels et al. 1992, 1993a, 1993b) essentially continues the tradition of using lagged prices and sometimes output. The study, which sets out to explain coffee, tea and cocoa supply management, uses a number of essential arguments to explain output. These include: (1) area harvested, which is believed to reflect maximum potential output; (2) a real producer price; (3) a lagged output, included to reflect relations between successive years; and, (4) a time trend in order to allow for trends in productivity. An important result in this study is that, in every case, area is correlated with output.¹² The authors also use acreage equations, which relate the area harvested to lagged prices, in order to arrive at a long-run investment equation.

Most of the other studies undertaken in relation to African exports have followed a similar approach, with supply response studied in terms of current and lagged prices. The finding for small African countries, particularly during the 1960s and 1970s, is that short-run elasticities are high for annual crops, while long-run elasticities are high for tree crops and minerals (Rwegasira 1984: 7–9). On the other hand, there has not been a great deal of recent work on aggregate commodity export functions. Indeed, a recent paper which assesses the export performance of sub-Saharan African countries calls for a disaggregated estimation of export supply functions, in order that light might be shed on factors influencing export volumes (Svedberg 1990: 32). However, work on specifying individual commodity supply function estimations is advancing (Gwyer 1971, Alibaruho 1974, Ghoshal 1974, Jones & Mutuura 1989, Eriksson 1993 and Dercon 1993 all represent good examples of such work)

The emphasis on supply in the above discussion need not imply that commodity models which explore the demand side – and hence its simultaneous determination – do not also exist. Indeed, such models not only avoid the simultaneous equation bias, which could arise from neglecting demand, but also challenge the small-country assumption, when constructed at a global level. Hence, this issue will be taken up again in section 4.4.

The link between supply theories and estimation technique: The relevance of the Error Correction Model (ECM)

Three features of the commodity supply theories discussed above may be linked to the ECM approach. These are: (1) the assumption that there exists a long-run relationship between the variables under study; (2) the hypothesis that this might have fairly distinct short- and longrun features, and (3) the assumed existence of some sort of in-built adjustment mechanism within this relationship. The first of these features would tend to suggest the need to run a co-integration test, since most series are I(1). In relation to the second of these features, the literature does not come out clearly one way or another. Thus, some writers take the short/long run as being synonymous with short/long lags. However, such an approach fails dismally, particularly in relation to short gestation commodities such as food and agricultural raw materials. In contrast, other studies rightly emphasize that the short run should refer to mean utilization of potential capacity, and the long run to an increase in potential output. However, since few of these studies use a relevant estimation technique,¹³ this type of study is not without its problems. The error correction model is a formal representation of dependent and independent variables, with explicit distinction being made between short-run variations (i.e., the immediate impact effect) and long-run aspects (associated with long-run level or a steady state relationship). Thus, it represents an appropriate technique for estimating supply functions based, as it is, on a distinction between the theory of potential, and utilization of potential, output. Moreover, since it explicitly includes an adjustment mechanism, an ECM approach will tend to incorporate the third feature of the theories reviewed above.

The theoretical model of this study

The above review provides insight into some important points in specifying export functions of primary commodities. Firstly, factors other than price are found to be important determinants of commodity supply. However, either lack of data or difficulty in quantifying this data, as well as having to focus on reduced forms, forces many researchers to use prices as the only explanatory variables. Secondly, a distinction across commodities, especially between annual and perennial crops, is essential in specifying supply functions. A third and relatively neglected point is the need to place the commodity market within a macro framework where the role of stockholding and the impact of macroeconomic variables is likely to be important.¹⁴

In this study, the export supply equation of a typical African economy is specified, in order to depict the behaviour of commodity producers, mediated through the government. We have further assumed that output of export commodities and exports respond to world prices in a broadly identical manner.¹⁵ Although government intervention within the export sectors of most African countries is obvious, it is assumed that the impact of world price, and especially the degree to which this price *changes*, will send similar signals both to the public sector in its capacity as a producer, and to individual producers. However, it is worth noting that individual producers are also likely to surrender a proportion of their income to government.

The response to price and other supply factors is likely to take two forms. In the short run, increased capacity utilization is important. Thus, *short-run* parts of the argument would be based on the commodity model of Goldstein and Khan (1978), Chu and Morrison (1986) and Hwa (1985). Here, *(latent) capacity utilization theory* is the underlying hypothesis. In the long run, producers are assumed to respond through change in potential output, or, in other words, through capacity creation. *Expected profitability theory*, following Chu and Morrison (1986),¹⁶ and other supply-inducing factors, such as capital formation and foreign inflow, are believed to explain this. However, foreign inflow may also result in 'Dutch disease' effects (see Chapter 5 for a discussion of this phenomenon). However, this remains largely an empirical question. Thus, I am essentially arguing that the different factors emphasized by different authors are basically complementary and hence, that these should explicitly be considered within any estimation. The lag structure will vary, depending on the nature of the commodity under consideration. For example, a longer lag structure should be used for beverages and minerals¹⁷ than for annual crops.¹⁸ This yields the following general theoretical model:

$$X_{si}^{ss} = a_0 + a_1 \left(\frac{ep_s}{p_d}\right) + a_2 \left(\frac{ep_s}{p_d}\right)_{t-1} +$$
[1]

Capacity utilization

$$a_{3}\left[\left(\frac{ep_{s}}{p_{d}}\right)_{i-i}-k^{-1}\sum_{i=1}^{k}\left(\frac{ep_{s}}{p_{d}}\right)_{i-i}\right]+$$

Expected profit

Expected profit

$$a_4(\Delta K)_{t-i} + a_5 FF_{t-i}$$

Capital formation

where: X is export supply, e is exchange rate, P_s is the export price of South, P_d is the domestic price, FF is foreign inflow, ΔK is the capital formation indicator, the term in brackets [...] is the expected profit indicator, the deviation of current price from k years moving average.

The theoretical export supply function and the data

In this section, the theoretical commodity export function in equation [1] is adjusted to fit the available data. In our database, investment by commodity sector is not available. However, the reparametrization of this theoretical formulation renders this possible. The following sets of equations are used for this purpose. The distinction between the short-run *capacity utilization* theory, and the long-run *capital formation* ar-

gument will be maintained. To avoid the complex variables in equation [1], we designate the eP_s/P_d by P and the expected profit indicator by Π^{σ} . Maintaining ΔK and FF as they are, a shorter version of equation [1] may be written as,

$$X_{Ai}^{ss} = \alpha_0 + \alpha_1 P_i + \alpha_2 P_{i-i} + \alpha_3 \Pi^{\sigma} + \alpha_4 \Delta K + \alpha_5 FF \qquad [2]$$

Our capital formation data refers to gross fixed capital formation, both public and private, at a national level. To overcome the lack of sectoral disaggregation of this data, the capital stock is specified in terms of factors that affect its formation. The possible factors, about which we have access to data, include prices, an indicator of expected profit and instability of prices (Π^{σ}), supply of domestic credit (*DCR*), supply of fertilizer, government investment $(I_g, indicator of infrastruc$ ture) and capital inflow (FF).¹⁹ Of these, factor prices are omitted, since they are specified in the above equation. This omission might be taken as implying that the lagged price effect for the current and certain other periods is direct. In other words, that this does not occur through capital formation. However, since the indicator of expected profit and instability is computed from lagged prices the effect is indirectly captured. In relation to the expected profit indicator, the ideal approach would be to use a measure of direct profit. Nevertheless, such profit data is hard to come by. Its approximation as a deviation from three-year movingaverage prices makes it similar to the price instability indicator. Hence, in specifying capital formation, it is included as one of the factors within the above equation. Its coefficient may then be interpreted both as a profit and price instability index. Thus, the capital stock may formally be given as,

$$\Delta K = \beta_0 + \beta_1 \Pi^{\sigma} + \beta_2 DCR + \beta_3 I_{\sigma} + \beta_4 FF$$
[3]

where: DCR is domestic credit, FF is capital inflow, I_g is government investment (GDI).

Equation [3] states that capital formation in the commodity sector is determined by prices, availability of domestic credit, public infrastructure provision and foreign inflow. Substitution of this capital formation equation into equation [2] yields the final adjusted equation for inclusion in the estimation,

$$X_{Ai}^{ss} = \phi_0 + \alpha_1 P_i + \alpha_2 P_{i-i} + \phi_1 \Pi^{\sigma} + \phi_2 DCR + \phi_3 I_g + \phi_4 FF$$
^[4]

where:

$$\phi_0 = \alpha_0 + \alpha_4 \beta_0$$
; $\phi_1 = \alpha_3 + \alpha_4 \beta_1$; $\phi_1 = \alpha_4 \beta_i$; $\phi_4 = \alpha_4 \beta_4 + \alpha_5$

At estimation stage, the linear formulation is transformed into a loglog form. This implies that the estimated coefficients are elasticities. However, specifically in relation to our own data, the expected profit indicator is found to be empirically problematic. In all cases, it is a stationary series and exhibits a very high correlation (usually greater than 0.80) with the price variables. Given the latter result, the former is entirely logical, since the moving average of profits is similar to a lagged price. Hence the series generated is nearly identical to the first difference of prices, which is stationary. This represents a serious multicolinearity problem and, therefore, may be interpreted as a serious shortcoming of previous studies. For this reason, in the estimation given below it is omitted from the functional form (see Appendix 4.2 for a discussion on these variable and estimation results). Finally, the foreign inflow variable (FF) might also result in a simultaneous equation bias²⁰ within one of the estimations, since it will affect the other capital formation indicators within the regression. However, this is likely not to represent a serious problem, since a very small proportion of foreign inflow is usually allocated to the primary sectors.

4.3.3 The Data

The relevant macro data used in this study is derived from a consistent macroeconomic database discussed in Chapter 3. Exports of commodities and price are compiled from the UNCTAD database (Annual Commodity Yearbook). However, certain changes are introduced, in order that this data may be adapted to the specificity of Africa.

- 1. Similar commodity categories are used for four commodity groups. These are:
 - (a) Agricultural raw materials (excluding synthetic) SITIC 2-22-27-28-233-244-266-277
 - (b) Tropical beverages SITIC 071.1+072+074.1
 - (c) Minerals, ores and metals SITIC 27+28+68+522.56
 - (d) Fuels SITIC 3.

129

2. The category *food* adopted in this study is slightly different from that used by UNCTAD. Thus, within the UNCTAD methodology, the category *all food* includes *food, tropical beverages, and vegetable oil seeds and oils.*²¹ However, in this study *food* is defined simply as all food items with the exception of tropical beverages.

Prices (export prices)

For each of the commodities detailed above, UNCTAD provides a price series running from 1970 onwards. These price indices are computed using the total developing country exports of these commodities for the years 1984–86 as weights. However, since no index is provided for the category *food*, as defined within our study, we have generated such an index by summing together prices from the UNCTAD *vegetable oilseeds and oils* and *food* categories, and weighted their respective prices based on their share of total developing country commodity exports.

A further major change which we make to UNCTAD's price series relates to the weights used in the construction of this series. Such changes are likely to be necessary since the weighting used by UNC-TAD may not be relevant for this study, in particular, and for Africa, in general. In other words, one might question whether the averaging method employed accurately reflects the world prices facing African economies. In order to answer this question the UNCTAD price is recalculated by weighting it by the level of each region's exports. This comparative analysis offers the following conclusions. Firstly, for Africa as a whole, the food price index is similar both in the UNCTAD study and in this study. For the NA region there was a difference between these two price series in 1970s and late 1980s. For the WCA region the UNCTAD series shows higher prices up to the early 1980s, at which time these prices fell into line with the price series used within the present study. For the ESA region, the UNCTAD series understates the regional indices. Secondly, for tropical beverages the UNCTAD price series is nearly identical with the regional price series computed in this study. Thirdly, for agricultural raw materials for Africa as a whole, the UNCTAD series overstates the actual price faced by African economies. This conclusion remains valid in relation to regional prices. Such overestimation is likely to be particularly great in relation to ESA countries. Fourthly, for mineral, ores and metals for the whole of Africa, the UNCTAD series overstates the actual price faced by African

Diagram 4.1 The modified and actual UNCTAD price series (WCA, 1985=100)



Minerals, ores & metals

Agricultural raw materials



Beverage



Food (excluding beverages)



UNCTAD commodity price series

UNCTAD price weighted by export of West and Eastern African region (using UN-WCA classification)

Diagram 4.2 The modified and actual UNCTAD price series (ESA, 1985=100)

Minerals, ores & metals

Beverage



Agricultural raw materials



Food (excluding beverages)



UNCTAD commodity price series

UNCTAD price weighted by export of East and Southern African region (using UN-ESA classification)

Diagram 4.3 The modified and actual UNCTAD price series (NA, 1985=100)



Minerals, ores & metals

Agricultural raw materials



Food (excluding beverages)





UNCTAD price weighted by export of Northern African region (using UN-N classification) economies. With the exception of the 1980s, such overstatement is severe throughout the period. However, for the ESA region, this index understates the price faced during the early 1970s and late 1980s. Finally, a comparison of the African regional price series with the one generated using the sample countries of this study reveals that these are nearly identical for all commodities, with the exception of minerals, for all regions. Diagrams 4.1, 4.2 and 4.3 illustrate these conclusions for the three regions, which are the subject of this study.

This result has profound implications, not only in relation to previous studies in Africa, but also to other studies which use world prices to analyse the impact of such prices at a regional and country level. Obviously elasticities computed from such studies will be biased if there is a variation between the regional and world prices (as listed in UNCTAD or IFS series), even if these have an identical trend.

4.3.4 Estimation

4.3.4.1 Estimation approach

Obviously the econometric specification may differ from this general theoretical specification. Based on recent innovations in time series econometrics, the estimation is, in fact, carried out by formulating an Error Correction Model (ECM; see Appendix 4.1). The estimation is undertaken by pooling the data from the sample countries in each region, for the period 1970–90. However, in order to ensure that the series are not unduly mixed, when a lag structure is used, careful treatment of end points is adopted. A constant country dummy is used in all cases when such a dummy is found to be statistically significant. Moreover, sample sizes are adjusted by excluding a country or countries, when such data violates almost all the relevant diagnostic tests.²² The choice of the variables is made after a search process that includes both diagnostic and co-integration tests. In all cases, the estimation is fully supported by the relevant diagnostic tests.

Preliminary estimations using current prices and capital formation indicators have shaped the approach followed in this section. That is, data has been allowed to inform theory, as per the approach advocated by Wuyts (1992a, 1992b). Such preliminary estimates reveal some interesting results. Firstly, short-run current price elasticities are generally positive, although not always statistically significant. For long gestation commodities, price is found mainly to affect capacity utilization. Secondly, long-run current price elasticities are found to be both positive and negative.²³ In some cases these do not differ significantly from zero. Thirdly, capital formation variables are, in general, positive and significant. Fourthly, within the WCA region, 10 to 25 per cent of any disequilibrum in the previous period is made up for in the current period, for all commodity categories, with the exception of minerals. This may be interpreted as indicating a generally low level of adjustment to disequilibrum or shocks. However, standing at 62 per cent, the adjustment coefficient for the minerals category remains very high. Similar adjustment coefficients, except for minerals, are also obtained for the ESA region. Finally, for the countries of the NA region, the level of these coefficients remains quite high, indicating the likely capacity of these economies to cope with shocks.

The inverse relationship, between exports and long-run current prices is also worth examining further. This relationship may be observed for agricultural raw materials, in all regions, as well as for minerals within the NA region. Based on such an examination, at least two propositions in relation to export supply and price dynamics may reasonably be set out. The first such proposition is that domestic prices may not always be important and, hence, that relative price may be crucial. In the macro context, this implies that domestic prices might be affected by capital inflow. This, in turn, underscores the importance of macro variables in the determination of the supply of commodities, so long as they affect the level of domestic price. Thus, in line with the general focus of this study, this implies the possibility that a sort of 'Dutch disease' effect may occur. This will be discussed at length in the following chapter. However, it is worth noting here that if a 'Dutch disease' effect is present, that a real appreciation in domestic currency is possible. Thus, Tables 4.5, 4.10 and 4.15 summarize information relating specifically to this inflationary pressure. For this purpose, a one-period lag is assumed, with nominal export price, rather than the real exchange rate being used. The rationale for using the nominal export price is that the domestic price is assumed to be affected by foreign inflows.²⁴ The use of debt stock data within an ECM formulation allows one to observe the 'Dutch disease' impact in the short run, since the change in the debt stock may be taken as a flow. This specification also allows one to examine the debt overhang problem in the long run, as well as having the advantage that it explicitly focuses on the impact of foreign inflows.

Examining the relationship between exports and long-run prices vields a second major proposition in relation to African export supply and price dynamics: specifically, that the nature of demand for foreign exchange²⁵ within most African economies results in these countries seeking to maximize their total export revenue in a situation of declining prices, usually by increasing volume of exports.²⁶ Given these working propositions arising out of an exploratory data analysis, the aggregate commodity groupings for each region are then estimated in three stages. Firstly, current price and capital formation indicators are taken as regressors. The results of this estimation, we call Model I. Secondly, the current price is replaced by a commodity-specific real exchange rate (in other words, the ratio of the regional commodity price measured in domestic currency to the domestic price of that commodity). This we call Model II. Finally current prices,²⁷ a capital formation indicator and foreign capital inflow are used as regressors. This is Model III.

For all three models the current price data used is the current commodity price computed for each region. For perennial crops and minerals two sets of lagged prices are used. A one-period lag is assumed to affect capacity utilization while the five-year lag is assumed to affect capacity creation. As a consequence of the ECM approach adopted, both period lags have short- and long-run effects. Variables for consumption of fertilizer, investment and capital stock are used interchangeably²⁸ as indicators of capital formation. Thus, the final estimations are the result of exhaustive search processes using both Hendery's general to specific (GS) approach and relevant diagnostic tests. The actual estimation is undertaken using E-views and TSP. The results of this estimation are discussed in the following section.

4.3.4.2 Results

East and Southern Africa (ESA)

A sample comprising $eight^{29}$ of the countries of the region is used. Before this estimation is carried out, a test for unit root of the variables of the three models (I, II, III) is undertaken. The result shows that the series are non-stationary (Table 4.1) and that the variables for each model are co-integrated (Table 4.2). The results of this estimation, obtained after an exhaustive search, are reported in Tables 4.3–4.5. The major findings³⁰ of this estimation are detailed below.

| d Southern Africa |
|-------------------|
| e for East and |
| n critical valu |
| % Mackinno |
| ADF test at 1 |
| analysis: an |
| nit root |
| 1 |

| Logarithms of | Symbol Used | Level of Integration |
|---|---|-------------------------|
| Food real (at 1985 price) | LFOODR | l(1) |
| Agricultural raw materials real (at 1985 price) | LAGRMR | I(1) |
| Agricultural raw materials as proportion of GDP | LAGRMG | (t) |
| Beverage real (at 1985 price) | LBEVR | (1) |
| Beverage as proportion of GDP | LBEVG | l(1) |
| Minerals real (at 1985 price) | LMMR | (1) |
| Minerals as proportion of GDP | LMMG | (1) |
| Price of food, agricultural raw material, minerals, beverage of West & Central Africa sample | LPFESAs, LPAESAs, LPBESAs & LPMESAs | I(1) |
| Aggregate bilateral, multilateral, private inflows and grants respectively; aggregate inflows, aggregate excluding private flows; foreign direct investment, fertilizer consumption | LBILAD, LMULTD, LPRIVD, LGRANT, LBMPGFF, LBMGFF, LFDI, LFERTZ | (1) |
| Capital stock as ratio of GDP, total, public, private | LKGDP, LKGVGDP, LKPGDP | (1) |
| Domestic credit as ratio of GDP: total, private, respectively | LDCRTG, LDCRPG | (1) |
| Government investment as proportion of GDP | LIGVGDP (at 10%) | l(0) border |
| Total, government and private investment as proportion of GDP | LIGDP, LIPGDP | (1) |
| (Regional) real exchange rate for food, agricultural raw materials, minerals and beverages [(regional commodity price X nominal exchange rate)/GDP deflator] | LRERF, LRERA, LRERM & LRERB | 1(1) |

| Africa |
|----------|
| Southern |
| and |
| r East |
| test fo |
| gration |
| Co-inte |
| 4.2 (|
| Table |

| L | Engel- | Mackinnon | Johansen | Mackinnon c | critical value |
|----------------------------|---------------------------|-------------------------|---------------------|-------------|----------------|
| Equation | Granger AUF statistics | critical value at 5% | IIKeIInood ratio | 5% | 1% |
| Food | | | | | |
| Model I | -3.929 | -3.795 | 87.21 | 29.68 | 35.65 |
| Model II | -3.57 | -3.489 ** | 38.20 | 29.68 | 35.65 |
| Model III | -3.971 | -3.862 ** | 121.68 | 47.21 | 54.46 |
| Agricultural raw materials | | | | | |
| Model I | -4.147 | -3.794 | 64.79 | 29.68 | 35.65 |
| Model II | -4.176 | -4.166 | 41.58 | 29.68 | 35.65 |
| Model III | -3.769 | -3.870 ** | 115.92 | 47.21 | 54.46 |
| Beverages | | | | | |
| Model I | -3.879 | -3.808 | 31.58 | 29.68 | 35.65 |
| Model II | -3.769 | -3.500 ** | 21.71 | 29.68 | 35.65 |
| Model III | -3.903 | -3.876 ** | 73.72 | 47.21 | 54.46 |
| Minerals | | | | | 1 |
| Model II | -3.274 | -3.511 ** | 31.92 | 29.68 | 35.65 |
| | | | | | |

** [*] Mackinnon critical value at 10% [1%] level of significance. The Johansen test used assumes linear deterministic trend in the data with intercept, no trend, in the test VAR (i.e., the co-integration equation, as a indicator of long-run equilibrium relation, has no trend).

Food: Within this region, the current price of food is found to have a strong and positive effect, in both the short and long run (for Model I) and in the long run (for Model III). Of these, the long-run effect is likely to be the stronger. When Model II is used, only the long-run relative price elasticity is found to be positive and significant. However, this remains significantly lower in magnitude than the elasticity for current prices. The results for all diagnostic tests are quite acceptable. A oneperiod lag is assumed for the impact of foreign inflows, using different foreign inflow indicators,³¹ while three years is allowed for the impact of capital formation indicators to be felt. These results indicate that not all variables are statistically significant. The results may also be taken as suggesting the likelihood that 'Dutch disease' possibilities cannot be sustained, the existence of a positive impact of capital formation and generally confirming the positive impact of prices in the long run.

Agricultural raw materials: The estimation for this commodity group confirms that capital formation indicators have a strong and statistically significant positive impact, both in the short and long run. Current prices in the long run are found either to be statistically significant and negative (Model I) or statistically not significantly different from zero. This result would tend to confirm the second proposition relating to the supply price relationship, discussed in the previous section (estimation approach). However, short-run relative price is found to be positive and statistically significant. Again, a 'Dutch disease' effect cannot be suggested by the data.

I(c) Tropical beverages: Only in Model I are current prices (with the two types of lags) found to be positive and statistically significant. Relative price, which indicates capacity creation in Model II, is also found to be statistically significant in the long run. The capital formation indicator yields an ambiguous result, having a negative value in Model II and positive one in Model III. Using Model III, of all capital inflow categories, only multilateral flows are found to be co-integrated with the other regressors. The lag for capital inflows is set one period behind that of price and capital formation indicators in order to allow for inflationary pressure formation. Botswana and Zambia are excluded from the analysis, since we are not in possession of beverage export data for these countries. With diagnostic tests yielding quite good results, all but one of the variables are found to be statistically significant in the short run. The exception to this is the negative value of the variable for long-lag capital inflows, indicating the presence of a 'Dutch dis-

| Table 4.3 Estimation with | n current price | s for East | and Souther | n Africa | | | | |
|-------------------------------|-------------------|------------|--------------|-----------|------------|----------|------|----------|
| Dependent→ | Food | | Agric. raw n | naterials | Bevera | ges | Min | erals |
| Regressors↓ | LFOODR | t-values | LAGRMR | t-values | LBEVR | t-values | LMMR | t-values |
| Short-run coefficien | ts (elasticities) | | | | | | | |
| Constant | -2.4 | -3.3*** | 4.16 | 3.0*** | | | | |
| ALP 1 | 0.29 | 2.06** | 0.02 | 0.05 | 0.14 | 1.77* | | |
| ALP_5 | | | | | 0.05 | 0.50 | | |
| | | | 0.16 | 1.77* | | | | |
| ALIGVGDP_1 ALKPGDP_5 | -0.21 | -1.38 | | | 0.03 | 0.25 | | |
| Long-run coefficient | ts (elasticities) | | | | | | | |
| IP 2 | 5 03 | 4 45*** | -1 38 | | 0.54 | 3 26*** | | |
| | 2 | 2 | | i | 0.30 | 1.83* | | |
| LIPGDP_2 | | | 0.44 | 1.91** | | | | |
| LIGVGDP_2 | -0.45 | -0.69 | | | | | | |
| LKPGDP_6 | | | | | 0.14 | 0.74 | | |
| Dependent_1 | -0.11 | -1.87** | -0.43 | -6.17*** | -0.34 | -3.79*** | | |
| Diagnostic test | | | | | | | | |
| ECM adjusted R ² | 0.18 | n=108 | 0.34 | n=85 | 0.09 | 06=U | | |
| Level adjusted R ² | 0.14 | | 0.57 | | 0.64 | | | |
| Jarque-Bera | 4.23** | | 0.4** | | 0.14** | | | |
| RESET(1) | 5.6 | Pr(0.02) | 1.4 | Pr(0.24) | 0.01 | Pr(0.91) | | |
| Chow | 1.4** | | 2.83*** | • | 2.77*** | | | |
| Degree of multicollinearity+ | Low | | Very low | | Very low | | | |
| Constant (country) dummy used | | | Botswana | -0.95 | Ethiopia | 0.59 | | |
| | | | Tanzania | 1.88 | Kenya | 0.80 | | |
| | | | | | Madagascar | 0.30 | | |
| | | | | | Tanzania | 0.46 | | |
| | | | | | Uganda | 0.64 | | |

140

| Dependent→ | Foo | в | Agric. raw n | naterials | Bevera | lges | Minera | als |
|-------------------------------|-----------------|----------|--------------|-----------|------------|----------|----------|----------|
| | LFOODR | t-values | LAGRMR | t-values | LBEVR | t-values | LMMR | t-values |
| Short-run coefficier | nts (elasticiti | es) | | | | | | |
| Constant | 0.84 | 2.85*** | 1.02 | 3.05*** | 1.84 | 3.47*** | 0.54 | 1.45 |
| ALP 1 | -0.02 | -0.15 | 0.33 | 2.25*** | | | 0.10 | 0.48 |
| ∆LP_5 | | | | | 0.08 | 1.18 | -0.42 | -1.7* |
| | | | 0.18 | 2.15** | | | | |
| ALIGVGDP_2 ALKPGDP_5 | -0.19 | -1.08 | | | | | 0.18 | 0.68 |
| Long-run coefficien | its (elasticiti | es) | | | | | | |
| LP 2 - | 0.13 | 1.6* | -0.18 | -1.24 | 0.10 | 1.19 | 1.26 | 1.97** |
| | 2 | 2 | 2 | | 0.23 | 3.16*** | 4.0 | -0.64 |
| LIPGDP_3 | | | 1.19 | 2.77*** | | | | |
| LIGVGDP_3 | -0.42 | -1.4 | | | | | | |
| LKPGDP_6 | | | | | -0.27 | -2.9*** | | |
| | | | | | | | 2.36 | 3.32*** |
| Dependent_1 | -0.34 | -4.5*** | -0.14 | -3.43*** | -0.50 | -4.42*** | -0.18 | -3.03*** |
| Diagnostic test | | | | | | | | |
| ECM adjusted R ² | 0.12 | n=108 | 0.18 | n=108 | 0.15 | 0=00 | 0.07 | n=95 |
| Level adjusted R ² | 0.34 | | 0.12 | | 0.81 | | 0.37 | |
| Jarque-Bera | 1.53** | | 2.85** | | 2.17 | | 74 | |
| RESET(1) | 0.17 | Pr(0.68) | 0.46 | Pr(0.50) | 1.4 | Pr(0.24) | 0.07 | pr(0.80) |
| Chow | 1.8** | | 2.98*** | | 2.3*** | | 1.55** | |
| Degree of multicollinearity+ | Low | | Very low | | Very low | | Low | |
| Constant (country) dummy used | Zambia | -0.84 | Botswana | -0.36 | Ethiopis | 0.77 | Botswana | 0.94 |
| | | | | | Kenya | 0.95 | Kenya | 0.52 |
| | | | | | Madagascar | -0.64 | | |
| | | | | | Uganda | 0.29 | | |

Table 4.4 Estimation with real exchance rate for East and Southern Africa

Determinants of Foreign Exchange Supply

141

| anie 4.0 Estilliation Wit | n capital III | INV IN ERS | | | | | | |
|-------------------------------|-----------------|------------|--------------|-----------|--------------|----------|-------|----------|
| Dependent→ | Foot | - | Agric. raw n | naterials | Bevera | ges | Miner | als |
| Regressors↓ | LFOODR | t-values | LAGRMR | t-values | LBEVR | t-values | LMMR | t-values |
| Short-run coefficie | nts (elasticiti | es) | | | | | | |
| Constant | -0.83 | -0.71 | 1.04 | 0.81 | -0.86 | -1.01 | | |
| ALP 1 | 0.11 | 0.59 | 0.35 | 0.00 | 0.10 | 0.80 | | |
| ALP 5 | | | | | 0.08 | 0.76 | | |
| | | | 0.22 | 2.33*** | | | | |
| | -0.15 | -0.75 | | | | | | |
| | | | | | -0.10 | -0.72 | | |
| ΔFF_2@ | 0.09 | 0.37 | 0.26 | 0.69 | -0.17 | -0.62 | | |
| AFF_6 | | | | | -0.24 | -1.83* | | |
| Long-run coefficie | nts (elasticiti | es) | | | | | | |
| LP 2 | 1.01 | 1.85** | -0.41 | -0.23 | 0.37 | 1.02 | | |
| LP_6 | | | | | 0.22 | 0.92 | | |
| LIPGDP_3 | | | 1.16 | 2.96*** | | | | |
| | 0.07 | 0.24 | | | 0,58 | £ 30*** | | |
| | 000 | 2 | 10 0 | L † 0 | 0.00 | 07.0 | | |
| FF_7 | 0.22 | ~LC.I | 0.27 | c/.0 | 0.05 0.05 | -3.4*** | | |
| Dependent_1 | -0.30 | -3.75*** | -0.17 | -3.33*** | -0.37 | -3.67*** | | |
| Diagnostic test | | | | | | | | |
| ECM adjusted R ² | 0.13 | n=119 | 0.10 | 06=U | 0.10 | n=84 | | |
| Level adjusted R ² | 0.25 | | 0.14 | | 0.68 | | | |
| Jarque-Bera | 7.13 | | 1.86** | | 1.27** | | | |
| RESET(1) | 0.13 | Pr(0.72) | 2.2 | Pr(0.14) | 0.01 | Pr(0.94) | | |
| Chow | 0.60** | | 2.05*** | Pr(0.01) | 1.99*** | | | |
| Degree of multicollinearity+ | Low | | Low | | Very Low | | | |
| Constant (country) dummy | Zambia | -0.95 | Ethiopia | 0.29 | Madagascar | -0.26 | | |
| used | | | | | Malawi | -0.32 | | |

Table 4.5 Estimation with capital inflow for East and Southern Africa

Notes to Tables 4.3-4.5:

The long-run coefficients and their t-values are using the Wickens and Breusch (1988) and Gurney (1989) approach. The long-run R2 is also taken from the same regression. _1, _2 ...etc. show one, two ...etc. periods lag; ~ weakly significant (closer to 10%)

* significant at 10%; ** at 5% and *** at 1% and less. For all of the equations the F value (over all fit) is significantly different from zero.

The Jarque-Bera _2 statistics at 5% level of significance for 2 degrees of freedom is 5.99. ** indicates significance at this level or better. Note, however, that it is relevant for large sample and visual inspection is important (see Mukherjee et al. 1997).

Chow break test is carried by using two country data as one series (when there is insufficient data by each country)

+ For multicollinearity simple correlation between regressors around <35 is assumed very low , 35-55 low & 55-65 acceptable.

The error correction form, for instance, for the food equation above can be given by the following (See Appendix 4.1):

△LFOODR=0.29△LP_1-0.21△LIGVGDP_1-0.11(LFOODR-5.03LP_2+0.45LIGVGDP_2+21.82). The long-run relationship being,

LFOODR=5.03LP_2-0.45LIGVGDP_2-21.82.

@ For food, bilateral debt is used (others are not co-integrated); for agr. raw material, the sum of all, except private inflows, is used.

 Table 4.6
 Unit root analysis: an ADF test at 1% Mackinnon critical value for North Africa

| Logarithms of | Symbol Used | Level of Integration |
|---|--|-------------------------|
| Food real export of food (at 1985 price) | LFOODR | i(1) |
| Agricultural raw materials real (at 1985 price) | LAGRMR | l(1) |
| Minerals real (at 1985 price) | LMMR | l(1) |
| Price of food, agricultural raw materials and minerals of North Africa sample | LPFNAs, LPANAs, and LPMNAs | l(1) |
| Aggregate bilateral, multilateral, pri- vate inflows and grants respectively; aggregate, aggregate excluding pri- vate flows; foreign direct investment, fertilizer consumption | LBILATD, LMULTD, LPRIVD, LGRANTS, LBMPGFF, LBMGFF,LFDI,LFERTZ | l(1) |
| Capital stock as ratio of GDP, total, public, private | LKGDP, LKGVGDP,LKPGDP | l(1)* |
| Domestic credit as ratio of GDP: total, public and private respectively | LDCRTG, LDCRGVG, LDCRPG | l(1) |
| Investment as proportion of GDP: total, public and private respectively | LIGDP, LIGVG, LIPG | l(1) |
| (Regional) real exchange rate for food, agr. raw materials and minerals [(regional commodity price X nominal exchange rate)/GDP deflator)] | LRERF, LRERA, and LRERM | l(1)* |

* For these variables the level of significance is 5%.

| Africa |
|-----------|
| North , |
| est for I |
| ration t |
| co-integ |
| 4.7 0 |
| Table |

| ; | Engel- | Mackinnon | Johansen | Mackinnon (| critical value |
|----------------------------|---------------------------|-------------------------|-----------------------|-------------|----------------|
| Equation | Granger ADF statistics | critical value at 5% | likelihood - ratio | 5% | 1% |
| Food | | | | | |
| Model I | -5.724 | -4.533 * | 63.11 | 29.68 | 35.65 |
| Model II | -6.320 | -4.532 * | 39.8 | 29.68 | 35.65 |
| Model III | -5.538 | -4.975 * | 68.27 | 47.21 | 54.46 |
| Agricultural raw materials | | | | | |
| Model II | -3.075 | -3.558 ** | 19.26 | 29.68 | 35.65 |
| Model III | -3.386 | -3.953 ** | 41.61 | 47.21 | 54.46 |
| Minerals | | | | | |
| Model I | -3.073 | -3.558 ** | 34.08 | 29.68 | 35.65 |
| Model II | -3.138 | -3.558 ** | 20.59 | 29.68 | 35.65 |
| Model III | -4.721 | -4.310 | 60.25 | 47.21 | 54.46 |

** [*] Mackinnon critical value at 10% [1%] level of significance. The Johansen test used assumes linear deterministic trend in the data with intercept, no trend, in the test VAR (i.e., the co-integration equation, as a indicator of long run equilibrium relation, has no trend).

ease' effect. The foreign inflow variable shows a statistically significant negative sign, in the long run, indicating the presence of a debt overhang, with a positive sign for the long- and short-lag aspects, respectively.

1(d) Minerals, ores and metals: Of the equations described in this section, the one relating to minerals, ores and metals is the most difficult to estimate. The real level of mineral exports and its relative price are found not to be co-integrated with ten types of capital formation indicator. Further, the co-integration Dickey-Fuller statistics for these two variables with that of investment in the private sector is 3.3, compared to a Mackinnon critical value of 3.5, at a 10 per cent significance level. However, given the well-documented uncertainty of this test in borderline cases, the estimation is undertaken with some reservation. Nevertheless, the results of this test point to a statistically significant negative impact of the short-run relative price relevant for capacity creation. In the long run, the impact of the price affecting capacity utilization is found to be statistically significant and positive. The capital formation indicator is also found to have a statistically significant positive impact. Long-run prices show a negative coefficient, although not at a statistically significant level. Adjustment to disequilibrium is also found to be very low, with only 18 per cent of the past error being made up for in the current period.

The adjustment coefficient varies from model to model within this region, ranging from about 10 per cent up to around 50 per cent in the case of beverage. This may be taken as indicating a generally low level of adjustment to disequilibrium or shocks for all commodity categories, with the exception of beverages. This result stands in contradiction to the intuitive expectation that short gestation items will be likely to adjust faster.

North Africa (NA)

Since no data is available relating to Libya and Morocco, only three countries are chosen as a sample for the North Africa region. These are Algeria, Egypt and Tunisia. A similar estimation procedure is followed in relation to the NA region, as was applied to the ESA region. However, since none of the sample countries are exporters of beverages, we undertake only three sets of estimations, using the three relevant models. Unit rate tests indicate that the series are non-stationary, but that they are co-integrated within each model (see Tables 4.6 and 4.7 for

details of this). Based on this analysis, the following observations may be made in relation to each commodity category.

2(a) Food: In general, the food function for the NA region suggests that current prices are not statistically significant in explaining exports of food, the only exception to this being the long-run value for Model III. Neither is the relative price effect satisfactory. However, the result does suggest a positive elasticity for short-run capital formation indicators within Model III. Indeed, the food equation in Model III would suggest that contrary to 'Dutch disease'/debt overhang hypothesis, capital inflows will have a positive and statistically significant impact, in the long run, and a positive but statistically insignificant impact in the short run.

2(b) Agricultural raw materials: Exports of agricultural raw materials are not found to be co-integrated with current prices and various capital formation indicators. Thus, the estimation is undertaken using Model II. This model indicates a positive elasticity for the real exchange rate, both in the short and long run. However, only the long-run elasticity is found to be statistically significant. The capital formation indicator is also found to have a positive elasticity is found to be statistically significant. The short and long run. However, only the short-run elasticity is found to be statistically significant. The short and long run. However, only the short-run elasticity is found to be statistically significant. Based on Model III, a positive and statistically significant coefficient for the short- and long-run capital formation indicators may be identified. Capital inflows are found to have a statistically significant positive impact in the short run and a negative impact in the long run. However, caution should be exercised in the relation to these results, since the Chow value for Model III is high.

2(c) Minerals, ores and metals: With the exception of the capacity utilization indicator in Model I, long-run prices are found to have a negative and statistically significant value. This would tend to support the hypothesis that a lower price will trigger exports, under a revenue-maximizing regime. The long-run capital formation indicator also shows a positive and statistically significant value within Model I, having a statistically significant negative value in the short run. This would tend to corroborate the revenue maximization argument detailed above. Moving on to examine Model II, in which an estimation is undertaken using the real exchange rate, all of the coefficients, with the exception of the constant, are found to be statistically insignificant in the short run. However, in the long run, the price that is assumed to show capacity creation is found to have a positive and statistically significant

value. In contrast, the price which indicates capacity utilization, while again positive, is not found to be statistically significant. Neither is the capital formation indicator found to be statistically significant within this model. However, when Model III is used, both of these types of prices are found to exhibit a negative and statistically significant result, in the long run. Long-run short-lag effects of capital inflows are also found to be positive and statistically significant, contrary to the 'Dutch disease' theory, while its long-run long-lag coefficient would tend to suggest a negative value. This result may be interpreted as showing that capital inflows are capable of ensuring capacity utilization but not capacity creation. (In fact, in the long run, 'Dutch disease' and debt overhang problems may be expected.)

In contrast to the two other regions, within the NA region, the adjustment coefficient is generally high. This would suggest that a good part of errors arising within the previous period may be made up for in the current period, which, in turn, points to a relatively higher capacity of the region to adjust to external shocks or, indeed, a deviation from the long-run equilibrium.

West and Central Africa (WCA) Region

A sample of ten countries has been selected from the WCA region, as defined in the UN-ECA (Economic Commission for Africa) data reporting system. These include Benin (Ben), Burkina-Faso (Bf), Cameroon (Cam), The Central African Republic (Car), Gabon (Gab), Ghana (Gha), Nigeria (Nig), Senegal (Sen), Sierra Leone (Ser), and Zaire/DR Congo (Zi). For this sample, four commodity export functions are estimated below. All variables used in the estimation are found to be non-stationary series. These variables are also integrated to the first order (Table 4.11). Thus, the estimation is carried out once a co-integration test has been undertaken for each of the three models. The results indicate that the variables in each equation are co-integrated, suggesting the existence of a long-run relationship between these (Table 4.12).

| Africa |
|------------|
| for North |
| t prices t |
| n current |
| on with |
| Estimatic |
| e 4.8 |
| Tabl |

| Dependent→ | Food | - | Agric. raw I | naterials | Miner | als |
|----------------------------------|-------------------|----------|--------------|-----------|------------------|----------|
| | LFOODR | t-values | LAGRMR | t-values | LMMR | t-values |
| Short-run coeffici | ents (elasticiti | es) | | | | |
| Constant | 2.1 | 0.84 | | | 8.52 | 3.95*** |
| 1 JLP 1 | -0.26 | -0.59 | | | 0.21 | 0.59 |
| JLP_5 | | | | | -0.25 | -0.93 |
| ALIPGDP_1 ALIGVGDP_5 | 0.54 | 0.86 | | | -0.56 | -2.24*** |
| Long-run coefficie | ents (elasticitio | es) | | | | |
| P 2 | 0.35 | 0.96 | | | -0.35 | -0.70 |
| P_6 | | | | | -0.91 | -3.54*** |
| | 0.08 | 0.18 | | | | |
| IGVGDP 6 | | | | | 1.14 | 5.05*** |
| Dependent_1 | -0.96 | -6.23*** | | | -0.71 | -9-9- |
| Diagnostic test | | | | | | |
| ECM adjusted R ² | 0.47 | n=57 | | | 0.60 | n=45 |
| Level adjusted R ² | 0.41 | | | | 0.60 | |
| Jarque-Bera | 273 | | | | 1.5** | |
| REŚET(1) | 1.78 | Pr(0.19) | | | 1.1 | Pr(0.30) |
| Chow | 2.7*** | | | | 2.03** | |
| Degree of multicollinearity+ | Low | | | | Low | |
| Constant (country) dummy used | Tunisia | 1.11 | | | Egypt Tunisia | 0.48 |

Notes for Tables 4.8 to 4.10: as given in Table 4.3.

Notes for Table 4.9: ++ The correlation coefficient between LIPG and LRERF = 0.82 (others are low).

Notes for Table 4.10: @ For agricultural raw material private debt with 1 period lag is used (since others flows are not co-integrated); ++ The correlation coefficient between LBMPGFF and LPMNASS =0.70 so is their difference; others are very low.

| Africa |
|--------|
| North |
| for |
| prices |
| real |
| with |
| ation |
| Estim |
| 4.9 |
| able |

| Dependent→FoodAgric. raw materialsRegressors↓LFOODRt-valuesLAGRMRt-valuesRegressors↓LFOODRt-valuesLAGRMRt-valuesShort-run coefficients (elasticities)0.65 -1.37 0.663 2.13^{m} Constant 4.88 4.4^{m} 1.09 2.13^{m} ALP 1 -0.52 -1.37 0.63 2.13^{m} Constant 4.88 4.4^{m} 1.09 2.13^{m} ALP 1 0.4 0.64 0.63 2.13^{m} ALP 5 0.13 0.3 1.32 2.08^{m} ALP 6DP 1 0.00 0.00 0.00 0.02 ALP 6DP 2 0.00 0.00 0.00 0.66 1.44 LP 2 0.13 1.32 2.4 1.7^{m} LP 2 0.00 0.00 0.00 0.66 1.44 LP 2 0.00 0.00 0.00 0.66 1.44 LP 2 0.13 1.32 2.4 1.7^{m} LP 2 0.00 0.00 0.00 0.03 0.13 LP 2 0.00 0.00 0.03 0.13 $n=56$ LP 2 0.13 $n=57$ 0.13 $n=56$ LP 2 0.00 0.03 0.03 0.03 0.077 Dependent 1 1.07 3.5^{m} 0.13 0.077 0.56 LP 2 0.046 0.56 0.13 0.077 Dependent 2 0.48 0.56 0.13 0.777 D | l able 4.9 Estimation v | vith real price | s for North | Atrica | | | |
|--|-------------------------------|-------------------|-------------|------------|-----------|---------------|----------------|
| Regressors LFOODR t-values LAGRMR t-values LAGRMR t-values 1 Short-run coefficients (elasticities) 0.52 -1.37 0.63 1.15 ALP_1 -0.52 -1.37 0.63 1.15 ALP_15 -0.52 -1.37 0.63 1.15 ALP_16 -0.52 -1.37 0.63 2.13** ALPGROVG_1 0.4 0.64 0.62 2.08** ALICKOGDP_5 0.3 1.32 2.4 1.7* LPCROVG_1 0.00 0.00 0.00 0.66 1.44 LPCSD_2 0.3 1.32 2.4 1.7* LPCSD_2 0.00 0.00 0.00 0.056 1.44 LPCSD_2 0.00 0.00 0.056 1.44 1.7* LPCSD_2 0.046 -1.02 -2.6*** 1.44 LPGDP_6 -1.02 -7.1** -0.22 -5.6** Dependent_1 -1.02 -7.1** -0.22 -5.6 | Dependent→ | Foo | P | Agric. raw | materials | Miner | als |
| Short-run coefficients (elasticities) | Regressors↓ | LFOODR | t-values | LAGRMR | t-values | LMMR | t-values |
| | Short-run coeffic | cients (elasticit | ies) | | | | |
| ALP_1 -0.52 -1.37 0.63 1.15 ALP 5 0.4 0.64 0.63 1.15 ALDCRGVG_1 0.4 0.64 0.62 2.08** ALICRGVG_1 0.3 0.3 1.32 2.08** ALICRGVG_1 0.3 1.32 2.4 1.7* LP_6 0.00 0.00 0.00 1.44 LPGDP_6 0.102 0.56 1.44 LIGNGPP_6 -1.02 -7.1** -0.22 -2.6*** Dependent_1 -1.02 -7.1** -0.22 -2.6*** Dependent_1 0.46 n=57 0.13 n=56 Jarque-Bera 0.34 Pr(56) 0.56 -44 Dependent_1 0.34 Pr(56) 0.56 -2.6*** Dependent_1 -1.02 -7.1** -0.22 -2.6*** Dependent_1 0.13 n=56 0.65 -6**** Dependent_1 0.102 0.13 n=56 -6**** Dependent_1 0.16 0.56 1.44 -7.1*** Dependent_1 | Constant | 4.88 | 4.4*** | 1.09 | 2.13** | 2.55 | 4*** |
| ALP.5 ALP.5 0.4 0.64 0.62 2.08** ALIDCRGVG_1 0.10 0.62 2.08** 0.64 0.62 2.08** ALIGVGDP_5 0.10 0.00 0.01 0.62 2.08** LP_2 0.3 1.32 2.4 1.7* LP_6 0.00 0.00 0.00 0.56 1.44 LPCRGVG_2 0.100 0.00 0.00 0.56 1.44 LPCRGVG_2 0.10 0.00 0.00 0.56 1.44 LPCRGVG_2 0.10 0.00 0.01 0.56 1.44 LIPGOP_2 0.10 0.00 0.03 0.13 n=56 LIPGOP_2 0.10 0.01 0.03 0.13 n=56 LIPGOP_2 0.46 n=57 0.13 n=56 Jarque-Bera 0.34 Pr(56) 0.56 1.44 Level adjusted R ² 0.46 n=57 0.13 n=56 Jarque-Bera 0.34 Pr(56) 0.56 1.44 Dependent_1 0.34 Pr(56) <td< td=""><td>ALP_1</td><td>-0.52</td><td>-1.37</td><td>0.63</td><td>1.15</td><td>0.48</td><td>1.42</td></td<> | ALP_1 | -0.52 | -1.37 | 0.63 | 1.15 | 0.48 | 1.42 |
| ALIPGDP_1 0.64 ALDCRGVG_1 0.62 ALIOCREVG_1 0.62 ALIOCREVG_1 0.65 ALIOCREVG_1 0.65 LP_2 0.3 LP_2 0.3 LP_2 0.3 LP_2 0.3 LP_2 0.3 LP_2 0.00 LP_2 0.3 LP_2 0.00 LP_6 1.44 LPCGOP_6 -1.02 LIGVGDP_6 -1.02 Dependen_1 -1.02 Dependen_1 -0.22 Diagnostic test 0.48 Level adjusted R ² 0.48 Jarque-Bera 0.34 Pr(56) 0.56 Jarque-Bera 0.34 Chow 3.5*** Degree of multicollinearity+ 1.66 Algeria -1.86 Algeria -2.2 | ∆LP_5 | | | | | 0.32 | 0.88 |
| ALDCRGVG_1 0.62 2.08** ALIGVGDP_5 0.00 0.62 2.08** LP_2 0.3 1.32 2.4 1.7* LP_6 0.00 0.00 0.00 0.56 1.44 LP2 0.3 1.32 2.4 1.7* LP6 0.00 0.00 0.00 0.56 1.44 LIPGDP_2 0.102 0.56 1.44 1.44 LIPGDP_6 -1.02 -7.1** -0.22 2.6*** Dependent_1 -1.02 -7.1** -0.22 2.6*** Dependent_1 0.103 n=56 0.13 n=56 Jarque-Bera 0.48 n=57 0.13 n=56 Jarque-Bera 0.34 Pr(56) 0.56 7.0 Degree of multicollinearity+ 1.86 0.09 Pr(0.77) Degree of multicollinearity+ 1.96 3.6**** 2.2 Chow 0.186 Algeria -1.86 Algeria -2.2 | | 0.4 | 0.64 | | | | |
| Long-run coefficients (elasticities) 2.4 1.7* LP_2 0.3 1.32 2.4 1.7* LP_6 0.3 1.32 2.4 1.7* LP6 0.00 0.00 0.00 0.56 1.44 LIPGDP_6 -1.02 -7.1** -0.22 2.6*** LIGVGDP_6 -1.02 -7.1** -0.22 2.6*** Dependent_1 -1.02 -7.1** -0.22 2.6*** Dependent_1 -1.02 -7.1** -0.22 2.6*** Dependent_1 0.48 n=57 0.13 n=56 Level adjusted R ² 0.46 0.56 0.6 56 Jarque-Bera 541 Pr(56) 3.6 56 56 Jarque-Bera 5.4 Pr(56) 3.6 56 56 56 Low 0.34 Pr(56) 3.6 56 56 56 56 56 56 56 56 56 56 56 56 56 | ALDCRGVG_1 ALIGVGDP_5 | | | 0.62 | 2.08*** | -0.19 | -0.61 |
| LP_2 0.3 1.32 2.4 1.7* LP_6 0.00 0.00 0.00 0.06 1.44 LPGDP_5 0.00 0.00 0.00 0.56 1.44 LIGVGDP_6 -1.02 -7.1** -0.22 -2.6*** Dependen_1 0.10 0.48 n=57 0.13 n=56 Dependent_1 0.34 n=57 0.13 n=56 Jarque-Bera 0.34 Pr(56) 0.56 0.03 Jarque-Bera 0.34 Pr(56) 0.09 Pr(0.77) Degree of multicollinearity+ High++ 1.86 Algeria -2.2 | Long-run coeffici | ients (elasticiti | es) | | | - | |
| LiPGDP_2 0.00 0.00 0.00 0.44 LIDCRGVG_2 -1.02 -7.1*** -0.22 -2.6*** LIGVGDP_6 -1.02 -7.1*** -0.22 -2.6*** Dependent_1 -1.02 -7.1*** -0.22 -2.6*** Dependent_1 -1.02 -7.1*** -0.22 -2.6*** Dependent_1 0.48 n=57 0.13 n=56 Level adjusted R ² 0.46 343 343 343 Jarque-Bera 541 Pr(56) 0.09 Pr(0.77) Chow 3.5**** -1.86 Algeria -2.2 | | 0.3 | 1.32 | 2.4 | 1.7* | 0.64 0.95 | 1.38 1 94** |
| LDCRGVG_2 LICKGVG_2 LIGVGDP_6 LIGVGDP_6 Dependent_1 -0.22 -2.6*** Dependent_1 -0.22 -2.6*** Dependent_1 -0.22 -2.6*** Dependent_1 -0.22 -2.6*** Dependent_2 -0.48 n=57 0.13 n=56 Jarque-Bera 541 -0.46 -0.56 0.56 Jarque-Bera 541 -0.09 Pr(0.77) Chow 3.5*** -0.09 Pr(0.77) Degree of multicollinearity + Hight -1.86 Algeria -2.2 | LIPGDP_2 | 0.00 | 0.00 | | | 2 | |
| Dependent_1 -1.02 -7.1*** -0.22 -2.6*** Dependent_1 -1.02 -7.1*** -0.22 -2.6*** Dependent_1 0.13 -0.22 -2.6*** Dependent_1 0.13 n=56 0.13 n=56 ECM adjusted R ² 0.46 n=57 0.13 n=56 Jarque-Bera 541 Pr(56) 0.09 Pr(0.77) Chow 3.5*** Low Low 2.2 Degree of multicollinearity+ High++ -1.86 Algeria -2.2 | LDCRGVG_2 | | | 0.56 | 1.44 | | |
| Diagnostic test 0.48 n=57 0.13 n=56 ECM adjusted R ² 0.48 n=57 0.13 n=56 Level adjusted R ² 0.46 0.56 0.56 Jarque-Bera 541 3.43 3.43 RESET(1) 0.34 Pr(56) 0.09 Pr(0.77) Chow 3.5*** 1.66 3.6**** 2.2 Degree of multicollinearity+ High++ 1.86 Algeria -2.2 | Dependent_1 | -1.02 | -7.1*** | -0.22 | -2.6*** | 0.02 -0.62 | -5.9*** |
| ECM adjusted R ² 0.48 n=57 0.13 n=56 Level adjusted R ² 0.46 0.56 0.56 343 Jarque-Bera 541 343 343 556 Jarque-Bera 541 0.34 Pr(56) 0.09 Pr(0.77) RESET(1) 3.5*** 1.5*** 3.6**** 1.0**** 1.0**** Defree <of multicollinearity+<="" td=""> High++ 1.86 Algeria -2.2 Constant (country) dummy Algeria -1.86 Algeria -2.2</of> | Diagnostic test | | | | | | - |
| Level adjusted R ² 0.46 0.56 Jarque-Bera 541 343 Jarque-Bera 541 343 Jarque-Bera 541 343 Chow 0.34 Pr(56) 0.09 Chow 3.5*** 3.6**** 5.6**** Degree of multicollinearity+ High++ Low -2.2 Constant (country) durmity Algeria -2.2 | ECM adjusted R ² | 0.48 | n=57 | 0.13 | n=56 | 0.49 | n=45 |
| Jarque-Bera 541 343 RESET(1) 0.34 Pr(56) 0.09 Pr(0.77) Chow 3.5*** 1.6*** Degree of multicollinearity+ High++ Low 2.2 Constant (country) durmny Algeria -2.2 | Level adjusted R ² | 0.46 | | 0.56 | | 0.40 | |
| RESET(1) 0.34 Pr(56) 0.09 Pr(0.77) Chow 3.5*** 3.5*** 3.6 3.6 Degree of multicollinearity+ High++ Low Constant (country) dummy Algeria -2.2 | Jarque-Bera | 541 | | 343 | | 0.06** | |
| Chow 3.5*** 3.6~*** 3.6~*** Degree of multicollinearity+ High++ Low Low Constant (country) dummy Algeria -1.86 Algeria -2.2 | RESET(1) | 0.34 | Pr(56) | 0.09 | Pr(0.77) | 0.28 | pr(0.60) |
| Degree of multicollinearity+ High++ Low Constant (country) dummy Algeria -1.86 Algeria -2.2 | Chow | 3.5*** | | 3.6~*** | | 4.5 | |
| Constant (country) dummy Algeria -1.86 Algeria -2.2 | Degree of multicollinearity+ | High++ | | Low | | Low | |
| | Constant (country) dummy | Algeria | -1.86 | Algeria | -2.2 | Algeria | -2.13 |

| h Africa |
|-----------|
| r North |
| flow fo |
| apital ir |
| with c |
| imation |
| 10 Est |
| ble 4.1 |

| Dependent→ | Foo | q | Agric. raw r | naterials | Miner | als |
|-------------------------------|------------------|----------|--------------|-----------|--------|----------|
| _ Regressors↓ | LFOODR | t-values | LAGRMR | t-values | LMMR | t-values |
| Short-run coefficie | ents (elasticiti | les) | | | | |
| Constant | -0.68 | -0.23 | -6.0 | -3.1*** | 7.8 | 3.6*** |
| ALP 1 | 0.05 | 0.09 | 0.61 | 1.12 | 0.40 | 1.24 |
| | | | | | -0.43 | -1.36 |
| ALIPGDP_1 ALIGDP_5 | 1.14 | 1.56~ | | | 0.31 | 0.82 |
| ALFERTZ 1 | | | 1.21 | 3.2*** | | |
| ΔFF_2@ | 0.98 | 1.35 | 0.68 | 2.19** | -0.39 | -0.97 |
| ∆FF_6 | | | | | -0.05 | -0.26 |
| Long-run coefficie | ents (elasticiti | es) | | | | |
| LP_2 | 06.0 | 2.21** | -0.4 | -0.6 | -1.02 | -2.2** |
| | | | | | -1.28 | -3.69*** |
| LIPGDP_2 | 0.17 | 0.39 | | | 30.0 | 99 V |
| LEERTZ 2 | | | 1.53 | 12*** | C7.0 | 0.00 |
| FF 3@ | 0.15 | 1.8* | -0.43 | -2.4*** | 0.58 | 2.54*** |
| FF_7 | | | | | -0.02 | 60'0- |
| Dependent_1 | -1.06 | -6.6*** | -0.60 | -5.5*** | -0.75 | -4.92*** |
| Diagnostic test | | | | | | |
| ECM adjusted R ² | 0.58 | n=50 | 0.40 | n=56 | 0.54 | n=39 |
| Level adjusted R ⁺ | | | 0.93 | | 0.75 | |
| Jarque-Bera | 34 | : | 67 | : | 0.60** | |
| RESET(1) | 0.49 | Pr(0.48) | 10.6 | Pr(0) | 0.06 | Pr(0.88) |
| Chow | 3.9~*** | | 6.5 | | 3.22** | |
| Degree of multicollinearity+ | Low | 1 53 | | FC C | High++ | 000 |
| constant (country) duming | Pilalia | °°.''- | Algena | -2.24 | Agena | 0.0- |
| nsen | | | | | | |

Chapter 4

| ğ |
|---|
| <u>.</u> |
| 4 |
| < |
| 7 |
| 5 |
| 3 |
| õ |
| Õ |
| Ξ. |
| g |
| 5 |
| |
| 3 |
| ٥, |
| 2 |
| |
| 2 |
| Ř |
| Φ |
| 3 |
| a |
| ŝ |
| 1 |
| 20 |
| 3 |
| ÷E |
| Ö |
| 5 |
| õ |
| ē |
| 2 |
| 2 |
| |
| <u>ن</u> |
| ac |
| Mac |
| 6 Mac |
| % Mac |
| 1% Mac |
| at 1% Mac. |
| at 1% Mac |
| st at 1% Mac. |
| est at 1% Mac. |
| test at 1% Mac |
| F test at 1% Mac |
| DF test at 1% Mac |
| ADF test at 1% Mac |
| n ADF test at 1% Mac |
| an ADF test at 1% Mac |
| an ADF test at 1% Mac. |
| is: an ADF test at 1% Mac |
| sis: an ADF test at 1% Mac |
| ysis: an ADF test at 1% Mac. |
| alysis: an ADF test at 1% Mac. |
| nalysis: an ADF test at 1% Mac. |
| analysis: an ADF test at 1% Mac. |
| it analysis: an ADF test at 1% Mac. |
| oot analysis: an ADF test at 1% Mac. |
| root analysis: an ADF test at 1% Mac. |
| it root analysis: an ADF test at 1% Mac. |
| nit root analysis: an ADF test at 1% Mac |
| Unit root analysis: an ADF test at 1% Mac. |
| Unit root analysis: an ADF test at 1% Mac |
| 1 Unit root analysis: an ADF test at 1% Mac. |
| 11 Unit root analysis: an ADF test at 1% Mac. |
| 4.11 Unit root analysis: an ADF test at 1% Mac. |
| 4.11 Unit root analysis: an ADF test at 1% Mac. |
| ole 4.11 Unit root analysis: an ADF test at 1% Mac. |

| Logarithms of | Symbol Used | Level of Integration |
|--|--|-------------------------|
| Food as proportion of GDP | LFOODG | (1) |
| Food real (at 1985 price) | LFOODR | (0) |
| Agricultural raw materials real (at 1985 price) | LAGRMR | (1) |
| Agricultural raw materials as proportion of GDP | LAGRMG | (1) |
| Beverage real (at 1985 price) | LBEVR | (1) |
| Beverage as proportion of GDP | LBEVG | (1) |
| Minerals real (at 1985 price) | LMMR | (1) |
| Minerals as proportion of GDP | LMMG | (1) |
| Price of food, agricultural raw material, minerals, beverage of West and Central Africa sample | LPFWCAs, LPAWCAs, LPBWCAs and LPMWCAs | 1(1) |
| Aggregate bilateral, multilateral, private inflows and grants; same excluding private flows; foreign direct investment | LBMPGFF, LBMGFF, LFDI | (1) |
| Total, government and private Investment as proportion of GDP | LIGNP, LIGGNP, LIPGNP | (1) I |
| (Regional) real exchange rate for food, agricultural raw materials, minerals and beverage [(regional commodity price X norminal exchange rate)/GDP deflator)] | LRERF, LRERA, LRERM and LRERB | l(1) |

| ' Africa |
|----------|
| Central |
| t and |
| West |
| est for |
| ition te |
| ntegra |
| Co-i |
| 94.12 |
| Table |

| ; | Engel- | Mackinnon | Johansen | Mackinnon (| critical value |
|----------------------------|---------------------------|-------------------------|----------|-------------|----------------|
| Equation | Granger AUF statistics | critical value at 5% | ratio | 5% | 1% |
| Food | | | | | |
| Model I | -6.534 | -4.362 * | 160.05 | 29.68 | 35.65 |
| Model II | -6.810 | -4.362 * | 66.53 | 29.68 | 35.65 |
| Model III | -6.612 | -4.734 * | 227.00 | 47.21 | 54.46 |
| Agricultural raw materials | | | | | |
| Model I | -4.551 | -4.362 * | 71.35 | 29.68 | 35.65 |
| Model II | -4.076 | -3.783 | 27.14 | 29.68 | 35.65 |
| Model III | -4.860 | -4.502 [@] | 94.4 | 47.21 | 54.46 |
| Beverages | | | | | |
| Model 1 | -4.142 | -3.795 | 58.41 | 29.68 | 35.65 |
| Model II | -3.485 | -3.491 ** | 29.58 | 29.68 | 35.65 |
| Model III | -4.074 | -3.867 ** | 83.25 | 47.21 | 54.46 |
| Minerals | | | | | |
| Model I | -3.537 | -3.485 ** | 69.32 | 29.68 | 35.65 |
| Model II | -4.089 | -3.790 | 109.34 | 47.21 | 54.46 |
| Model III | -4.610 | -4.163 | 60.31 | 47.21 | 54.46 |

③ is trend stationary, otherwise are not co-integrated. *** [*] Mackinnon critical value at 10% [1%] level of significance.

The Johansen test used assumes linear deterministic trend in the data with intercept, no trend, in the test VAR (i.e., the co-integration equation, as an indicator of long-run equilibrium relation, has no trend).

Notes for Tables 4.13-15: as given in Table 4.3. @ for beverages we used LDCRT (i.e. total domestic credit). Notes for Table 4.15:

A For minerals the foreign inflow used excludes private inflow since private investment is used as one of the repressors.

++ All other correlations are very low except that between LDCRT and LBMPGFF which is 0.79.

Tables 4.13–4.15 summarize the estimated results of the three models, for the four commodity categories within the WCA region. Based on these results, a number of observations can be made.

3(a) Food: It may be concluded that the aggregate food function exhibits a statistically significant positive coefficient for most variables³² in all models, in both the short and long run. However, the possibility of the existence of a 'Dutch disease' phenomenon cannot be inferred from the coefficient for short-term inflows. However, a longrun debt overhang problem may be inferred from the statistically significant negative coefficient for Model III.

3(b) Agricultural raw materials: The results of this estimation suggest a positive short-run elasticity in Model I and statistically insignificant values in the remaining cases. However, relative prices are found to be positive and statistically significant. Capital formation is also found to have a positive and statistically significant long-run value, within Model I. Indeed, contrary to the hypothesis of 'Dutch disease' and debt overhang, capital inflows show positive and statistically significant values in both the short and long run.

3(c) Tropical beverages: In relation to beverages, the estimation results confirm that prices, whether current or real, determine capacity utilization in both the short and long run. On the other hand, prices are found to be relatively unimportant as a factor determining capacity creation. The results also show capital formation indicators as having a positive and statistically significant long-run impact within Model II, while exhibiting a negative impact in Model III. Finally, capital inflows are also found to have a statistically significant positive impact, in both the short and long run.

3(d) Minerals, ores and metals: This estimation confirms the importance of capital formation indicators in the long run (within Model I), as well as the positive impact of current prices on capacity utilization, again in the long run (within Models I and III). Model II shows real exchange rate as having a statistically significant positive and negative long-run impact on capacity creation and utilization, respectively. Finally, the results of Model III suggest the possibility of 'Dutch disease' and debt overhang problems within the sector.
| Table 4.13 Estimation w | vith current pr | ices for W | est and Cen | tral Africa | | | | |
|-------------------------------|-------------------|-----------------------|--------------|-------------|---------|-----------|--------------|-----------|
| Dependent→ | Foo | q | Agric. raw n | naterials | Bevera | ges | Minera | als |
| Regressors↓ | LFOODR | t-values | LAGRMR | t-values | LBEVR | t-values | LMMR | t-values |
| Short-run coeffic | ients (elasticiti | es) | | | | | | |
| Constant | -0.68 | -0.74 | -0.85 | -1.6* | | | , | -0.32 |
| | 0.46 | 3.0*** | 0.36 | 1.5~ | 0.37 | 2.44*** | -0.60 | -0.66 |
| ΔLP_5 | | | | | 0.06 | 0.37 | -0.38 | -0.40 |
| ALFRTZ_1 | 0.11 | 1.63* | | | - | | | |
| ALIPGNP_5@ ALKG_2 | | | 0.12 | 0.24 | -0.07 | -0.60 | 0.49 | 1.18 |
| Long-run coeffici | ients (elasticiti | es) | | | | | | |
| LP 2 | 2.4 | 3.5*** | -0.16 | -0.35 | 0.84 | 2.5*** | 2.45 | 1.96** |
| LP_6 | | | | | 0.46 | 1.08 | -0.73 | -1.00 |
| LFRTZ_2 | -0.24 | -1.06 | | | | | | |
| | | | | | -0.01 | -0.04 | 1.03 | 3.15*** |
| LKG_3 | | | 1.02 | 5.6 | | | | |
| Dependent_1 | -0.20 | -3.01*** | -0.25 | -4.27*** | -0.25 | -3.94*** | -0.61 | -4.9*** |
| Diagnostic test | | - - - - - | - | | | | | |
| ECM adjusted R ² | 0.24 | n=114 | 0.11 | n=126 | 0.10 | n=108 | 0.27 | n=58 |
| Level adjusted R ² | 0.40 | | 0.25 | | 0.54 | | 0.83 | |
| Jarque-Bera | 1.19** | | 0.06** | | 4.5** | | 5.3** | |
| RESET(1) | 1.04 | pr.(0.31) | 0.07 | pr.(0.79) | 0.13 | pr.(0.71) | 0.04 | pr.(0.83) |
| Chow | 0.94** | | 1.93** | | 1.11**# | | 1.3** | |
| Degree of multicollinearity+ | Acceptable | | Acceptable | | Low | | Low | |
| Constant (country) dummy | Bť, | -0.65, | Gha, | -0.30 | Ben, | -0.89 | Ben, | -2.75 |
| used | Cam, | -0.37, | Sen | -0.32 | Car, | -0.62 | Car | -1.93 |
| | Car, | -0.88 | | | Gab | -0.97 | | |
| | Nig, | -0.35 | | | Sen, | -1.32 | | |
| | Ż | -0.81 | | | Ser | -0.57 | | |

154

Chapter 4

| Table 4.14 Estimation w | vith real exch | ange rate f | or West and | I Central Af | rica | | | |
|-------------------------------|-------------------|-------------|--------------|--------------|--------|-----------|----------|-----------|
| Dependent→ | Foo | q | Agric. raw r | naterials | Bevera | ides | Miner | als |
| Regressors↓ | LFOODR | t-values | LAGRMR | t-values | LBEVR | t-values | LMMR | t-values |
| Short-run coeffic | cients (elasticit | ies) | | | | | | |
| Constant | -0.28 | -0.72 | 0.04 | 0.10 | | | 0.13 | 0.14 |
| ALP_1 | 0.27 | 2.19** | 0.48 | 3.3*** | 0.18 | 1.96** | -0.12 | -0.19 |
| | | | | | 0.12 | 1.1 | 1.14 | 1.12 |
| <u>ALFRTZ_1</u> | 0.18 | 2.7*** | 0.01 | 0.26 | | | | |
| ALIPGNP_5@ | | | | | -0.04 | -0.31 | 0.28 | 0.50 |
| Long-run coeffici | ients (elasticiti | les) | | | | | | |
| LP 2 | 0.17 | 1.03 | 0.62 | 1.48~ | 0.50 | 2.32*** | -2.52 | -3.02*** |
| | | | | | 0.25 | 1.14 | 2.73 | 3.29*** |
| LFRTZ_2 | 0.42 | 2.06** | 0.03 | 0.21 | | | | |
| LIPGNP_6@ | | | | | 0.26 | 2.82*** | -1.44 | -1.39 |
| Dependent_1 | -0.15 | -2.88*** | -0.12 | -2.85*** | -0.26 | -3.64*** | -0.51 | -4.28*** |
| Diagnostic test | | | | | | | | |
| ECM adjusted R ² | 0.16 | n=111 | 0.09 | n≖133 | 0.09 | n=106 | 0.24 | n=58 |
| Level adjusted R ² | 0.26 | | 0.09 | | 0.65 | | 0.64 | |
| Jarque-Bera | 2.06** | | 0.05** | | 2.0** | | 3.29** | |
| RESET(1) | 0.45 | pr.(0.50) | 0.48 | pr.(0.50) | 0.95 | pr.(0.33) | 0.01 | pr.(0.91) |
| Chow | 0.58** | | 1.3** | | 0.85** | | 0.87** | |
| Degree of multicollinearity+ | Acceptable | | Very low | | Low | | Very low | |
| Constant (country) dummy | Sen | 0.33 | Ben | -0.15 | Car | -0.64 | Ben | -2.55 |
| used | z | -0.18 | Sen | -0.23 | Gab | -1.06 | Car | -2.65 |
| | | | | | Gha | 0.25 | | |
| | | | | | Nig | 0.76 | | |
| | | | | | Sen | -1.47 | | |

..... ¢ . . ; . . L ,

| Table 4.15 Estimation v | vith capital in | flow for W | est and Cen | tral Africa | | | | |
|-------------------------------|------------------|------------|--------------|-------------|---------------------|-----------|--------|-----------|
| Dependent→ | Foo | - | Agric. raw r | naterials | Bevera | lges | Miner | als |
| Regressors↓ | LFOODR | t-values | LAGRMR | t-values | LBEVR | t-values | LMMR | t-values |
| Short-run coeffici | ents (elasticiti | ies) | | | | | | |
| Constant | -2.28 | -2.07** | 1.08 | 1.28 | -1.29 | -0.84 | -3.84 | -0.87 |
| | 0.57 | 2.96*** | 0.35 | 1.26 | 0.37 | 1.55~ | 0.51 | 0.48 |
| ΔLP_5 | | | | | -0.03 | -0.17 | 0.05 | 0.05 |
| ALFRTZ_1 | 0.16 | 2.54*** | 0.03 | 0.58 | | | | |
| ALIPGNP_5@ | | | | | -0.25 | -1.96** | 0.47 | 1.16 |
| AFF_2^ | 0.04 | 0.65 | 0.21 | 2.55*** | 0.28 | 1.23 | -0.88 | -1.54~ |
| ∆FF_6^ | | | | | 0.19 | 1.7* | -0.37 | -0.75 |
| Long-run coeffici | ents (elasticiti | es) | | | | | | |
| LP 2 | 3.5 | 2.2*** | -1,14 | -1.45 | 1.35 | 1.42 | 4.0 | 1.96** |
| LP_6 | | | | | 0.05 | 0.09 | 0.23 | 0.18 |
| LFRTZ_2 | 0.50 | 2.39*** | 0.00 | 0.02 | | | | |
| LIPGNP_6@ | | | | | -0.86 | -2.16** | 0.58 | 0.82 |
| FF_3^ | -0.39 | -1.63* | 0.79 | 4.0*** | 0.62 | 1.57~ | -0.55 | -0.55 |
| FF_7^ | | | | | 0.67 | 1.7* | -0.54 | -0.67 |
| Dependent_1 | -0.14 | -3.72*** | -0.25 | -3.93*** | -0.24 | -2.9*** | -0.49 | -4.16*** |
| Diagnostic test | | | | | | | | |
| ECM adjusted R ² | 0.20 | n=106 | 0.11 | n=107 | 0.03 | n=97 | 0.18 | n=68 |
| Level adjusted R ² | 0.29 | | 0.28 | | 0.64 | | 0.70 | |
| Jarque-Bera | 0.13** | | 0.04** | | 1.47** | | 3.5** | |
| RESET(1) | 0.02 | pr.(0.88) | 0.38 | pr.(0.54) | 0.73 | pr.(0.39) | 1.06 | pr.(0.31) |
| Chow | 0.72** | | 1.3** | | 0.99** | | 1.25** | |
| Degree of multicollinearity+ | Low | | Low | | High | | Low | |
| Constant (country) dummy | Cen | 0.35 | a di | 30.0 | (0.79)++ Cam Gab | 05.06 | Bon | 36 |
| | 5 | 00.0 | | 24.0 | Che Nin | 0.00.0 | in C | |
| 7.00m | | | | i j | Sen Sen | 0.84 | 3 | r j |

156

In general, the estimation results for this region may be summarized as follows. For food and agricultural raw materials, both non-perennial and short gestation, 10–25 per cent of any disequilibrium in the previous period is made up for in the current period. However, in relation to beverages and minerals, this figure stands at the relatively higher levels of 25 per cent and 50–60 per cent, respectively. This result indicates a generally low level of adjustment to disequilibrium or shocks for the former, as well as a clear distinction between commodities that require a long and short gestation period. This stands in contradiction to the intuitively more appealing expectation that short gestation items are likely to adjust faster. (A possible explanation for this result is the importance of capacity utilization³³ in minerals and tree crops.)

4.3.5 Summing up

The results of the estimation are summarized in Table 4.16. I have reported only those values found to be statistically significant at a 10 per cent significance level or better. Within this table, the figures [1], [2] and [3] refer to estimations for Models I, II and III, respectively. In Model I, current price and capital formation indicators are used as regressors. In Model II, price is replaced by a commodity-specific regional real exchange rate. Finally, in Model III the basic estimation used for Model I is augmented by capital inflow indicators.

In this section, an attempt has been made to identify determinants of commodity export supply. This approach differs from previous studies in a number of important respects. Firstly, it emphasizes the role of other factors besides price in the determination of export supplies. Secondly, an error-correction model is used. And, thirdly it focuses exclusively on African countries.

Based on this analysis, a number of conclusions may be arrived at. Firstly, there would appear to be a clear distinction between short- and long-run elasticities. Hence, an ECM is likely to represent the relevant econometric technique in relation to commodity estimation. However, work in this area remains in its infancy. Secondly, estimation using relative prices (with a real exchange rate) yields largely satisfactory results. However, it is noted that the impact of relative prices is largely confined to capacity utilization and not to capacity creation. Thirdly, although capital formation indicators are neglected, within the literature, these are also found to have a positive and statistically significant impact, especially in the long run. As a result, specific parameters are ob-

| Table 4.16 Sumn | nary table of el | lasticities: A | UI Africa | | | | | |
|----------------------------|------------------------|------------------------------------|-------------------|-------------------|-------------------------------|-------------------------------|-----------|-----------------|
| | Current | : price | Relative | e prices | Capital fo | ormation | Foreign | inflows |
| | Shortrun | Long-run | Short-run | Long-run | Short-run | Long- run | Short-run | Long-run |
| West and | Central Africa (| NCA) | | | | | | |
| Food | 0.46[1] 0.57[3] | 2.4[1] 3.50[3] | 0.27 | | 0.11[1] 0.18[2] 0.16[3] | 0.42[2] 0.50[3] | | -0.39 |
| Agr. raw materials | 0.36[1] | | 0.48 | 0.62 | | 1.02[1] | 0.21 | 0.79 |
| Beverage | 0.37[1]* 0.37[3]* | 0.84[1]* | 0.18* | 0.50* | -0.25[3] | 0.26[2] -0.86[3] | 0.19# | 0.62* 0.67# |
| Minerals | • | 2.45[3]* | | -2.52* 2.73# | | 1.03[1] | -0.88* | |
| East and | Southern Africa | (ESA) | | | | | | |
| Food | 0.29[1] | 5.02[1] 1.01 [3] | | 0.13 | | | | 0.22 |
| Agr. raw materials | | -1.38[1] | 0.33 | | 0.18[1] 0.18[2] 0.22[3] | 0.44[1] 1.19[2] 1.16[3] | | |
| Beverage | 0.14[1]* | 0.54[1]* 0.30[1]# 0.7]3]# | | 0.12# | | 0.13[2] 0.58[3] | -0.24# | 0.58* -0.66# |
| Minerals | | | -0.42# | 1.27* | | 2.36[2] | | |
| North Afri | ica (NA) | | | | | | | |
| Food | | 0.90[3] | | | 1.14[3] | | | 0.15 |
| Agr. raw materials | | | | 2.40 | 0.62[2] | 1.53[3] | 0.68 | -0.43 |
| Minerals | | -0.91[1] -1.02[3]* -1.28[3]# | | 0.95# | -0.56[1] | 1.14[1] | | 0.58* |
| * = short lags (indicating | g capacity utilization | : | # = long lags (in | dicating capacity | creation) | | | 1 |

158

Chapter 4

tained for each region and for different commodity categories. This has an important policy implication in that, in addition to giving us one more policy handle other than price with which to work, it also implies boosting levels of capital formation (both internal and with aid), rather than providing welfare aid. This is likely not only to raise the level of exports, but also to have a multiplier effect on output working its way through imports. This result has particular relevance in the African context.

An important variation across regions may also be observed. Thus, if we examine the regional variation across commodity categories, the following picture emerges. For food, the short-run current price elasticity is found to be highest for the WCA region. In the long run, both WCA and ESA regions have higher price elasticities than the NA region. The short-run impact of capital formation indicators on food is also found to be strong for the NA region. A negative debt overhang effect is observed for WCA, but not for the other regions. In relation to agricultural raw materials, the short-run price effect is stronger within the WCA region, while the long-run price effect is found to be negative for the ESA region. A similar short-run phenomenon is observed for relative prices. A long-run relative price effect is found to be strongest within the NA region, followed by the WCA region. Capital formation indicators are strongest within the NA region, followed by the ESA region. A debt overhang problem is found to be present within the NA region, but not the WCA region. In relation to beverages, current and relative price and capital formation effects are strongest in the WCA region, followed by the ESA region. 'Dutch disease' and debt stress effects are observed for the ESA region, but not for the WCA region. Finally, in relation to minerals, a long-run current price effect is found to be strongest for the WCA region, and negative for the NA region.

The impact of the arguments used in the regression also varies across regions. Thus, the relative price effect is negative for the short lag of the WCA region and the short-run, long lag, of the ESA region. Longer lag, capacity creation, long-run effects are found to be strongest for the WCA region, followed by the ESA and NA regions. The impact of capital formation indicators is found to be strongest for the ESA region and nearly equal for the other two regions. Evidence of a 'Dutch disease' effect is apparent only within the WCA region. Based on the above observations, one might reasonably conclude that the response of different commodity groups varies across regions. Indeed, giving due attention to such differences is likely to be very important in analysing the pattern of trade and finance in Africa. In relation to the impact of foreign inflows on exports, no firm conclusion may be arrived at, since the results obtained are mixed. The results are also mixed *vis-à-vis* the relationship between long-run current export price and the supply of exports, when the impact of foreign inflows is explicitly considered within the model specification. Since it is not possible to arrive at any firm conclusion based on these results, this would tend to suggest the need for further research on this relationship, both at a regional and country level.

Although most published studies focus on output, rather than on export supply, nevertheless we will now attempt to compare the results of these studies with our own findings. Using relative prices, Bond (1987) reports aggregate supply price elasticities for Africa of -1.28 for food. 0.70 for agricultural raw material, and -1.89 for minerals. (For other global regions she found lower elasticities.) She attributes the apparently perverse relationship for food to population growth, as well as to the very large gap between world price and producer price. In an earlier study (Bond 1983) she found an average aggregate price elasticity of 0.12, based on a coefficient of the logarithm of real prices, with the figure for individual countries ranging between 0.03 and 0.22. Binswanger (1992) maintains that price elasticities for sub-Saharan Africa are not lower than for other areas and, in general, that long-run elasticities are higher than short-run ones. Binswanger's results show elasticity values ranging from 0.05 to 0.15, with the exception of Senegal and Burkina Faso, for which the figures are 0.54 and 0.22, respectively. The crosscountry result for this study stands at 0.06, long-run values fall in the range of 0.15 to 0.24 and no long-run elasticities were produced from the cross-country regression. Based on the results of his study of sisal in Tanzania, Gwyer (1971) found short-run price elasticity ranging from 0.24 to 0.29, with a long-run value of 0.48. For cotton in Uganda, Alibaruho (1974) found a short-run price elasticity ranging from 0.22 to 0.26 and a long-run value ranging from 0.44 to 0.66. Ghoshal (1974) found a one-period lag price elasticity of 1.16 to 1.71 in his study of rubber in Liberia. In their study of cotton in Kenva, Jones and Mutuura (1989) found short- and long-run elasticities of 1.33 and 1.71, respectively. Eriksson (1993) reviews a number of studies which examine supply price elasticity estimates for Tanzania. He reports elasticities falling in a range from 0.25 to 0.43 for perennials, 0.73^{34} to 2.4 in the

long run for cotton and a short-run value ranging from 1.5 to 2.3 for food crops. Also focusing on cotton in Tanzania, Dercon (1993) found that short-run elasticities for this crop ranged between 0.63 to 0.67. Ramanujam and Vines (1990) reported 'long-run' price³⁵ elasticities of 0.10 for food, 0.51 for beverages, 0.06 for agricultural raw materials and 0.33 for metals and minerals. The short-run price with no lag values stand at 0, 0.13, 0.08 and 0.31, respectively. Abebayehu (1990) estimated an export supply function for six African countries. His export price elasticity of supply is found to range from 0.50 to 1, if the exceptionally high value of 2.3 for Kenya is excluded.

Based on the above review, one might conclude that the findings detailed in this book stand in broad agreement with those of previous studies. More specifically, long-run price elasticities are generally found to be higher than short-run ones. However, the present study has an advantage over these other studies in that it explicitly clarifies what is meant by 'the long run' and 'the short run', using relevant estimation techniques. Moreover, the elasticity values generated in this study are stronger and more often positive than those arising out of previous studies. This may be attributed to the use of an error correction model and well-structured database with large degrees of freedom. Finally, since previous studies hardly touch upon other supply factors, a comparison with these studies in respect to such factors is not possible.

To sum up, the literature in relation to commodity export supply functions is characterized by explanatory variables that are either current or lagged relative prices. This study not only underlines the existence of other equally, or perhaps even more important factors, but also emphasizes the importance of their explicit incorporation within the estimation. Cross-sectional data for African countries is used for estimation purposes. Firstly, price-focused estimations are explored. Subsequent estimations are then undertaken by adding other relevant explanatory variables which may be rare within the literature. A number of important results are summarized in this section. Firstly, there is a clear difference between UNCTAD world prices and the regional price constructed for Africa. Hence previous work based on the former could have biased elasticities. Secondly, estimation using a real exchange rate resulted in statistically significant elasticity coefficients with a clear distinction between the long and short run. This underscores the importance of this particular specification as well as the use of an Error Correction Model (ECM) within this estimation. Thirdly, capital formation indicators are also found to be positive and statistically significant. Fourthly, the results of these estimations are found to vary across the three regions of Africa. And, finally, using the models detailed in this section, estimation with foreign inflow included yields mixed results both on the relationship between export prices and supply of exports and on the impact of aggregate foreign inflows.

4.4 AFRICAN EXPORTS IN A GLOBAL CONTEXT

4.4.1 Introduction

When data permits, one of the best ways of modelling a commodity market is to proceed along the lines outlined by Hwa (1985) and Ramanujam and Vines (1990). In these approaches, stock demand and supply is explicitly specified as a part of the system of consumption demand and supply equations, as set out in Chapter 2. The major problem in following this approach is a lack of commodity stock data, especially in relation to those disaggregated commodity categories used in this study. In order to overcome this problem, a simultaneous equation model of the demand for, and supply of, exports along the lines of Goldstein and Khan (1978), henceforth referred to as the GK model, is used as a starting point. Abebayehu's work on how this model may be applied to various African countries is also found to be useful, in this context (Abebayehu 1990). However, there are three main problems with the GK model. Firstly, the impact of stockholding is not explicitly incorporated within this model. Secondly, there is ambiguity in the use of a partial adjustment procedure in relation to the price equation of its disequilibrum model. And, thirdly, there may exist possible non-stationarity in relation to the series used for the estimation. I will return to examine some of these problems later in this study.

Most studies focusing on the export of commodities from developing countries take the individual national economy as their unit of analysis. Logically, such an approach will rest on the *small country assumption*. However, one of the major weaknesses of this approach is that it suffers from the *fallacy of composition* (Evans 1993) or the *adding-up* problem (Akiyama & Larson 1994). Thus, unless the modelling of exports for an individual country takes global effects into account, the results of such a modelling exercise may well be far from an accurate reflection on reality. The model proposed below attempts to resolve this problem by combining the global market of a commodity in question with the export market of the individual country or region. In this way the adding-up problem may explicitly be incorporated into the export function for individual or regional economies. In section 4.4.2 such a global model will be specified. Section 4.4.3 will be devoted to the estimation of the model. In this section some econometric problems of previous studies will be highlighted and a number of lessons drawn from this. Finally, section 4.4.4 will outline how this global model might be applied in relation to the macro supply situation in Africa.

4.4.2 The Global Commodity Model

The global demand for a commodity group *i* has two components. Firstly, a stock/inventory demand that is assumed to be determined by expected level of interest rate (i_w) and current prices. Agents in the North are assumed to hold commodities as part of their portfolio diversification scheme. The other part of aggregate demand is a consumption demand, be it industrial or otherwise. The latter is believed to be determined by the level of income in North $(Y_N)^{36}$ and by price of commodities (P_w) relative to an indicator of price for other trading partners (P_N) . This aggregate demand may be specified as,

$$\ln X^{dd} = a_0 + a_1 \ln Y_N + a_2 \ln i_w + a_3 \ln \left(\frac{P_w}{P_N}\right)$$
[1]

where: $a_1 > 0; a_2 < 0 a_3 < 0$

The supplies of exports are assumed to depend on price of commodities relative to domestic price in South (P_S), and on a productivity indicator taken to be the share of gross domestic investment in South (I_S) as percentage of the GDP in South (Y_S). This is given in equation [2],

$$\ln X^{ss} = b_0 + b_1 \ln \left(\frac{P_w}{P_s}\right)_{t-1} + b_2 \ln \left(\frac{I_s}{Y_s}\right)_{t-i}$$
[2]

where: b_1 and $b_2 > 0$.

This supply equation differs from that of the GK model in two respects. Firstly, it uses lagged, as opposed to current, price levels. Secondly, a lagged investment to GDP ratio, rather than current GDP is used as a productivity indicator. We opt for the use of lagged price as a determinant of supply on the assumption that current supply is affected by price signals for the previous period. Similarly, the use of an investment ratio is likely to overcome the problem of taking GDP as a productivity indicator, in a situation where the export sector might grow, owing to previous period investment. This remains true even if the overall performance of the economy, as depicted by GDP, is poor.

Given these refinements, the equilibrium model may be arrived at by simultaneously estimating equations [1] and [2] under the assumption that they are in equilibrium,

$$X^{dd} = X^{ss}$$

The disequilibrum model

A disequilibrium version of this model may be formulated by assuming a partial adjustment model, which is adapted from a similar earlier formulation within the export function, alongside the GK model (Houthakker & Taylor 1970). Thus, the change in exports are assumed to partially adjust to the gap between demand and actual flows in the previous period,

$$\Delta \ln X_t = \lambda \, \left(\ln X_t^{dd} - \ln X_{t-1} \right) \tag{4}$$

where λ is the coefficient of adjustment and is defined as $0 < \lambda < 1$, and Δ is a first difference operator representing change. Equation [4] states that the quantity of export adjusts to the condition of excess demand and that there is some degree of stickiness in the supply conditions. Substitution of equation [1] into [4] yields the following equation for exports:

$$\ln X_{t} = \alpha_{0} + \alpha_{1} \ln Y_{N} + \alpha_{2} \ln i_{w} + \alpha_{3} \ln \left(\frac{P_{w}}{P_{N}}\right) + \alpha_{4} \ln X_{t-1}$$
[5]

where:

$$\alpha_0 = \alpha_0 \lambda \ \alpha_1 = a_1 \lambda > 0 \ \alpha_2 = a_2 \lambda < 0 \ \alpha_3 = a_3 \lambda < 0 \ \alpha_4 = 1 - \lambda > 0$$

Normalizing equation [5] for world commodity price yields the estimating disequilibrum equation given by [5.1],

$$P_{w} = \beta_{0} + \beta_{1}Y_{N} + \beta_{2}i_{w} + \beta_{3}P_{N} + \beta_{4}X_{i} + \beta_{5}X_{i-1}$$
[5.1]

where:

$$\beta_0 = \frac{-\alpha_0}{\alpha_3} \ \beta_1 = \frac{-\alpha_1}{\alpha} > 0 \ \beta_2 = \frac{-\alpha_2}{\alpha_3} < 0 \ \beta_4 = -\frac{1}{\alpha_3} > 0 \ \beta_5 = -\frac{\alpha_4}{\alpha_3} > 0$$

In the version of the GK model which is applied to a sample of African countries by Abebayehu, the disequilibrum price equation is formulated following a similar procedure to that of the supply equation (Abebayehu 1990). That is, the change in price of exports adjusts to excess supply in a similar partial adjustment manner.³⁷

However, one of the major weaknesses of the GK model is the use of this type of partial adjustment model in relation to prices. If there is excess supply in the market, either the market clears for price or there must be some kind of non-price market clearing mechanism, such as a stock adjustment or, indeed, a related activity which might have the effect of neutralizing excess supply. Since a partial adjustment process in relation to prices is assumed within the GK model³⁸ the authors must have implicitly assumed that the excess supply, which is not taken care of by a movement in the level of prices, has somehow adjusted. However, Goldstein and Khan (1978) do not spell out what this mechanism might be. In contrast, since we have explicitly assumed the demand for stocks and inventory in the demand equation of our model, price will automatically clear that market. Thus, although the equation for export supply may follow a partial adjustment model, the price equation does not. As a result, our disequilibrum model is arrived at by simultaneously estimating equation [2] and equation [5], based on the equilibrium condition expressed in equation [3].

4.4.3 Estimation of the Model

4.4.3a Time series properties

A second major problem with the GK model, particularly as it is applied in Abebayehu (1990), is the assumption of stationarity. As can be inferred from Tables 4.17 and 4.18, all series, with the exception of real level of GDP in the North,³⁹ are found to be non-stationary. These series are also found to be I(1). This finding casts serious doubt over the validity of most supply and demand commodity models since the findings of these models may be spurious.

| Series (all in logarithms) | No trend & inter- cept term | With trend & intercept term | With intercept alone |
|---|---|--|---|
| 1966-1993 | | | |
| Real export Agricultural raw materials Beverage Food Minerals World interest rate (3 months LIBOR rate) Real GDP in North | 1.24 1.57 1.39 0.35 -0.63 1.36 | -1.88 -2.2 -2.53 -4.0 -2.34 -3.45 | -0.44 -0.43 -0.43 -3.46 -2.6 -1.47 |
| 1960-1993 World price Agricultural raw materials Beverage Food Minerals CDI to CDP ratio (South) | 1.05 0.25 0.55 0.65 | -1.9 -0.80 -2.4 -3.34 | -1.02 -1.41 -1.8 -2.0 |
| | 0.42 | 2.0 | 1.0 |

Table 4.17 A unit root analysis of the world commodity model and ADF test

Table 4.18 Mackinnon critical values for the ADF test

| No tre | nd & interce | ot term |
|----------|---------------|----------|
| 1% | 5% | 10% |
| -2.64 | -1.95 | -1.62 |
| With tre | end & interce | ept term |
| 1% | 5% | 10% |
| -4.27 | -3.56 | -3.21 |
| Witl | n intercept a | one |
| 1% | 5% | 10% |
| -3.65 | -2.96 | -2.62 |

There are two means by which this issue might be addressed. The first option is to conduct a co-integration analysis and apply an ECM model to this. However, the problem with this option is the limited size of the sample relating to the period 1960 (65) to 1993. Nevertheless, in spite of this shortcoming, our data is still relatively better than that used

in pre-existing long-run estimations. The most recent study of Ramamujam and Vines (1990) also uses data from 1960–86 for only a limited set of commodities. The advantage of this option is that it, at least, highlights a medium-term relationship. The second option is to transform all data to an I(0) series using the first difference values for this data.⁴⁰ The major shortcoming of this option is that information is lost in relation to the long-run (level) relationship. (Estimation results based on this approach are reported in Appendix 4.3.)

| Equation | Engel- Granger | Mackinnon critical | Johansen likelihood | Mackinnon d | critical value |
|--|-------------------|--|--|---|---|
| • | ADF statistics | value at 5% | ratio | 5% | 1% |
| Food | | | | | |
| Demand Supply Demand Supply ^d Supply ^e | -5.91 -2.72 | -5.86ª -4.14 ^b | 67.50 39.70 98.17° 50.72 ^d 56.87 ^e | 68.52 47.21 87.31 39.89 53.12 | 76.07 54.46 96.58 45.58 60.16 |
| Agricultural ra | w materials | | | | |
| Demand Supply | -3.36 -2.84 | -4.55 ^b -4.14 ^b | 84.42 43.12 | 68.52 47.21 | 76.07 54.46 |
| Beverage | | | | | |
| Demand Supply | -4.24 -4.38 | -4.97 ^⁵ -4.13 ^⁵ | 83.12 51.77 | 68.52 47.21 | 76.07 54.46 |
| Minerals | | | | | |
| Demand Supply | -7.01 -5.83 | -5.87° -5.39° | 92.60 60.16 | 68.52 47.21 | 76.07 54.46 |

Table 4.19 Co-integration test for world commodity model

Notes:

^b [^a] are Mackinnon critical values at 10% [1%] level of significance (others at 5%). The Johansen test used assumes a linear deterministic trend in the data with intercept, no trend, in the test VAR (i.e., the co-integration equation (CE), as an indicator of long run equilibrium relation, has no trend).

 $^\circ$ assumes linear deterministic trend in the data with intercept and trend in CE and no trend in test VAR.

^d assumes no deterministic trend in the data with no intercept or trend in the CE or test VAR.

^e assumes no deterministic trend in the data with intercept, no trend in CE and no intercept in test VAR.

AGRICULTURAL RAW MATERIALS

Demand

 $\Delta X = -0.174 \Delta P - 0.375 \Delta Y_N - 0.145 i_{W_{t-1}} - 0.673 (X_{t-1} + 0.708 P_{t-1} - 0.02 PY_{t-1} + 0.372 i_{W_{t-2}} - 5.504)$

TROPICAL BEVERAGE

Demand

 $\Delta X = -0.319 \Delta P + 0.90 \Delta Y_N - 0.003 i_{W_{t-1}} - 0.777 (X_{t-1} + 0.250 P_{t-1} - 0.322 Y_{t-1} - 0.110 i_{W_{t-2}})$

 $(-4.8)^{*} (2.2)^{+} (-0.04) (-3.2)^{*} (-5.1)^{*} (31.2)^{*} (1.64) \sim \mathbb{R}^{2} - \operatorname{adj} = 0.50 \text{ Jarque-Bera 0.12}^{*} \operatorname{Reset 1.07(0.59) LM 0.30(0.59) BG=0.33 (0.57) Sample 1968-93$

Supply

 $\Delta X = 0.003 \Delta P_{i-2} - 0.146 \Delta I_{i-5} - 0.590 (X_{i-1} - 0.098 P_{i-3} + 0.580 I_{i-6} - 6.087)$

FOOD

Demand

R²-adj = 0.92 Jarque-Bera 0.61* Reset 0.05(0.83) LM 0.23(0.64) BG=5.7 (0.01) Sample 1968-93 $\Delta X = -0.915 \Delta P - 0.356 \Delta Y_N - 0.194 i_{W_{l-1}} - 0.389 (X_{l-1} + 1.01 P_{l-1} - 0.583 PY_{l-1} + 0.049 i_{W_{l-2}} + 2.219)$ $(-14.0)^{*}$ (-1.5) $(-3.4)^{*}$ $(-2.3)^{*}$ $(-10.0)^{*}$ $(4.5)^{*}$ (-0.5) (-1.13)

 $\Delta X = -0.164 \Delta P_{i-1} - 0.738 \Delta I_{i-2} - 0.522 (X_{i-1} - 0.247 P_{i-2} + 0.246 I_{i-3} - 4.707)$ (3.6)* (-0.23) (1.32) (-3.7)* (-1.05) (-1.5) Supply

\mathbb{R}^{2} -adj = 0.30 Jarque-Bera 1.98* Reset 0.18(0.68) LM 8.0(0.01) BG=0.75 (0.49) Sample 1967-93

MINERALS

Demand

 $\Delta X = -0.312 \Delta P + -0.424 \Delta Y_N - 0.01 i_{W_{t-1}} - 0.527 (X_{t-1} + 0.646 P_{t-1} - 0.396 Y_{t-1} + 0.215 i_{W_{t-2}})$

$$(-2.8)*$$
 $(1.8)^{\circ}$ (-0.14) $(-2.65)*$ $(-7.3)*$ $(29.8)*$ $(-2.5)*$

R²-adj = 0.42 Jarque-Bera 1.74* Reset 5.6 (0.03) LM 1.2 (0.28) BG=1.74 (0.42) Sample 1968-93

Supply

 $\Delta X = 0.073 \Delta P_{t-1} + 0.487 \Delta I_{t-5} - 0.797 (X_{t-1} - 0.111 P_{t-2} - 0.607 I_{t-6} - 2.04)$ (1.09) (1.8) (-4.4)* (-2.8)* (2.03)⁺ (2.1)⁺

 \mathbb{R}^{2} -adj = 0.45 Jarque-Bera 1.58* Reset 0.56 (0.47) LM 0.66 (0.43) BG=0.66 (0.42) Sample 1967-93

Values in bracket are t-values while for diagnostic test P-values (See Appendix 4.1). The sign difference between the t and the coefficients in the ECM term is because the t values refer to the long run coefficient. * implies 1%, + 5% and ^ 10% level of significant. LM is hetroschedasticity test while BG is the Breuch-Godfery test for serial correlation. The 5% (2 degrees of freedom) value for Jarque-Bera is 5.99.

Although our data series are small,⁴¹ we have nevertheless undertaken a co-integration analysis using the available data. The results for this are reported in Table 4.18. It is noteworthy that for two of the models,⁴² the Engel-Granger test rejects the null hypothesis of co-integration while the Johansen test accepts this. This is mainly due to the limitations of the former test, in which a long-run co-integration regression is undertaken by taking the theoretically specified equation. Since the Johansen test, which is comparatively the better test, accepts all but one of these⁴³ estimations (albeit with some changes in assumptions with regard to the food equation), we accept the hypothesis of the existence of a long-run relationship. The ECM estimation which follows from this analysis is given in the following section.

4.4.3b The econometric model and estimation results

Table 4.20 shows the result of the ECM model fitted to data for the period running from 1960 to 1993. However, the export data relates only to the years between 1966 and 1993. Within this table, quantity of exports from South is denoted by X. Export figures for each commodity category relate to total exports from the South.⁴⁴ The price of each commodity group, which is derived from the UNCTAD world price, is denoted by P. Within the demand equation, the commodity price (P) is set relative to a general import unit price index of the North. The latter, following a similar approach to that adopted in the GK model, is taken as a proxy for the price of other trading partners. Within the supply equation, the commodity price (P) is set relative to domestic price in South (P_S) . The investment to GDP ratio in South is denoted by I and the two periods moving average three months dollar-based LIBOR rate by i_w . The expected interest rate at time t is assumed to be formulated on the basis of knowledge acquired in the preceding two periods. A rudimentary procedure of expectation formation based on a moving average value for the previous two periods is assumed.⁴⁵ Northern (OECD) real GDP is denoted by Y_N . All variables are in logarithms. Hence, coefficients may also be read as elasticities.

Our ECM model has an edge over similar previous models because we are estimating the model for different commodity categories which had previously been aggregated into one single 'total export' category. Moreover, previous estimations (Goldstein & Khan 1978, Abebayehu 1990, Ramanujam & Vines 1990) do not formally address the issue of spurious regression. The result in Table 4.20 broadly provides the expected signs. The diagnostic test for all estimations is quite acceptable. However, compared to the supply equations, the demand model is quite robust.⁴⁶ In addition to the benefits associated with the use of an appropriate methodology, our model also provides additional insight into the impacts of interest rate on commodity demand. (An alternative estimation using a short-run argument is provided in Appendix 4.3.)



Diagram 4.4 The structure of the commodity sub-bloc

4.4.4 Locating African Commodity Trade in the Global Market

In this section, an attempt is made to combine the global commodity model formulated above with the export supply models specified in section 4.3. The combined model works at two levels. Firstly, the supply of exports from all South together with the demand for such exports from the North will determine the world price of commodities. Here the market is assumed to be a flexi-price one, with price determination following a Marshallian market period in the very short run. The current world price, which is determined from this market, is taken as the relevant information signal for African producers, who are assumed to respond accordingly in the coming period. The impact of this world price on African supply may be envisaged in both the short and the long run. The error correction formulation for African supply reflects these separate effects. These two sub-models relating to African supply and the global market are combined in Chapter 6. Diagram 4.4 sketches how the commodity sub-bloc of our North-South model works.

4.5 CONCLUSION

The main concern in this chapter has been to outline the determinants of foreign exchange supply to Africa. We noted that there are two basic sources of foreign exchange for Africa: namely, capital inflows (coming both as FDI and as aid) and export earnings. The flows of capital to Africa are found to depend on the relative (to Other South) position of Africa in terms of economic, political and strategic self-interest indicators. This finding will be used in building the allocation of capital flows section of the North-South Model developed in Chapter 6.

The second important source of foreign exchange for Africa is derived from the export of primary commodities. Three competing models of export supply for the three regions and four commodity categories are analysed. Although the results are mixed, the model based on the real exchange rate (Model II) is found to be comparatively the best. This model underlines the importance of both relative prices and capital formation indicators. However, in section 4.3 I have argued that the global commodity market does, in fact, influence the level of prices with which African exporters are faced. In order to address these issues, global demand and supply equations for Southern exports are specified and estimated. This allows one to locate African export supply within a global commodity market framework. This formulation is pursued in the commodity sub-bloc of the model developed in Chapter 6. However, before proceeding to undertake such an analysis, the macroeconomic impacts of capital inflows will first be examined empirically in the following chapter.

172

5 External Finance and the African Economy: A Macro Approach

5.1 INTRODUCTION

In the previous chapter, some determinants of the supply of external finance were discussed. Such flows have a number of macroeconomic implications that require careful management. Specifically, three such implications may be outlined. The first of these relates to the recipients' fiscal response in the face of such flows. Section 5.2 will discuss this issue in the context of the wider literature on recipients' fiscal response. The second and related impact of such flows relates to the upward pressure, which they place on the level of domestic prices, and the resulting appreciation of the real exchange rate. As discussed in Chapter 2, this falls within the 'Dutch disease' literature. Section 5.3 will examine this issue in relation to Africa. In section 5.4 the positive impacts of external finance will be examined. In most African economies this relates to the possibility that external finance offers to finance imports. The latter, in turn, will have a positive effect on investment and growth, by relieving the import compression situation discussed in Chapter 3. An econometric analysis focusing on this issue is given in section 5.4. Finally, section 5.5 brings the chapter to a close by highlighting a number of conclusions, arising out of the preceding discussion.

5.2 THE FISCAL RESPONSE TO EXTERNAL FINANCE

5.2.1 Introduction

There is a growing consensus that capital inflows – which, in Africa, predominantly come in the form of aid – will have serious macroeconomic ramifications. In this section, I will focus on one aspect of this impact, relating specifically to the recipients' fiscal response to such flows. Critics of Chenery and Strout's (1966) two-gap model have argued that foreign capital inflows, and aid in particular, may represent a useful substitute for savings (Griffin 1970, Griffin & Enos 1970, Weisskopf 1972). Indeed, these commentators note that a large fraction of such flows is likely to be used to increase consumption. Papanek (1972), on the other hand, argues that these critics have mistaken association for causation.¹ This 'aid-saving debate' has continued for nearly three decades.² In his original work, Griffin (1970) suggested different avenues through which a decline in savings might take place. The first such avenue could arise as a consequence of capital inflow's negative impact on government's efforts to levy or collect tax. Indeed, such inflows might even provide an incentive for governments to reduce tax. Similarly, these inflows might encourage governments to increase public consumption. The second avenue, suggested by Griffin, relates to the presumed negative effect of capital inflows on private savings and, finally, to its stimulating impact on consumption of importables and exportables (Griffin 1970: 106-7). The first avenue, taken together with the nature of public expenditure in recipient countries, has been articulated around what White (1992) terms the 'fiscal response' literature. This literature builds on the works of Heller (1975) and its extension by Mosley et al. (1987). This section also addresses this issue by focusing on two types of taxes which, in previous studies, are usually aggregated, as well as on government consumption expenditure.

Heller's original contribution is articulated in a model that assumes that public sector decision-makers maximize a utility function comprising public investment, public consumption³ and borrowing from domestic sources (Heller 1975). By taking the variables as a deviation from their targeted values, the utility function is operationalized through a functional form that ensures diminishing marginal utility from each of the choice variables. The targets are further specified as a function of current and lagged values for a number of macroeconomic variables. These are imports, private investment, income and a social variable representing primary school enrolment. By maximizing this function, subject to the constraint that government expenditure should equal revenue (including foreign grants to the public sector and public foreign loans), Heller solves for the set of first-order equations that vield structural equations for the estimation. Using data from various African countries, the study highlights a number of broad findings. Firstly, the impact of foreign inflows (and especially those coming in grant form) on taxation levels is generally found to be negative. Secondly, foreign loans and other types of inflow are found to have an even greater negative impact on levels of domestic borrowing. Thirdly, inflows coming as grants are partially allocated to a consumption budget. Fourthly, foreign loans appear to lead both to an increase in total expenditure, and to the restructuring of the mode of domestic financing and pattern of public expenditure. (A full mathematical formulation of Heller's model is attached in Appendix 5.1.)

Mosley et al. (1987) have basically adapted and extended Heller's model by including aid's impact on growth of output and minor changes in the presentation of the public sector variable specification. Aid's impact on growth is considered by further specifying the target level of government investment. Thus, in Heller's original model, the target level of public investment is given as a function of lagged income and current private investment. Mosley and colleagues made two extensions to this model. Firstly, the private investment section of the model is further specified to include the price effects (a 'Dutch disease' type of relative price change) of aid inflows. Secondly, income is specified as being determined by a production function comprising public and private capital stocks and labour. Their finding, which is based on an OLS estimation, is that 'aid in the aggregate has no demonstrable effect on economic growth in the recipient countries'. This, they suggest. holds equally true both for the 1960s and the 1970s. Mosley and colleagues conclude that this may be due to the possibility of leakage into non-productive expenditure in the public sector as well as the transmission of negative price effects to the private sector.

One of the criticisms of Heller's model, which is addressed in the Mosley study, relates to the possible indirect positive impact of aid on taxes through its presumed positive impact on output (income). Other studies which adopt Heller's basic formulation have also been criticized along similar lines (McGillivray 1994, White 1994). Thus, White (1993) has attempted to reformulate Heller's model by explicitly considering these indirect effects. Closer examination of White's (1993) reformulation reveals that the new (demand-driven)⁴ model is based extremely heavily on the assumed existence of a Keynesian closure. (In other words, the model assumes the existence of a demand-constrained developing economy.) This assumption is taken almost as axiomatic, and hence, various conclusions, including the possibility that aid may raise taxes are drawn, based on this. However, it could be argued that

such an assumption cannot be sustained, in light of the substantial literature relating to the 'Dutch disease' effect. Further, the fact that most developing countries, particularly in Africa, are, in fact, supply-constrained,⁵ would tend to throw further doubt on this assumption. Moreover, even if we accept as reasonable the incorporation of this supply argument, then its possible impact on income is likely to be felt only after a certain lag. Within an agricultural economy this lag may be as long as a year for annual crops, and longer for perennials. Consequently, the effect of the previous period level of capital inflow will already be incorporated within the *current* level of income. This fact stands as the main justification for using such variables in equations [1] and [2] below of the model used in this study.

However, in spite of the above reservations, empirical results supporting the Heller-Mosley studies have been documented for many countries (see White 1992: 194). Notwithstanding this fact, two recent empirical studies have produced quite different results. Thus, Gang and Khan (1991) tested the above model using country data for India and a three-stages least-square technique. They conclude that the results of this analysis may be taken as confirming Heller's findings that in the presence of aid, there exists a negative relationship between taxes and government investment. However, Gang and Khan reject Heller's other main finding that grants and loans by and large go towards financing consumption. Research undertaken by Pack and Pack (1990) has also produced results which would appear not to accord with those of the Heller-Mosley studies. Based on an econometric analysis of a fourequation model depicting the fiscal posture of the Indonesian economy,⁶ they conclude that foreign aid actually stimulates total developmental public expenditure. Further, they note that nearly half of the revenue for such expenditure was raised by the Indonesian government from its own sources. However, the R^2 value reported, 0.99 per cent in almost all cases, leads one to question whether this result is free from spurious regression problems.⁷

To sum up, the recipient fiscal response literature holds that the government welfare (Mosley et al. 1987) or utility (Heller 1975) function may be maximized by narrowing the gap between the desired and actual performance of the government, subject to financing constraints, of which foreign inflows represents just one. One major shortcoming of this approach is the assumed existence of such a decision-making framework in the form of a public sector utility function. Even if one accepts this assumption à la Freidman (1953), nevertheless such a framework has a serious shortcoming, in that an additive assumption is included within the utility/welfare functions. Clearly such an assumption cannot hold if the arguments used in the utility/welfare functions are interdependent, as is the case in these functions.⁸ A further shortcoming of this approach stems from the fact that expenditure and revenue equations are set as binding constraints, inasmuch as the approach implicitly assumes that the utility function is the adjusting variable. However, by incurring a deficit, the budget constraint itself could, in practice, also become the adjusting variable.

Thus, a number of major drawbacks may be identified in relation to the fiscal response literature. Firstly, the decision-making framework is not plausible, not only because it is inconsistent within its own framework⁹ but also because decision-makers in most developing countries actually use a different framework, outlined below. Secondly, the econometric approach employed in these studies is extremely weak. Specifically, most series used in these studies have not been tested for the stationarity assumption. In fact, our own analysis of the series for various African economies reveals that these are not, in fact, stationary. Naturally, this would tend to imply that previous regression results reported in the literature might be spurious. Moreover, no diagnostic test is reported for these results. Indeed, for some of these cross-country studies, quite important tests, such as for parameter stability, have not been conducted. Thirdly, the tax variables used in the literature relate to aggregate taxes. Disaggregating taxes into different categories, such as direct and indirect, is likely to yield quite different results. Finally, the study relating to Africa as a whole not only is restricted in its sample size, but also fails to examine the variation across different African regions. The present study, therefore, represents an attempt to rectify these various shortcomings. Thus, in section 5.2.2 an alternative decision-making framework will be outlined and a relevant model and the equations used for the estimation will be specified. The model will focus on two types of taxes and government current expenditure. Section 5.2.3 will report on some of the estimated results and discuss their implications. Finally, section 5.2.4 will highlight a number of conclusions. based on the preceding discussion.

5.2.2 The Model

As has been argued above, the decision-making framework of utility maximization used in the fiscal response literature not only is inconsistent, but also does not reflect how decisions about financing are actually made in most African countries, in which foreign inflows are important. An alternative approach,¹⁰ which tallies more closely with the stylized facts in relation to Africa, may be outlined as follows. Firstly, policy-makers will have a targeted level of expenditure, based on projected growth and social development objectives. These policy-makers are confronted with three costly means of financing this desired level of expenditure. The first alternative is to finance this through foreign capital inflows. However, such an option brings with it implicit costs, in the form of policy conditionality, as well as the accumulation of a burden of debt.¹¹ The second alternative is to finance this expenditure by raising revenue domestically. The implicit costs of this option arise from the institutional problems as well as political implications associated with having to levy taxes. Finally, if the desired level of expenditure cannot be financed through these two mechanisms, governments may be forced to resort to deficit financing or expenditure reduction to the extent that is tolerable. However, such an option is likely to have costly political implications in the form of inflation and possible social unrest.

The choice of these alternative mechanisms moves recursively from the first to the third, depending on the availability of foreign finance as well as the associated implicit political/developmental cost attached to each by policy-makers.¹² If, for geo-political or economic supply reasons, foreign inflows are made easily available, then this will represent the least-cost financing mechanism, since it simultaneously minimizes the political risk associated with deficit financing¹³ and, at the same time, ensures the desired economic and social expenditure. However, such inflows might have the effect of discouraging domestic revenue collection, or even encouraging public expenditure,¹⁴ sometimes in an undesired direction, as a consequence of fungibility. Furthermore, the composition of public expenditure also matters. Thus, if the foreign exchange demand component of the public expenditure is high, then government may be faced with its own foreign exchange constraint, à la Chenery and Strout (1966). This, in turn, will put further pressure on the use of foreign inflows. In this recursive approach to the financing of public expenditure, which instrument of financing should be used at a particular point in time is entirely an empirical question. More specifically, this choice will depend, to a great extent, on the availability and timing of foreign inflows and the associated valuation of different instruments of financing by policy-makers.

Such a decision-making framework has a simple and clear implication for empirical work. From the point of view of the government of a developing country, the supply of foreign inflows may be assumed to be exogenously determined. However, such inflows are likely to affect both tax revenue and government expenditure. Thus, an econometric model with a recursive structure, which simultaneously allows for adjustment to past disequilibrum is likely to represent a relevant estimation technique. This will allow individual equations (numbered [1] to [3] below) to be estimated using Ordinary Least Squares (OLS), after their formulation within an Error Correction Model (ECM). Following this approach, three sets of equations, derived from a simple model that attempts to capture the recipients fiscal response to foreign inflow are estimated using pooled time series and cross-country data (1970-90) for the three regions covered in this study. Thus, in the following section, a model will be outlined which attempts to depict the fiscal posture of a typical African economy. This model comprises three behavioural equations and one closure. Although the closure may be used for simulation purposes, nevertheless its inclusion does not have a direct bearing on the estimated equations. The scope of the model here is limited to examining the impact of foreign inflow on taxes and government current expenditure.

(i) Direct tax

Direct tax (T_d) is set as a function of economic activity (Q) and capital inflow (F). This function in log-log form is given by equation [1]. The log transformation is used throughout in order to exploit its dual advantages of correcting skewed distribution and tackling possible heteroscedasticity.

 $\ln T_d = \alpha_1 + \beta_{11} \ln Q + \beta_{12} \ln F \text{ or}$ $T_d = AQ^{\beta_{11}} F^{\beta_{12}} \text{ where: } \alpha_1 = \ln A \qquad [1]$ where: $\beta_{11} > 0$; $\beta_{12} < 0$

By increasing the tax base, economic activity (Q) is believed to have a positive impact upon tax revenue. In the equations which are estimated below, the correlation coefficient between Q and F is checked to avoid possible multicolinearity. Only when such a correlation coefficient attains an acceptable value is the estimation carried. (Such a diagnostic check is also undertaken for equations [2] and [3] below.)

(ii) Indirect taxes

The level of indirect taxes in most African countries is closely related to external economic activity, and specifically to the level of exports (X) and imports (M). The level of these taxes is also directly related to private consumption expenditure (C_p) , since such taxes are usually imposed on privately consumed commodities. Finally, foreign inflows are also allowed to affect this type of taxation. This is given formally by equation [2]:

 $\ln T_{i} = \alpha_{2} + \beta_{21} \ln C_{p} + \beta_{22} \ln(X + M) + \beta_{23} \ln F$ where: $\beta_{21} > 0$; $\beta_{22} > 0$; $\beta_{23} < 0$ [2]

(iii) Government current expenditure

The level of current government expenditure (G) is assumed to be positively related to total revenue (T) and foreign inflows (F). It is also specified to be positively affected by external interest payments on both concessional (i_{w*}) and nonconcessional (i_{w}) loans. The final argument used in this function is a lagged value of the dependent variable. This is included in order to portray the persistence of previous patterns of expenditure. This is formally given in equation [3]:

$$\ln G = \alpha_1 + \beta_{31} \ln T + \beta_{32} \ln F + \beta_{33} \ln i_{w^*} + \beta_{34} \ln i_w + \ln G_{t-1} \qquad [3]$$

where: $\beta_{3i} > 0$; for i = 1...4

These three behavioural equations [1]–[3] may be closed by a public deficit [DEF] equation. This equates the difference between government total expenditure (government current expenditure, G, and government investment, I_g) and total revenue (which is the sum of total tax revenue, $T_d + T_i$, other government revenue, T_o , borrowing from domestic banks Z_b , and resource transfer from the private sector, Z_p) with the total capital inflow. The last two items, Z_b and Z_p , are balancing items that could be derived by subtracting public saving, current revenue minus current expenditure, and foreign savings, F_g , of the government from public investment, I_g .¹⁵ This closure is given as equation [4].¹⁶

$$Z = Z_b + Z_g = G + I_g - (T_d + T_i + T_O + F)$$
[4]

5.2.3 Estimation Results

Data sources

Government current expenditure is obtained from IMF Government Financial Statistics (GFS) and indirect taxes from World Bank (World Tables 1994). Direct taxes are derived as the difference between total government current revenue (World Bank, World Tables or African Development Indicators, ADI, and IMF's GFS) and the sum of indirect taxes and other tax revenues as defined in the IMF's GFS. Finally, capital inflow is derived by taking the difference between current and preceding year's stock of bilateral, multilateral and private debt stock plus grants (World Bank, World Tables). Grants are taken from the IMF's Balance of Payment data tape (and relate to lines 39 and 40 of IMF's Balance of Payment, or BoP. Yearbook, dealing with inter-official transfers). Concessional interest payments are taken from World Debt Tables (World Bank) while non-concessional interest payments are derived by multiplying loans from commercial banks by the three months LIBOR rate. All flows are considered since all are capable of placing pressure on tax and current expenditure variables.

The classification of countries is based on that set out by the United Nations Economic Commission for Africa (UN-ECA).¹⁷ As in the previous chapter, the data is based on eight countries within the ESA region, ten countries in the WCA region and three countries in the NA region. Each relevant variable is taken from a consistent macroeconomic database (1970–90) built along the lines discussed in Chapter 3.

| Series (all in landithma) | No trend | a & interce | pt term | With trer | id & interc | ept term | With | Intercept a | lone |
|----------------------------------|----------|-------------|---------|-----------|-------------|----------|-------|-------------|-------|
| Series (all in logariums) | ESA | ٩N | WCA | ESA | ٩N | WCA | ESA | NA | WCA |
| Capital flows: | | | | | | | | | |
| Total | 0.34 | 0.61 | -0.68 | -5.20 | -2.10 | 4.66 | -5.20 | -1.70 | -4.54 |
| Bilateral | -0.28 | 0.27 | 0.12 | 4.70 | -1.80 | 4.09 | 4.10 | -1.80 | -3.84 |
| Grants | -1.00 | -1.26 | -1.33 | -5.40 | -3.55 | -3.52 | -5.20 | -3.58 | -3.54 |
| Multilateral | -2.30 | 1.26 | 1.69 | -2.90 | -2.22 | -3.44 | -2.80 | -2.27 | -3.46 |
| Private | -0.27 | -0.10 | 0.20 | -3.90 | -5.20 | -3.49 | 4.10 | -4.14 | -3.32 |
| Concessional interest rate | -1.20 | 0.06 | -1.91 | 4.30 | -2.29 | -3.87 | -4.40 | -2.19 | -3.68 |
| Direct taxes | 0:90 | 0.67 | 0.62 | -1.75 | -2.80 | -2.23 | -1.65 | -2.09 | -2.19 |
| Export plus imports | 0.19 | 1.03 | -0.07 | 4.40 | -1.97 | -3.15 | -4.50 | -1.19 | -3.01 |
| Government expenditure | 3.00 | 1.83 | -0.09 | -1.94 | -3.59 | -3.35 | -2.40 | -3.60 | -3.45 |
| Indirect taxes | -0.03 | 0.91 | 0.02 | -3.80 | -2.06 | -3.61 | 4.00 | -0.98 | -3.12 |
| Non-concessional interest rate | -4.30 | -1.46 | -1.91 | -5.00 | -3.20 | -3.87 | -5.00 | -1.91 | -3.68 |
| Private consumption | -1.14 | 1.31 | -0.11 | -3.30 | -2.25 | -3.13 | -3.60 | -1.09 | -2.94 |

| ÷ | |
|--------|---|
| F tes | |
| AD | |
| or the | |
| es fc | |
| valu | - |
| itical | |
| n cr | |
| kinno | |
| Mac | |
| 5.1b | |
| ble 5 | |
| Ta | |

| No tren | d & no inter | cept term | With tre | end & interce | pt term | With | intercept ald | one |
|---------|--------------|-----------|----------|---------------|---------|-------|---------------|-------|
| 1% | 5% | 10% | 1% | 5% | 10% | 1% | 5% | 10% |
| -2.58 | -1.94 | -1.62 | 4.04 | -3.45 | -3.15 | -3.49 | -2.89 | -2.58 |

Table 5.1a A unit root analysis of the fiscal response model: an ADF test

| Equation # | Engel- Granger | Mackinnon critical value | Johansen likelihood | Mackin critical v | non /alue |
|--------------------|---------------------|-----------------------------|------------------------|----------------------|--------------|
| | ADF statis- tics | at 5% | ratio | 5% | 1% |
| ESA | | | | | |
| Direct taxes | -1.082 | -3.80 | 33.60 | 29.68 | 35.65 |
| With bilateral | -1.32 | -3.84 | 26.70 ++ | 24.31 | 29.75 |
| Indirect taxes | -4.34 | -4.19 | 81.40 + | 53.12 | 60.16 |
| Gov't expenditure | -1.82 | -4.22 | 57.42 | 47.21 | 54.46 |
| With grants | | | 75.40 + | 53.12 | 60.16 |
| NA | | | | | |
| Direct taxes | -3.86 | -3.63 ** | 32.83 | 29.68 | 35.65 |
| With multilateral | -3.90 | -3.63 ** | 28.96 | 29.68 | 35.65 |
| Indirect taxes | -3.85 | -3.64 ** | 40.80 | 29.68 | 35.65 |
| With private flows | -3.74 | -3.66 ** | 60.06 | 29.68 | 35.65 |
| Gov't expenditure | | | | | |
| With multilateral | -3.90 | -4.06 | 24.51 ++ | 24.31 | 29.75 |
| WCA | | | | | |
| Direct taxes | -2.75 | -3.54 | 43.60 | 29.68 | 35.65 |
| With bilateral | -2.53 | -3.59 ** | 51.50 ++ (+ also) | 24.31 | 29.75 |
| Indirect taxes | -2.39 | -3.51 ** | 49.38 | 29.68 | 35.65 |
| With private flows | -2.21 | -3.58 ** | 34.93 | 29.68 | 35.65 |
| With grants | -3.34 | -3.49 ** | 35.86 | 29.68 | 35.65 |
| Gov't expenditure | -2.04 | -3.95 ** | 78.00 + | 53.12 | 60.16 |

Table 5.1c Co-integration test for fiscal response model

** [*] are Mackinnon critical value at 10% [1%] level of significance (others at 5%).

The Johansen test used assumes linear deterministic trend in the data with intercept, no trend, in the test VAR (i.e., the co-integration equation (CE), as an indicator of long-run equilibrium relation, has no trend).

+ assumes intercept but no trend in CE and no intercept in test VAR.

++ assumes no deterministic trend in the data with no intercept or trend in the CE or test VAR.

+++ assumes no deterministic trend in the data with intercept, no trend, in CE and no intercept in test VAR.

The test is with aggregate flows (the sum of bilateral, multilateral, private and grants unless specified otherwise).

Results

Most econometric studies of the fiscal response literature fail either to report satisfactory diagnostic tests for their models or to test for the stationarity assumption. It is implicitly assumed either that the series are stationary or that they are co-integrated, without any formal investigation actually being undertaken. This casts doubt over the empirical results arising from studies based on such fiscal response models. In what follows, estimated results that correct for these deficiencies are reported. Tables 5.2–5.5 report results obtained from pooled time series (1970–90) and cross-country data for three regions of this study.

Prior to estimation, each series is tested for stationarity using the ADF unit root test. This is followed by a co-integration test between the hypothesized relationship using both the Engel-Granger (1987) and the Johansen (1988) methodologies. Only models that passed these tests are estimated and reported upon. The results of these tests suggest that a long-run relationship exists within our models (see Tables 5.1a-c). I have used Hendery's General to Specific (GS) methodology to arrive at parsimonious functional forms. Various diagnostic test results are also reported upon. Finally, since all the variables are again in logarithmic form, the coefficients obtained may also be interpreted as elasticities.

(i) Direct taxes

The estimation of direct taxes is pursued by first examining a second-order ECM equation. The second-order ECM for the ESA region is basically similar to its special case of first-order ECM, the only difference being that the short-run, one-period lagged value of GDP in second-order ECMs has a statistically significant elasticity coefficient of 0.47. Hence, the first-order ECM for the ESA region is reported in Table 5.2. Similarly, the second-order ECM for both the NA and WCA regions is nearly identical with the first-order ECM reported in Table 5.2. These results show that the restriction imposed in the second-order ECM is plausible.

As can be seen from Table 5.2, direct tax has a statistically significant positive elasticity with respect to GDP, especially in the long run. This elasticity is nearly identical in the ESA and NA regions, while relatively lower in WCA. The estimated equation is also characterized by the fact that adjustment to disequilibrium is very rapid in WCA, and successively slower in the NA and ESA regions. In general, the impact of total capital inflows is not found to be statistically different from zero. However, its short-run negative impact within the WCA region is significant at a 10 per cent level of significance, although with a low potency. Indeed, the long-run elasticity for the ESA region also suggests such a negative impact, at a 14 per cent level of significance. Notwithstanding its statistical insignificance, one notes a generally negative elasticity coefficient for direct taxes with respect to capital inflows. This clue leads one to investigate further the possible impact of different types of capital inflows.

| 6 |
|--------------------------------|
| × |
| .Ψ |
| ÷. |
| 77 |
| .2 |
| ÷. |
| ŝ |
| ΩÂ. |
| ~~ |
| ch l |
| - |
| Ch |
| ~ |
| ã. |
| ς. |
| 6 |
| 22 |
| ~ |
| 2 |
| Ψ. |
| 2 |
| .2 |
| ¥- |
| 1 |
| φ |
| 0 |
| ~ |
| 9 |
| - |
| S |
| ~ |
| <u> </u> |
| ō |
| ~ |
| 47 |
| 2 |
| |
| - |
| Ω, |
| <u>ت</u> ب |
| |
| - |
| ā |
| <u>o</u> e |
| api |
| capi |
| l capi |
| id capi |
| nd capi |
| and capi |
| and capi |
| s and capi |
| es and capi |
| es and capi |
| xes and cap |
| axes and cap |
| taxes and capi |
| taxes and cap |
| st taxes and cap |
| ict taxes and capi |
| ect taxes and cap |
| rect taxes and cap |
| lirect taxes and cap |
| Direct taxes and capi |
| 2 Direct taxes and cap |
| 2 Direct taxes and capi |
| 5.2 Direct taxes and capi |
| 5.2 Direct taxes and capi |
| e 5.2 Direct taxes and capi |
| le 5.2 Direct taxes and capi |
| ble 5.2 Direct taxes and capi |
| able 5.2 Direct taxes and capi |
| able 5.2 Direct taxes and capi |

| | Multc my ++ | /ery low (i) | | (II) Mon | | /ery low (iii) | |
|-----------|-----------------|--------------|------------|----------|------------|----------------|------------|
| ic check | Jarque- Bera | 1.7* / | | 7.3 | | 6.26 | |
| Diagnost | Reset | 3.6 | (0.06) | 0.48 | (0.49) | 91 | |
| | Chow | 1.7* | | 3.4* | | 1.3* | |
| R² | Level@ | 0.49 | n=76 | 0.50 | n=36 | 0.68 | n=40 |
| Adj. | ECM | 0.27 | | 0.38 | | 0.26 | |
| | Td_1 | -0.40 | 4 \$0 | -0.57 | -2.7* | -0.59 | 4.3* |
| -ong run | Ľ. | -0.21 | -1.5~ | -0.07 | -0.58 | -0.04 | -0.23 |
| - | σ | 0.77 | 9.0* | 0.37 | 1.6^ | 0.73 | 6.2* |
| _ | ΔF | -0.04 | -0.89 | -0.11 | -1.7^ | -0.00 | -0.12 |
| Short run | δΔ | 1.09 | 4.7* | 0.27 | 0.54 | 0.40 | 054 |
| •, | Con | 0 | | 1.05 | 0.81 | 0 | |
| | Region | ESA | (t-values) | WCA | (t-values) | NAI | (t-values) |

Notes:

Constant dummies used : (i) Uganda -1.1; Zambia 0.35 (ii) Cameroon 1.02; Gabon 1/37; Nigeria 2.1 and Burkina Faso 2.5 (iii) Algeria -2.15. ÷... N

* * Significant at 1% and better, + significant at 5% and ^ significant at 10%. ESA: East and Southern Africa, WCA: West and Central Africa, and NA: North Africa.

@ The long-run coefficients and their t-values are computed using the Wickens and Breusch (1988) and Gurney (1989) approach (see Appendix 4.1). The iong-run (level) R² is obtained from this same regression. _1 shows one period lag. All variables are as defined in equations 1 to 3; n in the column shows sample size. ė

++ For examining the degree of multicollinearity a simple correlation among regressors < 0.25 is considered Extremely Low (ELow), < 0.50 Very Low and < 0.60 Low (acceptable). (C) in this column indicates correction for multicolinearity of the general model by omitting the problem variable. 4

For the diagnostic tests, values in bracket are P-values.
 I For this region and in this table the F is the sum of bila

I For this region and in this table the F is the sum of bilateral and multilateral capital inflows only.

Similar to our discussion in Chapter 4, the ECM form of the models in Table 5.2 above may be given as follows (illustrated with ESA model):

$$\Delta T_d = -1.09 \Delta Q - 0.04 \Delta F - 40 \left(T_{d-1} - 0.77 Q_{-1} + 0.21 F_{-1} \right)$$

where the long-run equilibrium relation is given by $T_d = 0.77Q - 0.21F$.

Estimation of variation of elasticities across different financial instruments is undertaken by starting with a second-order ECM and working out towards a first-order ECM model, in search of parsimonious functional forms along the G-S approach. Only the bilateral inflowbased estimation for the ESA region is found to be important. This estimation is given below.

ESA with bilateral flows

$$\Delta Td = -0.64 \Delta Td_{t-1} + 0.89 \Delta y_t + 1.37 \Delta y_{t-1} - 0.10 \Delta F_t - (-4.7)^* (2.4)^* (3.2)^* (-1.8)^{-1}$$

$$0.13 \Delta F_{t-1} - 0.59 (Td_{t-2} - 0.77y_{t-2} + 0.3F_{t-2}) - (-1.94) + (-2.8)^* (14.0)^* (-3.3)^* - (-3.4)^* (2.7)^* [5.2.1]$$

$$R^2 - adj = 0.39 \qquad \text{Jarque-Bera 4.8}(0.79)$$

Reset 2.16(0.15) $\qquad \text{IM 2.04}(0.16)$

Reset 2.16(0.15) LM 2.04(0.16) R^{2} -level = 0.55 BG = 0.32(0.72)

Notes:

Values in bracket are t-values, while for diagnostic test P-values are given (see Appendix 4.1).

The sign difference between the t-values and the coefficients in the ECM term may be explained by the fact that the t-values refer to the long-run coefficient.

* implies 1%. + 5% and ^10% level of significance.

LM is a hetroscedasticity test while BG is the Breuch-Godfery test for serial correlation.

The 5% (2 degrees of freedom) value for Jarque-Bera is 5.99.

| are elasticities | |
|------------------|--|
| (coefficients | |
| instruments | |
| ss financial | |
| Variation acros | |
| Table 5.3 | |

| hort run Long | Long | Long | ^D | n | | Adj | . R | | Diagnos | stic checl | × | - End |
|---------------------------|---------|------------|--------------|----------|-------|------|------------|------|---------|-----------------|-------------|----------|
| Δ Q ΔF+ Q_1 | °-1 | 5 | | <u>1</u> | Td_1 | ECM | Level @ | Chow | Reset | Jarque- Bera | Multc ++ | м |
| 0.94 -0.13 0.75 | 3 0.75 | .75 | | -0.25 | -0.52 | 0.30 | 09.0 | 2.8* | 0.58 | 0.37* | Very low | Θ |
| 3.3* -2.7* 15* | * 15* | <u>ئ</u> | | -2.6* | 4.4* | | n=67 | | (0.45) | | | |
| 1.4 -0.01 1.3 | 1 1.3 | ų. | | -0.22 | -0.38 | 0.35 | 0.76 | 3.6* | 0.49 | 3.8* | Very low | • |
| 3.0* -0.17 10.0* | 7 10.0* | •0.0 | | -1.8^ | 4.4* | | n=39 | | (0.49) | | | |
| 0.07 -0.15~ 0.75 | 5~ 0.75 | .75 | | -0.08 | -0.60 | 0.34 | 0.68 | 3.4* | 4.1 | 36.0* | Very low | |
| 0.09 -1.5 8.0* | 8.0* | * 0 | | -0.15 | 4.4 | | n=39 | | (0.05) | | | |

Constant dummies used: (i) Uganda -1.4; Zambia 0.62; (ii) Burkina Faso =1.01; (iii) Algeria -2.15 Notes are as given under Table 5.2. + The value of the F variable is bilateral flows for ESA and WCA, and multilateral flows for NA. Notes: 1. Cor 3. + T

As can be seen from this estimation the diagnostic checks are quite acceptable with almost all values of coefficients being statistically significant. With a long-run elasticity of -0.30, these results clearly demonstrate that foreign inflows will have a negative impact in both the short and long run. For the other regions the second-order ECM does not differ from its first-order variant. Hence, only the latter is reported in Table 5.3.

Table 5.3 reports on the statistically significant results obtained by disaggregating capital inflows into the categories 'Bilateral', 'Multilateral', 'Grants' and 'Private inflows'. An exhaustive search process is undertaken before arriving at Table 5.3. For the ESA region estimation results are found to have statistically insignificant coefficients with all other capital inflow categories, with the exception of bilateral flows. A comparison with Table 5.2 indicates that, compared to aggregate flows, using bilateral flows within the estimation has a strong potency in the short run. For the WCA region both grants and multilateral flows are non-stationary series. Both failed a co-integration (Johansen 1988) test with GDP and direct taxes. However, the bilateral flow is found to be co-integrated with GDP and direct tax and also has a strong long-run elasticity, which was missing in the aggregate estimation for Table 5.2. The estimation for the NA region is less impressive in terms of its diagnostic test. However, the results of this test do suggest a short-run negative elasticity coefficient of direct tax with respect to multilateral flows. A model which included the other types of capital flows failed a co-integration test, and so could not be estimated.

2. Indirect taxes

Estimation of the indirect taxes equations is undertaken starting from a general specification of a second-order ECM. This general specification is similar to its first-order formulation for the ESA region. For the NA and WCA regions the second-order ECM performed best, and hence the results from this are reported below. The estimation of the second-order ECM for the WCA region, using the total inflow, bilateral and multilateral inflows, respectively, yields similar results to that obtained from the first-order ECM and, hence, only the first-order result is reported in Table 5.3. However, the parsimonious second-order ECM estimation using private inflows and grants, respectively, gave the best of the functional forms and so is reported below. WCA with private flows

 $\Delta T_{i} = 0.28 \Delta T_{i_{t-1}} + 0.60 \Delta C p_{t} + 0.27 \Delta C p_{t-1} - (-2.2) + (4.4)^{*} (2.0) + 0.27 (T_{i_{t-2}} - 1.03 C p_{t-2} + 0.12 F_{t-2} + 2.41) + 0.49 D G a b_{(-3.1)^{*}} (9.7)^{*} (-1.94)^{\wedge} (-3.4)^{*} (4.3)^{*} [5.2.2]$ $R^{2}\text{-adj} = 0.37 \qquad \text{Jarque-Bera } 1.4(0.50)$ Reset 0.90(0.35) LM 0.08(0.78) $R^{2}\text{-level} = 0.67 \qquad \text{BG} = 0.91(0.41)$ All variables are in natural logarithms. Notes as given in eq. [5.2.1].

WCA with grants

 $\Delta T_i = -0.08 \Delta T_{i_{t-1}} + 0.68 \Delta C p_t + 0.09 \Delta C p_{t-1} + 0.01 \Delta F_t + 0.07 \Delta F_{t-1} - 0.01 \Delta F_t + 0.07 \Delta$ (-0.82)(8.2)* (0.82)(0.13) (1.42) $0.24 (Ti_{t-2} - 0.95Cp_{t-2} + 0.13F_{t-2} + 1.71) + 0.249DGab$ (-3.7)* (11.6)* (-1.42) (-3.3)* (3.1)* [5.2.3] R^2 -adj = 0.43 Jarque-Bera 2.5(0.50) Jarque-Bera 2. LM 9.4(0.00) R^{-} adj = 0.43 Reset 0.00(0.99) R^{2} level = 0.64 BG = 3.2(0.04) R^{2} -level = 0.64 All variables are in natural logarithms. Notes as given in eq. [5.2.1].

These two equations show that both private inflows and grants have no short-run impact. However, both variables are found to have a statistically significant negative impact in the long term, this being greater in relation to private inflows than grants.

For the NA region, the second-order ECM is found to be the best specification. Two equations, identified in an exhaustive searching process, were found to be acceptable on statistical grounds and so are reported on below. The first of these uses total capital inflows, while the second uses private inflows to North Africa. Although both equations show strong negative elasticities with respect to capital inflows, the impact of private inflows is found to be important in the short run and to have a stronger impact in the long run. Given the relative importance
of private inflows in North African countries this result is entirely plausible.

NA with total inflows

$$\Delta T_{i} = -0.36\Delta T_{i}_{t-1} + 1.1\Delta XM_{t} + 0.61\Delta XM_{t-1} - 0.03\Delta F_{t} - (-1.3) (8.2)* (1.1) (-0.50) 0.07\Delta F_{t-1} - 0.80 (T_{i}_{t-2} - 0.93XM_{t-2} + 0.24F_{t-2}) (-1.03) (-3.7)* (26.4)* (-7.2) [5.2.4] R^{2}-adj = 0.27 Jarque-Bera 29(0.0) Reset 0.08(0.76) LM 2.4(0.13) R^{2}-level = 0.90 BG = 1.04(0.37) All variables are in natural logarithms.$$

NA with private inflows

$$\Delta T_{i} = -0.27 \Delta T_{i} t_{t-1} + 1.2 \Delta X M_{t} + 0.52 \Delta X M_{t-1} - 0.13 \Delta F_{t} - (-0.94) \quad (3.6)^{*} \quad (1.2) \quad (-3.0)^{*} \\ 0.14 \Delta F_{t-1} - 0.75 \ (T_{i} t_{t-2} - 0.94 X M_{t-2} + 0.28 F_{t-2}) \\ (-3.0)^{*} \quad (-3.7)^{*} \quad (23.6)^{*} \quad (-6.8)^{*} \qquad [5.2.5] \\ R^{2} - adj = 0.42 \qquad Jarque-Bera 5.8(0.05) \\ Reset 1.2(0.30) \qquad LM 2.4(0.14) \\ R^{2} - level = 0.88 \qquad BG = 0.89(0.42) \qquad n = 30 \\ All variables are in logarithms. Notes as given in eq. [5.2.1].$$

The estimation for North Africa clearly demonstrates that the impact of capital inflows on the level of taxes varies across financial instruments. Within the NA region, both total and private inflows have a strong negative impact on indirect taxes in the long run. A percentage change in inflows to the region leads to 0.30 per cent decline in the level of indirect taxes in the long run. The remainder of the first-order ECM estimations for the ESA and WCA regions, using total inflows are reported in Table 5.4.

| Desien | | Sho | rt run | | | Long | g run | |
|-------------------|----------------|--------------|--------------|----------------|------------------|----------------|----------------|----------------|
| Region | Con | ∆Cp | ΔXM | ΔF | C _{p_1} | XM_1 | F_1 | TĽ1 |
| ESA (t-values) | -0.85 -3.0* | 0.46 6.3* | 0.42 5.4* | -0.01 -0.75 | 0.63 5.2* | 0.73* 3.70* | -0.03 -0.63 | -0.21 -3.40 |
| WCA (t-values) | 0 | 0.64 8.2* | | 0.02 1.6^ | 0.64 27.0* | | 0.05 1 | -0.31 _ |

Table 5.4 Indirect taxes and capital inflow (coefficients are elasticities)

| | Ad | j. R² | | Diagnos | stic check | | Dum- |
|-------------------|------|--------------|-------|----------------|-----------------|---------|------|
| Region | ECM | Level@ | Chow | Reset | Jarque- Bera | Multc++ | my |
| ESA (t-values) | 0.67 | 0.84 n=87 | 1.7* | 0.19 (0.67) | 1.3* | Low | (i) |
| WCA (t-values) | 0.55 | 0.72 n=87 | 1.14* | 0.02 (0.90) | 2.07* | Low | (ii) |

Notes: See Table 5.2

Constant dummies used:

(i) Kenya 0.14; Tanzania 0.24

(ii) Burkina Faso -0.14; Gabon 0.28; Senegal 0.19

Table 5.4 shows that indirect taxes are explained by the level of private consumption and external trade (comprising the sum of imports and exports) both in the short and long run. Within the WCA region, nearly half of past errors are made up for in the current period. The corresponding value for the NA region is much higher, as can be seen from the above estimations, while the value for the ESA region stands at less than half of that. This finding tallies with the stylized fact that the NA region has a relatively stronger and more flexible economy than either the WCA or ESA regions. However, in contrast to what one might expect, based on the prevailing theoretical orthodoxy, the elasticity coefficient of capital inflows does not differ statistically from zero. although it does show a negative sign, especially in the long run. This last result is further investigated by exploring the impact of different types of capital inflows on indirect taxes. For the ESA region the indirect tax equation (eq. 2) is estimated by separately using the financial categories for bilateral, multilateral and private flows as well as grants. The diagnostic test results for the four equations are all quite acceptable, and, in all cases, the elasticity coefficient with respect to each of these flows is not found to be statistically different from zero. For the NA and WCA regions, the impact of private inflows is found to be statistically significant, as shown by the second-order ECMs set out above.

In sum, the major lesson which may be drawn from this analysis is that an examination of the impact of capital inflows on taxes requires the disaggregation of these taxes into different components. Indeed, it would be tempting to conclude that if the foreign capital obtained is then spent on imports, this might have a positive impact on indirect taxes, which are largely levied on the external sector in most African countries. This, in turn, may offset the possible negative impact of these tax collection efforts on the rest of the economy.

3. Government current expenditure

The estimation for the ESA region using a second-order ECM yields identical results to that of its first-order variant for all financial instruments, with the exception of grants. Thus, this parsimonious functional form is chosen and reported on in Table 5.5. The second-order ECM, using grants, also highlights a significant positive impact of these grants in the short run. This is given below.

| \sim |
|----------|
| Š, |
| 3 |
| <u>.</u> |
| St |
| ġ. |
| Ð |
| Ф |
| a |
| S |
| Ē |
| e. |
| <u>g</u> |
| ٩f |
| õ |
| છ |
| 2 |
| ō |
| Ę |
| |
| ā |
| ž |
| ğ |
| S |
| g |
| a |
| D) |
| S |
| Ë |
| ğ |
| e |
| 8 |
| G |
| * |
| 9 |
| E |
| 2 |
| τ |
| U. |
| Ĕ |
| ä |
| 5 |
| ž |
| 8 |
| Ċ |
| |
| 5 |
| ŝ |
| Ð |
| 2 |
| 100 |

| Doctor | | Sh | ort run | | | - | | ong run | | |
|-------------------|------|--------------|--------------|------|--------------|-----------------|--------------|---------|-------|---------------|
| lioifiau | Con | ΔT | Δiwn | Δiw | ΔF | Ŀ, | Ìwr1 | iw_1 | Ľ. | 5 |
| ESA (t-values) | 0 | | 0.15 2.2* | 0.03 | 0.01 | | 0.34 1 0+ | 0.07 | 0.21 | -0.19 2.0* |
| | 1 23 | | 1.1 | 10.0 | 20.0 | | | 0.0 | | |
| (private flows) | 3.0* | | 1.07 | | 5.5* 5.6* | | 1.15 | | 1.64^ | -2.7* |
| | Ad | l. R² | | | Diagnos | stic check | | | | |
| | - | | | | • | | | | | |
| Region | ECM | Level@ | Ŭ | N N | Reset | Jarque- Bera | Multc++ | | л м | |
| ESA (t-values) | 0.30 | 0.58 n=78 | 2.8 | *. | 0.05 | 0.38* | VLow (c) | 5 | | |
| WCA | 0.27 | 0.50 | 0.5 | * | 0.27 | 0.47* | VLow | | ~ | |
| (private flows) | | n=62 | | Ŭ | (09.0) | | | | | |

Notes: See Table 5.2

Constant dummies used: (i) Botswana -0.43, Kenya -0.06, Madagascar -0.2 (ii) Benin -0.54; Cameroon -0.26; Central African Rep. -0.46; Ghana -0.59; Senegal -0.30; Zaire -0.20.

ESA with grants

 $\Delta G_{I} = -0.18 \Delta G_{t-1} + 0.13 \Delta i_{w^{*}t} + 0.01 \Delta i_{w^{*}t-1} + 0.01 \Delta i_{wt} + (-1.33) \quad (1.8)^{\wedge} \quad (0.17) \quad (0.37)$ $0.05 \Delta i_{wt-1} + 0.06 \Delta F_{t} + 0.06 \Delta F_{t-1} \quad (1.8) + (1.7)^{\wedge} \quad (1.6) \sim -0.14 \quad (G_{t-2} + 0.03i_{w^{*}t-2} + 0.03i_{w^{t-2}} - 0.57F_{t-2}) \quad (-2.18) + (-015) \quad (-0.23) \quad (3.3)^{*} \qquad [5.2.6]$ $R^{2}\text{-adj} = 0.23 \qquad \text{Jarque-Bera} = 1.9(0.38)$ $Reset = 2.3(0.13) \qquad LM = 8(0.01)$ $R^{2}\text{-level} = 0.40 \qquad BG = 1.4(0.26) \qquad n = 79$ All variables are in logarithms. Notes as given in equation [5.2.1].

This result shows that although the short-run impact of grants is extremely weak, nevertheless its long-run elasticity remains very strong (at around 0.60). Similarly, the second-order ECM for the WCA region is also found to be the best functional form, and so is reported on below. The results also indicate that capital inflows within the region have a strong positive impact in both the short and long run.

WCA with total inflows

$$\Delta G_{I} = -0.40 \Delta G_{t-1} - 0.14 \Delta i_{w^{*t}} - 0.24 \Delta i_{w^{*t-1}} + 0.06 \Delta i_{wt} + (-3.0)^{*} (-1.3) (-2.7)^{*} (1.2) 0.14 \Delta i_{wt-1} + 0.10 \Delta F_{t} + 0.21 \Delta F_{t-1} (2.0)^{+} (2.2)^{*} (3.7)^{*} -0.39 (G_{t-2} + 0.004 i_{w^{*t-2}} + 0.15 i_{wt-2} - 0.36 F_{t-2}) (-5.4)^{*} (-0.02) (-2.7)^{*} (2.3)^{*} [5.2.7] R^{2}-adj = 0.43 Jarque-Bera = 6.5(0.04) Reset 1.1(0.29) LM = 0.05(0.81) R^{2}-level = 0.77 BG = 0.4(0.70) n = 59 At the interval of the other interval of interval of the o$$

All variables are in logarithms. Notes as given in equation [5.2.1].

The variables in the government expenditure function for North Africa are not co-integrated with any of the types of financial instruments, with the exception of multilateral flows. The second-order ECM using multilateral flows is found to be the best functional form and is given below.

NA with multilateral inflows

$$\Delta G_{I} = -0.81 \Delta G_{t-1} - 0.16 \Delta i_{w^{*}t} - 0.24 \Delta i_{w^{*}t-1} + 0.39 \Delta T_{t} + (-5.9)^{*} (-1.7)^{\wedge} (-2.0) + (3.8)^{*}$$

$$0.55 \Delta T_{t-1} + 0.08 \Delta F_{t} + 0.18 \Delta F_{t-1} - (4.2)^{*} (2.3)^{*} (4.3)^{*}$$

$$0.65 (G_{t-2} + 0.52 i_{w^{*}t-2} - +1.4 T_{t-2} + 0.23 F_{t-2}) - (-5.8)^{*} (-3.5)^{*} (15.0)^{*} (2.4)^{*}$$

$$0.21 DAlg - 0.26 Tun - (-2.1)^{+} (-2.5)^{*} [5.2.8]$$

 $\begin{array}{ll} R^2\text{-adj} = 0.75 & \text{Jarque-Bera} = 1.2(0.56) \\ \text{Reset} = 5.4(0.03) & \text{LM} = 0.37(0.55) \\ \text{R}^2\text{-level} = 0.95 & \text{BG} = 0.8(0.45) & \text{n} = 34 \\ \text{All variables are in logarithms. } DAlg \text{ and } DTun \text{ are dummies for Algeria} \\ \text{and Tunisia, respectively. Notes as given in equation [5.2.1].} \end{array}$

Contrary to what one might expect, the equation for the NA region has a statistically significant negative elasticity with respect to concessional interest rates. An observation of the plot of government current expenditure and concessional interest payments shows an upward trend for both series. An alternative estimation that excludes T (total revenue) from this functional form¹⁸ results in a similar coefficient for all, with the major difference that the concessional interest rate payment coefficient is not statistically different from zero, although still showing a negative coefficient. This, combined with a low RESET value, suggests that caution should be exercised in using this equation. Non-concessional interest payments are excluded from this equation, owing to lack of sufficient data points. Table 5.5 reports estimation results for equation [3], using the firstorder ECM. It shows that – with the exception of the WCA region, using the second-order ECM – aggregate capital inflows have no effect on government current expenditure in the short run. In the long run, however, such inflows are found to have a statistically significant positive impact in all regions. Within the ESA region, interest payments on concessional loans, both bilateral and multilateral, are found to have a significant and positive short- and long-run impact. However, this is not found to be statistically significant in the WCA region, especially in the long run. The adjustment coefficient is, in general, low, indicating the difficulty of adjusting to past disequilibrum on government current expenditure. However, there exists an important variation across regions, with the highest adjustment coefficient found for the NA region, followed by the WCA and ESA regions.

Following a similar procedure as for tax equations, various estimations for equation [3] based on different types of capital inflows are undertaken. The estimation for the ESA region, using bilateral, multilateral and private capital inflows, respectively, produces coefficients which are not found to be statistically significant. However, the estimation using grants exhibited statistically significant values and, hence, its second-order ECM estimation is given above. For the WCA region, on the other hand, the model based on grants could not pass the relevant co-integration test. The estimation using bilateral flows produces coefficients, which are not found to be statistically different from zero. However, the coefficient for private capital inflows, shows a statistically significant elasticity of 0.07 in the short run, and 0.20 in the long run. Indeed, the latter is lower than the aggregate elasticity obtained from the second-order ECM reported above. In relation to the NA region, the estimation using bilateral flows could not pass a cointegration test, while the estimation with grants, in addition to having poor Jarque-Berra statistics and suffering from high kurtosis, does not indicate a statistically significant elasticity. Private capital flows also indicate a statistically insignificant coefficient. However, the estimation using multilateral flows and a second-order ECM is found to be the best and so is reported on above.

5.2.4 Summing Up

In this section an attempt to quantify the fiscal response of African economies to external finance is made. The method adopted differs

from the existing literature at two levels. Firstly, and for the most part, it departs from the assumption that governments of developing countries will seek to maximize utility. This assumption stands as the cornerstone of most studies in this topic. However, in this book, we depart from this assumption in search of realism. Secondly, by using the latest developments in time series econometrics, the analysis presented in this section overcomes the other major shortcomings of the previous studies: namely, spurious regression and poor econometrics. The fact that almost none of the series in the sample are found to be stationary certainly raises questions about the validity of previous empirical studies.

By proposing an alternative decision-making framework based on the stylized facts in relation to Africa, several sets of equations are estimated, using an Error Correction Model. The results of the analysis are mixed. However, the following major points may be generalised from this section. Firstly, the impact of capital inflows on taxes varies across the type of inflows, the nature of taxes and between regions. In general, bilateral flows are found to have a negative impact on direct taxes. The long-run elasticity for direct taxes with respect to bilateral inflows ranges from -0.22, for the WCA region to -0.30 for the ESA region. On the whole, indirect taxes are found to have a statistically significant relationship with private inflows, although grants are also found to be important within the WCA regions. As a result, the longrun elasticity of indirect taxes with respect to private inflows varies from -0.12 for the WCA region, to -0.28 for the NA region. Secondly, capital inflows, in general, have a strong positive impact on current government expenditure in all regions. Such a positive impact is accentuated with the use of grants in the ESA region reaching an elasticity coefficient of 0.57. Private capital inflows within the WCA region also show a positive but less strong elasticity coefficient of 0.23.

A number of important tentative conclusions may be drawn from the above analysis. Primarily, the analysis presented in this section clearly demonstrates the importance of investigating each financial instrument for each of the different regions. This analysis also has important implications in relation to liberalization policies, such as structural adjustment programmes (SAPs), currently being implemented in Africa. Capital inflows to Africa generally came being conditional upon deficit reduction. However, these inflows have an inherent tendency to aggravate the deficit and so may result in governments drifting away from sustainable self-financing behaviour. A further important implication of the analysis relates to the impact of capital inflows on the distribution of income. By ensuring a reduction in direct taxes (as opposed to indirect taxes which usually are levied on ordinary consumer items), bilateral flows could influence the distribution of income in favour of those big firms, the commercial sector and the like, for which income taxes are important. Conversely, the opposite effect may arise from private inflows. Since the former types of inflow comprise the bulk of capital inflows to Africa, their effect is likely to be relatively the more important.

In sum, capital inflows to Africa may be thought of as beneficial, not only if such flows are worth their financial costs, but also as a consequence of their long-run effect on deficit financing and the distribution of income. This requires the computation of similar elasticities at each country level. And, perhaps more importantly, this also suggests that macroeconomic modelling in Africa should explicitly take this issue into account. Hence, the prototype model of Africa developed in Chapter 6 will explicitly incorporate these issues.

5.3 AID AND THE 'DUTCH DISEASE' IN AFRICA

5.3.1 Aid and the 'Dutch Disease': An Overview

From mid-1960s onward, the 'two-gap' model and its recent variant the 'three-gap' model, have emerged as the dominant analytical framework for the analysis of capital inflows to developing countries. The model helps to identify the capital requirement (labeled domestic and external 'gaps') necessary to achieve a target rate of growth designed à la Harrod and Domar. The major criticism of the two-gap model which is relevant in the context of the present study is its extreme structuralism, and, hence, the neglect of the impact of relative prices (Findlay 1973). It is this criticism which has led to emphasis being placed on the 'Dutch disease'¹⁹ literature. Historically, the 'Dutch disease' literature originated from a discussion of the problems associated with managing revenue from a booming sector of the economy.²⁰ Popularized, among others, by the works of Cordon and van Wijnbergen, the 'Dutch disease' concept has come to play an increasingly important role in the discussion of the macroeconomic impact of temporary resource discovery, in general, and foreign aid, in particular. This section will extend the discussion of this same issue presented in Chapter 3, in order to establish the empirical basis for the explicit incorporation of the 'Dutch disease' effect within the prototype African model developed in Chapter 6.

The argument runs as follows. Revenue is obtained from a booming sector. If part of this is spent on non-traded goods (the 'spending effect'), this leads to a real appreciation in the relative price of non-tradables relative to tradables. This, in turn, draws resources (the 'resource movement effect') out of the booming sector into the non-traded sector (Corden 1984, van Wijnbergen 1984). The growth effect of such a resource shift is considered to be negative, since traded sectors are characterized by 'learning by doing' (i.e., dynamic) externalities, which will have a higher and positive effect on growth (van Wijnbergen 1984 & 1986a, Edwards & van Wijnbergen 1989). It should be noted, however, that the exact form which this effect takes could also depend on the flexibility of prices within the factor market. Corden notes that if the effects of the boom

have raised the real wage in flexible-factor price model, then with a rigid real wage it would reduce unemployment instead, while if it would have reduced the real wage in the flexi-price model, it would generate unemployment in the fix-price case. (Corden 1984: 369)

This analysis has its origins in the 'dependent economy model' of Salter (1959) and Swan (1960). In both models, small economies are assumed to be price takers within international markets, and hence their terms of trade is taken as given. However, changes in the external economy (such as might be associated with a rise in overseas prices, or excess demand) may disturb the pre-existing equilibrium. This could result in a switching of demand from the traded to the non-traded sector. This, in turn, could entail a rise in the price of domestic, or non-traded goods and, hence, a subsequent supply reaction. Nonetheless, the final result will depend on a number of factors, including the relative speed and magnitude of different effects (Swan 1960: 55-62). Within Salter's formulation, the reaction to such changes, of, say, higher foreign prices or excess demand, will depend on two main factors. Firstly, on their impact on domestic price; and secondly, on the elasticity of substitution between the traded and the non-traded sector from the supply side, and the elasticity of demand substitution from the demand side (Salter 1959: 230). Focusing on these issues the 'dependent economy' model helps to understand such sectoral disequilibrum.

Edwards and van Wijnbergen (1989) argue for similarities between the discovery of natural resources and aid inflows and, hence, for similarities in the macroeconomic impacts of both. This similarity takes a number of specific forms. Firstly, both increase foreign exchange availability, with little or no additional use of domestic factors of production. Secondly, the impact of both is almost certainly temporary in nature. And finally, both come in the form of additional foreign exchange but will, at least partially, be spent on non-traded goods, thus placing upward pressure on the real exchange rate (Edwards & van Wijnbergen 1989: 1485). The policy conclusion of this observation is that subsidies to the traded sectors are essential since a 'Dutch disease' type of problem is likely to occur (van Wijnbergen 1986b: 130). Although their characterization of aid and the discovery of natural resources as temporary could be questioned, the other two observations made by Edwards and van Wijnbergen are important and justify the use of the 'Dutch disease' approach for analysing the impact of aid.

In order to link the relative price effect of capital inflows with the two-gap model, van Wijnbergen (1986b) derived his analysis of the 'Dutch disease' effect from a reinterpretation of the two-gap model. He emphasized that an ex ante wedge between the saving and trade gaps implies ex ante a home goods market disequilibrum. A binding trade gap corresponds to excess supply of home goods, and therefore, Keynesian unemployment; and a binding savings gap to excess demand for home goods, and therefore, classical unemployment.²¹ Using this basic structure. Wiinbergen studied the differential impact of aid which comes in the form of traded goods, or assets easily convertible into such goods, but which may also partly be spent on non-traded goods. This could have an effect on the home goods market and hence, on the real exchange rate. He concludes that substantial, and especially shortterm, aid flows in countries which have less access to foreign capital will place upward pressure on the real exchange rate (van Wijnbergen 1986b).

5.3.2 Aid, the 'Dutch Disease' and the African Context

A number of country-specific studies have been undertaken with the aim of assessing the possibility that a 'Dutch disease' effect may be present in various African countries. Younger (1992) presents an account for Ghana and shows the existence of 'Dutch disease' effect in relation to that country. In Younger's study aid not only leads to an appreciation in the real exchange rate, through inflation, but also to financial crowding out of the private sector through governments sterilization of inflows in the pursuit of a tight credit policy, which usually is imposed by international lenders. Benjamin et al (1989) constructed a CGE model for Cameroon with the objective of studying the impact of the oil boom and its 'Dutch disease' effect. Their results point to the important differential impact of a boom in the importable and exportable goods sector, owing to imperfect substitution between these. The impact in the latter, which is generally agricultural, being negative while that in the former, which is mainly manufacturing, could in fact be positive.²² Kavizzi-Mugerwa (1990) formulated a model of Zambia. which aims to examine the impact of copper prices within that country. The model's underlying theoretical formulation is the 'Dutch disease' literature and the author maintains that the 'Dutch disease' model is largely validated in the case of Zambia. A similar study, which analyses the Egyptian experience of an increase in oil exports within a 'Dutch disease' model framework, is reported in de Macedo (1982). In contrast, in examining the effects of the coffee boom in Tanzania, Musonda and Luvanda (1991) fail to accept the validity of the 'Dutch disease' model for Tanzania. Rather, they emphasize the importance of examining the pattern of the distribution of gains from the booming sector in determining its overall outcome. Love (1994) has shown the 'Dutch disease' effect of diamond production in the agricultural sector (excluding livestock)²³ of the Botswana economy.²⁴ Given this evidence of the existence of the 'Dutch disease' effect highlighted in individual studies. I will now examine the evidence using cross-section and time series data relating to Africa. However, before proceeding to the estimation. I will first examine how the theoretical connection to the A frican case is made.

Van Wijnbergen (1986a) uses the 'Dutch disease' model to analyse one of the effects of Aid in Africa. He notes that, by partially being spent on non-traded goods, aid places upward pressure on the real exchange rate. This leads to the contraction of the traded sector and the expansion of the non-traded one. This contraction – or 'de-industrialization', to use the original term – *and* real appreciation results from what is termed 'the spending effect' (Neary & Wijnbergen 1986: 15– 17). Within a typical 'Dutch disease' model, this spending effect not only raises the demand for 'specific factors', which are initially employed within the booming sector, but also, and perhaps more importantly, the demand for intersectorally mobile factors, such as labour. This requires either a rise in the wage rate or a fall in the relative price either a rise in the wage rate or a fall in the relative price of non-traded goods, the latter bringing about an increase in the level of unemployment. This is labelled the 'resource movement effect' and reinforces the spending effect (Neary & Wijnbergen 1986: 19).

According to Wijnbergen (1986a), the resource allocation consequences of the 'Dutch disease' in Africa is to shift labour from agricultural cash-crop production in rural areas to service employment, mostly in urban areas. The resulting increase in labour costs within the external sector effectively reduces its competitiveness. Applying this theory to data for a number of African countries, he concludes that 'increases in real volume of aid causes real appreciation.' However, White (1992) notes that Wijnbergen's estimations suffer from mis-specification (outperformed by a first-degree autoregressive process), multicollinearity, wrong t-values and auto correlation. In section 5.3.3, below, I will also correct for possible spurious regression problems, mis-specification and multicolinearity. Before that, however, it is worth questioning whether his theoretical description tallies with a typical African economy.

The contention in this section is that the theoretical reasons forwarded to describe the mechanism of casual links does not actually fit with the stylized facts for a typical African economy, even though the final conclusion arrived at could be similar.²⁵ Moreover, the approach followed contains a number of serious deficiencies, and hence, an alternative, empirically more robust, approach is required. Such an approach is discussed below. Firstly, I would argue that mobility of resources does not occur that easily, especially in the short run. Thus, in the short run, inflation arising from supply bottlenecks in the nontraded sector is likely to represent a more important influence than demand for labour, as hypothesized by van Wijnbergen. Thus, the spending effect is likely to be important in bringing about inflationary pressure. Secondly, in most urban areas in Africa there exists sufficient labour to meet the demand that may arise from the non-traded sectors, such as services. Thirdly, in the medium to long run, peasants may change their product mix. Thus, although the spending effect may result in a real appreciation in exchange rate, this will usually take place in the context of idle labour in urban areas and a sticky production structure in both the non-traded and traded sectors, especially for perennial crops and minerals. Hence, we cannot be sure how the market clears. However, within a dependent economy framework, these new

sets of rigidities could express themselves in the form of domestic inflation.

To sum up, the following major problems may be found with the existing literature. Firstly, as discussed above, the propagation mechanism through which the spending effect works may operate differently in Africa from other parts of the world. Secondly, while most studies are based on aggregated export data, disaggregation by commodity may actually result in differential intrasectoral variation. Thirdly, most time series studies could well be spurious since they do not formally test for the stationarity assumption. Indeed, the ECM model, which would have explicitly addressed the disequilibrum phenomena involved in this theory, is hardly employed in the literature. Thus, in order to rectify these shortcomings, the following simple, linear in parameter, model is formulated. This model, which explicitly emphasizes the spending effect, is formulated based on some stylized facts relating to Africa. 'Dutch disease' effects are examined empirically using data from various African economies.²⁶ The formulation deliberately emphasizes the spending effect and its inflationary consequences since this is more relevant within the African context. Thus, using equations [1] to [6] below, we will specify the hypothesized relationships.

The first important mechanism is that the foreign inflow (F), together with other factors (O), will positively affect the demand for nontradables (DD_{NT}) ,

$$DD_{NT} = f(F_A^a, O) \qquad \frac{\partial DD_{NT}}{\partial F_A^a} > 0 \qquad [1]$$

This demand for non-tradables can lead to the appreciation of the real exchange rate, defined as the ratio of foreign price (P_T) multiplied by the nominal exchange rate (e). (In other words the local currency value of the foreign to domestic price, P_{NT} .)

$$\frac{eP_T}{P_{NT}} = f(DD_{NT}), \quad f'(DD_{NT}) < 0$$
^[2]

where f' is the first derivative.

Combining equations [1] and [2] yields equation [3],

$$\frac{eP_T}{P_{NT}} = g(F_A^a, O^*), \quad \frac{\partial (ep_T/P_{NT})}{\partial F_A^a} < 0$$
[3]

Note that O^* within equation [3] also includes other determinants of the real exchange rate.

Although this is the general mechanism through which the impact of foreign inflows on the real exchange rate may be captured, nevertheless it does not represent an adequate specification for estimation, since it fails to clearly identify other determinants of the real exchange rate falling within O^* . Thus, notwithstanding the objective of this study, which is to isolate the impact of capital inflows on the real exchange rate, it is nevertheless necessary to locate this analysis within the broader framework under which the real exchange rate is determined. In order to accomplish this, we first identified what Edwards (1989) terms the equilibrium exchange rate 'fundamentals'. These variables are then used in specifying our empirical equation.

Three main determinants of the real exchange rate may be identified: firstly, the level of the equilibrium exchange rate (e^*) ; secondly, the deviation of the (index of) macroeconomic policy (Z) from its sustainable level (Z*); and thirdly, the nominal devaluation $(E_{t}-E_{t-1})$ (Edwards 1989: 133).²⁷ Using Edwards' formulation the dynamics of the real exchange rate behaviour is given by,

$$\Delta \log e_{t} = \theta \{ \log e_{t}^{*} - \log e_{t-1} \} - \lambda \{ Zt - Z_{t}^{*} \} + \varphi \{ \log E_{t} - \log E_{t-1} \}$$
 [4]

where: e is the actual exchange rate, Z an index of macro policy; , and are positive parameters designed to capture adjustment process.

In turn, the equilibrium real exchange rate is determined, *inter alia*, by external terms of trade (*ToT*), the level and composition of government consumption of non-tradables, technological progress, capital accumulation, and finally, capital, exchange and trade controls. Examining these in order, a deterioration in terms of trade brings about a depreciation due to the income effect on demand for non-tradables, which, in turn, leads to lowering of their prices. However, a substitution effect could work against this and, hence, what the net effect might be remains an empirical question. Secondly, the level and composition of government consumption of non-tradables will also depend on spending effects, which will raise the demand for non-tradables, as well

as income effects due to high taxes, which reduces demand for these. Thirdly, technological progress could have a negative impact on the relative prices of tradables to non-tradables through its effect on income, which raises the demand for non-tradables. However, this may be tempered, or even offset, by the factor augmentation impact of technology. Fourthly, the impact of capital accumulation will also depend upon within which sector it is carried. Thus, investment in the non-traded sector might lead to depreciation in the real exchange rate through its supply effect. And finally, capital, exchange and trade controls will also affect the equilibrium real exchange rate. Thus, by substituting these determinants of the equilibrium exchange rate into the real exchange rate equation set out above, the real exchange rate, in reduced form, may be specified as follows:

$$\log\left(\frac{eP_T}{P_{NT}}\right) = \gamma_1 + \gamma_2 \ y + \gamma_3 \ \log(G^c) + \gamma_4 \ Dev + \gamma_5 \ \log(ToT) + \gamma_6 \ \log(F_A^a)$$
[5]

where: is GDP growth assumed to show technological change; G^c government consumption expenditure assumed to show governments spending on the non-traded sector; *Dev* is nominal devaluation; *ToT* is terms of trade; *F* foreign inflow.

It should be noted that the constant term $_1$ captures, among other things, the impact of the variation of the macroeconomic policy from its sustainable level. With the exception of the coefficients for F and *Dev* (which are expected to have negative and positive coefficients, respectively), the sign for the other coefficients are indeterminate a priori. Hence, their sign will depend on the relative weight of substitution and income effects, as well as on income and factor augmentation impact, in the case of technology.

5.3.3 Estimation Results

5.3.3.1 Empirical approach

Estimation results for equation [5] in a reduced form, using crosssectional and time series data for 12 developing counters are reported in Edwards (1989). A modified version of Edwards' formulation is also used by van Wijnbergen (1986a)²⁸ for estimating the 'Dutch disease' effect of aid in six African countries. Both estimations are carried using a simple OLS and the results of these lend support to the hypothesis of the existence of a 'Dutch disease' problem. However, the estimations might well suffer from spurious regression, which is discussed below.

Tables 5.7 and 5.8 report estimation results based on a sample of countries from three regions of Africa. For each of these regions, one equation is specified for each aggregate commodity group. We have used four commodity groups, based on the modified UNCTAD price index constructed for each region. The results reported below differ from previous studies in at least four respects. Firstly, they are based on a consistent macroeconomic cross-section and time series data constructed for each country within each of the three regions. Secondly, they are commodity specific. Thirdly, the data is tested both for stationarity and for the existence of co-integration among the variables (see Tables 5.6a-5.6c). Depending on the results of these tests, an Error Correction Model (ECM) is fitted to the data. By starting from a second-order ECM, Hendery's General to Specific (GS) approach is followed in order to arrive at a parsimonious functional form. And finally, unlike in previous cross-country studies for Africa, a structural break (Chow) test is undertaken in order to test the validity of pooling the data across countries. The general first-order ECM equation estimated for each commodity group and region is given below. (See Appendix 4.1 for a specification of the econometric approach used.)

$$\Delta \log\left(\frac{eP_T}{P_{NT}}\right) = \alpha + \Delta \log\left(F_A^a\right)_{t-1} + \Delta \log(ToT) + \Delta \log(G^c) + \Delta Dev + \Delta y + \log\left(F_A^a\right)_{t-2} + \Delta \log(ToT)_{t-1} + \Delta \log(G^c)_{t-1} + Dev_{t-1} + y_{t-1} + \log\left(\frac{eP_T}{P_{NT}}\right)_{t-1}$$

$$\begin{bmatrix} 6 \end{bmatrix}$$

| | With interce |
|---|-----------------------------|
| 10del: an ADF test | With trend & infercent term |
| A unit root analysis of the 'Dutch disease' n | No trand & intercent term |
| Table 5.6a | |

| Sorice (all in locarithme) | No tren | d & interce | pt term | With tren | id & interc | ept term | With | intercept a | alone |
|----------------------------|---------|-------------|---------|-----------|-------------|----------|-------|-------------|-------|
| Jerres (an in ioganamics) | ESA | NA | WCA | ESA | NA | WCA | ESA | NA | WCA |
| Real capital flows | -0.43 | -0.06 | -0.92 | -5.32 | -1.75 | 4.48 | -5.17 | -1.40 | 4.36 |
| Real GDP growth @ | -8.70 | -5.30 | -9.90 | -8.66 | -5.24 | -10.0 | -8.64 | -5.29 | -9.90 |
| Terms of trade | 4.80 | -3.23 | -4.67 | -6.10 | -3.55 | -5.20 | -5.90 | -3.62 | -5.20 |
| Nominal exchange rate @ | 4.13 | -1.94 | -1.23 | -4.37 | -3.10 | -2.51 | -4.39 | -2.30 | -2.70 |
| Gov't consumption | -0.73 | 1.24 | -0.28 | -3.10 | -2.80 | -3.10 | -3.30 | -1.18 | -3.00 |
| Real exchange rate | | | | | | | | | |
| Agricultural raw material | -1.37 | | -0.88 | -2.39 | | -2.68 | -2.45 | | -2.36 |
| Beverage | -1.44 | | -1.00 | -2.41 | | -3.00 | -2.50 | | -2.50 |
| Food | -1.28 | -1.84 | -0.93 | -2.56 | -3.61 | -2.87 | -2.61 | -1.81 | -2.47 |
| Minerals | -1.27 | | -0.88 | -2.40 | | -2.64 | -2.45 | | -2.37 |
| | | | | | | | | | |

@ are not in logarithm.

| or the ADF test | |
|--------------------------|--|
| innon critical values fo | |
| le 5.6b Macki | |

| | ene | 10% | -2.58 | |
|----------------|---------------|-----|-------|--|
| | intercept alo | 5% | -2.89 | |
| | With i | 1% | -3.49 | |
| | pt term | 10% | -3.15 | |
| e ADF test | nd & interce | 5% | -3.45 | |
| alues for the | With tre | 1% | -4.03 | |
| on critical va | cept term | 10% | -1.62 | |
| Mackinno | & no interc | 5% | -1.94 | |
| Table 5.6b | No trend | 1% | -2.58 | |

| Equation | Johanssen | Mackinnon critical valu e | | |
|-------------------|-----------|---|--------|--|
| • | ratio | 5% | 1% | |
| ESA | | | | |
| Agr. raw material | 137.37 | 94.15 | 103.18 | |
| Beverage | 133.18 | 94.15 | 103.18 | |
| Food | 134.69 | 94.15 | 103.18 | |
| Minerals | 136.2 | 94.15 | 103.18 | |
| NA | | | | |
| Food | 110.32+ | 102.14 | 111.01 | |
| WCA | | | | |
| Agr. raw material | 179.74 | 94.15 | 103.18 | |
| Beverage | 181.46 | 94.15 | 103.18 | |
| Food | 172.35 | 94.15 | 103.18 | |
| Minerals | 175.5 | 94.15 | 103.18 | |

Table 5.6c Co-integration test for the 'Dutch disease' model

The Johansen test used assumes a linear deterministic trend in the data with intercept, no trend, in the test VAR (i.e., the Co-integration equation (CE), as an indicator of long run equilibrium relation, has no trend).

+ assumes intercept but no trend in CE and no intercept in test VAR.

5.3.3.2 The data

Real exchange rate is derived using a nominal exchange rate (World Bank, World Tables, 1994), price of aggregate commodities (re-computed from the UNCTAD series) and a domestic price (GDP deflator).²⁹ Terms of trade are derived using an export and import unit price index (World Bank, World Tables, 1994). Government consumption expenditure (IMF, GFS) is used as a proxy for government spending on non-tradables. Finally, capital inflow is derived by taking the difference between the current and preceding year's stock of bilateral, multilateral and private debt stock, plus grants (World Bank, World Tables, 1994). All flows are considered, since all may place pressure on the real exchange rate.³⁰ The real level of capital inflow, which is deflated by import unit price for each country, is also used. The results of the time series analysis and estimation results are given in Tables 5.6 to 5.9. These results will be discussed in the following section.

5.3.3.3 Estimation results

Tables 5.7 and 5.9 provide the first- and second-order estimation results for equation [6]. Both tables are presented in order to allow comparison

between the two formulations. Unless the two formulations are identical, or nearly identical, from an econometrics perspective, it is usually preferable to use the second-order ECM. Thus, in the context of this book the second-order ECM accommodates an extended foreign inflow and government spending lag structure, which is theoretically more plausible. Therefore, the following discussion is based on a secondorder ECM, with the exception of agricultural raw materials and beverage equations for the WCA region, which are based on the first-order ECM.

(a) East and Southern Africa (ESA)

The econometric results for this region confirm the existence of a 'Dutch disease' effect only within the food and beverage sectors. Although the second-order ECM for beverage suggests the possibility of 'Dutch disease' problems, as can be read from the coefficients of capital inflow indicator, its first-order ECM confirms its existence. Thus, it is noted that the impact of capital inflows will vary across commodities.

Two of the findings based on this estimation are worth examining further. Firstly, the apparent absence of 'Dutch disease' effects; and secondly, the finding that government consumption expenditure, which is assumed to show government spending on the non-traded sector, is negative and significant in almost all cases. Before arriving at Tables 5.7 and 5.9, a number of experiments based on different financial instruments were undertaken.³¹ The estimation resulting from these is basically similar to the one reported above. The use of both logarithms of government consumption and capital inflows as regressors within the same equations is undertaken after checking for possible correlation between these. The level of such correlation is found to be 0.15, which does not create a multicolinearity problem. The major finding arising from the results presented in this section is that government consumption has both short- and long-run elasticities ranging from -0.18 to -0.60. The use of a rigorous econometric specification and estimation yields different results from those arising out of previous studies, such as that undertaken by van Wijnbergen (1986a). Specifically, the direct effects of capital inflow on the real exchange rate are limited only to the food and beverage sectors. Furthermore, if capital inflows are used to finance government consumption expenditure, in almost all cases,

| e elasticities) (long-run values after | |
|---|--|
| First-order ECM results for Equation 5 (coefficients an transformation listed in Appendix 4.1) | |
| Table 5.7 | |

| Connoction ΔF_{-1} ΔToT ΔG ΔDev Δy F_{-2} ToT Agr. raw mat. 0.79 0.07 -0.07 -0.51 0.0004 -0.12 -0.07 -0.12 -0.01 -0.12 -0.01 -0.12 -0.01 -0.12 -0.01 -0.12 < | Commodity | | | Short | un | | | | - | Long run | | |
|--|---------------|-------|-------|-------|--------------|---------|-------|-------|-------|----------|--------|--------|
| ESA 0.79 -0.07 -0.51 0.0004 0.01 -0.12 0.01 | | Con | ΔF_1 | ΔΤοΤ | ΔG | ΔDev | Δy | F_2 | ToT_1 | م | Dev_1 | y_1 |
| Agr. raw mat. 0.79 -0.07 -0.51 0.0004 0.01 -0.2 R-values) 3.08° -0.07 -0.51 0.0004 0.12 -0.2 Beverage -0.18 0.002 0.85 -0.59 -0.24 0.01 -0.2 Food 1.00 0.08 0.07 -0.26 -0.23 -0.23 -0.23 -0.235 -0.235 -0.235 -0.235 -0.235 -0.22 -0.24 -0.35 -0.22 -0.22 -0.22 -0.26 -0.235 -1.28 -0.22 -0.26 -0.235 -0.22 -0.26 -0.235 -1.26 0.25 -0.26 -0.22 -0.26 0.23 -0.12 -0.26 0.23 -0.12 -0.26 0.23 -0.12 -0.26 0.23 -0.12 -0.26 0.23 -0.12 0.26 0.23 -0.12 0.26 0.23 -0.12 0.26 0.23 0.26 0.23 | ESA | | | | | | | | | | | |
| | Agr. raw mat. | 0.79 | | -0.07 | -0.51 | 0.0004 | | 0.01 | -0.20 | -0.45 | 0.0015 | |
| Beverage -0.18 0.002 0.85 -0.59 -0.24 -0.35 -3 (-values) -0.79 0.08 6.0° -2.6° -1.3 -1.5# 3. (-values) -0.79 0.08 6.0° -2.6° -1.3 -1.5# 3. (-values) 2.3° 1.9" -2.2° -0.20 0.25 0.25 0.25 0.25 0.25 0.26 0.25 0.26 0.146 0.1 Minerals 0.43 0.04 -1.13 2.12" 0.26 3.7" -0.1 Ma 0.55 -1.13 2.12" 0.26 3.7" -0.1 NA 0.55 -1.13 2.12" 0.26 3.7" -0.1 NA 0.55 -1.13 2.12" 0.26 3.7" -0.1 NCA 0.55 0.01 0.56 0.28 -0.14 -0.1 WCA 0.39 0.01 0.56 0.26 -1.5" -1.5" | (t-values) | 3.08* | | -0.90 | 4 .9* | 1.52~ | | 0.12 | -0.69 | -3.4* | 1.25 | -0.33 |
| (t-values) -0.79 0.08 6.0° -2.6° -1.3 -1.5# 3. Food 1.00 0.06 -0.37 -0.20 0.25 0.25 0.25 0.25 0.25 0.25 0.25 0.25 0.25 0.25 0.25 0.25 0.25 0.23 0.23 0.25 0.25 0.25 0.23 0.23 0.23 0.25 0.25 0.23 0.23 0.23 0.23 0.23 0.25 0.25 0.23 0.23 0.23 0.23 0.23 0.25 0.23 0.24 0.25 0.24 0.26 0.25 0.24 0.25 0.24 0.25 0.26 0.25 0.26 0.25 0.26 0.26 0.26 0.26 0.26 0.26 0.26 | Beverage | -0.18 | 0.002 | 0.85 | -0.59 | | -0.24 | -0.35 | 3.4 | 0.69 | 0.0025 | -1.28 |
| Food 1.00 0.06 -0.37 -0.20 0.25 0.2 (t-values) 2.3* 1.9* -2.2* -3.0* 0.23 0.25 0.2 Minerals 0.43 0.04 -0.22 0.0215 0.039 0.25 0.2 Minerals 0.43 0.04 -0.22 -0.22 0.039 0.23 0.1 Matues) 2.0* 2.3* -1.13 2.12** 0.26 0.23 0.1 MA 0.55 -1.13 2.12** 0.26 0.23 0.1 Kevalues) 2.0* 0.55 -1.13 2.12** 0.26 0.23 0.1 WCA 0.55 0.01 0.56 0.28 0.14 0.14 0.14 0.14 0.14 0.14 0.14 0.12 3.7* 0.01 0.25 0.12 0.12 0.12 0.12 0.12 0.12 0.12 0.14 0.12 0.12 0.12 0.12 0.12 0.12 0.12 | (t-values) | -0.79 | 0.08 | 6.0* | -2.6* | | -1.3 | -1.5# | 3.4* | 3.0* | 0.52 | -0.65 |
| (t-values) 2.3* 1.9** -2.2* -3.0* 1.46 0. Minerals 0.43 0.04 -2.2* -3.0* 1.46 0.23 (t-values) 2.0** 2.3* -1.13 2.12** 0.26 3.7* -0.1 Materials 0.43 0.04 -2.2* -3.0* -1.13 2.12** 0.26 3.7* -0.1 Materials 0.55 -1.13 2.12** 0.26 3.7* -0.1 National 0.55 -1.13 2.12** 0.26 3.7* -0.1 Kevalues) 2.0* 0.55 1.13 2.12** 0.26 3.7* -0.1 WCA 0.55 1.6* 0.28 -0.14 -1.5* <th>Food</th> <td>1.00</td> <td>0.06</td> <td>-0.37</td> <td>-0.20</td> <td></td> <td></td> <td>0.25</td> <td>0.17</td> <td>-0.21</td> <td>0.0026</td> <td>-0.32</td> | Food | 1.00 | 0.06 | -0.37 | -0.20 | | | 0.25 | 0.17 | -0.21 | 0.0026 | -0.32 |
| Minerals 0.43 0.04 -0.22 0.0215 0.039 0.23 -0.2 (t-values) 2.0* 2.3* -1.13 2.12** 0.26 3.7* -0.1 NA 0.55 -1.13 2.12** 0.26 3.7* -0.1 NA 0.55 -1.13 2.12** 0.26 3.7* -0.1 Food 0.55 -1.13 2.12** 0.26 3.7* -0.1 Ant 0.55 -1.13 2.12** 0.26 3.7* -0.1 Agr. raw mat. 0.39 -0.01 0.56 0.0027 -1.5* -3. VCA 3.5* -1.04 4.1* 3.5* -5.0* 17.7 Beverage 2.09 0 0.56 -0.12 5.0* 17.7 Beverage 2.09 0.48 -0.54 -1.28 5.0* 17.7 Revalues) 1.5# -1.28 -0.12 5.5* -1.28 5.0* Revalues < | (t-values) | 2.3* | 1.9** | -2.2* | -3.0* | | | 1.46 | 0.27 | -1.6# | 0.86 | -0.66 |
| (t-values) 2.0** 2.3* -1.13 2.12** 0.26 3.7* -0.1 NA 0.55 -1.13 2.12** 0.26 3.7* -0.1 NA 0.55 0.28 -0.14 -0.14 -0.14 -0.14 -0.14 -0.14 -0.14 -0.14 -0.14 -0.14 -0.14 -0.14 -0.14 -0.14 -0.14 -0.14 -0.14 -0.14 -0.15 -3.3 -0.14 -0.14 -0.14 -0.15 -3.3 -0.14 -0.15 -3.4 -1.15 -3.4 -1.15 -3.4 -1.15 -3.5 -0.12 -3.5 -0.12 -3.5 -0.12 -5.0* 17 -7.5 -3.5 -0.12 -5.0* 17 -7.26 6.6* -0.12 -1.128 -5.0* 17 -7.128 5.0* -1.128 5.5 -1.128 5.5 -1.128 5.5 -1.128 5.5 -1.128 5.5 -1.128 5.5 -1.128 5.5 -1.128 5.5 < | Minerals | 0.43 | 0.04 | | -0.22 | 0.0215 | 0.039 | 0.23 | -0.15 | -0.21 | 0.0324 | 0.31 |
| NA 0.28 -0.14 -0.15 -3.3 -0.14 -0.15 -3.3 -0.14 -0.12 -1.5 -3.3 -0.12 -3.5 -3.3 -0.12 -3.5 -3.3 -0.12 -3.5 -3.5 -3.5 -3.5 -3.5 -3.5 -3.3 -0.12 -3.5 | (t-values) | 2.0** | 2.3* | | -1.13 | 2.12** | 0.26 | 3.7* | -0.60 | -2.2* | 3.26* | 0.62 |
| Food 0.55 0.28 -0.14 -0.14 (t-values) 2.0** 1.6# -0.14 -0.14 WCA 0.39 -0.01 0.56 0.0027 -1.5- -3. WCA 0.39 -0.01 0.56 0.0027 -1.5- -3. WCA 0.39 -0.01 0.56 0.0027 -5.6* -3. Wratues) 2.4* -1.04 4.1* -5.6* -0.12 -5.0* 1.7 Beverage 2.09 0 0.97 -0.54 -0.12 5.0* 1.7 Food 0.75 0.48 -0.38 -0.12 5.0* 1.128 5.1 (t-values) 1.5# -0.28 -0.12 5.5 -0.12 2.5 -0.12 2.4 fourierals 1.5# -0.33 -0.28 -0.12 5.5 -0.47 1. | NA | | | | | | | | | | | |
| (t-values) 2.0** 1.6# -1.5~ -3. WCA 0.39 -0.01 0.56 0.0027 -1.5~ -3. WCA Agr. raw mat. 0.39 -0.01 0.56 0.0027 -2.6 6. Agr. raw mat. 0.39 -0.01 0.56 0.0027 -2.6 6. Agr. raw mat. 0.39 -0.01 0.56 0.027 -2.6 6. Beverage 2.09 0 0.97 -0.54 3.5* -0.12 2.0 Food 0.75 0.38 -0.38 -0.12 2.0 1.128 5.0 Food 0.75 0.38 -0.28 -0.12 2.1 -1.128 5.0 1.1 Minerals 1.5# -0.38 -0.28 -0.12 0.12 0.12 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05 < | Food | 0.55 | | | | 0.28 | | -0.14 | -0.79 | | 0.037 | 2.15 |
| WCA 0.39 -0.01 0.56 0.0027 -2.6 6. Agr. raw mat. 0.39 -0.01 0.56 0.0027 -2.6 6. Agr. raw mat. 0.39 -0.01 0.56 0.0027 -2.6 6. Reverage 2.09 0 0.97 -0.54 3.5* -0.12 2.1 Reverage 2.09 0 0.97 -0.54 3.5* -0.12 2.1 Food 0.75 0.38 -0.12 3.3* -2.5* -0.12 5.0* ft-values) 1.5# 0.38 -0.38 -0.12 5.4 -1.28 5.4 ft-values) 1.5# -0.38 -0.12 0.12 0.12 0.12 0.47 1. Aminerals 1.79 0.38 -0.22 0.00777 0.12 0.05 -0.47 0.05 -0.47 1. | (t-values) | 2.0** | | | | 1.6# | | -1.5~ | -3.5* | | 0.14 | 2.6* |
| Agr. raw mat. 0.39 -0.01 0.56 0.0027 -2.6 6. (t-values) 2.4" -1.04 4.1" 3.5" -5.0" 17 Beverage 2.09 0 0.97 -0.54 3.5" -5.0" 17 Beverage 2.09 0 0.97 -0.54 3.5" -0.12 2.1 Food 0.75 0.38 -0.12 3.5" -0.12 2.1 -0.12 5.0" 17 5.0" 17 5.0" 17 5.0" 17 5.0" 17 5.0" 11 5.0" 11 5.0" 11 5.0" 11 5.0" 11 5.0" 11 5.0" 11 5.0" 11 5.0" 11 5.0" 11 5.0" 11 5.0" 11 5.0" 11 5.0" 11 5.0" 5.0" 5.0" 5.0" 5.0" 5.0" 5.0" 5.0" 5.0" 5.0" 5.0" 5.0" 5.0" 5.0" </th <th>WCA</th> <td></td> | WCA | | | | | | | | | | | |
| (t-values) 2.4* -1.04 4.1* 3.5* -5.0* 1.7 Beverage 2.09 0 0.97 -0.54 -3.5* -5.0* 1.7 Reverage 2.09 0 0.97 -0.54 3.5* -0.12 2.1 Reverage 2.09 0 0.97 -0.54 3.2 2.1 -0.12 5. Food 0.75 0.48 -0.38 -0.12 5. -0.12 5. -0.12 5. -0.12 6. 1. -0.12 5. (t-values) 1.6# -0.17 0.12 0.07 1. -0.47 1. -0.47 1. -0.47 1. -0.47 1. -0.47 1. -0.47 1. -0.47 1. -0.47 1. -0.47 1. -0.47 1. -0.47 1. -0.47 1. -0.47 1. -0.47 1. -0.47 1. -0.47 1. -0.47 1. -0.47 1. < | Agr. raw mat. | 0.39 | -0.01 | 0.56 | | 0.0027 | | -2.6 | 6.16 | | 0.02 | |
| Beverage 2.09 0 0.97 -0.54 -0.12 2.1 (t-values) 4.5* 6.6* -4.0* -1.28 5. (t-values) 0.75 0.48 -0.38 -0.12 2.8 (t-values) 1.6# -0.38 -0.12 2.7 2.128 5. (t-values) 1.6# -0.38 -0.12 1.28 5. 2.12 1.12 </th <th>(t-values)</th> <td>2.4*</td> <td>-1.04</td> <td>4.1*</td> <td></td> <td>3.5*</td> <td></td> <td>-5.0*</td> <td>1.7#*</td> <td></td> <td>0.93</td> <td></td> | (t-values) | 2.4* | -1.04 | 4.1* | | 3.5* | | -5.0* | 1.7#* | | 0.93 | |
| (t-values) 4.5* 6.6* -4.0* -1.28 5. Food 0.75 0.48 -0.38 -0.28 -0.12 2 (t-values) 1.6# 3.3* -2.5* -0.077 -1.6# -0.47 1. Minerals 1.79 0.72 0.05 -0.12 2 2 | Beverage | 2.09 | 0 | 0.97 | -0.54 | | | -0.12 | 2.08 | | 0.0022 | -0.76 |
| Food 0.75 0.48 -0.38 -0.28 -0.12 2 (t-values) 1.6# 3.3* -2.5* -0.17 -1.6# -0.47 1. Minnerals 1.79 3.3* -2.5* 0.0177 0.12 0.05 -0.47 1. | (t-values) | 4.5* | | 6.6* | 4 • | | | -1.28 | 5.6* | | 0.72 | -6.01 |
| (t-values) 1.6# 3.3* -2.5* -1.6# -0.47 1. ininerals 1.79 -0.22 0.0177 0.12 0.05 -0. ininerals 1.79 -0.22 0.0177 0.12 0.05 -0. | Food | 0.75 | | 0.48 | -0.38 | | -0.28 | -0.12 | 2.6 | -1.21 | -0.03 | -2.21* |
| Minerals 1.79 -0.22 0.00177 0.12 0.05 -0. | (t-values) | 1.6# | | 3.3* | -2.5* | | -1.6# | -0.47 | 1.5~ | -2.33* | -2.4* | -7.09* |
| // | Minerals | 1.79 | | | -0.22 | 0.00177 | 0.12 | 0.05 | -0.03 | -0.03 | 0.0029 | 0.070 |
| (r-values) 3.0" -2.30 2.3 0.11 1.01 -0. | (t-values) | 3.8* | | | -2.58* | 2.9* | 0.77 | 1.07 | -0.19 | -0.40 | 1.3 | 1.02 |

(Continued)

| (pər |
|--------|
| ontinu |
| ğ |
| 5.7 |
| able |

| | | Adj. R² | | | Diagnost | ic check | |
|---------------|--------|---------|--------|-------|----------|-----------------|---------|
| Commodity | RER_1 | ECM | Level@ | Chow | Reset | Jarque- Bera | Multc++ |
| ESA | | | | | | | |
| Agr. raw mat. | -0.25 | 0.34 | 0.96 | 2.2* | 0.32 | 0.004* | ELow |
| (t-values) | -3.25* | | | | (0.58) | | |
| Beverage | -0.11 | 0.32 | 0.60 | 2.75* | 0.84 | 1.47* | ELow |
| (t-values) | -2.4* | | | | (0.36) | | |
| Food | -0.23 | 024 | 0.69 | 2.6* | 0.40 | 2.5* | |
| (t-values) | 4.0* | | | | (0.53) | | ELow |
| Minerals | -0.35 | 0.28 | 0.87 | 2.26* | 0.19 | 2.2* | |
| (t-values) | -4.3* | | | | (0.66) | | ELow |
| NA | | | | | | | |
| Food | -0.47 | 0.44 | 0.86 | 2.5* | 0.31 | 0.83* | VLow |
| (t-values) | -4.7* | | | | (0.58) | | |
| WCA | | | | | | | |
| Agr. raw mat. | -0.03 | 0.50 | 0.50 | 2.7* | 6.3 | 6.7 | VLow |
| (t-values) | -0.86 | | | | (0.02) | | |
| Beverage | -0.29 | 0.58 | 0.69 | 1.19* | 0.22 | 1.05* | VLow |
| (t-values) | -4.11* | | | | (0.64) | | |
| Food | -0.06 | 0.39 | 0.35 | 0.94* | 'n | 0.05* | VLow |
| (t-values) | -1.14 | | | | (0.09) | | |
| Minerals | -0.29 | 0.49 | 0.33 | 0.95* | 0.95 | 2.22* | VLow |
| (t-values) | 4.0* | | | | (0.33) | | |

| cant at 1% and better. icant at 5% and # signifi- 10% (~ closer to 10%). Iong run coefficients and alues are computed using kens and Breusch (1988) | x 4.1). g-run (level) R ² is obtained s same regression. show one, two period lags. bles are as defined in | amining the degree of linearity simple correlation regressors < 0.25 is con- regressors < 0.25 is con- regressors < 0.25 is con- regressors < 0.60 creatable). Careptable). Careptable). Careptable). Careptable). Careptable in an e-values in are P-values. | |
|--|---|--|--|
| fes: ignificant significant at 10% The long in t-value: Mickens d Gurney | e long-rui e long-rui m this sai m this sai mation 5 | The diagonal of the diagonal o | |

| | | | Cou | intry | | |
|--|---------------|-----------------|---------------|----------------------|----------------------|-----------------------|
| Commodity | Bot- swana | Ethiopia | Kenya | Mada- gascar | Tanza- nia | Zambia |
| ESA Agr. raw mat. Beverage Food Minerals | -0.75 | 0.17 -0.65 | 0.79 0.75 | 1.31 0.70 0.60 | 0.96 0.22 0.90 | 0.35 -0.45 0.31 |
| WCA Country Beverage | | Zaire -0.34 | | | | |
| NA Country FOOD | | Algeria 1.52 | Egypt 0.60 | | | |

 Table 5.8
 Country (constant) dummies used in the estimation of Table 5.7 (those significant at 1% or better)

this will be the main channel through which the 'Dutch disease' effect might operate. Since most aid flows are targeted at governments, this is likely to represent an important propagation mechanism.

Within the *mineral*-based estimation, contrary to our theoretical expectation, long-run capital inflows are found to have a statistically significant positive effect on real exchange rate within the first-order ECM, and a statistically insignificant but positive coefficient in the second-order ECM. As with the WCA region, the regional price index for the ESA region showed a general rising trend.³² This, combined with a statistically significant short- and long-run positive impact of devaluation, is likely to work against a direct 'Dutch disease' effect.

A number of other features relating to other determinants of the real exchange rate are also worth highlighting. Firstly, the terms-of-trade indicator is found to be statistically significant only for the beverage sector and, in the short run, the food sector. It has a positive coefficient, indicating the prevalence of the substitution effect. Devaluation is found to be important in the short run and only within the mineral sector. As an indicator of technology, real GDP growth is found to be important in the short run within the beverage and food sectors, and in the long run within mining. In all cases the coefficient exhibits a negative value, indicating the importance of the income rather than the factor augmentation effect.

| - |
|---|
| ~ |
| $\overline{\Omega}$ |
| ш |
| |
| 5 |
| യ |
| 0 |
| 5 |
| 0 |
| Ĵ. |
| σ |
| Ē. |
| õ |
| ĸ |
| 2 |
| Ψ |
| ŝ |
| - |
| õ |
| ~ |
| Ś |
| 2 |
| 5 |
| ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~ |
| ~ |
| ~ |
| œ |
| |
| |
| |
| σ |
| |
| ŝ |
| |
| <u>_</u> |
| 5 |
| 1 |
| -0 |
| - |

East & Southern Africa (ESA)

1. Agricultural raw materials

$$\Delta RER = -0.62\Delta G - 0.12\Delta G_{t-1} - (-6.8)^* (-1.15) \\ (-6.8)^* (-1.15) \\ 0.33(RER_{t-2} + 0.027F_{t-3} - 0.257ToT_{t-2} + 0.176G_{t-2} - 0.0007Dev_{t-2} + 0.196y_{t-2} - 1.76) + (-4.1)^* (-0.82) (1.28) (-1.6)^- (0.91) (-0.69) (2.5)^* \\ (-4.1)^* (-0.82DKen + 1.98DMad + 1.19DTan + 0.41DZam \\ 0.15DEth + 0.88DKen + 1.98DMad + 1.19DTan + 0.41DZam \\ (2.2)^* (3.6)^* (4.1)^* (3.9)^* (3.3)^* \\ (5.3.1)^* (5.3.1)^* (5.3.1)^* (5.3.1)^* \\ (5.3.1)^* (5.3.1)^* (5.3.1)^* (5.3.1)^* \\ (5.3.1)^* (5.3.1)^* (5.3.1)^* (5.3.1)^* \\ (5.3.1)^* (5.3.1)^* (5.3.1)^* \\ (5.3.1)^* (5.3.1)^* (5.3.1)^* \\ (5.3.1)^* (5.3.1)^* (5.3.1)^* \\ (5.3.1)^* (5.3.1)^* (5.3.1)^* \\ (5.3.1)^* (5.3.1)^* (5.3.1)^* \\ (5.3.1)^* (5.3.1)^* (5.3.1)^* \\ (5.3.1)^* (5.3.1)^* (5.3.1)^* \\ (5.3.1)^* (5.3.1)^* \\ (5.3.1)^* (5.3.1)^* \\ (5.3.1)^* (5.3.1)^* \\ (5.3.1)^* (5.3.1)^* \\ (5.3.1)^* (5.3.1)^* \\ (5.3.1)^* (5.3.1)^* \\ (5.3.1)^*$$

 R^{2} -adj = 0.44 Jarque-Bera 0.16* Reset 0.12(0.73) LM 0.60(0.44) BG = .60(0.50) Eth = Ethiopia, Ken = Kenya, Mad = Madagascar, Tan = Tanzania, Zam = Zambia. Notes:

Values in bracket are t-values. See Appendix 4.1 for diagnostic test P-values. The sign difference between the t-values

and the coefficients in the ECM term is due to the fact that t-values refer to the long run coefficient. * implies 1%, ** 5% and $^{\wedge}$ 10% (~ closer to 10%) level of significant. LM is hetroschedasticity test while BG is the Breuch-Godfery test for serial correlation. The 5% (2 degrees of freedom) value for Jarque-Bera is 5.99. D stands for constant country dummy (e.g. DKen is dummy for Kenya).

| UU. |
|-----|
| ົດ |
| 50 |
| - |
| e |
| > |
| ĺ۵. |
| ~ |
| ш |
| |
| |
| si. |
| |

$$\Delta RER = -0.60\Delta G - 0.56\Delta G_{r1} + 1.1\Delta To T + 0.20\Delta Tr_{r1} - 0.20\Delta y - (-1.5) - (-1.5) - (-1.5) - (-1.5) - (-1.5) - (-1.5) - (-1.5) - (-1.9)^* (-0.45) - (0.00711Dev_{r2} - 6.46y_{r2} + 1.07) + (-1.9)^{**} (-1.1) - (4.1)^* (1.5) - (1.3) - (3.4)^* (-0.45) - (0.45) - (0.75)^{**} (-1.1) - (4.1)^{**} (-1.1) - (4.1)^{**} (-1.5) - (-1.3) - (-1.3) - (-1.3) - (-0.45) -$$

ц.

$$\Delta RER = -0.16\Delta G + 0.266\Delta G_{r-1} + 0.0163\Delta Dev + 0.014\Delta Dev_{r-1} - (-1.03) (1.55) - (2.0)^{**} (1.4)$$

$$(-1.03) (1.55) - (2.0)^{**} (1.4)$$

$$0.37(RER_{r-2} - 0.0267F_{r-3} - 0.17To T_{r-2} - 0.0008G_{r-2} - 0.016Dev_{r-2} + 0.64y_{r-2} - 0.64) + (-4.7)^{*} (0.56) (0.75) (0.08) (1.3) (-1.9)^{**} (-1.3)$$

$$0.76DKen + 1.0DTan + 0.35DZam (1.3) (-1.9)^{**} (-1.9)^{**} (-1.3)$$

$$0.76DKen + 1.0DTan + 0.35DZam (3.3)^{*} (3.3)^{*} (3.3)^{*} (3.3)^{*} (3.7)^{*} (3.3)^{*} (3.3)^{*}$$

$$R^{2}$$
-adj = 0.35 Jarque-Bera 2.5^{*} Reset 0.23(0.64) LM 0.00(0.98) BG = 1.8(0.16)
Notes as given in Equation 5.3.1.
Corth Africa (NA)
$$Food$$

$$Food$$

$$Food$$

$$\Delta RER = -0.64 \Delta RER_{r-1} - 0.21 \Delta G + 0.29 \Delta G_{r-1} + 0.15 \Delta ToT + 0.28 \Delta y - (-4.2)* (-1.0) (1.4) (0.90) (1.3) (-4.2)* (-5.5)* (-5.6)* (-2.6)* (-3.1)* (0.90) (1.3) (-0.103 Dev_{t-2} + 0.12 y_{t-2} - 1.91) + 1.11 DAlg (-5.5)** (2.5)* (-2.6)* (-2.6)* (-3.1)* (0.51) (-0.31) (2.4)* (5.3)* [5.3.5] R^2 - adj = 0.50 Jarque-Bera 1.6* Reset 0.93(0.34) LM 0.90(0.35) BG = 1.33(0.28) Notes as given in Equation 5.3.1.$$

| ί. | . Food | |
|----|--|---------|
| | $\Delta RER = -0.40\Delta G - 0.22\Delta G_{r_1} + 0.75\Delta ToT + 0.35\Delta ToT_{r_1} - 0.25\Delta y - (-2.9)* (-1.53) - (3.9)* (2.04)** (-1.6)^{-1.53} - (3.9)* (-1.53) - (-0.823ToT_{r_2} + 0.592G_{r_2} - 1.72y_{r_2} - 9.37) - 0.18(RER_{r_2} + 0.062F_{r_3} - 0.823ToT_{r_2} + 0.592G_{r_2} - 1.72y_{r_2} - 9.37) - (-3.2)* (-0.69) (1.11) (-2.4)* (2.1)** (3.3)*$ | [5.3.6] |
| | R^2 -adj = 0.52 Jarque-Bera 0.32* Reset 0.00(0.99) LM 0.08(0.77) BG = 0.34(0.56) Notes as given in Equation 5.3.1. | |
| Ň | . Minerals | |
| | $\Delta RER = -0.272 \Delta RER_{i-1} + 0.016 \Delta F_{i-2} - 0.239 \Delta G - 0.168 \Delta y - (-2.9)^* (1.05) (-3.7)^* (1.8)^{\wedge}$ | |
| | $\begin{array}{llllllllllllllllllllllllllllllllllll$ | [5.3.7] |
| | R^2 -adj = 0.54 Jarque-Bera 1.0* Reset 2.12(0.15) LM 2.9(0.10) BG = 0.27(0.60) Notes as given in Equation 5.3.1. | |

West & Central Africa (WCA)

To sum up, a common phenomenon observed within the region is that adjustment coefficients are relatively higher than for the WCA region, with the exception of minerals. With the exception of beverage (for which the ECM coefficient stands at 9 per cent), the adjustment coefficient ranges from 27 to 37 per cent, meaning that 27–37 per cent of past errors are corrected for in the current period. Within all the models, values for the relevant diagnostic tests are quite acceptable. Perhaps more importantly, evidence for the 'Dutch disease' is not apparent within the minerals and agricultural raw materials sectors. However, government expenditure is, in general, found to have a statistically significant negative coefficient, which leads to a real appreciation. Thus, depending on how government expenditure is financed, there could be room for capital inflows to affect the real exchange rate indirectly.

(b) North Africa

In the database upon which this work is based, three commodity groups were specified for North Africa. These are *agricultural raw materials*, *food* and *minerals*. The application of the model to agricultural raw materials and a minerals-based real exchange rate could not pass the co-integration test. This indicates the absence of a long-run relationship between these variables. Thus, the estimation is undertaken only for the food-based real exchange rate, for which a co-integration relationship is found.

As summarized in Table 5.9, the estimation confirms the existence of 'Dutch disease' problem in the long run, although its potency is low. However, a strong long-run 'Dutch disease' effect could be expected, depending on how government consumption expenditure is financed. Long-run terms of trade are also found to have a negative elasticity, highlighting the importance of the income effect. An interesting aspect of this model is its high adjustment coefficient, which, standing at 77 per cent, indicates a rapid adjustment capability within the region.

(c) West and Central Africa

For the WCA region, we begin from the estimation for the *agricultural* raw materials-based real exchange rate. As can be seen from Table 5.7, and as might be suggested by the theory, the coefficient for capital inflows has a strong and statistically significant value in the long run. Terms of trade, on the other hand, has a positive effect on the real ex-

change rate, although its short-run effect is less potent. In common with the ESA region, this result would tend to indicate the importance of a substitution effect. As expected, nominal devaluation also has a statistically significant positive impact in the short run. However, caution should be exercised in relation to this result, since the lagged value of the real exchange rate is significant at only around a 20 per cent level. The results of the remaining diagnostic tests are all quite satisfactory. Hence, the findings of this model might reasonably be taken as suggesting the possibility of a 'Dutch disease' effect.

The results of the food-based real exchange rate model are very strong in relation to the entire range of diagnostic tests. The existence of a direct 'Dutch disease' effect arising from foreign inflows is not supported by the data. However, as in the previous cases, government consumption expenditure has statistically significant negative long- and short-run values. Depending on how the latter is financed, a 'Dutch disease' effect may be traced. The coefficient in relation to terms of trade is positive in the short run, suggesting that the substitution effect remains important. Technological progress, again measured as real GDP growth, has a negative effect in the short run, owing to the income effect, but a positive effect in the long run, due to its factor augmentation impact.

The estimation for the *beverage* category also produces similar results. Again, the diagnostic test results are very strong, with these results suggesting possible direct 'Dutch disease' effects in the long run. This suggestion is supported by the statistically significant negative value of current government expenditure, which could partly be financed by aid. The terms of trade effect shows similar results as that for food and agricultural raw materials, having a positive coefficient with a high potency, especially in the long run. The effect of devaluation is not found to be statistically significant, either in the short or long run. The long-run growth effect is negative, suggesting the importance of an income effect.

Contrary to our theoretical expectation, the indicator of capital inflows within the *minerals* equation shows a positive coefficient in the short run. However, these coefficients are not found to be statistically significant at conventional levels of significance. The price index of minerals, ores and metals for the region rose sharply between 1972 and 1980, and again from its lowest level in 1986 to its highest in 1989. As with the ESA region, this, combined with the nearly zero impact of devaluation, could work against a possible 'Dutch disease' effect. However, government consumption expenditure exhibits a statistically significant negative coefficient, in the short run. Terms of trade are found to have no effect within this model, while technological progress has a negative impact, associated with an income effect, in the short run. In relation to the beverage and mineral models, 26 per cent and 43 per cent, respectively, of past errors are made up for in the current period. This is comparatively high by the standard of the other two commodities and might be related to the capacity utilization possibility within these sectors, as noted in Chapter 4.

In sum, there would appear to be similarities between the results from the ESA region and the WCA region. Firstly, within this region the 'Dutch disease' effect is directly observed only in relation to the agricultural, and to some degree, the beverage sectors. However, in almost all cases, government current expenditure exhibits a statistically significant negative coefficient. If part of the latter is financed by foreign inflows, then the 'Dutch disease' effect could work its way through that. Secondly, with the exception of the food sector, in general, terms of trade is found to have a positive impact on the real exchange rate, with a particularly strong long-run impact. This underlines the importance of the substitution effect. And thirdly, in the case of food and agriculture-based real exchange rates, adjustment to equilibrium is found to be extremely slow. However, the results for the beverage and mineral-based real exchange rate suggest a faster adjustment, owing to capacity utilization, as described in Chapter 4.³³

5.3.4 Summing Up

In this section an attempt has been made to assess the empirical validity of the macroeconomic impact of aid on the real exchange rate. This analysis emerges from a discussion of the 'spending effect', within a classic 'Dutch disease' framework. The study not only proposes an alternative propagating mechanism relevant for Africa, but also attempts to assess the empirical validity of such a mechanism, using recently developed econometric techniques.

The results indicate that the 'Dutch disease' effect varies across regions and commodity categories. However, previous studies were not able to identify such variation, because these studies were based on an aggregate real exchange rate index. Moreover, most of these studies used simple OLS estimation techniques, without giving due attention to the non-stationarity of each series, nor to the co-integration behaviour among the variables. In contrast, this study explicitly adopts a procedure, which takes account of these issues, in conjunction with an ECM approach. This not only rectifies the shortcomings of previous studies, but also highlights the potential for new time series econometrics to explain the temporal and permanent impact of real exchange rate fundamentals, on top of providing an in built error correction mechanism.

Based on this analysis, one may conclude that a 'Dutch disease' effect can be observed in the food and beverage sectors of the ESA region, the food sector of the NA region and the agricultural raw materials and beverage sectors of the WCA region. Interestingly, in almost all cases, government spending on the non-traded sector is found to have a statistically significant negative elasticity. Thus, insofar as part of this spending is financed by foreign inflows, a 'Dutch disease' effect may be observed indirectly. This is entirely plausible since most capital flows in Africa are directed to the public sector. However, the policy implications of the above finding should not necessarily always be an across-the-board devaluation. Rather, for import-dependent economies (such as those examined in this study), policies such as devaluation should be analysed within a broader overall macroeconomic context. In other words, account should be taken of the contractionary impact of devaluation, as well as its implications for public expenditure. On the other hand, a case for protecting other traded sectors may also be made. The findings summarized within this section would also tend to support the view that there is a need for serious macroeconomic management, in the face of capital inflows, whether they come as aid, or in some other form. In this respect, an important instrument of macroeconomic management is government spending and how it is financed. Thus, African macro modelling should, ideally, take account of such considerations. Hence, the African prototype model developed in the following chapter will explicitly attempt to incorporate these considerations.

5.4 PRIVATE INVESTMENT AND IMPORT COMPRESSION: ECONOMETRIC RESULTS

Sections 5.2 and 5.3, focused on the possible macroeconomic problems associated with foreign inflows. It is clear, however, that foreign inflows do have a positive impact. Indeed, one of the most important positive effects of such inflows in relation to Africa stems from the possi-

bility that they offer to tackle the foreign exchange, or import, constraint which these countries face. This effect may be examined empirically by focusing on private investment and identifying the various determinants of such investment. This avenue of enquiry will be pursued within this section.

Determinants of private investment have been analysed using a variety of different theories. The accelerator model (Clark 1917, Keynes 1936, Chenery 1952), the Tobin-q model (Tobin 1969) and the user cost model (Jorgenson 1963) are the basic models upon which much of the analysis is based. Within the African context the application of the Tobin-q model is limited, since capital markets in that Continent remain extremely rudimentary. Thus, the competing models, in relation to this analysis, are Jorgenson's user cost model and a modified accelerator model, which incorporates the specificity of African economies. Jorgenson's user cost model is difficult to employ in Africa, partly as a consequence of data problems relating to depreciation and cost of capital and labour. Moreover, the model assumes substitution between these factors of production. However, this assumption is less plausible in Africa, where foreign exchange constraints are pervasive. The private investment function, set out below, is basically an acceleratorbased model (Chenery 1952), modified to accommodate the external constraint to private investment in developing countries (FitzGerald et al. 1992). FitzGerald et al.'s work is further extended by allowing the inclusion of other relevant variables, and formulating this within an Error Correction Model. This externally-constrained private investment function is specified below. Its estimation result, following a unit root and co-integration test, is reported below.

The Model

Private (p) investment (I) is defined as reflecting the adjustment of capital stock (K) to its desired level (K^*)

$$I_{p} = \lambda \left(K_{p}^{*} - K_{pt-1} \right)$$
[1]

where: λ is an adjustment coefficient.

The desired capital stock is further defined as,

$$K_{p}^{*} = b_{1}Y_{i} + b_{2}K_{gi-1} + b_{3}M_{A} + b_{4}J_{A} + b_{5}DG$$
^[2]

where M is level of import, J_A an indicator of capital flight and DG the level of public debt.

Substituting [2] into [1] and taking the first difference of the result, to circumvent the capital stock data problem, yields equation [3], which is estimated below,

$$I_{p} = \beta_{1} \Delta Y_{t} + \beta_{2} I_{gt-1} + \beta_{3} \Delta M_{A} + \beta_{4} \Delta J_{A} + \beta_{5} Z_{A} + I_{pt-1}$$
[3]

where $\beta_i = b_i$ and Z_A public borrowing.

Equation [3] is estimated using pooled data for the sample of countries in each of the three regions. The estimation is preceded by the analysis of the time series properties of the variables used (Tables 5.10a -5.10c). The series are, in general, nonstationary and co-integrated. Due to lack of sufficient data Z_A and J are omitted from the estimation. The results of this estimation are given below.

East and Southern Africa

$$\Delta \ln \left(\frac{I_p}{Y}\right) = 0.04 \Delta \ln \left(\frac{Y_t}{Y_{t-1}}\right) + 0.56 \Delta \ln \left(\frac{I_g}{Y}\right)_{t-1} + 0.68 \Delta \ln \left(\frac{\Delta M}{Y}\right)$$

$$(0.32) \quad (4.3)^* \quad (4.3)^*$$

Country constant dummies Mad = Madagascar, Uga = Uganda, CAR = Cent. African Republic, Gab = Gabon, Nig = Nigeria and Alg = Algeria

| Table 5.10a | A unit roo | t analysis | of the pr | ivate inve | stment mo | odel: an A | DF test | | | |
|--------------------|------------------|---|-------------|--------------|-----------|-------------------|------------------------------|--------------------------------|--------------------------------|------|
| Series (all in | logarithms | No tren | d & interce | ept term | With tren | td & interc | ept term | With | intercept a | lone |
| as the ratio o | f GDP) | ESA | NA | WCA | ESA | NA | WCA | ESA | NA | WCA |
| Private investme | ent | -0.43 | -0.23 | -0.87 | -3.3 | -2.6 | - <u>3.9</u> | -3.18 | -2.1 | -3.9 |
| GDP growth | | -8.7 | -5.3 | -9.9 | -8.66 | -5.24 | -10.0 | -8.64 | -5.29 | -9.9 |
| Import | | -1.08 | -1.1 | -1.01 | -3.6 | -2.45 | -3.7 | -3.62 | -0.59 | -3.5 |
| Public investme | nt | -1.3 | -0.41 | -1.09 | -3.2 | -2.96 | -5.3 | -3.21 | -3.0 | -4.8 |
| @ are not in logar | ithm. | - | | | | | | | | |
| Table 5.10b | Mackinno | n critical | values foi | the ADF | test | | | | | |
| No trend & | no intercept ter | E | With trend | & intercep | ot term | Ň | th intercep | ot alone | | |
| 1% | 5% 10% | ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~ | 1% | 5% | 10% | 1% | 5% | 109 | 9 | |
| -2.58 | -1.94 -1.6 | 1 | 4.03 | -3.45 | -3.15 | -3.49 | -2.89 | -2.5 | 80 | |
| Table 5.10c | Co-integra | ation test | for the pri | vate inve | stment mo | leb | | | | |
| | | | | Mackinnor | | - Notes: | | | | |
| Equation # | Jona | ihood | U | ritical valu | e | The Jo trend i | hansen test i n CF and no | used assume intercent in th | ss intercept n he test (i e | 0 |
| | E | tio | 5% | | 1% | the Co | -integration e | equation (CE) |), as an indi- tion has no | |
| ESA | 210 | .43 | 53.12 | | 60.16 | trend). | 5 | | | |
| NA | 02 | .65 | 53.12 | | 60.16 | + assu | mes No inter | cept or trend | in CE or tes | |
| WCA | 223 | .08+ | 39.89. | | 45.58 | VAR | | | | |

External Finance and the African Economy: A Macro Approach 223

West and Central Africa

$$\Delta \ln \left(\frac{I_p}{Y}\right) = 0.28 \Delta \ln \left(\frac{Y_t}{Y_{t-1}}\right) - 0.12 \Delta \ln \left(\frac{I_g}{Y}\right)_{t-1} + 0.74 \Delta \ln \left(\frac{\Delta M}{Y}\right)$$

$$(1.7)^{\wedge} \qquad (1.4) \qquad (4.5)^{*}$$

$$- 0.39 \ln \left[\left(\frac{I_p}{Y}\right)_{t-1} - 1.28 \ln \left(\frac{Y_t}{Y_{t-1}}\right)_{t-1} + 0.11 \ln \left(\frac{I_g}{Y}\right)_{t-2} - 3.0 \ln \left(\frac{\Delta M}{Y}\right)_{t-1} + 2.72 \right]$$

$$(-6.4)^{*} \qquad (-0.92) \qquad (4.7)^{*} \qquad (-9.0)^{*}$$

$$- 1.1CAR + 1.2Gab - 0.32Nig \qquad (-0.92) \qquad (4.7)^{*} \qquad (-9.0)^{*}$$

$$- 1.1CAR + 1.2Gab - 0.32Nig \qquad (-0.92) \qquad (-0.92) \qquad (-0.92) \qquad (-9.0)^{*}$$

$$R^{2} - adj = 0.34 \qquad \text{Jarque-Bera} = 0.10 \qquad \text{Chow} = 1.9 \qquad \text{Reset} = 2.2 \qquad \text{LM} = 1.5(0.23) \qquad \text{R}^{2} - \text{Ievel} = 0.59 \quad \text{BG} = 0.44(0.65) \qquad n = 114 \qquad \text{Values in bracket are t-values.}$$

* significant at 1% ** at 5% and ^ at 10%.

North Africa

$$\Delta \ln \left(\frac{I_p}{Y}\right) = -0.16 \Delta \ln \left(\frac{I_g}{Y}\right)_{t-1} + 0.29 \Delta \ln \left(\frac{\Delta M}{Y}\right)$$
(1.1)
(3.1)*
$$-0.19 \ln \left[\left(\frac{I_p}{Y}\right)_{t-1} + 0.89 \ln \left(\frac{I_g}{Y}\right)_{t-2} - 0.67 \ln \left(\frac{\Delta M}{Y}\right)_{t-1} + 3.8\right] + 0.144 lg$$
(-6.4)*
(-2.6)*
(0.81)
(0.81)
(3.1)*
(1.8)**
$$R^2 - adj = 0.33$$
Barque-Bera = 0.57
$$Chow = 3.4$$
Reset = 0.39
$$LM = 0.32(0.57)$$

$$R^2 - level = 0.36$$

$$BG = 2.8(0.07)$$

$$n = 57$$

Values in bracket are t-values.

* significant at 1% ** at 5% and ^ at 10%.

Two important features of the private investment functions, estimated above, are worth noting. Firstly, the result conclusively supports the hypothesis that private investment in Africa is externally constrained. Thus, it is reasonable to assume that the import compression argument is a valid one. These results also indicate that the long-run impact of import compression is stronger than its short-run impact. Hence, it is entirely plausible that short-run coefficients might show the role of intermediate imports while the long-run coefficient might imply capacity creation. Thus, research using disaggregated imports could represent a productive area for future study. The second noteworthy feature of the above estimation is that the results suggest crowding-in within the ESA region and crowding-out in the NA region. However, the coefficients for public investment in relation to the WCA region are not found to be statistically significant. This result underscores the variation in the impact of public investment, which exists across regions, as well as its positive impact in relatively less developed regions such as ESA

5.5 CONCLUSION

In this chapter, the macroeconomic impact of external finance in Africa was examined from three angles. Firstly, by way of the fiscal response literature; secondly through the 'Dutch disease' effect; and finally, in relation to the financing of imports and investment.

To date, the government's fiscal response to external finance has been studied exclusively within an optimization framework. This approach not only is inherently inconsistent, because of the additive assumption, but also diverges from the reality of the situation within developing countries, in general, and African economies in particular. Thus, section 5.2 proposes an alternative decision-making framework, based on a number of stylized facts in relation to Africa. By using an Error Correction Model (ECM), an empirical analysis of the fiscal response to external finance is made. The results indicate firstly, that the impact of capital inflows on taxes varies across different types of capital inflows and regions. Thus, in general, bilateral flows are found to have a negative impact on direct taxes. Secondly, capital inflows are also likely to have a strong positive impact on current government expenditure within the three regions identified in this study. Indeed, it is
argued that the combined effect of these may push the government away from a sustainable path of financing its deficit.

Section 5.3 questions both the empirical approach and certain aspects of the theoretical underpinnings forwarded in order to describe the propagating mechanism of the 'Dutch disease' in Africa. Building on this discussion, a new estimation of the 'Dutch disease' effect, using an ECM framework is undertaken. A number of conclusions may reasonably be drawn based on these results. Firstly, it is desirable to calculate a commodity-specific real exchange rate, since the impact of capital inflows will vary across commodities. Secondly, the empirical evidence confirms the existence of a 'Dutch disease' effect in the food and beverage sectors within the ESA, agricultural raw materials and beverage sectors within WCA, and, finally, the food sector within the NA region. And, thirdly, government current expenditure is found to lead to an appreciation in all regions, and for all commodity categories. Hence, if the latter is to be financed by external inflows, then the 'Dutch disease' effect will reveal itself through these.

Finally, section 5.4 describes the externally constrained nature of private investments in Africa. This result emphasizes the importance of financing imports in a bid to accelerate investment. No definitive conclusion may be arrived at in relation to the impact of public investment within the WCA region. However, a crowding-in and crowding-out effect does appear to be taking place within the ESA and NA regions, respectively. In sum, the results presented in this chapter underscore the need to take explicit account of a range of additional considerations in relation to macro modelling in Africa. Hence, such considerations will be incorporated into the prototype African model, developed in the next chapter.

6 Modelling Africa Within a Global Economic Framework

6.1 INTRODUCTION

In the preceding chapters, the trade and finance problems of Africa were examined from both theoretical and empirical angles. In this chapter, I will attempt to locate these problems within a global economy context. This is important for two main reasons. Firstly, it helps one to identify the position of Africa within the world economy. Secondly, it allows one to assess the extent to which Africa's entry into the world economy represents an obstacle to development efforts being undertaken in that continent. This exercise will be formally undertaken using the increasingly popular methodology of modelling North-South economic interaction. Notwithstanding the proliferation of such models, the modelling of the South remains, at best, rudimentary. Indeed, almost no examples of North-South models focusing specifically on Africa have been developed to date.¹

In this chapter, an attempt will be made to examine both the theoretical and practical efforts at modelling the South within a North-South framework. The chapter is organized as follows. Section 6.2 sketches the broader theoretical framework within which most current North-South models fall. In section 6.3 some of the more important of the existing North-South models are examined. Based on this review, section 6.4 attempts to draw out a number of lessons for improving the modelling of the South. These lessons form the basis for the development of an alternative North-South model focusing specifically on Africa. This model is fully developed in section 6.5.

6.2 BASIC THEORETICAL MODELS

6.2.1 The Mundell-Fleming Model

In open economy macroeconomics, modelling the interaction of finance and trade between countries and the analysis of related policy issues are formulated around what is termed the Mundell-Fleming model, or, following Vos (1994), the Meade-Mundell-Fleming model. As Vos (1994) notes, the model's foundation was laid by Robert Mundell (1960, 1963, 1968) and J. Marcus Fleming (1962), who extended the pioneering work of Mead (1951), focusing on the Keynesian internalexternal balance-adjustment analysis (see Vos 1994: 65–66). Following Vos (1994) and Hallwood and MacDonald (1994), among others, the basic structure of the model may be described as follows.

Assuming two countries, North (N) and South (S), their output supply and demand (equations 1 and 2) and similar demand and supply equation for credit (equation 3) may be given by,

$$Y_{N} = D_{N}(Y_{N}, r) + M_{S}(Y_{S}, \phi) - M_{N}(Y_{N}, \phi^{-1})$$
[1]

$$Y_{s} = D_{s}(Y_{s}, r) + M_{s}(Y_{s}, \phi) - M_{s}(Y_{s}, \phi^{-1})$$
[2]

$$B_{N} + B_{S} + H = L_{N}(Y_{N}, r) + L_{S}(Y_{S}, r)$$
[3]

where: D_i is aggregate domestic demand consisting of private consumption, investment and public expenditure that varies positively with income Y and negatively with the real interest rate r; $\phi = P_S/P_N$ which is the terms of trade; B_i the supply of domestic credit; L_i the demand for credit; H the world stock of reserves treated as an outside asset; M_i import demand.

Based on the model's further assumption, that a perfect capital market exists, interest rates tend to equalize across countries. Import demand (M_i) and, implicitly, export supply will depend on income and terms of trade. The model's solution also depends on the exchange rate regime which is in operation. With a fixed exchange rate the money supply may be taken as endogenous, while it is best considered as exogenous under a flexible exchange rate regime. Based on these assumptions, Diagram 6.1 summarizes the equilibrium condition for the world economy.





In Diagram 6.1 equilibrium for the world economy is attained at point *E*, where both the money (*LM*) and the product (*IS*) markets are in equilibrium within the two countries. This ensures a unique level of interest rate. Once such a framework is developed, it is quite straightforward to analyse the impact of different policies relating, say, to the impact of financial flows from North to South (see Vos 1994: 68–70 for such an analysis). Indeed, this basic formulation has been expanded upon and modified by a number of authors with the aim of overcoming some of the limitations of this basic formulation. Thus, according to Vos' survey, the basic model has been extended to include private and public sector demand in Frenkel and Razin (1987) while imperfect substitution between domestic and foreign assets is treated in Kouri and Porter (1974) and Dornbusch (1980) (see Vos 1994: 68–69).

6.2.2 The Theoretical Basis of Recent North-South Models

Notwithstanding the importance of the Meade-Mundell-Fleming model, the relevance of such an IS/LM approach in analysing North-South interaction remains limited. This is because the behaviour of agents and the structure of the economy, which is assumed to be the same in the two countries, could, in fact, vary significantly between them. A good example of this relates to differential access to capital markets (see Vos 1994: 70). This suggests the desirability of using different models for the economies of the North and South, which, in turn, implies that the interaction between the two economies may depend on the underlying theory about the structure of these economies. This also represents an important theoretical justification for the proliferation of North-South models in general. Thus, following the works of Ocampo (1986), Vos (1989b) and Dutt (1990)² the following discussion will outline the major theoretical classification of North-South models. This will help to locate the model, which we develop in section 6.3, within the relevant literature.

Most existing North-South models may be classified under three theoretical strands, which may be formulated by assuming a common Lewis-type economy in the South and combining this with either a neoclassical, Kaleckian or a Neo-Marxian North. Following the approach of Dutt (1990), I will first outline the incomplete general framework he has developed. The different theoretical approaches will then be defined according to their assumption of 'model closure' for the general framework. This thematic classification is also helpful in understanding the theoretical underpinnings of the recent North-South models, which are examined in section 6.3 and which will inform the version of the model developed in this book.

6.2.2.1 The general framework

In the general framework, the two economies, North (n) and South (s), are assumed to produce a single good N and S, respectively. Both use a Leontief technology, incorporating labour (L) and capital (K) as factors of production. The S good is assumed only to be a consumption good, while the N good is both a consumption and investment good, which can be used in both regions. These assumptions imply the following two quantity and price equations, given as:

$$X_{n} = c_{n}^{n}L_{n} + c_{n}^{s}L_{s} + g^{n}K_{n} + g^{s}K_{s}$$
^[1]

$$X_s = c_s^n L_n + c_s^s L_s$$
^[2]

$$P_n = W_n a_0^n + r_n P_n \left(\frac{K_n}{X_n}\right)$$
^[3]

$$P_s = W_s a_0^s + r_s P_n \left(\frac{K_s}{X_s}\right)$$
^[4]

with
$$\frac{K_i}{X_i} \ge a_1^i$$
, $i = n, s$ [5]

where a_1 is the technologically fixed capital-output ratio, c_j^i denotes the consumption of good *j* in region *i* per worker employed in region *i*, a_0 is the employment-output ratio, *g* is the rate of growth of capital, *r* the rate of profit, *W* is the money wage (rate) and *P* prices (measured in terms of a common currency with exchange rate fixed at 1).

It is further assumed that only profit earners save a constant fraction, s_n , of their profit. The total consumption expenditure of wage and profit earners is also assumed to be split between N and S goods, with a constant fraction, a, being spent on N goods. Workers in the South are assumed not to save and to consume only Southern goods, while Southern profit-earners are assumed to save a constant fraction, s_s , and spend a constant fraction, b, of their consumption expenditure on N good, the rest being spent on the S good. Labour and capital are assumed to be internationally immobile. These assumptions imply,

$$c_n^n L_n P_n = a \left[W_n L_n + (1 - s_n) r_n P_n K_n \right]$$
^[6]

$$c_s^n L_n P_s = (1-a)[W_n L_n + (1-s_n)r_n P_n K_n]$$
^[7]

$$c_{s}^{s}L_{s}P_{s} = W_{s}L_{s} + (1 - s_{s})(1 - b)r_{s}P_{n}K_{s}$$
[8]

$$c_s^s L_s P_n = (1 - s_s) b r_s P_n K_s$$
[9]

Substitution of equations 6 to 9 into equations 1 to 4 yields,

$$1 = a \left[\left(\frac{W_n}{P_n} \right) a_0^n + (1 - s_n) r_n \left(\frac{K_n}{X_n} \right) \right] +$$

$$\left[(1 - s_s) br_s + g^s \right] \left(\frac{K_n}{X_n} \right) k^{-1} + g^n \left(\frac{K_n}{X_n} \right)$$

$$1 = (1 - a) \left[\left(\frac{W_n}{P_n} \right) a_0^n \left(\frac{X_n}{K_n} \right) \left(\frac{K_s}{X_s} \right) k \pi^{-1} + (1 - s_n) r_n \left(\frac{K_s}{X_s} \right) k \pi^{-1} \right] +$$

$$\left(\frac{W_s}{P_s} \right) a_0^s + (1 - s_s)(1 - b) r_s \left(\frac{K_s}{X_s} \right)$$

$$1 = \left(\frac{W_n}{P_n} \right) a_0^n + r_n \left(\frac{K_n}{X_n} \right)$$

$$1 = \left(\frac{W_s}{P_s} \right) a_0^s + r_s \left(\frac{K_s}{X_s} \right) \pi^{-1}$$

$$[13]$$

where: $k = \frac{K_n}{K_s}$ and $\pi = \frac{P_s}{P_n}$

Assuming away capital flows, balance of payment requires balance of trade, which implies

$$(1-a)[W_nL_n + (1-s_n)r_nP_nK_n] = g^s P_nK_s + (1-s_s)br_sP_nK_s \qquad [14]$$

which, in turn, implies,

$$(1-a)\left[\left(\frac{W_n}{P_n}\right)a_0^n k + (1-s_n)r_n\left(\frac{K_n}{K_s}\right)\right] = g^s + (1-s_s)br_s \qquad [15]$$

Substitution of equations 15 and 3 into equation 2 implies³

$$s_n r_n = g^n \tag{16}$$

$$s_s r_s = g^s$$
^[17]

Equations 16 and 17 show that, with balanced trade, total income equals total expenditure, in each region, so that saving equals investment. The above framework can be represented by five independent equations, numbered 10, 12, 13, 16 and 17, comprising ten variables, W_n/P_n , W_s/P_s , r_n , r_s , g^n , $g^s K_n/X_n$, K_s/X_s , k and π and is clearly indeterminate. (In other words, the number of equations does not equal the number of variables.) Focusing on the long-run issues we may assume a given stock of capital in the short run and hence, k is assumed fixed. The condition for the long-run steady-state, equilibrium, where we will treat k as a variable to be determined, entails that K_n and K_s must grow at the same rate. Hence

$$g^n = g^s \tag{18}$$

Having substituted equation 18, which implies equations 12, 16 and 17 into 10, we arrive at equation 19. However, four more equations are required, in order to close the model. Sections 6.2.2.2 to 6.2.2.4. give us alternative closure rules, which allow us to arrive at a solution to the following general model:

$$k = \left[(1 - s_s) \left(\frac{b}{s_s} \right) + 1 \right] \left\{ (1 - a) \left[\left(\frac{1}{g} \left(\frac{K_n}{X_n} \right) \right) - 1 \right] \right\}^{-1}$$
[19]

where g is the common growth rate of capital stock.

6.2.2.2 The Solow-Lewis model

In this model the North is assumed to be a neoclassical economy, with perfect competition prevailing within the goods market. Full employment of labour and capital is assumed at all times, with growth determined by the natural rate, *n*. The production of the Southern good is constrained by the extent of capital accumulation. Assuming flexible and unlimited labour supply at a given real wage à la Lewis (1954), investment from profits will increase output in the long run. The fixed wage, however, may also be explained by a standard neo-Marxian argument. Thus, the biologically required, socially necessary labour time, taken in conjunction with Marx's description of moral and historical considerations implies a certain wage level below which workers will not work. At this wage level, these workers will join the reserve labour force (see Dutt 1990: 18). These assumptions imply,

$$\frac{K_n}{X_n} = a_1^n \tag{20}$$

$$\frac{K_s}{X_s} = a_1^s \tag{21}$$

$$g^n = n$$
 [22]

$$\frac{W_s}{P_s} = V_s \tag{23}$$

where V_s is the fixed real wage. Substitution of equations 21 and 23 into 13 implies,

$$\pi = V_s a_0^s + r_s a_1^s \tag{24}$$

which yields a relation between r_s and the terms of trade, π , shown by OT in panel (b) of Diagram 6.2. Equation 17 yields another relation between g^s and r_s , shown by OS in panel (d) while equation 16 gives the relation between g_n and r_n shown by ON in (c). Substitution of equation 20 into 12 yields,



Diagram 6.2 The Solow-Lewis model

$$1 = \frac{W_n}{P_n} a_0^n + r_n a_1^n$$
 [25]

which gives the Northern wage-profit frontier shown as AB in panel (a). From equation 22 the level of g^n is fixed at rate *n*, and panel (c) determines (* indicates equilibrium values) g^{n*} and r_n* , and r_s* , $g_s*\pi^*$ and $(W_n/P_n)^*$ are determined by diagram 6.2. K^* may be determined by substituting equations 20 and 22 into equation 10, in order to arrive at

$$k^* = \left[(1 - s_s) \left(\frac{b}{s_s} \right) \right] \left\{ (1 - a) \left[\left(\frac{1}{a_1^n n} \right) - 1 \right] \right\}^{-1}$$
[26]

 V_n^* (real wage in North) which equals $(W_n/P_n)(P_n/P_s)^{1-a}$, where *a* (the weight, given by $0 \le a \le 1$) is also determined since $(W_n/P_n)^*$ and π^* are determined. This will complete the model solution.

This model may be used to explore the effects of changes in the parameters of the model on economic interaction between the two regions. One notes, for instance, that technological change (understood as lower a_0^s and a_1^s) in the South results in a deterioration in terms of trade of the South. Conversely, within this model, Northern growth is found to improve the terms of trade of the South. Other similar parameter changes and their impact are discussed in detail in Dutt (see Dutt 1990: 163).

Models developed by Findlay (1980, 1981) and Alogoskoufis and Varangis (1992) may be broadly classified under this category. The World Bank's Global Economic model, Bank-GEM (Peterson et al. 1991, Pedersen 1994), may also be taken as falling under this category. This is because the model is based on the NIESR (National Institute of Economic and Social Research) and the London Business School global economic model, both of which incorporate similar features to the Solow-Lewis model, in the long run. However, the London Business School model uses an IS/LM framework to solve for its major endogenous variable, including trade volumes (see Allen & Vines 1994).⁴



237

6.2.2.3 The Kalecki-Lewis model

As with the Solow-Lewis model, within this framework the South is assumed to have a Lewis-type economy. Hence equation 21 is assumed. Along Kaleckian, or following Dutt, Kalecki-Steindl lines, the North is assumed as characterized by excess capacity. This implies that equation 20 should *not* be assumed. Instead, a strict inequality, given in equation 5, is applied for i = n. Thus, three equations are required in order to close the model. The first comes from the Lewisian assumption, which may be expressed by equation 24. The second is the Kalecki-Steindl desired accumulation function given by

$$g^{n} = g^{n} \left(r_{n} \frac{X_{n}}{K_{n}} \right)$$
[27]

with both partials being positive. The third is a Kaleckian mark-up pricing equation, given as

$$P_n = W_n a_0^n (1+z)$$
 [28]

where z is the fixed mark-up rate.

Substitution of equation 27 in equation 12 and solving for r_n gives

$$r_n = \left[\frac{z}{(1+z)}\right] \frac{X_n}{K_n}$$
[29]

which, when substituted into equation 27 implies g^n is a rising function of r_n , which we assumed to be concave and given by GG in panel (b) of Diagram 6.3. Its intersection with ON, given by equation 16, determines g^{n*} and r_n^* . As before, in panel (d), AB is the wage-profit frontier. However, because of excess capacity, the economy must lie inside this frontier, such as at point C, where $(W_n/P_n)^*$ is obtained using equation 28. Note also that at point C, the inequality of equation 5 (for i = n) is satisfied. OS in panel (a) shows equation 17 and solves for r_s^* and g_s^* . In panel (c) OT represents equation 24 and solves for π^* . Substituting r_n^* into equation 29 solves for $(X_n/K_n)^*$. Substituting this and the common rate of growth into equation 19 solves for k^* .

Analysis of parameter changes using this model reveals a number of interesting results. For instance, a rise in mark-up rate, which pushes the GG curve downward, is associated with a deterioration in terms of trade of the South. A similar deterioration in terms of trade may result from productivity growth associated with technological change in the South, while growth in the North may improve the terms of trade of the South. (See also Dutt 1990: 169 for a similar analysis, in relation to other parameter changes.)

Among the existing North-South models, some of which are examined in section 6.3 below, Taylor (1981, 1983, 1991), Darity and Fitz-Gerald (1982), IMF's MULTIMOD (Masson et al. 1990), Vos (1994), Sarkar (1994, 1996) may all be considered as falling under this broad category.

6.2.2.4 The Neo-Marxian North and a Lewis South model

Problems exist with the labelling of this type of model. The widely cited model developed by Vines (1984), as well as the published version, developed by Molana and Vines (1989), have both wrongly been termed '... along Kaldorian lines'. Molana and Vines' model essentially assumes a Lewis-type economy within the South.⁵ The North is also assumed to have surplus labour with an exogenous real wage and the level of output taken as supply-determined rather than Keynesian (Molana & Vines 1989: 444). Following Dutt (1990), the economy of such a North might better be termed Neo-Marxian than Lewisian. Certainly, it is wrong to label such models as Kaldorian, since Kaldor explicitly assumed that in the North, 'prices are administered' using mark-up, that there is excess capacity and hence, that output is demand determined through 'a stock adjustment mechanism' (Kaldor 1976: 705). After correcting for this oversight, Sarkar (1994, 1996) developed various North-South models. However, Sarkar's models are best understood as broadly falling under a Kalecki-Lewis classification, as described in section 6.2.2.3, above.

In terms of our equations, full capacity utilization in both North and South is assumed. This implies that equations 20 and 21 hold. The South is assumed to have a given real wage so that equation 23 also holds. Similar assumptions for the North implies,

$$\frac{W_n}{P_n} = V_n \left(\frac{P_s}{P_n}\right)^{1-a}$$
[30]

for a given V_n , which is fixed, say, for Neo-Marxian reasons.





Inserting equation 21 into equation 13 implies 24, as in the first model. This gives the OT line in panel (b) of Diagram 6.4. From equations 16 through 18 we get,

$$r_n = \left(\frac{S_s}{S_n}\right) r_s \tag{31}$$

which gives OR in panel (d). Substitution of 20 into 12 yields equation 25, which gives line AB in panel (c). In panel (a), ON shows equation 30 for a given V_n . Curve CD in panel (b) is derived from OR, AB and ON in the diagram and plots the locus of r_s and π , which satisfy equations 25, 30 and 31. The intersection of OT and CD in panel (b) determines r_n^* and π^* . r_s^* and $(W_n/P_n)^*$ can be read from the rest of the diagram. Substitution of the rates of profit into equations 16 or 17 determines g^* , and substitution of this and equation 20 into 19 implies,

$$k^* = \left[(1 - s_s) \left(\frac{b}{s_s} \right) \right] \left\{ (1 - a) \left[\left(\frac{1}{g^* a_1^n} \right) - 1 \right] \right\}^{-1}$$

Analysis of various parameter changes using this model is undertaken in Dutt (see Dutt 1990: 165). It is interesting to note that within this model, technical progress in the South (represented by lower a_0^s and a_1^s) implies a deterioration in its terms of trade, while a similar improvement in the North improves the terms of trade of the South. As noted above, the Vines (1984) and Molana and Vines (1989) models may broadly be taken as falling under this general category.

6.2.2.5 Some points about capital flows

The models outlined above are based on general macro and trade linkages. The emphasis on such a framework is justifiable for primary commodity-exporting countries like those in Africa, since such trade linkages are central in understanding macroeconomic performance. As demonstrated by Vos, '... capital flows do not alter the pattern of asymmetric adjustment' within such an economy (Vos 1989b: 130–38).

Following this line of argument, Vos (1989b) and Dutt (1990) have extended the above framework for the analysis of capital flows from North to South. Both assume that part of the capital stock in the South is owned by the North, and that the North earns and remits profit from that stock. Within this framework, a further assumption is that capital inflows to the South can be thought as a function of North-South profit differential. This implies, among other things, either the complete ownership of the South's capital by the North ('crowding-out') (Vos 1989b: 132, Dutt 1990: 177–81), or no involvement at all by the North (Dutt 1990: 177–81). The complete crowding-out of the South by the North is further accentuated if the saving elasticity of the North is larger than that of the South (see Dutt 1990: 177–81).

The main reason underlying this crowding-out is that, given the stylized fact of lower income elasticity of primary goods and a relatively high savings rate in the North, capital exports⁶ entail a faster growth in the supply of commodities than in Northern demand for these. This, in turn, leads to deterioration in the terms of trade of the South (Vos 1989b: 138). This result stands in stark contrast to the Ricardian H-O-S and the 'new trade theories' model prediction of gains from trade, discussed in Chapter 2. Rather, the result accords more closely with the predictions of non-orthodox models. Moreover, the fact that most African countries are becoming increasingly dependent on aid with comparatively limited access to international banks underlines the importance of theories of official capital flows and the credit-rationing theories discussed in Chapter 2. Vos (1989b, 1994) has formulated the credit-rationing argument in an international banks context. These theoretical avenues are briefly discussed below and are integrated into the model developed in this chapter.

To sum up, this section has clearly demonstrated that the view of the economy, or the stylized facts adopted by the researcher, implies different closure mechanisms and hence, more often than not, different results. Such a thematic classification helps to understand the existing models, which are examined in section 6.3 below. Thus, the following section will examine the practice of global modelling, focusing primarily on how the South is modelled, with the aim of arriving at lessons relevant to the modelling exercise undertaken in this chapter.

6.3 STANDARD NORTH-SOUTH LINKAGES

The standard linkages in North-South models may broadly be classified as *financial* and *trade*. These linkages serve as propagating mechanisms for the impact of macro-policy changes across the two regions, and usually run from North to South. The discussion of these linkages will proceed at two levels. Thus, financial linkages will be examined in section 6.3.1, and trade linkages in section 6.3.2.

6.3.1 Financial Linkages

Most North-South models take account of financial flows, linking North with South. However, the emphasis that is placed on finance, relative to trade linkages varies from model to model. For instance the models of, *inter alia*, Muscatelli and Vines (1991), Murshed (1990) and Vos (1994) place more emphasis on financial linkages compared with other North-South models such as MSG2 (McKibbin & Sachs 1991), the IMF's MULTIMOD (Masson et al. 1990) and Project LINK (Ball 1973). Hence, in this section I will examine how these financial linkages are established within a number of existing models.

In the UNCTAD model (see Ball 1973), which is also used by Project LINK as the model of the South, net factor payments are related to external deficits and export revenue. Important financial variables such as exchange rate and net foreign capital inflows are assumed to be determined exogenously. This exogenously given external finance is specified as affecting the level of reserves, which, in turn, affects the level of imports, and hence, investment. This specification allows for the possibility of depicting the phenomenon of 'import compression.' This is important, since import compression is reported to be widespread, not only in the relatively more advanced developing countries, such as Brazil (Fritsch 1988), but also throughout the less developed countries of Africa (Ndulu 1986, 1991 and Rattsø 1992b).

Within the MSG2 model (McKibbin & Sachs 1991), all new loans to the least developed countries (LDCs) are assumed to remain in historical proportions. All other capital inflows are derived from the consistency that requires world balance of payments to sum to zero. For the US, Japan, Germany, the rest of the OECD and OPEC, the current account is determined under the assumption that domestic agents have free, unrestricted access to international borrowing and lending at international rates. For LDCs the scale of borrowing is set exogenously, by consideration of country-specific risk. In modelling commercial flows to LDCs, the MSG2 model emphasizes the supply-determined nature of such flows, and hence the importance of credit rationing to the South.

In Beenstock's (1988) North-South econometric model, the financial linkage is established following the 'loanable-fund' doctrine. The

supply of such funds from OPEC and North is equated to the stock of LDC debt, as well as the North's public debt and capital requirement. This stock equilibrium, coupled with capital market specification for the North, determines the interest rate. Interest rates, in turn, affect the debt service of the South, which is included in the balance of payment equation for the South. The other financial linkage is aid, which is included in the balance of payment equation of the South, as being exogenously determined by the North. Indebtedness of the South is approached as follows. LDCs have a desired debt position in stock terms. In the short run the desired may differ from the actual, reflecting adjustment costs. Reserve movements are determined residually through the balance of payment identity. In the long run, the South could have a desired reserve that might be attained by controlling imports, stimulating exports or altering the exchange rate. Although the total supply of funds and its effect on interest rate is based on the classical 'loanablefund doctrine', the modelling of capital flows to South does not follow a specific financial theory. Rather, it is derived from a consistency rule, which, in the case of aid, is exogenously given. The Beenstock model also adheres to the assumption that capital flows to the South are largely demand driven. Hence, the model neglects supply conditions in the North which are emphasized in other models.

In the OECD model INTERLINK (OECD 1988), financial linkages are established through the use of capital flows. The capital flow to each country is a function of short-term interest rate differentials, expected rates of currency depreciation/appreciation, and domestic and foreign current balances. Foreign interest and exchange rates are taken as exogenous within the single country model. In the linked model, changes in interest rate, exchange rate and money supply of a country are specified as affecting macro variables in other countries, through their influence on capital flows. Hence, the effects of these variables may be felt throughout the whole system. These equations are subject to strong cross-country parameter restrictions, which, together with appropriate weighting matrices, ensure global consistency of net capital flows. Basically, the financial linkages specified in the INTERLINK model are based on portfolio choice theory. Indeed, the model's particular strength lies in the use of appropriately valued asset and liability stocks in the determination of capital flows.

The World Bank's Global Economic model, Bank-GEM (Peterson et al. 1991 and Pedersen 1994), focuses on three types of external cap-

ital flows. These are FDI, debt-creating flows and foreign-asset holdings of the private sector. FDI is set as being determined by the same factors as domestic investment: namely, foreign real rate of interest as an indicator of portfolio choice; domestic GDP as an indicator of market size or policy success; and the real exchange rate, which affects profitability. In modelling debt-creating flows, individual LDCs are assumed to receive a fixed share of official concessional funds, the exact size of which will depend on growth of income within OECD countries. Disbursement of official non-concessional loans will depend on GDP growth of the borrower (as a proxy for creditworthiness), the current account balance (representing financing requirements), and the previous period's net disbursements (as an indicator of borrower behaviour). Public commercial borrowing is projected by imposing the growth rate of official non-concessional borrowing, assuming that this falls within the general borrowing strategy of the LDC in question.

Finally, commercial borrowing of the private sector constitutes the residual component of the balance of payments. Portfolio returns and risks associated with portfolio holding, as well as risk differential by domestic and foreign investor, and hence, two-way flow approaches may be used to define the foreign assets of the private sector. Hence, these assets, sometimes referred to as 'capital flight', may be defined as a function of: (1) domestic income growth (which may be associated with lack of opportunities at home or policy credibility); (2) domestic and foreign inflation; and (3) the lagged ratio of foreign asset holdings to total wealth of the private sector within a 'normal' portfolio. The short-term capital flows of the private sector are set as a function of price and volume of imports (in order to show trade credits, which might finance imports), while those of the public sector are defined as the official foreign reserve flows of the balance of payment. The change in reserves is modelled as an adjustment to a desired stock level, which will depend on the dollar value of the exchange rate, as well as the price and volume of imports. Finally, the percentage change in nominal exchange rate is related to the percentage change in domestic and foreign inflation differentials. In general, the Bank-GEM financial modelling is informed by a Mundell-Fleming framework, since financial markets are assumed to clear for world interest rates in an integrated world capital market. Within this framework, private capital inflows are seen as being determined along the lines of a portfolio theory of finance (see Alemayehu 1994 for details about Bank-GEM).

Muscatelli and Vines' (1991) study also focuses on financial linkages. They begin their North-South model by outlining a number of possible causes of the debt crisis. Firstly, domestic policies pursued by the South; secondly, global macroeconomic shocks which result from changes in interest rates, the fiscal and monetary policy adopted by the North, as well as fluctuations in commodity and oil prices; thirdly, creditors' failure to assess risks; and fourthly, trade restrictions applied in the South. The debt crisis has resulted in a variety of expenditure switching and reducing effects. Of these, the former effect was found to have led to inflation, while the latter to import compression and fiscal adjustment, characterized by reduced public spending and high interest rates in the short run. The expenditure reducing effect was also found to have had supply-side effects on long-run growth and developmental potential, with implied demand-side impacts.

In order to locate the above problems in the context of North-South interaction, Muscatelli and Vines (1991) identify the macroeconomic linkages between the two regions. Hence, in the modelling approach adopted, the South is assumed to depend on North for its exports, import of capital goods, and finance. Conversely, the North is assumed to depend on the South in order to meet its demand for raw materials. Further, it is assumed, through the operation of international capital markets, that the economic behaviour of the South can influence that of the North, and vice versa. Thus, the perceived probability of default is likely to adversely affect capital inflows to the South, with a rise in interest rates also having the effect of depressing investment in the South. This, in turn, will affect the short-run demand for Northern goods, and may also have long-run supply effects, as the supply of commodities declines. Moreover, a rise in interest rate will have a negative effect on the debt service costs of the South. Recognition of these linkages has led various authors to focus their attention on three specific areas, in seeking a solution to the debt crisis. Firstly, domestic policies of the South; secondly, macroeconomic policies pursued by OECD countries; and thirdly, measures specifically aimed at reducing the level of outstanding debt.

Murshed (1990) also seeks to explain the debt crisis of the 1980s as representing a recent manifestation of ongoing North-South macroeconomic interactions. For him, the crisis as a whole, its emergence and possible solutions are all rooted in this interaction. Murshed notes a number of stylized features of North-South interaction in the 1980s.

Two such features are a contractionary monetary policy in the North and the intensification of a protectionist tendency toward Southern goods. Such policies have resulted in the collapse of commodity prices and the evolution of indebtedness into a fully blown debt crisis. Adjustment programs designed to tackle debt problems have resulted in entitlement losses and low growth rates. In his model, the North is depicted as functioning in a Keynesian fashion, with income assumed as being demand determined. In the goods market, the financial linkage is established by making Northern absorption partly a function of interest payments on Southern debt. Within this model, disposable income of the North is also set as being boosted by commercial tariffs levied on Southern exports. Capital flows from North to South are set to be a function of interest rates (as in a Mundellian or MacDougall approach), while the South's debt is related to balance of payments. Finally, the balance of payments itself is assumed to depend on macroeconomic conditions in the North.⁷ Having analysed the overall model, Murshed concludes that 'the debt crisis as a whole is best viewed as a painful spillover of Northern macroeconomic policy onto the South'.⁸ In general, Murshed's model focuses quite sharply on financial interactions relating to macro policy and macroeconomic conditions in the North and their repercussions on the South.

In MULTIMOD (Masson et al. 1990), a model developed and maintained by the IMF, the financial linkages of the finance-constrained developing country is formulated based on its capacity to service external debt. This capacity is measured by comparing the interest on debt to export ratios of LDCs, to an expected or targeted value for a similar ratio, which is exogenously given. In this specification, exchange rate is allowed to have an influence on capital flows to the South. Economic activities in the North will also affect the numerator and denominator, and hence, the overall value of this ratio. A sort of 'import compression' mechanism is predicted when the South's ratio passes the targeted threshold level. The theoretical basis of this approach is a rudimentary credit rationing mechanism based on solvency criteria. Finally, since exports are used in the computation of the relevant ratios, it is reasonable to conclude that MULTIMOD's approach may be used, not only to indicate insolvency, but also short-term liquidity problems.

In Vos' (1994) model, which is termed STAC, financial linkage is established by letting North have unrestricted access to capital markets, while the access of the South is rationed on the basis of an imperfect creditworthiness assessment. This assessment is based on perceived payment capacity, which is calculated by relating interest payment to exports, in a similar fashion to that undertaken in MULTIMOD. Moreover, financial flows from the North are linked to public and private investment by way of the deficit. Imports are determined as residual, thus capturing the 'import compression' phenomenon. Other official flows are taken as being exogenously determined. In Vos (1989b, 1994), as well as Darity and FitzGerald (1982), the theory of credit rationing by oligopolistically functioning international banks is applied in modelling the linkages between the regions. The basic argument of these approaches is that the behaviour of the suppliers, who are basically the international and commercial banks of the North, needs to be taken account of in the determination of such flows. Hence, in relation to lending to the South, the focus is mainly on credit rationing. In Taylor (1981), and Marquez and Pauly (1987), the South's dependence on capital inflows from the North is recognized and understood to result from technological or capital input dependency. The extent of this dependency, which is equal to the regions trade deficit, is assumed to be determined by political factors. Indeed, following Vos (1989b), Taylor (1991) has attempted to model these financial linkages. Finally, the models of Findlay (1980, 1981) and Molana and Vines (1989) do not consider capital flows at all.

To sum up, recent global models have attempted to incorporate different aspects of financial flows. However, the emphasis, the selection of variables and the specification adopted will vary depending on the theoretical approach chosen or the accounting framework adopted. Most existing models which focus on the North emphasize the Mundell-Fleming and portfolio theories of capital movements in modelling financial linkages. However, there are also a growing number of North-South models which are based on theories of credit rationing and oligopolistic banking behaviour. Indeed, the latter approach is quite plausible in explaining private bank flows to African countries, which are usually considered to be the worst risk. Thus, in modelling these flows to Africa a theoretical framework based on oligopolistic banking behaviour will be very important. Hence, this approach, together with some other specific features of African macroeconomics not considered in existing North-South models, will be taken into account in building the model set out in this chapter.

| North-South models | Uses interest rate from integrated capital market | Uses dis- aggregated capital flows |
|---|--|--|
| Beenstock 1988 | Yes | Limited |
| Darity & FitzGerald 1982 | No | Limited |
| INTERLINK, OECD, 1988 | Yes | No |
| MULTIMOD (IMF) Masson et al. 1990) | No | No |
| Murshed 1990 | Yes | No |
| Muscatelli & Vines 1991 | No | No |
| MSG2 (McKibbin & Sachs 1991) | No, but yes across N | No |
| STAC (Vos 1994) | No | Limited |
| Taylor 1981; Marquez & Pauly 1987 | No | No |
| Findlay 1980, 1981; Molana & Vines 198 | No | No |
| UNCTAD, Project LINK (1973) | No | No |
| Bank-GEM (Peterson et al. 1991, Pedersen 1994) | Yes | Yes |
| North-South models | Focus on debt, default and N macro policy effect on S | Focus on import compression |
| Beenstock 1988 | Yes | No |
| Darity & FitzGerald 1982 | No | No |
| INTERLINK, OECD, 1988 | No | No |
| MULTIMOD (IMF) Masson et al. 1990) | Limited | Yes |
| Murshed 1990 | Yes | No |
| Muscatelli & Vines 1991 | Yes | Yes |
| MSG2 (McKibbin & Sachs 1991) | No | No |
| STAC (Vos 1994) | Yes | Yes |
| Taylor 1981; Marquez & Pauly 1987 | No | No |
| Findlay 1980, 1981; Molana & Vines 1989 | No | No |
| UNCTAD, Project LINK (1973) | No | Yes |
| Bank-GEM (Peterson et al. 1991, Peder- sen 1994) | No | No |

Table 6.1 Summary of financial linkages

(Continued)

| North-South models | Comprise credit rationing behavior in N | Focus on supply side determination of flows (liquidity, risk) | Focus on the demand side determination of flows |
|---|---|---|--|
| Beenstock 1988 | No | No | No |
| Darity & FitzGerald 1982 | Yes | Yes | Yes |
| INTERLINK, OECD, 1988 | No | Limited | Limited |
| MULTIMOD (IMF) Masson et al. 1990) | Yes | Yes | Νο |
| Murshed 1990 | No | No | Yes |
| Muscatelli & Vines 1991 | No | Yes | Yes |
| MSG2 (McKibbin & Sachs 1991) | Yes | Yes | Yes, through consistency rule |
| STAC (Vos 1994) | Yes | Yes | Yes, through deficit financing |
| Taylor 1981; Marquez & Pauly 1987 | No | Yes, | Yes, through deficit financing |
| Findlay 1980, 1981; Molana & Vines 1989 | No | No | No |
| UNCTAD, Project LINK (1973) | No | No | No |
| Bank-GEM (Peterson et al. 1991, Pedersen 1994) | No | Yes, but limited | Yes |

Table 6.1 (continued)

6.3.2 Trade Linkages

At least two basic approaches have been used in modelling trade linkages between North and South: firstly, one which ensures *endogenous determination* of the relevant trade variables (i.e., exports, imports and their prices), and secondly, an approach based on trade or market *share matrices*. In relation to the first of these approaches, three important aspects may be outlined. The first of these relates to the nature of the market, with the North usually assumed to specialize in industrial goods production and the South in primary goods. The market for industrial goods clears for quantity, or, in some models, for price, while that of primary goods clears only for price. Secondly, it is noteworthy that some models move beyond this simple but illuminating classification, and disaggregate commodities into at least four to five commodity groups. Thirdly, a number of models focus explicitly on the impact of financial variables on the primary commodity market. The second approach, using trade share metrics, also has a number of variants. Thus, bilateral trade may be estimated directly, from a linearized Armington formulation (Hickman & Lau 1973), by using an estimation of total exports, by allowing variation in the share matrices (Klein & van Peterson 1973), or by emphasizing measures of competitiveness (Samuelson & Kurihara 1980).

A very common approach, used in most recent global economic models, is to establish the link between North and South through the specification of the supply of exports from one region, with the demand for imports in the other. This approach has been adopted by, among others, Krugman (1979a), Taylor (1981, 1983, 1991), Marquez and Pauly (1987), Masson et al. (1990), Vos (1994), Muscatelli and Vines (1991) and Moutos and Vines (1988). In none of these models, with the exception of MULTIMOD, are commodities disaggregated. Rather, usually an aggregate manufactured good is specified for the North, and an aggregate primary commodity for the South. However, as described below, the models differ as to how they specify their trade linkages, as well as the assumptions made about the nature of the goods market.

In the UNCTAD model, Project LINK, an attempt is made to model the South. However, the specification of the import/export equation, or trade linkage, for the South is carried in a very simple manner. Thus, exports are simply related to an index of world trade and a ratio of exports to world price, while imports are related to GDP, lagged real foreign inflows, deflated by import price, and the real exchange rate. The major trade linkage is established through market shares. The initial LINK exercise was to estimate the import function of each country, for a given level of world trade. Subsequently, world-level consistency is achieved by making the total world level of trade an adjusting variable. This is termed the Mini-Link. This approach assumes that the level of world trade is the only adjusting variable. An alternative approach, termed the Maxi-Link, is employed in the final version of the Project LINK model. The Maxi-Link is a modified Armington formulation, and is estimated by relating actual to estimated exports. The latter are estimated based on share matrices, estimated imports, a partial adjustment specification of such estimates in the preceding period, and a trend term (see Klein & van Peterson 1973). When constructing models the size of Project LINK and Bank-GEM, the use of a simplified structure is understandable. However, it should be pointed out that this simplicity might come at the expense of ensuring the use of theoretically and empirically sound adjustment mechanisms.

| North-South models | | Uses trade share matrices | Uses endogenous determination of trade variables |
|---|---|--|--|
| Alogoskoufis & Varangis 1992 | | No | Yes |
| Beenstock 1988 | | No | Yes |
| Darity & FitzGerald 1982 | | No | Yes |
| INTERLINK, OECD, 1988 | | Yes | No |
| MULTIMOD (IMF) (Masson et al, 1 | 1990) | No | Yes |
| Murshed 1990 | | No | Yes |
| Muscatelli & Vines 1991 | | No | Yes |
| MSG2 (McKibbin & Sachs 1991) | | No, | Yes |
| STAC (Vos 1994) | | No | Yes |
| Taplin 1973 | | Yes | No |
| Taylor 1981; Marquez & Pauly 198 | 37 | No | Yes |
| Findlay 1980, 1981; Molana & Vine | es 1989 | No | Yes |
| UNCTAD, Project LINK (1973) | | Yes | No |
| Bank-GEM 1991, 1994 | | Yes | No |
| North-South models | Directly lini macro variables t commodit market | ks Usa disag o gat y comn itie | In goods es market the gre- North is ed quantity nod- clearing while s South is price clearing |
| Alogoskoufis & Varangis 1992 | Yes | N | o No ^a |
| Beenstock 1988 | Yes | N | o No [⊳] |
| Darity & FitzGerald 1982 | No | N | o Yes |
| INTERLINK, OECD, 1988 | No | Ye | s No ^c |
| MULTIMOD (IMF) (Mas. et al '90) | No | Ye | s Yes |
| Murshed 1990 | Limited | N | o Yes |
| Muscatelli & Vines 1991 | Limited | N | |
| MSG2 (McK. & Sachs 1991) | No | N | |
| STAC (Vos 1994) | No | N | Yes |
| Taplin 19/3 | NO | Ye | S NO |
| Taylor 1981; Mar. & Paul 1987 | NO | N |) Yes |
| Find. 1980, '81; Mol. & V. 1989 | NO | N | D NO [™] |
| UNCIAD, Project LINK (1973) Ropk CEM 1001 1004 | NO | Ye | s No ^b |
| Dalik-GEW 1991, 1994 | INO | re | 5 INU |

Table 6.2 Summary of trade linkages

^a interest also clears ^b both are price clearing ^c price clearing in the North is limited

The Bank-GEM model followed two approaches in dealing with trade linkages. The first was to use a modified Armington-type specification. The modification relates to the inclusion of price pass-through effects, a time trend⁹ and the use of an error correction method within the estimation. This approach is then used in the specification of trade in manufactures (SITC 5-9). For trade in food and beverage (SITC 0+1), raw materials (SITC 2+4) and energy (SITC 3), a similar approach to that used in Project LINK is proposed. However, it is not clear which version of the LINK's linkage mechanism is to be used.¹⁰ Neither does the Bank-GEM model clearly explain the market structure used in its commodity market. For manufactured goods it seems that a middle path between monopoly and perfect competition is chosen. However, there is no guarantee that the price specification used will yield the intended results (see Peterson et al. 1991: 18). Indeed, within such a set-up, price is more likely to be flexible than fixed.

A global trade model, very much in the tradition of the export share matrix, is that of Taplin's (1973) model of world trade.¹¹ Taplin's model (discussed at length in Chapter 2) is based on the 'Expanded World Trade Model (EWTM)', developed by the Research Department of the IMF for the short-term forecasting of trade flows and analysis of economic policies. The version discussed here treats the world as divided into 27 countries and regions, comprising each of the 25 developed countries, the CEMA (the former 'socialist' countries) and the rest of the world (RW). The RW encompasses the developing countries in total. The level of imports for each developed country is determined by an import function. Economic activity (measured as autonomous spending) and relative prices are both represented as explanatory factors within the import function. However, due to a shortage of relevant information, economic activity and exports are used to determine the imports from the CEMA region. The imports of the rest of the world are set as a function of current and past foreign exchange receipts. The level of exports of each country and region are obtained by distributing the forecasted imports by an export share matrix. In common with Project LINK, export share matrices lie at the heart of the system. Thus, using Armington's approach, the share matrices are estimated by regressing the change in market shares onto a proportionate change in price ratio.

Another widely cited global model is MSG2, in which the linkage between North and South is set in a vertically integrated fashion. The North produces consumption and investment goods, while the South supplies primary inputs, including oil, for use in the North's production process. Thus, the demand for goods from the South is a derived demand for Northern goods. Output of the North is specified, in a Cobb-Douglas form, as a function of value added and primary inputs. Value added is further specified, also in Cobb-Douglas form, using capital and labour as inputs, while primary inputs are defined as a Cobb-Douglas function of oil and non-oil primary inputs. Oil is then specified as a Cobb-Douglas function of domestic production and imports from OPEC. For goods from OPEC and the LDCs, a single world uniform price is assumed. Prices of commodities from the South and OPEC are defined by a mark-up over a basket of OECD goods. By making this mark-up, a positive function of the demand for OPEC and LDC exports - in effect, the price of OPEC and the South's commodities - are assumed to be demand determined. However, the downside of this assumption is that it overlooks the important distinction that the oil market clears for output and the primary commodity market for price, which is taken as a stylized fact in most recent North-South models. However, a useful development incorporated into the MSG2 model is the use of an integrated or nested production structure. Although the MSG2 specified such relations at an aggregate level, the method may also be applied in relating sectoral supply to sectoral demand, within the two regions or countries. The specification of such sectoral linkages would allow for the examination of the sectoral implications of macro linkages in general, and of trade linkages in particular.

Based on Vernon's (1966) theory of 'product cycle', Krugman (1979a) has formulated trade patterns that emerge from technological change within a North-South framework. He assumes a world of two countries, consisting of an innovating North and a non-innovating South. Innovation takes the form of developing new products, which could, after a lag, be produced in the South. The lag in the adoption of new technology by the South gives rise to trade. Interestingly, Krugman concludes that the

North always exports new products and imports old products... Each good is at first produced in and exported by the North; then when technology becomes available to South, the industry moves to lower wage country. Case studies in such a world reveal a Vernon-type product cycle. (Krugman 1979a: 260) Although his theory cannot usefully be applied to the South in general, Krugman's general point remains valid for particular countries, particularly in East Asia. However, its relevance to low-income, institutionally-weak African countries remains very limited.

One of the trade linkages used in the North-South model developed by Alogoskoufis and Varangis (1992) relates to an attempt to link the fiscal condition of the North with the primary commodity price of the South. This is a very small model, which is analytically similar in structure to that of the Mundell-Fleming model. In the section of the model dealing with the North, for a given set of supply and demand parameters, macro equilibrium depends on the relative price of commodities to manufactures, and on the real interest rate. A similar equilibrium structure is also maintained in the section dealing with the South. Finally, the two models are combined and solved for the equilibrium values of real interest rate and relative prices. The model conveys an important point, that the relationship between world interest rate and primary commodity prices could be negative or positive, depending on the origin of shocks. Thus, if demand and supply shocks originate, say, from the North, the relationship between real interest rate and relative prices is negative. This result indicates the presence of a strong linkage between the fiscal deficit in the North and price of primary commodities. The model developed by Beenstock (1988) also emphasizes such commodity and financial market linkages.

In the IMF's MULTIMOD (Masson et al. 1990), trade in goods is disaggregated into three basic commodity groups: oil, primary commodities and manufactures. Trade in oil is assumed to take place at a unique world price, which is determined exogenously. For the industrialized countries a domestic demand equation determines oil consumption, with oil imports determined as a residual, once domestic sources have been tapped. For the two groups of LDCs specified in the model, namely capital importing and capital exporting, exports of oil are taken as endogenous and world demand is equated with world supply. Any increase in demand is shared between the two regions in fixed proportions, with production passively responding to demand at a given price. Primary commodities produced by the South are assumed to have perfectly flexible prices, which will ensure market clearing. Relative price changes in favour of primary products, or profitability, are assumed to induce a shift of resources into this sector, with associated positive repercussions. Manufactured products are assumed to be produced by all countries and are mark-up priced. With the exception of high-income oil earners, the import and export of manufactures is taken as endogenous to the model.

The trade linkage is established with the North's export of manufactures, through an export equation that comprises foreign absorption and real effective exchange rate as explanatory factors. Another trade linkage is established with the commodity imports from the South using an import function, which contains the variables for Northern GDP and real exchange rate as arguments. Owing to the endogenous determination of the main linkages,¹² the model is characterized as a dynamic version of the Mundell-Fleming model. An important feature of MULTIMOD is its market structure. Specifically, manufactured goods are (fix-price) quantity clearing, while primary commodities are (flexiprice) price clearing.

The North-South models of Taylor (1981, 1983, 1991), Darity and FitzGerald (1982), Marquez and Pauly (1987), Vos (1994) and Sarkar (1994, 1996), among others, are also based on a similar market structure, with the Northern goods market clearing for output, and commodity markets for price. These models fall under the same theoretical grouping as the Kalecki-Lewis models discussed in the previous section. In contrast, the trade linkages within the models of Findlay (1980, 1981), as well as Molana and Vines (1989), describe a perfectly competitive market within both regions. These markets accommodate goods produced using a neoclassical production function.¹³ Thus, in these models, the demand for imports within each region will depend on relative price and real income.

In the OECD's INTERLINK model, volume of imports and price of exports are determined endogenously, with domestic demand, costs and international prices used as explanatory variables. However, with the exception of energy suppliers, non-OECD countries are taken as price takers, with their export prices assumed to follow that of competitors. Food and raw materials are taken as a function of demand within the OECD. Such single-country estimates are passed to the trade linkage model where a consistent export volume (by way of share equations)¹⁴ and import price estimates are determined. This is undertaken by allocating country-based estimates of global import demand among individual exporting countries, which, in turn allows for the determination of import prices for each country. The main determinant of the export share is the export market growth elasticity which, itself, depends on

past trends, competitiveness and the commodity export composition of a country. These export and import prices are, in turn, passed to the individual country models, as revised inputs. The process is then iterated, in order to arrive at an internationally consistent model solution. Global consistency between import and export volume is enforced by combining the structure of the system, as described above, with parameter restrictions across a set of export volume equations. In general, this trade linkage is similar to the one used in Project LINK and proposed for use by Bank-GEM. However, while in LINK, consistency is achieved by estimating exports, allowing for possible factors to influence the share-based estimation, parameter restriction plays a significant role in INTERLINK.

6.4 SUMMING UP

So far, in this book, I have examined both the theory and practice of North-South modelling. The theoretical section of this discussion has shown that the assumed stylized facts, or view of the economy, will determine the structure of the North-South models which are developed. As a result, the implications of particular shocks or policies could vary across different types of model.

An interesting conclusion may be arrived at when these theoretical models are contrasted with the stylized facts in relation to Africa. As described in Chapter 1, African countries are price takers within the international commodity market, with the market for their commodities characterized by price clearing. An 'unlimited' labour supply is also found to represent a plausible hypothesis for most of these countries. Moreover, African nations will tend to trade with (Northern) countries, which have excess capacity, and which are dominated by big firms with price-making capacity. In fact, as demonstrated by Yeats (1991), for some commodities, such as metals, African countries will tend to pay an average premium of 23 per cent above the unit value for other developing countries (Yeats 1991: 201). The discussion presented in Chapter 1 also shows the extent to which African countries are dependent on aid. These countries are also likely to be the worst risk for private lenders, and commercial banks in particular. These stylized facts underscore the relevance of the Kalecki-Lewis model in depicting analytically the incorporation of these national economies into the world economy. However, this incorporation would not be complete without, first,

describing the financial interaction of these economies. Hence, the discussion relating to official inflows presented in Chapter 2, combined with an international credit-rationing hypothesis, will need to be integrated into the Kalecki-Lewis model, in order to strengthen the relevance of this model for Africa. This line of argument stands as the main justification for adopting this structure in the model constructed in the following section.

Two other important issues arise from the survey of models presented in this chapter. Firstly, undertaking a survey of existing models is extremely important in identifying the specific linking mechanisms, which will need to be incorporated during model construction. Hence, the linkages identified within these models will be used to guide the construction of the model set out below. Secondly, the above survey demonstrates that the modelling of the South remains extremely rudimentary. Indeed, in the case of Africa, such modelling has been neglected almost entirely. The major implication of these findings is that insufficient account has been taken of the basic macro features of the South within existing North-South models. This implies a need to identify the salient macro features of African economies, as was undertaken in Chapter 3, and to integrate these within a North-South modelling framework. Hence, the following section will be devoted to such an exercise.

6.5 AFRICA WITHIN A NORTH-SOUTH FRAMEWORK: A NEOSTRUCTURALIST GLOBAL ECONOMETRIC MODEL (AFRIMOD)

The specification of the North-South model outlined below is based on the theoretical North-South models of Taylor (1981, 1983, 1991), Darity and FitzGerald (1982), Kanbur and Vines (1986), Murshed (1990) and Vos (1994), as well as on the empirical models of Marquez and Pauly (1987) and Masson et al. (1990). In relation to the theoretical discussion set out above, the basic characteristic of the model is its effort to combine a Kaleckian/Keynesian North, a Lewis-type economy for the South, and an oil-producing and -supplying region (representing the OPEC countries). The North and OPEC will be dealt with in a very limited fashion, with an attempt to develop and elaborate upon the modelling of Africa taken as the prime objective. This approach stands in opposition to the current practice in North-South modelling, in which the focus has, mainly, been on elaborating upon the section of the model dealing with the North. All parameters, with the exception of two equations, and some ratios, have been econometrically estimated. This means that the model may also be interpreted as an empirical model. The discussion will be organized as follows. In sections 6.5.1 to 6.5.5, a North-South model will be specified. Although the specification set out below takes a linear form, all estimations are undertaken using a log-log model. However, in order to ensure compatibility with the identities specified in the model, appropriate conversion of the loglog model is undertaken. Section 6.5.6 outlines some practical analytical solutions, which help in understanding the major issues, which are addressed within the model. Finally, a glossary of symbols used and coefficients estimated is given at the end of the chapter.

6.5.1 The North

The modelling of the North adopted in this book is not an elaborated one. There are two main reasons for this: firstly, the focus of the book is mainly on the modelling of the South, and Africa in particular; secondly, if need be, an elaborated model of the South developed in this book may be combined with the existing models which have an elaborated North. Notwithstanding the existence of a number of models incorporating an elaborated North, I have specified below a less elaborated model of North, based on Taylor (1981). The use of this less elaborated model for the North has two desirable consequences. Firstly, it allows one to specify the North along theoretical lines, which are best suited to the purposes of this book. Secondly, incorporation of this less elaborated model for the North allows one to understand the interaction of the two regions and hence, subsequent feedback effects of the North on the South.

6.5.1.1 The major macro aggregates sub-bloc

The North is assumed to be an economy with excess capacity, with output taken as being demand determined, along Keynesian/Kaleckian lines. This is given by equation 1, which is the aggregate equilibrium condition, and the basic closure of the model for the North.

$$Q_N = C_N + I_N + G_N + X_N - M_N$$
[1]

The demand component comprises a simple consumption function that depends on income (equation 2), an investment function, which depends on income and world interest rate (equation 3) and exports and imports (given in equations 4 and 5, respectively). By making the two regions, OPEC and the South, the destination and source for exports and imports of the North, respectively, equations 4 and 5 ensure a trade linkage among the three regions.

$$C_N = a_2 + b_2 Q_N \tag{2}$$

$$I_N = a_3 + b_3 Q_N - c_3 i_W$$
[3]

$$X_N = PM_N^{fg} \left(MM_O^{fg} + MM_A^{fg} \right) + M_{OT} + otX_N$$
^[4]

$$M_N = MR_N^S + P_{oil} \left(MO_N^O + XO_{NA} + XO_{WC} \right) + otM_N$$
^[5]

The Northern (manufactured) price equation (equation 6) is based on a Kaleckian mark-up pricing approach, which assumes an oligopolistic market structure in the North. The specification follows recent works of Taylor (1981, 1983, 1991), Darity and FitzGerald (1982), Marquez and Pauly (1987) and Vos (1994). It is positively related to prices of intermediate inputs, comprising wages, imported raw materials and oil price, and inversely related to excess capacity in the North.

$$PM_N^{Jg} = a_6 + b_6 P_{oil} + c_9 w_N + d_6 P_R - e_6 EXCAP$$
^[6]

Following a Kaleckian/Kaldorian tradition, private saving is assumed to come from profit and non-wage income, which, in turn, depends on the mark-up pricing scheme specified in equation 6. Following Murshed, the level of saving is further augmented by factor income from abroad as well as interest income from Southern debt. Since such income usually accrues to multinational firms and banks, their propensity to consume out of these factor incomes is assumed to be negligible. This is given in equation 7.

$$S_{N} = s_{N} \left[\tau \left(w_{N} b_{N} + P_{oil} m_{N} \right) \right] Q_{N} + (1 + \mu_{s}) i_{w} B_{s} + i_{w*} F_{s} + i_{w} \left(D C_{N} - J_{N} - J_{OT} \right) + f p_{s}$$
[7]

Factor payment receipts from the South is formulated based on a historical trend of factor payments and foreign direct investment (FDI)

to South. Thus, the coefficients in equation 8 below are average ratios of factor payments to FDI for the years 1983–90.

$$fp_S = a_8 FDI_{OT} + b_8 FDI_{NA} + c_8 FDI_{WC} + d_8 FDI_{ES}$$
[8]

The North also interacts with the OPEC (oil-producing and -exporting countries) through its oil consumption function, which is given in equation 9, as well as subsequent demand for oil imports, which is given by equation 10. Consumption of oil in the North is related to price of oil, the price of a substitute, such as coal, and the level of economic activity in the North.

$$COIL_N = a_9 - b_9 P_{oil} + c_9 P_{coal} + d_9 Q_N$$
 [9]

$$MO_N^O = (C_{oil_N} - Q_{oil_N}) - XO_{NA} - XO_{WC}$$
[10]

6.5.1.2 The official capital flows sub-bloc

Following Vos (1994), the deficit of the North may be specified as in equation 11, where it is financed by issuing bonds, or by assuming that the government in the North is a borrower, having preferential access to international capital markets.

$$DEF_N = \Delta DC_N = G_N + i_w DC_N + \Delta F_S - t_N (1 - P_{oil} m_N) Q_N$$
[11]

However, the Vos version of this is extremely aggregated and does not specify how inflows to the South are determined and allocated. Hence, a major extension of Vos' approach is undertaken by further specifying official capital flows to the South, using equations 12 and 13. As discussed in Chapter 2, given the total inflow to the South (in equation 11) as a policy variable in the North, or determined by the North's affordability criteria, the allocation of the total inflow across the South, given by equation 12, is based on the theoretical approaches of, *inter alia*, Ruttan (1992), OECD (1985), Mikesell (1968), Mosley (1987), Dudley and Montmarquette (1976), and McGillivary and White (1993). This literature emphasizes economic, strategic and political self-interest, as well as developmental/humanitarian considerations. Of these, economic considerations are represented by the first two arguments within equation 12, strategic and political self-interest by the third argument and developmental/humanitarian considerations by the
fourth argument. This specification is extended to include indicators of relative repayment capacity (represented by the fifth argument in equation 12), since this represents another important economic variable not considered in the literature. As shown in Chapter 4, the formulation of equation 12, using an ECM, allows the error correction term to capture not only the impact of recent past budgetary practices but also previous historical/colonial ties.

$$\Delta F_{A} = a_{12} - b_{12} \left(\frac{M_{OT}}{M_{A}} \right) - c_{12} \left(\frac{FDI_{OT}}{FDI_{A}} \right) + d_{12}SAP + e_{12} \left(\frac{Y_{OT}^{P}}{Y_{A}^{P}} \right) + f_{12} \left(\frac{DSR_{OT}}{DSR_{A}} \right)$$
[12]

6.5.1.3 The foreign direct investment sub-bloc

Following the discussion presented in Chapter 2 and the empirical analysis of Chapter 4, equation 13 is specified using a simple 'eclectic' approach to the determination of foreign direct investment (Dunning 1993) and theory of industrial organization (Helleiner 1989). The first argument in this equation shows market size, the second is used to capture the concentration of FDI in the mining sectors of most African countries, the third argument is a dummy, which indicates the number of countries who followed a programme of deregulation, or adopted a structural adjustment package or similar policy incentive, and finally, the lagged dependent variable is used to capture other historical considerations, as well as 'economies of specialization' or internalization used in pursuing FDI. In the first two arguments the variables are computed relative to other South, in order to indicate the choice faced by Northern agents.

$$FDI_{A} = a_{13} - b_{13} \left(\frac{Q_{OT}}{Q_{A}} \right) - c_{13} \left(\frac{MIN_{OT}}{MIN_{A}} \right) +$$

$$d_{13}SAP + e_{13} (FDI_{A})_{t-1}$$
[13]

Both equations 12 and 13 are fitted to the three regions, which form the focus of this study.

6.5.1.4 The commodity sub-bloc

The commodity sub-bloc arises out of the empirical analysis presented in Chapter 4. Ideally the modelling of the commodity could have been undertaken along the lines of Hwa (1985) and Ramanujam and Vines (1990), both of which are based on the theoretical work of Adams and Behrman (1976).¹⁵ However, this could not be carried out due to the absence of available data on stockholding. As discussed in Chapter 4, the commodity bloc specified in equations 14 and 15 indirectly addresses the stockholding behaviour of Northern agents. Following an econometric analysis, the equilibrium version is found to be relatively the better and hence, this version below is used.

$$X_{i}^{dd} = a_{14} + b_{14}Q_{N} - c_{14}i_{w} - d_{14}\left(\frac{P_{R_{i}}}{PQ_{N}}\right)$$
[14]

Solving equation 14 for P_{Ri} and taking the log of the result gives equation 14.1. This equation is used for estimating the simultaneous equations-based model (given in Appendix 4.3). Results reported in Table 8.2 below refer to the ECM-based estimation of equation 14.

$$P_{R_i} = a_{14} + b_{14}Q_N - c_{14}i_w + d_{14}PQ_N - e_{14}X_i^{dd}$$
[14.1]

$$X_{i}^{ss} = a_{15} + b_{15} \left(\frac{P_{Ri}}{PQ_{s}} \right)_{t-1} + c_{15} \left(\frac{I_{s}}{Y_{s}} \right)_{t-k}$$
[15]

where: k is the number of lag period and i stands for commodity i.

$$X_i^{dd} = X_i^{ss}$$
 [16]

Equations 14 and 15 are used for four commodity categories (i = 1...4) consisting of food, beverage, agricultural raw material, and, finally, metals, ores and minerals. As discussed in Chapter 4, as well as in Bond (1987), this distinction is found to be empirically important. Thus, through equation 16, the global commodity market for each commodity category clears for price (i.e., flexi-price). As discussed in Chapter 2, this is in line with various studies relating to the functioning of commodity markets, starting from the classic works of Prebisch (1962) and Singer (1950). Following a lag period, the price derived in this market, in turn, determines the supply of exports from Africa

(equation 33). Moreover, through the identity given in equation 50, supply from Africa has an effect on global supply, which is given in equation 15. Thus, as illustrated in Diagram 4.1 of Chapter 4, the global commodity market and African supply are dynamically linked.

6.5.2 The South: Africa

6.5.2.1 The major macro aggregates sub-bloc

Within this sub-bloc, the South is divided into two regions, comprising Other South (OT) and Africa (A). The latter, in turn, is further divided into three sub-regions, based on the UNECA classification. These are East & Southern Africa (ESA), North Africa (NA) and West & Central Africa (WCA).¹⁶ As discussed in Chapter 1, an historic and economic justification for such a classification, based on the nature of each region's interaction with Western Europe, may be found in Nzulu et al. (1979) and Amin (1972). The model specified below is a prototype model used for each region. The rest of the World and Other South categories are, on the whole, assumed to be exogenous to the model, the major features of which are outlined below.

Unlike the North, the South is assumed to be a supply-constrained economy. In general, the long-run, supply-constrained nature of such economies may be depicted by a production structure characterized by full-capacity utilization (Taylor 1981, 1983). However, in Africa, the past two decades have witnessed some vital sectors of the economy being faced with under-capacity utilization, arising from problems relating to the supply of imported inputs. This has been especially true of manufacturing, which is dependent on intermediate imports, as well as those sections of the agricultural sector which are dependent on imports of fertilizer and transport facilities. This theme has recently been taken up in the 'import compression' literature in general, and by Ndulu (1986, 1991) and Rattsø (1992a) in particular. Hence, I have explicitly incorporated these considerations in modelling the South through equations 18, 19 and 20. All type of foreign inflows, except foreign direct investment (FDI), are specified as determining the level of imports.

In relation to public and private investment, these are linked to the level of imports through equations 18 and 19. As discussed in Chapter 3, as well as in FitzGerald et al. (1992), the investment functions are based on a theoretical framework that depicts the externally constrained

nature of investment in developing countries. Given the flow of foreign finance (whether based on trade or on external borrowing), as well as outflows of foreign exchange (with debt service payment included), the level of imports is defined (by way of equation 20) as an accommodating variable. The level of reserves is also assumed to be zero, which is not an unreasonable assumption, given the precarious nature of reserves in Africa. A further simplifying assumption which is employed is that (after settling external financial obligations, including capital flight) all types of aid are best considered as representing an additional source of external finance, which will enhance the level of imports. Although this assumption is very limiting in the context of 1970s Africa, where project aid was important, it is quite relevant for the 1980s and 1990s. Moreover, equation 20 is also central in incorporating the basic themes summarized in the import compression literature. This equation depicts a situation in which imports are squeezed by foreign obligations, of which debt servicing is central.

Thus, equations 17–23 together summarize the major macro bloc of the South. Beginning with equation 17, this equation illustrates the supply of output from the South (the estimation procedure for the capital output ratios is discussed below).

$$Q_A = a_{17} + \sigma_{AP} I_{AP} + \sigma_{Ag} I_{Ag}$$
^[17]

where: $\sigma_{AP} = \frac{\Delta Q_A}{\Delta K_{AP}}; \quad \sigma_{Ag} = \frac{\Delta Q_A}{\Delta K_{Ag}}$

$$\Delta K_{Ag} = I_{Ag}^* = a_{18} + b_{18}Q_A + c_{18}M_A + d_{18}I_{Ag_{t-1}}$$
[18]

subject to: $I_{Ag}^f \leq \alpha M$

where: * shows desired level of investment, is the share of M that went to the public sector investment, superscript f is the foreign exchange-constrained level of investment.

$$\Delta K_{AP} = I_{AP}^{*} = a_{19} + b_{19}Q_A + c_{19}M_A + d_{19}I_{Ag_{l-1}}$$
subject to: $I_{AP}^{f} \le (1 - \alpha)M$
[19]

$$M_{A} = X_{A} + \Delta F_{A} + \Delta B_{A} + FDI_{A} + i_{W}J_{S} - \Delta J_{A} -$$

$$(1+\mu)i_{W}B_{A} - i_{W*}F_{A} - k_{8}FDI$$
[20]

where: k = b, c and d

$$C_{oil_A} = a_{21} + b_{21}P_{oil} + c_{21}\Delta F_A + d_{21}Q_A$$
^[21]

Equation 21 summarizes consumption of oil by the South. However, unlike Marquez and Pauly (1987), in which oil consumption of the South is considered to be a substitute for capital in the context of cost minimization, we argue here for its complementary. Thus, it has a relatively rigid (inelastic) demand that depends on its own price, availability of external finance and the South's level of output. The econometric estimation would also appear to lend support to this position.

$$MM_A^{fg} = a_{22} + b_{22} PM_N^{fg} + c_{22} Q_A + d_{22} \Delta F_A$$
[22]

$$XO_A = a_{23} + b_{23}P_{oil} + c_{23}FDI_{A_{l-1}}$$
^[23]

Equation 22 gives manufactured imports of South, which is specified as a function of own price, domestic demand and availability of foreign inflows. The export of oil function, specified in equation 23 is not used in relation to the ESA region, since that region does not produce oil. Rather, exports of oil are assumed to depend on the relative (to coal) price of oil, economic activity in the North and supply inducing factors such as FDI. Since the econometric analysis did not produce a statistically significant result in relation to the price of coal and Northern economic activity, they are omitted from equation 23. The importance of supply factors in oil export functions is discussed in Oshikoya's (1989) model for Nigeria. He uses output of oil as a regressor. However, since the relationship between output and exports are proportional, this is tantamount to regressing a variable on itself. Thus, in order to rectify this shortcoming, lagged (domestic) investment in the sector is used within this model. The latter, however, turns out to be statistically insignificant while FDI remains significant. Hence, this serves as justification for the inclusion of FDI within equation 23.

6.5.2.2 The fiscal response sub-bloc

The fiscal response of the South to external finance is modelled using equations 24-32. Considering the empirical findings of Chapter 5, government fiscal response in most African countries is assumed to depend on the influence of an uncertain external economic environment (particularly in relation to foreign inflows), as well as the political economy of deficit financing. Following the discussion presented in Chapter 5, tax revenue is modelled using equations 24 and 25, which deal with direct tax and indirect taxes, respectively. Government expenditure and borrowing, as well as the relevant identities, for this sub-bloc are modelled using equations 27-32, where Z represents borrowing from banks along with resource transfers from the private sector, $Z_b + Z_{pr}$ (see Alemavehu et al. 1992 and FitzGerald 1993).

$$T_A^d = a_{24} + b_{24}Q_A + c_{24}\Delta F_A$$
 [24]

$$T_A^{iex} = a_{25} + b_{25}C_p + c_{25}(X+M)_A + d_{25}\Delta F_A$$
[25]

$$T_A = T_A^d + T_A^{iex} + otGR_A$$
^[26]

$$G_A = a_{27} + b_{27}T_A + c_{27}\Delta F_A + d_{27}G_{A(-1)}$$
^[27]

$$S_{Ag} = T_A + otGR_A - G_A$$
^[28]

$$S_{Ap} = a_{29} + b_{29}Q_A$$
 [29]

$$S_A = S_{Ag} + S_{Ap}$$
[30]

$$\Delta F_A = \Delta F_{Ag} + \Delta F_{Ap} \tag{31}$$

$$Z_A = Z_B + Z_{pr} = I_{Ag} - S_{Ag} - \Delta F_{Ag}$$
[32]

6.5.2.3 The commodity export supply sub-bloc

The supply of exports from Africa is modelled based on the analysis presented in Chapter 4. As described in this chapter, the real exchange rate-based model fits the data best and, hence, it is this version of the model, which is employed here. The ECM version of the export supply equation (equation 33) shows both the short-run and a long-run impact of price and capital formation indicators. The short-run parts of the argument are based on (latent) capacity utilization theory (Wickens & Greenfield 1973, Goldestin & Khan 1978, Chu & Morrison 1986 and Hwa 1985). The last argument in equation 33 is used to indicate the problem of capital formation (investment) in the sector. The lag structure (i.e., k) is five to six years when i is beverage or minerals while one to two years for food and agricultural raw materials.

$$XR_{A_{i}} = a_{33} + b_{33} \left(\frac{eP_{R_{i}}}{PQ_{A}}\right)_{i-k} + c_{33}K_{i-k}$$
[33]

6.5.2.4 The 'Dutch disease' sub-bloc

The interaction between equations 33, 34 and 35 describes a 'Dutch disease' possibility, which was discussed in detail in Chapter 5. Using the alternative framework set out in this book, foreign inflows could lead to an appreciation in the exchange rate, through the demand that might be created in the non-traded sector. Indeed, the latter could further affect the performance of the traded sector, through equation 33 above. Thus, the econometric result obtained in Chapter 5 is based on the reduced form equations derived from the broader framework given in Edwards (1989). However, in order to make the specification consistent within the North-South model specified in this chapter, the following two steps are followed. Firstly, in equation 34, domestic price is set as a function of excess demand and other determinants of price. At the second stage the real exchange rate is defined, in equation 35, using the nominal exchange rate, which is assumed fixed, as well as the price of exports, which was determined in the commodity sub-block of the model.

For economies with an important food sector, Taylor (1983) has described the formation of prices using excess demand. Within a national macroeconomic framework, and when data permits, the disaggregation of excess demand into food and non-food sectors has quite profound implications (see Taylor 1983: 39–48). However, in the estimation given below, an aggregate version of a similar equation is used. This equation incorporates the possibility of imported inflation, or a mark-up pricing effect, and also explicitly employs foreign inflows. Moreover, by using an ECM, price stickiness is also assumed. The basic specification is given by equation 34.

$$PQ_A = a_{34} + b_{34}\Delta F_A + c_{34}MUP$$
 [34]

$$RER_{Ai} = \frac{eP_{R_i}}{PQ_A}$$
 $i = 1..4$, for the four commodity categories [35]

6.5.3 Major Oil Exporting Countries (OPEC)

The total output of OPEC is assumed to depend on demand for oil both in the North and South which, in turn, depends on the level of output for each region, as shown in equation 36.

$$Q_O = m_{No}Q_N + m_{So}Q_S$$
^[36]

Given the import of oil to the North (equation 10) and that of Africa (equation 21), oil imports for the rest of the South are derived as residual (equation 37). Although imports and consumption of oil may safely be assumed to be equal in the ESA region and the majority of countries in WCA (equation 21), this assumption does not hold for NA and the major oil producers of WCA (Gabon, Cameroon and Nigeria). Thus, oil consumption for oil-producing countries actually equals production less exports.

$$MO_{OT}^{O} = Q_O - MO_N^O - C_{oil_A}$$
^[37]

$$MM_O^{fg} = m_{oN} P_{oil} Q_O$$
^[38]

$$\Delta J_O = P_{oil}Q_O - (PM_N^{fg} \cdot MM_O^{fg}) - otM_O + i_w J_O$$
^[39]

Finally, equation 38 summarizes the portion of OPEC's income that is spent on Northern goods, with the remainder, including interest income on savings, saved in international banks, as per equation 39.

6.5.4 International Banking

This section is based on the work of Darity and FitzGerald (1982) and Vos (1994). In this literature, in contrast to the pure theory of capital movement discussed in Chapter 2, international banks are assumed to work on a segmented capital market, characterized by credit rationing.¹⁷ In order to capture this phenomenon, we begin by specifying the basic closure of this bloc (equation 40),

$$\Delta B^{s+p_N} = \Delta J_{OT} + \Delta J_N + \Delta J_O + \Delta J_A - \Delta DC_N$$
^[40]

$$\Delta J_N = a_{41} + b_{41} J_N + c_{41} (i_w - r_N)_{t-1}$$
[41]

$$\Delta J_{A} = a_{42} + b_{42} J_{A}$$
 [42]

Following Vos (1994), equation 40 equates the supply of funds from the three regions, net of Northern government demand for deficit financing, with the demand for credit by South, as well as the private sector of the North. Equations 39, 41 and 42 summarize the supply of funds. The credit rationing is effected as follows. First, priority will be given to Northern firms, as specified in equation 43.

$$\Delta B_{pN}^{d} = a_{43} + b_{43} I_{N}$$
[43]

The remainder of funds are, then, directed to the South. The supply of credit to the South could, in principle, be specified using two approaches.

In Vos' model, suppliers' perception of default is believed to be inferred from a critical outstanding stock of bank debt to export earning ratio (Vos 1994: 218). From the borrower's perspective Vos' model essentially maintains that there is a certain debt management scheme, which is employed by Southern governments. Hence, the demand for bank loans in the South is limited by a maximum targeted level for the interest to foreign exchange earnings ratio.¹⁸ Thus, the ratio essentially implies the possibility of downward adjustment of demand for foreign inflows by Southern borrowers, based on their own assessment or debt management scheme, irrespective of supply. Here, demand is not necessarily supply determined, and neither is it perfectly elastic.

On the other hand, the IMF's MULTIMOD (Masson et al. 1990) follows quite a different formulation, comparing an interest-to-export ratio with a certain critical maximum as an indicator of repayment ca-

pacity. Thus, the ratio basically determines supply, with demand, by the South, implicitly assumed to be perfectly elastic. Both suppliers and borrowers are assumed to use the same indicator. By ensuring that this ratio lies well below the target level, borrowers may ensure an increase in the supply of funds.

Based on the experience of Africa, detailed in Chapter 1, the absence of debt management and the existence of nearly perfectly elastic demand are likely to represent the more realistic set of assumptions.¹⁹ Hence, MULTIMOD's approach is followed. Moreover, since the South represents an aggregate category, Vos' (1994) model does not detail a mechanism which may be used to show how bank flows are allocated among borrowers in the South. Consequently, Vos' model is extended to incorporate the assumption that Africa represents a poorer risk category than the rest of the South and hence, that it is only allocated surplus capital, once demand for such capital from the rest of the South has been met. The latter, in turn, will be determined by its historical maximum share, which is assumed to be exogenous within this model. There are two main reasons why this is likely to represent a realistic picture for African countries. Firstly, as documented by Eaton and Gersovitz (1980) and Odedokun (1996), bank flows to these countries are likely to be supply constrained. Secondly, Kasekende et al. (1995) describe how such flows to Africa are characterized by a high level of supply volatility. Hence, these flows are given as in equation 44.

$$\Delta B_A = \Delta B_N^{s+pN} - \Delta B_{pN}^d - \Delta B_{OT}$$
^[44]

where: $\Delta B_A \leq \Delta \overline{B}$, $\Delta \overline{B}$ is the historically determined maximum.

The second extension to Vos' model relates to the allocation of funds across Africa. In this study, the debt service ratio (equation 47) is used as the relevant indicator of both solvency and liquidity in the African context. This is consistent with the findings of Kasekende et al. (1995) and other recent studies which focus on short-term flows to Africa. Given the debt service ratio, the allocation mechanism developed in this study is based on the relative position of each African region's solvency indicator to the target level for the same ratio set by suppliers. Thus, it is assumed that although Africa is a worst-risk category among Southern borrowers, allocation of inflows destined for Africa will depend on the relative level of the debt service ratio for each region. This is formally given as follows,

$$\Delta B_{Ai} = \eta_i \Delta B_A \tag{45}$$

$$\eta_{i} = \frac{(\lambda^{*} / \lambda_{i})}{\sum_{i=1}^{n} \left(\frac{\lambda^{*}}{\lambda}\right)_{i}}$$
[46]

$$\lambda_{i} = \frac{i_{W}(1+\mu_{S})B_{Ai} + i_{W}*F_{A} + \Pr(in_{Ai} + otDS_{i})}{X_{Ai}}$$
[47]

where: λ^* is exogenously determined maximum debt service ratio (formulated by suppliers). λi is debt service ratio of the i^{th} risk group (region) in Africa. The sum of in equation 46 over *n* adds to 1.

Once this credit rationing is effected, the financial market eventually clears for interest rate, by way of equations 43 and 3.

6.5.5 Identities Used

$$XR_{Ai} = XR_{ESi} + XR_{NAi} + X_{WCi}$$
^[48]

where: i = 1...4, the four commodities.

$$X_{Ai} = \sum_{i=1}^{4} XR_{Ai} + otX_A + P_{oil}XO_A$$
[49]

$$X_i^{SS} = XR_{A_i} + XR_{OT_i}$$
^[50]

$$\Delta J_A = \Delta J_{ES} + \Delta J_{NA} + \Delta J_{WC}$$
^[51]

$$Q_A = Q_{ES} + Q_{NA} + Q_{WC}$$
^[52]

6.5.6 Some Partial Equilibrium-based Analytical Solutions

In this section some basic and partial analytical solutions to the above model are set out. The results are partial, since general equilibrium effects and stability conditions of the model are not discussed. However, although partial, this discussion does help to synthesize some of the major theoretical underpinnings of the proposed model.

The South

Owing to the focus of the study, the South is left as relatively disaggregated, with total output determined using the output-capital ratio, given in equation 17. Since output in the South is assumed to be both private and public capital constrained, an increase in capital will raise output. However, such capital flows do have macroeconomic effects, one of which is the fiscal response.

The fiscal response

An important aspect of the South part of the model is the attempt to show the fiscal posture of the South within a North-South framework. Thus, inserting equations 24, 25 (using identity 26), 27 and 18 into a definition of public deficit (I_{Ag} - S_{Ag}) from equation 32 and solving for Z_S results in,

$$Z = (a_{18} + a_{27} - a_{24} - a_{25}) + (b_{18} - b_{24})Q_A + (c_{18} - c_{25})M_A + (c_{27} - c_{24} - d_{25})\Delta F_A - otGR_A + [6.7a]$$

$$b_{27}T_A + d_{18}I_{Ag_{I-1}} - b_{25}C_p - c_{25}X + d_{27}GA_{I-1}$$

The fiscal deficit in the South is financed either through foreign inflows, which are usually predetermined in the North, or through domestic financing (Z). Thus, the public deficit is adjusted by a partial financial crowding-out process, presumed in Z, that would affect Ip, either directly or through the imposition of an inflationary tax. This phenomenon is not examined further in this study. However, crowding-in/out, in its rudimentary form may be seen in the model by examining the effect of Ig on Ip.

From the above equation, leaving aside the equilibrium effects, economic activity might induce a deficit through the demand that it creates for investment, but may also dampen it through its effect on government revenue. The net effect will depend on the two elasticities (b_{18} and b_{24}). Similar effects may also be read from the coefficients of imports. This partial result shows that foreign inflows have a negative impact on the deficit, through its expenditure-inducing (c_{27}) and revenuedepressing effects (the sum of c_{24} and d_{25}). However, the general equilibrium effect could be different, since foreign inflow, through imports, may induce a rise in output. Foreign trade, expansion in private consumption (openness of the economy) and government efforts in enhancing non-tax revenue could also help to reduce the government deficit. On the other hand, the deficit is positively related to the levels of investment and government consumption.

The commodity market

The unique feature of this North-South model is the inclusion of a commodity model sub-bloc, which is commonly built independent of a global macroeconomic model. This market is assumed to be a flexiprice market that encompasses Northern agents running a stock in order to meet demand/supply for commodities. The market functions through the equilibrium condition outlined in equation 16. Using equations 14– 16, the partial model for price could be given by,

$$P_{R_{i}} = -\left[a_{15} - a_{14} + b_{15}\left(\frac{PR_{i}}{PQ_{s}}\right)_{i-k} + c_{15}\left(\frac{I_{s}}{Y_{s}}\right)_{i-k} - b_{14}Q_{N} + c_{14}i_{w}\right]\frac{PQ_{N}}{d_{14}} \qquad [6.7b]$$

It can be read from the first-order partial derivative of the above equation that for a given level of demand, an increase in supply of commodities will negatively affect the price level. This value is given by,

$$\frac{\partial P_{\mathrm{R}_{i}}}{\partial I_{\mathrm{S}_{i-k}}} = -\frac{c_{15}}{d_{14}Q_{\mathrm{S}_{i-1}}}PQ_{N} \qquad [6.7c]$$

This is a consequence of the flexi-price structure of markets under which commodity producers of the South operate. Moreover, Southern price is negatively related to current and lagged real exchange rate, world interest rate, profitability and the possibility of financing investment within the export sector. Thus, notwithstanding the hypothesis that a high level of supply leads to high income, owing to a strong belief in 'the small country assumption' or ultimately Say's law, the adoption of a global framework helps in identifying the possibility of a deterioration in terms of trade. This solution not only highlights the 'fallacy of composition in policy advice' but also the point that, unless the production/marketing structure is changed, individual country efforts could ultimately be self-defeating at the aggregate level.

The 'Dutch disease'

A characteristic observed in many African countries' interactions with the North is the phenomenon of the 'Dutch disease'. As described in Chapter 5, the focus here is on what is termed the 'spending effect', which, in turn, will affect the supply of exports. Thus, foreign inflow, through its spending effect, will raise the demand for non-tradables, which will affect the real exchange rate, as defined in equations 35–34. Following a one-period lag, this, in turn, will have an effect on the traded sector, which may be depicted by equation 33. Hence, incorporating equation 34, by inserting the lagged values of equation 35 into equation 33 and taking its first-order partial derivative with respect to foreign inflow, gives,

$$\frac{\partial XR_{A}}{\partial F_{A}} = \frac{-b_{33}c_{34}ePR_{i_{l-k}}}{\left(a_{34} + b_{34}F + c_{34}MUP\right)^{2}}$$
[6.7d]

where: k is periods of lag.

This partial equilibrium result shows that the 'Dutch disease' impact of foreign inflows for a given level of nominal exchange rate may be detrimental to exports, if it is large enough to offset its impact on capital formation within the export sector. Generally, the non-traded sector is positively related to the current and lagged level of foreign inflows and inversely related to export-enhancing factors. This highlights the policy dilemma of export promotion and domestic production, especially of food.

Southern macro balance

The macro balance of the South is derived by combining the private and public investment, saving and foreign inflows components within a single ex-post identity, which equates the saving gap with foreign inflow. Thus, imports of the South (equation 20) are substituted into the two investment equations (18 and 19). This ensures that the import compression phenomenon is taken into account. By inserting equation 17 into equation 29, Southern savings may also be linked with the capital-constrained output of the South. Finally, the investment and saving gap is set to match the level of foreign inflows, coming in the form of aid, FDI and bank loans, in order to yield the macro balance equation for the South. This latter equation is then solved for the Southern level of output. Since the resulting algebra is extremely long, the analysis presented in this section is based on some of the values of first-order partial derivatives.

$$\frac{\partial Q_A}{\partial M_A} = \frac{c_{25}}{(b_{18} + b_{19} - b_{24})}$$
[6.7e]

$$\frac{\partial}{\partial} \frac{Q_A}{i_w} = \frac{-(c_{19} + c_{18})J + (c_{18} + c_{19})(1 + \mu_s)B_A}{(b_{18} + b_{19} - b_{24})}$$
[6.7f]

$$\frac{\partial}{\partial} \frac{Q_A}{\Delta B_A} = \frac{1}{(b_{18} + b_{19} - b_{24})}$$

and $\frac{\partial}{\partial} \frac{Q_A}{B_A} = \frac{(c_{18} + c_{19})(1 + \mu_S)i_w - c_{18} - c_{19}}{(b_{18} + b_{19} - b_{24})}$
$$\frac{\partial}{\partial} \frac{Q_A}{\Delta F_A} = \frac{(1 + c_{24} + d_{25} - c_{18} - c_{19} - c_{27})}{(b_{18} + b_{19} - b_{24})}$$

(6.7g]

and
$$\frac{\partial Q_A}{\partial F_A} = \frac{(c_{18} + c_{19})t_{w^*}}{(b_{18} + b_{19} - b_{24})}$$
 [6.7h]

From equation 6.7e it can be seen that, given the investment and saving elasticity, the level of output in the South is directly related to the level of imports. Thus, when imports are scarce, as is the case in the 1980s, the level of investment, and consequently output, will also be adversely affected for a given level of the capital output ratio.

Similarly, from equation 6.7f, if (a) the demand impact of Southern output on investment is larger (or smaller) than its impact on savings, and (b) import elasticity, with respect to the cost of borrowing from banks, is larger than its elasticity with respect to interest revenue from capital flight, then this could indicate the existence of a direct (or inverse) relationship between world interest rate and Southern level of output.

The impact of capital inflows and debt stock on the level of output may be read from equations 6.7g and 6.7h. The partial results suggest that for a given level of income elasticity of investment and saving, the impact of debt stock on output of the South will depend on the level of interest rate (both concessional and non-concessional) as well as the sensitivity of investment to imports. In the case of capital flows, their fiscal impact also matters, and this may be inferred from the values of c_{24} , c_{25} and c_{27} .

The North

As indicated above, the Northern model is kept at a highly aggregated level, owing to the fact that the focus of this study is mainly on Africa. Hence, the partial analytical solution for the North is limited to the discussion of North's macro balance. Equilibrium in North may be attained by ensuring that the North's macro balance holds. That is, the investment-saving gap, including the fiscal deficit, should equal the external gap, given as imports, plus net factor and current transfer receipts less exports. Combining the deficit, as defined in equation 11, with equations 3, 4, 5 and 7 in the North's macro balance equation (i.e., I-S + DEF = M-X + NFP + NCTr), and solving the result for one important macro variable in the North, say investment, can help to highlight the partial implication of the model in the North. Since the resulting algebra is extremely long the following discussion is, again, based on the first-order partial derivative of such an investment equation.

$$\frac{\partial I_N}{\partial i_w} = s_N (B + B\mu_S + DC_N - J_N - J_{OT}) - DC_N \qquad [6.7i]$$

$$\frac{\partial I_N}{\partial Q_N} = (1 - m_N P_{oil}) t_N + s_N \tau (m_N P_{oil} + w_N b_N)$$
[6.7j]

Equation 6.7i shows that the ex-post level of investment is positively related to interest-yielding lending. It is also inversely related to the cost of borrowing. In both cases the saving rate is important. The latter result underscores the importance of saving on net lending. As could be expected in a Keynesian set-up, investment is positively related to the level of output (equation 6.7j). However, a rise, either in the level of intermediate inputs or in its price, may depress investment, if saving from the mark-up based profit, excluding the wage based component, is less than the tax revenue levied on intermediate imported inputs.

The international financial market

The credit-market component of the model highlights the segmented nature of this market. It also highlights how credit rationing is put into practice, with the South often being the last to be allocated such credit. However, eventually, the market will clear for world interest rate, as can be shown using equations 3, as well as equations 39–43. We may arrive at the equilibrium solution for this market by letting excess demand clear for world interest rate. Solving this final equation for world interest rate gives the following partial equilibrium solution,

$$i_{w} = \frac{a_{43} + a_{3}b_{43} + b_{3}b_{43}Q_{N} + B_{A} + B_{OT} + \Delta DC_{N} - J_{N} - J_{O} - J_{A} - J_{OT}}{c_{3}b_{43}}$$
[6.7k]

The result indicates that interest rate is positively related to investment demand as well as the demand for credit in both North and South. However, it is inversely related to the deposit and crowding-out effects of cost of investment financing (c_3) .

6.5.7 Parameters of the Model

Two sets of parameters are used in the above model. In general, the model is based on the single-equation ECM-based estimation of the functions specified above. The main results of this econometric study are discussed in Chapters 4 and 5 of this book. However, for a few equations, parameters are derived using different sources and methods. Hence, section 6.5.7.1 outlines this latter type of parameter, while section 6.5.7.2 discusses the estimation approach used and lists the complete parameter values of the model.

6.5.7.1 Parameters derived and used from previous studies

(a) Capital output ratio

The capital output ratio used is derived from an estimation of a similar equation to that of Sarmad (1990) and Khan and Reinhart (1990). The approach is based on a production function, designed to show the impact of public and private investment, together with other productivity indicators captured by the constant term on growth of real output. Although, in terms of the relevant diagnostic test, the function is not satisfactory, nevertheless it represents a better approximation than existing estimates such as that undertaken by Sarmad (1990), since his sample contains only two countries from Africa. Table 6.3 summarizes the estimated coefficients derived from equation 17 for each region.

| Region/ Parameter | a ₁₇ | (t- values) | σΑΡ | (t- values) | σ Ag | (t- values) |
|----------------------|-----------------|----------------|------|----------------|-------------|----------------|
| ESA | -0.36 | (-7.5) | 0.37 | (1.5) | 1.62 | (4.0) |
| NA | -0.17 | (-2.2) | 0.46 | (2.3) | 0.38 | (1.0) |
| WCA | -0.36 | (-7.5) | 1.09 | (4.1) | 1.96 | (4.5) |

Table 6.3 Regional capital-output ratio parameters

(b) Capital flight from Africa

The values used in relation to this analysis are based on those of Hermes and Lensink (1992). In their econometric estimation using an 'exclusive' definition of capital flight (Vos 1990),²⁰ they report the following results, which have similar regressors as the equation specified in our model.

$$\Delta J/Y_s = -0.06 + 0.92 (\Delta F_s/Y_s) + 0.02 REXH84 - 0.02 FINC$$

(-3.57) (10.97) (2.83) (-0.076)
n = 66 adj R² = 0.66 F = 25.31

where: ΔJ = capital flight; Y = GDP; ΔF_s = annual flow of long-term debt; *FINC* = exchange rate adjusted interest rate differential between domestic and US financial asset; and *REXH84* = the overvaluation of the effective exchange rate with 1984 as equilibrium year.

Vos' re-estimation of the same data, with the aim of determining the coefficients of stock adjustment (Vos 1994: 339) resulted in marginal coefficients ranging from 0.18 to 0.24, with a constant term lying between -0.03 and 0.01. Based on the latter, the constant parameter 0.01 and the marginal coefficient of 0.24 are used. The findings of Hermes and Lensink (1992) in relation to the impact of capital inflows are also worth noting. However, my own estimation – using stock of capital-based outflows, generated using flow data running from 1970, but excluding the financial incentive variable for 18 African countries – indicates an insignificant coefficient for the capital inflow variable. The stock adjustment coefficient also lies in the range of 0.05 to 0.11. However, owing to the possible problem with my stock data formulation, I have opted to use the estimation based on Hermes and Lensink (1992), re-computed in Vos (1994: 339).

 $\Delta J_A / Y_A = 0.01 + 0.24 (J_A / Y_A)_{t-1}$ (0.19) (6.38) R² = 0.36 F = 107.7 n = 65 (t-values in brackets)

6.5.7.2 Estimated parameters of the model

Assuming no specification problem a system estimation for all equations would ideally have rendered efficient estimates. However, this entails a major problem for a model of our size. Hence, a more manageable, three-step approach is adopted in this study. Firstly, a number of essential equations, such as those relating to the global commodity market, are simultaneously estimated. Secondly, the potential use of bloc recursivity is exploited when that is possible. (For example, the result of the global commodity sub-bloc is incorporated into the African supply of commodities, as discussed in Chapter 4.) And finally, for other estimations, resort is made to a single equation error correction model (ECM).²¹

Unless otherwise expressed, all the estimations are undertaken using a log-log model. Thus, coefficients reported in Table 6.4 below may be interpreted as elasticities. The econometric estimation of the model starts from a second-order ECM. However, when the second-order ECM produces similar results to that of its first-order variant, I have opted for its first-order form, on the grounds of parsimony. All parameters of the model are reported in Table 6.3. When the coefficient reported is not an estimated one, this is noted. In general, two estimated coefficients for one explanatory variable are given. These values refer to the short-run (change, column 2) and long-run (level, column 4) values for the regressor in question. The lag length of the regressors is reported in columns 3 and 5. Column 6 provides remarks about the degree of the ECM and other relevant information. The following example illustrates how Table 6.4 may be read.

In the main text of the model, detailed in this section, each equation is set out in a standard format. Thus, equation number 2, is given by,

$$C_N = a_2 + b_2 Q_N$$

The first-order ECM estimation for this equation is formulated after transforming C_N and Q_N into logarithmic form and rewriting it as

$$\Delta \ln C_N = a_2 + b_2 \Delta \ln Q_N + b_2^* \ln Q_{N_{l-1}} - c_2 \ln C_{N_{l-1}}$$

Thus, in Table 6.4, the first column lists all coefficients of the regressors, including the lag-dependent coefficient, given by c_2 in the above example. Column two summarizes the short-run values of these regressors (b_2 in the example above), with the level of lags listed in the third column. The fourth column gives the long-run value of the regressors (b^*_2 in the above example) and fifth column lists the level of lag for the latter (-1 in the example above). The last coefficient (c_2) gives the error correction term, which is labelled as lag dependent in the table.

| Parameters | Short-I | run | Long-r | run | Remark/reference |
|------------------------|------------------|-----|--------|-----|---|
| Parameters | Value | Lag | Value | Lag | on the estimation |
| a2 | 16.40 | | | | OLS estimation from Mar- guez & Pauly (1987) model |
| b ₂ | 0.60 | | | | |
| a 3 | 0.27 | | | | OLS estimation; see App. 6.1 |
| b3 | 0.89 | | | | |
| C3 | -0.09 | | | | |
| a ₆ | 1.50 | | | | OLS estimation from Mar- guez & Pauly (1987) model |
| <i>b</i> 6 | 0.09 | | | | |
| C6 | 0.33 | | | | |
| d ₆ | 0.35 | | | | |
| 0 6 | -0.16 | | | | |
| a ₈ | 0.50 | | | | average ratio; see App. 6.1 |
| b ₈ | 0.48 | | | | |
| C ₈ | 1.42 | | | | |
| d ₈ | 3.50 | | | | |
| a, | -2.60 | | | | OLS estimation; see App. 6.1 |
| D ₉ | -0.58 | | | | |
| C9 | 0.89 | | | | |
| 0 ₉ | 0.30 | | | | see Chapter 4, section 4.2.2 |
| a12 | 0.00 | | 1 01 | 2 | see Chapter 4, section 4.2.2 |
| D12 C12 | | | -0.51 | -2 | |
| Lag depend | | | -0.51 | _1 | |
| ang dopond. | 4 40 | | -0.00 | - • | see Chapter 4, section 4.2.2 |
| b ₁₂ | -2.04 | -1 | | | |
| f ₁₂ | -0.13 | -1 | | | |
| a ₁₂ | 5.50 | | | | |
| b ₁₂ | -1.56 | -1 | -2.29 | -2 | see Chapter 4, section 4.2.2 |
| d ₁₂ | 2.02 | | | | |
| e ₁₂ | 3.13 | -1 | | | |
| Lag depend. | | | -0.86 | -1 | |
| a ₁₃ | -1.25 | | | | see Chapter 4, section 4.2.1 |
| D ₁₃ | -5.50 | -1 | -0.17 | -2 | |
| C13 | o 7 0 | | -1.50 | -2 | |
| <i>d</i> ₁₃ | 0.73 | | 4.00 | | |
| Lag depend. | 0.62 | | -1.00 | | and Charter 4 contine 4.0.4 |
| a ₁₃ | 0.63 | | 0.66 | 4 | see Chapter 4, section 4.2.1 |
| 013 Cin | | | -0.00 | -1 | |
| Lag depend | | | -1.11 | -1 | |
| aug depend. | -0.2 | | 0.71 | -1 | see Chapter 4 section 4.2.1 |
| b13 | -0.2 | | -0 19 | -1 | 000 chuptor 4, 000ton 4.2.1 |
| - 13 C13 | | | -0.09 | -1 | |
| - 10 | | | 0.00 | • • | 1 |

Table 6.4 Estimated parameters of the model

The North (N)

| | Short-ı | run | Long-r | un | Remark/reference |
|--------------------------|---------|-----|--------|-----|-----------------------------|
| Parameters | Value | Lag | Value | Lag | on the estimation |
| Lag depend. | | | -0.82 | -1 | |
| a₁₄ (i=Agr.) | -3.70 | | | | see Chapter 4, section 4.4 |
| b14 | 0.38 | | 0.01 | -1 | |
| C14 | -0.15 | -1 | -0.25 | -2 | |
| d14 | -0.17 | | -0.48 | -1 | |
| Lag depend. | | | -0.67 | -1 | |
| a₁₅ (i=Agr.) | -0.01 | | | | from Maximum Likelihood (FI |
| b15 | 0.08 | -1 | | | estimation, see App. 4.3 |
| C15 | 0.30 | -1 | | | |
| a ₁₄ (i=Bev.) | | | | | see Chapter 4, section 4.4 |
| b14 | 0.90 | | 0.25 | -1 | |
| C14 | -0.003 | -1 | 0.09 | -2 | |
| d ₁₄ | -0.32 | | -0.19 | -1 | |
| Lag depend. | | | -0.78 | -1 | |
| a ₁₅ (i=Bev.) | 3.6 | | | | see Chapter 4, section 4.4 |
| b15 | 0.003 | -2 | 0.06 | | |
| C15 | -0.15 | -5 | -0.57 | | |
| Lag depend. | | | -0.59 | -1 | |
| a ₁₄ (i=Food) | -0.86 | | | | see Chapter 4, section 4.4 |
| b14 | 0.36 | | 0.23 | -1 | |
| C14 | -0.19 | -1 | -0.02 | -2 | |
| d14 | -0.92 | | -0.39 | -1 | |
| Lag depend. | | | -0.39 | -1 | |
| a ₁₅ (i=Food) | 2.4 | | | | see Chapter 4, section 4.4 |
| b ₁₅ | -0.16 | -1 | 0.13 | -2 | |
| C15 | -0.74 | -2 | -0.13 | -3 | |
| Lag depend. | | | -0.52 | -1 | |
| a ₁₄ (i=Min.) | | | | | see Chapter 4, section 4.4 |
| b14 | -0.42 | | 0.21 | -1 | |
| C14 | -0.01 | -1 | -0.11 | -2 | |
| d. | -0.31 | | -0.34 | -1 | |
| Lag depend | 0.01 | | -0.53 | -1 | |
| a ₁₅ (i=Min.) | 1.62 | | 0.00 | • | see Chapter 4, section 4.4 |
| b15 | 0.07 | | 0.09 | -2 | |
| C15 | 0.49 | -5 | 0.48 | -6 | |
| Lag depend | 0.40 | , s | -0.80 | -1 | |
| | | | 0.00 | • | |

| The | North | (N) |
|-----|-------|-----|
| | | • • |

| | Short- | run | Long-I | un | Remark/reference |
|--------------------------|--------|-----|--------|-----|-------------------------------|
| Parameters | Value | Lag | Value | Lag | on the estimation |
| a ₁₈ | | | | | see Chapter 5, section 5.4 |
| b ₁₈ | | | 0.02 | | |
| C18 | 0.62 | | 0.20 | -1 | |
| Lag depend. | | | -0.21 | -1 | |
| a ₁₉ | 2.9 | | | | |
| b ₁₉ | | | -0.11 | -1 | see Chapter 5, section 5.4 |
| C19 | 0.68 | | 0.51 | -1 | |
| d ₁₉ | 0.56 | | 0.21 | -2 | |
| Lag depend. | | | -0.25 | -1 | |
| a ₂₁ | -0.80 | | | | First-order ECM, see App. 6.1 |
| D ₂₁ | -0.92 | | -0.25 | -1 | |
| C ₂₁ | | | 0.05 | -1 | |
| Lag depend. | 0.05 | | -0.70 | -1 | First order ECM acc App 6.1 |
| a22 | 0.85 | | 0.02 | 4 | First-older ECM, see App. 6.1 |
| D ₂₂ | -0.25 | | -0.03 | - 1 | |
| C ₂₂ | 0.74 | | 0.07 | -1 | |
| U ₂₂ | 0.03 | | 0.03 | -1 | |
| Lag uepenu. | 0.60 | | -0.44 | -1 | see Chanter 5 section 5.2 |
| a24 | 1 10 | | 0.31 | 1 | |
| D24 | 1.10 | | -0.08 | -1 | |
| Lag depend | | | -0.00 | -1 | |
| Lag depend. | -0.70 | | -0.40 | - 1 | |
| has | 0.46 | | 0.13 | -1 | see Chapter 5, section 5,2 |
| D25 Cos | 0.40 | | 0.15 | -1 | |
| da: | 0.12 | | -0.01 | -1 | |
| Lag depend. | | | -0.21 | -1 | |
| 827 | -0.40 | | 0.21 | • | see Chapter 5, section 5.2 |
| depend.(ecm2) | | | -0.49 | -1 | |
| b ₂₇ | 0.58 | | | | |
| b ₂₇ | 0.56 | -1 | 0.51 | -2 | |
| C ₂₇ | 0.01 | | 0.01 | -2 | |
| Lag depend. | | | -0.53 | -2 | |
| a ₂₉ | 0 | | | | First-order ECM, see App. 6.1 |
| b ₂₉ | 0.96 | | 0.45 | -1 | |
| Lag depend. | | | -0.73 | -1 | |
| and (i=Agr.) | 0.35 | | 0.70 | | see Chapter 4, section 4.3 |
| b33 | 0.33 | -1 | -0.03 | -2 | |
| C33 | 0.18 | -2 | 0.17 | -3 | |
| Lag depend. | | - | -0.14 | -1 | |
| a33 (i=Bev.) | 0.40 | | | | see Chapter 4, section 4.3 |
| b33 | -0.05 | -2 | | | • • |
| C ₃₃ | | | -0.06 | | |
| Lag depend. | | | -0.41 | -1 | |
| a ₃₃ (i=Food) | 0.16 | | | | see Chapter 4, section 4.3 |
| | | | | | |

East & Southern Africa (ESA)

| | Short-run | | Long-r | un | Remark/reference |
|--------------------------|-----------|-----|--------|-----|----------------------------|
| Parameters | Value | Lag | Value | Lag | on the estimation |
| b ₃₃ | | | 0.04 | | |
| Lag depend. | | | -0.34 | -1 | |
| a ₃₃ (i=Min.) | 0.20 | | | | see Chapter 4, section 4.3 |
| b33 | 0.23 | -2 | | | |
| C33 | | | 0.43 | -6 | |
| Lag depend. | | | -0.18 | -1 | |
| a ₃₄ | | | | | ECM, see App. 6.1 |
| depend.(ecm2) | | | 0.60 | -1 | |
| b34 | 0.004 | -1 | 0.01 | -2 | |
| C ₃₄ | 0.14 | | 0.02 | -2 | |
| C ₃₄ | -0.10 | -1 | | | |
| Lag depend. | | | -0.02 | -2 | |

East & Southern Africa (ESA)

North Africa (NA)

| Descenations | Short-run | | Long-r | un | Remark/reference |
|-----------------|-----------|-----|--------|-----|-------------------------------|
| Parameters | Value | Lag | Value | Lag | on the estimation |
| a ₁₈ | -0.40 | | | | see Chapter 5, section 5.4 |
| b ₁₈ | | | -0.02 | -1 | |
| C ₁₈ | | | 0.11 | -1 | |
| Lag depend. | | | -0.25 | -1 | |
| a19 | -0.20 | | | | see Chapter 5, section 5.4 |
| b ₁₉ | 0.29 | | | | |
| C19 | | | 0.12 | -1 | |
| d ₁₉ | 1 | | -0.17 | -2 | |
| Lag depend. | | | -0.19 | -1 | |
| a ₂₁ | -1.22 | | | | First-order ECM, see App. 6.1 |
| b ₂₁ | -0.97 | | -0.14 | -1 | |
| C ₂₁ | | | | | |
| d ₂₁ | } | | 0.16 | -1 | |
| Lag depend. | | | -0.64 | -1 | |
| a22 | 3.33 | | | | First-order ECM, see App. 6.1 |
| b22 | -0.55 | | 0.33 | -1 | |
| C ₂₂ | 0.43 | | | | |
| Lag depend. | | | -0.52 | -1 | |
| a23 | -0.40 | | | | First-order ECM, see App. 6.1 |
| b ₂₃ | -0.71 | | 0.19 | -1 | |
| Lag depend. | | | -0.50 | -1 | |
| a24 | 0.56 | | | | see Chapter 5, section 5.2 |
| b ₂₄ | | | 0.45 | -1 | |
| C ₂₄ | -0.15 | | -0.05 | -1 | |
| Lag depend. | | | -0.60 | -1 | |

| Domonostono | Short-i | un | Long-r | un | Remark/reference |
|--------------------------|---------|-----|--------|-----|----------------------------|
| Parameters | Value | Lag | Value | Lag | on the estimation |
| a ₂₅ | -2.10 | | | | see Chapter 5, section 5.2 |
| b ₂₅ | | | | | |
| C ₂₅ | 0.92 | | 0.36 | -1 | |
| d ₂₅ | | | -0.04 | -1 | |
| Lag depend. | | | -0.48 | -1 | |
| a ₂₇ | -0.05 | | | | see Chapter 5, section 5.2 |
| depend.(ecm2) | | | -0.54 | -1 | |
| b ₂₇ | 0.52 | | 0.32 | -2 | |
| b ₂₇ | 0.38 | -1 | | | |
| C ₂₇ | 0.08 | -1 | 0.02 | -2 | |
| Lag depend. | | | -0.33 | -2 | |
| a ₃₃ (i=Agr.) | -2.05 | | | | see Chapter 4, section 4.3 |
| b33 | 0.63 | -1 | 0.52 | -2 | |
| C ₃₃ | 0.61 | -1 | 0.12 | -2 | |
| Lag depend. | | | -0.22 | -1 | |
| a ₃₃ (i=Food) | 0.20 | | | | see Chapter 4, section 4.3 |
| b ₃₃ | -0.52 | -1 | 0.31 | -2 | |
| C33 | | | -099 | -1 | |
| Lag depend. | | | | | |
| a ₃₃ (i=Min.) | -1.04 | | | | see Chapter 4, section 4.3 |
| b ₃₃ | 0.48 | -1 | 0.39 | -2 | |
| b33 | | | 0.59 | -6 | |
| Lag depend. | | | -0.62 | -1 | |
| 8 ₃₄ | -0.40 | | | | ECM, see App. 6.1 |
| b34 | -0.003 | -1 | 0.05 | -2 | |
| C34 | 0.54 | | 0.18 | -1 | |
| Lag depend. | | | -0.08 | -1 | |

North Africa (NA)

West & Central Africa (WCA)

| Parameters | Short-run | | Long-run | | Remark/reference |
|-----------------|-----------|-----|----------|-----|-------------------------------|
| | Value | Lag | Value | Lag | on the estimation |
| a ₁₈ | -0.50 | | | | see Chapter 5, section 5.4 |
| b ₁₈ | 0.27 | | 0.17 | -1 | |
| C18 | 0.64 | | 0.31 | -1 | 1 |
| Lag depend. | | | -0.33 | -1 | |
| a19 | 4.00 | | | | see Chapter 5, section 5.4 |
| b ₁₉ | 0.28 | | 0.50 | -1 | |
| C ₁₉ | 0.74 | | 1.15 | -1 | |
| d19 | | | -0.04 | | |
| Lag depend. | | | -0.39 | -1 | |
| a21 | 0.10 | | | | First-order ECM, see App. 6.1 |
| b ₂₁ | -1.10 | | -0.10 | -1 | |

| | Short- | un | Long-r | un | Remark/reference |
|--------------------------|--------|-----|--------|-----|-------------------------------|
| Parameters | Value | Lag | Value | Lag | on the estimation |
| C ₂₁ | 0.03 | | 0.02 | -1 | |
| Lag depend. | | | 0.18 | -1 | |
| a ₂₂ | 1.80 | | | | First-order ECM, see App. 6.1 |
| b22 | -1.01 | | -0.03 | -1 | |
| C ₂₂ | 0.33 | | 0.06 | -1 | |
| d ₂₂ | 0.04 | | 0.03 | -1 | |
| Lag depend. | | | -0.50 | -1 | |
| a23 | -1.60 | | | | First-order ECM, see App. 6.1 |
| b ₂₃ | | | 0.54 | -1 | |
| C ₂₃ | 0.17 | | | | |
| Lag depend. | | | -0.30 | -1 | |
| a ₂₄ | -1.00 | | | _ | see Chapter 5, section 5.2 |
| D ₂₄ | 1.41 | | 0.49 | -1 | |
| C ₂₄ | | | -0.09 | -1 | |
| Lag depend. | | | -0.38 | -1 | |
| a ₂₅ | -0.40 | | | | see Chapter 5, section 5.2 |
| b ₂₅ | 0.63 | | 0.20 | -1 | |
| d ₂₅ | 0.02 | | 0.01 | -1 | |
| Lag depend. | | | -0.31 | -1 | |
| a ₂₇ | -0.20 | | | | see Chapter 5, section 5.2 |
| depend. (ecm2) | | | -0.25 | -1 | |
| D ₂₇ | 0.85 | | 0.15 | -2 | |
| b ₂₇ | 0.36 | -1 | | - | |
| C ₂₇ | 0.02 | | 0.03 | -2 | |
| C ₂₇ | 0.02 | -1 | | | |
| Lag depend. | | | -0.22 | -2 | |
| a ₂₉ | -0.70 | | | | First-order ECM, see App. 6.1 |
| D ₂₉ | 1.55 | | 0.35 | -1 | |
| Lag depend. | | | -0.33 | -1 | |
| a ₃₃ (I=Agr.) | -0.50 | | | - | see Chapter 4, section 4.3 |
| D ₃₃ | 0.48 | -1 | 0.08 | -2 | |
| Lag depend. | | | -0.12 | -1 | |
| a ₃₃ (I=Bev.) | -1.00 | | | | see Chapter 4, section 4.3 |
| D ₃₃ | 0.18 | -1 | 0.13 | -2 | |
| D ₃₃ | 0.13 | -5 | 0.07 | -6 | |
| C ₃₃ | | | 0.06 | -6 | |
| Lag depend. | | | -0.26 | -1 | |
| a ₃₃ (i=Food) | -0.65 | | | | see Chapter 4, section 4.3 |
| b33 | 0.27 | | 0.03 | -2 | |
| C ₃₃ | 0.18 | -1 | 0.06 | -2 | |
| Lag depend. | | | -0.15 | -1 | |
| a ₃₃ (i=Min.) | -0.3 | | | | see Chapter 4, section 4.3 |
| b ₃₃ | | | -1.28 | -2 | |
| b ₃₃ | | | 0.39 | -6 | |
| a ₃₄ | | | | | ECM, see App. 6.1 |
| depend. (ecm2) | | | 0.38 | ļ | |

West & Central Africa (WCA)

| Parameters | Short-run | | Long-run | | Remark/reference |
|-----------------|-----------|-----|----------|-----|-------------------|
| | Value | Lag | Value | Lag | on the estimation |
| b ₃₄ | 0.02 | | 0.005 | -2 | |
| b34 | -0.02 | -1 | | | |
| C34 | 0.22 | -1 | 0.03 | -2 | |
| Lag depend. | | | -0.02 | -2 | |

West & Central Africa (WCA)

OPEC and International Banking

| Demonsterer | Short-run | | Long-run | | Remark/reference |
|-----------------|-----------|-----|----------|-----|----------------------------|
| Parameters | Value | Lag | Value | Lag | on the estimation |
| m _{NO} | 0.00038 | | | | Average using data 1983-90 |
| m _{so} | 0.0008 | | | | Average using data 1983-90 |
| mon | 0.22 | | | | Average using data 1983-90 |
| 8 ₄₁ | -1.38 | | | | See section 8.8.1b above |
| b41 | 6 | | 0.27 | -1 | ECM, see App. 6.1 |
| C41 | -0.02 | | | | |
| Lag depend. | | | -0.41 | -1 | |
| a42 | 0.01 | | | | Vos (1994), see section |
| b42 | 0.24 | | | | 6.5.7.1b above |
| a 43 | 18 | | | | |
| b43 | 0.34 | | | | Average using data 1983-90 |
| μs | 0.40 | | | | Vos (1994) |
| λ* | 30% | | | | Assumed |

Note: The values of coefficients are estimated results without any sign change. The sign in the model specification is theoretical expected sign. Thus, in equation 3 the interest rate coefficient is $-c_3 = -0.09$.

Glossary of symbols used

| N | North |
|----------------------------|--|
| S | South |
| А | Africa |
| ESA | East and Southern Africa |
| NA | North Africa |
| WC | West and Central Africa) |
| OT | Other South (excluding Africa) |
| λ | Debt service ratio |
| τ | Mark-up rate in North (Price) |
| δ_{AP}, δ_{Ag} | Output-capital ratio of the private and public sector, respectively |
| ΔB_A | Net commercial bank credit flow to Africa |
| $\Delta B^{\hat{d}}_{nN}$ | North private sector demand for bank credit |
| ΔB_{OT} | Net commercial bank credit flow to Other (non-African) South |
| ΔB_N^{S+pN} | Net commercial bank credit flow to South and North's private sector |
| ΔDC_{M} | Northern government debt (flow) |
| ΔF_A | Foreign inflows (net of commercial bank lending and FDI) to A frica |
| ΔF_{4-} | Foreign inflow to the public sector of A frica |
| ΔF_{AB} | Foreign inflow to the private sector of A frica |
| ΔF_{OT} | Foreign inflows (net of commercial bank lending and FDI) to Other (non-Africa) South |
| ΔF_S | Foreign inflows (net of commercial bank lending and FDI) to South |
| $\Delta J_{O, A,,OT, N}$ | Flow of saving deposit in international banks by OPEC, Africa Other South and North |
| μ_S | Bank mark-up rate over world interest rate on Southern borrowers |
| b_N | The labour output ratio in North |
| C_N | Consumption in North |
| C_p | Private consumption in Africa |
| ĊOILA | Consumption of oil in Africa |
| $COIL_N$ | Consumption of oil in North |
| DC_N | Northern government outstanding debt |
| | |

| Chapter | 6 |
|---------|---|
|---------|---|

| DEF_N | Deficit in North |
|----------------------------------|--|
| е | Nominal exchange rate (Africa) |
| DSR OT, A | Debt service ratio of Other South and Africa |
| | respectively |
| EXCAP | Excess capacity in North |
| F_A | Stock of foreign inflows (net of commercial bank |
| | lending and FDI) to Africa |
| $FDI_{S}, FDI_{A,OI}$ | Direct foreign investment, FDI to South, Africa and |
| | Other South |
| F_{OT} | Stock of foreign inflows (net of commercial bank |
| | lending and FDI) to Other (non-Africa) South |
| fpS | Factor payments for South |
| F_S | Stock of foreign inflows (net of commercial bank |
| | lending and FDI) to South |
| G_A | Government consumption expenditure in Africa |
| GDP_{mi} | GDP in mining sector |
| G_N | Government expenditure in North |
| H^{d}_{Ni} | Commodity stock demand (North) |
| $H^{s}{}_{Ni}$ | Commodity stock supply (North) |
| I _{A, Ag, Ap} | Investment in Africa, Africa public sector and private |
| | sector, respectively |
| I_N | Investment in North |
| I_S | Investment in South |
| Y_S | GDP in South |
| i _w | World interest rate |
| i _w * | World interest rate on concessional |
| Ј _{О, А,ОТ, N} | Saving deposit in international banks by OPEC, Africa, |
| | Other South and North |
| K _{Ag} , _{Ap} | Capital stock in public and private sectors in Africa |
| ММ ^{Јg} _{О, А} | Import of manufactured goods by OPEC and Africa, |
| | respectively. |
| M_N, M_{OT} | Import of North and other South respectively |
| m_{No} | Import propensity of oil consumption in North |
| m_{Ns} | Import propensity of oil consumption in South |
| m_{oN} | import propensity of OPEC for Northern goods |
| MO^o_{N} | Import of oil by North |
| MR [°] _{Ni} | Import of primary commodities by North from Africa |
| | i = 14 |
| MUP | Manufactured unit price index |
| | |

| othDSi | Other debt service component (discrepancy) |
|-------------------------------|---|
| otGR _A | Other government revenue |
| otM_N | Other imports of North |
| otX_N | Other exports of North |
| P _{coal} | Price of coal |
| $PM^{fg}{}_N$ | Price of manufactured goods of North |
| Poil | Price of oil |
| PO_A | Domestic price in Africa |
| $P\widetilde{Q}_N$ | Domestic price in the North |
| $P\widetilde{R}_i$ | Export price of primary commodities from Africa |
| Prin _{Ai} | Principal payment by Africa |
| O_N, O_{OT}, O_A | Output in North, other South and Africa, respectively |
| Qoiln | Output of oil in North |
| \tilde{o} | Growth rate of A frican output |
| $\tilde{S}_{4\pi}$ $S_{4\pi}$ | Saving of A frica public and private respectively |
| SN SA | Saving of North & Africa (total public and private) |
| $\omega_N, \omega_A, sg, sp$ | respectively |
| ty | Tax rate in North |
| T , $T^{d, iex}$, | Total revenue Direct and indirect tax revenue |
| -A, - A | respectively in A frica |
| Way | Wage rate in North |
| X^{dd} s: | Global supply of primary commodities from South |
| X_{N} | Export of the North |
| X_{4} | Export of A frica (total) |
| XOA | Export of oil by Africa (total) |
| XONA XOWC | Export of oil by North Africa and West and Central |
| <i>MA</i> , <i>W</i> C | Africa, receptively |
| XR Ai | Export Supply of primary commodities from Africa |
| X ^{ss} _{Si} | Global demand for primary commodities from South |
| $Y^{p}_{OT,A}$ | Per capita income of other South and Africa |
| 01,11 | respectively |
| Z_A | Total resource transfer from Africa |
| Z_B | Public borrowing from banks |
| Z_{pr} | Resource transfer from private to the public sector |
| - | |

The Effect on Africa ofExternal Shocks Generated in Developed Countries

7.1 INTRODUCTION

In this chapter, an attempt will be made to assess the impact of a range of global shocks on African economies, using the model specified in Chapter 6. The chapter does not exhaust the full potential of AFRIMOD for the analysis of external shocks and economic policies. Rather, the model will be used to examine a number of crucial external shocks and policy simulations. The chapter is organized as follows. Section 7.2 describes the accounting framework used in the model. This framework serves both as a consistent database of the world and as a base run value in the calibration of the model. Section 7.3 briefly discusses the model solution algorithm employed, the stability conditions of the model and calibration procedure followed. In section 7.4 the results of selected external shocks and policy simulations are discussed. Finally, section 7.5 brings the chapter to a close by highlighting a number of conclusions arising out of the preceding discussion.

7.2 WORLD ACCOUNTING MATRICES (WAM) AS DATABASE FOR AFRIMOD

Vos (1988, 1989a), Luttik (1992) and de Jong and Vos (1995b) have developed a global accounting framework, which organizes trade and financial data by source and destination. This framework, termed the World Accounting Matrix (WAM), consists of four sub-matrices. These are labelled (1) current, (2) capital account transaction by origin and destination, (3) gross domestic investment, and (4) gross national saving. These sub-matrices are supplemented by an assessment of stocks of international assets and liabilities as well as GNP (see Diagram 7.2a below).

| | 1 | 2 | | n | | 1 | 2 | | n | |
|--------|------------------------------------|------------------------------------|-----|------------------------------------|----|------------------------------------|------------------------------------|---|----------------|----------------|
| 1 2 | A ₁₁ A ₂₁ | A ₁₂ A ₂₂ | | A _{1n} A ₂₁ | X1 | l ₁ | I2 | | | I1 |
| n | A _{n1} | A _{n2} | | Ann | Xm | In | | • | | I _n |
| | M ₁ | | | Mn | Ст | l ₁ | | | l _n | I |
| | | | | | | - | | | | |
| 1 2 | S1 | S₂ | | | S1 | B ₁₁ B ₂₁ | B ₁₂ B ₂₂ | | B1n B21 | C1 |
| n | | | • | Sn | Sn | B _{n1} | B _{n2} | | Bnn | Cn |
| | S ₁ | | ••• | Sn | S | D ₁ | | | Dn | F |

Diagram 7.2a A World Accounting Matrix for n countries

Source: Luttik (1992: 34).

The WAM approach assumes the existence of a direct link between current and capital account flows, the domestic resource gap and balance of payments. Employed within a global context, this implies that world export of goods, services and transfers should equal world imports of goods, services and transfers, and, thus, that the world current account should sum to zero. World foreign saving should also sum to zero, since some countries will be in surplus, while others in deficit, in each accounting year. Similarly, global saving should equal global investment. For the world economy, comprising *i* countries, this may be summarized¹ as follows (see Vos 1988 & 1989a, Luttik 1992, de Jong & Vos 1995b).

$$S_i - I_i \equiv E_i - M_i + R_i \equiv \Delta F A_i - \Delta F L_i + (\Delta R E S_i + E O_i)$$
[7.1]

$$\sum_{i}^{n} (S_{i} - I_{i}) \equiv \sum_{i}^{n} (E_{i} - M_{i} + R_{i}) \equiv 0$$
[7.2]

$$\sum_{i}^{n} \left[(\Delta FA_i - \Delta FL_i) + (\Delta RES_i + EO_i) \right] \equiv 0$$
[7.3]

where: S = gross national saving, I = gross domestic investment, E = export of goods and non-factor services, M = import of goods and non-factor services, R = net factor income and current transfer from abroad, FA = change in total external financial assets. FL = change in total

FA = change in total external financial assets, FL = change in total external financial liabilities, RES = change in reserves, EO = errors and omissions.

The WAM rests on equation [7.1], which is the basic accounting framework for each country. For the world economy, the conditions under equations [7.2] and [7.3] will then follow logically.

These equations form the basic identities for the world economy. Based on these relationships, countries and regions are linked within the WAM through trade and financial matrices, organized by origin and destinations (Luttik 1992: 22). This WAM framework in matrix format is set out in Diagram 7.2a. The Northeast quadrant shows gross domestic investment, the Northwest quadrant summarizes the current account, the Southwest quadrant, national savings and the Southeast quadrant, the flow of funds account. These flow categories may be combined with satellite matrices/vectors and applied to stock data (see Luttik 1992 for details).

The WAM improves upon existing global accounting frameworks in a number of specific respects. Firstly, it maintains consistency at a global level; secondly, it explicitly allows for aggregation or disaggregation within its framework; and thirdly, it shows the origin and destination of transactions. Owing to these attributes, the WAM may usefully serve as a database for the analysis of trade and finance within a global framework.² Hence, in this study, this framework, with its 1990 value, will be used to adjust our regional data in order that it may accord with the 1990 WAM format.

Table 7.2b shows the WAM for 1990. The WAM values summarized in this table will be used as the accounting framework for the model developed in Chapter 6. The year 1990 is taken as the base year, since this is the latest year for which a WAM is available. The use of this WAM ensures global consistency, as explained in equations [7.1]– [7.3]. Thus, the final solution of the model is calibrated around the 1990 WAM values. The African data in the WAM is grouped according to whether it relates to North Africa (NA) or Sub-Saharan Africa (SSA). The NA data is used directly, while an adjustment is made to the SSA data. Thus, in relation to SSA, a two-step process is adopted. Firstly, regional data for all countries within WCA and ESA regions is compiled, based on the UN-ECA classification. Then, since this regional data obviously does not tally with the WAM-based data, the WAM-based SSA data is allocated between WCA and ESA regions, based on the proportion of the two derived from the regional data compiled.

7.3 MODEL SOLUTION ALGORITHM AND STABILITY CONDITIONS

7.3.1 Model Solution

The model is solved with the Guass-Seidel algorithm, using EVIEWS software. Guass-Seidel is a fairly standard, and widely used algorithm for large-scale simultaneous equation models. The beauty of Guass-Seidel lies in its ability to render a solution without going through the difficult processes of derivatives, matrix inversion and similar procedures usually required in arriving at a numerical solution. Such procedures are extremely difficult in large-scale models. Rather, Guass-Seidel employs an iterative scheme, starting from an initial guess, until some specified tolerance level is reached. Obtaining convergence will usually depend on a variety of factors (see Heien et al. 1973, Press et al. 1989 for more details on these).³

Having solved the model using this algorithm, the solution for the year 1990 is calibrated using only the constants, in order to yield the actual values for 1990.⁴ Each endogenous equation is included in the model in its full ECM dynamic form. For almost all equations the solution reproduces substantially closer values to that of the base run. In order to ensure a full reproduction of the base run, the constant values for each endogenous equation are calibrated. This method of calibration maintains the slope coefficients, which are the behavioural and structure indicators, intact. The latter coefficients are assumed to be more important than the constant terms.

| of US\$) |
|------------|
| billions c |
| (in |
| r 1990 |
| 9 (|
| (WAM) |
| Matrix |
| Accounting |
| World |
| Balanced |
| Table 7.2b |

| | | | | | Current | Account | | | | | | |
|------------------|---------|---------|--------|--------|---------|---------|---------|-------|-------|--------|-------|---------|
| | NSA | JAP | GERM | ¥ | opc | EEU | LDCot | ٩N | SSA | OBC | INT | TOTAL |
| Current Account | | | | | | | | | | | | |
| NSA | 0.00 | 106.56 | 40.00 | 60.74 | 248.58 | 6.03 | 169.53 | 6.51 | 5.63 | 22.42 | 2.09 | 668.09 |
| JAP | 103.74 | 0.00 | 24.76 | 27.34 | 107.95 | 4.00 | 110.22 | 2.64 | 6.42 | 25.41 | 9.34 | 421.83 |
| GERM | 51.44 | 17.99 | 0.00 | 52.76 | 305.65 | 19.10 | 52.76 | 5.32 | 4.93 | 6.01 | 6.85 | 522.81 |
| Ę | 67.30 | 39.71 | 39.19 | 0.00 | 184.17 | 6.85 | 33.92 | 2.18 | 8.29 | 6.68 | 2.19 | 390.46 |
| opc | 240.82 | 95.07 | 286.28 | 195.99 | 735.68 | 34.24 | 167.76 | 23.62 | 27.80 | 26.45 | 23.12 | 1856.82 |
| EEU | 4.81 | 4.71 | 16.71 | 3.86 | 37.35 | 38.61 | 26.46 | 1.81 | 1.01 | 0.62 | 0.15 | 136.11 |
| LDCot | 198.40 | 113.95 | 55.43 | 32.71 | 168.90 | 21.64 | 125.65 | 5.56 | 7.25 | 61.46 | 3.63 | 794.56 |
| NA | 6.14 | 1.21 | 6.38 | 2.51 | 29.51 | 1.91 | 6.03 | 1.07 | 0.45 | 0.35 | 5.77 | 61.33 |
| SSA | 13.58 | 2.94 | 5.65 | 2.85 | 30.03 | 0.57 | 3.89 | 0.29 | 4.27 | 0.68 | 2.14 | 66.89 |
| OBC | 40.85 | 17.51 | 10.59 | 13.94 | 17.26 | 1.17 | 53.19 | 0.96 | 1.41 | 8.11 | 3.92 | 168.91 |
| INT | 6.05 | 2.75 | 6.08 | 4.11 | 24.07 | 0.34 | 12.83 | 0.99 | 1.48 | 0.48 | 16.15 | 75.34 |
| Subtotal | 733.12 | 402.40 | 491.07 | 396.81 | 1889.14 | 134.47 | 762.23 | 50.97 | 68.93 | 158.68 | 75.34 | 5163.15 |
| Canital Account | | | | | | | | | | | | |
| USA | 814 00 | | | | | | | | | | | 814 00 |
| JAP | | 993.82 | | | | | | | | | | 993.82 |
| GERM | | | 380.82 | | | | | | | | | 380.82 |
| | | | 10.000 | 167 64 | | | | | | | | 167.64 |
| 000 | | | | | 1164 53 | | | | | | | 1164.53 |
| EEU | | | | | | 318 14 | | | | | | 318.14 |
| LDCot | | | | | | | 766.44 | | | | | 766.44 |
| NA | | | | | | | | 33.68 | | | | 33.68 |
| SSA | | | | | | | | | 17.59 | | | 17.59 |
| OBC | | | | | | | | | | 44.38 | | 44.38 |
| [NT | | | | | | | | | | | 0.0 | 0.00 |
| Subtotal | 814.00 | 993.82 | 380.82 | 167.64 | 1164.53 | 318.14 | 766.44 | 33.68 | 17.59 | 44.38 | 0.00 | 4701.04 |
| Change in reserv | es | | | | | | | | | | | |
| Errors & omissio | ns | | | | | | | | | | | |
| Total | 1547.13 | 1396.21 | 871.89 | 564.45 | 3053.67 | 452.61 | 1528.67 | 84.65 | 86.52 | 203.06 | 75.34 | 9864.19 |
| | | | | | | | | | | | | |

296

Chapter 7

| | | | | | ပီ | apital Acco | ount | | | | | | |
|--|--------|---------|--------|--------|---------|-------------|--------|-------|----------------|-------|------------|---|---|
| | NSA | JAP | GER | ¥ | SOC | EEU | LDCot | NA | SSA | OBC | INI | Subtot | Total |
| Current Account USA JAP JAP JAP JAP OR ODC EEU LDCot LDCot SSA SSA | 879.03 | 974.38 | 349.08 | 173.99 | 1196.85 | 316.50 | 734.11 | 23.31 | 19.63 19.63 | 34.15 | | 879.03 974.38 349.08 173.99 1196.85 316.50 734.11 23.31 23.31 23.31 23.31 23.31 23.31 23.31 23.31 | 1547.13 1396.21 871.89 564.45 3053.67 452.61 1528.67 84.65 86.52 86.52 203.06 |
| INT Subtotal | 879.03 | 974.38 | 349.08 | 173.99 | 1196.85 | 316.50 | 734.11 | 23.31 | 19.63 | 34.15 | 0.0 0.0 | 0.00 4701.04 | 75.34 9864.19 |
| Capital Account | 000 | | 0 | | 10.1 | | | | | | | | |
| ASU | 0.00 | 14.99 | 9.53 | 28.49 | 44.65 | 0.73 | 2.04 | 1.26 | 0.14 | -3.19 | 1.69 | 100.32 | 914.33 |
| JAP | 2.61 | 0.00 | 16.33 | 32.83 | 50.75 | 1.32 | 8.72 | 1.29 | 0.45 | 1.37 | -1.16 | 114.52 | 1108.34 |
| GERM | 3.24 | 0.00 | 0.00 | 21.18 | 31.63 | 0.04 | 2.65 | -0.07 | 0.30 | 3.92 | 1.35 | 64.24 | 445.06 |
| ž | 4.47 | 6.94 | 25.45 | 0.00 | 89.45 | -0.05 | 7.21 | 2.05 | 1.23 | 3.31 | 1.02 | 141.09 | 308.73 |
| ODC | 29.67 | 62.04 | 50.42 | 55.43 | 176.36 | 0.85 | 69.14 | 6.95 | 3.74 | 11.87 | 6.94 | 473.39 | 1637.91 |
| EEU | 0.02 | -0.14 | 4.46 | -2.35 | -1.58 | -0.41 | 0.10 | 0.00 | 0.00 | -0.12 | 0.60 | 0.58 | 318.72 |
| | 20.52 | 16.79 | 1.89 | 3.55 | 16.52 | -0.51 | -0.02 | -0.01 | 0.01 | 2.45 | 9.80 | 70.97 | 837.41 |
| NA | 0.16 | -0.10 | 0.16 | -0.03 | -0.25 | -0.03 | 0.23 | 0.00 | 0.00 | 0.00 | 0.98 | 1.12 | 34.80 |
| SSA | 0.18 | -0.07 | 0.15 | 0.64 | 2.98 | 0.06 | 0.38 | 0.00 | -0.02 | -0.02 | 3.40 | 7.68 | 25.27 |
| OBC | 4.49 | 2.00 | 0.43 | 0.89 | 3.99 | 0.03 | 0.31 | 0.03 | 0.03 | 0.07 | 0.16 | 12.44 | 56.82 |
| INT | 2.08 | 5.08 | 2.80 | 4.08 | 11.01 | 0.13 | 2.57 | 0.28 | 0.20 | 0.30 | 0.62 | 29.16 | 29.16 |
| Subtotal | 67.44 | 107.52 | 111.62 | 144.72 | 425.51 | 2.16 | 93.33 | 11.77 | 6.07 | 19.96 | 25.40 | 1015.52 | 5716.55 |
| Cnange in reserves | -0.19 | -0.34 | -0.68 | 0.01 | 0.08 | 1.01 | -1.52 | -0.36 | -0.21 | -0.07 | 2.28 | 00.0 | 0.00 |
| Errors & | | • | | | | | | | | | | | |
| omissions | -31.95 | 26.77 | -14.96 | -9.98 | 15.48 | -0.95 | 11.49 | 0.07 | -0.22 | 2.77 | 1.48 | 0.0 | 0.00 |
| Total | 914.33 | 1108.34 | 445.06 | 308.73 | 1637.91 | 318.72 | 837.41 | 34.80 | 25.27 | 56.82 | 29.16 | 5716.55 | 15580.75 |

External Shocks Generated in Developed Countries
Notes to Table 7.2b: USA=United States of America, JAP=Japan, GERM=Germany, UK=United Kingdom, ODC=other developed countries (i.e., Canada, France, Italy, Belgium, Luxembourg, Denmark, Greece, Ireland, The Netherlands, Portugal, Spain, Austria, Iceland, Norway, Sweden, Switzerland, Finland, Australia, New Zealand, Israel and South Africa), EEU=Easter Europe, LDCot=Developing Countries excluding Africa, OBC=Official Banking Centers, INT=International Organizations; NA=North Africa and SSA=Sub-Saharan Africa.

Sources: WAM Database (electronic), Money and Finance Research Group, ISS, The Hague, The Netherlands. See also Jong et al (1993) for details of the country classification.

7.3.2 Stability Conditions

The stability condition of the model depends on the four excess demand functions. The first of these, for the commodity market, is a flexiprice market, and, hence, the excess demand in the market is assumed to clear for world commodity price. The second demand function, for the non-traded sector in South, is assumed to clear for domestic price P_S . The third function, for the Northern goods market, clears for quantity of output, Q_N , owing to our assumption of a Keynesian framework. And, finally, the excess demand for bank loans, is assumed, after credit rationing, to clear for world interest rate. Equations [7.8] and [7.9] depict the Northern government bond market and macro determination of imports in South, respectively. These excess demand functions, in their general form, may be given by the following equations (see Vos 1994).

$$\frac{dP_R}{dt} = f \Big[ED_C \left(Q_N, Q_{S_1} i_w, P_R, PQ_A \right) \Big]$$
[7.4]

$$\frac{dQ_N}{dt} = f\left[ED_N\left(Q_N, Q_A, i_w, P_R, PQ_A\right)\right]$$
[7.5]

$$\frac{di_w}{dt} = f\left[ED_B\left(Q_N, Q_A, i_w, P_R, PQ_A\right)\right]$$
[7.6]

$$\frac{dP_A}{dt} = f\left[ED_A\left(Q_N, Q_A, i_w, P_R, PQ_A\right)\right]$$
[7.7]

$$DEF_N(Q_N, i_w, P_R) - \Delta DC_N = 0$$
[7.8]

$$M_{A} - X_{Ai} - \Delta F_{A}^{net} = 0$$
 [7.9]

where: ED_A , ED_B , ED_C and ED_N , indicate excess demand in African non-traded goods, international banking, commodity, Northern goods markets, respectively; *t* is time, and ΔF_A^{net} net capital inflows to Africa (see the glossary of symbols used in Chapter 6).

Vos (1994) has analysed the general linear and nonlinear stability conditions of his STAC model. (This model has similar theoretical underpinnings to the model developed in this book.) Having analysed the trace of the matrix and the determinant of the Jacobian, he concludes that these stability conditions do, in fact, hold (Vos 1994: 305–9). Since AFRIMOD is completely linear, the Jacobian for the model is equal to the determinants of the coefficient matrix.⁵

The existence of a numerical solution for AFRIMOD suggests that the Jacobian is non-singular and, hence, that there is no local stability problem. Nevertheless, in spite of the theoretical analysis of the trace of the matrices (reported in Vos 1994), as well as the numerical exploration undertaken as part of this study, it is worth examining the validity of the model's stability conditions using a structural sensitivity analysis. Such an analysis is usually undertaken in two stages. Firstly, the symmetry and linearity of the model is examined, with a view to assessing the stability of multipliers. And secondly, the sensitivity analysis itself is undertaken (Zellner & Peck 1973: 154-62, Jamshidi 1989: 48–67).⁶ The latter is computationally expensive for large models like AFRIMOD and also requires the use of specially designed software. Hence, such an analysis is not pursued here. However, various runs (using Eviews) are experimented with, in order to assess the sensitivity of the model. The results of this analysis indicate that the model is fairly stable. Additional supporting evidence with respect to the stability of the model is also provided by symmetry and linearity tests for selected variables. These are reported in Table 7.3 below.

Table 7.3 shows the Zellner-Peck test of symmetry (SYM) and linearity (LIN) for the two 'types' of simulations (relating to policy and external shocks) analysed using the model. The result is based on the formula given below (see Zellner & Peck 1973, Jamshidi 1989).

Let the deviation from the base run for a variable y in period t be given by,

 $\delta y_t \Delta = y_{t,\Delta} - y_{t,0}$

where: $y_{t,0}$ is the value of y in period t for the base run and y_t , is value of y in period t after policy or external shock change by units.

Then,

T

$$SYM(\Delta, -\Delta) = \frac{\sum_{t=1}^{T} |\delta y_t, \Delta + \delta y_t, -\Delta|}{\frac{1}{2} \sum_{t=1}^{T} (|\delta y_t, \Delta| + |\delta y_t, -\Delta|)}$$
[7.10]

$$LIN(k\Delta, \Delta) = \frac{\sum_{i=1}^{T} |\delta y_i, k\Delta - k\delta y_i, \Delta|}{\frac{1}{2} \sum_{i=1}^{T} (|\delta y_i, k\Delta| + |k\delta y_i, -\Delta|)}$$
[7.11]

where: k is an amplification factor (assumed to have a value of 2 in the test below), SYM and LIN stand for symmetry and linearity, respectively.

If the model's response is characterized by symmetry and linearity, the value of SYM and LIN coefficients will approximate zero since the numerator will be zero (see Zellner & Peck 1973, Jamshidi 1989). The upper limit is usually difficult to establish. For instance, Jamshidi (1989) took 0.5 as the upper threshold, while Zellner and Peck consider a numerical value of around 0.3 as the threshold. Following the original formulation adopted by Zellner and Peck, the best judgment about the values may be made by conceptualizing the results (SYM and LIN) as percentage deviations for mean absolute distance from the base run.

The first three columns of Table 7.3 report the test results for a Northern policy simulation of a $\pm 5\%$ change in the level of aid to the South. The remaining columns report test results for an external shock simulation of a ± 1 point change in world interest rate. The values reported are results obtained using the formula of symmetry and linearity, given as equations [7.10] and [7.11] for t = 9 (1990–98). As can be read from Table 7.3, these values are, on the whole, close to zero. However, some variation does exist across variables. Thus, the export variables are found to depart from linearity and symmetry, while, for the public deficit, the results show perfect linearity and symmetry. Such disparity is possible because each variable reacts differently, both as a consequence of its own dynamics, as given by the equation which represents it, and due to the general equilibrium effect.⁷ Based on the results summarized in Table 7.3, it is fair to conclude that the model is linear and symmetrical and that it, at least, has local stability. This, in turn, en-

sures the stability of multipliers. However, as noted by McKibbin and Bok (1993), the main problem with many large macroeconometric models is that they are unstable. Indeed, this would appear to be true for a number of the AFRIMOD variables, such as those relating to exports. In such a situation, McKibbin and Bok (1993) suggest a careful constraining of parameter estimates as a solution.

| Policy simulation (aid to the South) | | | External shock simulation (interest rate change) | | |
|--------------------------------------|-------------|--------------------------|---|----------|--------------------------|
| Variable | SYM (±0.05) | LIN (k=2) [on ±0.05] | Variable | SYM (±1) | LIN (k=2) [on ±0.05] |
| PRA | 0.131849 | 0.000685 | PRA | 0.127286 | 0.003372 |
| PR | 0.020638 | 0.000480 | PR | 0.002363 | 0.001555 |
| PR _F | 0.008677 | 0.000652 | PRF | 0.003087 | 0.002829 |
| PRM | 0.112067 | 0.00064 | PRM | 0.198016 | 0.003620 |
| PDEF_ES | 0 | 0 | PDEF_ES | 0 | 0 |
| PDEF_NA | 0 | 0 | PDEF_NA | 0 | 0 |
| PDEF_WC | 0 | 0 | PDEF_WC | 0 | 0 |
| XR_ES | 1.424321 | 1.145035 | XR_ES | 0.062771 | 0.388333 |
| XR_NA | 1.138572 | 1.140491 | XR_NA | 1.175849 | 1.179492 |
| XR_WC | 1.996231 | 1.987624 | XR_WC | 1.924767 | 1.896718 |
| Q_ËS | 0.327078 | 0 | Q_ËS | 0.218054 | 0.001760 |
| Q_NA | 1.128924 | 0 | Q_NA | 0.229353 | 0.000218 |
| Q_WC | 0.82428 | 0 | Q_WC | 0.310863 | 0.000127 |

Table 7.3 Zellner and Peck symmetry and linearity test

Note: PR stands for export price; subscript *A*, *B*, *F* and *M* stand for agr. raw material, beverage, food and mineral, respectively. The suffix *_ES*, *_NA* and *_WC* stand for ESA, NA and WCA regions. *PDEF* is public deficit and Q is real output.

7.4 ANALYSIS OF EXTERNAL SHOCKS AND POLICY SIMULATIONS USING AFRIMOD

This section reports the results of external shocks and policy simulations carried using AFRIMOD. The following simulations are run by invoking a permanent shock in the model for eight consecutive periods, starting from the base year (1990). The eight-year period is chosen arbitrarily, on the assumption that this will reflect the medium-term nature of the model. A further justification for the choice of eight simulation periods is that the maximum lag length in the model is seven years. The effects of this shock are reported as percentage deviations from the base run. The base run simulation is obtained by running the model for the eight consecutive periods without including any policy changes or external shocks. Thus, this base run may be interpreted as representing the effect of the model's own dynamics. Although the dichotomy between the impact of external shock and Northern policy is sometimes blurred, the simulations carried below are categorized using such a dichotomy, if only to simplify the presentation. Three policy shocks (two from the North and one from the South) and three external shock simulations (one each from the North, OPEC and the rest of the developing countries) are conducted.

Policy simulations

(i) Doubling the level of aid from its current level

This simulation is financed by (a) bonds issued in North's market and (b) budget financed (i.e., government expenditure in North is reduced by the amount of aid given to the South). The level of aid to the South in the base run is set at around 0.3 per cent of the GDP of OECD countries. Experience has shown that the DAC target of 1 per cent is found to be practically difficult for most OECD countries. However, the experience of the best performers, such as the Netherlands, could be reproduced by all OECD members if the doubling of the existing level of aid was possible. Hence, the simulation undertaken in this section will examine such a scenario.

(ii) A 66.7 per cent debt stock reduction

This simulation examines the possible impact of the World Bank and IMF's recent, vigorously publicized⁸ Highly Indebted Poor Countries (HIPC) initiative. This initiative comprises two phases. Phase one requires a three-year record of compliance with an IMF program. This leads to the decision point for acceptance into the HIPC initiative. Acceptance requires having indicators that show the country be beyond debt sustainability thresholds. These are defined as a 20–25 per cent debt service ratio and present value of debt to export ratio of between 200 to 250 per cent. If countries find themselves above these thresholds they are eligible for the Nepal terms, under which they are entitled to a two-third reduction in debt stock. This reduction is applied to all types of debt, including multilateral. Phase two of the new initiative comprises the implementation of another three-year IMF program and, if

accepted within the HIPC category, an 80 per cent debt stock reduction. Since it is widely believed that few African countries are likely to reach this phase in the foreseeable feature, the simulation below is for the first phase only (see Oxfam, 1997 for a critical review of this issue).

(iii) Annual 10 per cent increases in public investment in Africa In this simulation aid is financed by other government revenue of African countries.

External shocks simulations

(iv) A one-point rise in world interest rate

This approximates the fluctuation of the six-month US dollar-based LIBOR from 1988 to 1990.⁹ Since world interest rate is an endogenous variable, the external shock simulation is conducted by changing the constant term (a_{43} in equation [6.7k] of Chapter 6). One possible interpretation for such a shock could be an exogenous rise in Northern demand for private credit, due, say, to an institutional change in Northern banking systems. Alternatively, the provision of government incentives for those who invest (for example, in Eastern Europe) could serve to explain such a shock.

(v) A 175 per cent increase in world oil price

This simulation reproduces a shock similar to the second oil price shock of 1978 to 1979. I have opted for this assuming the magnitude (nearly 260 per cent) of the first oil shock of 1973 is not likely to be repeated, given the lessons drawn from that experience. Moreover, the second oil price shock is a relatively recent and a mild one, which might feasibly be carried by OPEC.

(vi) A 10 per cent increase in export supply of commodities by the South, excluding Africa

The average rate of export growth of the America's (western hemisphere) was 11 per cent in 1990 and 10 per cent in 1991, while the corresponding rates for South Asia were 16 and 14 per cent, respectively (UNCTAD 1993). Thus, the 10 per cent simulation attempts to approximate this scenario. This simulation is useful in understanding the fallacy of composition in policy advice, such as an across-the-board devaluation by all countries, as well as the 'add-up problem' of commodity exports.

7.4.1 An Increase in the Level of Aid to the South

Three main issues are clarified by simulating an increase in the level of aid to the South. Firstly, the effects of such an increase on the different variables selected for inclusion is made clear. The second issue upon which light is thrown relates to the question of whether the financing mechanism of aid matters or not. Finally, the simulation helps to clarify whether the effects of an increase in the level of aid is likely to vary across regions and commodities.

Diagrams 7.4.1(a) to 7.4.1(c) show the impact of an increase in aid on real commodity exports and price of commodities. The results indicate that, in general, both prices and exports show a negative deviation from the base run.¹⁰ The potency of the effect is highest when aid is bond financed. This effect manifests itself through three main channels. Looking first at bond/deficit-financed aid, this is likely to have a double effect. Firstly, by raising the demand for credit, upward pressure is placed on interest rate. The latter affects the price of commodities directly, through the speculative demand placed on particular commodities, and, indirectly through its investment, and hence, aggregate demand-depressing effect in the North. The combined effect of this is to depress demand for commodities and hence prices and exports. This effect is less potent when the aid is budget financed, chiefly because the interest rate effect is minimized. This is shown in Diagrams 7.4.1b and c by the fact that the broken line, representing budget-financed aid, lies above the solid lines, representing bond-financed aid.

The oscillating pattern for minerals, which is also observed for all other simulations set out in this chapter, results from using both the short-run and two-periods lag long-run effect of interest rates together within the same simulation. Thus, in the base run, although the interest rate rises steadily, the rate at which it does so is not constant. Since the long-run interest rate is not significant in relation to other commodities, the pattern with respect to interest rate for these other commodities may reasonably be assumed to result only from the short-run impact. The sharp decline in exports for the ESA region results from the long-lag effects, which are felt after five years. Similarly, the decrease in food price (Diagram 7.4.1c) in the period of the shock results from a comparatively strong short-run one-period lag effect of interest rate. This effect is particularly strong for bond-financed aid due to the impact of the latter on interest rate. Two important conclusions may be drawn, based on the above analysis. Firstly, compared to budget financing, deficit financing is likely to have an adverse effect on commodity prices and African export revenue. Thus, the method of financing of aid matters. This result would tend to support the conclusions of Vos' (1994) study.¹¹ Hence, if bond/ deficit financing is the preferred means of financing in the North, there could emerge an asymmetric relationship in policy choice between the South and the North. Secondly, the effect of this policy will depend on the particular dynamics of each commodity. Thus, depending on the composition of exports of different regions, and their sensitivity to interest rate and Northern demand, different regions may be affected in different ways by the same policy shock. This underscores one of the underlying theoretical bases of this book, which holds that the structure of an economy matters. In relation to Africa, the export structure for the ESA region is found to be comparatively the most vulnerable.

Diagrams 7.4.1d and e show the impact of aid on public revenue and expenditure. In line with the econometric results presented in Chapter 5, the impact of aid on public expenditure and revenue is found to be positive and negative, respectively. This result shows the working of the fiscal response effect. The result is also interesting because it takes account of the general equilibrium effect of aid.¹² However, the effect of this aid on the public deficit is found to be negative, with public revenue, in general, declining. This effect is very small for the WCA region and highest for the NA region. This is attributed to the fact that elasticity of government revenue to foreign inflows are found to be relatively higher, in both the long and short run, in NA than in the two other regions. In the WCA region, the foreign inflow effect is significant only in the short run, and hence, its potency is low. The nature of financing of aid is found to have a negligible effect on public revenue, with identical results being obtained for both. This results from the negligible impact of world interest rate on public revenue in the South. This contrasts strongly with its effects on public expenditure, shown in Diagram 7.4.1e, for which, deficit-financed aid is found to have a strong potency. Hence, expenditure is found to be sensitive to the nature of financing, especially in the NA region. This is brought about by higher interest payments in response to a rise in interest rate, which results from the deficit financing of aid. The strength of this effect within the NA region may be explained by the high private debt composition within that region.

Finally, Diagram 7.4.1f shows the impact of aid on real output. The general decline in output would tend to accord with the discussion set out in the previous paragraph. Only within the NA region is the nature of financing of aid found to have a differential impact. Although both deficit- and budget-financed aid depresses output in the North, the impact of the latter is found to be comparatively stronger. The main reason underlying this result is the fact that the demand effect of a cut in public expenditure is higher than the interest rate-induced lower investment-demand (and hence aggregate-demand) effect. This, combined with the commodity price effect of deficit financing, brings to light the asymmetric relationship in policy choice of how to finance aid from the North and South's perspective. In contrast to some global models,¹³ the results of which suggest that aid leads to a rise in South's GDP, the impact of such aid inflows on the level of real output in Africa is found to vary across regions. This may be explained both by the level of disaggregation, and the inclusion of the macroeconomic effect of aid, within the model developed in this book. Thus, although aid might initially result in a rise in the level of imports and hence output. its macroeconomic ('Dutch disease' and 'fiscal response') and terms of trade effects work against this initial positive impact. Thus, aid would appear to have no noticeable effect on output within the NA region but a declining effect in the WCA region. In the ESA region, aid is initially found to have a positive effect, only to decline later. In this region, two years after initial capital inflows, a rise in public expenditure is observed. This is mainly attributed to a strong two-year lag short-run impact of imports, which is made possible through this inflow. This, in turn, has the effect of crowding-in private investment. The combined effect is a gentle rise in output over the six years of the simulation. The decline in output within the WCA region is attributed to the offsetting effects of export earnings. This arises as a consequence of the heavy weight of minerals and agricultural raw materials in the export basket for this region, with the price of the latter falling strongly, following the shock. Moreover, although aid-financed imports could have led to a rise in output through its positive impact on investment, the lack of a positive relationship between public and private investment in this region is likely to have limited such possibilities.

Two important conclusions may be drawn, based on the above simulation. Firstly, aid to Africa is likely to have profound macroeconomic effects. This manifests itself in the negative effect, which such aid has on the price of commodities and on Africa's fiscal balance. On the positive side, such aid may also provide relief for countries facing an import compression problem. The second important point is that deficit, rather than budget, financing of aid is likely to have an adverse impact on the South. However, this method of financing will have a comparatively lesser effect on Northern output. This may highlight a possible policy choice conflict between the two regions, owing to the structure of their interaction.









Diagram 7.4.1c

The effect of a 100 per cent increase in bond (solid lines) and budget (broken lines) financed aid to the South on African commodity prices





Diagram 7.4.1d The effect of a 100 per cent increase in budget financed aid to the South on government revenue in Africa

Diagram 7.4.1e

The effect of a 100 per cent increase in bond (solid lines) and budget (broken lines) financed aid to the South on government expenditure in Africa





7.4.2 Debt Cancellation to Africa

There is a heated debate among donors and recipients alike as to the possible impact of debt cancellation on Africa. However, the generally held expectation is that debt cancellation will alleviate some of the major economic problems facing African nations. In order to test this policy the 'Nepal terms', which is incorporated into the HIPC initiative, is simulated. As noted above, this initiative entitles participating nations to a two-thirds total debt stock reduction. The impact of this policy on debt service ratios and private capital flows to Africa is illustrated in Diagrams 7.4.2a and b, respectively.

The results of the simulation indicate that this policy has virtually no effect on the real level of output, and hence these results are not reported here. The main channel through which this output effect might be expected to manifest itself is by way of lower interest payments, which allows a higher level of imports and hence higher public and private investment. Increased private investment will also be facilitated by resolving the debt overhang problem. The mere reduction in interest payments is found to have too weak an effect to counteract the model's own dynamics, which are heavily influenced by commodity price, and other effects. However, the results of the simulation indicate that reducing interest payments does have an effect in reducing the burden of debt servicing. This is illustrated in Diagram 7.4.2a. From this diagram, it can be seen that the effects of a reduction in interest payments are most pronounced in the relatively less developed regions, such as ESA. Indeed, using MULTIMOD, Soludo (1993) has obtained similar results for the South in general. This is not a surprising result, since the relative debt burden of poor regions is very high

However, a less clear picture emerges with respect to private capital inflows to Africa, once the 'debt overhang' problem has been solved by debt cancellation. As can be seen from Diagram 7.4.2b, immediately following the shock, private inflows will increase to the ESA region and decline in the other regions. However, following this immediate shock, such inflows are likely to remain fairly stable, thereafter. This upward shift in the level of inflows to the ESA region results from an impressive improvement in the debt service ratio (shown in Diagram 7.4.2a), following the cancellation of debt. As demonstrated by Soludo (1993), this may be explained by the forward-looking assessment of the South's debt servicing capacity by Northern agents. Since the model developed here employs MULTIMOD-type debt-repayment assessment criteria, it is reasonable to assume that a similar inflow will result. However, the above simulation represents an advance on previous models, in that it highlights how this effect could vary across regions. The model also highlights the important result that it is not simply a decline in the debt service ratio which matters, in terms of attracting private inflows. Rather, the degree to which it declines and the resulting relative position of a particular region in relation to the rest of Africa, and the South,¹⁴ in general, are both important. This finding has particular relevance for both NA and WCA regions.

To sum up, the above simulation indicates that debt cancellation is not a panacea to the African debt problem. It is also shown that the impact of such a cancellation is likely to vary markedly across regions. However, debt cancellation can alleviate the pressure on the fiscal balance of government, both directly, by reducing the expenditure burden, and indirectly, by providing positive signals to private lenders.



Diagram 7.4.2a The effect of a 66.7 per cent bilateral and multilateral debt cancellation on Africa's debt service ratio

Diagram 7.4.2b The effect of a 66.7 per cent bilateral and multilateral debt cancellation on private capital flows to Africa



7.4.3 An Increase in Public Investment

AFRIMOD may also be used to simulate domestic policy options. Thus, in order to illustrate the value of the model in analysing domestic policies within a global framework, a 10 per cent permanent, annual increase in public investment, financed by other government revenue, is assumed. The impact of this policy on real output, private investment and real exports is given in Diagrams 7.4.3a, b and c, respectively.

In general, public investment is found to have a positive impact on the level of output, as shown in 7.4.3a. However, some variation exists across regions. Thus, the impact of public investment is strongest in the ESA region, owing to a higher capital output ratio and crowding-in. This is illustrated in Diagram 7.4.3b. The relatively weaker impact in the WCA and NA regions is attributed to lack of crowding-in and the evidence of crowding-out, respectively, for the two regions. These results follow logically from the findings presented in Chapter 5. The relatively higher level of output in WCA, compared to NA, results from differences in the capital output ratio of public investment in the two regions. The general equilibrium effect of this policy on Northern output is also shown in Diagram 7.4.3a. The demand for Northern goods generated in Africa owing to this policy shock has a positive, albeit less potent, effect on Northern output.

The impact of this policy on the export of commodities is, in general, insignificant. However, a slight trend, at first declining and thereafter recovering, may be noted. The declining trend is partly a dynamic result that operates through the impact of public investment on productivity and exports. This results in an increase in supply and hence, a fall in price, through the operation of the global market. The subsequent supply response following a lag period is a decline in real exports. As discussed in Chapter 4, such an effect is found to be strongest in relation to the agricultural raw material sector of the NA region and the food-exporting sector of the WCA region. For the ESA region, such an effect is dampened by an offsetting impact of private investment on exports and hence, a sharp rise in such exports. Only the mineral sector of this region is found to be affected, after a lag, in a similar fashion to the other regions. This is shown by a slight fall in supply in the sixth period.



Diagram 7.4.3a financed) public investment on real output

Diagram 7.4.3b The effect of an annual 10 per cent exogenous increase in (public revenuefinanced) public investment on private investment







7.4.4 An Increase in the World Interest Rate

In order to assess the effects of world interest rate, a one-point exogenous rise is invoked within the model. The first impact of this is to raise the debt service ratio. As can be seen from Diagram 7.4.4a, this has an adverse effect in all regions. The variation between regions may be explained by a divergence in export performance¹⁵ following the shock. This result indicates that an increase in interest rate will raise the debt burden of Africa, both directly and indirectly, through the operation of the world commodity market. An increase in interest rate is also likely to have an adverse effect on private inflows.¹⁶

As shown in Diagram 7.4.4b, the impact of an increase in interest rate on commodity price will vary between commodities. Thus, the effect of this external shock is most severe for agricultural raw materials and minerals, whose global demand is relatively more sensitive to changes in world interest rate. Hence, regions which have a larger composition of these commodities in their export basket are likely to suffer more, for an equal level of external shock. This would tend to lend



Diagram 7.4.4a The effect of a 1-point permanent exogenous rise in the world interest rate, 3 months LIBOR on the debt service ratio

Diagram 7.4.4b

The effect of a 1-point permanent exogenous rise in the world interest rate, 3 months LIBOR on commodity price



support to the general proposition raised in relation to the previous simulation, that the effect of an external shock will depend on the production structure of a region. To sum up, the important point, which emerges from this simulation is that Northern macro policy, such as that relating to world interest rate, will have a serious impact on commodity prices and export earnings of African producers. Such policies may also have an impact on private capital flows to Africa. In all cases the model suggests the likelihood that this change in world interest rate will have an adverse macroeconomic effect.

7.4.5 An Increase in Oil Price

In order to analyse the impact of the actions of oil producing and exporting countries, a 175 per cent increase in oil price, equivalent in magnitude to the second oil price shock, is invoked within the model. This is likely to have a number of effects, which are illustrated in Diagram 7.4.5a. Firstly, such a shock will probably create a recessionary situation in the North. It could also result in an initial rise, and subsequently, a secular decline of bank credit available to Northern agents and Southern borrowers. And finally, it is likely to bring about a general decline in world interest rate, with a recovering trend as time passes. These results differ from most outcomes reported in the literature, but accord in certain respects with those produced by Vos (1994). This may be explained by the fact that Vos' model has similar theoretical underpinnings to the one developed here, particularly in relation to how it deals with credit rationing.

Vos' model indicates the likelihood that an oil shock will result in a general decline in interest rate. This is explained by the fact that a rise in oil price will lead to an increase in deposits by OPEC member states into international banks, which, in turn, will push interest rates downward. However, in contrast to Vos' (1994) model, the simulation presented here suggests that the interest rate will have a tendency to rise over time.¹⁷ In other global models, such as those developed by Bruno and Sachs (1985), McKibbin and Sachs (1991), MULTIMOD (Masson et al. 1988) and the World Bank's GEM (Allen & Vines 1992), interest rate is found to rise after the shock. In these models, although a rise in oil price results in an increase in the supply of funds from OPEC, this will also have an adverse impact on savings and hence, on the supply of funds from the North, due to the recessionary impact of the shock. In spite of the recessionary impact on investment, this is likely to result in

a rise in world interest rate. The results of AFRIMOD suggest a reconciliation of these two features. Thus, in the early period of the shock the supply of funds from OPEC will have a negative impact on interest rate. However, as time elapses, the recessionary impact starts to take effect, which results in a slow recovery in the interest rate. However, since Vos' (1994) model is not dynamic, this effect is not captured within that model.

The impact on the level of imports of each region, shown in Diagrams 7.4.5b and c, will depend on a number of factors. Thus, as shown in Diagram 7.4.5b, whether the region is oil exporting and/or has the financial means to secure the necessary imports may be important, as it is for the WCA and NA regions. Similarly, whether the region is able to adjust to the shock through import compression may also be important. Thus, as illustrated in Diagram 7.4.5c, in the ESA region, such a shock is likely to have an adverse impact on output through its import compression effect. Further disaggregation of the regions, say into oil importers and oil exporters, would be useful in showing, for example, whether import compression holds for particular types of non-oil-exporting economy.¹⁸

Diagram 7.4.5a The effect of a 175 per cent permanent increase in oil price on interest rate, oligopolistic lending and Northern output





Diagram 7.4.5b The effect of a 175 per cent permanent increase in oil price on imports to Africa

Diagram 7.4.5c The effect of a 175 per cent permanent increase in oil price on imports to East Africa



7.4.6 An Increase in Supply of Commodity Exports by Other South

The final simulation exercise experimented with is a 10 per cent increase in the supply of commodity exports by other developing countries of the South to Africa. As shown in Diagrams 7.4.6a and b, the immediate impact of this shock is reflected in world commodity price and exports. From the first of these diagrams, it is clear that the small country assumption fails dismally to hold for all commodity categories. The fall in price is worst, however, for food. This may be explained by the fact that the supply of food from 'other South' is nearly 40 times larger than for Africa. As discussed in relation to the previous simulation, the oscillation observed in mineral price, and hence, by extension, export revenue and the vulnerability indicator, may be attributed to the particular dynamics of that market.¹⁹ However, in spite of these oscillations, the overall effect of this supply shock is a secular downward trend within this market. This general decline in prices triggers various effects. The most important of these, illustrated in Diagram 7.4.6b, is a decline in the level of real exports. This, in turn, will have knock-on effects on the capacity to import, and hence, may result in import compression.

Given this simulation result, it is possible to compute the amount of capital inflow required to maintain the same level of imports and hence, the same level of output and output growth as was in place, prior to the shock. This hypothetical level of capital inflow, measured as a proportion of the base run level of annual bilateral and multilateral flows, is given in Diagram 7.4.6c. In plain English, this diagram illustrates the vulnerability of each region to indebtedness, owing to an external shock such as a deterioration in terms of trade.

This takes us back to the central proposition, put forward at the beginning of this book, that the African debt problem is deeply rooted in the position of the continent as a small commodity supplier within the world economy. If we accept this proposition as accurate, then it follows logically that Africa will remain vulnerable to debt unless this fundamental problem is resolved. Although all three regions are found to be vulnerable, the degree of vulnerability varies across regions, depending on the composition of commodity exports. Hence, those regions whose export composition comprises commodities, which are relatively vulnerable to world market shocks, will suffer the most. As shown in Diagram 7.4.6b, the most vulnerable region is NA. This is followed by the WCA region, which appears to suffer from various degrees of vulnerability to indebtedness, as illustrated in the hypothetical situation set out in Diagram 7.4.6c. The high degree of vulnerability to debt, shown in this diagram for the ESA and NA regions, may be explained by the level of export earnings and the projected base run value of flows, which are, themselves, a function of export performance.

7.5 CONCLUSION

This chapter has shown not only the potential of the model developed in this study, for the analysis of external shocks and policy simulation, but also the implication of such shocks for African economies. The model is essentially one of the new generation of macroeconometric models consisting of econometrically estimated parameters and a globally consistent accounting framework. The model is solved using a Guass-Seidel algorithm. Both the theoretical analysis of the trace of the matrices for the excess demand function, and the symmetry and linearity test undertaken, indicate that the model is fairly stable.







Diagram 7.4.6b The effect of a 10 per cent increase in supply by other (non-Africa) South on real commodity exports from Africa

Diagram 7.4.6c Vulnerability to debt: The effect of a supply-induced decline in world commodity price on flows of official finance to Africa



However, it is worth noting some important limitations of this model. The first of these relates to the fact that, while the modelling of Africa is quite sophisticated, by the standards of the existing literature, the modelling of the North, applied here, is generally rather crude. This sets a limit as to the extent to which it is possible to undertake an indepth analysis of North-South interaction. Naturally, this is a consequence of the fact that the focus of this study is explicitly on Africa. Nevertheless, it would be desirable for future work in this area to combine African models, such as the one developed in this book, with more sophisticated Northern models. The second major limitation of the study is that the other developing countries, and the rest of the world, are generally assumed as exogenous. While some attempt has been made to simulate²⁰ the effects of other developing countries on Africa, nevertheless the modelling of 'other South' is not as dynamic as that undertaken in relation to Africa. Again, this is due to the scope and focus of the study, which is explicitly on the interaction between the North and Africa. Hence, useful insights might result from future work, which attempts, explicitly, to model these regions. A final weakness of this study is that the econometric estimation adopted is based on a single-equation ECM. This implicitly assumes that the theoretically specified functions are the most relevant long-run cointegration vector. To the extent that this assumption departs from reality, the estimation technique clearly has limitations.

However, in spite of these various limitations, the various simulations conducted, although not exhaustive, do highlight some of the structural and policy issues facing African economies in their interaction with the North. These have been discussed at some length in this chapter. In the concluding chapter, I will summarize the major findings of the study as a whole, and highlight some policy implications of these various findings.

8 Concluding Remarks

This study has attempted to model the macroeconomic interaction between Africa and the North. There are two main reasons why such an exercise might be considered important. Firstly, building a macroeconomic model of Africa which is based on stylized facts and econometric estimation within a world economy framework is an exercise that has value in itself. This claim may be justified since no such models of Africa may currently be found within the literature. Indeed, a number of policy questions and external shocks, not discussed in this book, may be analysed using the model developed here. Nonetheless, models are usually built with specific objectives in mind. In this respect, the model developed in this book is no exception. Hence, the second main justification for the present study, which is to examine the questions outlined in Chapter 1, using the simulation exercise set out in Chapter 7. Chapter 7 also outlined various policy implications of the model, as well as how such a model might be used to analyse external shocks and policy questions by locating these within a North-South macroeconomic framework. The main points arising out of this exercise are summarized below.

The first point worth noting is that Northern macro polices are likely to have a dual impact on Africa, both through finance and trade.¹ Hence, any analysis of the impact of external shocks and Northern policy on Africa will be incomplete if this finance-trade linkage is not adequately explored. Examining these two aspects within a global economic framework allows one to explore those dimensions of development problems which would otherwise have remained obscured were the national economy to be taken as the sole unit of analysis. Further, such an analysis is crucial in revealing the context under which individual national economies function. Letting these various insights guide experimentation with the model developed in this book allows for a number of major issues to be clarified. These may be summarized as follows:

1. The debt problem in Africa is essentially a commodity problem. Hence, a decline in world commodity price must be counteracted by a proportional inflow of foreign capital if a decline in output (which is, itself, capital constrained) is to be avoided. However, such an inflow of capital will imply a corresponding increase in debt.

As the analysis in Chapter 1 has shown, the historical interaction of African economies with the North has left them with an external sector dominated by the export of commodities and import of manufactured goods. This has resulted in a secular deterioration in the terms of trade of Africa. Both rising and declining commodity prices are shown to create the pre-conditions for indebtedness. Rising commodity prices improve creditworthiness of the debtors in the eyes of lenders, as well as capacity to pay in the eyes of debtors. Declining prices, combined with the belief in the existence of a cyclical commodity price pattern, has also led to indebtedness, associated with attempts to overcome what is perceived as a 'temporary' foreign exchange problem. However, when such a pattern of trade and indebtedness is structural, the debt problem appears as a systemic phenomenon embedded in the macroeconomic interaction of North and South, in general, and the effects of the commodity market, in particular.

The simulation of an increase in supply of commodities by other developing countries clearly demonstrates the likelihood that an increase in the supply of commodities will lead to a decline in commodity prices.² Thus, a 10 per cent increase in the supply of commodities by other developing countries leads to a 10–50 per cent fall in the price of agricultural raw materials, a 50–75 per cent fall in the price of beverages, an 80–90 per cent fall in the price of food, as well as a fall in the price of minerals, ranging between 15 and 80 per cent, for the period under simulation. If this fall in price is to be compensated by borrowing, for a given level of export volume, then this implies an increase of 10–20 per cent on the base run level of capital flows into WCA, for the period under simulation. The corresponding figures in relation to the NA and ESA regions range between 60 and 160 per cent, and 5 and 180 per cent, respectively.

The vulnerability to indebtedness or aid can emerge even from apparently positive actions, as shown by the first simulation (doubling the level of aid to the South). The simulation of an increase in aid to the South also underscores the differential impact, which the manner of financing of aid by the North, has on Southern economies. For instance, doubling the existing level of aid leads to a 0.5-4 per cent decline in exports from the ESA region when this is financed by a Northern deficit. However, when such aid is financed through budgetary cutbacks in the North, the corresponding decline in exports from ESA stands at 0.25-3 per cent. In relation to the NA region, the use of bond-financed aid implies a decline in exports of around 1 per cent, while budget-financed aid brings about a reduction of only around 0.1 per cent. No major change is observed for the WCA region due to the nature of its financing. Thus, based on the specific results above, it is shown that from Africa's perspective budget-financed aid is to be preferred. However, from the North's perspective, such budget financing leads to a decline in Northern output of between 1 and 2.5 per cent, with a corresponding reduction of 1 per cent, when the aid is bond/deficit financed.

Various other specific points may be read from the simulations presented in Chapter 7. However, the general finding arising from these various simulation experiments is that the debt problem is intricately linked with the trade problem. In summary, based on the above analysis, it may be concluded that the place of African economies in the world economy is such that they are sellers in a flexiprice market and buyers in a mark-up price market. Given these structural features, an increase in Southern productivity, a rise in world interest rate and Northern recession are shown to work against the South in general, and Africa in particular.

2. African economies are extremely sensitive to world interest rate. This effect is twofold. Firstly, high interest rates will have an adverse impact on debt servicing, and hence place fiscal pressure on these economies. Secondly, by affecting world demand for commodities, an increase in the level of interest rates will have an impact on African commodity markets.

The simulation in relation to world interest rate has shown two propagation mechanisms for Northern policy on Africa: a direct effect, by which interest rates will affect the size of debt burden and subsequent private capital inflows; and an indirect effect, through the demand for commodities. The latter is found to take two forms. Thus, a rise in world interest rate depresses commodity prices directly, through its commodity speculative demand effect, and indirectly, by affecting Northern income and hence, depressing demand. The combined effect in Africa is import compression, and when loans are obtained to redress the foreign exchange problem, debt accumulation. Specifically a one point increase in world interest rate leads to a 2.5 per cent fall in world commodity price of food and agricultural raw materials, as well as a 0.5 to 3.9 per cent fall in the price of minerals. The effect on beverage price is found to be negligible, chiefly as a result of the low interest elasticity for this commodity. Similarly, a rise in world interest rate leads to an increase in the debt service ratio of 2.5 per cent, 3.5 per cent and 4 to 5.5 percent for NA, WCA and ESA regions, respectively. The policy implication of this analysis is that debt alleviation initiatives will need to pay special attention to macroeconomic interaction, regional variations and the commodity price implications of Northern policy, if they are to have any chance of succeeding.

3. Contrary to the impression given by multilateral donor institutions, the public deficit in Africa is not merely a domestic phenomenon. Rather, it is very much linked to foreign inflows, their method of financing in donor countries, commodity price dynamics and the latter's impact on the external trade of Africa.

In general, capital flows to Africa are found to have negative and positive impacts on public revenue and expenditure, respectively. However, there would appear to be some variation in this across regions. Thus, doubling the existing level of aid leads to a 1.5-2.5 per cent decline in public revenue, from the base-run level, within the ESA region, and a fall of around 4 per cent in NA.³ In contrast. the same change in aid levels leads to a 2.25 per cent increase in public expenditure in NA, while in the ESA and WCA regions, the level of such expenditure remains fairly constant, at around 0.25 per cent.⁴ The first simulation in Chapter 7 has shown what the fiscal response of African economies would be to an increase in aid. This simulation has also demonstrated that the method of financing is likely to have a direct bearing on public expenditure. Thus, aid financed by a Northern deficit is likely to depress commodity prices more than aid financed through budget cuts. More specifically, a bond-financed doubling of the existing level of aid leads to a 0.25-2 per cent decline in the price of minerals, while budget financing may lead to a small increase of around 0.2 per cent, over the simulation period. The corresponding figures in relation to the price of agricultural raw materials are a decline of around 1 per cent, when bond financed, and around 0.2 per cent when budget financed. For beverage prices, the method of financing does not appear to make a significant difference, while for food prices, bond financing may result in a decline in price from the base run of up to 1.2 per cent. Thus, from the African perspective, budget financing of aid is to be preferred. However, output in the North is found to be depressed more heavily by budget financing, which entails a 1 to 2.5 per cent decline from the base run. This highlights the existence of a conflict in policy choice between the two regions and hence, underscores the need to consider the global macroeconomic impact of Northern policy on Africa when policy proposals for fiscal discipline in Africa are made.

- 4. Within our simulation, domestic policy options could represent an important instrument for mitigating structural problems. However, the impact of public investment on private investment varies not only across regions, but also across time. Thus, a 10 per cent increase in public investment in the ESA region leads to a 5-10 per cent increase in private investment in that region. In NA and WCA the initial rise in private investment of around 2 per cent quickly fell to a low of 15 per cent by the end of the simulation period. This policy has led to an increase in output of 5-10 per cent in ESA, 4-10 per cent in WCA, and nearly 1 per cent in NA. In less developed regions, such as ESA, public investment has a positive impact on private investment due to its crowding-in effect. Hence, flexible and timely evaluation of such domestic macro policy options is crucial. On the other hand, although the magnitude is small, such policies may lead to an increase in the level of output in the North through a rise in demand for Northern goods.⁵ This indicates the possibility that Global Keynesianism is in operation.
- 5. For similar policy shocks, although the general pattern is similar, different regions may respond in quite different ways. This underlines one of the important points in the structuralist argument, that the specific structure of a particular region is important.

In the particular case of the three regions which form the subject of this book, these would appear to exhibit a common general pattern, following policy/external shocks. However, within that broader pattern, these regions also exhibit their own unique reaction. This is shown in all simulations undertaken in Chapter 7. This point may be illustrated by examining the impact of a Northern deficit-financed doubling of aid to Africa. Thus, this policy results in an initial rise in output of up to 0.4 per cent leading to an eventual 1 per cent decline in ESA, a decline in output of 0.5-1.4 per cent in WCA and around a 0.1 per cent increase in output of NA. The impact of a doubling of deficit-financed aid on public revenue is also found to vary across regions. Thus, this policy was found to have almost no effect in WCA, while leading to a 2-2.5 per cent fall in public revenue in ESA and a fall of about 4 per cent in NA. Similarly a twothird debt stock cancellation leads to a decline in debt service ratio of nearly 80 per cent for ESA, while bringing about only a 25 per cent and 12 per cent reduction in WCA and NA, respectively. A range of other, regionally-specific, variations may also be observed. based on these simulations. These variations may be attributed to three main factors. Firstly, variation in the composition of commodity exports as well as the price dynamics of commodities for each region. Secondly, the differential operation of fiscal response and 'Dutch disease' effects of external finance in each region. And thirdly, variation in the composition of private and multilateral debt stock, as well as current flows to each region. The major implication of these variations is that future research by regional organizations, such as the Economic Commission for Africa, should focus on developing macro models of each country and linking these with Northern macro models. Such an exercise would provide an effective tool for the analysis of the macroeconomic impacts of Northern policy and external shocks on Africa.

To sum up, the analysis undertaken in this book has shown that there exists a complex and dynamic interaction between North and South. This is characterized by an economic structure which makes the African economy extremely vulnerable to the North's macro policies and external shocks.⁶ It has also been shown that international finance mechanisms operate in quite a unique fashion in relation to agrarian economies, which are exporters of primary commodities, importers of manufactured goods and on the periphery of global capital markets. The implication of this finding is that international finance studies in agrarian economies need to explicitly link considerations relating to commodity trade and external finance.

The Neostructuralist macroeconomic approach of building macroeconomic analysis on the specific economic structure of an economy, which, itself, is the result of a specific historical process, represents a fruitful avenue for macroeconomic research in Africa. However, macroeconomic studies of Africa undertaken to date may be grouped into two, quite different categories. Firstly, a number of studies have been undertaken, at higher level of abstraction and theoretical contour, not based on rigorous empirical work but with a good sense of history.⁷ Secondly, a new generation of economists has attempted to undertake empirically rigorous studies devoid of the historical focus which is becoming common in the economics departments of many African universities. Although the two approaches may be of great value, in their own right, a fruitful study of macroeconomic issues in Africa calls for a combining these two approaches. Indeed, the two approaches may be seen as complementary, since the former provides us with historicallyformed stylized facts, while the latter helps us to investigate these empirically. The Neostructuralist's dictum of 'historically formed economic structures matters' can, as has been attempted in this book, be seen as a helpful analytical framework in this endeavour. This approach has paramount significance in Africa, where the burden of history, which moulded its economy, is still felt today. However, the need to articulate this structure empirically represents a necessary and important prerequisite for drawing up an appropriate course of action.

Appendices

Appendix 1 Country Classification

1. North

Comprises OECD countries, World Bank definition including Greece and Portugal: Australia, Austria, Belgium, Canada, Denmark, Finland, France, Germany, Greece, Iceland, Ireland, Italy, Japan, Luxembourg, Netherlands, New Zealand, Norway, Portugal, Spain, Sweden, Switzerland, United Kingdom and United States. In terms of regional classification it is North America (USA and Canada), EEC, Australia, Japan and New Zealand.

2. SOUTH

All developing countries (World Bank, World Tables definition).

3. OTHER SOUTH

All developing countries excluding Africa (World Bank, World Tables definition).

4. OPEC

OPEC members excluding African and Western Hemisphere suppliers: Bahrain, Kuwait, Qatar, Saudi Arabia, United Arab Emirates (UAE) Iran and Iraq.

5. AFRICA*

5.1 East and Southern Africa (ESA)

Economic Commission for Africa (ECA) definition including Sudan: <u>Angola</u>, Botswana, <u>Comoros</u>, <u>Djibouti</u>, Ethiopia, Kenya, Lesotho, Madagas-

car, Malawi, Mauritius, <u>Mozambique</u>, Seychelles, Somalia, Sudan, Swaziland, Tanzania, Uganda, Zambia and Zimbabwe. *Samples used for econometric estimation of ESA model:* Botswana, Ethiopia, Kenya, Madagascar Malawi, Tanzania, Uganda, Zambia.

5.2 North Africa (NA)

ECA definition, excluding Sudan: Algeria, Egypt, Libya, Morocco and Tunisia. *Samples used for econometric estimation of NA model:* Algeria, Egypt and Tunisia.

5.3 West and Central Africa (WCA, ECA definition)

West Africa: Benin, Burkina Faso, <u>Cape Verde</u>, Cote d' Ivoire, Gambia, Ghana, <u>Guinea</u>, Guinea Bissau, <u>Liberia</u>, Mali, Mauritania, Niger, Nigeria, Senegal, Sierra Leone and Togo.

Central Africa: Burundi, Cameroon, Central African Republic, Chad, Congo, <u>Equatorial Guinea</u>, Gabon, Congo, Rwanda, <u>Sao Tome & Principe</u>, Zaire and D.R. Congo.

Samples used for econometric estimation of WCA model: Benin, Burkina Faso, Gabon, Ghana, Nigeria, Senegal, Sierra Leone Cameroon, Central African Republic, Zaire and D.R. Congo.

* Due to lack of complete data, countries which are underlined are not included in the regional database used in the model.
Appendix 4.1 The Econometric Approach Followed

1. AN OVERVIEW

The estimation in this book is undertaken using an Error Correction Model (ECM). This approach is based on the assumption that a long-run (equilibrium) relationship exists between variables in a specified equation. This, in turn, is assumed to be captured by a cointegration analysis. Following Engel and Granger (1987), two or more time series are cointegrated of order d, b, denoted CI(d, b), if they are integrated of order d, but there exists a linear combination of them that is integrated of order b, where $b \le d$. There are two ways of testing the existence of cointegration between variables of interest and formulating an ECM model. The first approach (the Engel-Granger two-stage approach) begins by testing whether the variables of interest are stationary or not. If variables contemplated in the model follow an I(1) process, then, in the first stage, estimates of the long-run equilibrium equation (using OLS) is made. An ADF test on the residual of the long-run equation will then be conducted. This is to determine whether the variables in question are cointegrated (whether the error term follows a stationary process). If the error term is stationary (taken as proof of cointegration) in the second stage, we could combine the error term with the first difference of the variables (short-run indicators) to estimate the final model. This approach has a number of shortcomings, particularly when there are more than two variables in an equation (see Engel & Granger 1991, Banerjee et al. 1993, Enders 1995, among others). This has led to the popularity of Johansen's approach which is better at handling multivariate systems (see Johansen 1988, 1991; Johansen & Juselius 1990). Following Johansen (1988, 1991) we may consider a VAR model given by:

$$X_{t} = \Pi_{1}X_{t-1} + \dots + \Pi_{k}X_{t-k} + \mu + \phi D_{1} + \varepsilon_{t} \quad (t = 1, \dots, T)$$
[4.1a]

In general, economic time series are non-stationary processes, and VAR systems like equation 4.1a can be expressed in the first differenced form. Using =1-L, where L is the lag operator, equation 4.1a can be rewritten as,

$$\Delta X_{t} = \Gamma_{1} \Delta X_{t-1} + \dots + \Gamma_{k-1} \Delta X_{t-k+1} + \Pi X_{t-1} + \mu + \phi D_{1} + \varepsilon_{t}$$
[4.1b]

where:

 $\Gamma_i = -(I - \Gamma_1 - ... - \Gamma_i)$ (*i* = 1,..., *k* - 1) and $\Pi = -(I - \Pi_1 - ... - \Pi_k)$

Model 4.1b is a traditional first-difference VAR model except for the term

X _{*t*-1}. The Johansen procedure is based on an examination of matrix Π , which contains information about the long-run relationship. The analysis of the long-run relationship in the model is based on examining the rank of this matrix. If this matrix has a full rank, the vector process X_t is stationary. If the rank equals zero, the matrix is a null matrix and equation 4.1b remains a traditional VAR. The third and most interesting possibility is when $0 < \operatorname{rank}(\Pi) = r < P$, which implies there are $p \times r$ matrices and such that $\Pi = -r$. The cointegration vector has the property that X_t is stationary even though X_t itself is non-stationary. The Johansen procedure helps to determine and identify the cointegrating vector(s). The empirical study in this book used both the Engel-Granger and the Johansen approaches.

2. ERROR CORRECTION MODEL (ECM) FORMULATION

Once the variables in question are cointegrated, we can formulate an ECM, following the Granger representation theorem (Engel & Granger 1987), which states that cointegrated variables have an ECM representation. A general representation of ECM with a one-period lag and one exogenous variable (see Banerjee et al. 1993, Thomas 1993, Kennedy 1992, Enders 1995, among others) may be written as,

$$\Delta y_{t} = \alpha_{0} + (\alpha_{1} - 1)(y_{t-1} - x_{1t-1}) + \beta_{10} \Delta x_{1t} + (\alpha_{1} - 1 + \beta_{10} + \beta_{11})x_{1t-1}$$

If we add and subtract $(\beta_{10} + \beta_{11})x_{1/-1}$ on the right-hand side, we arrive at the equation used for estimation, which may be given by,

$$\Delta y_{t} = \alpha_{0} + (\alpha_{1} - 1)(y_{t-1} - kx_{t-1}) + \beta_{10}\Delta x_{1t} + \varepsilon_{t}$$

where k, the long-run coefficient, is $\frac{(\beta_{10} + \beta_{11})}{(1 - \alpha_1)}$.

At a general level, if we have a long-run relationship given by,

$$Y_t = \alpha_0 + \alpha_1 X_t + \alpha_2 Z_t \qquad [4.2a]$$

The ADL for this can be written as,

$$Y_{t} = \alpha_{0} + \sum_{i=1}^{k} \alpha_{3} Y_{t-i} + \sum_{i=0}^{k} \alpha_{4} X_{t-i} + \sum_{i=0}^{k} \alpha_{5} Z_{t-i} + \varepsilon_{1t}$$
[4.2b]

By adding and subtracting terms appropriately (see above), the ECM formulation of equation 4.2b can be given as,

$$\Delta Y_{t} = \alpha_{0} + \sum_{i=1}^{k-1} \alpha_{6i} \Delta Y_{t-i} + \sum_{i=0}^{k-1} \alpha_{7i} \Delta X_{t-i} + \sum_{i=0}^{k-1} \alpha_{8i} \Delta Z_{t-i} + \alpha_{9} [Y - \beta_{1} X - \beta_{2} Z]_{t-k} + \varepsilon_{2t}$$

where:

$$_{6i} = (_{3}-1)_{i} + _{3,i+1} + ... + _{3,k-1}$$
 for $i = 1,...k-1$ and
 $_{j,i} = _{j,i} + _{j+1,i+1} + ... + _{5,k-1}$ for $j = 4, 5$ and $i = 1, ...k-1$
 $_{9} = _{3,i} + _{3,i+1} + ... + _{3,k}$ for $i = 0, ..., k$ and
 $_{1} = (_{4,i} + _{4,i+1} + ... + _{4,k}) / _{9}$ and
 $_{2} = (_{5,i} + _{5,i+1} + ... + _{5,k}) / _{9}$ and $_{2i}$ is a white noise process.

The terms in square brackets [] show the long-run model.

If we adapt this general ECM formulation with a one-period lag and six (a reasonable figure, in the context of most of our equations) exogenous variables we arrive at,

(a) The long-run relationship

given as,

$$y = \gamma_0 + \gamma_1 x_1 + \gamma_2 x_2 + \gamma_3 x_3 + \gamma_4 x_4 + \gamma_5 x_5 + \gamma_6 x_6$$

(b) The ADL formulation

The Autoregressive-Distributed Lag (ADL), denoted ADL (m.n,p) [where m represents the number of lags of the endogenous, n the number of lags

for the exogenous and p the number of exogenous variables] for our formulation will be ADL (1,1,6). This representation (from which an ECM can be formulated) is given as,

$$y = \beta_0 + \beta_{11}x_1 + \beta_{12}x_{1t-1} + \beta_{21}x_{2t} + \beta_{22}x_{2t-1} + \beta_{31}x_{3t} + \beta_{32}x_{3t-2} \dots + \alpha_1y_{t-1}$$

(c) The ECM representation

The ADL representation, as set out above, helps us to derive the general ECM representation. This ECM representation is given by,

$$\Delta y_{t} = \beta_{0} + (\alpha_{1} - 1) \left[y_{t-1} - \sum_{j=1}^{6} x_{jt-1} \right] + \sum_{j=1}^{6} \beta_{j0} \Delta x_{jt} + \sum_{j=1}^{6} (\alpha_{1} - 1 + \beta_{j0} + \beta_{j1}) x_{jt-1}$$

A first- and second-order ECM formulation is chosen in this study in order to minimize the complexity of the lag structure. Ideally one needs to start from a generous lag structure and test down to arrive at a parsimonial equation. The large size of my model meant that I was not able to follow this approach strictly.

(d) Inferring the long-run values

It may be generalized that the following formula gives the long-run coefficient for each of the exogenous terms and *i* different lags,

The constant term: γ_0 =

$$=\frac{\rho_0}{1-\sum_{i=1}^n\alpha_i}$$

R

Coefficient of the exogenous variables:

$$\gamma_j = \frac{\sum_{i=1}^{n=1} \beta_{ji}}{1 - \sum_{i=1}^n \alpha_i}$$

.. 1

where: j = exogenous variable jn = maximum lag level (1st order, 2nd order...)

Most of my coefficients are estimated using panel data. Applied time series analysis using such data is not yet well developed. Thus, the estimated coefficients may have a number of shortcomings. However, this approach is widely used in applied work. To tackle possible shortcomings, I conducted relevant tests for each country and generally found that the models show cointegration. This points to the need to conduct further study when the new technique is refined. Interested readers may refer to Anderson and Hsiao (1982), Levin and Lin (1993), Pesaran and Smith (1995), Pedroni (1995), McCoskey, Kao and Chiang (1998), Kao (1997, 1999), and Njuguna (1999) on the newly emerging issues of time series analysis in panel data. Njuguna (1999) in particular provides a summary of the newly developed tests.

(e) T-values of the long-run coefficients

The t-values of the long-run coefficients should be generated using the Wickens-Breusch transformation (see Gurney 1989: 89–90). These may be computed by regressing

 $\frac{y - [1 + (\alpha_1 - 1)]y_{t-1}}{(\alpha_1 - 1)} \text{ on all except } Y_{t-1}$

3. HYPOTHESIS OF DIAGNOSTIC TEST

This section briefly summarizes the test used in the analysis.

(a) Cointegration test

I use both the Engel and Granger and the Johansen tests.

 H_0 = Unit-root (of the error term) exists / the error term is non-stationary H_1 = The error term (of the cointegration equation) is stationary

For the Johansen test, the null is that no cointegration relationship exists. Mackinnon critical values are used in both cases.

(b) Specification test

I use Rumsy RESET (1)

 $H_0 = u \sim N(0, \sigma^2 I)$

Coefficient of fitted vector = $0 \Rightarrow$ no specification problem

 $H_1 = u \sim N(\mu, \sigma^2 I)$ Coefficient of fitted vector 0 => specification problem

Moreover, RESET (2) is also tested for all the equations and the result is similar, and often better, than the RESET (1).

(c) Normality test

I used a Jarque-Bera, which tests for the hypothesis of the value of 0 and 3 for skewness and Kurtosis, respectively. Although this applies for large samples (n>1000), nevertheless it may serve as a good indicator. Other visual inspection, which is done in estimation, is also important, even if this statistic may reject normality (see Mukherjee et al. 1998).

 H_0 = Normality $H_1 = H_0$ is not correct

(d) Parameter restriction test

I used a Chow break test.

$$H_0 = \frac{1}{1} = b_{1j} x t$$

where: 1 is the restricted coefficient and b_{1i} = the unrestricted coefficients.

 $H_1 = H_0$ is not correct.

Appendix 4.2 Stationarity Condition of the Expected Profit Indicator (Π)

Following the theoretical discussion presented in section II, the expected profitability/price instability index is computed for each of the commodity groups and regions. The index is computed by subtracting the three-year moving average of real exchange rate from each current level of real exchange rate,

$$\Pi^{\sigma}{}_{t} = \left(\frac{eP_{s}}{P_{d}}\right)_{t} - k^{-1} \sum_{i=1}^{k} \left(\frac{eP_{s}}{P_{d}}\right)_{t-1}$$

where k is assumed to be three years. Thus, the expected long-term average price is formulated by taking the evolution in past three years of the real exchange rate. This variable is tested for stationarity using the Augmented Dickey-Fuller Unit Root Test. The result is reported below.

| Region | Food | Agr. raw material | Beverage | Minerals, ores and metals | 1% Mackin- non critical value |
|--------------|--------|----------------------|----------|---------------------------------|-------------------------------------|
| North Africa | -5.7 | -8.04 | NA | -9.25 | -2.6 |
| W&C Africa | -11.66 | -11.67 | -9.67 | -10.53 | -2.57 |
| E&S Africa | -9.1 | -8.83 | -9.34 | -8.33 | -2.58 |

Augmented Dickey-Fuller Unit Root Test on $\Pi\sigma$

As this table shows, the series are clearly stationary. On the other hand all the other explanatory variables are non-stationary, there being I(1) variables. This indicates the absence of a long-run bivariate relationship. Esti-

mating variables with different levels of integration throws up a number of problems (Thomas 1993: 164, Banerjee et al. 1993, Ch. II) in bivariate analysis, although not in some multivariate systems. However, given the nature of our model, which is a multivariate system in which a subset of the regressors are cointegrated, we have estimated the model using this variable.

Some remark about the transformation of the data is in order. As all variables are in logarithmic form, taking the log of the price instability indicator will hide the negative deviation side of the story. Hence, one ideally should use the actual variable. However, this creates a problem of the functional form, which is log-linear. The best transformation, with a loglog functional form, and, at the same time, not losing sight of the negative deviations, is to take the difference between the log of real price and log of the moving average of the price. Formally, this may be expressed as,

 $X = f(P, P^*, Z)$

where: P is the relative price, P^* the price instability indicator (comprising a moving average of lagged P) and Z represents other regressors.

 $P^* = P - [\{Pt_1 + P_2 + P_3\}/3]$ Definition $Log(P^*) = Log(P) - Log([...])$ Taking logarithm of all variables $Log(P^*) = Log(P / [...])$

This will allow both the desired functional form (log-log) and the information about negative deviation (when [] > P) to be shown simultaneously. It is this value which is used for estimation purposes.

Estimation result

This series has exhibited a very high correlation (in most cases higher than 0.80) with the price variables in all regions. In spite of the multicolinearity problem involved, the estimation is carried using relative prices and capital formation indicators as regressors. For the ESA region, the estimation shows high R-square and insignificant t-values, indicating the likelihood of multicollinearity. Moreover, the estimation is undertaken using only the profit and capital formation indicators, with the former effectively replacing the real exchange rate, especially for minerals and agricultural raw materials. For food, the profit indicator shows a short-run negative coefficient while, for minerals, a positive long-run one. In relation to the other commodity categories, the coefficient is not significant. Similarly, for the

WCA region, the estimation, in general, was not found to be statistically significant. The profit indicator shows a positive value only for the capacity utilization aspect of minerals, although estimation with profit and capital formation indicators shows the profit indicator effectively replacing the real exchange rate. Neither are most of the estimates for the NA region found to be significant. However, for minerals, the profit indicator showed a negative coefficient, both in the short and long run, while for food it shows a positive short-run coefficient. To sum up, the lesson from this exercise is that the profit indicator is nearly identical with the real exchange rate, and, hence creates a serious multicollinearity problem. Therefore, it is appropriate to use either of the two, but not both. This highlights a serious shortcoming of previous studies, which commonly used both.

Appendix 4.3 Results Using First Difference Values in the Global Commodity Model

In this appendix I report the result of the estimation of the global commodity market, using the first difference of the series in the model. Both the equilibrium and disequilibrum versions of the model are estimated. The equilibrium model essentially uses equations 1 and 2 set out in section 4.4.2. In this model, adjustment is instantaneous. In the disequilibrum model equations 2 and 5, where stickiness is assumed in the supply of exports, are used. In both cases the equilibrium condition for equation 3 is imposed.

1. THE EQUILIBRUM MODEL

The equilibrium model is estimated using the Full Information Maximum Likelihood (FIML) estimation method. The method has the advantage of taking all restrictions into account while estimating. Moreover, it has excellent asymptotic properties. On the down side, the method is sensitive to specification problems in any one of the equations, since that situation will affect the entire system. The estimation proceeds by maximizing the likelihood function of the model. In this model, the estimation is undertaken by imposing a constraint on the world and domestic prices in the supply equations. The constraint ensures an identical coefficient, with opposite signs, for the two regressors (see Table 3 below for the unconstrained version). Results of this equilibrium model are given in Table 1.

Table 1 Estimates for the Constrained Equilibrium Model, FIML& 3SLS

Agricultural Raw Materials (FIML)

$$\Delta \ln X_{t} = -0.01 + 0.08 \ln \left(\frac{\Delta P_{w}}{\Delta P_{s}}\right)_{t-1} + 0.30 \Delta \ln \left(\frac{I_{s}}{Y_{s}}\right)_{t-1}$$
(0.02) (0.07) (0.27)

 $\Delta \ln P_{w_{t}} = -0.19 + 1.45 \Delta \ln Y_{N_{t}} - 0.23 \Delta \ln i_{w_{t}} + 3.1 \Delta \ln P_{N_{t}} - 0.98 \Delta \ln X_{t}$ $(0.10) \quad (0.58) \quad (0.19) \quad (1.4) \quad (1.65)$

Beverage (FIML)

$$\Delta \ln X_{t} = -0.01 + 0.10 \ln \left(\frac{\Delta P_{w}}{\Delta P_{s}}\right)_{t-1} + 0.30 \ln \Delta \left(\frac{I_{s}}{Y_{s}}\right)_{t-4}$$
(0.02) (0.07) (0.33)

$$\Delta \ln P_{w_{t}} = -0.17 + 0.76\Delta \ln Y_{N_{t}} - 0.36\Delta \ln i_{w_{t}} + 2.5\Delta \ln P_{N_{t}} - 0.08\Delta \ln X_{t}$$
(0.14) (0.83) (0.25) (1.9) (1.6)

Food (3SLS) (FIML which could not converge)

$$\Delta \ln X_{t} = 0.06 + 0.07 \ln \left(\frac{\Delta P_{w}}{\Delta P_{s}}\right)_{t-1} + 0.001 \ln \Delta \left(\frac{I_{s}}{Y_{s}}\right)_{t-2}$$
(0.05) (0.19) (1.02)

$$\Delta \ln P_{w_{t}} = -0.16 + 1.2\Delta \ln Y_{N_{t}} - 0.07\Delta \ln i_{w_{t}} + 2.9\Delta \ln P_{N_{t}} - 0.55\Delta \ln X_{t}$$
(0.11) (0.78) (0.21) (1.5) (0.30)

Minerals, Metals and Ores (2SLS) (FIML failed)

$$\Delta \ln X_{t} = -0.11 + 0.33 \ln \left(\frac{\Delta P_{w}}{\Delta P_{s}}\right)_{t-1} + 2.8 \ln \Delta \left(\frac{I_{s}}{Y_{s}}\right)_{t-1}$$
(0.13) (0.75) (2.7)
$$\Delta \ln P_{w} = -0.08 + 1.5 \Delta \ln Y_{w} - 0.03 \Delta \ln i_{w} + 1.13 \Delta \ln P_{w} - 0.02$$

 $\Delta \ln P_{w_{t}} = -0.08 + 1.5\Delta \ln Y_{N_{t}} - 0.03\Delta \ln i_{w_{t}} + 1.13\Delta \ln P_{N_{t}} - 0.02\Delta \ln X_{t}$ (0.12)(0.76)
(0.25)
(1.9)
(0.11)

Notes:

Figures in bracket are standard errors. In almost all cases the values are similar to the hetroschedastic consistent standard errors.

 R^2 and other indicators such as D.W. are not reported, due to their ambiguity in the simultaneous model. For instance the value of R^2 is bounded by (- ∞ , 1), not (0, 1). Hence small values are not a good indicator of 'poor fit' (see Goldstein & Khan 1978: 278).

For minerals, both estimations methods (FIML and 3SLS) yield nearly identical results. Hence we report only the FIML-based results.

2. THE DISEQUILIBRUM MODEL

The estimation of the disequilibrum model using equations 2 and 5.1 was not successful. The estimation for agricultural raw materials and minerals also failed to converge. In Table 4.2, estimation results of food and beverage, respectively, are reported. The model is estimated using a Three Stage Least Square Iterative (3SLSI) procedure, by fully taking into account the non-linear constraints of the entire structure of the model. As a generalized least square method this takes the residual of the reduced form equation, estimated by the Two Stage Least Square (2SLS) method, as well as their covariance matrices for transforming the estimating equations iteratively, until convergence is achieved. The procedure is asymptotically Full Information Maximum Likelihood. The estimated results following this procedure are reported below.

Table 2 The Disequilibrum Model: 3SLSI

Food

 $\Delta \log X_{t} = 0.05 + 0.02 \Delta \log P_{w_{t-1}} + 0.02 \log \Delta P_{s_{t-1}} - 0.07 \log \Delta \left(\frac{I_{s}}{Y_{s}}\right)_{t-2}$ (0.05) (0.0.19) (0.12) (0.83)

 $\Delta \log P_{w_t} = 24.8 - 172.4 \Delta \log Y_{N_t} - 46.2 \Delta \log i_{w_t} +$

$$18.8\Delta \log P_{N_t} - 435.6\Delta \log X_t - 52.3X_{t-1}$$

$$\begin{array}{c} \lambda = 0.88 \ a_0 = -0.06 \ a_1 = 0.45 \ a_2 = 0.12 \ a_3 = -0.003 \ a_4 = -0.05 \\ (0.83) \ (0.07) \ (3.4) \ (1.01) \ (0.02) \ (0.7) \end{array}$$

Beverage

$$\Delta \log X_{t} = -0.07 + 0.16 \Delta \log P_{w_{t-1}} + 0.46 \log \Delta P_{s_{t-1}} + 0.40 \log \Delta \left(\frac{I_{s}}{Y_{s}}\right)_{t-4}$$
(0.06) (0.12) (0.26) (0.44)

$$\Delta \log P_{w_{t}} = -0.84 + 0.93\Delta \log Y_{N_{t}} - 1.29\Delta \log i_{w_{t}} + 10.51\Delta \log P_{N_{t}} + 21.55\Delta \log X_{t} + 12.X_{t-1}$$

$$\lambda = 0.44 \ a_{0} = -0.09 \ a_{1} = 0.10 \ a_{2} = 0.10 \ a_{3} = -0.13 \ a_{4} = 1.1 \\ (0.38) \ (0.07) \ (0.58) \ (0.21) \ (0.08) \ (0.91)$$

Notes: As given under Table 1.

Figures in bracket are standard errors and refer to structural coefficients. In almost all cases the values are similar to heteroschedastic consistent standard errors. The plausibility of the parameters should be inferred from the coefficients of the structural equations.

3. THE UNCONSTRAINED ESTIMATION

Results reported in this section are based on an unconstrained model. That is, the world price and the domestic price coefficients, unlike in Table 1, are not constrained to have identical values.

Table 3 Estimates of the Equilibrium Model, FIML& 3SLS

Agricultural Raw Materials (FIML)

$$\Delta \ln X_{t} = -0.02 + 0.04 \Delta \ln P_{w_{t-1}} + 0.12 \ln \Delta P_{s_{t-1}} + 0.30 \Delta \ln \left(\frac{I_{s}}{Y_{s}}\right)_{t-1}$$
(0.02) (0.09) (0.09) (0.29)

$$\Delta \ln P_{w_{t}} = -0.17 + 1.4\Delta \ln Y_{N_{t}} - 0.23\Delta \ln i_{w_{t}} + 2.9\Delta \ln P_{N_{t}} - 0.71\Delta \ln X_{t}$$
(0.10)(0.57) (0.19) (1.37) (1.32)

Beverages (3SLS)

$$\Delta \ln X_{t} = -0.01 + 0.10 \Delta \ln P_{w_{t-1}} + 0.11 \ln g \Delta P_{s_{t-1}} + 0.32 \ln \Delta \left(\frac{I_{s}}{Y_{s}}\right)_{t-4}$$
(0.04) (0.09) (0.13) (0.45)

$$\Delta \ln P_{w_t} = -0.20 + 0.59 \Delta \ln Y_{N_t} - 0.50 \Delta \ln i_{w_t} + 3.4 \Delta \ln P_{N_t} - 1.56 \Delta \ln X_t$$
(0.14) (0.83) (0.25) (1.9) (0.57)

Food (3SLS)

$$\Delta \ln X_{t} = 0.06 + 0.06 \Delta \ln P_{w_{t-1}} + 0.006 \ln \Delta P_{s_{t-1}} + 0.002 \ln \Delta \left(\frac{I_{s}}{Y_{s}}\right)_{t-2}$$

$$(0.08) \quad (0.19) \qquad (0.29) \qquad (1.02)$$

$$\Delta \ln P_{w_{t}} = -0.16 + 1.2\Delta \ln Y_{N_{t}} - 0.08\Delta \ln i_{w_{t}} + 2.9\Delta \ln P_{N_{t}} - 0.59\Delta \ln X_{t}$$

$$(0.11) \quad (0.77) \qquad (0.20) \qquad (1.4) \qquad (0.29)$$

Minerals and Ores (FIML)

$$\Delta \ln X_{t} = 0.05 - 0.48 \Delta \ln P_{w_{t-1}} - 0.08 \ln \Delta P_{s_{t-1}} + 3.9 \ln \Delta \left(\frac{I_{s}}{Y_{s}}\right)_{t-1}$$
(0.27) (1.1) (1.1) (3.8)

 $\Delta \ln P_{w_{I}} = 0.51 + 0.59 \Delta \ln Y_{N_{I}} - 0.23 \Delta \ln i_{w_{I}} + 0.18 \Delta \ln P_{N_{I}} - 0.11 \Delta \ln X_{I}$ (0.40) (0.82) (0.16) (1.4) (0.18)

Notes: As given under Table 1.

Appendix 5.1 The Structure of Heller's Model

Heller's model is set as a maximization problem of a utility function [U], given as equation 1a, subject to two budget constraints, given as equations 2a and 3a.

$$U = \alpha_{0} + \alpha_{1}(I_{g} - I^{*}_{g}) - \frac{\alpha_{2}}{2}(I_{g} - I^{*}_{g})^{2} + \alpha_{3}(G_{s} - G^{*}_{s}) - \frac{\alpha_{4}}{2}(G_{s} - G^{*}_{s})^{2} + \alpha_{5}(G_{c} - G^{*}_{c}) - \frac{\alpha_{6}}{2}(G_{c} - G^{*}_{c})^{2} - [1a]$$

$$\alpha_{7}(T - T^{*}) - \frac{\alpha_{8}}{2}(T - T^{*})^{2} - \alpha_{9}(B - B^{*}) - \frac{\alpha_{10}}{2}(B - B^{*})^{2}$$

where: I_g is government investment, G_s government 'socioeconomic' consumption, G_c government expenditure on civil consumption, T taxation and B is borrowing. Asterisks (*) indicate a target level for each variable. This functional form is chosen to ensure diminishing marginal utility from each of the arguments.

$$I_g = B + (1 - \rho_1)T + (1 - \rho_2)A_1 + (1 - \rho_3)A_2$$
 [2a]

$$G_s + G_c = \rho_1 T + \rho_2 A_1 + \rho_3 A_2$$
 [3a]

where: A_1 and A_2 are total foreign grants and loans, respectively, and ρ 's are budget constraint parameters.

Before optimizing this function, each of the desired levels for the choice variables, are further specified, as given below:

$$I_g^* = \alpha_{11} Y_{t-1} + \alpha_{12} I_p$$
 [4a]

where: Y is total output and Ip real private investment.

$$T^* = \alpha_{13}Y_t + \alpha_{14}M_{t-1}$$
 [5a]

where: *M* is total real imports.

$$G_{c}^{*} = \alpha_{15} G_{c,t-1}$$
 [6a]

$$G_s^* = \alpha_{16}E + \alpha_{17}Y_t + \alpha_{18}(Y_t - Y_{t-1})$$
[7a]

where: E is primary school enrolment.

$$B^* = 0$$
 [8a]

Having arrived at the desired level of the choice variables, using equations 4a to 8a, equation 1a is optimized with respect to current policy variables, subject to the budget constraint 2a and 3a. The structural equations required for the estimation are obtained from the first-order condition of the optimization, using Lagrangian multipliers associated with the constraints. This is the basic Heller model. See also Mosley et al. (1987) for an extension of this basic model, and White (1993) for a reduced version of the model. Binh and McGillivray (1993) present a justification for avoiding the linear term in Heller's model.

Appendix 5.2 Estimation Using Other Approaches

In this appendix we will briefly outline the results of the ECM-based estimation, using an alternative specification. I have used the main variables included in van Wijnbergen's (1986a) estimation, based on the African data. The explanatory variables are limited to real GDP and a one-period lag capital inflow indicator. We could not use current and one-period lag capital, as was done in van Wijnbergen (1986a), due to problems of multicolinearity.

The result of the estimation varies both across regions and commodities. The estimation using the ESA data yields a number of interesting results. With some ambiguity, it would appear that the 'Dutch disease' effect is evident for beverage and, to a greater extent, agricultural raw materials. However, this does not hold for the other commodities. In relation to minerals, a relatively high cost of adjustment, or the contractual nature of mineral extraction activities by transnational corporations, might represent alternative explanations for the absence of a 'Dutch disease' effect. Moreover, in the case of both minerals and foods, the rise in the real exchange rate might also be partially explained by the level of devaluation. In the specification used in the text of this book, such an effect is assumed to be captured by explicitly specifying devaluation as one of the regressors. The findings in relation to the WCA region support the hypothesis of the existence of a 'Dutch disease' effect for beverages, and also suggests its possibility for food. However, this hypothesis was not found to be statistically acceptable in relation to other commodities in that region. For the NA region the 'Dutch disease' effect is confirmed for minerals, and there is a strong empirical suggestion that this effect is also evident for food. However, for agricultural raw materials no such effect would appear to be in evidence.

These results should be interpreted with some caution, for a number of reasons. It has been observed that, in a situation where the regression coefficient of the real exchange rate (RER) with foreign inflows (FF) is either

positive or not statistically different from zero, a separate regression of the domestic price on FF shows a positive and statistically significant sign. Under such circumstances, a positive relationship in the model value may be possible, if there is an offsetting increase, either in the foreign price of commodities, or the level of the nominal exchange rate arising from a devaluation. Thus, the inclusion of terms of trade and devaluation as regressors within this study may account for this effect.

In Chapter 4, we have used debt stock data within the ECM. In such a formulation, the short run value for change in debt stock shows the flow variable in which 'Dutch disease' effects are expected, while its long-run value, or level, may indicate a debt overhang. Estimation for the three regions, using such an approach yields mixed results. The estimation suggests a negative relationship between capital inflows and exports of beverages, in all regions. This might implicitly suggest the existence of a 'Dutch disease' effect.

| | | | | | | | | | | · normali | | |) | |
|------|--------------|--------|-----------|----------|-------|--------|----------|----------|------------|-----------|--------------|-------------|----------|------------|
| Eq. | Const. | | Short-run | t-values | | | -ong-run | t-values | | 12 | u U | L | 1 | Estimat. |
| no. | value a * | ø | q | c | q | q | J | q | ECM | auj. K | оп | L | 5 | method |
| en σ | 06- | v27 0. | 94 | 4 3 | 1 34 | 50.8 | 5.0 | | | 0.93 | 0.76 0.19 | 1632 7 9 | 55 20 | ors ors |
| ESA |) i | | P | þ. | 2 | | | | | | 2 | 2 | 2 | 0.0 |
| 21 | 1.8 | 6.9 | -11.0 | 0.88^ | | 4.9 | 1.8 | | -8.7 | 0.79 | 0.22 | 37.3 | 79 | ECM1 |
| 2 | 2.4 | 5.3 | -1.71 | 6.9 | 1.9 | -0.67^ | 2.1 | 2.0 | -6.4 | 0.43 | 0.14 | 7.3 | 94 | ECM1 |
| 29 | | | 1.91 | | | 5.5 | | | -9.2 -9 | 0.35 | 0.61 | 8.8 | 72 | ECM1 |
| 34 | | ecm2 | | o | clag | | | | | | | | | |
| | | 7.31 | 0.80^ | 2.8 | -1.9 | 1.7+ | 1.9 | | -2.9 | 0.47 | 0.05 | 14.8 | 96 | ECM2 |
| AA | | | | | | | | | | | | | | |
| 3 | | | -7.9 | | 0.54^ | -2.0 | | 6.0 | -0.8 | 0.77 | 0.25 | 26.7 | 39 | ECM1 |
| 22 | 3.3 | 2.4 | -1.9 | 2.0 | | 2.7 | -0.1^ | | -5.0 | 0.30 | 0.20 | 4.4 | 58 | ECM1 |
| 33 | 1.8 | 5.6 | -7.1 | | | 1.7+ | | | 9.9 9 | 0.75 | 0.22 | 28.3 | 56 | ECM1 |
| 29 | | | -0.62^ | | | 5.7 | | | -5.6 | 0.36 | 0.64 | 11.3 | 55 | ECM1 |
| 34 | -0.40 | -1.9 | -0.3^ | 4.0 | | 0.4^ | 2.32 | | -2.02 | 0.30 | 0.06 | 4 | 38 | ECM1 |
| WCA | | | | | | | | | | | | | | |
| 21 | 0.41 | 2.2 | -11.2 | 1.5- | | -1.7+ | 1.0 | | -4.5 | 0.65 | 0.32 | 27.6 | 115 | ECM1 |
| 22 | 2.7 | 5.4 | -2.5 | 4.0 | 4.1 | -0.51^ | 3.7 | 2.3 | -5.7 | 0.41 | 0.16 | 7.11 | 68 | ECM1 |
| 33 | 4.4 | 6.1 | -7.6 | 1.4^ | | 0.02^ | | | 4.4 | 0.84 | 0.23 | 25.8 | 33 | ECM1 |
| 29 | -1.02 | -2.8 | 6.7 | | | 3.9 | | | 4 | 0.36 | 0.59 | 18.8 | 96 | ECM1 |
| 34 | | ecm2 | • | b lag | O | | | | | | | | | |
| 34 | | 4.4 | 1.7+ | -2.0 | 1.5~ | 0.55^ | 2.2 | | -1.8 | 0.48 | 0.11 | 9.9 | 58 | ECM2 |
| 41 | -1.16 | -7.7 | 11.02 | | | 2.53 | -1.9 | | -6.6 | 0.57 | 0.30 | 39.4 | 117 | ECM1 |
| | | | | | | | | | | | | | | |

Test statistics of estimated parameters of the model not reported in other chapters Appendix 6.1 * The coefficient of the constant before the calibration; All are significant at 5% or better. ^A not significant; + (-) significant at 10% (or closer). Adj.R² is adjusted R², SE= standard error of regression; F= F-value, n=number of observation; ECM1 (ECM2) first (or second) order error correction model.

Appendix 6.1

Notes

Chapter 1

- 1. Naturally, these aggregated figures show only an 'average' scenario for African nations in general. However, there are significant exceptions to the picture painted by these statistics. Specifically, Burundi and Guinea-Bissau, in the WCA region, had a debt service ratio of 40 and 94 per cent, respectively, by 1992, while Uganda and Madagascar in the ESA region had ratios ranging from 40 to 70 per cent and from 50 to 60 per cent, respectively, from the mid-1980s. In relation to debt to GNP, Mozambique recorded a ratio ranging from 300 to 580 per cent from the mid-1980s to the early 1990s, while Guinea-Bissau had a debt to GNP ratio of between 130 and 300 per cent from 1980–90. Finally, the debt to GNP ratio for Congo and Cote d'Ivoire stood at close to 200 per cent during the mid-1980s (source: World Bank, World Debt Tables, electronic, 1994).
- 2. However, the Bank acknowledges that many African nations were faced with unfavorable terms of trade during the early 1980s.
- 3. In subsequent publications notably Africa's Adjustment and Growth in the 1980s, published jointly with UNDP – the Bank argues forcefully that Sub-Saharan Africa has been in relatively 'good shape' compared to other parts of the developing world and that policy mistakes have been the principal cause of its economic crisis. However, the ECA (1989b) argues that the Bank has based its conclusions on 'pseudo-statistics' and selective reporting. Re-examination of the data by ECA analysts would tend to suggest that the Bank's argument cannot be substantiated (see ECA 1989b and Mosley & Weeks 1993 for a brief summary).
- 4. However, according to the Bank, the effects of the protectionist policies of developed nations may be rendered less significant due to the low capacity of African manufacturing, an inability to produce temperate products as well as the continent's preferential status within the

EEC. See also Amjadi et al. (1996) for a recent argument along these lines, as well as proposals for a possible policy conditionality plan for privatizing African shipping lines.

- 5. See White (1996a) for a review of this debate.
- 6. In contrast, Collier and Gunning (1999) argue that lack of openness represents one of the major causes of poor performance of African economies.
- 7. This basically includes the system of government, public enterprises, the private sector, domestic markets, research and development, forces of nature and climate, ethnicism and society's value system, external commodity markets and finance and transnational corporations.
- 8. Collier (1991) cites the Zambian economy and copper price as a classic example of negative shocks. In Collier's opinion two errors are made. Firstly, the price fall was treated as temporary, and, secondly, foreign exchange shortages were handled by rationing. Notwithstanding an acknowledgment of the effect of negative shocks, he emphasized poor policies in what he called 'controlled' economies as representing a major problem. However, it could be argued that the root cause of these policy problems lies in the structure of the economy of these countries, and in their external trade in particular. Taken in this light, policy problems, per se, may be of only secondary importance.
- 9. However, Ghura (1993) is extremely optimistic in stating that judicious macro and trade policies may stimulate growth in Africa, even if external conditions do not improve. This viewpoint is essentially similar to the types of empirical studies undertaken in support of banktype policies.
- 10. This is measured as the divergence in the rate of growth of a country's exports from that of the world as a whole, over the period under study, multiplied by the total exports of the country in question. This is taken from a simple model, which specifies the different factors affecting exports (see Stein 1977: 106).
- 11. See Leys (1996) and Ofuatey-Kodjoe (1991) for critiques of dependency theory in the African context.
- 12. Makandawire (1989 cited in Elbadawi et al. 1992) summarizes the two contending views about the cause of African crisis as structuralist and neoclassical. He notes:

The structuralist view is one which highlights a number of features and 'stylized facts' that almost every point contradicts the neoclassical view ... class based distribution of income rather than marginal productivity based distribution of income; oligopolist rather than the laissez-faire capitalist market; increasing returns or fixed proportion functions rather than 'well-behaved' production functions with decreasing returns and high rates of substitution; non-equivalent or 'unequal exchange' in the world rather than competitive, comparative advantage based world system; low supply elasticities rather than instantaneous response to price incentives (Makandawire 1989). (Elbadawi et al. 1992)

- 13. See Stewart (1993) for a discussion of this issue.
- 14. See, however, Helleiner (1993) who argues for an emerging consensus on this issue.
- 15. See Amin (1974, Chapter Two) on the mercantilist period.
- 16. Amin (1972) has termed this the pre-mercantile period.
- 17. Wallerstein characterizes the trade of the period as trade in 'luxuries', with such trade being undertaken between external arena and not in an integrated world economy framework. Wallerstein and Amin define luxuries as those goods, the demand for which comes from the part of the profit that is consumed. Suraffa defines luxuries as goods which are not used in the production of other goods. He, however, took it as trade/exchange in which 'each can export to the other what is in his system socially defined as worth little in return for the import of what in its system is defined as worth much'. Or, in Alpers' phrase, 'trade from which each side believed itself to be profiting' (Wallerstein 1976: 31 and footnote 3).
- 18. Maghreb refers to North Africa.
- 19. This stands in sharp contrast to the current categorization of North Africa as geographically and economically distinct from Sub-Saharan Africa. For justification of this view see Sommers and Assefa (1992) and various World Bank/IMF classification schemes for Africa.
- 20. The original work is written in 1933.
- 21. The commonly argued case that since Ethiopia was not colonized, it represents a 'counter factual' for how other parts of Africa might have developed in the absence of colonialism, is a very weak one. Firstly, a good part of the history of Ethiopia has been a history of wars under the ideology of either religion, region, nationality or a combination of these. This has created a serious crisis in the agricultural sector (see

Gebrehiwot 1917). Secondly, Ethiopia's history has been characterized by two clearly distinct antagonistic classes: the landed aristocracy and the peasantry, with corresponding state structures (see Gebru 1995). Given the history of conflict which characterizes Ethiopia's history, the main preoccupation of the landed aristocracy and church has been to maintain its power. Thirdly, colonialism had the effect of disrupting the dynamic caravan trade which linked the Southwest parts of Ethiopia to the rest of the East African region. And, finally, Ethiopian independence was basically a besieged one. Since it was encircled by hostile and powerful colonial forces, naturally this had an influence on the political and economic structure of the country. More specifically, Ethiopia developed as a militaristic nation, with a dependent economy based on the export of commodities and import of manufactures.

- 22. First by the Portuguese, and later by the British, Dutch, Germans and Scandinavians.
- 23. In describing the impact of underdeveloped nations' interaction with Western Europe, Baran noted

[the population of these nations] found themselves in the twilight of feudalism and capitalism enduring the worst features of both worlds. Their exploitation is multiplied, yet its fruits were not to increase their productive wealth; these went abroad or served to support parasitic bourgeoisie at home. They lived in abysmal misery, yet they had no prospect of a better tomorrow. They lost their time-honored means of livelihood, their arts and crafts, yet there was no modern industry to provide new ones in their place. They were thrust into extensive contact with advance of the West, yet remained in a state of the darkest backwardness. (Baran 1957: 144)

Perhaps we should not be surprised that Baran's description, written nearly four decades ago, remains relevant today.

- 24. Imports of palm oil by Britain, groundnuts by France, palm kernels (for cattle cake) by Germany and (for the manufacturing of margarine) by the Dutch represented the main items traded during the 19th century, prior to the onset of formal colonialism at the end of that century. (For a description of this, see particularly Chapter 4 of Hopkins 1973.)
- 25. These were prompted by the so called 'cotton famine' in Europe, following the American civil war.
- 26. The motives underlying colonialism represent a widely debated topic. For instance, Austen (1987) argues that

within [the] general context of intense multifaceted international competition, the economic rational for African colonization was to a considerable extent pre-emptive, designed to assure access to potential rather than actual markets and commodities as well as trade routes ... to Asia. (Austen 1987: 116)

- 27. There are many examples of Africans being forced into cash crop production. This occurred in Tanganyika (today's mainland Tanzania), in the Portuguese colonies, in French Equatorial Africa and French Sudan (today's Mali). In Congo Brazzaville the French enforced cotton cultivation by banning traditional agricultural activities. These policies of coercion were resisted to the extent possible. The revolts in Tanganyika and Angola represent cases in point (see Rodney 1972: 172–81, Austen 1987: 140–42).
- 28. This was the policy followed by Germany in what is now called Namibia. Indeed, the extermination of the Africans was so extensive that, when they discovered diamond, the Germans had to look for migrant labour for mining from other regions (see Longmire 1990: 203-4).
- 29. The English translation appeared in 1979.
- 30. See also Amin (1972) for a political and social analysis of how the region's commodity production and exports were controlled.
- 31. Pim places this at the center of his investment analysis and argues that the main investment was in areas with extensive mineral wealth, plantation possibilities and a mass of unskilled labour. This involved heavy expenditure in communications, which required an expansion of the export sector for its finance. The latter, in turn, required a large labour supply, which was secured by direct and indirect compulsion, affecting every aspect of native life (Pim 1977: 229).
- 32. France was in possession of such a protected market in West Africa. The protectionist policy was the result of pressure from French metallurgical, textile and chemical industries, which had difficulty competing with Britain (Hopkins 1973: 160). Portuguese industrialists had also created such protected markets in Africa, especially for their textile industry (Seleti 1990: 36).
- 33. Portuguese colonialism does not qualify as 'matured' in his analysis.
- 34. In virtually all African countries, one to three commodities account for 50–90 per cent of total exports. Indeed, in the period 1982–86, in 13 African countries 1 product, in 8 countries 2 products, in 6 countries 3

products, and finally, in 8 African countries 4 products accounted for over 75 per cent of export earning (see Adedeji 1993 for details).

Chapter 2

- 1. See also Vos (1994) and Helleiner (1989: 1451) for a detailed discussion on this.
- 2. In contrast to this approach, another strand of the literature focuses on 'immiserization of the recipient'. In their study of the implication of capital flows Brecher and Diaz-Alejandro (1977) showed that capital inflows to small open economies would reduce the recipients welfare, measured in terms of consumption. This condition is vividly illustrated when foreign profit remittance absorbs the increase in total output due to capital inflow (Brecher & Diaz-Alejandro 1977: 319). Quibria (1986), using a simple model of 'North' (growing à la Solow and saving in a Keynesian way) and 'South' (saving in a Kaleckian way), concluded that capital flows (even in a situation of no distortion) may result in immiserization of the recipient.
- 3. Magdoff (1992) provides some statistics on FDI for the recent past. Thus, in 1960, 47 per cent of FDI originated from the US, 45 per cent from Europe and 0.7 per cent from Japan. The figures for 1973 stood at 48, 39 and 4.9 per cent respectively, and at 28, 50.2 and 11.5 per cent respectively, for 1989. Magdoff also notes that, contrary to popular belief, the bulk of FDI actually goes to industrialized rather than Third World nations, with the latter's share declining from 31 per cent in 1967 to 19 per cent in 1989. In relation to the structure of FDI, in 1990 investments in finance, insurance and banking in LDCs were almost one-third higher than in manufacturing (40.1 per cent versus 31.3 per cent of total investments).
- 4. See also Jungnickel (1993) on the recent pattern towards investment in services.
- 5. This optimization is described as the borrowing problem in finite horizon and is given by:

max $U(C_1, C_2 \dots C_T)$ Subject to: $Q_t = F(K_t L_t)$ $K_{t+1} = K_t(1-d) + I_t$

$$C_{t} = (Q_{t} - rD_{t}) - I_{t} + (D_{t+1} - D_{t})$$
$$D_{t} \le \max \sum_{i=1}^{T} (1 + r)^{(t-i-1)} (Q_{i} - I_{i})$$

where: Q = output, K = capital, L = labour, d = depreciation, I = investment, C = consumption, D = stock of debt, r = discount rate and t = time.

- 6. This can be understood as 'the absence of a legal framework to limit the borrowers liability and safeguard the lender's right' (Folkerts-Landau 1985: 326). This differs from domestic credit, where the lender is protected by bankruptcy law or collateral.
- 7. It should be noted that such an approach was not intended to address the risk of individual projects, as had been the case with earlier approaches. Rather, this approach was characterized by public guarantees of debts and 'cross-default clauses' which were related to the ability of the country, as opposed to individual borrowers within the country, to pay the external debt. This was enforced by *syndication* of lending banks. Folkerts-Landau refers to these syndication and cross-default clauses as *the new financial innovations*.
- Indeed, a number of theories focus on the behavior of bank managers, specifically in relation to loan disbursement and risk-taking (see for instance Darity & Horn 1988).
- 9. Even such categorization is difficult since some of the studies focus on both explanations. Nevertheless, it is legitimate to divide these studies along this line based on the degree of emphasis the authors placed on these alternative explanations.
- 10. However, as Gasper (1992) points out, the aid literature has traditionally been dominated by economists, who, as a discipline, have traditionally been less concerned with ethics. For this reason, he argues, ethical concerns have not been rigorously dealt with (Gasper 1992: 10).
- 11. As Mosley et al. note: (a) the right have argued against the allocative case for aid, while (b) the left argue that there has been an improper focus on the distribution of income rather than on issues of power. Mosley maintains that, while the latter proposition is a matter of value judgment, the former represents a testable proposition (Mosley et al. 1987: 14). Gasper (1992) argues that Mosley et al.'s (1987) study is governed by the degree of faith he places in the power of empirical analysis to identify effects as opposed to informing value judgments.

He concludes 'this positivism helps explain weaknesses in identifying the structure of the evaluative arguments on aid' (Gasper 1992: 9). In effect, Gasper seems to be arguing in favour of making a priori value judgments before moving on to research the quantifiable aspect of a problem. His criticism would have been reasonable if it was limited to the degree of emphasis placed, as opposed to the important issue of studying the quantifiable (or what he labels the positivist) aspect.

- 12. However, Miksell appears to discount humanitarian considerations in noting how the US acted against President Nasser of Egypt, in spite of an undertaking to help the poverty-stricken people of that nation (Miksell 1968: 7).
- 13. The authors group the studies into six categories based on implicit restrictions imposed on a general form, which includes three of the stated motives for aid (excluding the humanitarian one). These sub-groups are: the recipient need/donor interest; hybrid (which comprehensively models the above three motives); bias (which deals with small country and income bias in allocation); developmental (which is similar to the recipient-need model, i.e., allocation is based on the developmental needs of the recipient, but does not include donor interest); administrative/incremental (in which allocation is based on past experience), and, limited dependent variable studies (which take into account eligibility for aid).
- 14. However, some economists, such as Kindleberger, have argued that there is nothing 'new' in the new trade theories. Indeed, they argue that some of the ideas presented in these theories had already been discussed, for instance in Ohlin (1933) (see Krugman 1992: 425). There is also no clear consensus as to which theories should be included in the 'new', with Stewart (1984), for instance, including technological gap models within this category.
- 15. Greenway and Milner (1986) classify intra-industry trade theories under a number of broad categories. Firstly, Neo-Heckscher-Ohlin, intraindustry trade models in which factor endowment is seen as important. Secondly, Neo-Chamberlinian models of monopolistic competition, in which intra-industry trade is analysed in terms of a monopolistic competition framework. And, finally, Neo-Hotelling models of monopolistic competition, which is essentially similar to the latter, but differs in terms of its treatment of product differentiation (Greenway & Milner 1986: 1-22).

16. Indeed, Emmanuel poses this question even more graphically, in stating

The copper of Zambia or the Congo and the gold of South Africa are no more primary than coal, which was only yesterday one of the chief exports of Great Britain, sugar is about as much manufactured than Scotch Whisky or the great wines of France; before they are exported coffee, cocoa and cotton have to undergo a machine processing no less considerable, if not more so, than in the case of Swedish or Canadian timber; bananas and spices are no more primary than meat or dairy products. And yet the price of the former decline while those of the latter rise, and the only common characteristic in each case is that they are, respectively, the product of poor countries and the product of rich countries. (Emmanuel 1972: xxx).

- 17. See Bloch and Sapsford (1996) for recent evidence along similar lines.
- 18. The Ethiopian economist Gebrhiwot (1917) has already discovered the terms of trade deterioration three decades before Prebisch and Singer
- 19. Prebisch notes how full factor mobility, especially of labour, would have lowered the high wages of US workers in his study. However, this finding is academic, for he states that 'the protection of this [USA's] standard of living, attained by great effort, should have prevailed over the uncertain advantage of an academic concept'. Thus he concludes, 'the classic rules of the game [free international trade and gains from trade] form an indivisible whole, and, if one is eliminated, the other cannot logically serve as absolute standards governing relations between the center and the periphery' (Prebisch 1962: 7).
- 20. Patnaik (1996) shows that even if export earnings are used for capacity creation in the primary sector, trade causes retrogression in this sector. This is because contraction of the manufacturing sector reduces the stimulus for expansion in the primary sector, and hence the expansion of the primary sector requires a larger level of public expenditure. However, the latter is adversely affected by contraction in the manufacturing sector, with further dynamic repercussion in the primary sector (Patnaik 1996: 215–16).
- 21. Unlike the classical economists, and even Keynes (1912, cited in Thirlwall 1991), who believed in diminishing return and hence an improvement in terms of trade to primary commodity producers, it was Kindleberger who first noted a secular decline in the terms of trade of primary commodity producers (Kindleberger 1943 cited in Thirlwall 1991; see also Sarkar 1986 and Thirlwall 1991 for details.)

- 22. See also Thirlwall (1991) for a brief summary and a concise survey of this empirical debate.
- 23. However, recently it is argued that the construction of this series might have an upward bias (see Sapsford & Balasubramanya 1994, footnote 2).
- 24. Based on this reconstructed data, the deterioration is 0.5 per cent per annum, rather than 0.7–0.8 per cent, as originally calculated by Prebisch.
- 25. The formulation of this equation differs from the original formulation of the model (Goldestin & Khan 1978) since it includes exchange rate, lagged price effect and supply shock, which is broadly defined to include other factors which influence exports from region R.
- 26. However, it should be noted that the equilibrium model applied by Bond incorporates those variations from the basic model, as listed in footnote 25 above.
- 27. This earlier work on short-run modelling is included at the end of this section, specifically within equations [6] and [7], in order that this might combine with the medium-run framework.
- 28. More specifically, these relate to the effects of current prices on future supply through investment and, in turn, the effect of change in capacity on commodity prices.
- 29. Thus, according to this classification scheme: (i) 'food' includes wheat, rice, soya beans, ground nut oil, coconut oil, palm oil, fish meal, sugar, meat and bananas; (ii), 'beverage' covers coffee, cocoa and tea; (iii) 'agricultural raw materials' includes cotton, wool, rubber, hides, jute, sisal, tobacco and timber; and finally (iv) 'metals and minerals' covers copper, iron ore, tin, aluminium, zinc, nickel, lead and phosphate.
- 30. Since there is no basic difference between the work of Adam and Behrhman, the model specified here, and that of Hwa, we will not describe Adam and Behrman's work. Rather, we hope that the model which we do describe, along with Haw's model, show the same basic features as contained within Adam and Behrman's (1976) model.
- 31. They have estimated the expected price equation by presuming that the excess of the expected rate of return, over the sum of the carrying cost and the interest rate obtainable on alternative financial assets must be an increasing function of the outstanding stock of commodities. This is done on the assumption that commodity stockholders may be compen-

sated for the risk associated with holding these stocks. Finally, a number of adjustments are made to this estimate, using time series data (see pp. 14-18).

- 32. Palaskas and Gilbert (1990) argue that storage disequilibrum is an implausible basis for a price adjustment theory. This discussion is postponed to Chapter 4 where this issue is revisited at the estimation stage.
- 33. Details on the production function form is reported to appear in the original research, which we do not have access to. However, for our own purposes a crude production function concept is sufficient.
- 34. Project LINK is an ongoing project, which works on connecting all world macro models (see Chapter 6 for further details on this).
- 35. This system is given in a linear matrix form as follows: $M_i = c_i X_i + Z_i$. By transforming the vector of c_i 's into a diagonal matrix C and representing the vector of f.o.b.-c.i.f. adjustment factors $1/\Lambda_j$ by ϕ , the model becomes:

$$M = CX + Z \dots (1')$$
 $X = A\phi M \dots (2')$ and $B = X - M \dots (3')$.

The solution to this system can be given by:

| $M = (I - CA\phi)^{-1}Z$ | (1") |
|--|------|
| $X = A \left(I - CA\phi \right)^{-1} Z$ | (2") |
| $B = (A - I)(I - CA)^{-1}Z$ | (3") |

I denotes the identity matrix and the superscript -1 is an operation for a matrix inversion.

- 36. For different commodity groupings, the functional form may vary slightly. Thus, imported consumption goods may become directly related to autonomous spending, which indirectly affects income, income affects consumption and consumption, in turn, determines imports. Raw material imports may also have a similar indirect link but, in this case, through the demand for inputs. In other words, autonomous spending affects demand, demand affects output and output, in turn, determines inputs. In relation to manufactured imports, these are generally directly linked to the import content of investment (Taplin 1973: 183–84).
- 37. While estimating the import function, due to lack of fit, statistical problems and gaps in data, some deviation from the assumed framework has had to be made. Specifically, for CEMA, imports are made a function of GDP, while for New Zealand, GNP is used, with previous

period exports being employed in the case of Turkey. For developing countries as a whole, imports are set to depend on foreign exchange earnings, defined as exports plus invisible balance, including capital inflow, and transfers in the current and preceding years.

- 38. Using a regression technique, market share elasticities are computed. This technique is used as follows. In general, the approach is to regress the change in current market share on proportionate changes in the current price ratio. Taplin (1973) applies such an approach both to total world trade, and to individual commodity groups. The commodity grouping results vary slightly from those obtained using the total. For instance, based on total trade, the elasticity is found to be approximately 0.43, while for commodity groups it varies from 0.15 to 1.275 (see Taplin 1973, Table 3, p. 193).
- 39. See Labys in Adam and Behrman (eds) 1978, Chapter 8.

Chapter 3

- 1. A detailed version of this model and its estimation method is reported in Olayide et al. (1980). A shorter explanation of aspects of the model structure, particularly in relation to supply, as well as a full specification of model structure are reported in Olafin and Iyaniwura (1983). The discussion here is based on the latter reference.
- 2. Boutros-Ghali and Taylor (1980) also propose a SAM-based macro model for Egypt. However, since the formulation of this model is governed by 'basic needs' considerations, which are not the objective of this study, we do not examine this here.
- 3. A modified version of this model is found in Benjamin (1990). However, this model differs from the previous one in a number of important respects. Firstly, in its treatment of investment behavior. Secondly, in how it deals with formal and informal financial markets. Thirdly, in the fact that it disaggregates sectors of the economy by institutions (both public and private). And, finally, in that it allows fiscal balance to affect interest rates and money supply. The model is then closed by making nominal exchange rate and foreign capital inflow exogenous. In this model, the rate of interest (as cost of borrowing for investment and reward to saving) is given a strong role in allocation of resources, although the income effect and the intertemporal prices effect on saving and consumption is believed to offset its effect. However, the rudimentary nature of most African countries' money markets and rigidity in mobility of resources seems to contradict this basic mechanism of

the 1990 model. Despite this setback the model underlines the importance of observing sectoral adjustment over time, which could differ at different stages of the sectoral adjustment process. (For example, in a situation of booming sector like that of oil in Cameroon, manufactures can contract in the short run but expand in the coming years).

- 4. See also Harton and McLaren (1989) for a shorter version of this model and its application to the analysis of alternative adjustment policies.
- 5. The tradable sector is, itself, further divided into manufacturing (importable) and exportable mineral resources.
- 6. In relation to the income identity, real income is calculated as absorption plus exports, net of imports, in real terms. Change in money supply is equated to government expenditure, net of its revenue, exports net of imports and change in private domestic credit and capital inflow.
- 7. Two exceptions to this include a fairly disaggregated but extremely simplistic Keynesian model developed by Asmerom and Kocklaeuner (1985), as well as quite a promising supply-driven model developed by Lemma (1993).
- 8. A similar CGE model for Rwanda, focusing on demand for working capital, domestic credit (as opposed to import) rationing and financial repression has been developed by Decaluwé and Nsengiyumva (1994).
- 9. Although the authors claim that theirs is the first generation of models in Africa, Harris' (1985) survey and the discussion in this section shows this not to be the case.
- 10. In fact, it could be argued that it would be timely to update Trap's (1993) work, based on subsequent experience of stabilization and structural adjustment efforts, as well as a number of other studies undertaken with the aim of assessing adjustment in Africa.
- 11. Trap labels these economists the 'neoclassical-structuralists'. However, it could be argued that this label represents a contradiction in terms.
- 12. Their survey begins by examining problems in using an aggregate regression approach to explore growth. This is tackled by examining the response of economic agents (i.e., rural households and manufacturers) in the face of risk. This analysis is augmented by examining the market in which these agents function. In all cases, agents' response is more often found to be behavioural rather than structure-determined in nature.

- 13. This represents a departure from Ndulu's (1986) original assumption of mark-up pricing. However, it could be argued that Ndulu's, rather than Rattsø's, assumption is the more realistic.
- 14. Ndulu (1991) is clear that the exogeneity of growth (i.e., dependence of growth on exogenous factors) will depend on the relative importance of these factors (say, barter terms of trade) *vis-à-vis* the endogenous one (say, volume of exports which he and Svedberg take as policy related). By reconstructing Svedberg's (1990) decomposition analysis in order to isolate these factors, for the period 1980–88, rather than 1970–85, originally used by Svedberg, he found, unlike Svedberg, a strong role for barter terms of trade.
- 15. Griffin's hypothesis that there exists a negative relationship between aid and domestic savings has been severely criticized from a host of different angles. His approach is found to be weak, owing to misspecification in the definition of savings and consumption within his model. Thus, consumption is assumed to be a function of income plus aid, while saving is not. This assumption effectively overlooks the feedback effect of aid on savings via income through the multiplier effect (see White 1992 for a full survey of these criticisms).
- 16. This is basically similar to the time period used in the above-cited works.
- 17. To my knowledge, there is no global econometric model for Africa. In fact, econometric-based global models are extremely rare in the literature. Hence, this initial attempt could serve as a useful starting point for further research in this extremely important area, which is vital in the ongoing globalization process.
- 18. This has its historical roots in the work of the French 'physiocrat' Quesnay. A rigorous discussion of national accounting was also undertaken in England, during the 17th century, based on the work of William Petty. This was followed by the work of Gregory King (1696, cited in Luttik 1992) who developed national income estimates for England and Wales showing per capita income, expenditure and saving. Numerous other writers have built on this work, over the centuries, culminating in the pioneering work of R. Stone and J. Meade, in collaboration with J.M. Keynes. The UN has also played a useful role in standardizing this work, through its recommendation of a 'System of National Accounts (SNA)', which was first introduced in 1953 (Luttik 1992: 16–17). The new SNA (UN et al. 1993) tackled many of the problems which have been discussed in this section and, it would

appear that analytical consistency between BOP and SNA has finally been achieved.

19. Electronic data for 21 African nations constructed on the basis of these alternative assumptions may be obtained from the author.

Chapter 4

- 1. Private capital flows will be examined in Chapter 6, using a creditrationing framework. Such grants are assumed to follow a similar pattern to official flows.
- 2. The econometric methodology adopted here is further discussed in Appendix 4.1.
- 3. This is defined as the sum of multilateral and bilateral aid flows.
- 4. In some of these countries, principal rescheduling began to occur again from the mid-1980s onward.
- 5. It is noted that for equal levels of per capita income, countries with a high population might receive a higher level of inflows. This could be interpreted as suggesting the need to use population size as one of the indicators. The functional form adopted (log-log model) entails that this will collapse to mere GDP ratio, if the population ratio for the two regions is included. However, the use of the per capita ratio for the regions allows for implicit recognition of this fact and hence this is the approach followed in this study.
- 6. This may be measured either using a proxy indicator, such as the number of commercial banks, or directly, in terms of the total credit disbursed.
- 7. According to Palaskas, the Ady (1968) model for cocoa and the French model (cited in Palaskas 1986) for apples are the only two such models published prior to 1960. During this period, although the supply response literature was extensive, specification of supply functions was rare (see Palaskas 1986: 12).
- 8. See Lim Lin Shu (1975: 67–69) for a comparison of these and relevant algebra.
- 9. A typical formulation of such an investment function is one which relates current investment I_i (which is the difference between the current and lagged level of capital stock, $K_i - K_{i-1}$) with that of the desired

 (K^*) and actual (K) level of capital stock with an adjustment variable α included. Formally this is given as,

 $I_t = K_t - K_{t-1} = \alpha (K^*_t - K_{t-1})$

(See Palaskas 1986 for an analysis of an alternative specification of K^* .)

- 10. At estimation stage these 'other' factors took different forms. For food and beverage, price of fertilizer and lagged stock level are used. For agricultural raw materials, a time trend and price of oil (as representing input price) is used. For metals, lagged stock level and interest rate (indicating cost of capital) are used.
- 11. See Chapter 3 of Guvenen et al. (1991) for an excellent survey of these developments.
- 12. This is an important result since one can use output figures as a proxy to area, data for which is hard to come by.
- 13. A simple OLS being the most common one.
- 14. See section 4.4 and Chapter 6. Hwa (1985) represents a good example of a study, which includes stockholding variables within commodity models, while Alogoskoufis and Varangis (1992) include macro variables in their commodity model.
- 15. Indeed, most studies use output of a commodity instead of exports on the assumption that change in output is easily convertible to change in exports. Export supply models, on the other hand, employ exports for this purpose, on the assumption that change in exports result from change in output.
- 16. The expected profit indicator applied by Chu and Morrison (1986) is again used, with minor changes. This may also be interpreted as a price instability index, since it is computed as the deviation of current price from a moving average price (at a specified lag), which is assumed to show the long-run average price.
- 17. In this study, a five-year lag is assumed.
- 18. See Bond (1987) for a discussion of the empirical validity of this argument.
- 19. As with FDI, capital inflows and other official flows are important for the mining sector.
- 20. Although, in a simple level-based OLS estimation, simultaneous equation bias may also arise from a lack of a complete demand and supply

model, the complete demand and supply model may be estimated simultaneously (as undertaken by Goldstein & Khan 1978). However, the limitation of simultaneous equation bias might not arise in relation to the use of the ECM approach. (See Kennedy 1994: 251, for example. See also Hamilton 1994 and Rao 1994 for a formal treatment of this issue.)

- 21. According to the UNCTAD classification, the category 'food' covers wheat, maize, rice, sugar, beef and veal, bananas, pepper and soybean and fishmeal. 'Tropical beverages' are defined as including coffee, cocoa and tea. Finally, 'vegetable oil seeds and oils' is taken as referring to soybeans, soybean oil, sunflower oil, groundnut, groundnut oil, copra, coconut oil, palm kernels, palm kernel oil and palm oil.
- 22. Specifically, Chow, Jarque-Bera and Reset along visual inspection.
- 23. And significantly negative for agricultural raw materials in the ESA region.
- 24. In fact, an ECM regression of domestic price (GDP deflator) on foreign inflows was run separately, in order to justify this proposition. The result indicates a strong positive relationship, with a likelihood of 'Dutch disease' effects for multilateral debt. Such debt represents an important flow to Africa.
- 25. This is likely to be determined by, amongst other things, the import content of industries, pattern of consumption of a section of the urban population, and levels of debt servicing and capital flight.
- 26. However, one problem with this hypothesis is in explaining why a rise in price is associated with lower exports, if the demand for foreign exchange is so elastic. See Bevan et al. (1987) for an alternative explanation for this apparently perverse relationship, based on the concept of rationing.
- 27. Current price is chosen in Model III, cognizant of the assumed correlation between foreign inflow and domestic price (which is the denominator in the relative price computation).
- 28. Subject to a cointegration test and exclusion of the possibility of multicoliniarity.
- 29. These are Botswana, Ethiopia, Kenya, Madagascar, Malawi, Tanzania, Uganda and Zambia.
- 30. See Alemayehu (1996) for further details on these.
- 31. Bilateral, multilateral, grants and private flows are all tried for this purpose.
- 32. The exceptions being the long-run capital formation in Model I and short-run capital inflow in Model III.
- 33. An example of this being the 'recovery rate' in tree crops.
- 34. This would appear to be a short-run value, although that is not stated clearly.
- 35. The 'long run' being defined here as long lags of five years. In spite of an error in their glossary in defining real and current prices, it would seem that this study is, in fact, based on real prices.
- 36. Although the coefficient of this variable is likely to be positive, the possibility of a negative coefficient cannot be ruled out. Specifically, this may occur if an increase in Northern income is associated with faster growth in domestic production than in consumption of importables (see Goldstein & Khan 1978: 276 and the reference cited there).
- 37. In the GK model the change in price is assumed to be represented by a partial adjustment process of the form,

 $\Delta \log P_{w} = \gamma \left[\log X_{t} - \log X_{t}^{ss} \right] \quad \gamma > 0$

By substituting a supply equation into this formulation and solving it for various prices, an equation for estimating the disequilibrum model, taken in conjunction with equation 5, is arrived at. GK and Abebayehu (1990) estimated these equations using a FIML method. My attempt to estimate a similar model rendered an extremely inferior result. This raises the question as to whether GK and Abebayehu's models were free from spurious regression. This issue will be re-examined in section 4.4.3.

- 38. Based on their empirical results, the model is invariably found to have a partial adjustment coefficient of less than one.
- 39. This might be trend stationary at a weak 10 per cent significance level.
- 40. Apart from the obvious fact that de-trended values hardly overcome the non-stationarity problem, we note that some de-trended series, such as indicators of domestic prices are found to be an I(2) series. For this reason a transformation based on first difference is preferred.
- 41. UNCTAD has reported such data from 1966 onwards.
- 42. Specifically, those models relating to demand for agricultural raw materials and beverage.

- 43. The exception relates to the supply of agricultural raw materials.
- 44. Defined, according the UNCTAD definition, as non-oil-exporting developing countries.
- 45. A more involved formulation using a Koyck transformation (1954) or the adaptive expectation formulation of Cagan (1956) requires the construction of an underlying model, in order to transform the unobserved series based on expected values, into terms comparable to observed values. However, this approach is not adopted in this study, since the underlying model is integrated into the aggregate demand equation and hence, is not explicit.
- 46. Future work using a longer series and, where possible, explicit stock data may shed light on this issue. However, I was not able to pursue this option, for lack of available data.

- 1. The association which Papanek is referring to is between low saving and high foreign capital inflow for common exogenous reasons. Bowles (1987) has attempted to investigate the causality question using the 'Granger Causality' concept. His analysis suggests that casual relationships are not universal and that, in countries where such a relationship does hold, it is found to depend on the structure of aid, whether from bilateral and multilateral sources or not. Gupta (1975) has used a simultaneous equation model which allows for both the direct and indirect effects of aid. He suggests that the negative effects of capital inflows have been grossly overestimated and that the total indirect and direct effects of such inflows may also be positive. However, his estimation of the savings function shows a negative coefficient, both with aid and with foreign capital inflows in general.
- 2. See White (1992) for a recent survey on this debate.
- 3. Public consumption is further disaggregated into civil consumption (comprising government administration, public debt servicing, security and diplomacy expenditure) and 'socio-economic' consumption (comprising non-capital current expenditure).
- Contrary to such demand optimism, it can be argued that the short-run macroeconomic impact of aid from the demand side could be contractionary (see Bhaduri & Skarstein 1996 for the theoretical derivation of such a possibility).

- 5. These may either be imported intermediate or capital formation.
- 6. The Pack and Pack model attempts to test the displacement hypothesis by theoretically relating it to the nature of indifference curves that the government is assumed to face. Thus, in its broadest sense it is a variant of the analysis, which is based on the utility maximization framework.
- 7. Since the time series variables which form the basis for this study have not been tested for the plausibility of the stationarity assumption, and since no other diagnostic tests have been reported, it is not possible to determine whether this doubt is, in fact, justifiable.
- 8. Although the constraint is meant to handle such a trade-off, the functional form adopted is similar to utility functions related to linear expenditure systems which are based on the additive assumption, which, in turn, requires independence among choice variables (see Alchian 1971, Koutsoyiannis 1975 for a discussion and further reference).
- 9. This inconsistency stems from the additive assumption upon which the framework depends.
- 10. The philosophical underpinnings of such an approach may broadly be labelled *critical realism* (see Lawson 1989 and Lawson et al. 1996).
- 11. This inevitably assumes the existence of a stable political structure as well as a government concerned with long-term issues. However, the prevalence of government short-termism in most African countries may result in a lower priority being attached to such long-term issues.
- 12. In principle, this may be captured by using a cost minimization function, subject to the desired growth/developmental expenditure target. This approach is similar to those cited above, except that the concern for budgetary equilibrium has been replaced by a concern for attaining the desired (downward adjustable) expenditure level, even if that implies incurring a budget deficit. However, such a function is difficult to quantify since these costs are political in nature and could vary, depending on the valuation placed on them by policy makers in different countries. However, in this study, an ordinal rating of the associated costs beginning with foreign inflows, for which costs are lowest, followed, in ascending order, by domestic revenue and deficit financing has been followed. This is assumed to be the relevant stylized fact in the African context. Formally, since such a cost function takes the form of $C = f(P_i, F_i)$ where i = 1,2,3 and P_i is the cost of using one of the financing mechanisms F_i subject to the constraint $Ig^* = g(F_i)$

where Ig^* is desired level of growth, the Lagrange-based optimization equation $f(f(P_i, F_i) - \lambda[g(F_i) - Ig^*])$ will yield an identical equation structure to the one outlined here if it is assumed that $P_1 < P_2 < P_3$.

- 13. A 'Dutch disease' type of problem is not usually anticipated.
- 14. See Romer (1989) for an analysis of such funds and their macroeconomic implications for countries, which depend on commodity aid and associated counterpart funds. (This aspect of such funds will not be discussed in this section.)
- 15. For details of the derivation of the resource transfer (Z) from a consistent macroeconomic framework, see FitzGerald (1993).
- 16. This equation can show the impact of deficit on the level of inflation if the deficit is financed, say, by an increase in the money supply.
- 17. As described in Chapter 1, a historical justification for such a classification can be found in Nzula et al. (1979) and Amin (1972).
- 18. This is excluded because the correlation coefficient between T and interest payments on concessional loans stands at 0.66.
- 19. This term was first coined in print in *The Economist* of 26 November 1977 (see Corden 1984). For the discussion of its effects in The Netherlands, see Ellman (1981).
- 20. Which, in the case of The Netherlands, is the natural gas sector.
- 21. This can be seen from the following accounting identity given by van Wijnbergen (1986b). Y is income, A absorption, n non-tradables and t tradables.

(ex ante) saving gap = Y - A

$$= Y_n + Y_t - A_n - A_t$$

= $(Y_n - A_n) + Y_t - A_t$
= $(Y_n - A_n) + (\text{ex ante})$ trade gap

22. Fardmanesh (1991a, 1991b) has undertaken a similar analysis for oil exporters. He notes that contrary to the core 'Dutch disease' model, the manufacturing sector of oil-exporting developing countries has in fact expanded following the oil boom. Indeed, a similar pattern is observed for Egypt and Nigeria. Neary and Wijnbergen (1986) have accommodated this phenomenon within the core model by relaxing the assumption of free trade. They stressed that in many countries, these sectors are protected (i.e., semi-traded). Benjamin et al. (1989) have explained

this through the imperfect substitution assumption, while Fardmanesh (1991a, 1991b) prefer to attribute this to a rise in the world price of manufacturing that was witnessed in the wake of the two oil price hikes. However, he does not commit himself to a possible mark-up pricing scheme, which his analysis might otherwise suggest. Rather, he bases his analysis on the assumption that a rise in manufacturing price, possibly due to a mark-up pricing scheme, leads to a rise in output. Nonetheless, following the analysis presented in Chapter 3, the main reason for expansion could also relate to the new foreign exchange earnings, which relieve the import compression problem, which previously had represented a constraint in the manufacturing sector.

- 23. Love (1994) has undertaken an excellent political economy analysis of the spending effect from the booming sector in Botswana. He notes how government expenditure unduly favors livestock development in an effort to facilitate the transition by a traditional dominant group of cattle owners to a contemporary capitalist class of commercial farmers. He observes how the interests of this group is maintained and legitimized by the new orthodoxy in economic thinking.
- 24. A number of PhD theses and other studies aiming to describe the 'Dutch disease' effect of oil in Nigeria have also been undertaken, mainly in North America.
- 25. Such a distinction is important because policy interventions at different stages of the mechanism can be effective when the propagating mechanisms are realistically captured.
- 26. This section will not concern itself with examining the final impact of aid or, indeed, the boom itself on sectoral performance. A simple model incorporating such an analysis is presented in Fardmanesh (1991a). Within the latter model the manufactured, agricultural and the non-traded sectors' value added are regressed on relative price and revenue from the booming sector. The negative coefficient of the latter may be taken as indicating the existence of a 'Dutch disease' problem. However, this approach is not adopted in the present study, since establishing an appreciation in the real exchange rate is sufficient to prove the point. This is because it has already been established in Chapter 4 that commodity exports are positively related to the real exchange rate.
- 27. The theoretical analysis which precedes such an empirical specification is developed in Chapters 2 and 3 of Edwards (1989).

- 28. See Appendix 5.2 for an ECM estimation using van Wijnbergen's form of specification.
- 29. Different types of prices are used as a proxy for the price of traded and non-traded goods. Thus, in a study on Brazil, Krugman (1988) uses world export price and Brazilian wholesale prices, respectively. US wholesale price or any trading partners' prices may also be used as an indicator of foreign price. Van Wijnbergen (1986a) uses import prices and the consumer price index (CPI). Although CPI and GDP deflators may be used interchangeably as an indicator of domestic prices, they are found to be good proxies within some countries and very poor proxies in others (see Dornbusch & Helmers 1988). Due to the lack of a complete series for CPI, I have used a GDP deflator in this study instead.
- 30. Indeed, experiments using the sum of bilateral and multilateral flows and grants lead to no major difference in results.
- 31. In particular, experiments are made based on real levels of bilateral, multilateral and grant flows.
- 32. More specifically, the price index for the region rose sharply, during the period from 1972 to 1979. Then, following a dramatic fall between 1979 and 1987, it rose sharply again in 1990. (For more details, see Chapter 4.)
- 33. It is worth noting that while food and agricultural commodities have a short gestation period, beverage and minerals have a longer one.

- 1. The theoretical model developed by Rattsø (1992b), along the lines of Taylor (1981) and the non-formal analysis of Kiss (1984) may be the only exceptions to this generalization.
- 2. Unless otherwise stated, the discussion in section 4.2.2 is based primarily on Dutt (1990, particularly Chapters 2 and 8) and, to a lesser extent, on Ocampo (1986). In the discussion which follows, in order to avoid unnecessary repetition, frequent citation of these studies will not be made.
- 3. Solving equation 15 for g^n , inserting the result into equation 2 solving it again for g^n , and multiplying the result by (P_n/P_n) (with interpretation of the result) will, in a fairly straightforward fashion, yield equation 16. Similar steps may be taken in order to arrive at equation 17.

- 4. In the latest version of this model, the South is taken as comprising Latin America, the newly industrialized countries and Africa. However, the author was not able to obtain access to this (unpublished) African model.
- 5. This section of the model is rightly specified along Kaldorian line (see Kaldor 1976: 705).
- 6. These may be perceived either as passive finance, within a Kalecki-Lewis model, or assumed Northern ownership in the South, within a Solow-Lewis model.
- 7. For example, through the effects of monetary policy on interest rates, as well as terms of trade between North and South.
- 8. Murshed specified his North-South model at both comparative static and dynamic levels. It seems that the two versions provide different conclusions about the impact of the North's macro policy. Thus, while the former underscores its negative impact on the South's debt, the latter does not. Murshed did not comment on this apparently contradictory result (see Murshed 1990: 87).
- 9. This is included in order to approximate convergence of consumers' test or preference, especially between developed countries and the newly industrialized countries.
- 10. Ideally, the Maxi-LINK should be used since this is relatively the stronger linkage mechanism.
- 11. This model, which was discussed in detail in Chapter 2, serves as a useful illustration of how market share models are formally modelled.
- 12. These are: price and volume of goods traded, exchange rate and interest rate.
- 13. However, Molana and Vines (1989) assumed production technology of a fixed-coefficient type within the North, since labour is assumed to be in surplus. For the South, their Cobb-Douglas function has three arguments: land, labour and capital.
- 14. For each category of good and service, a weighted (by import volume demand) market growth variable is constructed. In order to ensure system consistency, the elasticities of this market growth variable are, in general, imposed in the region of unity.

- 15. See Alemayehu (1995) for a discussion of some theoretical implications of the incorporation of such models into a global model of Africa.
- 16. However, the Sudan is excluded from NA and included in the ESA region, since its economic structure is similar to the latter.
- 17. See Vos (1994, and Chapter 4 in particular) for an excellent historical analysis of the segmented nature of the global capital market.
- 18. This maximum/critical ratio not only limits borrowing but also indicates a high probability of default (Vos 1994: 216).
- 19. However, non-market restrictions on the flows of bank credit to Africa could be responsible for limiting the total level of inflows. Within the model, this is ensured by limiting the market-based flow to Africa to its historical maximum level.
- 20. The definition of capital flight (CF) used is:

 $CF = \Delta D + FDI - CA - \Delta RES - \Delta B$

where ΔD is flows of debt, FDI foreign direct investment, *CA* current account balance, ΔRES change in reserve, and ΔB the change in domestic banking claims on foreign banks.

21. Most global models use estimations from other studies. However, the recent IMF model (MULTIMOD) uses a single-equation (first-order) ECM-based estimation. Our approach is similar to the latter but improves on it in two respects. Firstly, it allows for simultaneous estimation to be undertaken in some of the blocs. And, secondly, it starts its estimation from a second-order ECM (following a General to Specific approach). Finally, as noted in Chapter 4, the use of an ECM approach helps to tackle the problem of simultaneous equation bias.

- 1. The global consistency of balance of payment statistics was discussed more than two decades ago by Mundell. He explored, in general terms, the point that the balance of payment position for all countries should add to zero. This is also referred as 'Cournot's law' (see Mundell 1968: 143–47).
- 2. See de Jong and Vos (1995a) for a recent analysis of trade and finance, using a WAM. An earlier analysis is also contained in FitzGerald and Luttik (1991).

- 3. Most notably, Heien et al. (1973) provide a simple and powerful analysis of the algorithm, as well as problems associated with its use.
- 4. Since this is a painstaking process, it is advisable to run the model by bloc or by equation, and subsequently to combine these blocs and then to run the model one final time. This procedure is extremely helpful not only in detecting errors but also for calibration purposes. Moreover, it is noteworthy that ECM estimation is extremely powerful in reproducing the base run values, with only a minor calibration.
- 5. Suppose a linear function is given in matrix form by AX = D where A is an nxn matrix of a_{ij} coefficients, X an nxn matrix of x_i variables and D an nx1 vector of constant values. If we take each of the rows of the left hand side variables as functions denoted by y_i , the Jacobian is given by,

$$|J| = \frac{\partial (y_1, y_2, \dots, y_n)}{\partial (x_1, x_2, \dots, x_n)} \text{ which in turn} = |A|$$

Thus, for a linear model the Jacobian is essentially the determinant of A and hence, as long as A is non-singular, the stability condition holds.

- 6. See Jamshidi (1989) for an excellent and simple methodological discussion of structural sensitivity analysis and an illustration using MULTIMOD.
- 7. Similar results relating to the implicit GDP deflator of the USA are reported in Zellner and Peck (1973).
- 8. This was launched at a joint IMF-World Bank press conference held in September 1996.
- 9. The value of this stood at 8.13, 9.27 and 8.35 per cent, for the years 1988, 1989 and 1990, respectively (IMF 1994).
- 10. The only exception to this being exports in WCA and price of beverage.
- 11. However, while Vos' conclusions are derived from a model that has a single aggregate category of South, the above findings relate only to Africa.
- 12. The so-called indirect impact on output and income is modelled, in this book through imports.
- 13. For example, MULTIMOD (see Jamshidi 1989: 38).

- 14. The relative position of each region to the South in general is not examined here.
- 15. The best export performance is observed in the NA region, and the worst in ESA.
- 16. Private inflows are modeled as being sensitive to this ratio, as shown in the debt cancellation simulation.
- 17. The difference between the results of Vos' model and the one presented here may be attributed to the use of econometrically estimated parameters and a higher degree of disaggregation in this model. Moreover, the model developed in this book is a dynamic one, both in terms of time and the inclusion of short-run and long-run effects together. In contrast, the dynamic element in Vos' model lies in the stock adjustment and inclusion of current responses to lagged variables.
- 18. Such a disaggregation would be particularly relevant in relation to WCA and NA regions.
- 19. See also section 7.4.1 for a discussion of this phenomenon.
- 20. For example, an increase in the export of commodities from the rest of the South is simulated.

- 1. The financial impacts relate to debt, debt service, financial inflows and their macroeconomic effect, while the trade-related impacts operate mainly through commodity prices.
- 2. This occurs by way of the 'adding-up' effect.
- 3. However, such a doubling of aid appears to have no significant effect on public revenue within the WCA region.
- 4. However, by the end of the simulation, this effect collapses to nearly zero in the WCA region.
- 5. A 0.3 per cent increase in Northern output, from its base run value, due to a 10 per cent increase in public investment in Africa.
- 6. Specifically a flexi/mark-up price market, imported capital-constrained investment in South, a Southern fiscal posture and domestic price which is sensitive to the external sector.
- 7. Examples of such an approach are the works of Samir Amin (1972, 1996) and Bade Onimode (1988).

References

(Note: Ethiopian names are cited using 'first' names, without comma.)

- Abebayehu Tegene (1990) 'The Supply and Demand for Exports in Some African Countries: A Simultaneous Equation Model', *Journal of Quantitative Economics*, 6(2): 395-410.
- Adam, Christopher (1995) 'Review Article: Adjustment in Africa: Reforms, Results and the Road Ahead', *World Economy*, 18 (5): 729-35.
- Adams, F. Gerard & J.R. Behrman (1976) Econometric Models of World Agricultural Commodity Markets: Cocoa, Coffee, Tea, Wool, Cotton, Sugar, Wheat, Rice. USA: Ballinger Publishing Co.
- Adams, F. Gerard & J.R. Behrman (eds) (1978). Econometric Models of World Commodity Policy. Toronto: Heath & Company.
- Adedeji, Adebayo (1993) 'Introduction: Governance, Aid and Africa', in Douglas Rimmer (ed.) Action in Africa: The Experience of People Involved in Government, Business and Aid. London: James Currey and The Royal African Society.
- Ady, P. (1968) 'Supply Functions in Tropical Agriculture', Bulletin of the Oxford University Institute of Economics and Statistics.
- Agarwal, J.P. (1980) 'Determinants of Foreign Direct Investment: A Survey', Weltwirtschaftliches Archiv-Review of World Economics, 116(4): 739-73.
- Akiyama, Takamas & D.F. Larson (1994) 'The Adding-Up Problem: Strategies for Primary Commodity Exports in Sub-Saharan Africa', World Bank Policy Research Working Paper 1245, Washington DC: The World Bank.
- Alchian, Armen A. (1971) 'The Meaning of Utility Measurement', in William Breit & Harold M. Hochman (eds) *Readings in Microeconomics*, 2nd edn, Hinsdale: Druden Press (also in *American Economic Review*, 1953 (March): 26-50).
- Alemayehu Geda (1995) 'Africa in North-South Economic Model: A Theoretical Analysis of The External Sector' (memo).

- Alemayehu Geda (1996), 'Determinants of Aggregate Primary Commodity Export Supply: Econometric Results form African Countries' ISS Working Paper Sub-Series on Money, Finance and Development, No. 66, Institute of Social Studies, The Hague.
- Alemayehu Geda, E.V.K. FitzGerald, K. Saramad & R. Vos (1992)
 'Trends of Public and Private Capital Account Variables in Developing Countries', Institute of Social Studies, The Hague (mimeo).
- Alemayehu Geda (1992) 'Macroeconomic Interaction Between North and South: A World Accounting Matrix (WAM) based Global Model Approach with Special Emphasis on Africa', Ph.D. Research Seminar, Erasmus University, Rotterdam.
- Alemayheu Geda (1994) 'A Comparative Look at International Linkages of the World Banks Global Economic Model (Bank-GEM) and Other North-South Models', *ISS Working Paper, Sub-Series on Money, Finance and Development, Number 58,* Institute of Social Studies, The Hague.
- Ali, Liaqat (1984) 'Africa's Commodity Trade and Structure of Marketing and Distribution of Primary Commodities', *Foreign Trade Review*, 19 (3): 367-89.
- Alibaruho, G. (1974) 'Regional Supply Elasticities in Uganda's Cotton Industry and the Declining Level of Cotton Output', *Eastern African Economic Review*, 6: 35-56.
- Allen, Chris & David Vines (1992) 'Some Key Global Adjustment Scenarios and Their Effects on Major Developing Country Regions', A Report to the United Nations, No. F2S/038.
- Allen, Chris & David Vines (1994) 'Should Clinton Cut the Deficit or is there a Global Paradox of Thrift' in David Vines and David Currie (eds) *Macroeconomic Interaction Between North and South*. Cambridge: Cambridge University Press.
- Allen, Polly R. & Peter B. Kenen (1980) Asset Markets, Exchange Rates, and Economic Integration: A Synthesis. Cambridge: Cambridge University Press.
- Alogoskoufis, George & Panos Varangis (1992) 'OECD Fiscal Policies and the Relative Prices of Primary Commodities', *Policy Research Working Papers in International Trade, WPS 955.* Washington DC: The World Bank.
- Amin, Samir (1972) 'Underdevelopment and Dependence in Black Africa: historical origin', *Journal of Peace Research*, 9: 105-19.
- Amin, Samir (1974a) 'Accumulation and Development: A Theoretical Model', *Review of African Political Economy*, 1: 9-26.

- Amin, Samir (1974b) Accumulation on a World Scale: A Critique of the Theory of Underdevelopment, Vol. II. London: Monthly Review Press.
- Amin, Samir (1996) 'On the Origin of the Economic Catastrophe in Africa' in Sing C. Chew & Robert A. Denmark (eds) The Underdevelopment of Development. London: Sage Publication.
- Amjadi, Azita, Ulrich Reinke & Alexander Yeats (1996) 'Did External Barriers Cause the Marginalization of Sub-Saharan African in World Trade', Policy Research Working Paper No. 1586, Washington DC: The World Bank.
- Anderson, T.W. & Cheng Hsiao (1982) 'Formulation and Estimation of Dynamic Models Using Panel Data', *Journal of Econometrics*, 18: 47-82.
- Anyadike-Danes, M.K & M.N. Anyadike-Danes (1992) 'The Geographic Allocation of the European Development Fund Under the Lome Convention', World Development, 20 (11): 1647-61.
- Arab Republic of Egypt Ministry of Planning (1977) The Five Year Plan 1978-1982: The General Strategy for Economic and Social Development, Vol. 1.
- Arab Republic of Egypt Ministry of Planning (1982) The Detailed Framework of the Five Year Plan for Economic and Social Development 1982/83-1986/87, Part I Principal Components.
- Ardeni, Giorgio & Brain Wright (1992) 'The Prebisch-Singer Hypothesis: A Reappraisal Independent of Stationarity Hypotheses', *The Economic Journal*, 102 (413): 803-12.
- Armington, P.A. (1969) 'A Theory of Demand for Products Distinguished by Place of Production', *IMF Staff Papers*, 16: 159-78.
- Asante, S.K.B. (1991) African Development: Adebayo Adedeji's Alternative Strategies. London: Hans Zell Publishers.
- Askari, Hosein & J.T. Cummings (1976) Agricultural Supply Response: A Survey of the Econometrics Evidence. New York: Praeger Publisher.
- Asmerom Kidane & Gerhard Kocklaeuner (1985) 'A Macroeconometric Model for Ethiopia: Specification, Estimation, Forecast and Control', *Eastern African Economic Review*.
- Austen, Ralph A. (1987) African Economic History: Internal Development and External Dependency. London: James Currey.
- Bacha, E.L. & C.F. Diaz Alejandro (1982) International Financial Intermediation: A Long and Tropical View. New Jersey, International Finance Section, Department of Economics, Princeton University.
- Bacha, Edmar (1978) 'An Interpretation of Unequal Exchange From Prebisch-Singer to Emmanuel', *Journal of Development Economics*, 5: 319-30.

- Bacha, Edmar (1984) 'Growth with Limited Supplies of Foreign Exchange:
 a reappraisal of the two-gap model', in Moshe Syrquin, Lance D. Taylor
 & Larry E. Westphal (eds) Economic Structure and Performance: Essays
 in Honour of Hollis B. Chenery. Orlando: Academic Press.
- Bacha, Edmar (1990) 'A Three-Gap Model of Foreign Transfers and the GDP growth rate in developing countries', *Journal of Development Economics*, 32: 279-96.
- Balassa, Bela (1967) Trade Liberalization among Industrial Countries. New York: McGraw-Hill.
- Balassa, Bela (1983) 'Policy Response to External Shocks in Sub-Saharan African Countries', *Journal of Policy Modeling*, 5 (1): 75-105.
- Balassa, Bela (1984) 'External Shocks and Policy Response in Sub-Saharan Africa, 1973-78', *Finance and Development*, 21 (1): 10-12.
- Ball, R.J. (ed.) (1973) *The International Linkage of National Economic Models*. Amsterdam: North Holland Pub Co.
- Banerjee, Anindya, J. Dolado, J.W. Galbraith & D.F. Hendry (1993). Co-Integration, Error-correction, and the Econometric Analysis of Non-Stationary Data. Oxford: Oxford University Press.
- Bank of Ghana (various years: 1970-1986) Report of the Board.
- Bank of Sierra Leone (various years 1970-1990) Annual Report and Statement of Accounts.
- Bank of Sierra Leone. Balance of Payments: 1970-74. Freetown.
- Bank of Tanzania (various years 1970-1990) Economic and Operation Reports, Dare Salaam.
- Bank of Zambia (various issues: 1986-1988) Quarterly Financial and Statistical Review. Lusaka.
- Bank of Zambia (various years: 1970-1986) Report and Statement of Accounts. Lusaka.
- Baran, Paul (1957) The Political Economy of Growth. London: Monthly Review Press.
- Baran, Paul & Paul M. Sweezy (1966) Monopoly Capital: An Essay on the American Economic and Social Order. England: Penguin Books Ltd.
- Bardhan, Pranab (1967) 'Optimum Foreign Borrowing', in Karl Shell (ed.) *Essays on the Theory of Optimal Economic Growth*. Massachusetts: MIT Press.
- Bateman, M.J. (1965) 'Aggregate and Regional Supply Functions for Ghanaian Cocoa', *Journal of Farm Economics*, 47 (May): 384-401.
- Beenstock, Michael (1988) 'An Econometric Investigation of North-South Interdependence', in David Currie & David Vines (eds) *Macroeconomic Interactions between North and South*. Cambridge: Cambridge University Press.

- Beenstock, Michael (1980) 'Political Econometry of Official Development Assistance', World Development, 8 (2): 137-144.
- Berhrman (1978) 'On Commodity Modelling' in Adams, F. Gerard & J.R. Behrman (eds) *Econometric Models of World Commodity Policy*. Toronto: Heath and Company.
- Benjamin, Nancy (1990) 'Investment, the Real Exchange Rate, and Dutch Disease: A Two-Period General Equilibrium Model of Cameroon', *Journal of Policy Modelling*, 12 (1): 77-92.
- Benjamin, Nancy, S. Devara, R. Weiner (1989) 'The Dutch Disease in Developing Country: Oil Revenue in Cameroon', Journal of Development Economics, 30: 71-92.
- Berhanu Nega (1994) 'A Peripheral-Socialist Model of the Ethiopian Economy' in Berhanu Abegaz (ed.) *Essay on Ethiopian Economic Development*. Aldershot: Avebury.
- Bernstein, Henry, T. Hewitt & A. Thomas (1992) 'Capitalism and the Expansion of Europe' in T. Allen & A. Thomas (eds) Poverty and Development in The 1990s. Oxford: Oxford University Press and The Open University.
- Bettelheim, C. (1972) 'Theoretical Comments' in A. Emmanuel. Unequal Exchange: A Study of the Imperialism of Trade. New York: Monthly Review Press.
- Bevan, D.L., A. Bigsten, P. Collier & J.W. Gunning (1987) 'Peasant Supply Response in Rationed Economies', World Development, 15(8): 431-39.
- Bhaduri, Amit & Rune Skarstein (1996) 'Short-Period Macroeconomic Aspects of Foreign Aid', *Cambridge Journal of Economics*, 20(2): 195-206.
- Binh, Tran-Nam Binh & Mark McGillivray (1993) 'Foreign aid, taxes and public investment: a comment', *Journal of Development Economics*, 41: 173-76.
- Binswanger, Hans P. (1992) 'Determinants of Agricultural Supply and Adjustment Policies', *Agriculture Supply* (September).
- Blaug, Mark (1992) The Methodology of Economics or How Economists Explain, 2nd edn, Cambridge: Cambridge University Press.
- Bloch, Harry & David Sapsford (1996) 'Trends In The International Terms of Trade Between Primary Producers and Manufactures', *Journal of International Development*, 8(1).
- Bond, Marian E. (1983), 'Agricultural Responses to Prices in Sub-Saharan African Countries', *IMF Staff Papers*, 30(4): 703-26.

- Bond, Marian E. (1987) 'An econometric Study of Primary Commodity Exports from Developing Country Regions to the World', *IMF Staff Papers*, 34(2): 77-92.
- Boutros-Ghali, Youssef & Lance D. Taylor (1980) 'Basic Needs Macroeconomics: Is it manageable in the case of Egypt', *Journal of Policy Modelling*, 2(3): 409-36.
- Bowles (1987) 'Foreign Aid and Domestic Savings in Less Developed Countries: Some Tests for Causality', *World Development*, 15(6): 789-96.
- Brecher, R.A. & C.F. Diaz Alejandro (1977) 'Tariffs, Foreign Capital and Immiserizing Growth', *Journal of International Economics*, 7: 317-22.
- Brooks, George E. (1993) Landlords and Strangers: Ecology, Society and Trade in Western Africa: 1000-1630. Oxford: Westview Press.
- Brown, Richard P.C. (1992) Public Debt and Private Wealth: Debt, Capital Flight and the IMF in Sudan. London: Macmillan.
- Bruno, M. & J.D. Sachs (1985) *Economics of World-Wide Stagflation*. Oxford: Basil Blackwell.
- Bryant, R.C., D. Currie, J. Frenkel, P. Masson & R. Portes (eds) (1989) *Macroeconomic Policy in an Interdependent World*, Washington DC: IMF and The Brooking Institution.
- Buckley, P.J. & M. Casson (1976) The Future of Multinational Enterprise. London: Macmillan.
- Bulow, Jeremy & Kenneth Rogoff (1989) 'A Constant Reconstructing Model of Sovereign Debt', *Journal of Political Economy*, 97(1): 155-78.
- Burdette, Marcia (1990) 'Industrial Development in Zambia, Zimbabwe and Malawi: The Primacy of Politics', in Z.A. Koncazcki, J.L. Parpart & T.M. Shaw (eds) Studies in the Economic History of Southern Africa Vol. 1, The Front-line States. London: Frank Cass Ltd.
- Cagan, P.H. (1956) 'The Monetary Dynamics of Hyperinflation', in M. Freedman (ed.) *Studies in the Quarterly Theory of Money*. Chicago: University of Chicago Press.
- Carlisle, Donald (1954) 'The Economics of a Fund Resource with Particular Reference to Mining', *The American Economic Review*, 44(4): 595-616.
- Central Bank of Egypt (various years: 1980/81, 1982/83, 1983/84, 1984/ 85, 1985/86, 1986/87, 1988/89, 1990/91). Annual Reports, Cairo.
- Central Bank of Kenya (various years: 1989-1991) Economic Report, Nairobi.
- Central Bank of Libya, Economic Research Division (1974) Economic Bulletin, Vol. XIV, No. 7-12.

- Central Bank of Libya (various years: 1976-1984) Annual Report of the Board of Directors.
- Central Bank of Nigeria (various years: 1970-1992) Annual Reports and Statement of Accounts.
- Central Bureau of Statistic, Ministry of Planning and National Development (various years: 1979-1988 and 1992) *Economic Survey*, Nairobi.
- Chenery, H. (1952) 'Overcapacity and the Acceleration Principle', Econometrica, 20(1): 1-28.
- Chenery, H. & A. Strout (1966) 'Foreign Assistant and Economic Development', *The American Economic Review*, 56(4): 679-733.
- Cheng, Leonard (1984) 'International Trade and Technology: A Brief Survey of the Recent Literature', *Weltwirtschaftliches Archiv*, 120: 165-89.
- Chowdhury, Aris & Iyanatul Islam (1993) The Newly Industrializing Economies of East Asia. London: Routledge.
- Chu, Ke-Young & T.K. Morrison (1984) 'The 1981-82 Recession and Non-Oil Primary Commodity Prices', *IMF Staff Papers*, 31 (March): 93-140.
- Chu, Ke-Young & T.K. Morrison (1986) 'World Non-Oil Primary Commodity Markets: A Medium-Term Framework of Analysis', *IMF Staff Papers*, 33(1): 139-84.
- Clark, J. Maurice (1917) 'Business Acceleration and the Law of Demand: A Technical Factor in Economic Cycles', Journal of Political Economy, 25(1): 217-35 (reprinted in American Economic Association, 1944, Reading in Business Cycle Theory. Philadelphia: Blakiston Co. pp. 234-54).
- Cline, William R. (1983) 'International Debt and the Stability of the World Economy', *Policy Analyses in International Economics No. 4*. Washington DC: Institute for International Economics.
- Cline, William R. (1995) International Debt Re-examined. Washington DC: Institute for International Economics.
- Coase, Ronald H. (1937) 'The Nature of the Firm', *Economica*, 4 (November): 386-405.
- Cohen, Daniel (1987) 'External and Domestic Debt Constraints of LDCs: A theory with Numerical Application to Brazil and Mexico' in R.C. Bryant & R. Portes (eds) Global Macroeconomics: Policy Conflict and Co-operation. London: Macmillan.
- Cohen, Daniel (1994) 'Growth and External Debt' in Frederick van der Ploeg (ed.) *The Handbook of International Macroeconomics*. Oxford: Blackwell.
- Collier, P. (1991) 'Africa's External Economic Relations: 1960-90', African Affairs, 90: 339-56.

- Collier, P. & W. Gunning (1999) 'Explaining African Economic Performance', Journal of Economic Literature, 37(1): 64-111.
- Collins, Susan (1992) 'Aid, Allocation and Availability: Capital Flows to Developing Countries?' Paper presented to the World Bank Annual Conference on Development Economics, Washington DC.
- Cooper, Richard N. & J.D. Sachs (1985) 'Borrowing Abroad: The Debtor's Perspective' in Smith & Cuddington (eds) *International Debt and The Developing Countries*. Washington DC: The World Bank.
- Corden, Max (1989) 'Debt Relief and Adjustment Incentives' in Jacob A. Frenkel, Michael P. Dooley & Peter Wickham (eds) *Analytical Issues in Debt*. Washington DC: International Monetary Fund.
- Corden, Max (1984) 'Booming Sector and Dutch Disease Economics: Survey and Consolidation', Oxford Economic Papers, 36: 359-80.
- Cornia, Giovanni Andrea (1992) An Alternative Development Paradigm of Africa's Recovery in the 1990s' Economic Research Seminar, Institute of Social Studies, The Hague.
- Cuddington, John (1992) 'Long-run trends in 26 primary commodity prices: A disaggregated look at the Prebisch-Singer hypothesis', *Journal of Development Economics*, 39(2): 207-227.
- Cuddington, John & Carlos Urzua (1989) 'Trends and Cycles in the Net Barter Terms of Trade: A New Approach', *Economic Journal*, 99(396): 426-42.
- Currie, David & David Vines (eds) (1988) Macroeconomic Interactions between North and South. Cambridge: Cambridge University Press.
- Darity, W. & B. Horn (1988) The Loan Pushers. The Role of Commercial Banks in the International Debt Crisis. Cambridge (Mass.): Ballinger.
- Darity, William & E.V.K. FitzGerald (1982) 'A Kalecki-Keynes Model of World trade, Finance, and Development', *ISS Working Paper Sub-Series* on Money, Finance and Development No. 7, ISS, The Hague.
- Davies, Rob, Jorn Rattsø & Rangar Torvik (1994) 'The Macroeconomics of Zimbabwe in the 1980s: a CGE-Model Analysis', *Journal of African Economies*, 3(2): 153-98.
- Decaluwe, Bernard & Fabien Nsengiyumva (1994) 'Policy Impact under Credit Rationing: a Real and Financial CGE Model of Rwanda', *Jour*nal of African Economies, 3(2): 262-308.
- Dercon, Stefan (1993) 'Peasant Supply Response and Macroeconomic Policies: Cotton in Tanzania', Journal of African Economies, 2(2): 157-94.
- Dick, Herman, S. Gupta, T. Mayer & D. Vincent (1983) 'The Short-run Impact of Fluctuating Primary Commodity Prices on Three Developing Economies: Colombia, Ivory Coast and Kenya', World Development, 11 (5): 405-16.

- Dickson, K.B. (1977) 'Trade Patterns in Ghana at the Beginning of the Eighteenth Century', in Z.A. Koncazcki & J.M. Knoxzacki, An Economic History of Tropical Africa: Vol. I, The Pre-Colonial Period. London: Frank Cass.
- Dixit, A. & V. Norman (1980) *Theory of International Trade*. Cambridge: Cambridge University Press.
- Dixit, Avinash K. & Joseph E. Stiglitz (1977) 'Monopolistic Competition and Optimum Product Diversity', *The American Economic Review*, 67 (3): 297-308.
- Don, F.G.H. & G. Gallo (1987) 'Solving Large Scale Systems of Equation', Journal of Forecasting, 6(3): 167-80.
- Dornbusch, Rudiger (1980) Open Economy Macroeconomics. New York: Basic Books.
- Dornbusch, Rudiger, F. Leslie & C. Helmers (eds) *The Open Economy: Tools for Policy Makers in Developing Countries.* Oxford: Oxford University Press.
- Dudley, Leonard & C. Montmarquette (1976) 'A Model of the Supply of Bilateral Foreign Aid', *The American Economic Review*, 66(1): 132-42.
- Dunning, John H. (1993) Multinational Enterprises and the Global Economy. England: Addison-Wesley Publishing Company.
- Dutt, Amitava Krishna (1990) Growth, Distribution and Uneven Development. Cambridge: Cambridge University Press.
- Easterly, William & Ross Levine (1996) 'Africa's Growth Tragedy: A Retrospective 1960-89' Washington D.C.: The World Bank Policy Research Department (manuscript, September).
- Eaton, Jonathan & Mark Gersovitz (1980) 'LDC Participation in International Financial Markets: Debts and Reserves', *Journal of Development Economics*, 7 (March): 3-21.
- Eaton, Jonathan & Lance D. Taylor (1986) 'Developing Country Finance and Debt', *Journal of Development Economics*, 22: 209-65.
- Eaton, Jonathan & Mark Gersovitz (1981) 'Debt with Potential Repudiation: Theoretical and Empirical Analysis', *The Review of Economic Studies*, 48(2): 289-309.
- Eaton, Jonathan, Mark Gersovitz & Joseph E. Stiglitz (1986) 'The Pure Theory of Country Risk', *European Economic Review*, 30(3): 481-513.
- Economic Commission for Africa (ECA) (1989a) African Alternative Framework to Structural Adjustment Programs for Socioeconomic Recovery and Transformation (AAF-SAP). Addis Ababa: ECA.
- ECA (1989b) Statistics and Policies: ECA Preliminary Observation on the World Bank Report: Africa's Adjustment and Growth in the 1980s. Addis Ababa: ECA.

- Edwards, Sebastian (1989) Real Exchange Rates, Devaluation, and Adjustment: Exchange Rate Policy in Developing Countries. Cambridge: MIT Press.
- Edwards, Sebastian & S. van Wijnbergen (1989) 'Disequilibrum and Structural Adjustment', in H. Chenery & T.N. Srinivasan (eds) Handbook of Development Economics, Vol. II. Amsterdam: North Holland.
- Eichengreen, Barry & Richard Portes (1986) 'Debt and Default in the 1930s: Causes and Consequence', *European Economic Review*, 30(3): 599-640.
- Elbadawi, I.A., D. Ghura & G. Uwujaren (1992) 'World Bank Adjustment Lending and Economic Performance in Sub-Saharan Africa in the 1980s: A comparison with other low-income countries', *Policy Re*search Working Papers, WPS 1000. Washington DC: The World Bank.
- Ellman, M. (1981) 'Natural Gas, Restructuring and Re-industrialization: The Dutch Experience of Industrial Policy', in T. Barker & V. Brailovsky (eds) *Oil or Industry?* London: Academic Press.
- Emmanuel, A. (1972) Unequal Exchange: A Study of the Imperialism of Trade. New York: Monthly Review Press.
- Enders, Walter (1995) Applied Econometric Time Series. New York: John Wiley & Sons, Inc.
- Engel, R.F. & C.W.J. Granger (1987) 'Co-integration and Error-correction: Representation, Estimation and Testing', *Econometrica*, 55: 251-76.
- Engel, R.F. & C.W.J. Granger (eds) (1991) Long-run Economic Relationships: Readings in Co-integration. Oxford: Oxford University Press.
- Eriksson, Gun (1993) 'Peasant Responses to Price Incentives in Tanzania: A Theoretical and Empirical Investigation, Research Report No. 91, The Scandinavian Institute of African Studies, Uppsala.
- Ethier, Wilfred J. (1982) 'National and International Returns to Scale in the Modern Theory of International Trade', *American Economic Review*, 72(3): 389-405.
- Evans, David (1993) 'Trade Policy Reform and Tropical Beverage: The Fallacy of Composition in Policy Advice', *IDS Discussion Paper No.* 326, University of Sussex, Brighton.
- Ezekiel, M. (1938) 'The Cobweb Theorem', Quarterly Journal of Economics, 52: 255-80.
- Ezenwe, Uka (1991) 'The African Debt Crisis and the Challenge of Development', *Intereconomics*, 28 (January/February): 35-43.
- Falvey, Rodney (1981) 'Commercial Policy and Intra-Industry Trade', Journal of International Economics, 11(4): 495-511.
- Fantu Cheru (1992) 'Structural Adjustment, Primary Resource Trade and Sustainable Development in Sub-Saharan Africa', World Development, 20(4): 497-512.

- Fardmanesh, Mohsen (1991a) 'Dutch Disease Economics and the Oil Syndrome: An Empirical Study', *World Development*, 19(1): 711-17.
- Fardmanesh, Mohsen (1991b) 'Terms of Trade Shocks and Structural Adjustment in Small Open Economy: Dutch Disease and Oil Price Increase' Journal of Development Economics, 34(1/2): 339-53.
- Feder, Gershon & L.V. Uy (1985) 'The Determinants of International Creditworthiness and their Policy Implications', *Journal of Policy Modelling*, 7(1): 133-57.
- Federal Republic of Nigeria, Federal Ministry of Economic Development, Central Planning Office (1975) The Third National Development Plan: 1975-80, Vol. I, Lagos.
- Fernandez-Arias, Eduardo & P.J. Montiel (1996) 'The Surge in Capital Flows to Developing Countries: An Analytical Overview' *The World Bank Economic Review*, 10(1): 51-77.
- Feyerabend, Paul (1975) Against Method: Outline of An Anarchist Theory of Knowledge. London: Routledge.
- Fieldhouse, D.K. (1986) Black Africa 1945-1980: Economic Decolonization and Arrested Development. London: Allen and Unwin.
- Findlay, Ronald (1973) International Trade and Development Theory. New York: Colombian University Press.
- Findlay, Ronald (1980) 'The Terms of Trade and Equilibrium Growth in the World Economy', *The American Economic Review*, 70(3): 291-99.
- Findlay, Ronald (1981) 'The Fundamental Determinants of the Terms of Trade', in Sven Grassman & Erik Lundberg (eds) *The World Economic Order: Past and Prospects*. London: Macmillan.
- FitzGerald, E.V.K. (1993) The Macroeconomics of Development Finance: A Kaleckian Analysis of the Semi-industrialized Economy. London: St. Martin's Press.
- FitzGerald, E.V.K. & J. Luttik (1991) 'A Note on the Measurement of Trade and Financial Linkages in the WAM', *ISS Working Paper-Sub-Series on Money, Finance and Development, No. 40*, The Hague: Institute of Social Studies.
- FitzGerald, E.V.K., K. Jansen & R. Vos (1992) 'External Constraints on Private Investment Decisions in Developing Countries', *ISS Working Paper, Sub-Series on Money, Finance and Development, No. 43*, The Hague, The Netherlands.
- FitzGerald, E.V.K.& R.Vos (1989) Financing Economic Development: A Structural Approach to Monetary Policy. Aldershot: Gower.
- Fleming, J.M. (1962) 'Domestic Financial Policies Under Fixed and Floating Exchange Rates', *IMF Staff Papers*, 9(3): 369-79.

- Folkerts-Landau, David (1985) 'The Changing Role of International Bank Lending in Development Finance', *IMF Staff Papers*, 32(2): 317-63.
- Fosu, Augstin Kwasi (1991) 'Capital Instability and Economic Growth in Sub-Saharan Africa', Journal of Development Studies, 28(1): 74-85.
- Frankel, Herber (1977) 'Some Aspects of Investment and Economic Development in the Continent of Africa', in Z.A. Koncazcki & J.M. Knoxzacki, An Economic History of Tropical Africa: Vol. II The Colonial Period. London: Frank Cass.
- Frenkel, J. & A. Razin (1987) 'The Mundell-Fleming Model a Quarter of a Century Later: A Unified Exposition', *IMF Staff Papers*, 34 (4): 567-620.
- Frey, Burno (1984) International Political Economy. Cambridge: Basil Blackwell.
- French, B.C. & J.L. Matthews (1971) 'A Supply Response Model for Perennial Crops', American Journal of Agricultural Economics, 53 (August): 478-90.
- Friedman, Milton (1953) 'The Methodology of Positive Economics', in Daniel M. Hausman (ed.) (1990) The Philosophy of Economics: an Anthology. Cambridge: Cambridge University Press.
- Fritsch, Winston (1988) 'Brazil's Growth prospects: Domestic Saving, External Finance and OECD Performance Interactions', in David Currie & David Vines (eds) Macroeconomic Interactions between North and South. Cambridge: Cambridge University Press.
- Galil, Eisa Abdalla Abdel (1994) 'Towards an Understanding of Sudan's Balance of Payments Difficulties', unpublished MA thesis, Institute of Social Studies, The Hague.
- Gang, Ira N. & Haidar Khan (1991) 'Foreign Aid, Taxes and Public Investment', Journal of Development Economics, 34: 355-69.
- Gasper, Des (1992) 'Development Ethics An Emergent Field? A look at scope and structure with special reference to the ethics of aid', *ISS Working Paper Series No. 134*. The Hague: Institute of Social Studies.
- Gebrehiwot Baykedagne (1917) Mengestena Yehizb Astedader (in Amharic), English translation in Tenker Bonger (1995) The State and Economy in Early 20th Century Ethiopia: Prefiguring Political Economy. London: Karank House.
- Gebru Mersha (1995) 'State Farms in State-Centered Accumulation Strategies in Socialist Africa: The Rationale and Critique', Ph.D. dissertation, Institution of Social Studies, The Hague.
- Ghoshal, A. (1974) 'The Supply Response of Liberian Rubber Farmers', Malayan Economic Review 19 (October): 105-125.

- Ghura, Dhaneshwar (1995) 'Macro Policies, External Forces, and Economic Growth in Sub-Saharan Africa', *Economic Development and Cultural Change*, 43(4): 759-77.
- Goldstein, Morris & M.S. Khan (1978) 'The Supply and Demand for Exports: A Simultaneous Approach', *The Review of Economics and Statistics*, 60(1): 275-86.
- Gray, L. (1914) 'Rent Under the Assumption of Exhaustibility', *Quarterly Journal of Economics*, May: 464-79.
- Greenway, David & Chris Milner (1986) The Economics of Intra-industry Trade. New York: Basil Blackwell.
- Grier, Kevin B. & G. Tullock (1989) 'An Empirical Analysis of Cross-National Economic Growth: 1951-80, Journal of Monetary Economics, 24: 259-76.
- Griffin, Keith (1970) 'Foreign Capital, Domestic Savings and Economic Development', Bulletin of the Oxford University Institute of Economics and Statistics 32: 99-112.
- Griffin, Keith & J. Enos (1970), 'Foreign Assistance, Objectives and Consequences', *Economic Development and Cultural Change*, 18(3): 313-37.
- Grilli, Enzo R. & Maw Cheng Yang (1988) 'Primary Commodity Prices, Manufactured Goods Prices, and the Terms of Trade of Developing Countries: What the Long Run Shows', *The World Bank Economic Review*, 2(1): 1-48.
- Grubel, Herbert (1970) 'The Theory of Intra-Industry Trade', in I.A. Mc-Dogall & R.H. Snape (eds) *Studies in International Economics*. Amsterdam: North-Holland Publishing.
- Gupta, Kanhaya (1975) 'Foreign Capital Inflows, Dependency Burden, and Saving Rate in Developing Countries: A Simultaneous Equation Model, *Kyklos*, 28(2): 358-74.
- Gurney, Andrew (1989) 'Obtaining Estimates for the Standard Errors of Long-run Parameters' (Macro Modelling Notes), *National Institute Economic Review*, 128 (May): 89-90.
- Guvenen, O., W.C. Labys & J-B Lesourd (eds) (1991) International Commodity Market Models: Advances in Methodology and Applications. London: Chapman and Hall.
- Gwyer, G.D. (1971) 'Long and Short-Run Elasticities of Sisal Supply', Eastern Africa Economic Review, 3: 19-29.
- Haggard, Stephan (1990) Pathways from the Periphery: The Politics of Growth in the Newly Industrialized Countries. Ithaca: Cornell University Press.

- Hallwood, C. Paul & Ronald MacDonald (1994) International Money and *Finance* (2nd edn). Oxford: Blackwell Publishers.
- Hamilton, James (1994) *Time Series Analysis*. Princeton: Princeton University Press.
- Harris, John (1985) 'A Survey of Macroeconomic Modelling in Africa' Paper presented to a meeting of the Eastern and Southern Africa Macroeconomic Research Network, December 7-13, Nairobi, Kenya.
- Harris, Laurence (1986) 'Conceptions of the IMF's Role in Africa', in Peter Lawrence (ed.) World Recession and The Food Crisis in Africa. London: James Curry.
- Harton, Susan & J. McLaren (1989) 'Supply Constraints in the Tanzanian Economy: Simulation Results from a Macro-econometric Model', *Journal of Policy Modelling*, 11(2): 297-313.
- Harvey, Charles (1985) Macroeconomics in Africa: Revised Pan-African Edition. London: Heinemann.
- Heckscher, E.F. (1919) 'The Effect of Foreign Trade on the Distribution of Income', *Ekonomisk Tidskrift*, 497-512.
- Heien, Dale, Jim Matthews & Abner Womack (1973) 'A Methods Note on the Guass-Seidel Algorithm for Solving Econometric Models', *Agricultural Economics Research*, 25(3): 71-80.
- Helg, Rodolfo (1990) 'A Note on the Stationarity of the Primary Commodities Relative Price Index', *Development Studies Working Papers*, *No. 31 (September)*. Oxford: University of Oxford, Queen Elizabeth House.
- Helleiner, G.K. (1981) Intra-Firm Trade and The Developing Countries. London: Macmillan Press Ltd.
- Helleiner, G.K (1989) 'Transnational Corporations and Direct Foreign Investment' in H. Chenery & T.N. Srinivasan (eds) *Handbook of Development Economics*, Vol. II, Amsterdam, North Holland.
- Helleiner, G.K. (1993) 'From Adjustment to Development in Sub-Saharan Africa: Consensus and Continuing Conflict', in G.A. Cornia & G.K. Helleiner (eds) From Adjustment to Development in Africa: Conflict, Controversy, Convergence, Consensus? New York: St. Martin's Press.
- Heller, P.S. (1975) 'A Model of Public Fiscal Behaviour in Developing Countries: Aid, Investment and Taxation', *The American Economic Review*, 65(3): 429-45.
- Helpman, Eihanan (1981) 'International Trade in the Presence of Product Differentiation, Economies of Scale and Monopolistic Competition: A Chamberlain-Heckscher-Ohlin Approach', Journal of International Economics, 11(3): 305-40.

- Herfindahl, O.C. (1955) 'Some Fundamentals of Mineral Economics', Land Economics, 31 (May): 131-38.
- Hermes, Niels & R. Lensink (1992) 'The Magnitude and Determinants of Capital Flight: The Case for Six Sub-Saharan African Countries', *De*velopment Economics Seminar Paper No. 92-1/3, Institute of Social Studies, The Hague.
- Hickman, B.G. & L.J. Lau (1973) 'Elasticities of Substitution and Export Demands in a World Trade Model', *European Economic Review*, 4: 347-80.
- Holtham, Gerald & Martine Durand (1987) 'OECD Economic Activity and Non-oil Commodity Prices: Reduced-Form Equations for INTER-LINK', OECD Department of Economics and Statistics Working Papers No. 42.
- Hopkins, A.G. (1973) An Economic History of West Africa. London: Longman.
- Hotelling, H. (1931) 'The Economics of Exhaustible Resources', Journal of Political Economy, 39 (April): 137-75.
- Houthakker, H.S. & L.D. Taylor (1970) Consumer Demand in the United States: Analysis and Projections. Cambridge: Harvard University Press.
- Hufbauer, G.C. (1966) Synthetic Material and the Theory of International Trade. Cambridge, MA: Harvard University Press.
- Husain, Ishrat (1994) ' Structural Adjustment and the Long-term Development of Sub-Saharan Africa', in van der Hoeven & F. van der Kraaij Structural Adjustment and Beyond in Sub-Saharan Africa. London: James Currey.
- Hwa, Erh-Cheng (1985) 'A Model of Price and Quantity Adjustment in Primary Commodity Markets', *Journal of Policy Modelling*, 7(2): 305-38.
- Hymer, Stephen (1976) The International Operation of National Firms: A Study of Foreign Direct Investment. Cambridge: MIT Press.
- International Monetary Fund (IMF) (1994) Internationals Financial Statistics. Washington DC: International Monetary Fund.
- Jamshidi, Ahmad (1989) 'Evaluating Global Macroeconomic Models: A Case Study of MULTIMOD', World Bank Policy, Planning and Research Working Paper Series WPS 298, Washington DC: The World Bank.
- Johansen, Søren (1988) 'Statistical Analysis of Cointegration Vectors', Journal of Economic Dynamics and Control, 12:231-54 (also in Engel & Granger, eds, 1991, Chapter 7).

- Johansen, Søren (1991) 'Estimation and Hypothesis Testing of Cointegration Vectors in Gaussian Vector Autoregressive Models', *Econometrica*, 59(6): 1551-80
- Johansen, Søren & K. Juselius (1990) 'Maximum Likelihood Estimation and Inference on Cointegration: With Applications to the Demand for Money', Oxford Bulletin of Economics and Statistics, 52(2): 169-210.
- Johnston, M.A. (1980) 'Unequal Exchange and the Flow of Surplus in the World Capitalist System', in Richard Peet (ed.) An Introduction to Marxist Theories of Underdevelopment. Canberra: The Australian National University, Research School of Pacific Studies.
- Jones, Eugene & J. Mutuura (1989) 'The Supply Responsiveness of Small Kenyan Cotton Farmers', *Journal of Developing Areas*, 23(4): 535-44.
- Jong, N. de & R. Vos (1994) 'Theoretical and Empirical Approaches to Direct Foreign Investment: A Survey of the Literature', ISS Working Paper Sub-Series on Money, Finance and Development No. 57, The Hague: Institute of Social Studies.
- Jong, N. de & R. Vos (1995a) 'Regional Blocs or Global Markets? A World Accounting Approach to Analyze Trade and Financial Linkages', Weltwirtschaftliches Archive, 131(4): 748-73.
- Jong, N. de. & R. Vos (1995b) 'Trade and Financial Flows in a World Accounting Framework: A Balanced WAM for 1990', *Review of Income* and Wealth, 41(2): 139-59.
- Jong, N. de, R. Vos, T. Jellema & H. Zebregs (1993) 'Trade and Financial Flows in A World Accounting Framework: Balance WAMs for 1989-1990', Paper prepared for UN-DESIPA and Project LINK, Institute of Social Studies, The Hague and GDUI, Erasmus University, Rotterdam.
- Jorgenson, D.W. (1963) 'Capital Theory and Investment Behavior', American Economic Review, Supplement 53(2): 247-59.
- Jungnickel, Rolf (1993) 'Recent Trends in Foreign Direct Investment', Intereconomics, 28(May/June): 118-25.
- Kalaitzandonakes, Nicholas G. & J.S. Shonkwiler (1992) 'A State-Space Approach to Perennial Crops Supply Analysis', *American Journal of Agricultural Economics*, 74(2): 343-52.
- Kaldor, Nicholas (1976) 'Inflation and Recession in the World Economy', *Economic Journal*, 86(344): 703-14.
- Kanbur, S.M.R. & David Vines (1986) 'North-South Interaction and Commodity Control', *Journal of Development Economics*, 23(2): 371-87.
- Kanbur, S.M.R. & P. Svedberg (1991) Trade and Development in Sub-Saharan Africa. Manchester: Manchester University Press.
- Kasekende, Lous, Damoni Kitabire & Matthew Martin (1995) 'Capital Inflows and Macroeconomic Policy in Sub-Saharan Africa', mimeo.

- Kao, C. (1999) 'Spurious Regression and Residual-based Tests for Cointegration in Panel Data', *Journal of Econometrics*, 90: 1-44.
- Kao, C. & M. Chiang (1998) 'On the Estimation and Inference of a Cointegrated Regression in Panel Data, mimeo, Syracuse University.
- Kay, Geoffrey (1975) Development and Underdevelopment: A Marxist Analysis. London: The Macmillan Press Ltd.
- Kayizzl-Mugerwa, Steve (1990) 'Zambia: A Note on the Macroeconomic Impacts of Copper Prices', *Eastern Africa Economic Review*, 6(2): 143-47.
- Kennedy, P. (1992) A Guide to Econometrics. Oxford: Blackwell.
- Keynes, J.M. (1936) The General Theory of Employment, Interest and Money. London: Macmillan.
- Khan, Mohsin S. & Carmen M. Reinhart (1990) 'Private Investment and Economic Growth in Developing Countries', *World Development*, 18 (1): 19-27.
- Khan, Mohsin S. & M. Knight (1983) 'Sources of Payment Problems in LDCs', *Finance and Development*, 20(4): 2-5.
- Kindleberger, Charles (1969) American Business Abroad: Six Lectures on Direct Investment. New Heaven: Yale University Press.
- Kiss, Judit (1984) 'What Kind of Future for Africa: Lesson from an Econometric World Trade Model for Africa', *Studies on Developing Countries*, No. 113. Budapest: Institute for World Economy of the Hungarian Academy of Science.
- Klein, L.R. & A. van Peterson (1973) 'Forecasting World Trade within project LINK', in R.J. Ball (ed.) *The International Linkage of National Economic Models*. Amsterdam: North-Holland Publishing.
- Kletzer, Kenneth (1984) 'Asymmetries of Information and LDC Borrowing with Sovereign Risk', *Economic Journal*, 94(374): 287-307.
- Kofman. Paul (1991) 'Managing Primary Commodity Trade: On The Use of Future Markets', PhD dissertation, Erasmus University, Rotterdam.
- Kojima, Kiyoshi (1982) 'Macroeconomic Versus International Business Approach to Direct Foreign Investment', in Peter J. Buckley (ed.) (1990) *International Investment*. England: Edward Elgar Publishing Limited.
- Koncazcki, Z.A (1977) 'Portugal's Economic Policy in Africa', in Z.A. Koncazcki & J.M. Knoxzacki, An Economic History of Tropical Africa: Vol. II The Colonial Period. London: Frank Cass.
- Koncazcki, Zbigniew (1990) 'Socioeconomic Formations of the Southern African Iron Age: An Overview', in Z.A. Koncazcki, J.L. Parpart & T.M. Shaw (eds) *Studies in the Economic History of Southern Africa Vol. 1, The Front-line States.* London: Frank Cass.

Koutsoyiannis, A. (1975) Modern Microeconomics. London: Macmillan.

- Kouri, P. & M. Porter (1974) 'International Capital Flows and Portfolio Equilibrium', *Journal of Political Economy*, 82(3): 443-68.
- Koyck, L.M. (1954) Distributed Lags and Investment Analysis. Amsterdam: North-Holland.
- Kravis, Irvine (1971) 'The Current Case for Import Limitation' in Commission on International Trade and Investment Policy. United States Economic Policy in an Interdependent World. Washington DC: US Government Printing.
- Krugman, Paul (1979a) 'A Model of Innovation, Technology Transfer, and The World Distribution of Income', *Journal of Political Economy*, 87(2): 253-66.
- Krugman, Paul (1979b) 'Increasing Returns, Monopolistic Competition, and International Trade', *Journal of International Economics*, 9(4): 469-79.
- Krugman, Paul (1980) 'Scale Economies, Product Differentiation, and the Pattern of Trade', *The American Economic Review*, 70(5): 950-59.
- Krugman, Paul (1981) 'Intra-industry Specialization and the Gains from Trade', Journal of Political Economy, 89(5): 959-73.
- Krugman, Paul (1988) 'External Shocks and Domestic Policy Response', in Rudiger Dornbusch, F. Leslie & C. Helmers (eds) The Open Economy: Tools for Policy Makers in Developing Countries. Oxford: Oxford University Press.
- Krugman, Paul (1989) 'Market-Based Debt-Reduction Schemes', in JacobA. Frenkel, Michael P. Dooley & Peter Wickham (eds) *Analytical Issues* in Debt. Washington DC: International Monetary Fund.
- Krugman, Paul (1992) 'Does the New Trade Theory Require a New Trade Policy', *The World Economy*, 15(4): 423-41.
- Krumm, Kathie L. (1985) 'The External Debt of Sub-Saharan Africa' World Bank Staff Working Papers No. 741, Washington DC: The World Bank.
- Kwack, S.Y. (1989) 'Overview of the Prototype African Model', in D. Salvatore, African Development Prospects: A Policy Modelling Approach. New York: Taylor & Francis.
- Labys. W.C. (1978) 'Commodity Modelling', in F. Gerard Adams & J.R. Behrman (eds) *Econometric Models of World Commodity Policy*. Toronto: Heath and Company.
- Labys, W.C. & P.K. Pollak (1984) Commodity Model for Forecasting and Policy Analysis. London: Croom Help.
- Lall, Sanjaya (1973) 'Transfer-Pricing by Multinational Manufacturing Firms', Oxford Bulletin of Economics and Statistics, 35(3): 173-95.

- Lall, Sanjaya (1995) 'Structural Adjustment and African Industry', World Development, 23(12): 2019-31.
- Lancaster, Kelvin (1980) 'Intra-industry Trade Under Perfect Monopolistic Competition', *Journal of International Economics*, 10(2): 151-75.
- Larsen, Flemming, J. Llewllyn & S. Potter (1883) 'International Economic Linkages', OECD Economic Studies, No. 1 (Autumn).
- Lawrence, Peter (1986) 'Introduction: Word Recession and the Food Crisis in Africa', in Peter Lawrence (ed.) World Recession and the Food Crisis in Africa. London: James Curry.
- Lawson, Clive, M. Peacock & S. Pratten (1996) 'Realism, Understanding and Institutions', *Cambridge Journal of Economics*, 20(1): 137-51.
- Lawson, Tony (1989) 'Abstraction, Tendencies and Stylised Facts: a Realist Approach to Economic Analysis', in T. Lawson, J.G. Palma & J. Sender (eds) Kaldor's Political Economy. London. Academic Press Limited.
- Lee, S. Hun (1993) 'Are the Credit Ratings Assigned by Bankers Based on the Willingness of LDC Borrowers to Repay', *Journal of Development Economics*, 40(2): 349-59.
- Lemma Mered (1993) 'Modelling the Ethiopian Economy: Experience and Prospects', mimeo.
- Lensink, Robert (1996) Structural Adjustment in Sub-Saharan Africa. London: Longman.
- Lensink, Robert & P.A.G. van Bergeijk (1991) 'The Determinants of Developing Countries' Access to the International Capital Markets', *The Journal of Development Studies*, 28(1): 86-103.
- Leontief, Wassily (1954) 'Domestic Production and Foreign Trade: The American Capital Position Re-examined', *Economia Internazionale* 2 (1): 3-32 (reprinted in Richard Caves & Harry G. Johnson (1980) *Readings in International Economics*. London: George Allen & Unwin, pp. 503-27).
- Leontief, Wassily (1956) 'Factor Proportion and the Structure of American Trade: Further Theoretical and Empirical Analysis', *Review of Economics and Statistics*, 38: 386-407.
- Levin, A. & C.-F. Lin (1993) 'Unit Root Test in Panel Data: New Results' Department of Economics Discussion Paper, UC-San Diego.
- Lewis, A.W. (1954) 'Economic Development with Unlimited Supply of Labour', *The Manchester School*, 22 (May): 139-91.
- Leys, Colin (1996) The Rise and Fall of Development Theory. London: James Currey.
- Linder, S.B. (1961) An Essay on Trade and Transformation. New York: John Wiley & Sons.

- Lipton, Peter. (1991) Inference to the Best Explanation. London: Routledge.
- Lipumba, Nguyuru, B. Ndulu, S. Horton & A.J. Plourde (1988) 'A Supply Constrained Macro-Econometric Model of Tanzania', *Economic Modelling*, 5(4): 354-80.
- Longmire, Philip (1990) 'Land and Labour in the Namibian Economy', in Z.A. Koncazcki, J.L. Parpart & T.M. Shaw (eds) Studies in the Economic History of Southern Africa, Vol. 1, The Front-line States. London: Frank Cass Ltd.
- Love, Roy (1994) 'Drought, Dutch Disease and Controlled Transition in Botswana Agriculture', *Journal of Southern African Studies*, 20(1): 71-83.
- Lucas, Robert E.B. (1993) 'On the Determinants of Direct Foreign Investment: Evidence form East and Southeast Asia', World Development, 21 (3): 391-406.
- Luttik, Joke (1992) *Trade and Finance in a World Accounting Matrix*. Unpublished dissertation, Institute of Social Studies, The Hague and University of Amsterdam, Amsterdam.
- MacDougall, D. (1960) 'The Benefits and Costs of Private Investment from Abroad: A theoretical Approach', *Economic Record*, 36: 13-55.
- Macedo, Jorge Braga de (1982) 'Currency Diversification and Export Competitiveness: A Model of the Dutch Disease in Egypt', *Journal of Development Economics*. 11: 287-306.
- Mack, Andrew (1974) 'Theories of Imperialism: The European Perspective', The Journal of Conflict Resolution, 18 (3): 514-35.
- Magdoff, Harry (1969) The Age of Imperialism: The Economics of US Foreign Policy. London: Monthly Review Inc.
- Magdoff, Harry (1992) 'Globalization To What End?', in Ralph Miliband & Leo Panitch (eds) Socialist Register. London: The Merlin Press.
- Maizels, Alfred, Robert Bacon & George Mavrota (1992) 'The Potential for Supply Management of Commodities Exported by Developing Countries: The Case of Cocoa', Report prepared at Queen Elizabeth House, Oxford University.
- Maizels, Alfred, Robert Bacon & George Mavrota (1993a) 'The Potential for Supply Management of Commodities Exported by Developing Countries: The Case of Coffee', Report prepared at Queen Elizabeth House, Oxford University.
- Maizels, Alfred, Robert Bacon & George Mavrota (1993b) 'The Potential for Supply Management of Commodities Exported by Developing Countries: The Case of Tea', Report prepared at Queen Elizabeth House, Oxford University.

- Malawi National Statistical Office (1971) Balance of Payments: 1970. Zomba: Government Publishing House.
- Malley, J.R. et al. (1991) 'The Specification, Estimation and Simulation of a Small Global Macroeconomic Model', *Economic Modelling*, 8(4), Butterworth-Heinemann.
- Mamdani, Mohammed (1994) 'A Critical Analysis of the IMF Program in Uganda', in Ulf Himmelstrand, Kabiru Kinyanjui & Edward Mburugu (eds) African Perspective on Development. London: James Currey Ltd.
- Marquez, J.R. & P. Pauly (1987) 'International Policy Co-ordination and Growth Prospects of Developing Countries: An Optimal Control Application', *Journal of Development Economics*, 25: 89-104.
- Masson, P., S. Symanski, R. Hass & G. Merdith (1990) MULTIMOD Mark II: A Revised and Extended Model. Washington DC: IMF.
- Masson, P., S. Symanski, R. Hass & M. Dooley (1988) 'MULTIMOD A Multi-region Econometric Model', in *IMF Staff Studies for the World Economic Outlook.* Washington DC: IMF.
- Mbelle, Ammon & Thomas Sterner (1991) 'Foreign Exchange and Industrial Development: A Frontier Production Function Analysis of Two Tanzanian Industries', *World Development*, 19(4): 341-47.
- McCoskey, Suzanne & Chihwa Kao (1997) 'A Residual-based Test of the Null of Cointegraion in Panel Data', memo, University of Syracuse and United States Naval Academy.
- McDonald, D. (1960) 'The Benefits and Costs of Private Investment from Abroad: A Theoretical Approach', *Economic Record*, 36: 13-55.
- McGillivary, Mark (1994) 'The Impact of Foreign Aid on the Fiscal Behaviour of Asian LDC Governments: A Comment on Khan and Hoshino (1992), World Development, 22(12): 2015-17.
- McGillivary, Mark & H. White (1993) 'Aid Principles and Policy: An Operational Bias for the Assessment of Donor Performance', Paper prepared for the Workshop on Aid Policy and Performance, EADI VIIth General Conference, Technical University of Berlin, Berlin, 15-18 September.
- McKibbin, W.J. & J.D. Sachs (1991) Global Linkage: Macroeconomic interdependence and Co-operation in the World Economy. Washington DC: The Brooking Institution.
- McKibbin, Warwick & T.J. Bok (1993) 'The Asia-Pacific Region in an Integrated World Economy', in Asia-Pacific Economic Modelling Conference – 93 (August 25-28) organized by Economic Modelling Bureau of Australia and Australian National University.

- McKinley, R.D. & R. Little (1979) 'The US Aid Relationship: A Test of The Recipient Need and The Donor Interest Models, *Political Studies*, 27(2): 237-50.
- Meade, J. (1951) The Theory of International Economic Policy, Vol I: The Balance of Payments. Oxford: Oxford University Press.
- Memedovic, Olga (1994) 'On The Theory and Measurement of Comparative Advantage: An Empirical Analysis of Yugoslav Trade in Manufacturing with the OECD Countries, 1970-1986', *Tinbergen Institute Research Series No. 65.* Rotterdam: Erasmus University. Tinbergen Institute.
- Mikesell, Raymond (1968) The Economics of Foreign Aid. Chicago: Aldine Publishing Company.
- Mistry, Percy S. (1988) African Debt: The Case for Relief for Sub-Saharan Africa. Oxford University, Queen Elizabeth House: Oxford International Associates.
- Mistry, Percy S. (1991) African Debt Revisited: Procrastination or Progress? The Hague: FONDAD.
- Mistry, Percy S. (1994) Multilateral Debt: An Emerging Crisis. The Hague: FONDAD.
- Mistry, Percy S. (1996) Resolving Africa's Multilateral Debt Problem: A Response to the IMF and The World Bank. The Hague: FONDAD.
- Mohammed, Baba G. (1989) 'Oil boom and Structure of Nigerian Economy 1970-87', unpublished MA thesis, Institute of Social Studies, The Hague.
- Molana & Vines (1989) 'North-South Growth and Terms of Trade: A Model on Kaldorian Lines', *The Economic Journal*, 99(396): 443-53.
- Montiel, P. (1991) 'Dynamic Response to Policy and Exogenous Shocks in an Empirical Developing Country Model with Rational Expectations', *Economic Modelling*, April.
- Mosley, Paul (1985) 'The Political Economy of Foreign Aid: A Model of the Market for a Public Good', *Economic Development and Cultural Change*, 33(2): 373-94.
- Mosley, Paul & J. Weeks (1993) 'Has Recovery Begun? Africa's Adjustment in the 1980s Revisited', World Development, 21(10): 1583-606.
- Mosley, Paul, Turan Subasat & John Weeks (1995) 'Assessing Adjustment in Africa', World Development, 23(9): 1459-73.
- Mosley, Paul, J. Hudson & S. Horrell (1987) 'Aid, The Public Sector and the Market in Less Developed Countries', *The Economic Journal*, 97(181): 616-41.
- Motamen, H. (ed.) (1988) Economic Modelling in OECD Countries. London: Chapman & Hall.

- Moutos, Thomas & David Vines (1988) 'The Simple Macroeconomics of North-South Interaction', *American Economic Association (AEA) Papers* and Proceedings, Vol. 79, No. 2.
- Moyo, Nelson P. & N. Amin (1992) 'The Development Experience of Post-Colonial Sub-Saharan Africa', in Willem L.M. Adriaansen & J.G. Waardenburg (eds) A Dual World Economy: Forty Years of Development Experience. Bombay: Oxford University Press.
- Mukherjee, Chandan & Marc Wuyts (1991) 'Data Analysis in Development Research: An Argument', *ISS Worrking Paper No. 103.* The Hague: Institute of Social Studies.
- Mukherjee, Chandan, Howard White & Marc Wuyts (1998) Econometrics and Data Analysis for Developing Countries. London: Routledge.
- Mulalu, Sally Linda (1987) 'The Role of Foreign Finance in Alleviating Zambia's Balance of Payment Problem', unpublished MA thesis, Institute of Social Studies, The Hague.
- Mundell, Robert (1960) 'The Monetary Dynamics of International Adjustment Under Fixed and Flexible Rates', *Quarterly Journal of Eco*nomics, 74 (May): 227-57.
- Mundell, Robert (1963) 'Capital Mobility and Stabilization Policy under Fixed and Flexible Exchange Rate', *Canadian Journal of Economics* and Political Science, 29 (November): 475-85.
- Mundell, Robert (1968) International Economics. New York: Macmillan.
- Munro, J. Forbes (1976) Africa and the International Economy 1800-1960. London: J.M. Dent & Sons Ltd.
- Murshed, S. Mansoob (1990) 'Stabilization Policy, Commercial Policy and Debt in a North-South Framework', in Graham Bird (ed.) *The International Financial Regime*, London: Surrey University Press.
- Muscatelli, Vito A. & David Vines (1991) 'Third World Debt and Macroeconomic Interaction Between the North and South', *Journal of Development Studies*, 27(3): 146-66.
- Musonda, F. Mndeme & Eliab Luvanda (1991) 'The Consequences of the 1976-77 Coffee Boom on the Tanzanian Economy: A Test of the Dutch Disease Model', *Eastern Africa Economic Review*, 7(2): 1-16.
- Mwale, Charles (1983) 'Domestic Adjustment to External Shocks of the 1970s: The Case of Zambia', Unpublished MA thesis, Institute of Social Studies, The Hague.
- National Bank of Ethiopia (various years: 1970-1990) Annual Report. Addis Ababa.
- Ndulu, Benno J. (1986) 'Investment, Output Growth and Capacity Utilization in an African Economy: The Case of Manufacturing Sector in Tanzania', *Eastern Africa Economic Review*, 2(1): 14-31.

- Ndulu, Benno J. (1991) 'Growth and Adjustment in Sub-Saharan Africa', in Ajay Chhibber & Stanley Fischer (eds) *Economic Reform in Sub-Saharan Africa*. Washington DC: The World Bank.
- Neary, J. Peter & S. van Wijnbergen (1986) 'Natural Resource and Macroeconomy: A Theoretical Framework', in J. Peter Neary & S. van Wijnbergen (eds) *Natural Resource and Macroeconomy*. Oxford: Basil Blackwell.
- Negishi, Takashi (1989) History of Economic Theory. Amsterdam: North Holland.
- Nerlove, Marc (1958) The Dynamics of Supply: Estimation of Farmers' Response to Price. Baltimore: Johns Hopkins University Press.
- Neumark, Daniel S. (1977) 'Trans-Sahara Trade in the Middle Ages', in Z.A. Koncazcki & J.M. Knoxzacki (eds) An Economic History of Tropical Africa: Vol. I, The Pre-Colonial Period. London: Frank Cass.
- Ngwenya, A.R. & P.K. Bugembe (1987) 'The Origins and Causes of the African Social and Economic Crisis', in International Conference on Africa: the Challenge of Economic Recovery and Accelerated Development, Abuja, Nigeria (15-19 June).
- Njuguna, Angelica E. (1999) 'Growth and Convergence in A Disequilibrium Solow-Swan Model: The Case of ASEAN Countries 1960-1995', Ph.D. dissertation, University of New England, Australia.
- Nzula, A.T., I.I. Potekhin & A.Z. Zusmaovich (1979) Forced Labour in Colonial Africa. London: Zed Press.
- Organization of African Unity (OAU) (1981) Lagos Plan of Action for the Economic Development of Africa: 1980-2000. Addis Ababa: OAU.
- Ocampo, Jose Antonio (1986) 'New Development in Trade Theory and LDCs', Journal of Development Economics, 22: 129-70.
- Odedokun, M.O. (1996) 'Analysis of Commercial Bank Lending to African Countries', *African Development*, 21(1): 99-121.
- OECD (1985) Twenty-five Years of Development Co-operation: A Review. Paris: OECD.
- OECD (1988) OECD INTERLINK System: Reference Manual. Paris: OECD.
- Ofuatey-Kodjoe (1991) 'African International Political Economy: An Assessment of the Current Literature', in Craig N. Murphy & Roger Tooze (eds) *The International Political Economy*. Boulder: Lynne Rienner Publisher.
- Ohlin, B. (1933) Interregional and International Trade. Cambridge: Harvard University Press.
- Olafin, Sam (1989) 'Obstacles to African Development', in D. Salvatore, African Development Prospects: A Policy Modelling Approach. New York: Taylor & Francis.

- Olfin S. & Iyaniwura (1983) 'From Oil Shortage to Oil Glut: Simulation of Growth Prospects in Nigerian Economy', *Journal of Policy Modelling*, 5 (3): 363-78.
- Onimode, Bade (1988) A Ppolitical Economy of the African Crisis. London: Zed Books Ltd.
- Opeskin, Brain R. (1996) 'The Moral Foundations of Foreign Aid', World Development, 24 (1): 21-44.
- Oshikoya, W.T. (1989) 'Foreign Borrowing, Agricultural Productivity, and the Nigerian Economy: A Macro-sectoral Analysis', *Journal of Policy Modelling*, 11(4): 531-46.
- Oxfam (1997) 'Poor Countries Debt Relief: False Dawn or New Hope for Poverty Reduction?', Oxfam Policy Papers (<u>http://www.oneworld.org/</u> <u>Oxfam/papers/debdawn.htm</u>).
- Pack, Howard & Janet Rothenberg Pack (1990) 'Is Foreign Aid Fungible? The Case of Indonesia', *Economic Journal*, 100 (March): 188-94.
- Pal, Suresh (1992) 'Agricultural Exports of India: Issues of Growth and Instability', *Indian Journal of Agricultural Economics*, 47(2): 185-95.
- Palaskas, Theodosis (1986) 'An Econometric Analysis of Factors Influencing the Production of Dried Vine Fruits in Greece and Export Markets', Ph.D. dissertation, Brasenose College, University of Oxford.
- Palaskas, Theodosis & C.L. Gilbert (1990) 'Consistent Estimation of a Disequilibrium Model of Primary Commodity Price Adjustment', *Applied Economics*, 22: 1417-26.
- Papanek, Gustav (1972) 'The Effect of Aid and Other Resources Transfers on Saving and Growth in Less Developed Countries', *The Economic Journal*, 82(3): 934-50.
- Parish, F. (1938) 'Causes of Changes in Gold Supply', *Economica*, 5 (Nov-ember): 379-409.
- Patnaik, Prabhat (1996) 'Trade as a Mechanism of Economic Retrogression', in Henry Bernstein & Tom Brass (eds) Agrarian Questions: Essays in Appreciation of T.J. Byers. London: Frank Cass (also in special issue of The Journal of Peasant Studies, Vol. 24, Nos. 1 and 2, October 1996).
- Pedersen, Karsten N. (1994) 'The Prototypical Developing Country Model of the World Bank Global Economic Model', in Scobie, H.M. (ed.) *Economic Progress and Growth*. London: Chapman & Hall.
- Pedroni, P. (1995) 'Panel Cointegration: Asymptotic and Fine Sample Properties of Pooled Time Series Test with an Application to the PPP Hypothesis', memo, Department of Economics, Indiana University.

- Pesaran, M. Hashem & Ron Smith (1995) 'Estimating Long-run Relationships from Dynamic Heterogeneous Panels', *Journal of Econometrics*, 68: 79-113.
- Peterson, Christian, K.N. Pedersen, E.J. Riordan, R.A Lynn & T. Bradly (1991) 'The Structure of a World Bank Global Economic Model', Paper presented at the Conference on Modelling Global Economic Interdependence, Korean Development Institute (May 27-28).
- Pim, Alan (1977) 'Capital Investment in Sub-Saharan Africa', in Z.A. Koncazcki and J.M. Knoxzacki (eds) An Economic History of Tropical Africa: Vol. II, The Colonial Period. London: Frank Cass.
- Pleskovi, Boris (1989) 'Inter-industry Flows in a General Equilibrium Model of Fiscal Incidence: An Application to Egypt', *Journal of Policy Modelling*, 11(1): 157-77.
- Posner, M.V. (1961) 'International Trade and Technical Change', Oxford Economic Papers, 13: 323-41.
- Prebisch, Raul (1962, earlier 1950) 'The Economic Development of Latin America and its Principal Problems', *Economic Bulletin for Latin America*, 7(1): 1-22.
- Press, H. William, B.P. Flannery, S.A. Teukolsky & W.T. Vetterling (1989) Numerical Recipes: The Art of Scientific Computing (FORTRAN Version). Cambridge: Cambridge University Press.
- Quibria, M.G. (1986) 'A Note on Foreign Investment, The Savings Function and Immiserization of National Welfare', *Journal of Development Economics*, 21(2): 361-72.
- Ramanujam, Prathap & D. Vines (1990) 'Commodity Prices, Financial Markets and World Income: A Structural Rational Expectations Model', *Applied Economics*, 22: 509-27.
- Rao, B. Bhaskara (ed.) (1994) Cointegration for The Applied Economist. London: Macmillan.
- Rattsø, Jørn (1992a) 'The Asymmetric Relation Between Sub-Saharan Africa and The Rest of the World: A Theoretical Analysis of the Role of Import Compression', *Eastern Africa Economic Review*, 8(1): 1-11.
- Rattsø, Jørn (1992b) 'Import Compression, Investment Determination and Macroeconomic Adjustment: A Macroeconomic Model Relevant for Sub-Saharan Africa', *Development Economics Seminar Paper NR. 92-*1/4a. The Hague: Institute of Social Studies.
- Rattsø, Jørn (1994) 'Medium-run Adjustment under Import Compression: Macroeconomic Analysis Relevant for Sub-Saharan Africa', *Journal of Development Economics*, 45: 35-54.
- Reinhart, Carmen M. & Peter Wickham (1994) 'Commodity Prices: Cyclical Weakness or Secular Decline?', *IMF Staff Papers*, 41(2): 175-213.
- Republic of Ghana, Central Bureau of Statistics (1981) *Economic Survey:* 1977-1980. Accra.
- Republic of Kenya (1989) National Development Plan: 1989-1991. Nairobi.
- Reserve Bank of Malawi (various years: 1970-1992) Reports and Accounts. Zomba.
- Ricardo, David (1817) On the Principle of Political Economy and Taxation. Reprinted in Sraffa & Dobb (eds) The Works and Corresponds of David Ricardo, Vol I: David Ricardo on the Principle of Political Economy and Taxation. Cambridge: Cambridge University Press.
- Riddel, R. (1987) Foreign Aid Reconsidered. London: James Curry.
- Rodney, Walter (1972) How Europe Underdeveloped Africa. London: Bogle L'Ouveture Publishing.
- Romer, Michael (1989) 'Macroeconomics of Counterpart Funds Revisited', World Development, 17 (June): 795-808.
- Ruttan, Vernon (1992) 'Foreign Economic Assistance and Economic Development', in L.B. Fletcher (ed.) World Food in the 1990s: Production, Trade and Aid. Boulder: Westview press.
- Rwegasira, Delphin (1984) 'Exchange Rate and Management of the External Sector in Sub-Saharan Africa', *African Development Bank Economic Research Papers No. 3.* Abidjan: ADB.
- Sachs, Jeffrey D. (1984) 'Theoretical Issues in International Borrowing', *Princeton Studies in International Finance, No. 54*, Department of Economics, Princeton University.
- Salter, W.E.G (1959) 'Internal and External Balance: The Role of Price and Expenditure Effect', *The Economic Record*, 226-238.
- Salvatore, D. (1989) African Development Prospects: A Policy Modelling Approach. New York: Taylor & Francis.
- Samuelson, L.W. & E. Kurihara (1980) 'OECD Trade Linkage Methods Applied to EPA World Econometric Model', *Economic Bulletin, No. 18.* Japan: Economic Research Institute.
- Samuelson, P.A. (1948) 'International Trade and The Equalization of Factor Price', *Economic Journal*, 58(230): 163-85.
- Sapsford, David & V.N. Balasubramanyam (1994) 'The Long-run Behaviour of the Relative Price of Primary Commodities: Statistical Evidence and Policy Implications', *World Development*, 22(11): 1737-45.
- Sarkar, Prabirjit (1986) 'The Singer-Prebisch Hypothesis: A Statistical Evaluation', *Cambridge Journal of Economics*, 10(4): 355-71.
- Sarkar, Prabirjit (1992) 'Long-term Behaviour of Terms of Trade of Primary Products vis-à-vis Manufactures: A Critical Review of Recent Debate', *Economic and Political Weekly*, 25 (June): 1612-14.

- Sarkar, Prabirjit (1994) 'North-South Terms of Trade and Growth: A Macroeconomic Framework on Kaldorian Lines', *World Development*, 22(11): 1711-15.
- Sarkar, Prabirjit (1996) 'Terms of Trade of the South in the Process of Growth: What the Neo-Kaleckian Macro Model Shows', *Development* Studies Working Papers, No. 99 (November), Oxford: University of Oxford, Queen Elizabeth House.
- Sarmad, Khwaja (1990) 'Public and Private Investment and Economic Growth', ISS Working Paper Series on Money, Finance and Development No. 34, Institute of Social Studies, The Hague, The Netherlands.
- Sau, R. (1978) Unequal Exchange, Imperialism and Underdevelopment: An Essay on the Political Economy of World Capitalism. Calcutta: Oxford University Press.
- Saunders, Paul & Andrew Dean (1986) 'The International Debt Situation and Linkages Between Developing Countries and the OECD', OECD Economic Studies, No. 7, 156-203.
- Schneider, Frederick & Bruno S. Frey (1985) 'Economic and Political Determinants of Foreign Direct Investment', World Development, 13(2): 161-75.
- Schumpeter, J. (1954) *History of Economic Analysis*. London: George Allen & Unwin.
- Seleti, Yonah N. (1990) 'The Development of Dependent Capitalism in Portuguese Africa', in Z.A. Koncazcki, J.L. Parpart & T.M. Shaw (eds) Studies in the Economic History of Southern Afric, a Vol. 1, The Frontline States. London: Frank Cass Ltd.
- Shaikh, Anwar (1984) 'The Laws of International Exchange', in E.J. Nell, Growth, Profits and Prosperity. Cambridge: Cambridge University Press.
- Shu, David Lim Lin (1975) Supply Response of Primary Producers. Kuala Lumpur, Malaya: Panerbit University.
- Sierra Leone Ministry of Development and Economic Planning, Central Planning Unit (1981) Annual Plan 1977-78. Freetown: Sierra Leone Government.
- Sierra Leone Ministry of Development and Economic Planning, Central Planning Unit (1981) *The National Development Plan 1981/82-1985/86*. Freetown: Sierra Leone Government.
- Singer, H.W. (1950) 'US Foreign Investment in Underdeveloped Areas: The Distribution of Gains Between Investing and Borrowing Countries', *The American Economic Review: Papers and Proceedings*, 40(2): 473-85.
- Smith, Adam (1776) *The Wealth of Nations*. Reprinted 1991, London: David Campbell Publishers Limited.

- Smith, G.W. & J. Cuddington (1985) (eds) International Debt and Developing Countries. Washington DC: World Bank.
- Sodersten, Bo & Geoffrey Reed (1994) International Economics, 3rd edn. London: Macmillan.
- Solink, Bruno (1974) 'An Equilibrium Model of the International Capital Market', *Journal of Economic Theory*, 8: 500-524. Reprinted in Peter J. Buckley (ed.) (1990) *International Investment*. England: Edward Elgar Publishing Limited.
- Soludo, Charles (1993) 'Simulating Global Policy Scenarios', in a CEPR Conference on Macroeconomic Interaction Between North and South, Oxford (23/24 April).
- Sommers, L.M. & Assefa Mehertu (1992) 'Trade Patterns and Trends in the African-European Trading Area: Lesson for Sub-Saharan Africa from the Era of the Lome Accords 1975-1988', *Africa Development*, 17 (2): 5-26.
- Sparks, Donald L. (1991) 'Economic Trends in Africa South of the Sahara, 1990', in *Africa South of the Sahara 1991, 20th edn*. London: Europa Publications Limited.
- Spraos, John (1983) Inequalising Trade? A Study of Traditional North/South Specialisation in the Context of Terms of Trade Concepts. Oxford: Clarendon Press.
- Stafanski, Bogdan (1990) 'An Alternative Development Strategy for Africa', *African Bulletin No. 36.* Warsaw: University of Warsaw, Institute of Developing Countries.
- Stein, Leslie (1977) 'Export Trends in East Africa', Development and Change, 8(1): 103-16.
- Stein, Leslie (1984) Trade and Structural Change. London: Croom Helm.
- Stewart, F. (1984) 'Recent Theories of International Trade: Some Implication for the South', in Henry Kierzkowski (ed.) Monopolistic Competition and International Trade. Oxford: Clarendon Press.
- Stewart, Frances (1993) 'Are Short-term Policies Consistent with Longterm Development Needs in Africa', in G.A. Cornia & G.K. Helleiner, From Adjustment to Development in Africa: Conflict, Controversy, Convergence, Consensus? New York: St. Martin's Press.
- Stiglitz, Joseph & Andrew Weiss (1981) 'Credit Rationing in Markets with Imperfect Information', *The American Economic Review*, 71(3): 393-410.
- Stock, J.H. (1987), 'Asymptotic Properties of Least Square Estimators of Co-integrating Vectors', *Econometrica*, 55: 1035-56.
- Stopler, Wolfgang F. & Paul A. Samuelson (1941) 'Protection and Real Wages', *Review of Economic Studies*, 9: 58-73.

- Sutcliffe, Bob (1986) 'Africa and the World Economic Crisis', in Peter Lawrence (ed.) World Recession and the Food Crisis in Africa. London: James Curry.
- Svedberg, Peter (1990) 'The Export Performance of Sub-Saharan Africa', in Frimpong-Ansah, Jonathan & P. Svedberg (eds) *Trade and Development in Sub-Saharan Africa*. Manchester: Manchester University Press.
- Svedberg, Peter (1991) 'The Export Performance of Sub-Saharan Africa', *Economic Development and Cultural Change*, 39(3): 549-66.
- Swan, T.W. (1960) 'Economic Control in a Dependent Economy', *The Economic Record*, (March): 51-66.
- Tait, Alan A. & P.S. Heller (1982) 'International Comparisons of Government Expenditure', Occasional Paper No. 10. Washington DC: International Monetary Fund.
- Taiwo, I.O. (1991) 'Commodity Prices and Debt Crisis in Sub-Saharan Africa', Eastern Africa Economic Review, 7 (2): 39-50.
- Taplin, G.B. (1973) 'A Model of World Trade', in R.J. Ball (ed.) The International Linkage of National Economic Models. Amsterdam: North-Holland Publishing Company.
- Taylor, J. (1988) 'Treatment of Expectations in Large Multi-country Models' in C. Ralph et al. (eds) *Empirical Macroeconomics for Interdependent Economies*. Washington DC: The Brooking Institution.
- Taylor, Lance D. (1981) 'South-North Trade and Southern Growth: Bleak Prospects from the Structuralist Point of View', *Journal of International Economics*, 11(4): 589-602.
- Taylor, Lance D. (1983) Structuralist Macroeconomics: Applicable Models for the Third World. New York: Basic Books.
- Taylor, Lance D. (1991) Income Distribution, Inflation and Growth: Lectures on Structuralist Macroeconomic Theory. Cambridge, Mass: The MIT Press.
- The Socialist Peoples Libyan Arab Jamahiria Secretariat of Planning (1978) Economic and Social Achievements: 1970-1977, Tripoli.
- Thirlwall, A.P. (1991) 'The Terms of Trade of Primary Commodities, Debt and Development', in Paul Davidson & J.A. Kregel (eds) *Economic Problems of the 1990s: Europe, the Developing Countries and the United States.* England: Edward Elgar Publishing Ltd.
- Thomas, R.L. (1993) Introductory Econometrics: Theory and Application. London: Longman.
- Tobin, J. (1969) 'A General Equilibrium Approach to Monetary Theory', Journal of Money, Credit and Banking, 1 (February): 15-29.

- Trap, Finn (1993) Stabilization and Structural Adjustment: Macroeconomic Frameworks For Analysing The Crisis in Sub-Saharan Africa. London: Routledge.
- UN (1949) International Capital Movements During the Inter-War Period. New York: United Nations, Department of Economic Affairs.
- UN (1954) The International Flow of Private Capital 1946-1952. New York: United Nations, Department of Economic Affairs.
- UN, OECD, IMF, WB, CE (1993) System of National Accounts. New York: United Nations.
- UNCTAD Staff (1973) 'Models for Developing Countries', in R.J. Ball (ed.) *The International Linkage of National Economic Models*. Amsterdam: North Holland Publishing Co.
- UNCTAD (1987, 1988, 1989, 1990, 1991, 1992, 1994) Commodity Yearbook. Geneva: UNCTAD.
- UNCTAD (various years) Monthly Commodity Price Bulletin. Geneva: UNCTAD.
- UNCTAD (1989) Monthly Commodity Price Bulletin: Supplement 1970-1989, November. Geneva: UNCTAD.
- UNCTAD (1993) Handbook of International Trade and Development Statistics. Geneva: UNCTAD.
- United Republic of Tanzania, Ministry of Planning and Economic Affairs (1983) *The Economic Survey: 1982*, Dar es Salaam.
- van Arkadie, Brain (1986) 'Some Realities of Adjustment: An Introduction', *Development and Change*, 17(3): 371-86.
- van der Hoeven, Rolph (1993) 'External Dependence, Structural Adjustment and Development Aid in Sub-Saharan Africa', in *ISS International Policy Workshop on International Capital Flows and Economic Adjustment in Developing Countries.* The Hague: Institute of Social Studies.
- van Frausum, Yves & D.E. Sahn (1993) 'An Econometric Model for Malawi: Measuring the Effects of External Shocks and Policies', *Journal of Policy Modelling*, 15(3): 313-27.
- van Wijnbergen, Sweder (1984) 'The "Dutch Disease": A Disease After All?', *The Economic Journal*, March: 41-55.
- van Wijnbergen, Sweder (1986a) 'Aid, Export Promotion and the Real Exchange Rate: An African Dilemma', *Macroeconomics Division Development Research Department*, The World Bank.
- van Wijnbergen, Sweder (1986b) 'Macroeconomic Aspects of The Effectiveness of Foreign Aid: On the Two-Gap Model, Home Goods Disequilibrum and Real Exchange Rate Misalignment', *Journal of International Economics*, 21: 123-36.

- Vansina, J. (1977) 'Long-distance Trade-routes in Central Africa', in Z.A. Koncazcki & J.M. Knoxzacki (eds) An Economic History of Tropical Africa: Vol. 1: The Pre-Colonial Period. London: Frank Cass.
- Vernon, Raymond (1966) 'International Investment and International Trade in the Product Cycle', *Quarterly Journal of Economics*, 80(2): 190-207.
- Vines, D.A. (1984) 'A North-South Growth Model along Kaldorian Lines', *CEPAR Discussion No. 26*, London, Centre for Economic Policy Research.
- Vines, D.A. & S.M. Kanbur (1984) 'North-South Interaction and Commod Control', *CEPAR Discussion No. 8*, London, Centre for Economic Policy Research.
- Vos, Rob (1988) 'Accounting for the World Economy', *Working Paper* Sub-Series on Money, Finance and Development No. 26, The Hague: Institute of Social Studies.
- Vos, Rob (1989a) 'Accounting for the World Economy', *Review of Income* and Wealth, 35(4): 389-408.
- Vos, Rob (1989b) 'International Finance, Global Adjustment and North-South Relation', in E.V.K. FitzGerald & Rob Vos (eds) *Financing Economic Development: A Structural Approach to Monetary Policy*. Aldershot: Gower.
- Vos, Rob (1990) 'On the Measurement of Private Foreign Asset Accumulation: The Case of the Philippines', *ISS Working Paper Series on Money Finance and Development, No. 33*, Institute of Social Studies, The Hague, The Netherlands.
- Vos, Rob (1994) Debt and Adjustment in the World Economy: Structural Asymmetries in North-South Interactions. London: Macmillan Press Ltd.
- Wallerstein, Immanuel (1976) 'The Three Stages of African Involvement in The World Economy', in Wallerstein, Immanuel & C.W. Gutkind (eds) *The Political Economy of Contemporary Africa*. London: Sage Publication.
- Weisskopf, Thomas E. (1972) 'The Impact of Foreign Capital Inflow on Domestic Savings in Underdeveloped Countries', Journal of International Economics, 2: 25-38.
- Wheeler, David (1984) 'Sources of Stagnation in Sub-Saharan Africa', World Development, 12(1): 1-23.
- White, Howard (1992) 'The Macroeconomics Impact of Development Aid: A Critical survey', *Journal of Development Studies*, 28(2): 163-240.
- White, Howard (1993) 'Aid and Government: A Dynamic Model of Aid, Income and Fiscal Behaviour', *Journal of International Development*, 5 (3): 305-12.

- White, Howard (1994) 'Foreign Aid, Taxes and Public Investment: A further comment', Journal of Development Economics, 45:155-63.
- White, Howard (1996a) 'Adjustment in Africa: A Review Article', Development and Change, 27(4): 786-815.
- White, Howard (1996b) 'Zambia in 1990s as the Case of Adjustment in Africa', *Economic Research Seminar*, The Hague: Institute of Social Studies.
- White, Howard & Mark McGillivray (1992) 'Descriptive Measures of the Allocation of Development Aid', *ISS Working Paper Series No. 125*. The Hague: Institute of Social Studies.
- White, Howard & Mark McGillivray (1993) 'Explanatory Studies of Aid Allocation Among Developing Countries: A Critical Survey', *ISS Working Paper Series No. 148.* The Hague: Institute of Social Studies..
- Wickens, M.R. & Greenfield (1973) 'The Econometrics of Agricultural Supply: An Application to the World Coffee Market', *Review of Economic and Statistics*, 55(2): 433-40.
- Wickens, M.R. & T.S. Breusch (1988) 'Dynamic Specification, the Longrun and the Estimation of Transformed Regression Model', *The Economic Journal*, 98 (Conference, Supplement): 189-205.
- Williamson, John (1983) The Open Economy and the World Economy: A Textbook in International Economics. New York: Basic Books.
- World Bank (1978) Arab Republic of Egypt: Economic Management in the Period of Transition, Vol. IV, Report No. 1815-EGT.
- World Bank (1983) Arab Republic of Egypt: Current Economic Situation and Growth Prospects, Report No. 4498-EGT.
- World Bank (1981) Accelerated Development in Sub-Saharan Africa: An Agenda for Action. Washington DC: The World Bank..
- World Bank (1989) Sub-Saharan Africa: From Crisis to Sustainable Growth- A Long-term Perspective Study. Washington DC: The World Bank.
- Word Bank (1994) Adjustment in Africa: Reforms, Results, and the Road Ahead. Washington DC: The World Bank.
- World Bank (1994) World Debt Tables, Electronic Data (STARS).
- World Bank (1994) World Tables, Electronic Data (STARS).
- World Bank (2000) *Global Development Finance*, CD-ROM. Washington DC: The World Bank.
- World Bank, *DEC Analytical Database* (International Economics Department, Analysis and Prospects Division).
- World Bank, World Bank Economic and Social Database (BESD), Electronic form.

- Wuyts, Marc (1992a) 'Theory, Fact and Method, Unit 2', in School of Oriental and African Studies (SOAS). *Research Methods in Financial Economics*. London: University of London.
- Wuyts, Marc (1992b) 'Thinking with Data', Unit 5', in School of Oriental and African Studies (SOAS) *Research Methods in Financial Economics*. London: University of London.
- Yeats, Alexander (1991) 'Do African Countries Pay More for Imports? Yes', in Chhiber & Fischer (eds) *Economic Reform in Sub-Saharan Africa*. Washington DC: World Bank.
- Younger, S.D. (1992) 'Aid and the Dutch Disease: Macroeconomic Management When Everybody Loves You', *World Development*, 20(11): 1787-59.
- Zellner, Arnold & S.C. Peck (1973) 'Simulation and Experiments with a Quarterly Macroeconomic Model of the U.S. Economy', in A.A. Powell & R.A. Williams (eds) *Econometric Studies of Macro and Monetary Relations*. Amsterdam: North Holland.

Index

Abebayehu Tegene 161, 162, 165, 170 accelerator model 221 accounting framework for macroeconomic analysis of Africa 105-9 Accounting for the World Economy xvii accumulation balance 107 Adam, C. 11 Adams, F.G. 59, 70, 75, 263 adaptive expectation scheme 60, 121 adding-up problem 162, 163, 303 Adedeji, A. 13 adjustment 11 adventurer companies 26 Ady, P. 123 AERC (African Economic Research Consortium) 101 African Alternative Framework to Structural Adjustment Programmes 10 African Caribbean and Pacific (ACP) region 49 African Development Indicators 181 Africa's Adjustment and Growth in the 1980s 353n3 AFRIMOD external shock simulations 303 glossary of symbols used 289-91 identities used 272 international financial market 278 limitations 323 linearity 299-300 model solution algorithm 295-301 neostructuralist global econometric model 258-91 North 277-8 parameters 278-88 policy simulations 302-3 stability conditions 298-301 World Accounting Matrices (WAM) as a database 292-5 Agarwal, J.P. 35, 36 Agenda for Action 10

agricultural products 17 agricultural raw materials 62, 70, 71, 125, 130.159 East and Southern Africa 139, 213 ECM estimation 168 North Africa 146 West and Central Africa 153, 217 agricultural subsidy policies 11 aid 2-3 Africa 257 allocation 50 allocative function 45 and domestic savings 104, 366n15 and Dutch disease 198-220 Africa 201, 202 empirical studies 48-50 geographical distribution 49 and growth xix policy simulation 302, 304-10 as a public good 47-8 and real exchange rate 219-20 reasons for 45-6 redistributive function 45 and savings 174 simulation of increased level 304-10 stabilization function 45 and taxes 175 aid flows 44 Ake 15 Akiyama, T. 162 Alejandro, C.F.D. 23, 41 Algeria see North Africa Alibaruho, G. 160 Ali, L. 13 Allen, C. 236, 317 Alogoskoufis, G. 236, 255 Amin, S. 15, 16, 17, 18, 21, 22, 25, 26, 27, 28, 56, 264 Amjadi, J. 354n4 Anderson, T.W. 338 Angola 19, 21 Anyadike-Danes, M.K. 49, 116

Anyadike-Danes, M.N. 49, 116 Ardeni, G. 58 Armington formulation 251 Armington, P.A. 82 Asante, S.K.B. 15 Askari, H. 122 Asmerom Kidane 99 Augmented Dickey-Fuller Unit Root Test 340 Austen, R.A. 17, 18, 22, 26 Autoregressive-Distributed Lag (ADL) formulation 336-7 available resources 12 Bacha, E.L. 23, 41, 55, 101 **Balanced World Accounting Matrices** (WAM), for 1990 296-8 balance of payments 108 Malawi 94 Balance of Payments (BOP) statistics 107 balance of payments crisis 101 balance of payments deficit 12 balance of payments problems 10-11 Balassa, B. 53 Balasubramanyam, V.N. 58 Ball, R.J. 243 Banerjee, A. 334, 335, 341 Bank-GEM model 236, 244-5, 253, 317 bank lending, Africa 51 bank lending theories 38-9 bank managers 359n8 Bansfled 45 Baran, P. 20, 37, 356n23 Bardhan, P. 38 bargaining-theoretical approaches 41 Bateman, M.J. 122 Beenstock, M. 44, 49, 243, 244, 255 Behrman, J.R. 59, 70, 75, 122, 263 Belgian Congo 25 Benin see West and Central Africa Benjamin, N. 92, 99, 201 Berg Report 10 Berhanu Nega 96, 99 Bernstein, H. 19 Bettelheim, C. 56 beverages 62, 70, 71, 124, 130 East and Southern Africa 139, 145, 213 ECM estimation 168 West and Central Africa 153, 218 bilateral aid 46 bilateral debt xviii, 8 bilateral flows 2 Binh, T.-N. 349

Binswanger, H.P. 120, 160 Bismarck 21 Blaug, M. 29 Bok, T.J. 301 bond/deficit finance 3, 326 bond finance 41, 328 Bond, Goldstein and Khan, commodity model 62-6 Bond, M.E. 62, 66, 85, 123, 160, 263 Botswana 374n23 Dutch disease 201 Boutros-Ghali, Y. 99 Brazil 243 British East Africa 22 British and French West Africa 25 British Guinea 22 Brooks, G.E. 17, 18 Bruno, M. 317 Buckley, P.J. 36 budget constraint 177 budget financing 328 buffer stock system xix Bugembe, P.K. 13 Bulow, J. 40 Burdette, M. 23, 26 Burkina Faso 160 Burundi 353n1 Cagan, P.H. 121 Cameroon CGE model 92 Dutch disease 201 capacity 2,9 capacity expansion 101 capacity utilization 101, 102, 123, 134, 239 capacity utilization theory 126, 127, 268 capital xviii capital flight 245 from Africa 279 capital flows xviii, 2, 34, 35, 46, 50, 172, 244-5, 247 economic and geo-political factors 44-6 in models 241-2 official 44-50, 46 capital formation 128, 159, 161 capital inflow 2, 197-8 and direct taxes 185 and distribution of income 198 and government current expenditure 193 and indirect taxes 191

capitalism 15, 46 capital markets 23 access 230 segmentation 43 capital movement, portfolio theory of 38 capital output ratio 279 caravan trade 18 Carlisle, D. 121 cartel xix cash cropping 20, 22 Casson, M. 36 Central Africa 26 Central African Republic see West and Central Africa CGE (computable general equilibrium) model Cameroon 92 Dick et al. 76-8 Zimbabwe 97 Chenery, H. 104, 174, 178, 221 Chow test 206 Chu, K.-Y. 66, 67, 84, 121, 123, 126, 127, 268 civil strife 11 Clark, J.M. 221 Cline, W.R. 2, 40, 41 Coase, R.H. 36 Cobb-Douglas form 254 'cobweb theorem' 66, 121 cocoa supply model, Ghana 122 Cohen, D. 41, 42 co-integration 113, 145 Collier, P. 11, 13, 100, 354n8 colonial Africa, macro regions 25-8 colonial era 1, 3, 16 colonialism 13, 21 colonization 16 commodities 13 and debt 325-6 commodity exports 23-5, 303 history 25 commodity market, Africa 274-5 commodity model Bond, Goldstein and Khan 62–6 Hwa, E. 72-5 Ke-Young Chu and T.K. Morrison 66-9 commodity models 61-85 commodity prices xix, 2, 3, 16, 304 commodity problem xviii commodity stock holding 70 commodity supply modelling 124

commodity trade in the global market Africa 171-2 'comparative disadvantage' 37 concessionaire companies 26 concessionality 9 Congo 353n1 constant elasticity of substitution (CES) 92 consumer price index (CPI) 94, 103 consumption 11, 40, 81, 98, 174, 176 Cooper, R.N. - 39 copper, Zambia 354n8 copper prices model, Zambia 95-6 Corden, M. 104, 198, 199 Cote d'Ivoire 353n1 cotton 20, 21, 22, 28 experiments 21 Kenya 160 Tanzania 161 Uganda 160 'Counot's Law' 377n1 country classification 332-3 credit rationing 43, 242, 248, 270, 278, 317 credit rationing theories 38 crisis theory 15 crops 122, 123 supply response model 122 crowding-in 102, 273, 306 crowding-out 102, 242, 273, 278 CSAE (Centre for the Study of African Economies) 100 Cuddington, J. 39, 40, 58 Cummings, J.T. 122 Darity, W. 32, 239, 248, 256, 258, 260, 270, 359n8 Davies, R. 97, 99 debt xvii, 3, 8-16 background 8-10 and commodities 325-6 Malawi 94-5 sustainability 51 Debt and Adjustment in the World Economy xvii debt cancellation 3 simulation 310-12 debt crisis 246, 247 origins 16 debt indicators 4-7 debt overhang 9, 311 debt problem, institutional response 41 debt relief Laffer curve 42, 43

416

debt-repayment feasibility model 40 debt service ratio 271 Decaluwé, B. 99 decision-making framework 176-7, 178-9, 197 default, risk of 39 deficit financing 305, 307 deficit reduction 2 demand elasticity, food 56 dependence 12, 13, 15, 16, 248 dependency theory 15 dependent economy model 199 Dercon, S. 125, 161 devaluation 212 developmental considerations 44 Dick et al., computable general equilibrium model 76-8 Dick, H. 76, 85 Dickey-Fuller test 58 Dickson, K.B. 22 direct investment 41 direct taxes 179-80, 184-6, 188 and capital inflows 185 disaggregation 3 disequilibrium model 64, 72–5 distribution of income, and capital inflows 198 distribution of wealth 13 diversification 11, 13 Dixit, A. 53 domestic borrowing, and foreign loans 175 domestic capital formation, Tanzania 94 domestic demand 123 domestic macroeconomic conditions 99 domestic policies 11, 328 domestic savings, and aid 104, 366n15 donors 44, 45, 46, 47, 48, 49 Dornbusch, R. 230 DR Congo see West and Central Africa drought 11 Dudley, L. 44, 48, 261 Dunning, J.H. 36, 262 Dutch disease xviii, xix, 2, 3, 87, 103, 127, 135, 159, 173, 329, 350 in Africa 275 and aid 198-220 Botswana 201 Cameroon 201 Egypt 201 Ghana 200-1 literature 104, 198 model 92, 268-9

North Africa 217 Tanzania 94 variation 219-20 West and Central Africa 153, 217-19 Zambia 95, 201 Dutch disease model co-integration test 208 data 208 ECM results 210-11 estimation results 205-19 unit root analysis 207 Dutt, A.K. 29, 30, 32, 230, 234, 236, 239, 240, 242 East Africa, export trends 14 Easterly, W. 11 East and Southern Africa (ESA) 18-19, 20, 25, 136-42, 184, 186, 192, 332-3 agricultural raw materials 139, 213 beverages 139, 145, 213 co-integration test 138 Dutch disease, econometric results 209, 212-15, 217 exports 304 FDI 112 food 139, 213 government consumption expenditure 209 minerals 212, 215 ores and metals 145 official capital flows 116-17 private investment model 222 real exchange rate 212 unit root analysis 137 Eaton, J. 39, 41, 42, 271 ECA (Economic Commission for Africa) 10, 12, 15, 192, 353n3 ECLAC (Economic Commission for Latin America and the Caribbean) 86 eclectic approach to FDI 111 'eclectic paradigm' 36-7 ECM see Error Correction Model econometric models 59-60 Malawi 94-5 econometric optimization models 122 Economic Commission for Africa (ECA) 10, 12, 15, 192, 353n3 economic crisis, causes 10-16 economic interest of donors 44 economic linkage with the North, history 16-29 Economic and Social Progress in Latin America (ESPLA) 111, 115

'economies of specialization' 36 education 11 Edwards, S. 104, 199, 200, 204, 205, 268 Egypt 364n3 Dutch disease 201 general equilibrium model 91-2 North Africa 146 elasticities 56, 170 Africa 158 Elbadawi, I.A. 11, 354-5n12 Emmanuel, A. 52, 55 Enders, W. 334, 335 endogenous determination 250 Engel-Granger test 170, 184 Engel, R.F. 334, 335 Enos, J. 104, 174 environmental factors 14 equilibrium 59 equilibrium exchange rate 204 equilibrium model 64 Eriksson, G. 125, 160 Error Correction Model (ECM) 111, 125-6, 134, 161, 179, 196, 197, 206, 209, 225, 280, 334-5, 351 equation 184, 186 formulation 335-8 ESPLA see Economic and Social Progress in Latin America estimation 350-2 estimation approach 134-6 Ethier, W.J. 53 Ethiopia 18, 355-6n21 macroeconomic model 96-7 Eurodollar 16 European Community (EC) 49 European control of Africa 19-20 European market xix Evans, D. 162 EVIEWS software 295 exchange rate 14 exogeneity of growth 102 exogenous foreign borrowing 90 Expanded World Trade Model (EWTM) 78-9, 81, 253 expansionary policies xix expected profitability theory 127 expected profit indicator 368n16 stationarity condition 340-2 export demand 62-3 export prices 130-4 exports xvi, xx, 2, 3, 9, 10, 11, 13, 268 Africa 162-72 primary commodities 14, 20, 56

literature 120-9 export shares matrix 81-3 export supply 63-4 Africa, determinants 119-62 external balance 107 external debt 1 external economic relations xvii external finance 14 fiscal response to 173-98 macroeconomic effects of 103-5 external finance problems, Africa 1-2, 3, 8.34 external sector 10, 12, 16 external shocks 10, 11, 12, 13 effect on Africa 292-323 impact 99 external shock simulations, AFRIMOD 303 external trade sector 93 Ezekiel, M. 66, 121 factor endowment differentials 52 factor mobility assumptions 52 'fallacy of composition' xvii, 162, 275 Falvey, R. 53 Fantu Cheru 13, 14 FDI (foreign direct investment) 8, 23, 26, 35-8, 50, 262, 264, 266, 358n3 determinants of allocation 111-14 Feder, G. 38 Fernandez-Arias, E. 111 Feyerabend, P. 29 Fieldhouse, D.K. 21 'Finance and Development' research group xvii finance theories 35-51 financial flows 243, 248 determinants 111-19 financial linkages, North-South 243-50 financial outflows 9 Findlay, R. 198, 236, 248, 256 fiscal response 2, 3, 267, 273-4, 329 literature 174-7 model 178-81 co-integration test 183 estimation results 181-96 results 183-96 unit root analysis 182 to external finance 87, 173-98 FitzGerald, E.V.K. 30, 32, 106, 221, 239, 248, 256, 258, 260, 264, 267, 270 fixed import coefficient 101 Fleming, J.M. 228 flexi price structure 101

f.o.b.-c.i.f. problem 83-4 Folkerts-Landau, D. 41-2 food 62, 70, 71, 159, 268 demand elasticity 56 East and Southern Africa 139, 213 ECM estimation 168 North Africa 215 UNCTAD classification 369n21 West and Central Africa 153, 216, 218 food prices 130 North Africa 146 foreign aid 1 foreign capital 99 foreign direct investment see FDI foreign exchange earnings 99 foreign exchange reserves 9 foreign loans, and domestic borrowing 175 Fosu, A.K. 14 France 20 Frankel, H. 22 free market international system 15 'free rider' problem 40 French, B.C. 122 French Equatorial Africa 25 Frenkel, J. 230 Frey, B. 44, 48 Friedman, M. 177 Fritsch, W. 243 Full Information Maximum Likelihood (FIML) method 66, 343 Fundamental Theorem of Unequal Exchange 55 Fyfe 21 Gabon see West and Central Africa game theory 48 Gang, I.N. 176 gap models 101 Gasper 44, 45 GDP xvi, 9, 90, 98 Gebrehiwot Baykedagne 356n21, 361n18 Geda, A. 106, 245, 267 Geenway, D. 360n15 general equilibrium model, Egypt 91-2 geographical distribution of aid 49 geo-political factors in capital flows 44-6 geo-political interest of donors 44 Germany 21 Gersovitz, M. 39, 271 Ghana cocoa supply model 122 Dutch disease 200

Ghoshal, A. 125, 160 Ghura, D. 11, 14, 354n98 Gibson, B. 55 Gilbert, C.L. 124 global commodity model 2, 163-6 co-integration test 167 commodity sub-block 171, 172 disequilibrium model 164-5, 345-6 econometric model and estimation results 170 - 1equilibrium model 343-5 estimation 165-71 results using first difference values 343-7 time series properties 165-70 unconstrained estimation 346-7 unit root analysis 166 global economic model, London Business School 236 'global financial architecture' xix global financial market 43 Global Keynesianism 328 GNP 81 19 gold Goldstein Khan model 162, 163, 164 Goldstein, M. 62, 64, 65, 66, 84, 123, 126, 162, 170, 268 governance xvi government current expenditure 180-1, 192-6 and capital inflow 193 government expenditure 2, 174 Government Financial Statistics, IMF 181 government policy 11 Granger Causality concept 371n1 Granger, C.W.J. 334, 335 grants 8, 181, 194, 245 Gray, L. 121 great depression 24 Greenfield 121, 123, 268 Grier, K.B. 11 Griffin, K. 104, 105, 174 Grilli, E.R. 58 groundnuts 21 growth 11, 104 and aid xix Grubel, H. 53 Guass-Seidel algorithm 295, 321 Guinea 19 Guinea-Bissau 353n1 Gunning, W. 11, 100 Gurney, A. 338 Guvenen, O. 59, 61

Gwyer, G.D. 125, 160 Hallwood, C.P. 228 Harberger 91 Harris, J. 88 Harris, L. 15 Harrod-Domar growth model 104 Harton, S. 99 Harvey, C. 105 Heckscher, E.F. 52 Heckscher-Ohlin-Samuelson theory of trade 36 Heien, D. 295 Helg, R. 58 Helleiner, G.K. 36, 37, 53, 54, 262 Heller, P.S. 105, 174, 175, 176 Heller's model 348-9 Helpman, E. 53 Hendery's General to Specific (GS) methodology 184, 206 Herfindahl, O.C. 121 Hermes, N. 279, 280 Hickman, B.G. 251 HIPC (Highly Indebted Poor Countries) initiative xviii, 302-3, 310 history, economic 16-30 Hopkins, A.G. 17, 18, 21, 26, 27 Horn, B. 359n8 Hotelling, H. 121 Houthakker, H.S. 164 Hsiao, C. 338 Hufbauer, G.C. 52 humanitarian considerations 44, 119, 360n12 humanitarian motives 50 human resource development 13, 15 Hwa, E. 70, 84, 85, 124, 126, 162, 263, 268 commodity model 72-5 Hymer, S. 35, 36 IMF xvi, xix, 8, 15, 78, 100, 247, 253, 255 Government Financial Statistics 181 MULTIMOD 3, 239, 243, 247, 255, 270-1, 317 'immiserization of the recipient' 358n2 imperfect competition 52, 53 imperialism 15 imperial self-sufficiency in raw materials scheme 20, 21-2 import compression 87, 101-3, 243, 247, 248, 264, 276, 307, 327

and private investment, econometric results 220-5 Zimbabwe 97 'import compression problem' 9 import equations 80-1 imports 3, 9, 27 import substitution 54 income 13 income distribution 15 increasing returns 53 independence 1, 13 India 176 indirect taxes 180, 188-92 and capital inflow 191 Indonesia 176 industrial organization theory -36 industrial production 46 industrial revolution 19 inflation 12, 103 infrastructure 1, 11, 123 innovation 52 input-output models 61 insolvency 40 instability 1 institutional response, debt problem 41 Inter-American Development Bank 111, 115 interest rates xix, 14, 16, 42, 244, 246, 277, 303, 304 and debt servicing 311 and oil prices 317 world interest rate 326-7 increase, simulation 315-17 inter-industry coefficients 90 inter-industry relations 61 INTERLINK model, OECD 24, 256 international banking 270-2 international bank lending 41 international financial market, AFRIMOD 278 international financial systems 23 international firms 36 international trade theories, implications for Africa 52-85 intra-firm trade 53, 54 intra-industry trade 36, 53, 360n15 investment xx, 1, 9, 56, 98 iron 20 Italy xvi Iversen 35 Iyaniwura 90, 99 Jamshidi, A. 3, 299

Jarque-Berra statistics 196 Johansen, S. 334 Johansen test 170, 184 Johnstone, M.A. 56 Jones, E. 125, 160 Jong, N. de 292, 293 Jorgenson, D.W. 35, 221 Kalaitzandonakes, N.G. 123 Kaldor, N. 239 Kalecki-FitzGerald model 96 Kalecki-Lewis model 237, 238-9 Kalecki, M. 57 Kanbur, S.M.R. 258 Kao, C. 338 Kasekende, L. 111, 271 Kay, G. 56 Kayizzi-Mugerwa, S. 95, 99, 201 Kennedy, P. 335 Kenya 20 cotton 160 Keynesian closure 175 Keynes, J.M. 106, 221 Ke-Young Chu and T.K. Morrison, commodity model 66-9 Khan, H. 176 Khan, M.S. 62, 64, 65, 66, 84, 123, 126, 162, 170, 268, 279 Kindleberger, C. 35 Kjekshus 18 Klein, L.R. 251 Kletzer, K. 41 Kocklaeuner, G. 99 Kojima, K. 37 Koncazcki, Z.A. 18, 22, 27 Korean war 23 Kouri, P. 230 Koyck, L.M. 122 Kravis, I. 53 Krugman, P. 36, 42, 52, 53, 251, 254, 255 Kurihara, E. 251 Kwack, S.Y. 98 labour supply 257 Labys, W.C. 59, 60, 61, 70, 84 Lagos Plan of Action 12, 15 Lall, S. 11, 53, 54 Lancaster, K. 53 'land question' 25 Larson, D.F. 162 Lau, L.J. 251 Lawrence, P. 14

Lawson, T. 29, 30 least developed countries (LDCs) 243-4 Lee, S.H. 38 Lemma Mered 99 Lensink, R. 105, 279, 280 Lesotho see East and Southern Africa Levin, A. 338 Levine, R. 11 Lewis, A.W. 234 Lewis type economy 238, 239-41 Leys, C. 18 liberalization 100 liberalization policies 197 Liberia, rubber 160 LIBOR (London Inter Bank Rate) 98, 170, 181, 303 Libya see North Africa Lin, C.-F. 338 Linder, S.B. 53 linearity, AFRIMOD 299-300 Lipton, P. 29 Lipumba, N. 92, 99 liquidity model 123 Little, R. 48 'loanable fund' doctrine 243, 244 loans 24 local industrialization, blockage 21-2 Lome system xix London Business School, global economic model 236 London club 9 Longmire, P. 21 Love, R. 201, 374n23 low-level disequilibrium xvi Luttik, J. xvii, 292, 293, 294 Luvanda, E. 201 luxury trade 355n17 McCoskey, S. 338 MacDonald, R. 228 MacDougall, D. 35, 38 MacDougall model 35 Macedo, J.B. de 201 McGillivray, M. 44, 47, 49, 50, 116, 175, 261, 349 Mack, A. 56 McKibbin, W.J. 243, 301, 317 McKinley, R.D. 48 McLaren, J. 99 macroeconomic effects 2, 3 of external finance 103-5 macroeconomic model, Ethiopia 96-7 macroeconomics, Africa 100-9

macro models Africa 88-100 type and category 88-9 Tanzania 92-4 Madagascar 353n1 Magdoff, H. 37, 46, 358n3 Maizels, A. 124 Maji-Maji uprising 20 Malawi balance of payments 94 debt 94-5 econometric model 94-5 external shocks and policies 94 monetary sector 95 prices 95 Mamdani, M. -15 manufactured goods 18 marginalization xvi market imperfection 36 market mechanism 15 market model 59 markets 14 perfection in 35 market segmentation 41 mark-up 238-9, 326 Marquez, J.R. 248, 251, 256, 258, 260, 266 Marxist orientation 10 Marxist theories of FDI determination 37, 38 Masson, P. 243, 247, 251, 255, 258, 270, 317 Matthews, J.L. 122 Meade, J. 228 Meade-Mundell-Fleming Model 228, 229 mercantilist period 16 metals 70, 257 methodology 29-31 Mikesell, R. 44, 46, 261, 360n12 Milner, C. 360n15 minerals 25, 102, 121-2, 130, 159, 304 East and Southern Africa 145, 215 ECM estimation 169 exports 10 extraction 24, 27 North Africa 146–7 West and Central Africa 153, 216, 218 mixed (heterodox) factor-based models 123-5 modelling within a global economic framework 227-91 Molana 239, 240, 248, 256 monetary policy 247 monetary sector, Malawi 95

monetization 102 Montiel, P.J. 111 Montmarquette, C. 44, 48, 261 Morocco see North Africa Morrison, T.K. 66, 67, 84, 121, 123, 126, 127, 268 Mosley, P. 11, 44, 45, 46, 47, 105, 174, 175, 176, 261, 349, 353n3 Moutos, T. 251 Moyo, N.P. 15 Mozambique 353n1 MSG2 model 243, 253-4 Mukherjee, C. 29, 30, 339 multilateral debt xviii. 8 MULTIMOD model, IMF 3, 239, 243, 247, 255, 270-1, 317 multinational enterprises (MNEs) 53, 54 multiple regression equation 11 Mundell-Fleming Model 228-30 Mundell, R. 228, 377n1 Munro, J.F. 20, 21, 23, 24 Murshed, S.M. 243, 246, 258, 260 Muscatelli, V.A. 243, 246, 251 Musonda, F.M. 201 Mutuura, J. 125, 160 national income accounting 106 Ndulu B.J., 87, 101, 102-3, 243, 264 Neary, J.P. 201, 202 needs 12, 13 neoclassical approach 35, 100 neoclassical trade theories 52 neo-Marxian North and Lewis South model 239 - 41neostructuralist global econometric model, AFRIMOD 258-91 Nepal terms 302, 310 Nerlove, M. 67, 121 Neumark, D.S. 17 new trade theories 53, 54, 86, 360n14 Ngwenya, A.R. 13 NIESR (National Institute of Economic and Social Research) 236 Nigeria 266 macroeconomic model 90-1 Njuguna, A.E. 338 non-orthodox (critical) school 54 non-orthodox models 54-8 Norman, V. 53 North Africa 17, 145-7, 333 agricultural raw materials 146 capital inflows 189-90 co-integration 144

North Africa - continued Dutch disease 147, 217 estimation with capital inflow 150 estimation with current prices 148 estimation with real prices 149 112 FDI food 215 food prices 146 minerals 146-7 official capital flows 118 unit root analysis 143 Northern countries (the North) 1, 332 Africa's economic linkage, history 16 - 29in AFRIMOD 277-8 modelling 259-62 and the South 3 Northern macro policies, impact on Africa 324 North-South, financial linkages 243-50 North–South framework 36 within a North-South framework 258-91 North–South modelling 35–51 North-South models 52, 103, 227, 258 financial linkages 249-50 standard linkages 242-57 theortical bases 230-42 trade linkages 252 North-South trade 54, 86 in primary commodities, models 61-85 Nsengiyumva, F. 99 Nzula, A.T. 18, 25, 26, 27 OAU (Organization of African Unity) 12 Ocampo, J.A. 36, 55, 230 Odedokun, M.O. 271 OECD 44, 45, 170, 256, 261 INTERLINK model 24, 256 official capital flows 44-50, 46, 261-2 Africa 114-19 official inflows 258 Ofuatey-Kodjoe 15 Ohlin, B. 52 oil 254, 255–6, 261, 266, 373*n*22 oil prices 10, 246, 303 increase, simulation 317-19 oil price shocks 16 oil revenue, impact 99 Olafin, S. 90, 99 Onimode, B. 15, 21 OPEC 244, 254, 261, 269, 317, 332 operative forces 12-13

Opeskin, B.R. 45 optimal borrowing 38-9 optimization models 60 Ordinary Least Squares (OLS) 179 orthodox macroeconomic management 11 Oshikoya, W.T. 90, 99, 266 Oskawe 15 output 3, 11 'ownership-specific, location and internalisation' (OLI) 36 Oxfam 303 Pack, H. 176 Pack, J.R. 176 Pal, S. 123 Palaskas, T. 123, 124 Papanek, G. 104, 105, 174 Paris club 9 Parish, F. 121 partial equilibrium-based analytical solutions 273-8 Patnaik, P. 57 Pauly, P. 248, 251, 256, 258, 260, 266 peasant cropping 24-5 Peck, S.C. 299 Pedersen, K.N. 236, 244 Pedroni, P. 338 perfect competition, 59 perfection in markets 35 Pesaran, M.H. 338 Peterson, C. 236, 244, 253 Pim, A. 357n31 plantations 18, 24, 26, 27 Senegal 21 Pleskovi, B. 91, 99 polarization trends 14 policy 178 policy failure 11 policy modelling, Africa 99 policy shocks 3, 328-9 policy simulations 302-3 public investment 303 political instability 11 Pollack, P.K. 59, 60, 61 Popperian approach 29 population growth 11 population policy 15 Porter, M. 230 portfolio approach 35 portfolio holding 245 portfolio investment 23 portfolio theory 36, 38, 248 Posner, M.V. 52

post-independence period 16 potential supply approach 121 poverty 45 poverty alleviation 9 Prebisch, R. 55, 56, 57, 58, 263 Prebisch–Singer hypothesis 56–7 and empirical evidence 57-8 pre-colonial trade, Africa 17-19 pre-mercantilist period 16 Press, H.W. 295 160 price elasticities price-focused supply models 120. 121-3 prices, Malawi 95 prices (export prices) 130-4 primary commodities 13, 52, 57 exports 14, 20, 56 literature 120-9 North-South trade models 61-85 quantitative trade models 59-85 trade in 54-8 primary school enrollment 174 private investment determinants 221 and import compression, econometric results 220-5 private investment model 221-5 unit root analysis 223 process models 59 product cycle theory 35-6, 254 production capacity 13 productivity 13 Project LINK 79, 243, 251, 253 prototype model, Africa 98-9 public deficit, Africa 327-8 public expenditure 16, 178 public infrastructure development 25 public investment xix, 9 increase, simulation 313-15 policy simulation 303 public and private sectors 13 public sector 102 public sector deficit requirement 103 public utilities 20, 23–5 quantitative models of trade in primary commodities 59-85 'Quantity theory' of money 103 Quibria, M.G. 358n2 'race to the bottom' xvii Ramanujam, P. 70, 85, 124, 161, 162, 167, 170, 263 Ramanujam and Vines, structural rational expectation model 70-2

Rattsø, J. 87, 101, 243, 264 raw materials 20, 21, 46, 56 Razin, A. 230 real exchange rate 101, 102, 200, 204, 205, 208, 268, 350 and aid 219-20 East and Southern Africa 212 realist approach 29 recipient need models 48-9 recipients fiscal response 105 recursive approach 178 regional complementarities 17 regional development 18 regional variation 159 regions 3 Reinhart, C.M. 58 rescheduling 9 resource allocation 15 resources 12, 13 restrictive practices xix revenue maximization 146 Rheinhart, C.M. 279 Ricardo, D. 52 Riddell, R. 44 risk of default 39 road infrastructure 120 Rodney, W. 17, 19, 22, 28 Rogoff, K. 40 rubber, Liberia 160 Ruttan, V. 44, 45, 261 Rwanda 365n8 Rwegasira, D. 125 Sachs, J.D. 38, 39, 40, 41, 243, 317 Sahn, D.E. 94, 99 Salter, W.E.G., 199 SAM framework 91-2 Samuelson, L.W. 251 Samuelson, P.A. 52 Sapsford, D. 58 Sarkar, P. 58, 239, 258 Sarmad, K. 279 Sau, R. 56 savings 108 and aid 174 scale economies 53 sectoral adjustment 100 segmentation of the capital market 43 Seleti, Y.N. 26, 27 self-sufficiency, in food 15 Senegal 21, 160 plantations 21 share matrices 250

shocks xviii external 10, 11, 12, 13 financial and trade xvi policy 3 Shu, D.L.L. 122 Sierra Leone see West and Central Africa simultaneous equation model 162 Singer, H.W. 55, 56, 57, 263 slave-trade 16, 19 small country assumption 162 Smith, A. 52 Smith, G.W. 39, 40 Smith, R. 338 socio-political organization problems 12 Solink, B. 38 Solow-Lewis model 234-7 Soludo, C. 311 South 273, 332 in AFRIMOD 264-6 commodity exports simulation 320-1 macro balance 275-7 modelling 264-9 Southern and Eastern Africa 27 South–South trade 53–4 sovereign lending 51 sovereign lending models 39-43 'sovereign risk' 41 spatial equilibrium and programming models (SEPMs) 60 Spraos, J. 56 STAC model 247-8 stagnation 14 static/dynamic capital movement theories 38 stationarity 165, 177, 203 stationarity condition, expected profit indicator 340-2 stationary series 129 Stefanski, B. 13, 14 Stein, L. 14 Stewart, F. 52, 53 Stiglitz, J.E. 38, 53 stock holding 71 71-2 stock identity Stopler, W.F. 52 storage disequilibrium 363n32 Streeten, P. 44, 45 Strout, A. 104, 174, 178 structural adjustment programmes (SAPs) 8, 12, 100 structuralist school 55, 56-7, 59 structural macroeconomics school 30

structural rational expectation model, Ramanujam and Vines 70-2 sub-Saharan Africa 14, 102, 160 'Sudanic economy' 18 sugar 22 supply capacity 11 supply and demand 90 supply determination 67-8 supply price elasticities, Tanzania 160 supply theories, and estimation technique 125-6sustainable development 13 Sutcliffe, B. 14.15 Svedberg, P. 125 Swan, T.W. 199 Sweezy, P.M. 37 syndication 40, 41, 359n7 system dynamics models (SDMs) 59-60 Tanzania 18, 20, 102, 357n27 coffee boom 201 cotton 161 domestic capital formation 94 Dutch disease 94 macro model 92-4 sisal 160 supply price elasticities 160 Taplin, G.B. 82, 85, 253 model of world trade 78-85 taxes 2, 26, 27, 92, 98, 102, 174, 177 and aid 175 direct 179-80, 184-6, 188 indirect 180, 188-92 Taylor, L. 30, 32, 39, 41, 42, 64, 99, 101, 164, 239, 248, 251, 256, 258, 259, 260, 264, 268, 269 technological change 28, 236, 239, 254 technological gap models 52, 86 technology 205 and world trade patterns 52 technology transfer 36 terms of trade xix 1, 3, 10, 13, 14, 38, 55, 56, 239, 275 external 204 textile trade 17, 18 third period 16 Thirlwall, A.P. 361n22 Thomas, R.L. 335, 341 three gap model 198 time series econometrics 58, 134 time series properties, global commodity model 165-70 tobacco 21, 22, 62

Tobin, J. 35, 221 Tobin-q model 221 total trade model 79-80 tradables 205 trade pre-colonial 17-19 in primary commodities 54-8 gantitative models 59-85 trade and financial data, disaggregation 108 - 9trade linkages, in North-South models 250 - 7trade relations xix trade restrictions 11 transfer pricing 36 Trap, F. 87, 100, 105 'triangular trade' - 19 Tullock, B. 11 Tunisia see North Africa t-values 338 two-gap model 103, 104, 105, 198, 200 Uganda 353n1 cotton 160 UNCTAD 2, 129, 130, 161, 206, 303 Annual Commodity Yearbook 108 model 243 price series 131-3 underdevelopment 10, 15 unequal exchange theory 55-6 United Nations 10, 23, 24 United Nations Economic Commission for Africa (UN-ECA) 181 United Nations System of National Accounts 106 Urzua, C. 58 user cost model 221 utility function 105 utilization rate approach 121 Uy, L.V. 38 Van Arkadie, B. 11 94, 99 van Frausum, Y. van Peterson, A. 251 Vansina, J. 17 van Wijnbergen, S. 104, 198, 199, 200, 201, 205, 209, 350 Varangis, P. 236, 255 VAR model 334-5 Vernon, R. 35, 52, 254 Vines, D.A. 70, 85, 124, 161, 162, 167, 170, 236, 239, 240, 246, 248, 251, 256, 258, 263, 317

Vos, R. xvii, 30, 32, 42, 47, 228, 230, 239, 240, 242, 243, 248, 251, 256, 258, 260, 261, 270, 271, 279, 280, 292, 293, 305, 317, 318 wages, and unequal exchange theory 55 Wallerstein, I. 20, 21 WAM (World Accounting Matrices) 292 - 511 war Korean war 23 World War I 23 World War II 23, 24 wealth, distribution 13 Weeks, J. 353n3 Weiss, A. 38 Weisskopf, T.E. 174 West Africa 25-6 pre-colonial trade 17 West and Central Africa 19, 147, 184, 189, 194, 333 agricultural raw materials 153, 217 beverages 218 co-integration test 152 Dutch disease 153, 217–19 estimation with capital flow 156 estimation with current prices 154 estimation with real exchange rate 155 FDI 113 food 153, 216, 218 minerals 153, 216, 218 official capital flows 118-19 private investment model 224-5 tropical beverages 153 unit root analysis 151 Western Europe 356n23 White, H. 11, 44, 47, 49, 50, 116, 174, 175, 176, 202, 261, 349 Wickens-Breusch transformation 338 Wickens, M.R. 121, 123, 268 Wickham, P. 58 Williamson, J. 38 working hypothesis 29-30 World Accounting Matrices (WAM), as a database for AFRIMOD 292-5 World Bank xviii, 10, 11, 15, 100, 114, 181, 208 Bank-GEM (global economic model) 236, 244-5, 253, 317 world commodity model see global commodity model world debt tables 181

world economy xvi, 1 modelling xvii world energy demand 64 world interest rate 315–17, 326–7 world population xvi world trade, structural changes 10 world trade model, Taplin, G.B. 78–85 world trade patterns, technology 52 Wright, B. 58 Wuyts, M. 29, 30, 134

Yang, M.C. 58

Yeats, A. 257 Younger, S.D. 200 Zaire *see* West and Central Africa Zambia 11 copper 201, 354*n*8 copper prices model 95–6 Dutch disease 95, 201 Zellner, A. 299 Zimbabwe CGE model 97 import compression 97